

Special Issue Reprint

Financial Analysis, Corporate Finance and Risk Management

Edited by Eulália Mota Santos and Margarida Freitas Oliveira

mdpi.com/journal/risks



Financial Analysis, Corporate Finance and Risk Management

Financial Analysis, Corporate Finance and Risk Management

Guest Editors

Eulália Mota Santos Margarida Freitas Oliveira



Basel • Beijing • Wuhan • Barcelona • Belgrade • Novi Sad • Cluj • Manchester

Guest Editors Eulália Mota Santos Mathematics and Natural Sciences Polytechnic of Leiria Leiria Portugal

Margarida Freitas Oliveira Management, Marketing and Finance ISCAC–Polytechnic University of Coimbra Coimbra Portugal

Editorial Office MDPI AG Grosspeteranlage 5 4052 Basel, Switzerland

This is a reprint of the Special Issue, published open access by the journal *Risks* (ISSN 2227-9091), freely accessible at: www.mdpi.com/journal/risks/special_issues/2C4N8644RT.

For citation purposes, cite each article independently as indicated on the article page online and using the guide below:

Lastname, Firstname, Firstname Lastname, and Firstname Lastname. Article Title. *Journal Name* Year, *Volume Number*, Page Range.

ISBN 978-3-7258-3690-1 (Hbk) ISBN 978-3-7258-3689-5 (PDF) https://doi.org/10.3390/books978-3-7258-3689-5

© 2025 by the authors. Articles in this book are Open Access and distributed under the Creative Commons Attribution (CC BY) license. The book as a whole is distributed by MDPI under the terms and conditions of the Creative Commons Attribution-NonCommercial-NoDerivs (CC BY-NC-ND) license (https://creativecommons.org/licenses/by-nc-nd/4.0/).

Contents

About the Editors
Preface ix
Eulália Santos and Margarida Freitas OliveiraSpecial Issue "Financial Analysis, Corporate Finance and Risk Management"Reprinted from: Risks 2025, 13, 43, https://doi.org/10.3390/risks130300431
Fernando Tavares, Eulália Santos, Margarida Freitas Oliveira and Luís AlmeidaDeterminants of Corporate Indebtedness in Portugal: An Analysis of Financial BehaviourClustersReprinted from: <i>Risks</i> 2024, 12, 91, https://doi.org/10.3390/risks12060091
 Davood Askarany, Soleil Jafari, Azam Pouryousof, Sona Habibi and Hassan Yazdifar Board Gender Diversity and Risk Management in Corporate Financing: A Study on Debt Structure and Financial Decision-Making Reprinted from: <i>Risks</i> 2025, <i>13</i>, 11, https://doi.org/10.3390/risks13010011 25
Ermanno C. Tortia Dividend-Based Labor Remuneration and Tradable Shares in Worker Cooperatives Reprinted from: <i>Risks</i> 2024, <i>13</i> , 5, https://doi.org/10.3390/risks13010005
Helmi Jedidi and Georges DionneNonparametric Testing for Information Asymmetry in the Mortgage Servicing MarketReprinted from: Risks 2024, 12, 192, https://doi.org/10.3390/risks1212019269
Marco Desogus, Enrico Sergi and Stefano Zedda The Role of Credit Consortia in the Financial Structure of Sardinian Companies During the SARS-CoV-2 Crisis Reprinted from: <i>Risks</i> 2024 , <i>12</i> , 190, https://doi.org/10.3390/risks12120190
Samar F. Abou Ltaif, Simona Mihai-Yiannaki and Alkis Thrassou The Role of Personal Remittances in Economic Development: A Comparative Analysis with Foreign Direct Investment in Lebanon Reprinted from: <i>Risks</i> 2024 , <i>12</i> , 176, https://doi.org/10.3390/risks12110176
Dumisani Selby Nkambule, Bhekisipho Twala and Jan Harm Christiaan Pretorius Effective Machine Learning Techniques for Dealing with Poor Credit Data Reprinted from: <i>Risks</i> 2024 , <i>12</i> , 172, https://doi.org/10.3390/risks12110172
Mona Parsaei, Davood Askarany, Mahtab Maleki and Ali Rahmani Risk Management in Product Diversification: The Role of Managerial Overconfidence in Cost Stickiness—Evidence from Iran Reprinted from: <i>Risks</i> 2024 , <i>12</i> , 150, https://doi.org/10.3390/risks12100150
Narjis Abdlkareem Ibrahim, Mahdi Salehi, Hussen Amran Naji Al-Refiay and Mahmoud Lari Dashtbayaz A Financial Stability Model for Iraqi Companies Reprinted from: <i>Risks</i> 2024 , <i>12</i> , 140, https://doi.org/10.3390/risks12090140
Stefan Ionescu, Gabriel Dumitrescu, Corina Ioanăș and Camelia Delcea Mapping the Landscape of Key Performance and Key Risk Indicators in Business: A Comprehensive Bibliometric Analysis Reprinted from: <i>Risks</i> 2024 , <i>12</i> , 125, https://doi.org/10.3390/risks12080125

Abbas Ali Daryaei, Davood Askarany and Yasin Fattahi Impact of Audit Fees on Earnings Management and Financial Risk: An Analysis of Corporate Finance Practices
Reprinted from: <i>Risks</i> 2024 , <i>12</i> , 123, https://doi.org/10.3390/risks12080123
Samar F. Abou Ltaif, Simona Mihai-Yiannaki and Alkis Thrassou Lebanon's Economic Development Risk: Global Factors and Local Realities of the Shadow Economy Amid Financial Crisis Reprinted from: <i>Risks</i> 2024 , <i>12</i> , 122, https://doi.org/10.3390/risks12080122
Ndonwabile Zimasa Mabandla and Godfrey Marozva Government Borrowing and South African Banks' Capital Structure: A System GMM Approach Reprinted from: <i>Risks</i> 2024, 12, 112, https://doi.org/10.3390/risks12070112
Guy Burstein and Inon Zuckerman Uncertainty Reduction in Operational Risk Management Process Reprinted from: <i>Risks</i> 2024 , <i>12</i> , <i>77</i> , https://doi.org/10.3390/risks12050077
Ramona Rupeika-Apoga, Stefan Wendt and Victoria Geyfman Shareholders in the Driver's Seat: Unraveling the Impact on Financial Performance in Latvian Fintech Companies Reprinted from: <i>Risks</i> 2024, 12, 54, https://doi.org/10.3390/risks12030054
Marcos Escobar-Anel and Yiyao Jiao Robust Portfolio Optimization with Environmental, Social, and Corporate Governance Preference Reprinted from: <i>Risks</i> 2024, 12, 33, https://doi.org/10.3390/risks12020033
José Sequeira, Cláudia Pereira, Luís Gomes and Armindo Lima Features of the Association between Debt and Earnings Quality for Small and Medium-Sized Entities Reprinted from: <i>Risks</i> 2024, 12, 32, https://doi.org/10.3390/risks12020032
Iveta Grigorova, Aleksandar Karamfilov, Radostin Merakov and Aleksandar Efremov Simulation of Dynamic Performance of DeFi Protocol Based on Historical Crypto Market Behavior Reprinted from: <i>Risks</i> 2023 , <i>12</i> , 3, https://doi.org/10.3390/risks12010003

About the Editors

Eulália Mota Santos

Eulália Mota Santos received her PhD in Mathematics from the University of Aveiro, her master's degree in Mathematics from the University of Aveiro, and her degree in Applied Mathematics from the University of Évora. She is an Invited Adjunct Professor at the School of Education and Social Sciences of the Polytechnic Institute of Leiria, Portugal. She has authored and co-authored several journal articles and book chapters in the following areas: optimization and operational research, finance, management, tourism, wine tourism, human resources, and marketing. She is interested in analysis and data processing in different areas such as management, tourism, and marketing.

Margarida Freitas Oliveira

Margarida Freitas Oliveira received her PhD in General and Strategic Management and Business Development from ISCTE in 2016. She has held the title of Specialist in Human Resources Management—Scientific Area Management & Administration since 2020. She is an Adjunct Professor at Coimbra Business School |ISCAC—Polytechnic University of Coimbra. She is a Researcher at the Management Center of the Business Research Unit (BRU-IUL) ISCTE—Instituto Universitário de Lisboa, as well as a Researcher at the Center for Organizational and Social Studies of the Polytechnic of Porto (CEOS.PP). She has published several publications in specialized scientific journals and book chapters indexed in Web of Science and Scopus and has more than 20 years of business experience in the areas of management, human resources, strategy, and marketing. She exercises management functions in business groups with different business areas. Her scientific interests include management, human resources management, strategy, innovation, marketing, entrepreneurship, and organizational behavior.

Preface

The reprint *Financial Analysis, Corporate Finance and Risk Management* combines eighteen articles that explore the key aspects of financial analysis, corporate financing strategies, and risk management. This Book offers new approaches to understanding and tackling financial challenges in a constantly changing economic landscape.

The articles within this reprint discuss topics such as corporate debt in Portuguese firms, the relationship between debt and earnings quality in small- and medium-sized enterprises, and how gender diversity and managerial behavior influence financial decisions. Additionally, this reprint explores how risk management can be enhanced with new methodologies and technologies, such as artificial intelligence.

Other themes covered include sustainability in finance, financial market volatility, and the impact of economic changes in emerging countries. These contributions are vital for improving financial decision-making and promoting stability within organizations.

This reprint is an essential read for researchers, professionals, and policymakers interested in innovative and resilient financial practices.

Eulália Mota Santos and Margarida Freitas Oliveira *Guest Editors*





Editorial Special Issue "Financial Analysis, Corporate Finance and Risk Management"

Eulália Santos ^{1,*} and Margarida Freitas Oliveira ²

- ¹ School of Education and Social Sciences, Polytechnic of Leiria, Campus 1, Rua Dr. João Soares, Apartado 4045, 2411-901 Leiria, Portugal
- ² CEOS.PP, ISCAC, Polytechnic University of Coimbra & Business Research Unit (BRU-IUL), Quinta Agrícola, Bencanta, 3045-601 Coimbra, Portugal; msoliveira@iscac.pt
- * Correspondence: eulalia.santos@ipleiria.pt

1. Introduction

In an era of rapid economic transformation, evolving regulatory landscapes, and technological advancements, the need for robust financial analytics, effective corporate financial strategies, and risk management frameworks has never been more critical. Companies must navigate financial uncertainty, optimize their capital structures, and implement sustainable financial practices to ensure long-term stability and success. In response to these challenges, this Special Issue presents a collection of eighteen high-quality research contributions that explore innovative methodologies and emerging trends in these areas.

Financial analysis plays a crucial role in informed decision-making, enabling the assessment of corporate performance and understanding of an organization's financial health. In today's context, where markets face constant challenges, the evolution of financial practices and the adoption of new approaches are essential to ensuring business resilience and competitiveness.

2. A Short Review of the Contributions to This Special Issue

This Special Issue brings together eighteen significant contributions that address various aspects of financial analysis and its practical applications.

2.1. Financial Analysis and Capital Structure

The contribution titled "Determinants of Corporate Indebtedness in Portugal: An Analysis of Financial Behaviour Clusters" (Tavares et al. 2024) investigates corporate debt patterns in Portuguese firms, while "Features of the Association between Debt and Earnings Quality for Small and Medium-Sized Entities" (Sequeira et al. 2024) explores the relationship between debt and earnings quality in small and medium-sized enterprises.

2.2. Corporate Finance Strategies and Risk Management

Corporate finance strategies continue to evolve, incorporating insights from behavioral finance and sustainable finance practices. In this context, "Board Gender Diversity and Risk Management in Corporate Financing" (Askarany et al. 2025) and "Risk Management in Product Diversification: The Role of Managerial Overconfidence in Cost Stickiness" (Parsaei et al. 2024) delve into how governance structures and managerial behavior influence capital structure decisions and risk management.

Risk management remains a cornerstone of financial stability, requiring sophisticated models to measure and mitigate uncertainties. Several papers in this Special Issue contribute to this discussion, including "Mapping the Landscape of Key Performance and



Received: 19 February 2025 Accepted: 24 February 2025 Published: 26 February 2025

Citation: Santos, Eulália, and Margarida Freitas Oliveira. 2025. Special Issue "Financial Analysis, Corporate Finance and Risk Management". *Risks* 13: 43. https:// doi.org/10.3390/risks13030043

Copyright: © 2025 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https://creativecommons.org/ licenses/by/4.0/). Key Risk Indicators in Business" (Ionescu et al. 2024) and "The Role of Credit Consortia in the Financial Structure of Sardinian Companies During the SARS-CoV-2 Crisis" (Desogus et al. 2024), both of which highlight innovative approaches to managing financial risks in volatile environments. Advances in quantitative risk management (QRM) are also explored in "Uncertainty Reduction in Operational Risk Management Process" (Burstein and Zuckerman 2024), which presents novel methodologies for assessing and mitigating financial uncertainties.

2.3. Information Asymmetry and Financial Risk

Information asymmetry also stands out as a critical factor in financial markets. "Nonparametric Testing for Information Asymmetry in the Mortgage Servicing Market" (Jedidi and Dionne 2024) investigates the presence of information asymmetry in the mortgage market, while "Impact of Audit Fees on Earnings Management and Financial Risk: An Analysis of Corporate Finance Practices" (Daryaei et al. 2024) assesses the relationship between audit fees, earnings management, and financial risk.

2.4. Macroeconomics and Financial Stability

Macroeconomic issues and their implications for financial stability are explored in "The Role of Personal Remittances in Economic Development: A Comparative Analysis with Foreign Direct Investment in Lebanon" (Abou Ltaif et al. 2024a) and "Government Borrowing and South African Banks' Capital Structure: A System GMM Approach" (Mabandla and Marozva 2024), highlighting the impact of economic policies in emerging markets. "Lebanon's Economic Development Risk: Global Factors and Local Realities of the Shadow Economy Amid Financial Crisis" (Abou Ltaif et al. 2024b) examines economic risks in Lebanon in the face of the shadow economy.

2.5. Sustainable Finance and Technological Innovation

Sustainable finance and its role in shaping corporate financial decisions is another key theme in this Special Issue. "Robust Portfolio Optimization with Environmental, Social, and Corporate Governance Preference" (Escobar-Anel and Jiao 2024) investigates how sustainability considerations are increasingly integrated into financial decision-making, reinforcing the growing recognition that sustainable finance practices contribute to long-term financial resilience.

Technological advancements, particularly in artificial intelligence and machine learning, are transforming financial risk assessment. "Effective Machine Learning Techniques for Dealing with Poor Credit Data" (Nkambule et al. 2024) examines how machine learning models improve credit risk assessment, enhancing predictive accuracy and decision-making processes in financial analysis and risk management.

On the other hand, empirical studies on financial market volatility and its implications for investment strategies are explored in "Simulation of Dynamic Performance of DeFi Protocol Based on Historical Crypto Market Behavior" (Grigorova et al. 2024). This paper highlights the increasing importance of understanding market volatility, particularly in emerging sectors like decentralized finance (DeFi), and its impact on investment strategies. "A Financial Stability Model for Iraqi Companies" (Ibrahim et al. 2024) develops a financial stability model tailored to the specificities of the Iraqi economic context. "Dividend-Based Labor Remuneration and Tradable Shares in Worker Cooperatives" (Tortia 2025) examines how financial behavior and capital structure decisions impact firm outcomes and economic performance. Finally, "Shareholders in the Driver's Seat: Unraveling the Impact on Financial Performance in Latvian Fintech Companies" (Rupeika-Apoga et al. 2024) investigates how the presence of shareholders in management positions affects the financial performance of financial performance.

3. Conclusions

This Special Issue reflects the diversity of approaches and methodologies used in financial analysis, contributing to a broader understanding of the challenges and opportunities shaping the global economic landscape. The 18 articles presented here reinforce the need for innovative strategies in optimizing financial planning, risk management, and business sustainability. We are confident that this Special Issue will serve as a valuable resource for researchers, industry professionals, and policymakers, strengthening the development of more robust and resilient financial practices.

Author Contributions: Conceptualization, E.S. and M.F.O.; writing—original draft preparation, E.S. and M.F.O.; writing—review and editing, E.S. and M.F.O.; supervision, E.S. and M.F.O.; project administration, E.S. and M.F.O. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable—This editorial does not involve human or animal studies.

Informed Consent Statement: Not applicable—This editorial does not involve human studies.

Data Availability Statement: This editorial did not involve the collection or analysis of original research data.

Acknowledgments: We are grateful to MDPI for the invitation to be Guest Editors of this Special Issue and deeply appreciate the Editorial Team of the journal *Risks* for their excellent professionalism, cooperation, and dedication.

Conflicts of Interest: The authors declare no conflicts of interest.

References

- Abou Ltaif, Samar F., Simona Mihai-Yiannaki, and Alkis Thrassou. 2024a. The Role of Personal Remittances in Economic Development: A Comparative Analysis with Foreign Direct Investment in Lebanon. *Risks* 12: 176. [CrossRef]
- Abou Ltaif, Samar F., Simona Mihai-Yiannaki, and Alkis Thrassou. 2024b. Lebanon's Economic Development Risk: Global Factors and Local Realities of the Shadow Economy Amid Financial Crisis. *Risks* 12: 122. [CrossRef]
- Askarany, Davood, Soleil Jafari, Azam Pouryousof, Sona Habibi, and Hassan Yazdifar. 2025. Board Gender Diversity and Risk Management in Corporate Financing: A Study on Debt Structure and Financial Decision-Making. *Risks* 13: 11. [CrossRef]
- Burstein, Guy, and Inon Zuckerman. 2024. Uncertainty Reduction in Operational Risk Management Process. Risks 12: 77. [CrossRef]
- Daryaei, Abbas Ali, Davood Askarany, and Yasin Fattahi. 2024. Impact of Audit Fees on Earnings Management and Financial Risk: An Analysis of Corporate Finance Practices. *Risks* 12: 123. [CrossRef]
- Desogus, Marco, Enrico Sergi, and Stefano Zedda. 2024. The Role of Credit Consortia in the Financial Structure of Sardinian Companies During the SARS-CoV-2 Crisis. *Risks* 12: 190. [CrossRef]
- Escobar-Anel, Marcos, and Yiyao Jiao. 2024. Robust Portfolio Optimization with Environmental, Social, and Corporate Governance Preference. *Risks* 12: 33. [CrossRef]
- Grigorova, Iveta, Aleksandar Karamfilov, Radostin Merakov, and Aleksandar Efremov. 2024. Simulation of Dynamic Performance of DeFi Protocol Based on Historical Crypto Market Behavior. *Risks* 12: 3. [CrossRef]
- Ibrahim, Narjis Abdlkareem, Mahdi Salehi, Hussen Amran Naji Al-Refiay, and Mahmoud Lari Dashtbayaz. 2024. A Financial Stability Model for Iraqi Companies. *Risks* 12: 140. [CrossRef]
- Ionescu, Ștefan, Gabriel Dumitrescu, Corina Ioanăș, and Camelia Delcea. 2024. Mapping the Landscape of Key Performance and Key Risk Indicators in Business: A Comprehensive Bibliometric Analysis. *Risks* 12: 125. [CrossRef]
- Jedidi, Helmi, and Georges Dionne. 2024. Nonparametric Testing for Information Asymmetry in the Mortgage Servicing Market. *Risks* 12: 192. [CrossRef]
- Mabandla, Ndonwabile Zimasa, and Godfrey Marozva. 2024. Government Borrowing and South African Banks' Capital Structure: A System GMM Approach. *Risks* 12: 112. [CrossRef]
- Nkambule, Dumisani Selby, Bhekisipho Twala, and Jan Harm Christiaan Pretorius. 2024. Effective Machine Learning Techniques for Dealing with Poor Credit Data. *Risks* 12: 172. [CrossRef]

Parsaei, Mona, Davood Askarany, Mahtab Maleki, and Ali Rahmani. 2024. Risk Management in Product Diversification: The Role of Managerial Overconfidence in Cost Stickiness—Evidence from Iran. *Risks* 12: 150. [CrossRef]

Rupeika-Apoga, Ramona, Stefan Wendt, and Victoria Geyfman. 2024. Shareholders in the Driver's Seat: Unraveling the Impact on Financial Performance in Latvian Fintech Companies. *Risks* 12: 54. [CrossRef]

Sequeira, José, Cláudia Pereira, Luís Gomes, and Armindo Lima. 2024. Features of the Association between Debt and Earnings Quality for Small and Medium-Sized Entities. *Risks* 12: 32. [CrossRef]

Tavares, Fernando, Eulália Santos, Margarida Freitas Oliveira, and Luís Almeida. 2024. Determinants of Corporate Indebtedness in Portugal: An Analysis of Financial Behaviour Clusters. *Risks* 12: 91. [CrossRef]

Tortia, Ermanno C. 2025. Dividend-Based Labor Remuneration and Tradable Shares in Worker Cooperatives. Risks 13: 5. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.



Article Determinants of Corporate Indebtedness in Portugal: An Analysis of Financial Behaviour Clusters

Fernando Tavares ^{1,2,*}, Eulália Santos ³, Margarida Freitas Oliveira ⁴ and Luís Almeida ^{5,6}

- ¹ REMIT—Research on Economics, Management and Information Technologies, Department of Economics and Management, Universidade Portucalense, 4200-027 Porto, Portugal
- ² Instituto Superior Miguel Torga, Largo da Cruz de Celas No. 1, 3000-132 Coimbra, Portugal
- ³ Higher School of Education and Social Sciences, CI&DEI, Polytechnic of Leiria, Apartado 4045, 2411-901 Leiria, Portugal; eulalia.santos@ipleiria.pt
- ⁴ CEOS.PP—Polo Coimbra (ISCAC) & Business Research Unit (BRU-IUL-ISCTE), Polytechnic Institute of Coimbra, Coimbra Business School, Quinta Agrícola-Bencanta, 3045-231 Coimbra, Portugal; msoliveira@iscac.pt
- ⁵ Higher Institute of Accounting and Administration of Aveiro, Aveiro University, 3810-193 Aveiro, Portugal; gomesalmeida@ua.pt
- ⁶ GOVCOPP Unit Research, Aveiro University, 3810-193 Aveiro, Portugal
- Correspondence: ftavares@upt.pt

Abstract: Corporate indebtedness is a powerful tool in determining a company's financial health with impacts on its image and reputation. The main objective of this research is to study the determining factors in corporate indebtedness in Portugal. It also has the secondary objectives of creating clusters of companies' behaviour in relation to the use of credit and verifying their differences in relation to the characteristics of the companies. It uses a quantitative methodology based on a questionnaire survey of 1957 Portuguese companies. The results of the factor analysis show the formation of six determining factors in corporate indebtedness, namely the negotiating relationship with banks, financing, cycle and indebtedness, company operating performance, guarantees used to obtain bank financing and financing risk analysis as well as secondary forms of bank financing. The application of cluster analysis to the six factors formed led to the classification of companies into three clusters: the resilient financial cluster, the operational excellence cluster and the strategic financial cluster. There are several statistically significant differences in the corporate financing factors in relation to the clusters to which they belong. The evidence of the factors and clusters explaining company financing provides insights for improving credit access practices and for implementing public policies that facilitate access to credit and promote economic development.

Keywords: financial management; financing strategies; credit evaluation; financial structure; alternative forms of financing

1. Introduction

Financial ratios are strong indicators of a company's likelihood of bankruptcy (Magnanelli and Izzo 2017). According to Pacheco et al. (2019), the ratios that demonstrate the greatest predictive power of bankruptcy are indebtedness, solvency, return on equity, return on assets, financial autonomy and the ratios Working Capital/Total Assets; EBIT/Total Assets and Turnover/Total Assets. In addition to the financial variables, the authors refer to two non-financial variables, namely exports and the number of employees, which are negatively related to company bankruptcy in the construction sector in Portugal. This indicates that companies in this sector with a greater presence abroad are less likely to go bankrupt, as is the size effect.

For Tavares et al. (2021) and Santos et al. (2022), image risks are linked to the company's reputation, which it transfers abroad. On the other hand, these authors consider



Citation: Tavares, Fernando, Eulália Santos, Margarida Freitas Oliveira, and Luís Almeida. 2024. Determinants of Corporate Indebtedness in Portugal: An Analysis of Financial Behaviour Clusters. *Risks* 12: 91. https:// doi.org/10.3390/risks12060091

Academic Editor: Steven Haberman

Received: 5 April 2024 Revised: 12 May 2024 Accepted: 29 May 2024 Published: 31 May 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

5

that financial risks are associated with the economic and financial capacity of companies and the way they carry out their business, an idea corroborated by Almeida (2023). Financial risks are distinguished between short-term and medium- and long-term risks. In the short term, the importance of liquidity is emphasised in order for companies to maintain a positive and balanced net cash position. In the medium and long term, there is a focus on solvency, which reduces the likelihood of bankruptcy (Santos et al. 2022). According to Tavares et al. (2021), strategic planning is essential for companies, as they need to have a clear vision of the economic, social and political changes that are taking place in the ecosystem in which they operate.

Many family businesses survive over generations when management is professionalised and there is a separation between ownership and control of the company, which increases the company's chances of growth (Santos et al. 2022). The perception of operational risk positively influences the perception of financial risk in family businesses. In a family business, its longevity leads to the consolidation of the link between the company name and the family name (Tavares et al. 2015; Santos et al. 2022).

The aspects that are most evident in the perception of risk management planning are having a management plan that includes the relationship with customers, suppliers and employees (medium and long term) and also an up-to-date security plan (Tavares et al. 2023). There are statistically significant differences in risk management planning, with the larger the organisation, the greater the importance given to risk management planning (Forti et al. 2011).

Authors such as Mutamimah et al. (2021), Zhang et al. (2022), and Ogunmokun et al. (2024) argue that the provision of credit to companies is based on the analysis of corporate information to assess repayment capacity. However, the information asymmetry between banks and companies poses a challenge, making credit risk a critical concern. These same authors point to the relationship between banks and companies as one of the factors reducing information asymmetry, creating better conditions for companies to obtain credit.

Bhatia and Kumar (2023) investigated the relationship between the debt efficiency, operational efficiency and performance of Indian companies in the period between 2013 and 2019. The authors found evidence through a GLS regression model of a positive association between operational efficiency and business performance, as well as the cost of capital. They highlighted operational efficiency as a relevant factor in corporate financial performance and debt acquisition, and these findings are supported by Aziz and Abbas (2019), Boshnak (2023), and Ye et al. (2023).

The main objective of this work is to study the determining factors in corporate indebtedness in Portugal. Its secondary objectives are to create clusters of companies' behaviour in relation to the use of credit and to verify their differences in relation to the characteristics of the companies. The present study represents a significant advancement in the investigation of financing for Portuguese SMEs, notable for its originality in employing cluster analysis. By addressing existing gaps in the literature concerning this market, it provides a broader and more detailed perspective on SME financing dynamics, contributing to a more robust and informed knowledge base in this field of the Portuguese economy. In order to achieve the proposed objectives, this work is divided into five sections. In addition to this introduction, the next section reviews the literature. Then, the methodology is presented, and in the fourth section, the results are presented and discussed. Finally, the conclusions are presented.

2. Literature Review

2.1. Corporate Reputation and the Business Relationship with Banks

For Mehtap and Kokalan (2013), corporate reputation is an increasingly emphasised issue in management and organisational studies and is considered a very valuable intangible asset that is determined by the evaluations and perceptions of the parties. Thus, according to the authors, different groups and stakeholders can perceive a company's reputation from different points of view and different perspectives.

Companies whose boards are larger become less effective because problems and parasitism arise and decisions are more difficult to make in a timely manner (O'Sullivan et al. 2016). But, on the other hand, with a larger board it is easier to detect the sometimes opportunistic behaviour of managers. However, too many directors and managers become difficult to control and will jeopardise performance (Forti et al. 2011). The presence of women on boards of directors reduces debt costs and perceptions of default risk by fund providers, while board independence and board size do not have a significant impact on debt costs (Aksoy and Yilmaz 2023). The results indicate that corporate social responsibility reporting has a positive association with debt access and a negative association with its cost of debt (Uyar et al. 2024).

2.2. Financing, Growth Cycle and Indebtedness

According to Rodrigues et al. (2018), earnings before interest on taxes (EBIT) are not relevant to indebtedness, but they are relevant to investment. For the authors, the same can be said for return on equity (ROE). However, the tangibility of the asset, the cost of debt and the value of the asset are relevant when it comes to debt. Rosa et al. (2018) assessed the impact of corporate social performance (CSP) on the cost of debt capital and leverage. The authors found that the level of corporate social performance significantly affects the cost of debt capital and leverage in times of stability, because in times of crisis, companies are expected to focus their efforts on maintaining profitability and the manifestation of corporate social responsibility is generally understood by creditors as a strategy that companies adopt to hide negative performance. According to Rodrigues et al. (2018), the tangibility of assets is important for accessing debt. The cost of debt (interest payable) restricts access to debt. According to Rosa et al. (2018), the higher the social performance, the better they are able to grow, in their opinion, because they disclose more information about future strategies. Creating a more transparent and cooperative corporate culture reduces suspicions of opportunistic behaviour, creates trust between stakeholders and promotes involvement, thereby reducing information asymmetries, agency costs and capital restrictions. The relationship between the cost of debt and indebtedness naturally involves the company's ability to pay. Although companies always seek the best composition of the capital structure, their freedom to achieve an optimal capital structure faces problems due to the set of restrictions that characterise imperfect capital markets (Rodrigues et al. 2018).

Rosa et al. (2018) observed that companies with higher CSP not only have a stronger corporate image but also create expectations of better ongoing financial performance and lower exposure to risk. In times of economic stability, lenders tend to perceive CSP as a strategic project and companies that adopt this approach can obtain loans at lower interest rates. According to Lemma (2015), high levels of perceived corruption lead companies to use higher levels of perceived corruption lead companies to use higher levels of short-term leverage, lower levels of long-term leverage and debt with shorter maturities.

For Ghouma et al. (2018), a stronger board of directors (in terms of composition and structure) ensures greater protection of investors' rights and therefore reduces agency problems within the company. These reduced agency problems within the company represent an important aspect that allows companies to enjoy a lower cost of debt financing. Lower information asymmetry, through the company's disclosure policy, also explains the negative relationship between the cost of debt and the quality of corporate governance (Tavares et al. 2023). Rosa et al. (2018) also found a positive correlation between company size and corporate social performance and that the impact of corporate social performance on the cost of debt capital is more significant for small companies. The authors understand that lenders perceive that small companies are generally less inclined to intervene in corporate social performance due to budgetary constraints, so they are more willing to finance small companies when they decide to engage in corporate social performance.

Magnanelli and Izzo (2017) conclude that there is a positive relationship between corporate social performance and the cost of debt, demonstrating that corporate social responsibility is a value factor with an impact on the company's risk profile. Companies in countries with higher levels of perceived leverage in terms of corruption respond to the weaknesses of enforcement institutions through greater concentration of ownership and greater participation in the control block (Chai et al. 2019). Corruption makes law enforcement institutions ineffective and therefore distorts corporate governance (Lemma 2015).

Magnanelli and Izzo (2017) recommend publishing and disseminating non-financial information about the company, as this strengthens relations between key stakeholders and can stabilise the market. The effective implementation of corporate social responsibility activities has an effect on risk reduction (Pacheco et al. 2019; Tavares et al. 2021). Magnanelli and Izzo (2017) conclude that the higher the risk of companies, the higher the cost applied by banks, and the better the reputation for responsibility or performance of companies, the lower the risk and consequently the cost of debt financing.

2.3. Company Operating Performance

According to Forti et al. (2011), performance measured by return on assets (ROA) and market value assessed by the market-to-book ratio are positively affected by companies' cash retention. They also consider, with regard to the relationship between cash management and business performance, that effective cash management is an important contribution to maximising operating profit. In this way, companies that adopt efficient cash management that is consistent with a contemporary view of the subject have a better chance of prospering in today's competitive environment.

According to Soares et al. (2012), companies' financial statements and public information can be used to determine the credit ratings of Brazilian companies with reasonable accuracy. According to the authors, an increase in the generation of operating income over total financial expenses reduces credit risk and increases the credit rating.

The characteristics of loans to small- and medium-sized enterprises (size of the loan, length of the loan, information required that is not always complete) require concern on the part of financial institutions in order to assess their creditworthiness (Chai et al. 2019). Thus, because of the asymmetric information and nature of loans to companies, it is very difficult for financial institutions to accurately assess their credit risk, and restricted access to finance hinders the development of companies.

Yazdanfar and Öhman (2015) confirm that debt ratios in terms of trade credit, shortterm debt and long-term debt negatively affect the performance of companies in terms of profitability. On the other hand, a high debt ratio increases agency costs and the risk of losing control of the company, owners and managers of small and medium enterprises (SMEs), which tend to finance their businesses with equity to a fairly high degree, according to Rosa et al. (2018). Chai et al. (2019) believe that due to the very different nature of the various sectors and the great variation in the size and terms of loans between the different sectors, it is not appropriate to assess credit risk for all companies using the same credit rating system. These authors have shown that for small companies, non-financial factors are more influential in predicting default risks than financial factors.

Yazdanfar and Öhman (2015) confirmed that profitable SMEs tend to use equity and retained earnings more than external financing. They also found that SMEs use short-term debt more than medium- and long-term debt, which can reduce costs related to information asymmetry and agency conflicts. The lower the level of leverage, the lower the agency costs of external investments, the lower the debt and the higher the company's profitability (Rosa et al. 2018). According to Yazdanfar and Öhman (2015), there is a negative relationship between age and profitability, which can be explained by the life cycle theory.

Ugwueze et al. (2019) point out that in the presence of asymmetric information, a company is financed more by internally generated funds than by external funds. The authors conclude that long-term debt and retained earnings are not statistically significant in

turnover, unlike short-term debt. To reduce information asymmetry and moral hazards between SMEs and financial institutions such as banks, SME owners and managers should also be aware of the importance of transfer measures that can improve their relationship with financial institutions in the business environment (Yazdanfar and Öhman 2015). According to Amarna et al. (2024), in a study conducted on European companies, investors negatively value Environmental Social Governance disclosure, leading to an increase in the cost of equity capital, while creditors positively value it, resulting in a reduction in the cost of debt.

2.4. Guarantees Used to Obtain Bank Financing

The deficiency in the process of analysing companies' credit leads to the existence of non-performing loans (Manuel et al. 2018). According to Jucá et al. (2016), the higher the levels of assets that can be used as collateral, the higher the level of financial leverage; the higher the level of non-current assets (fixed assets), the higher the maturity of the debts; the higher the level of current assets (stocks and accounts receivable), the lower the maturity of the debts and the assets that can be used as collateral are more relevant for companies that contract bank debts. Thus, the authors conclude that assets that can be used as collateral are of fundamental importance in contracting long-term debt, while stocks and receivables are more geared towards making short-term debt viable. These assets are more relevant for companies that do not have access to the capital markets, i.e., that rely on bank debt (Tavares et al. 2015). Sometimes some investment projects do not meet the necessary conditions for approval, but when it comes to projects linked to state initiatives, public banks are called upon to support them (Manuel et al. 2018). In their study, Ogunmokun et al. (2024) found that during periods of economic restructuring, the lack of government intervention and the presence of banks have a negative and significant impact on loans to small businesses.

Ono and Uesugi (2009) consider that the use of collateral is complementary to relationshipbased financing and point out that collateral plays an important role in facilitating the relationship between borrowers and lenders. In this way, they understand that banks whose claims are backed by collateral monitor more intensively, and that borrowers who have a long-term relationship with their main banks are more likely to offer collateral. Unlike collateral, personal guarantees have a weaker complementarity with relationship-based financing (Manuel et al. 2018). For Calcagnini et al. (2014) the presence of guarantees, mainly collateral, affects the cost of credit for Italian companies, systematically reducing interest rates on loans. Guarantees are a more powerful instrument for riskier borrowers than for safer ones (Jucá et al. 2016).

Ono and Uesugi (2009) found that personal guarantees are more often required of smaller, more owner-orientated companies, with the aim of limiting the risk of mixing the personal assets of business representatives. According to Calcagnini et al. (2014), the likelihood of secured loans increases with the company's higher risk perception and during economic crises. Thus, the probability of posting guarantees is lower the longer the relationship between banks and clients and the greater the number of banks for each company.

Ono and Uesugi (2009) found that the use of collateral was more likely to be associated with the more frequent submission of relevant documents and a longer-term relationship with the bank. For Mayordomo et al. (2021), the use of personal guarantees and collateral by a major Spanish banking group and its subsidiaries increased to improve the credit risk of loans and meet regulatory capital requirements. Personal guarantees were required more often than collateral, especially in subsidiaries with lower capital ratios (Ugwueze et al. 2019). Mayordomo et al. (2021) documented that subsidiaries with lower capital ratios significantly increased the use of personal guarantees after the changes in the collateral policy, highlighting the importance of these changes for the financial institution as a whole and for each of the subsidiaries. On the other hand, Calcagnini et al. (2014) found that during the financial crisis, guarantees helped banks to improve their screening of companies,

which resulted in lower interest rates for guaranteed loans. Interest rate variability is higher among larger banks (Manuel et al. 2018).

2.5. Analysing Financing Risk

Tavares et al. (2015) analysed the forms of financing and products most used by SMEs in the Porto district of Portugal. In their study, they found that the most commonly used sources of finance are bank credit, self-financing and, to a lesser extent, European Union funds. As for the importance given to the products used by companies, they concluded that the most important products identified by companies are bank current accounts, medium- and long-term loans, leasing and/or renting, financing of foreign operations, bank guarantees and short-term loans, among other things.

There are signs that better governance practices reduce credit constraints for companies, i.e., companies that adopt better governance practices have greater access to resources in the financial market (Cicogna et al. 2007). The continuous innovation of financial products is a general reality in the financial services industry and is important for maintaining competitiveness (Li and Li 2021). Van Osnabrugge (2000) concludes that venture capitalists are very concerned about exiting their investments in the future and also have higher expected and partially realised rates of return on their venture investments than business angel investors. With regard to the relevance of guarantees for obtaining financing, Tavares et al. (2015) found that the most commonly used guarantees are: the promissory note subscribed by the company with a personal guarantee, the promissory note subscribed by the company, the mutual guarantee, the mortgage guarantee and the pledge of securities.

2.6. Secondary Forms of Bank Financing

Gonçalves et al. (2018), in a study carried out in Brazil, found that self-financing, followed by indebtedness, is the most-used primary source of financing. According to Myers and Majluf (1984), financing begins with the use of equity, followed by external financing through the issue of debt and finally the issue of shares. This hierarchy is justified by the fact that internally generated resources (retained earnings) have no transaction costs for the company to finance itself.

Increased credit risk reduces bank profits (Manuel et al. 2018). Managers should therefore pay more attention to the role of cost efficiency and risk-taking (Belkhaoui et al. 2020).

According to Mura and Buleca (2012), in order to secure investment projects and increase performance, the company needs to finance its own activity. If the company does not have sufficient sources of its own, it will try to obtain it from external sources (Santos et al. 2022).

3. Methodology

3.1. Characterisation of the Population and Sample

The target population for this study is Portuguese companies with a turnover of more than EUR 500,000.00, whose emails were obtained from the SABI database (Iberian Balance Sheet Analysis System). The sampling method used was non-probabilistic for convenience, as only companies included in this database were used. It should be noted that although the sampling method is not probabilistic, all companies that received the invitation by email have the same probability of responding to the questionnaire. The study population comprises Portuguese companies with sales volumes exceeding EUR 500,000.00 in the year 2021, totalling approximately 60,000 companies, from which 1957 valid responses were obtained.

The sample consists of 1957 Portuguese companies that responded to the questionnaire. Regarding the characterisation of the questionnaire respondents, they were mostly male (62.1%, n = 1216) and aged between 20 and 85 years, with an average of approximately 46 years (SD = 10.35). Regarding the roles they perform in their companies, 33.3% (n = 652) were managing partners, 26.3% (n = 514) were administrative, 23.2% (n = 455) were administrators or managers and 17.2% (n = 336) were directors. On average, these companies have been on the market for approximately 25 years (SD = 19.36), with 50% of companies operating on the market for 21 years or less.

Table 1 presents the elements that characterise the responding companies. Thus, of the 1957 companies that participated in the study, the majority (50.6%, n = 991) are characterised by being small companies. The companies belonged to different areas of activity, with commerce, services and industry representing 67.9% (n = 1330) of responding companies, but only 17.3% (n = 340) of companies operated in the international market.

Ta	ble	1.	Cł	naract	erisa	tion	of	com	panies.
----	-----	----	----	--------	-------	------	----	-----	---------

Variables	Categories	n	%
	Less than 10	633	32.3
Number of employees	From 10 to 49	991	50.6
Number of employees	From 50 to 249	275	14.1
	More than 250	58	3.0
_	Business	522	26.7
	Services	449	22.9
	Manufacturing industry	359	18.3
	Construction	343	17.5
Sector / area of activity commerce	Tourism	96	4.9
Sector / area of activity commerce	Transport	68	3.5
	Restoration	46	2.4
	Agriculture	34	1.7
	Gas, electricity, water	27	1.4
	Extractive industry	13	0.7
_	National market	870	44.5
Main market where the company has	Regional market	401	20.5
the highest sales volume	Local market	346	17.7
	International market	340	17.3
Tune of company	Family	1328	67.9
Type of company	Unfamiliar	629	32.1
_	Quota company	1603	81.9
Type of society	Anonymous society	349	17.8
Type of society	Cooperative	3	0.2
	Limited partnership	1	0.1
	Rapid growth	35	1.8
Company life cycle phase	Healthy and growing	776	39.7
Company me cycle phase	Stabilised	967	49.4
	Decline	173	8.8
Accounts cortified by an external entity	Yes	1247	63.7
Accounts certified by an external entity	No	710	36.3

The majority of companies surveyed were family-owned (67.9%, n = 1328) and the two types of company that stand out most are the limited liability company (81.9%, n = 1603) and the public limited company (17.8%, n = 349). As for companies' life cycle phases, in general, they were stabilised (49.4%, n = 967) or healthy and fast growing (41.5%, n = 811). Regarding certification, the majority (63.7%, n = 710) had accounts certified by an external entity.

3.2. Data Collection Instruments

The instrument used was a questionnaire survey composed of two parts. In the first part, the respondents were characterised (age, sex and role they perform in the company) and the companies were characterised (number of employees, sector or area of activity, year of founding of the company, main market where it has the largest sales volume, type of company, type of society, phases of the company's life cycle, certification). In the second part, we analysed the determining factors in corporate debt.

To measure the importance of determining factors in corporate debt, 40 items were used (see Table 2 for a more detailed understanding of the items and corresponding variables), constructed while taking into account the works of (Tavares et al. 2015, 2021, 2023) and using a 5-point Likert scale (1—Not at all important; 5—Very important).

Table 2. Matrix of factor loadings of the determining factors in corporate debt.

	Loadings
Factor 1: Business relationship with banks ($\alpha = 0.913$)	
R4. Increase negotiating power with banks	0.792
R6. Have the option to choose which bank you want to work with	0.770
R3. Possibility of reducing interest rates charged	0.756
R7. Caution about a financial institution	0.752
R2. Diversity of advice to support the activity	0.746
R5. Possibility of obtaining higher amounts of financing	0.732
R8. Increase acceptance of credit applications	0.723
R1. Variety of products and services offered	0.718
R9. Share the risk across more than one bank	0.695
Factor 2: Financing, life cycle and debt ($\alpha = 0.887$)	
F4. The debt capital/own capital ratio of other companies in the same sector of activity	0.772
F5. The business life cycle phase	0.741
F7. The fact that new financing signals to competitors that the company will not reduce installed	0 733
capacity or production	0.755
F3. The fact that the use of foreign capital is limited provides customers and suppliers with a stable	0 720
financial situation	0.720
F6. The volatility of cash flows	0.709
F8. A new level of debt signals the company's ability to grow results	0.680
F1. The fact that the level of debt discourages potential buyers	0.657
F9. The rating classification	0.649
F2. The nature of your business	0.642
Factor 3: Company operational performance ($\alpha = 0.849$)	
D3. Standard operating procedures are helpful to employees at work	0.823
D5. Instructions for the operations to be carried out are important for employees at work	0.818
D4. Harmonious working relationships are important to the company	0.807
D2. Company regulations inform employees about what is expected of them	0.746
D1. Details of job requirements and instructions are important	0.650
D6. Preserving public image is one of the company's main policies	0.635
Factor 4: Guarantees used to obtain bank financing ($\alpha = 0.838$)	
G2. Promissory note signed by the company with personal guarantee	0.784
G4. Mutual guarantee	0.782
G1. Declaration signed by your company	0.731
G5. Financing using mutual guarantee	0.681
G3. Mortgage guarantee	0.645
Factor 5: Financing risk analysis ($\alpha = 0.820$)	
A2. Analysis of managers' hobbies	0.820
A3. Way of life of company managers	0.818
A4. Analysis of the political exposure of company managers	0.707
A5. Company succession assured	0.701
A1. Duration of the relationship with suppliers	0.645
Factor 6: Secondary forms of bank financing ($\alpha = 0.800$)	
FS2. Loan discount	0.740
FS1. Factoring	0.738
FS3. Discount on promissory notes	0.733
FS5. Confirming	0.700
FS4. Financing/foreign operations	0.633

Source: Own elaboration.

3.3. Data Collection and Analysis Procedures

After a literature review, a questionnaire was created and subjected to a pre-test. After carrying out the pre-test, the proposed suggestions were taken into consideration and the questionnaire was created using the Google Forms tool. Subsequently, between the months of March and May 2021, the questionnaire link was sent to the companies that make up the population under analysis, accompanied by a short text, inviting the companies to participate in the present study. In this invitation, the objectives of the study were defined and the anonymity and confidentiality of the information provided were guaranteed.

After data collection, the IBM SPSS Statistics 28 software was used. To characterise the company representatives in the responses to the questionnaire and characterise the companies, descriptive statistics techniques were used (descriptive measures and absolute and relative frequencies). According to Kline (2016), the sensitivity of the items was initially analysed using the asymmetry ($|Sk| \leq 3$) and flatness ($|Ku| \leq 10$) coefficients.

The suitability of applying an exploratory factor analysis (AFE) to items that assess the determining factors in corporate debt was assessed using the KMO—the Kaiser–Meyer– Olkin index—and Bartlett's sphericity test. According to Hair et al. (2014), KMO values greater than 0.8 reveal a good adequacy of the sample and a statistically significant Bartlett's test of sphericity (p < 0.05) indicates that there are sufficient correlations between the items to continue the analysis. To obtain the determining factors in corporate debt, EFA was applied, using the main components method and the Kaiser criterion (eigenvalues greater than 1) and the varimax method were used to extract the factors. According to Hair et al. (2014), factor loadings must be above 0.50 to translate more than 25% of the variable's variance.

The reliability of the factors obtained with the application of EFA were evaluated using the calculation of the Cronbach's alpha value. Values greater than 0.7 are considered acceptable (Hair et al. 2014).

After defining the determining factors of corporate debt, cluster analysis was applied in order to group companies into groups with common characteristics. Finally, to compare, in the clusters found, the determining factors of corporate debt between the various characteristics of the companies, Student's *t*-test for independent samples and Welch's test were applied. In cases where the intention is to compare two independent samples, Student's *t*-test was used, and in cases where it is intended to compare more than two independent samples, ANOVA or Welch's test was used in cases of the violation of the assumption of homogeneity of variances (Marôco 2018).

4. Results and Discussion

4.1. Determining Factors in Business Debt

The Kaiser–Meyer–Olkin (KMO) sampling measure and Bartlett's sphericity test revealed a very good sample adequacy ($\chi^2(741) = 37830.95$, p < 0.001, KMO = 0.902) for the application of exploratory factor analysis to the 39 items that make up the determining factors in business debt (Hair et al. 2014).

Table 2 shows that six factors were retained, which together explain 58.50% of the total variance. The first factor explains 13.70% of the variance and is the business relationship with banks (items R1, R2, R3, R4, R5, R6, R7, R8 and R9), as it is composed of items that address aspects related to the importance that companies give to the closeness of their relationship with banks.

The second factor refers to items related to a company's financing decision, are related to the company's life cycle and also take into account debt at the beginning and end of the financing operation (items F1, F2, F3, F4, F5, F6, F7, F8 and F9). This factor explains 12.42% of the variance and is known as financing, cycle and debt.

The third factor is the company's operational performance (items D1, D2, D3, D4, D5 and D6) and explains 9.23% of the variance. The items in this factor address aspects related to the importance that companies give to their operational part and to their good structuring and consolidation. The fourth factor explains 7.98% of the variance and involves guarantees used to obtain bank financing (items G1, G2, G3, G4 and G5). The items in this

factor refer to essential guarantees, or of first concern, used in company financing. The fifth factor is financing risk analysis (items A1, A2, A3, A4, A5 and A6) and explains 7.76% of the variance. The items in this factor are related to what companies do not consider important when analysing the risk of financing a company, that is, they are aspects that companies tend to underestimate when taking out financing.

The sixth, and last, factor is secondary forms of bank financing (items FS1, FS2, FS3, FS4 and FS5) and explains 7.41% of the variance. The items in this factor relate to the use of bills and promissory notes, factoring and other forms of financing.

Regarding the loading values of all items, these are greater than 0.5 (minimum value of 0.633), which is in accordance with Hair et al. (2014). Regarding the reliability of each factor, this is considered, according to Hair et al. (2014), very good for the factor of business relationship with banks (factor 1: $\alpha = 0.913$) and good for the remaining factors (factor 2: $\alpha = 0.887$; factor 3: $\alpha = 0.849$, factor 4: $\alpha = 0.838$, factor 5: $\alpha = 0.820$ and factor 6: $\alpha = 0.800$).

These results aggregate the determining factors of corporate debt. In short, it can be seen that there are six factors that as a whole are considered when companies resort to debt. Thus, when companies resort to debt, the company's business relationship with banks, its life cycle and its operational performance, as well as the guarantees required by the bank, are decisive, with the need to carry out a risk analysis financing, as well as secondary forms of bank financing.

4.2. Cluster Analysis

The dendrogram in Figure 1 shows that the six determining factors in corporate debt lead to the formation of three clusters.



Figure 1. Dendrogram of clusters related to the determining factors in corporate indebtedness. Source: Own elaboration.

Cluster 1 in Figure 2 contains the largest number of companies (n = 946, 48.3%) and presents high levels in the factors of business relationship with banks (M = 3.88, SD = 0.56) and company operational performance (M = 4.09, SD = 0.49). As for the financing factor, the cycle and debt factor presents reasonable levels (M = 3.28, SD = 0.66). In the remaining factors, the levels are low, falling below the midpoint of the scale used (three points). Thus, cluster 1 will be called the "Resilient Financial Cluster", as it has a strong relationship with banks and excellent operational performance. The authors Mehtap and Kokalan (2013) and O'Sullivan et al. (2016) mentioned that corporate reputation is a very valuable intangible asset and this helps to maintain a good relationship with banks. Soares et al. (2012) and Chai et al. (2019) refer to the importance of the operational part of the company being healthy, and that in small- and medium-sized companies, non-financial factors are more influential than financial factors.



Figure 2. Averages of the determining factors in corporate debt by cluster. Legend: Cluster 1— Resilient Financial Cluster; Cluster 2—Operational Excellence Cluster; Cluster 3—Strategic Financial Cluster; NBR—negotiating relations with banks; FLCI—financing, life cycle and indebtedness; COP— Company's operational performance; GUBF—guarantees used to obtain bank financing; FRA financing risk analysis; SFBF—secondary forms of bank financing. Source: Own preparation.

Cluster 2 is made up of 388 (19.8%) companies and presents a high level only in the company's operational performance factor (M = 3.88, SD = 0.72). In all other factors, the levels are below the midpoint of the scale. Therefore, cluster 2 will be called the "Operational Excellence Cluster", reflecting its prominence only in the company's operational performance factor, while the other factors present levels below average. Santos et al. (2022) state that the perception of operational risk positively influences the perception of financial risk. Also, for Soares et al. (2012) an increase in operating income over total financial expenses reduces credit risk and increases the credit rating.

Cluster 3 consists of 623 (31.8%) companies and is characterised by presenting high average levels in four of the determining factors: business relationship with banks (M = 4.09, SD = 0.50); financing, cycle and debt (M = 3.51, SD = 0.61); company operational performance (M = 4.09, SD = 0.60); and guarantees used to obtain bank financing (M = 3.54, SD = 0.66). The factor of financing risk analysis is slightly above the midpoint of the scale (M = 3.09, SD = 0.71) and the factor of secondary forms of bank financing presents low levels (M = 2.77, SD = 0.91). Taking the above into account, cluster 3 will be called the "Strategic Financial Cluster", as it stands out by presenting high average levels in four determining factors, indicating a strategic and balanced approach in relation to financial and operational aspects. Strategic planning is essential for companies. Authors such as Tavares et al. (2023), Rodrigues et al. (2018) and Pacheco et al. (2019) allude in their work to strategic planning and risk management, as well as their alignment with the companies' life cycle.

It should be noted that in all clusters the company's operational performance factor presents high levels.

4.3. Comparative Analysis by Clusters

In Table 3, after applying Student's *t*-test for independent samples, it was observed that, in the Resilient Financial Cluster, there are statistically significant differences in the financing risk analysis factor between family and non-family companies (t = 2.052, p < 0.05), with family companies being those that attribute greater importance to this factor. In the Strategic Financial Cluster, there are statistically significant differences in the factor of secondary forms of bank financing between family and non-family companies (t = -3.022,

p < 0.01), with non-family companies being those that attribute greater importance to this factor. This verification is in line with what was mentioned by Santos et al. (2022).

Table 3. Comparative analysis of the determining factors in corporate indebtedness between family and non-family companies by cluster.

Clusters	Factors		Family			t-Test		
	-	n	M	SD	п	M	SD	
	NBR	641	3.89	0.56	305	3.86	0.57	0.663
	FLCI	641	3.30	0.66	305	3.23	0.66	1.576
Cluster	COP	641	4.10	0.49	305	4.08	0.48	0.574
Financial	GUBF	641	2.54	1.00	305	2.47	1.00	1.022
Resilient	FRA	641	2.29	0.84	305	2.17	0.76	2.052 *
	SFBF	641	1.81	0.72	305	1.90	0.70	-1.843
	NBR	247	2.81	1.09	141	2.98	1.00	-1.517
Cluster of	FLCI	247	2.38	0.95	141	2.44	0.94	-0.648
Evcellence	COP	247	3.84	0.75	141	3.94	0.67	-1.350
Operational	GUBF	247	1.34	0.42	141	1.36	0.50	-0.466
operational	FRA	247	1.82	0.81	141	1.75	0.77	0.829
	SFBF	247	1.29	0.50	141	1.27	0.47	0.408
	NBR	440	4.09	0.50	183	4.10	0.51	-0.336
Cluster	FLCI	440	3.52	0.61	183	3.51	0.63	0.155
Financial	COP	440	4.09	0.59	183	4.07	0.64	0.344
Strategic	GUBF	440	3.52	0.66	183	3.59	0.66	-1.228
Strategic	FRA	440	3.07	0.69	183	3.15	0.75	-1.204
	SFBF	440	2.70	0.88	183	2.94	0.96	-3.022 **

Legend: NBR—negotiating relations with banks; FLCI—financing, life cycle and indebtedness; COP—company's operational performance; GUBF—guarantees used to obtain bank financing; FRA—financing risk analysis; SFBF—secondary forms of bank financing. Note: ** p < 0.01; * p < 0.05. Source: Own preparation.

In Table 4, we intend to compare the determining factors of corporate debt between the ages of the companies, by cluster. To this end, two groups were considered, one encompassing companies aged less than 25 years and the other with the remaining companies (the value of 25 is justified as it is the average value of the companies' ages). Thus, as we want to compare two age groups, we used Student's *t*-test for independent samples, and it was observed that, in the Operational Excellence Cluster, there were statistically significant differences in the financing, cycle and debt factor (t = -1.975, p < 0.05), with companies under 25 years old being those that attribute greater importance to this factor. In the Strategic Financial Cluster, there were statistically significant differences in the secondary forms of bank financing factor (t = -2.117, p < 0.05), with companies under 25 years old being those that attribute greater importance sunder 25 years old being those that attribute greater importance sunder 25 years old being those that attribute greater importance to this factor.

In Table 5, we intend to compare the determining factors of corporate debt between certified and non-certified companies, by cluster. Student's *t*-test for independent samples was applied, as the aim was to compare two groups of companies (certified and non-certified). The company's operational development factor is the only one that presents statistically significant differences in the three clusters under analysis, with certified companies being those that attribute greater importance to this factor.

				Compa	ny Age				
Clusters	Factors	Un	der 25 Years	Old	25	Years or Old	ler	t-Test	
	-	n	М	SD	n	М	SD		
	NBR	571	3.88	0.58	375	3.87	0.54	-0.272	
	FLCI	571	3.29	0.64	375	3.26	0.68	-0.638	
Cluster	COP	571	4.09	0.49	375	4.09	0.48	0.092	
Financial	GUBF	571	2.52	0.98	375	2.51	1.03	-0.154	
Resilient	FRA	571	2.24	0.83	375	2.27	0.79	0.576	
	SFBF	571	1.87	0.74	375	1.80	0.67	-1.585	
	NBR	226	2.88	1.09	162	2.87	1.03	-0.079	
Cluster of	FLCI	226	2.48	0.94	162	2.29	0.96	-1.975 *	
Excellence	COP	226	3.86	0.71	162	3.90	0.74	0.497	
Operational	GUBF	226	1.37	0.47	162	1.30	0.42	-1.602	
Operational	FRA	226	1.78	0.77	162	1.82	0.83	0.461	
	SFBF	226	1.30	0.46	162	1.26	0.53	-0.805	
	NBR	387	4.08	0.52	236	4.10	0.48	0.522	
Cluster	FLCI	387	3.53	0.60	236	3.49	0.64	-0.797	
Cluster	COP	387	4.10	0.61	236	4.07	0.59	-0.564	
Stratogia	GUBF	387	3.56	0.62	236	3.51	0.72	-0.833	
Sualegic	FRA	387	3.11	0.71	236	3.06	0.71	-0.937	
	SFBF	387	2.83	0.93	236	2.67	0.86	-2.117 *	

Table 4. Comparative analysis of the determining factors in corporate indebtedness between compa-nies under 25 years old and 25 years old or more by cluster.

Legend: NBR—negotiating relations with banks; FLCI—financing, life cycle and indebtedness; COP—company's operational performance; GUBF—guarantees used to obtain bank financing; FRA—financing risk analysis; SFBF—secondary forms of bank financing. Note: * p < 0.05. Source: Own preparation.

Table 5. Comparative analysis of the determining factors in corporate indebtedness between certified and non-certified companies by cluster.

Clusters	Factors		Yes			<i>t-</i> Test		
	-	п	M	SD	п	M	SD	-
	NBR	606	3.91	0.54	340	3.83	0.59	1.954
	FLCI	606	3.30	0.63	340	3.25	0.71	0.909
Cluster	COP	606	4.14	0.47	340	4.01	0.51	3.982 ***
Financial	GUBF	606	2.50	1.01	340	2.55	0.99	-0.725
Resilient	FRA	606	2.23	0.77	340	2.28	0.89	-0.880
	SFBF	606	1.87	0.71	340	1.79	0.72	1.841
	NBR	227	2.90	1.05	161	2.84	1.08	0.512
Cluster of	FLCI	227	2.36	0.93	161	2.46	0.97	-1.058
Excellence	COP	227	3.96	0.68	161	3.75	0.77	2.840 **
Operational	GUBF	227	1.34	0.45	161	1.35	0.46	-0.212
Operational	FRA	227	1.82	0.78	161	1.77	0.82	0.548
	SFBF	227	1.29	0.52	161	1.28	0.45	0.201
	NBR	414	4.09	0.52	209	4.10	0.46	-0.298
Cluster	FLCI	414	3.54	0.63	209	3.47	0.57	1.203
Financial	COP	414	4.13	0.60	209	4.01	0.60	2.273 *
Strategic	GUBF	414	3.55	0.67	209	3.51	0.64	0.667
Suategic	FRA	414	3.10	0.72	209	3.07	0.69	0.541
	SFBF	414	2.74	0.90	209	2.82	0.92	-0.970

Legend: NBR—negotiating relations with banks; FLCI—financing, life cycle and indebtedness; COP—company's operational performance; GUBF—guarantees used to obtain bank financing; FRA—financing risk analysis; SFBF—secondary forms of bank financing. Note: *** p < 0.001; ** p < 0.01; * p < 0.05. Source: Own preparation.

From the analysis of Table 6 and applying Student's *t*-test for independent samples, it is observed that, in the financing risk analysis factor, there are statistically significant differences in both the Resilient Financial Cluster (t = -3.470, p < 0.01) and the Strategic Financial Cluster (t = -2.288, p < 0.05), with companies with less than 50 employees being those that attach greater importance to this factor. In the Resilient Financial Cluster, there are also statistically significant differences in the business relationship with banks factor (t = 2.325, p < 0.05), with companies with 50 or more employees being those that attribute greater importance to this factor. In the Strategic Financial Cluster, there were statistically significant differences in the guarantees used to obtain bank financing factor (t = 2.095, p < 0.05), with companies with 50 or more employees attaching greater importance to this factor. It appears that small companies are more focused on resilience and larger companies are more focused on strategic planning.

Table 6. Comparative analysis of the determining factors in corporate indebtedness between companies with less than 50 employees and companies with 50 or more by cluster.

Clusters	Factors	Less	Than 50 Wo	rkers	50 c	or More Worl	kers	<i>t-</i> Test
	-	п	M	SD	п	M	SD	-
	NBR	788	3.86	0.57	158	3.97	0.51	2.325 *
~	FLCI	788	3.29	0.67	158	3.24	0.57	-0.957
Cluster	COP	788	4.08	0.49	158	4.15	0.47	1.600
Financial	GUBF	788	2.52	0.99	158	2.52	1.04	-0.053
Kesilient	FRA	788	2.28	0.85	158	2.08	0.62	-3.470 **
	SFBF	788	1.84	0.72	158	1.88	0.67	0.661
	NBR	326	2.84	1.08	62	3.05	0.93	1.540
Cluster of	FLCI	326	2.41	0.96	62	2.32	0.91	-0.694
Excellence	COP	326	3.86	0.74	62	3.95	0.61	0.814
Operational	GUBF	326	1.35	0.47	62	1.30	0.36	-0.823
Operational	FRA	326	1.82	0.83	62	1.67	0.56	-1.790
	SFBF	326	1.29	0.51	62	1.24	0.36	-0.979
	NBR	510	4.09	0.52	113	4.11	0.45	0.389
Cluster	FLCI	510	3.51	0.62	113	3.55	0.58	0.688
Financial	COP	510	4.08	0.62	113	4.11	0.54	0.466
Stratogic	GUBF	510	3.51	0.66	113	3.65	0.65	2.095 *
Juacegie	FRA	510	3.12	0.72	113	2.96	0.65	-2.288*
	SFBF	510	2.78	0.92	113	2.74	0.83	-0.338

Legend: NBR—negotiating relations with banks; FLCI—financing, life cycle and indebtedness; COP—company's operational performance; GUBF—guarantees used to obtain bank financing; FRA—financing risk analysis; SFBF—secondary forms of bank financing. Note: ** p < 0.01; * p < 0.05. Source: Own preparation.

Table 7 shows that, by applying Student's *t*-test for independent samples, there are statistically significant differences in the company's operating performance factor in both the Resilient Financial Cluster (t = -2.846, p < 0.01) and the Strategic Financial Cluster (t = -2.327, p < 0.05), with companies whose life cycle is in the stabilised or declining phases attaching greater importance to this factor. In the Resilient Financial Cluster, there are also statistically significant differences in the business relationship with banks factor (t = -2.010, p < 0.05), with companies whose life cycle is in the stabilised or declining phases being those that attribute greater importance to this factor. In the Strategic Financial Cluster, there were statistically significant differences in the business relationship with banks factor (t = -2.010, p < 0.05), with companies whose life cycle is in the stabilised or declining phases being those that attribute greater importance to this factor. In the Strategic Financial Cluster, there were statistically significant differences in the factor of guarantees used to obtain bank financing (t = 3.157, p < 0.01), with companies whose life cycle is in the phases of rapid and healthy growth being those that attribute greater importance to this factor. It appears that the importance given to the guarantees required to obtain bank financing is essential for companies undergoing rapid and healthy growth.

Clusters	Factors	Fast a	nd Healthy G	Growth	Stabi	t-Test		
		n	M	SD	п	M	SD	
	NBR	539	3.85	0.55	405	3.92	0.58	-2.010 *
	FLCI	539	3.31	0.67	405	3.25	0.64	1.376
Cluster	COP	539	4.05	0.48	405	4.14	0.49	-2.846 **
Financial	GUBF	539	2.52	1.01	405	2.52	0.99	-0.002
Resilient	FRA	539	2.28	0.80	405	2.22	0.84	1.079
	SFBF	539	1.84	0.72	405	1.85	0.71	-0.193
	NBR	238	2.85	1.07	150	2.91	1.05	-0.525
Cluster of	FLCI	238	2.47	0.96	150	2.28	0.92	1.925
Excellence	COP	238	3.87	0.70	150	3.89	0.76	-0.268
Onerational	GUBF	238	1.36	0.47	150	1.32	0.42	0.806
Operational	FRA	238	1.80	0.83	150	1.80	0.74	-0.020
	SFBF	238	1.30	0.52	150	1.26	0.44	0.858
	NBR	363	4.08	0.49	256	4.11	0.53	-0.845
Cluster	FLCI	363	3.53	0.63	256	3.50	0.59	0.522
Tinanaial	COP	363	4.04	0.61	256	4.15	0.59	-2.327 *
Financial	GUBF	363	3.61	0.67	256	3.44	0.63	3.157 **
Strategic	FRA	363	3.11	0.74	256	3.07	0.66	0.656
	SFBF	363	2.82	0.94	256	2.70	0.86	1.638

Table 7. Comparative analysis of the determining factors in corporate indebtedness between the phases of the company's life cycle by cluster.

Legend: NBR—negotiating relations with banks; FLCI—financing, life cycle and indebtedness; COP—Company's operational performance; GUBF—guarantees used to obtain bank financing; FRA—financing risk analysis; SFBF—secondary forms of bank financing. Source: Own preparation. ** p < 0.01; * p < 0.05.

In Table 8, after applying Welch's test as an alternative to ANOVA, the assumption of homogeneity of variances was violated, and it can be seen that, in the secondary forms of bank financing factor, there are statistically significant differences both in the Resilient Financial Cluster (F = 7.357, p < 0.001) and in the Strategic Financial Cluster (F = 3.380, p < 0.05), with companies operating in the national market attaching greater importance to this factor. The Games–Howell multiple comparison test showed that, in the Resilient Financial Cluster, there are significant differences between companies operating in the local market and those operating in the national market (p < 0.001); between those operating in the local market and those that operate in the international market (p < 0.05); and between those that operate in the regional market and those that operate in the national market (p < 0.01). In the Strategic Financial Cluster, there are significant differences only between companies operating in the regional market and those operating in the national market (p < 0.05), with companies operating in the regional market and those that operate in the national market (p < 0.05), with companies operating in the regional market and those operating in the national market (p < 0.05), with companies operating in the regional market and those operating in the national market (p < 0.05), with companies operating in the regional market and those operating in the national market (p < 0.05), with companies operating in the national market and those operating in the national market are significant differences only between companies operating in the regional market and those operating in the national market (p < 0.05), with companies operating in the national market being those that attribute greater importance to the factor secondary forms of bank financing.

In the Resilient Financial Cluster, there are also statistically significant differences in the financing risk analysis factor (F = 5.203, p < 0.01), with companies operating in the local market being those that attribute greater importance to this factor. Using the Games–Howell multiple comparison test, significant differences were verified between companies operating in the local market and those operating in the international market (p < 0.05) and between those operating in the regional market and those operating in the international (p < 0.05). It was therefore observed that companies operating in the national market give greater importance to secondary forms of bank financing, which is in line with what was mentioned by Tavares et al. (2015).

In Table 9, after applying Student's *t*-test to compare two independent samples (private limited company and public limited company), it was observed that there were no statistically significant differences in the determining factors of corporate debt between companies that have limited liability companies and public limited companies in any of the clusters.

		Markets in Which Companies Operate												
0.	F (Lo	ocal Ma	rket	Reg	ional M	larket	National Market			International Market			Welch's
Clusters	Factors	n	Μ	SD	n	Μ	SD	n	Μ	SD	n	Μ	SD	rest
	NBR	158	3.83	0.61	208	3.88	0.57	408	3.89	0.55	172	3.91	0.55	0.636
	FLCI	158	3.24	0.75	208	3.29	0.62	408	3.30	0.65	172	3.27	0.61	0.273
Cluster	COP	158	4.06	0.53	208	4.07	0.45	408	4.12	0.47	172	4.07	0.52	1.078
Financial	GUBF	158	2.37	0.98	208	2.48	1.01	408	2.54	0.97	172	2.66	1.05	2.396
Resilient	FRA	158	2.39	0.89	208	2.37	0.87	408	2.19	0.79	172	2.13	0.71	5.203 **
	SFBF	158	1.67	0.71	208	1.75	0.70	408	1.94	0.70	172	1.88	0.73	7.357 ***
	NBR	82	2.85	1.11	85	2.91	1.08	149	2.88	1.04	72	2.85	1.05	0.063
Cluster of	FLCI	82	2.41	0.99	85	2.28	0.89	149	2.45	0.98	72	2.43	0.90	0.664
Excellence	COP	82	3.98	0.64	85	3.85	0.58	149	3.87	0.75	72	3.82	0.90	0.856
Operational	GUBF	82	1.37	0.42	85	1.28	0.42	149	1.34	0.49	72	1.38	0.45	0.834
Operational	FRA	82	1.77	0.80	85	1.88	0.80	149	1.82	0.83	72	1.69	0.70	0.926
	SFBF	82	1.25	0.43	85	1.34	0.59	149	1.30	0.49	72	1.21	0.41	1.212
	NBR	106	4.11	0.55	108	4.09	0.52	313	4.10	0.49	96	4.05	0.47	0.268
Cluster	FLCI	106	3.43	0.60	108	3.45	0.55	313	3.55	0.63	96	3.56	0.64	1.586
Einangial	COP	106	4.04	0.65	108	4.07	0.59	313	4.10	0.60	96	4.10	0.57	0.310
Stratogic	GUBF	106	3.58	0.66	108	3.46	0.59	313	3.54	0.69	96	3.56	0.61	0.861
Strategic	FRA	106	3.19	0.72	108	3.09	0.73	313	3.10	0.69	96	2.97	0.71	1.527
	SFBF	106	2.67	0.86	108	2.60	0.85	313	2.88	0.94	96	2.72	0.88	3.380 *

Table 8. Comparative analysis of the determining factors in corporate indebtedness between the markets in which companies operate by cluster.

Legend: NBR—negotiating relations with banks; FLCI—financing, life cycle and indebtedness; COP—company's operational performance; GUBF—guarantees used to obtain bank financing; FRA—financing risk analysis; SFBF—secondary forms of bank financing. Note: *** p < 0.001, ** p < 0.01, * p < 0.05. Source: Own preparation.

Table 9. Comparative analysis of the determining factors in corporate indebtedness between types of society by cluster.

Clusters	Factors	Limited	l Liability Co	ompany	And	onymous Soc	iety	t-Test
	-	n	Μ	SD	п	M	SD	
	NBR	763	3.87	0.58	181	3.93	0.49	-1.592
	FLCI	763	3.27	0.67	181	3.35	0.58	-1.594
Cluster	COP	763	4.09	0.49	181	4.11	0.47	-0.579
Financial	GUBF	763	2.51	0.98	181	2.56	1.08	-0.559
Resilient	FRA	763	2.26	0.84	181	2.20	0.69	1.057
	SFBF	763	1.85	0.73	181	1.82	0.66	0.441
	NBR	322	2.84	1.09	64	3.07	0.92	-1.730
Cluster of	FLCI	322	2.38	0.97	64	2.51	0.85	-1.085
Excellence	COP	322	3.85	0.73	64	3.99	0.71	-1.345
Operational	GUBF	322	1.35	0.46	64	1.31	0.40	0.721
Operational	FRA	322	1.82	0.84	64	1.71	0.50	1.483
	SFBF	322	1.30	0.51	64	1.23	0.36	1.329
	NBR	518	4.08	0.51	104	4.15	0.46	-1.214
Cluster	FLCI	518	3.51	0.60	104	3.55	0.64	-0.605
Financial	COP	518	4.09	0.58	104	4.05	0.70	0.687
Stratogic	GUBF	518	3.52	0.64	104	3.61	0.75	-1.168
Strategic	FRA	518	3.11	0.71	104	3.01	0.72	1.360
	SFBF	518	2.76	0.91	104	2.78	0.89	-0.163

Legend: NBR—negotiating relations with banks; FLCI—financing, life cycle and indebtedness; COP—company's operational performance; GUBF—guarantees used to obtain bank financing; FRA—financing risk analysis; SFBF—secondary forms of bank financing. Source: Own preparation.

5. Conclusions

The main objective of this work was to study the determining factors in corporate indebtedness in Portugal. It also had the secondary objectives of creating clusters of companies' credit behaviour and verifying their differences in relation to company characteristics. Thus, in this study, we can see that the determining factors of corporate indebtedness in Portugal are the negotiating relationship between companies and banks; the intended financing, the company's level of indebtedness and the stage of its life cycle; the company's operating performance; the guarantees used to obtain bank financing; the analysis of financing risk; and secondary forms of bank financing. It was also found that the six determining factors in corporate indebtedness lead to the formation of three clusters of companies. One was called the "Resilient Financial Cluster", which has a strong relationship with banks and an excellent operational performance. The cluster called the "Operational Excellence Cluster" reflects the company's excellent operational performance, and the cluster called the "Strategic Financial Cluster" is characterised by high average levels in four determining factors, namely strong relationships with banks; desired financing, level of indebtedness and stage of the company's life cycle; the company's operational performance; and guarantees used to obtain bank financing.

Through comparative cluster analysis, it is concluded that there are statistically significant differences as follows:

- In the "Strategic Financial Cluster", in secondary forms of bank financing, where non-family companies attach greater importance to this factor when compared to non-family companies;
- In the "Operational Excellence Cluster", companies under 25 years old attach greater importance to the financing, cycle and debt factor;
- The "operational development of the company" factor is the only one that presents statistically significant differences in the three clusters, with certified companies attaching greater importance to this factor;
- In the "financial risk analysis" factor, in the "Resilient Financial Cluster" and in the "Strategic Financial Cluster", companies with less than 50 employees attribute greater importance to this factor;
- In the "operational performance" factor, there are differences in the "Resilient Financial Cluster" and the "Strategic Financial Cluster", with companies that are in the stabilised or declining phase of the life cycle being those that attach greater importance to this factor;
- In the "business relationship with banks" factor, in the "Resilient Financial Cluster", companies whose life cycle is in the stabilised or declining phases attribute greater importance to this factor;
- In the factor "secondary forms of bank financing", there are differences in the "Resilient Financial Cluster" and the "Strategic Financial Cluster", with companies operating in the national market being those that attach greater importance to this factor;
- In the "financing risk analysis" factor, the "Resilient Financial Cluster" presents differences, with companies operating in the local market being those that attach greater importance to this factor.

Practical Implications for Companies

A practical implication of this study is that understanding the determinants of corporate indebtedness is crucial for companies to grasp the factors influencing their debt and cost of capital, enabling them to gather the necessary information to gain advantages. This study provides data on this research field for market segmentation and financing strategies, through the formation of three distinct clusters. This study points to the need to segment the market and adapt financing strategies according to the specific characteristics of each cluster. Thus, different business groups should have differentiated approaches. A detailed and personalised analysis is required, depending on the cluster to which companies belong. The operational development factor of a company emerges as a key consideration in all clusters, with certified companies assigning greater importance to this aspect. Additionally, the relationship with banks is a crucial element, especially for companies in specific stages of their life cycle. Therefore, strategies aimed at improving operational performance and strengthening bank relations may be prioritised for financial and operational success. For example, family and non-family businesses may have different perspectives on secondary forms of bank financing, while younger companies may prioritise financing and the company's life cycle. These differences should be taken into account when developing financing and risk management strategies. This work also provides information for policymakers to formulate policies supporting access to credit by Portuguese SMEs. In summary, companies in Portugal should adopt strategic and personalised approaches to financing and risk management, taking into account the determinants of corporate indebtedness and the specific characteristics of each market segment identified by clusters.

It is hoped that this work will contribute to a better understanding of the factors determining bank financing by companies, the banking sector, the community in general and the academic world in particular. It is also expected that it can contribute to the implementation of public policies by the state, so as to better be able to create lines that facilitate access to credit for companies and thus promote economic development.

Author Contributions: Conceptualisation, F.T. and E.S.; methodology, F.T. and E.S.; software, F.T. and E.S.; validation, F.T., E.S., M.F.O. and L.A.; formal analysis, F.T., E.S., M.F.O. and L.A.; investigation, F.T., E.S., M.F.O. and L.A.; resources, F.T., E.S., M.F.O. and L.A.; data curation, F.T.; writing—original draft preparation, F.T. and E.S.; writing—review and editing, F.T., E.S., M.F.O. and L.A.; visualisation, F.T., E.S., M.F.O. and L.A.; supervision, F.T. and E.S. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Data Availability Statement: The data presented in this study are available on request from the corresponding author. The data are not publicly available due to what they achieved through surveys of companies.

Conflicts of Interest: The authors declare no conflicts of interest.

References

- Aksoy, Mine, and Mustafa Kemal Yilmaz. 2023. Does board diversity affect the cost of debt financing? Empirical evidence from Turkey. *Gender in Management* 38: 504–24. [CrossRef]
- Almeida, Luís. 2023. Risk and Bankruptcy Research: Mapping the State of the Art. *Journal of Risk and Financial Management* 16: 361. [CrossRef]
- Amarna, Khayria, Raquel Garde Sánchez, Maria Victoria López-Pérez, and Mahmoud Marzouk. 2024. The effect of environmental, social, and governance disclosure and real earning management on the cost of financing. Corporate Social Responsibility and Environmental Management, 1–13. [CrossRef]
- Aziz, Sohail, and Ulfat Abbas. 2019. Effect of debt financing on firm performance: A study on non-financial sector of Pakistan. *Open Journal of Economics and Commerce* 2: 8–15. [CrossRef]
- Belkhaoui, Samir, Naif Alsagr, and Stefan F. van Hemmen. 2020. Financing modes, risk, efficiency and profitability in Islamic banks: Modeling for the GCC countries. *Cogent Economics & Finance* 8: 1750258. [CrossRef]
- Bhatia, Prince, and Rahul Kumar. 2023. Do debt, and operating efficiency contributes to corporate performance? *International Journal of System Assurance Engineering and Management* 15: 1203–9. [CrossRef]
- Boshnak, Helmi. 2023. The impact of capital structure on firm performance: Evidence from Saudi-listed firms. *International Journal of Disclosure and Governance* 20: 15–26. [CrossRef]
- Calcagnini, Giorgio, Fabio Farabullini, and Germana Giombini. 2014. The impact of guarantees on bank loan interest rates. *Applied Financial Economics* 24: 397–412. [CrossRef]
- Chai, Nana, Bi Wu, Weiwei Yang, and Baofeng Shi. 2019. A multicriteria approach for modeling small enterprise credit rating: Evidence from China. *Emerging Markets Finance and Trade* 55: 2523–43. [CrossRef]
- Cicogna, Maria Paula Vieira, Rudinei Toneto Junior, and Maurício Ribeiro do Valle. 2007. O impacto da adesão a padrões mais elevados de governança sobre o financiamento empresarial. *Revista de Administração* 42: 52–63.
- Forti, Cristiano Augusto Borges, Fernanda Maciel Peixoto, and Kellen Silva Freitas. 2011. Retenção de caixa, desempenho operacional e valor: Um estudo no mercado de capitais brasileiro. *Revista de Contabilidade e Organizações* 5: 20–33. [CrossRef]

Ghouma, Hatem, Hamdi Ben-Nasr, and Ruiqian Yan. 2018. Corporate governance and cost of debt financing: Empirical evidence from Canada. *The Quarterly Review of Economics and Finance* 67: 138–48. [CrossRef]

Gonçalves, Tadeu Junior, Anderson Roberto Pires, and Pedro Borges Junior. 2018. A forma de financiamento das companhias abertas sob os pressupostos da pecking order theory e trade off theory. *Revista Evidenciação Contábil & Finanças* 6: 101–18. [CrossRef]

Hair, Joseph F., William C. Black, Barry J. Babin, and Rolph E. Anderson. 2014. *Multivariate Data Analysis*, 7th ed. Edinburgh: Pearson. Jucá, Michele Nascimento, Anderson Luís Saber Campos, Douglas Dias Bastos, and Eduardo Alexandre Mendes. 2016. Endividamento e estrutura de ativos: Evidências no Brasil. *Revista de Financas Aplicadas* 7: 1–19.

Kline, Rex B. 2016. Principles and Practice of Structural Equation Modeling. New York: Guilford Press.

Lemma, Tesfaye T. 2015. Corruption, debt financing and corporate ownership. Journal of Economic Studies 42: 433-61. [CrossRef]

Li, Luya, and Hongxun Li. 2021. Analysis of financing risk and innovation motivation mechanism of financial service industry based on Internet of things. *Complexity* 2021: 5523290. [CrossRef]

Magnanelli, Barbara Sveva, and Maria Federica Izzo. 2017. Corporate social performance and cost of debt: The relationship. *Social Responsibility Journal* 13: 250–65. [CrossRef]

Manuel, Benedito, Fernando Oliveira Tavares, and Luís Pacheco. 2018. Decisões de financiamento bancário por parte da banca angolana: O caso do Banco de Poupança e Crédito. e3-Revista de Economia, Empresas e Empreendedores na CPLP 4: 55–80. [CrossRef]
 Marôco, João. 2018. Análise Estatística com o SPSS Statistics 25, 7th ed. Lisboa: ReportNumber.

Mayordomo, Sergio, Antonio Moreno, Steven Ongena, and Maria Rodriguez-Moreno. 2021. Bank capital requirements, loan guarantees and firm performance. *Journal of Financial Intermediation* 45: 100825. [CrossRef]

Mehtap, Ozge, and Ozgur Kokalan. 2013. The relationship between corporate reputation and organizational citizenship behavior: A comparative study on TV companies and banks. *Quality & Quantity* 47: 3609–19. [CrossRef]

Mura, Ladislav, and Ján Buleca. 2012. Evaluation of financing possibilities of small and medium industrial enterprises. *Procedia Economics and Finance* 3: 217–22.

Mutamimah, Mutamimah, Mochammad Tholib, and Robiyanto Robiyanto. 2021. Corporate governance, credit risk, and financial literacy for small medium enterprise in Indonesia. *Business: Theory and Practice* 22: 406–13. [CrossRef]

Myers, Stewart C., and Nicholas S. Majluf. 1984. Corporate financing and investment decisions when firms have information that investors do not have. *Journal of Financial Economics* 13: 187–221. [CrossRef]

Ogunmokun, Olapeju Comfort, Oluwasoye Mafimisebi, and Demola Obembe. 2024. Bank lending behaviour and small enterprise debt financing. *Journal of Entrepreneurship in Emerging Economies* 16: 675–97. [CrossRef]

Ono, Arito, and Iichiro Uesugi. 2009. Role of collateral and personal guarantees in relationship lending: Evidence from Japan's SME loan market. *Journal of Money, Credit and Banking* 41: 935–60. [CrossRef]

O'Sullivan, Jennifer, Abdullah Mamun, and M. Kabir Hassan. 2016. The relationship between board characteristics and performance of bank holding companies: Before and during the financial crisis. *Journal of Economics and Finance* 40: 438–71. [CrossRef]

Pacheco, Luís Miguei, Raquel Rosa, and Fernando Oliveira Tavares. 2019. Risco de falência de PME: Evidência no setor da construção em Portugal. *Innovar* 29: 143–57. [CrossRef]

Rodrigues, Santiago Valcacer, Elton Eustáquio Casagrande, and David Ferreira Lopes Santos. 2018. A interdependência das decisões empresariais de investimento e endividamento na América Latina. *Enfoque: Reflexão Contábil* 37: 1–20. [CrossRef]

Rosa, Fabio, Giovanni Liberatore, Francesco Mazzi, and Simone Terzani. 2018. The impact of corporate social performance on the cost of debt and access to debt financing for listed European non-financial firms. *European Management Journal* 36: 519–29. [CrossRef]

Santos, Eulalia, Vasco Tavares, Fernando Oliveira Tavares, and Vanessa Ratten. 2022. How is risk different in family and non-family businesses? A comparative statistical analysis during the COVID-19 pandemic. *Journal of Family Business Management* 12: 1113–30. [CrossRef]

Soares, Gustavo de Oliveira Godoy, Eduardo Senra Coutinho, and Marcos Antônio de Camargos. 2012. Determinantes do rating de crédito de companhias brasileiras. *Contabilidade Vista & Revista* 23: 109–43.

Tavares, Fernando Oliveira, Eulália Santos, and Vasco Capela Tavares. 2021. Risk categorization in Portuguese organizations in times of the COVID-19 pandemic—An exploratory statistical analysis. *Journal of Entrepreneurship and Public Policy* 10: 306–22. [CrossRef]

Tavares, Fernando Oliveira, Eulália Santos, Vasco Capela Tavares, and Vanessa Ratten. 2023. Risk Planning and Management in Portuguese Companies—A Statistical Approach. *Journal of Risk and Financial Management* 16: 314. [CrossRef]

Tavares, Fernando Oliveira, Luís Pacheco, and Emanuel Ferreira Almeida. 2015. Financiamento das pequenas e médias empresas: Análise das empresas do distrito do Porto em Portugal. *Revista de Administração* 50: 254–67. [CrossRef]

Ugwueze, Catherine Amaka, C. E. Nwoha, and Uche Lucy Onyekwelu. 2019. Effect of financing decisions on operational performance of pharmaceutical firms in Nigeria. *Academic Journal of Current Research* 6: 3244–5621.

Uyar, Ali, Nouha Ben Arfa, Cemil Kuzey, and Abdullah S. Karaman. 2024. Do creditors appreciate CSR transparency and credibility in emerging markets? *Journal of Accounting Literature ahead-of-print*. [CrossRef]

Van Osnabrugge, Mark. 2000. A comparison of business angel and venture capitalist investment procedures: An agency theory-based analysis. *Venture Capital: An International Journal of Entrepreneurial Finance* 2: 91–109. [CrossRef]

Yazdanfar, Darush, and Peter Öhman. 2015. Debt financing and firm performance: An empirical study based on Swedish data. *Journal* of Risk Finance 16: 102–18. [CrossRef]

Ye, Zhiwei, Yasir Shahab, Yasir Riaz, and Collins G. Ntim. 2023. Strategic deviation and the cost of debt financing. *Economic Modelling* 125: 106371. [CrossRef]

Zhang, Wen, Shaoshan Yan, Jian Li, Xin Tian, and Taketoshi Yoshida. 2022. Credit risk prediction of SMEs in supply chain finance by fusing demographic and behavioral data. *Transportation Research Part E: Logistics and Transportation Review* 158: 102611. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.



Article



Board Gender Diversity and Risk Management in Corporate Financing: A Study on Debt Structure and Financial Decision-Making

Davood Askarany ¹, Soleil Jafari ^{2,*}, Azam Pouryousof ³, Sona Habibi ² and Hassan Yazdifar ^{4,5}

- ¹ Department of Accounting and Finance, Business School, University of Auckland, Auckland 1010, New Zealand; d.askarany@auckland.ac.nz
- ² Department of Accounting, Gonbad-e-Kavoos Branch, Islamic Azad University, Gonbad-e-Kavoos 4971799151, Iran; habibi@pnu.ac.ir
- ³ Faculty of Management, Economics & Accounting, Payame Noor University, Tehran 19395-4697, Iran; pouryousof@pnu.ac.ir
- ⁴ College of Business, Law and Social Sciences, University of Derby,
- Derby DE22 1GB, UK; h.yazdifar@derby.ac.uk
- ⁵ Gulf Finance Centre, Gulf University for Science and Technology, Hawally 32093, Kuwait
- * Correspondence: soleil@iau.ac.ir

Abstract: Purpose: This study examines the role of board gender diversity in shaping corporate financial decisions, particularly in terms of debt structure and risk management. Focusing on the Tehran Stock Exchange, it explores how female representation on boards influences long-term and short-term leverage decisions, focusing on the moderating effect of board compensation. Design/Methodology: Utilising a quantitative ex post facto design, the study analyses data from 114 companies listed on the Tehran Stock Exchange between 2017 and 2021. Multivariate regression techniques, including year- and industryfixed effects, are employed to investigate the relationship between board gender diversity, debt structure, and risk-taking behaviour. Findings: The results reveal a significant negative relationship between female board representation and long-term debt, suggesting that companies with more female directors tend to adopt more conservative debt structures, thereby reducing risk. Additionally, the findings demonstrate that board compensation moderates this relationship by curbing managerial risk-taking, further improving financial decision-making. Originality/Value: This research provides novel insights into the intersection of board gender diversity and risk management in financial decision-making, particularly in the context of a developing economy like Iran. It also offers practical implications for firms seeking to optimise their debt structures while maintaining sound risk management practices.

Keywords: board gender diversity; debt structure; risk management; corporate financing; Tehran Stock Exchange

1. Introduction

The role of gender diversity on corporate boards has become a pivotal area of interest as firms strive for balanced financial decision-making and effective risk management strategies. This study builds on this discourse by examining how female representation on corporate boards influences debt structure and risk-taking in financing decisions, particularly in the context of emerging markets such as Iran. While traditional finance theory emphasises the importance of strategic debt management to maintain liquidity and mitigate



Academic Editors: Eulália Mota Santos, Margarida Freitas Oliveira and Dayong Huang

Received: 25 September 2024 Revised: 20 November 2024 Accepted: 11 December 2024 Published: 13 January 2025

Citation: Askarany, Davood, Soleil Jafari, Azam Pouryousof, Sona Habibi, and Hassan Yazdifar. 2025. Board Gender Diversity and Risk Management in Corporate Financing: A Study on Debt Structure and Financial Decision-Making. *Risks* 13: 11. https://doi.org/10.3390/ risks13010011

Copyright: © 2025 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https://creativecommons.org/ licenses/by/4.0/).
agency conflicts, this research investigates whether the presumed risk aversion attributed to female directors contributes to more conservative debt structuring. Through exploring board diversity within Iran's distinct socio-cultural and regulatory environment, this study highlights the potential of gender-diverse boards to foster financial resilience and ethical governance practices in response to market uncertainties.

The impact of gender diversity on corporate decision-making, particularly concerning financial decisions and risk management, has received growing attention in recent years. In today's dynamic business environment, strategically managing debt and risk is crucial to ensuring corporate financial sustainability. This study builds on this growing body of research by examining the influence of board gender diversity on financial risk management, particularly in the context of corporate debt structure in Iran.

Rooted in traditional finance theories, firms must carefully manage various risk factors, such as liquidity and refinancing risks, to achieve optimal financial outcomes. Debt maturity decisions are crucial in managing financial risks by balancing short-term obligations with long-term financial stability. This study investigates whether female directors, perceived as more risk-averse, prefer debt structures that reduce financial risk and mitigate potential agency conflicts between shareholders and managers.

By exploring the moderating role of board compensation, this research aims to provide further insights into how gender diversity and financial incentives influence corporate risk management. Iran's unique socio-cultural and regulatory context presents an opportunity to contribute to the broader literature on gender diversity and risk management in emerging markets. The findings of this study are expected to offer practical implications for corporate governance and risk management frameworks, aligning with the themes of the special issue on financial analysis and risk management.

2. Literature Review

The impact of female directors on corporate decision-making has garnered extensive scholarly attention, particularly in promoting diversity and enhancing governance practices. This study builds upon this foundation by investigating the influence of female board representation on financing decisions and debt structure choices within the unique context of Iran. Unlike Western countries, where corporate governance structures and cultural norms differ significantly, Iran presents a distinctive setting to explore these dynamics. The gender ethics framework suggests that women exhibiting higher moral standards may prefer shorter-maturity debt to align with shareholder interests, thus mitigating debt-related agency problems. Additionally, behavioural perspectives propose that female directors, presumed to exhibit greater risk aversion, may opt for longer-maturity debt to effectively manage liquidity and refinancing risks.

Rooted in traditional finance theories, corporations navigate various factors to determine the optimal debt maturity structure, including liquidity/refinancing risk, mispricing risk, and under/overinvestment problems. This choice is crucial for financial policy and serves as a regulatory mechanism to address manager–shareholder agency conflicts. Long-term debt financing allows managers to shield themselves from constant monitoring, potentially deviating from the optimal debt maturity choice and imposing costs on shareholders.

In recent years, scholars have extensively probed the impact of female directors on corporate decision-making across various domains, including dividend policy, executive compensation, and mergers and acquisitions. Studies have shown that female directors tend to emphasise monitoring and accountability, which can influence decisions related to debt maturity structure. The gender ethics framework posits that women, exhibiting higher moral standards, are more attuned to ethical concerns than men. Applying this framework to debt maturity structure choices suggests that female directors may display less self-interest, potentially preferring longer-maturity debt to align with shareholder interests and mitigate debt-related agency problems.

Additionally, considering managerial risk aversion, it has been proposed that female directors, presumed to exhibit greater risk aversion, may make safer financial decisions than their male counterparts. Short-term debt is considered costlier for risk-averse executives due to higher liquidity/refinancing risk costs and increased external monitoring through refinancing.

This research seeks to unravel the impact of board gender diversity on companies' financing decisions and debt maturity structure choices, explicitly focusing on Iran's unique socio-cultural and regulatory environment. It questions whether female representation on the board decreases long-term leverage, aligning with the gender ethics framework, or increases it, aligning with the behavioural perspective. Additionally, the study explores the moderating role of board compensation in this relationship.

In contrast to prior studies like Abobakr and Elgiziry (2016) and Harris (2014), which focus on how board characteristics influence corporate financial leverage, our study brings a unique perspective by examining not only the direct effect of board gender diversity on debt structure but also how this relationship is moderated by board compensation. This additional layer of analysis provides novel insights into how compensation structures shape the impact of gender diversity on financial decision-making. This topic has not been extensively covered in prior research. Furthermore, our focus on the Tehran Stock Exchange (TSE) offers valuable context from a developing economy, contributing to the broader literature on how cultural and regulatory environments influence the dynamics of corporate governance, particularly in non-Western contexts. This focus on the Iranian market adds depth to the discussion by examining how women in leadership positions function within a patriarchal society, providing insights distinct from those of the studies mentioned.

Empirical testing was conducted on data from 114 companies listed on the Tehran Stock Exchange (TSE) from 2017 to 2021, utilising multivariate regression with year- and industry-fixed effects. The results align with the gender ethics framework, revealing that female representation on the board reduces long-term leverage. Furthermore, the findings indicate that board compensation strengthens this relationship, underscoring the positive impact of higher board compensation on aligning management and shareholder interests and reducing managerial risk-taking.

This research contributes to the literature by examining the impact of board gender diversity on financing decisions and debt maturity structure choices in Iran, a developing country with distinct cultural and governance practices. Additionally, it pioneers in exploring the moderating role of board compensation, providing valuable insights into the complex dynamics of corporate governance and financial decision-making in a non-Western context.

Gender diversity has received substantial attention in the business literature (Labelle et al. 2015; Richards 2003; Rosser et al. 2024). The literature on board gender diversity and its impact on corporate decision-making has burgeoned in recent years, reflecting a growing recognition of the importance of diversity in governance structures (Duppati et al. 2020; Peng et al. 2023; Said et al. 2022; Ullah et al. 2020). Studies have examined various aspects of corporate behaviour influenced by female board representation, including dividend policy, executive compensation, and mergers and acquisitions.

One area of particular interest is the relationship between board gender diversity and debt maturity structure choices. Li and Zeng (2019) found systematic differences in debt maturity structure choices with the presence of female directors, suggesting that female directors may prefer short-term debt to align managerial and shareholder interests. This finding aligns with the gender ethics framework, which posits that women exhibiting higher

moral standards may be more attuned to ethical concerns and less self-interested in their decision-making than men.

Adams and Ferreira (2009) also highlight the positive role of female directors in monitoring and holding managers accountable, which can influence decisions related to debt financing. The emphasis on monitoring and accountability may lead female directors to prefer shorter-maturity debt, which subjects managers to more frequent external monitoring through refinancing.

From a behavioural perspective, Brockman et al. (2010) suggest that short-term debt may be costlier for risk-averse executives due to higher liquidity/refinancing risk costs. Female directors, presumed to exhibit greater risk aversion, may opt for longer-maturity debt to mitigate these costs and make safer financial decisions.

The literature suggests that female board representation can influence debt maturity structure choices through various mechanisms, including monitoring, accountability, and risk aversion. The empirical evidence supports the notion that board gender diversity is associated with differences in corporate financial decision-making, highlighting the importance of diversity in governance structures for effective risk management and shareholder value creation.

In the study by Datta et al. (2021), the relationship between board gender diversity and financing decisions is also examined. However, the leverage decision in our research is different. In Datta et al. (2021), decisions related to debt maturity were considered. The data were used as follows. Compustat item codes are provided in parentheses: ST3: Percentage of debt maturing in three years or less divided by total debt. ST5: Percentage of debt maturing in five years or less divided by total debt. However, in our research, the basis is on long-term and short-term debt according to the balance sheet. Although Iran has banned interest-based activities according to Islamic laws, the bank interest rates (and market rates) differ for short-term and long-term financing, with higher bank interest rates for short-term facilities. In Datta et al. (2021), the impact of compensation on the relationship between board gender diversity and financing decisions was also addressed. Still, they defined compensation based on Compustat as executive equity-based compensation. The sum of the value of new stock options (using the modified Black–Scholes method) is granted to the firm's CEO and CFO as a percentage of the total compensation paid to them. However, in Iran, it is not possible to use data related to stock-based compensation because they include only a few companies, reducing the sample size, and it is not usually disclosed. We used board compensation disclosed in general financial reports. Therefore, the differences presented in the reporting structure in Iran may affect research findings. Given Iran's economic and political environment, such as the lack of effective governance mechanisms, weak risk management practices, and a weak culture of responsibility, it seems that examining the impact of board gender diversity on long-term debt (specifically in the present study) and the moderating effect of board compensation on this relationship is valuable.

3. Theoretical Framework and Hypotheses

3.1. Debt Structure

This study is situated within the Agency Theory paradigm, a foundational corporate finance framework that examines the relationships and conflicts of interest between principals (shareholders) and agents/management (Daryaei et al. 2022; Zadeh et al. 2022; Pouryousof et al. 2022; Shandiz et al. 2022; Zadeh et al. 2023). Within this context, exploring debt structure choices, financial decisions, and corporate governance, particularly regarding board gender diversity, is inherently linked to addressing agency conflicts. By investigating how the composition of the board, specifically the inclusion of women, influences decision-making processes, managerial risk-taking behaviour, and the alignment of inter-

ests between directors and shareholders, this study seeks to contribute to understanding agency dynamics in corporate settings.

According to traditional finance theory, companies must carefully weigh various factors, such as liquidity/refinancing risk, mispricing risk, and underinvestment/overinvestment problems, to achieve an optimal debt maturity structure. The choice between longand short-term debt can serve as a robust monitoring mechanism for reducing agency conflicts within organisations. Short-term debt, for example, can effectively lower agency costs by subjecting managers to more frequent external monitoring due to the necessity of refinancing (Rajan and Winton 1995; Stulz 2001). However, separating ownership and control may incentivise managers to deviate from the optimal debt maturity structure to avoid monitoring, often at the expense of the shareholders.

Debt structure, a critical determinant of corporate success and growth, is pivotal in a company's ability to thrive and survive in competitive markets (Mohammadi and Zanjirdar 2020). Abor (2008) underscores the heightened risk of bankruptcy associated with elevated debt levels, highlighting the delicate balance required in managing debt obligations. In addition to shareholder–manager conflicts, a fundamental conflict of interest exists between shareholders and creditors, further complicating the decision-making process surrounding debt maturity structure. This complexity is compounded by agency problems, information asymmetry, firm size, and financial distress, significantly influencing companies' optimal debt structure choices (Huang et al. 2016).

Research findings on the relationship between leverage and debt maturity (leverage structure decisions such as short-term or long-term debt) present diverse perspectives and conclusions. Diamond (1991), for instance, argues that companies with high leverage may prefer long-term debt to mitigate the risk of inefficient liquidation, thereby providing additional time for debt repayment. Conversely, Flannery (1986) posits that higher debt levels enable companies to manage refinance risk effectively by issuing long-term debt securities. However, Shan (2019) presents a contrasting viewpoint, suggesting a negative impact of leverage on debt maturity and proposing that agency costs may be alleviated by reducing leverage or shortening debt maturity periods (Hosni Al-Ghar and Rahimian 2018). The interplay between leverage and debt maturity, influenced by firm value, market conditions, and regulatory environments, remains a nuanced and evolving study area within corporate finance.

The focus of the paper is on debt structure. Our research refers to debt structure as using short-term versus long-term debt. We have used the structure in the balance sheet for long-term debt and the long-term debt ratio (long-term debt to assets) to measure leverage structure decisions. Thus, we do not focus on debt maturity. However, some arguments regarding debt maturity are cited. According to the theoretical and empirical foundations presented in the article, three different arguments about the impact of women on the board of directors on debt structure are provided, including risk aversion view, ethical view, and the view that gender diversity of the board has no relationship with debt structure decisions. If the impact of board gender diversity on long-term leverage is positive, the risk aversion view of women is confirmed. If negative, the ethical view of women is confirmed. The third view (no relationship) is confirmed if there is no significant relationship. The article text was reviewed for clarification, and additional explanations were provided.

This study's theoretical framework draws upon the rich literature within Agency Theory and traditional finance theory to elucidate the complex dynamics of debt structure choices, financial decision-making processes, and corporate governance practices. By examining the impact of board gender diversity on these phenomena, this research aims to contribute valuable insights to academic scholarship and practical policymaking in corporate finance and governance.

3.2. Board Gender Diversity

Gender theory posits that women contribute uniquely to economic success, stemming from their distinct moral development and values. Empirical evidence consistently supports that women often exhibit higher ethical standards, demonstrating greater sensitivity to social relationships and more vital adherence to rules and regulations. Within corporate governance contexts, gender diversity on boards has been recognised as a significant determinant of decision-making processes, particularly in corporate risk-taking behaviours (Barnett et al. 1994; Geiger and O'Connell 1998; Chun 2005; Datta et al. 2021; Levi et al. 2014; Linley et al. 2007; Peterson and Seligman 2003).

Female representation on corporate boards has emerged as a critical factor influencing various dimensions of corporate governance, financial performance, and risk management. Research indicates that the presence of female directors contributes to enhanced corporate governance practices, reduced risk-taking, and improved financial performance outcomes (Catalyst 2017; Chen et al. 2017; Sila et al. 2016). From an Agency Theory perspective, female directors are perceived as effective monitors of managerial behaviour, often demanding greater accountability and exhibiting lower tolerance for opportunistic behaviours (Adams and Ferreira 2009; Sepasi and Abdoli 2015).

The value of gender diversity within corporate boards extends beyond mere representation to encompass a broader range of competencies, perspectives, and decision-making styles. Studies have shown that diverse boards, including those with a significant proportion of female members, are likelier to engage in thorough deliberation, consider a more comprehensive array of viewpoints, and make more informed decisions. This diversity of perspectives is precious in complex and dynamic business environments, where innovative solutions and adaptive strategies are essential for organisational success.

Furthermore, research suggests that gender-diverse boards are better equipped to navigate and mitigate various forms of corporate risk. Women, on average, exhibit greater risk aversion than men, leading to more cautious decision-making processes and a reduced propensity for engaging in high-risk activities (Sun and Zhang 2021). This risk-averse behaviour can serve as a valuable counterbalance to the potentially overconfident and risk-seeking tendencies often associated with male-dominated leadership structures (Adams and Ferreira 2009). As such, the presence of female directors can contribute to a more balanced and prudent approach to risk management within organisations.

Moreover, gender diversity on corporate boards has been linked to improved stakeholder relations and enhanced corporate reputation. Companies with diverse boards are perceived as more inclusive, socially responsible, and attuned to the needs and preferences of various stakeholder groups, including employees, customers, and investors (Terjesen et al. 2016). This positive perception can translate into tangible organisational benefits, including increased brand loyalty, employee morale, and investor confidence.

In addition to its effects on corporate governance and risk management, gender diversity on boards can also have significant implications for organisational innovation and performance. Research suggests that diverse teams, including individuals with diverse gender identities, are more innovative and better able to adapt to changing market conditions (Carter et al. 2017). By bringing together individuals with different backgrounds, experiences, and perspectives, gender-diverse boards can foster a culture of creativity, collaboration, and open-mindedness, thereby driving organisational innovation and competitive advantage.

However, despite the growing recognition of the benefits of gender diversity on corporate boards, women continue to be underrepresented in leadership positions within many organisations. The "glass ceiling" phenomenon, systemic barriers, and unconscious biases often impede women's advancement to senior leadership roles (Hoyt and Murphy 2016). Addressing these barriers and promoting greater gender diversity within corporate leadership structures requires concerted efforts from organisations, policymakers, and society.

In conclusion, gender diversity on corporate boards plays a crucial role in shaping organisational outcomes, including corporate governance practices, risk management strategies, stakeholder relations, innovation, and performance. By harnessing female directors' unique perspectives, experiences, and decision-making styles, organisations can enhance their effectiveness, resilience, and competitiveness in today's dynamic business environment. Efforts to promote greater gender diversity within corporate leadership structures are a matter of social justice and equality and a strategic imperative for sustainable business success.

3.3. Board Gender Diversity and Debt Structure

Research exploring the intersection of board gender diversity and debt structure offers valuable insights into the complex dynamics of corporate decision-making processes. Evidence suggests that women, on average, exhibit greater moral and ethical sensitivities than their male counterparts. Surveys and studies across various professions, including accounting, auditing, sales, and among students, consistently indicate higher levels of moral development, conservative ethical viewpoints, and greater adherence to ethical standards among women (Bernardi and Arnold 1997; Larkin 2000; O'Donnell and Johnson 2001; Dawson 1997; Pierce and Sweeney 2010; Albaum and Peterson 2006).

The choice of a company's leverage and debt maturity structure is a critical decision that can significantly impact its financial health, performance, and risk profile. According to Agency Theory, managers may be motivated by self-interest and select a debt maturity structure that is not aligned with the best interests of shareholders. Specifically, managers may opt for long-term debt to avoid the frequent external monitoring associated with refinancing short-term debt (Brockman et al. 2010; Datta et al. 2005).

In the context of board gender diversity, the presence of women in executive and boardroom positions introduces a potentially moderating influence on debt structure decisions. Assuming that female executives possess higher moral and ethical sensitivities, they may be more inclined to make decisions that align with shareholders' interests and corporate governance principles. As such, female executives may prefer short-term debt, which offers enhanced transparency, accountability, and alignment with shareholder interests, without fearing market scrutiny or external monitoring.

However, the relationship between board gender diversity and debt structure is multifaceted and influenced by various factors, including organisational culture, industry dynamics, and regulatory frameworks. While the presence of women in leadership roles may contribute to more ethical and transparent decision-making processes, it is essential to recognise that gender diversity alone may not guarantee optimal outcomes. Effective governance mechanisms, robust risk management practices, and a culture of accountability are equally crucial for ensuring sound debt structure decisions and safeguarding shareholder interests.

Moreover, the impact of board gender diversity on debt structure may vary across different contexts and organisational settings. Cultural norms, societal expectations, and institutional factors can shape how gender diversity influences company decision-making processes. For example, in patriarchal societies where traditional gender roles are deeply entrenched, the presence of women in leadership positions may face more severe resistance (resistance of other leadership officials of the organisation) or encounter implicit biases that hinder their influence on strategic decisions, including debt structure choices (Datta et al. 2021).

However, it is essential to note that although the presence of women in strategic roles may improve ethical and transparent decision-making processes, this impact requires the presence of effective governance mechanisms, robust risk management practices, and a culture of responsibility. In other words, the effect of board gender diversity on leverage decisions may vary across different contexts and organisational settings, including cultural norms, social expectations, and institutional factors. In Iran, we witness weak corporate governance, risk management, and a culture of responsibility. Additionally, cultural norms, such as the patriarchal society of Iran, may prevent women from significantly influencing financial decisions, like leverage decisions, despite their physical presence in strategic decision-making processes. Therefore, this raises the question of whether the presence of women on the board of directors in Iran, based on the existing literature and empirical evidence, can lead to more ethical and transparent financing decisions or whether organisational settings and cultural norms prevent this impact.

Empirical research on the relationship between board gender diversity and debt structure has yielded mixed findings, highlighting the need for further investigation and nuanced analysis. While some studies suggest a positive association between female representation on boards and adopting more conservative debt policies (Datta et al. 2021; Choi et al. 2017; Ho et al. 2015; Larkin 2000), others find no significant relationship or even observe contradictory results (Bajtelsmit and VanDerhei 1997; Hinz et al. 1997; Jianakoplos and Bernasek 1998; Levin et al. 1988; Sundén and Surette 1998). These inconsistencies underscore the complexity of the underlying mechanisms and the importance of considering contextual factors in understanding the impact of gender diversity on corporate decision-making.

Future research could explore the mediating and moderating factors influencing the relationship between board gender diversity and debt structure. Longitudinal studies tracking changes in board composition and debt policies over time could provide valuable insights into the causal mechanisms underlying observed associations. Qualitative research methods, such as interviews and case studies, could offer deeper insights into the decision-making processes and dynamics within boardrooms and executive suites.

In conclusion, the relationship between board gender diversity and debt structure is a complex and multifaceted phenomenon that warrants careful examination and analysis. While the presence of women in leadership positions may contribute to more ethical, transparent, and shareholder-aligned decision-making processes, the impact of gender diversity on debt policies is contingent upon various contextual factors and organisational dynamics. By advancing our understanding of these dynamics, researchers can contribute to developing more effective governance practices and strategies for promoting diversity, equity, and inclusion within corporate boardrooms.

3.4. Hypotheses Development

Exploring the intricate relationship between board gender diversity and debt structure involves synthesising various theoretical perspectives and empirical evidence to formulate hypotheses that capture the nuanced dynamics. This section delineates three main arguments guiding the investigation into the impact of board gender diversity on debt structure choices. Subsequently, it proposes hypotheses grounded in theoretical frameworks and empirical observations.

First Argument: Ethical Inclinations and Debt Structure

The first argument posits that female executives with higher ethical inclinations and moral standards tend to make decisions that align more closely with shareholder interests, particularly in debt structure choices. Drawing from the gender ethics framework, which suggests that women are more attuned to ethical considerations such as disclosure, integrity, and conflicts of interest (Ho et al. 2015; Larkin 2000), it is hypothesised that firms with female directors will exhibit a preference for shorter-term debt. Female executives, presumed to be less self-interested than their male counterparts, are expected to opt for shorter-term debt to avoid the scrutiny associated with long-term financing arrangements. This expectation stems from the notion that shorter debt maturity structures impose more immediate accountability and transparency, mitigating potential agency conflicts between management and shareholders (Choi et al. 2017). Consequently, female executives are anticipated to encounter fewer debt-related agency problems, leading to a more optimal alignment of debt maturity structures with shareholder interests.

Second Argument: Risk Aversion and Debt Maturity Structure

The second argument centres on the risk tolerance of directors, mainly focusing on the risk-averse nature often attributed to female executives. Building upon the premise that risk-averse individuals exhibit a preference for safer financial decisions and tend to avoid exposure to uncertainty (Bajtelsmit and VanDerhei 1997; Hinz et al. 1997; Jianakoplos and Bernasek 1998; Levin et al. 1988; Sundén and Surette 1998), it is hypothesised that female directors will opt for a more conservative approach to debt maturity structure.

Under this perspective, female executives are expected to favour longer-term debt arrangements to mitigate liquidity and refinancing risks associated with short-term financing. Short-term debt instruments, characterised by higher liquidity/refinancing risk costs and increased external monitoring through refinancing, may be perceived as less favourable by risk-averse female directors (Brockman et al. 2010). Therefore, it is posited that female directors will exhibit a propensity towards more extended debt maturity structures, reflecting their risk-averse nature and preference for safer financial strategies.

Third Argument: Competing Perspectives and Firm Performance

The third argument acknowledges the existence of mixed evidence regarding the impact of board gender diversity on firm performance, introducing competing perspectives that warrant further exploration. While some studies have suggested a positive relationship between board gender diversity and various performance metrics (e.g., Catalyst 2017; Chen et al. 2017; Sila et al. 2016), others have found inconclusive or even negative associations (e.g., Adams and Ferreira 2009).

In light of these conflicting findings, it is imperative to consider the potential moderating effects of factors such as directors' compensation on the relationship between board gender diversity and firm performance (Jane Lenard et al. 2014; Datta et al. 2021; Cumming et al. 2024; Gormley et al. 2023). Directors' compensation is a way to match the interests of management and shareholders and influences managerial risk-taking behaviours (Brockman et al. 2010; Datta et al. 2021). Therefore, it is hypothesised that the relationship between board gender diversity and financial leverage is contingent upon the level of board compensation, with higher compensation levels strengthening the association between gender diversity and leverage decisions.

3.5. Formulated Hypotheses

H1. Board gender diversity has a significant effect on the long-term leverage of companies.

The literature suggests that female executives with higher ethical inclinations and lower self-interest are more likely to favour shorter debt maturity structures, aligning better with shareholder interests. Consequently, it is hypothesised that firms with greater board gender diversity will prefer shorter-term debt arrangements, leading to lower longterm leverage ratios compared to those with less diverse boards.

H2. Board compensation strengthens the relationship between gender diversity and financial leverage.

Directors' compensation is a mechanism to align management and shareholders' interests and influences managerial risk-taking behaviours. Higher levels of board compensation are expected to reinforce the impact of board gender diversity on financial leverage decisions, with firms offering higher compensation packages exhibiting a stronger association between gender diversity and leverage choices.

The hypotheses in this study are developed to explore the nuanced impacts of board gender diversity on debt structure choices, informed by ethical and risk aversion theories. Female directors are often associated with heightened ethical standards and a preference for conservative financial strategies. This suggests that board gender diversity may reduce reliance on long-term debt. This aligns with the hypothesis that a gender-diverse board structure reduces agency conflicts by enhancing transparency and aligning with shareholder interests. Additionally, by investigating the moderating role of board compensation, this study extends prior research to assess whether compensation structures amplify the risk-averse inclinations of female directors, potentially curbing excessive leverage. These hypotheses contribute to a deeper understanding of how board diversity shapes financial risk management, particularly within developing economies.

4. Research Methodology

4.1. Design

The research methodology adopted for this study is a quantitative ex post facto design chosen to rigorously investigate the formulated hypotheses regarding the relationship between board gender diversity and debt structure choices among companies listed on the Tehran Stock Exchange (TSE). This design facilitates examining causal relationships between variables by analysing data collected after the events of interest have occurred (Nassirzadeh et al. 2023).

Methodological Choices and Exclusion of Certain Industries

This study adopted a quantitative ex post facto design, which allowed the authors to investigate the causal relationships between board gender diversity and debt structure by analysing data after the events of interest have occurred. This approach provided reliable insights into the effects of gender diversity on financial decision-making.

- a. Exclusion of Certain Industries: The study explicitly excluded financial institutions, investment companies, banks, insurance firms, leasing companies, and financial intermediaries from its sample. These sectors were left out to maintain data homogeneity because these industries operate under distinct regulatory and financial conditions that could skew the analysis. Financial institutions, for instance, often have different debt structures due to capital adequacy regulations and specific leverage rules that do not apply to non-financial firms. By focusing on operating companies, the study ensured that its findings regarding gender diversity and debt decisions were relevant and comparable across industries without distortions caused by sector-specific factors.
- b. Contextual Relevance: The study drew on a sample from the Tehran Stock Exchange (TSE)-listed companies over five years (2017–2021). The selected companies were required to have consistent fiscal years and active trading without significant interruptions, ensuring the analysis reflected the firms' ongoing business activities. By selecting firms that have not undergone prolonged trading halts, the study ensured that the data were not biased by irregular trading patterns that could misrepresent a firm's financial stability.

4.2. Data Collection

Data for this study were primarily sourced from the Rahavard Novin 3.8 software and the audited financial statements released by the TSE. These sources provided comprehen-

sive and reliable information for thoroughly analysing the investigated variables. Audited financial statements ensured the accuracy and credibility of the financial data used in the study (Eghbal et al. 2024; Zadeh et al. 2023).

4.3. Population and Sample

The population under scrutiny encompasses all companies listed on the TSE during the five years from 2017 to 2021. To ensure the sample's representativeness and validity findings, specific criteria were employed for sample selection:

Excluding certain entity types, such as investment companies, financial intermediaries, holdings, banks, insurance firms, and leasing companies, from the sample maintained homogeneity and focus on operating companies relevant to the study's objectives.

Consistency in activity and fiscal year: The selected companies needed to maintain consistent activity and fiscal year throughout the observed period to facilitate accurate comparison and analysis of financial data.

Accessibility of information: Adequate information required for calculating the research variables needed to be accessible for each selected company to ensure the feasibility of data analysis.

Exclusion of instances with prolonged trading halts: Instances of trading halts lasting longer than three months were considered exclusionary to avoid potential data distortions due to irregular trading activity irregularities.

Fiscal year-end alignment: The selected companies' fiscal year-end needed to align with the calendar year-end (20 March) to ensure consistency in reporting periods and facilitate comparability of financial data across companies.

Based on these criteria, a systematic sample comprising 114 companies was derived, totalling 570 firm years of observation. The sample selection process ensured the inclusion of companies that are representative of the population under study while adhering to the specified criteria for sample inclusion.

4.4. Analytical Approach

The analytical approach employed in this study involved Ordinary Least Squares (OLS) regression analysis, incorporating year- and industry-fixed effects. OLS regression is a widely used statistical technique for estimating the relationships between variables in regression models. By including year- and industry-fixed effects in the regression model, the analysis controlled for potential time and industry-specific factors that could influence the variables of interest, thereby enhancing the robustness and reliability of the results.

4.5. Hypotheses Testing

The hypotheses regarding the relationship between board gender diversity and debt structure choices were rigorously tested using the OLS regression analysis. The regression model allowed for examining the hypothesised relationships while controlling for potential confounding variables. Statistical tests, such as *t*-tests and F-tests, were conducted to assess the coefficients' significance and the regression model's overall fit.

4.6. Validity and Reliability

Ensuring the validity and reliability of the study findings was paramount to drawing accurate conclusions and making meaningful interpretations. To enhance internal validity, rigorous data cleaning procedures were implemented to eliminate errors and inconsistencies in the collected data. Additionally, sensitivity analyses were conducted to assess the robustness of the results to different model specifications and assumptions.

The research methodology employed in this study was designed to provide a comprehensive and rigorous analysis of the relationship between board gender diversity and debt structure choices among companies listed on the Tehran Stock Exchange. By adopting a quantitative ex post facto design, utilising reliable data sources, employing stringent sample selection criteria, and employing advanced statistical techniques, the study aimed to generate valuable insights into the factors influencing debt structure decisions in the context of board gender diversity.

4.7. Model and Variables

Following Datta et al. (2021) and as it is described in Table 1, the following model was used to test the hypotheses:

 $Leverage_{it} = \alpha + \beta_1 Female_{it} + \beta_2 Comp + \beta_3 Comp * Female_{it} + \beta_4 Size_{it} + \beta_5 MTB_{it} + \beta_6 FixedAssets_{it} + \beta_7 Profitability_{it} + \beta_8 ROA_{it} + \beta_9 AGE_{it}$ (1) + $\beta_{10} SalesGrowth_{it} + \varepsilon_{it}$

Variable Label Measurement Long-term debt divided by the market Dependent Leverage structure Leverage value of total assets Equal to 1 if there is at least one female Board gender diversity Female board member and 0 otherwise Independent Equal to 1 if board compensation is greater Board compensation Comp than the industry average, and 0 otherwise Firm size Size Natural log of total assets Market-to-book ratio MTB Market-to-book value of total assets Plant, property, and equipment (PPE) Fixed asset ratio Fixed assets divided by total assets Control Profitability Profitability Income reported for the current period Earnings before interest and tax divided by ROA Return on assets total assets AGE Firm age Current fiscal year minus first fiscal year

Table 1. Variables and measurements.

4.8. Findings

The descriptive statistics presented in Table 2 offer valuable insights into the variables under examination. Net financing, represented by debt, demonstrates a mean of 9,028,997, indicating considerable variability in the funding amounts across the sample companies. The maximum observed value of 399,976,695 in 2018 for Iran Khodro Company highlights the substantial financing undertaken by certain firms, while the minimum value of 45,543 in 2016 for Piazar Agro Industrial Company underscores the diversity in financing needs among companies.

Table 2. Descriptive statistics of the variables.

Variable	Mean	Median	SD	Max.	Min.	Skewness	Kurtosis
Leverage	0.5568	0.5501	0.206704	1.27394	0.03143	0.12557038	0.01155
Female	0.0323	0	0.164572	1	0	3.7551189	4.21278
MTB	6.11	3.21	2.89	51.68	0.250	0.9149	1.2670
Fixed Assets	43,335,584	442,779	16,492,437	18,476,568	9737	5.967334	4.77706

Variable	Mean	Median	SD	Max.	Min.	Skewness	Kurtosis
Profitability	2,078,434	257,788	156,487	382,775,011	-155,050,040	-3.165395	4.5467
ROA	0.15	0.12	0.1631	0.83	-0.40	0.5437	0.8525
AGE	25.66	22.00	13.6914	66	6	1.15041	0.32694
Sales Growth	0.47	0.36	0.8552	15.28	-0.91	3.59613	0.67899

Table 2. Cont.

Regarding debt structure choice, as measured by leverage, the mean value stands at 0.5568, indicating the average proportion of debt to total assets across the sample companies. The range of Leverage values is considerable, with a maximum of 1.27394 observed for Iran Khodro Company in 2017 and a minimum of 0.03143 for Shahdiran Company in 2019. These variations in leverage reflect differences in companies' debt management strategies and risk preferences.

In terms of board gender diversity, represented by the proportion of female directors (Female), the mean value is 0.0323, indicating limited diversity in board composition across the sample companies. Out of 570 firm years and 114 companies over five years, only 59 instances exhibit gender diversity on the board, reflecting the underrepresentation of women in corporate leadership roles in the context of the Tehran Stock Exchange.

An analysis of skewness and kurtosis coefficients for all variables indicates positive skewness, suggesting that the data distribution is skewed towards higher values, and kurtosis coefficients are within an acceptable range, indicating normal distribution. Notably, the composition of the board of directors exceeds the industry average in 38% of the sample, showing variations in board diversity across companies.

Following a thorough assessment of regression assumptions, multivariate regression analysis incorporating year- and industry-fixed effects was conducted to explore the relationship between board gender diversity, long-term leverage, and the moderating role of board compensation, as presented in Table 3.

Leverage _{it} =α+β ₁ Female _{it} +β ₂ Comp+β ₃ Comp*Female _{it} +β ₄ Size _{it} +β ₅ MTB _{it} +β ₆ FixedAssets _{it} + β ₇ Profitability _{it} +β ₈ ROA _{it} +β ₉ AGE _{it} +β ₁₀ SalesGrowth _{it} +ε _{it}					
Variable	Coefficient	SE	t-Statistic	Sig.	VIF
Constant	-3.53	4.7	-0.75	0.45	-
Female	-1.09	0.49	-2.23	0.02	1.57
Comp	0.056	0.34	3.14	0.035	2.65
Comp*Female	0.089	0.47	4.04	0.018	3.18
Size	0.2	0.089	2.24	0.04	1.156
MTB	0.14	0.078	1.84	0.065	2.03
Fixed Assets	-0.15	0.075	-2.07	0.038	1.78
Profitability	-3.4	0.72	-5.97	0.00	2.5
ROA	-0.63	0.63	-1.00	0.31	1.79
AGE	-0.18	0.64	-0.27	0.78	2.38
Sales Growth	0.15	0.64	0.23	0.82	2.59
	0.38	Adjusted R ² 0.30		0.30	
F-statistic (<i>p</i> -value)	44.76 (0.00)		Durbin-Wats	on statistic	2.12

Table 3. Summary estimation results.

The reported F-statistic, with a *p*-value smaller than 0.05 (*p*-value < 0.05), signifies the overall model's significance, indicating that the independent variables jointly explain a significant portion of the variance in the dependent variable. The coefficient of determination (\mathbb{R}^2) of 0.38 means that the variables included in the regression model explain 38% of the observed variation in long-term leverage.

The regression coefficient for board gender diversity (Female) is estimated at -1.09, with a *p*-value of 0.0254. This negative coefficient suggests that an increase in board gender diversity is associated with a decrease in long-term leverage, supporting the first hypothesis. The statistically significant *p*-value indicates that this relationship is unlikely to have occurred by chance alone, providing robust evidence for the negative association between board gender diversity and long-term leverage.

Furthermore, the regression coefficient for the interaction term between board gender diversity and board compensation (Comp*Female) is estimated at 0.089, with a *p*-value of 0.018. This positive coefficient suggests that board compensation amplifies the impact of board gender diversity on long-term leverage, supporting the second hypothesis. The statistically significant *p*-value indicates that the moderating effect of board compensation on the relationship between board gender diversity and long-term leverage is unlikely due to random variation, providing strong evidence for the moderating role of board compensation.

Overall, the multivariate regression analysis findings support the hypothesised relationships between board gender diversity, board compensation, and long-term leverage among companies listed on the Tehran Stock Exchange. These findings contribute to a deeper understanding of the factors influencing debt structure decisions in the context of board diversity and compensation practices, with implications for corporate governance and financial management strategies.

4.9. Robustness Tests: Addressing Endogeneity

Endogeneity issues, such as omitted variable bias and reverse causality, must be addressed to improve the credibility of the findings. One approach would be introducing an instrumental variable (IV) analysis to help mitigate these concerns. For example, external factors, such as changes in gender equality policies or legal requirements for female board representation, could serve as valid instruments. These instruments would isolate the exogenous variation in gender diversity, thus addressing concerns of reverse causality between board diversity and leverage. Additionally, the paper should discuss how factors like firm culture or the external regulatory environment may have been omitted from the analysis and how these factors could potentially bias the results.

We used robustness tests to address potential endogeneity issues affecting the relationship between board gender diversity and leverage decisions. We applied Heckman's (1979) model combined with an instrumental variable to address potential selection bias.

In the Heckman model, the base model we tested (in Table 3) deals with endogenous issues related to "self-selection", meaning that female directors may join companies with specific characteristics. We used a conditional logistic regression model to estimate the probability of creating gender diversity as a function of firm-specific characteristics (X variables include size, profitability, growth opportunities, leverage, abnormal returns, fixed asset ratio, and the standard deviation of stock returns over the fiscal year).

Table 4 tests for endogeneity to strengthen the reliability of its conclusions.

$$Female_{it} = \alpha + Y_s + \tau_t + \beta_1 Gender \ Equality_{it} + \theta X_{it} + \varepsilon_{it} \tag{2}$$

The results in Table 4 indicate that the coefficient of gender diversity, without considering the dependent variable used and without fixed effects, is positive and statistically significant at the 1% level. Therefore, our results support previous findings in favour of the first hypothesis, confirming that gender diversity is an important and influential factor in a company's leverage decisions. Including fixed effects, the coefficient remains positive and statistically significant at the 1% level, further supporting our findings that gender diversity is a crucial characteristic impacting corporate leverage decisions.

Variable	Coefficient (t-Statistic)			
Female	2.37 (0.01)			
Size	-0.087 (0.67)			
Profitability	0.056 (0.045)			
MTB	-0.035 (0.098)			
Lev	-0.078 (0.108)			
ABN_Earn	-0.091 (0.99)			
Fixed-Asset	-0.187 (0.98)			
RET_STD	0.87 (0.98)			
Adjusted R ²	0.246			

Table 4. Instrumental variables regression and Hausman test for endogeneity.

The endogeneity of the board gender diversity variable was examined. The gender diversity of the board of directors is an endogenous variable, and the average gender diversity in the industry can be an instrumental (exogenous) variable. Therefore, the endogeneity of this variable was checked using the 2SLS method.

$$[[Female]] _it = \alpha + \beta_1 [[Ind-Female]] _it + \sum Controls + \varepsilon_it$$
(3)

In this model, Ind-Female is the average gender diversity of the industry, and \sum Controls are the control variables of the model (1).

Fisher's statistic in this estimate is 2.87, and its significance level is 0.003, which indicates the effect of the instrumental variable (gender diversity of the industry) on the independent variable (gender diversity).

In the second stage, the dependent variables (Leverage) are regressed on the predicted values obtained from the gender diversity of the first stage along with the control variables:

 $[Leverage] _it = \alpha + \beta_1 [Female-Predicted] _it + \beta_2 Comp +$ $\beta_3 Comp*Female-Predicted] _it + \sum Controls + \varepsilon_it$ (4)

This stage examines the impact of the exogenous component of board gender diversity on debt structure, isolating any potential endogeneity. If the coefficient on the predicted board gender diversity is statistically significant in the second stage, it indicates a causal impact on debt structure. The coefficient β_1 (Female-Predicted) has an estimation of -0.98, and its significance level and t-statistics are 0.002 and 2.43, respectively, which indicates the significant impact of gender diversity on the debt structure. Also, the β_3 coefficient (Comp*Female-Predicted) is 0.073, and its significance level and t-statistics are 0.034 and 3.48, respectively, which shows that the remuneration of the board of directors highlights the effect of gender diversity on the debt structure.

4.10. Ethical Considerations

The study recognises the need for ethical diligence throughout the research process. Although no explicit participant interaction or sensitive personal data are involved in the study, ethical issues related to data privacy and confidentiality are paramount. The study utilises publicly available audited financial statements and data from the Rahavard Novin software, which compiles financial data for listed companies on the TSE.

Potential ethical issues stem from using secondary data related to corporate governance, such as board compensation details and debt structure decisions, which could reflect corporate strategies. As such, the researchers ensure that no confidential or proprietary information that could harm individual firms or stakeholders is disclosed. Data anonymisation and compliance with the ethical guidelines of financial research, such as respecting the corporate governance laws in Iran and abiding by data protection regulations, are upheld throughout the study.

Additionally, the research navigates the socio-cultural sensitivity surrounding gender roles in Iran, ensuring that the results are interpreted objectively without cultural bias despite the patriarchal context in which the study is situated.

5. Discussion

The role of a company's board of directors is crucial in overseeing executive management and safeguarding shareholder interests. Extensive research has highlighted the value of board diversity, noting that female members bring unique experiences and ethical perspectives. Board diversity can manifest through observable characteristics such as race, gender, and age as well as intangible qualities like education and work experience. Our study supports the notion that female directors exhibit a heightened focus on monitoring activities and tend to prefer short-term debt instruments to align the interests of managers and shareholders. This finding underscores the belief that increased female representation enhances corporate risk management practices. From both social and economic perspectives, it is imperative to advocate for gender equality in management positions, as discriminatory practices can hinder effective human resource utilisation and impede organisational performance.

Our study has revealed a negative association between board gender diversity and long-term leverage, which aligns with the gender ethics framework. Contrary to behavioural perspectives suggesting women's risk aversion influences debt structure decisions, our findings suggest otherwise, consistent with prior research. We found that board compensation strengthens the relationship between board gender diversity and debt structure. This indicates that higher levels of board compensation contribute to better alignment of management and shareholder interests while mitigating managerial risk-taking behaviours. These results align with the literature by Brockman et al. (2010) and Datta et al. (2021), highlighting the importance of board composition and compensation practices in shaping corporate financial decisions.

Agency Theory provides a robust framework for describing the results of this study. Agency Theory elucidates the conflicts of interest between principals (shareholders) and agents (managers). This theory is particularly apt for examining corporate governance issues, such as board composition and its impact on financial decisions. Agency Theory posits that agents may act in their self-interest, potentially at the expense of principals. Female directors, hypothesised to exhibit higher ethical standards, can mitigate these conflicts by aligning debt structure decisions more closely with shareholder interests.

The study supports the hypothesis that companies with female directors prefer shorter-term debt, reducing long-term leverage. This decision aligns with the principal's interest by enhancing transparency and accountability and mitigating agency conflicts.

Agency Theory also addresses risk-taking behaviours in corporate decision-making. Female directors, being more risk-averse, are expected to favour financial strategies that minimise risk, thereby protecting shareholder value. Female directors' preference for longer-term debt reflects a conservative approach to debt management, reducing liquidity and refinancing risks, which aligns with shareholders' desire for stability and lower risk exposure.

Agency Theory highlights the complexity of incentive structures. Directors' compensation plays a crucial role in aligning the interests of managers and shareholders, influencing risk-taking behaviours and financial decisions. The study reveals that directors' compensation moderates the impact of board gender diversity on financial leverage. Higher compensation levels strengthen the relationship between gender diversity and conservative leverage decisions, indicating that well-compensated female directors are more effective in reducing agency costs.

In summary, Agency Theory comprehensively explains the study's findings. Female directors on corporate boards contribute to more ethical, risk-averse decision-making and enhanced alignment of managerial actions with shareholder interests. The moderation effect of directors' compensation further underscores the importance of incentive structures in mitigating agency conflicts and achieving optimal corporate governance outcomes. This theoretical framework supports the empirical results and offers valuable insights into the dynamics of board composition and financial decision-making in the context of gender diversity. Here are some of the contributions of the current study:

5.1. Novel Insights in Financial Decision-Making and Risk Management

- a. Gender Influence on Corporate Financing: This study provides new evidence on how female board representation impacts financial decisions, specifically in debt structure and risk management. The research shows that female directors tend to favour more conservative financial strategies, such as reducing long-term debt. This risk-averse behaviour supports the view that gender diversity can act as a moderating force in corporate risk-taking, promoting stability during periods of financial decision-making. This is particularly valuable in corporate governance frameworks where risk mitigation is critical.
- b. Moderating Role of Board Compensation: A key finding of this study is the moderating effect of board compensation on the relationship between gender diversity and debt structure. This insight is novel because it highlights how compensation structures can influence the extent to which female directors' risk-averse behaviours affect financial decisions. It suggests that well-compensated female directors may be better positioned to ensure financial prudence and alignment with shareholder interests.
- c. Application to a Developing Economy: The study's focus on Iran adds value to the existing literature by exploring gender diversity in a non-Western context. Iran's distinct socio-cultural and regulatory environment, which includes Islamic finance principles and corporate governance limitations, offers a unique setting to understand how gender diversity influences financial behaviour. The research expands the discussion beyond Western economies, providing insights into how female board rep-

resentation could influence risk management strategies in emerging markets. This is particularly relevant for economies where corporate governance structures might not be as robust as those in more developed countries.

- 5.2. Contribution to the Broader International Literature
- a. Broader Implications for Gender Diversity: By focusing on a developing economy like Iran, the paper offers significant contributions to the global literature on gender diversity and corporate governance. It highlights how cultural and institutional factors affect the role of women in decision-making, offering comparative insights for other developing economies with similar socio-economic characteristics. For example, the patriarchal norms in Iran may limit female directors' influence. Still, the study shows that where they can exert influence, they can significantly impact financial risk management decisions.

This study adds value to the global discussion on gender diversity and corporate governance by illustrating how these factors play out in an Iranian context. The Iranian market, characterised by complex regulatory structures and societal norms regarding gender roles, provides an ideal setting to explore the broader impact of female participation in board decisions. The study extends the conversation beyond the typical Western-centric perspectives, where much of the literature has focused. It highlights how board gender diversity affects financial risk-taking and management, which may differ based on cultural and regulatory environments. The research underscores the necessity of promoting gender diversity in corporate governance to improve decision-making processes, especially in emerging markets. It highlights the potential benefits of including women in key decisionmaking roles to enhance risk management strategies in corporate finance.

- b. Expanding the Scope of Corporate Governance Studies: Most studies on gender diversity have concentrated on developed economies, with Western governance structures at the forefront. This paper contributes to the global discussion by showing that even in countries with weaker governance mechanisms, gender diversity still plays a crucial role in shaping ethical financial decisions. The unique challenges faced by Iranian corporations—such as economic instability and regulatory limitations—allow for a broader understanding of how board gender diversity can drive better risk management outcomes across different economic and cultural landscapes.
- c. Contextual Significance: Iran's Islamic finance system, where interest-based transactions are prohibited, creates a unique backdrop for studying financial decisionmaking. This context further emphasises the need for ethical governance and conservative financial practices—to which female directors can contribute significantly. The findings suggest promoting gender diversity in such contexts could lead to more ethical and transparent financial management. This provides a valuable case study for international governance bodies and policymakers looking to enhance global diversity and risk management practices.

In conclusion, the paper provides new perspectives on how board gender diversity intersects with corporate financial risk management in emerging markets like Iran. Its insights into the role of female directors in promoting conservative debt structures and its focus on the moderating role of compensation systems enrich the global discourse on corporate governance, diversity, and risk management.

6. Conclusions

This study highlights the critical role of board gender diversity in corporate risk management and financial decision-making. The findings indicate that female directors contribute to more conservative debt structures, reducing overall financial risk, mainly through shorter-term debt instruments. Furthermore, the moderating effect of board compensation strengthens the alignment between management and shareholder interests, enhancing risk mitigation and improving the ethical governance of the firms.

From a practical perspective, the results underscore the importance of promoting gender diversity on corporate boards to optimise financial decision-making and manage risk effectively. Companies prioritising diversity can benefit from improved risk management practices and more sustainable financial structures. Policymakers and regulators may also consider these findings when developing frameworks to encourage gender diversity and robust governance practices in corporate boards, particularly in emerging markets.

6.1. Study Limitations and Their Impact on Generalizability

6.1.1. Geographical and Cultural Context

The study is conducted within Iran's specific socio-cultural and regulatory framework, a developing economy with distinct cultural and institutional norms. These norms, particularly regarding gender roles and corporate governance, differ significantly from those in Western and other global economies. Iran's patriarchal context may impose implicit biases and resistance against female board members, which might not be as pronounced in different regions. Therefore, while the findings offer valuable insights into the influence of board gender diversity on debt structure and risk management, they may not be readily generalisable to countries with different governance standards, regulatory frameworks, and cultural settings.

6.1.2. Exclusion of Specific Industries

The study excludes several vital industries, including financial institutions, banks, insurance firms, and investment companies. Due to regulatory and financial constraints specific to their sectors, these industries tend to have distinct capital structures, leverage practices, and risk management approaches. The exclusion ensures data homogeneity but limits the scope of the findings. Thus, the results might not apply to industries where sector-specific regulations influence leverage and debt decisions. Further studies that include a broader range of industries could provide more comprehensive insights.

6.1.3. Timeframe of the Study

The data collection spans five years (2017–2021), a relatively short period to capture long-term trends in board gender diversity and financial decision-making. While the study does account for fixed-year effects, longer-term studies might reveal different patterns or changes in gender dynamics and risk management over time, especially in crisis periods or periods of economic recovery. Extending the time limit could also mitigate the influence of short-term economic conditions, such as market fluctuations or changes in government policies, which may have influenced the results within the selected period.

6.1.4. Corporate Governance and Board Compensation

The study highlights board compensation as a moderating factor in the relationship between board gender diversity and debt structure. However, the specific governance mechanisms in place in Iran, such as weaker corporate governance structures and lack of transparency, may further limit the applicability of the results to other countries with more robust governance frameworks. In economies with better-developed accountability and transparency standards, the moderating effect of board compensation on financial decision-making might operate differently.

6.1.5. Data Source and Availability

The reliance on the Rahavard Novin software and audited financial reports ensured the study's accuracy. However, these sources may not fully capture the nuanced dynamics of board discussions or the informal influences in boardrooms. Additionally, this reliance on secondary data limits the study's ability to incorporate qualitative insights into how gender diversity shapes financial decision-making.

6.1.6. Recommendations for Future Research

Future research could benefit from expanding the geographical scope to include crosscountry comparisons, especially in countries with differing levels of board gender diversity and corporate governance structures. Additionally, longitudinal studies could assess the long-term effects of female board representation on financial decision-making, especially during economic volatility. Researchers could also explore qualitative data, such as interviews or case studies, to better understand the interpersonal dynamics within diverse boards. The study's limitations, such as its focus on companies listed on the Tehran Stock Exchange, suggest that future research should explore the broader application of these findings across different industries and regions. Further exploration of other factors, such as economic instability and regulatory frameworks, could provide deeper insights into the relationship between board composition, risk management, and corporate finance.

Author Contributions: Conceptualisation, S.J., A.P. and S.H.; methodology, S.J., A.P. and S.H.; software, S.J., A.P. and S.H.; validation, S.J., A.P., D.A. and S.H.; formal analysis, S.J., A.P. and S.H.; investigation, S.J., A.P. and S.H.; resources, S.J., A.P. and S.H.; data curation, S.J., A.P. and S.H.; writing—original draft preparation, S.J., A.P. and S.H.; writing—review and editing, D.A. and H.Y.; visualisation, D.A. and H.Y. supervision, A.P.; project administration, S.J., A.P., S.H., D.A. and H.Y. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Data Availability Statement: Companies' data can be purchased via the TSE website at https://www. ceicdata.com/en/iran/tehran-stock-exchange-market-capitalization/market-capitalization-tehran-st ock-exchange-tse (accessed on 10 December 2024). The relevant data are called 'rahavard-novin', are available under the 'products' category, and can be accessed at: https://mabnadp.com/products/ rahavard-novin (accessed on 10 December 2024).

Conflicts of Interest: The authors declare no conflicts of interest.

References

- Abobakr, Mohamed Galal, and Khairy Elgiziry. 2016. The effect of board characteristics and ownership structure on the corporate financial leverage. *Accounting and Finance Research* 5: 1–14. [CrossRef]
- Abor, Joshua. 2008. Agency theoretic determinants of debt levels: Evidence from Ghana. *Review of Accounting and Finance* 7: 183–92. [CrossRef]
- Adams, Renée B., and Daniel Ferreira. 2009. Women in the boardroom and their impact on governance and performance. *Journal of Financial Economics* 94: 291–309. [CrossRef]
- Albaum, Gerald, and Robert A. Peterson. 2006. Ethical Attitudes of Future Business Leaders: Do They Vary by Gender and Religiosity? Business & Society 45: 300–21. [CrossRef]
- Bajtelsmit, Vickie L., and Jack L. VanDerhei. 1997. A Risk Aversion and Pension Investment Choices. In *Positioning Pensions for the Twenty-First Century*. Edited by Michael S. Gordon, Olivia S. Mitchell and Marc M. Twinney. Philadelphia: University of Pennsylvania Press.
- Barnett, Tim, Gene Brown, and Ken Bass. 1994. The Ethical Judgments of College Students regarding Business Issues. *Journal of Education for Business* 69: 333–38. [CrossRef]
- Bernardi, Richard A., and Donald F. Arnold Sr. 1997. An Examination of Moral Development within Public Accounting by Gender, Staff Level, and Firm. Contemporary Accounting Research 14: 653–68. [CrossRef]
- Brockman, Paul, Xiumin Martin, and Emre Unlu. 2010. Executive Compensation and the Maturity Structure of Corporate Debt. *The Journal of Finance* 65: 1123–61. [CrossRef]

- Carter, Mary Ellen, Francesca Franco, and Mireia Gine. 2017. Executive Gender Pay Gaps: The Roles of Female Risk Aversion and Board Representation. *Contemporary Accounting Research* 34: 1232–64. [CrossRef]
- Catalyst. 2017. 2016 Catalyst Census: Women and Men Board Directors. Available online: https://www.catalyst.org/research/2016 -catalyst-census-women-and-men-board-directors/ (accessed on 6 December 2024).
- Chen, Jie, Woon Sau Leung, and Marc Goergen. 2017. The impact of board gender composition on dividend payouts. *Journal of Corporate Finance* 43: 86–105. [CrossRef]
- Choi, Jaewon, Dirk Hackbarth, and Josef Zechner. 2017. Granularity of Corporate Debt. Available online: https://ssrn.com/abstract= 3055884 (accessed on 5 April 2024).
- Chun, Rosa. 2005. Ethical Character and Virtue of Organisations: An Empirical Assessment and Strategic Implications. *Journal of Business Ethics* 57: 269–84. [CrossRef]
- Cumming, Douglas, Xiaohu Guo, and Lukai Yang. 2024. REITs board gender diversity: The spillover effect of the Big Three campaign. Journal of Economics and Finance 48: 889–923. [CrossRef]
- Daryaei, Abbas Ali, Yasin Fattahi, Davood Askarany, Saeed Askary, and Mahdad Mollazamani. 2022. Accounting Comparability, Conservatism, Executive Compensation-Performance, and Information Quality. *Journal of Risk and Financial Management* 15: 489. [CrossRef]
- Datta, Sudip, M. A. I. Iskandar-Datta, and Kartik Raman. 2005. Managerial Stock Ownership and the Maturity Structure of Corporate Debt. *The Journal of Finance* 60: 2333–50. [CrossRef]
- Datta, Sudip, Trang Doan, and Francesca Toscano. 2021. Top executive gender, board gender diversity, and financing decisions: Evidence from debt structure choice. *Journal of Banking & Finance* 125: 106070. [CrossRef]
- Dawson, Leslie M. 1997. Ethical differences between men and women in the sales profession. *Journal of Business Ethics* 16: 1143–52. [CrossRef]
- Diamond, Douglas W. 1991. Debt Maturity Structure and Liquidity Risk. The Quarterly Journal of Economics 106: 709–37. [CrossRef]
- Duppati, Geeta, Narendar V. Rao, Neha Matlani, Frank Scrimgeour, and Debasis Patnaik. 2020. Gender diversity and firm performance: Evidence from India and Singapore. *Applied Economics* 52: 1553–65. [CrossRef]
- Eghbal, Maryam, Farzaneh Nassirzadeh, and Davood Askarany. 2024. The Relationship Between Non-additivity Valuations, Cash Flows and Sales Growth. *Computational Economics* 64: 429–59. [CrossRef]
- Flannery, Mark J. 1986. Asymmetric Information and Risky Debt Maturity Choice. The Journal of Finance 41: 19–37. [CrossRef]
- Geiger, Marshall A., and Brendant T. O'Connell. 1998. Accounting Student Ethical Perceptions: An Analysis of Training and Gender Effects. *Teaching Business Ethics* 2: 371–88. [CrossRef]
- Gormley, Todd A., Vishal K. Gupta, David A. Matsa, Sandra C. Mortal, and Lukai Yang. 2023. The big three and board gender diversity: The effectiveness of shareholder voice. *Journal of Financial Economics* 149: 323–48. [CrossRef]
- Harris, Cindy K. 2014. Women directors on public company boards: Does a critical mass affect leverage? Paper presented at 2014 Proceedings of the Northeastern Association of Business, Economics, and Technology, State College, PA, USA, October 23–24.
- Heckman, James J. 1979. Sample selection bias as a specification error. Econometrica: Journal of the Econometric Society, 153-61. [CrossRef]
- Hinz, Richard P., David D. McCarthy, and John A. Turner. 1997. Are women more conservative investors? Gender differences in participant directed pension investments. In *Positioning Pensions for the Twenty-First Century*. Philadelphia: University of Pennsylvania Press, pp. 91–103.
- Ho, Simon S. M., Annie Yuansha Li, Kinsun Tam, and Feida Zhang. 2015. CEO Gender, Ethical Leadership, and Accounting Conservatism. *Journal of Business Ethics* 127: 351–70. [CrossRef]
- Hosni Al-Ghar, Masoud, and Nezamoddin Rahimian. 2018. The effect of CEO overconfidence on debt maturity structure in companies listed on the Tehran Stock Exchange. *Asset Management and Financing* 6: 98–106. [CrossRef]
- Hoyt, Crystal L., and Susan E. Murphy. 2016. Managing to clear the air: Stereotype threat, women, and leadership. *The Leadership Quarterly* 27: 387–99. [CrossRef]
- Huang, Ronghong, Kelvin Jui Keng Tan, and Robert W. Faff. 2016. CEO overconfidence and corporate debt maturity. *Journal of Corporate Finance* 36: 93–110. [CrossRef]
- Jane Lenard, Mary, Bing Yu, E. Anne York, and Shengxiong Wu. 2014. Impact of board gender diversity on firm risk. *Managerial Finance* 40: 787–803. [CrossRef]
- Jianakoplos, Nancy Ammon, and Alexandra Bernasek. 1998. Are Women More Risk-Averse? Economic Inquiry 36: 620–30. [CrossRef]
- Labelle, Réal, Claude Francoeur, and Faten Lakhal. 2015. To Regulate Or Not To Regulate? Early Evidence on the Means Used Around the World to Promote Gender Diversity in the Boardroom. *Gender, Work & Organisation* 22: 339–63. [CrossRef]
- Larkin, Joseph M. 2000. The Ability of Internal Auditors to Identify Ethical Dilemmas. Journal of Business Ethics 23: 401–9. [CrossRef]
- Levi, Maurice, Kai Li, and Feng Zhang. 2014. Director gender and mergers and acquisitions. *Journal of Corporate Finance* 28: 185–200. [CrossRef]
- Levin, Irwin P., Mary A. Snyder, and Daniel P. Chapman. 1988. The Interaction of Experiential and Situational Factors and Gender in a Simulated Risky Decision-Making Task. *The Journal of Psychology* 122: 173–81. [CrossRef]

- Li, Yiwei, and Yeqin Zeng. 2019. The impact of top executive gender on asset prices: Evidence from stock price crash risk. *Journal of Corporate Finance* 58: 528–50. [CrossRef]
- Linley, P. Alex, John Maltby, Alex M. Wood, Stephen Joseph, Susan Harrington, Christopher Peterson, Nansook Park, and Martin E. P. Seligman. 2007. Character strengths in the United Kingdom: The VIA Inventory of Strengths. *Personality and Individual Differences* 43: 341–51. [CrossRef]
- Mohammadi, S. A., and M. Zanjirdar. 2020. Cash Flow Risk and Capital Structure Decisions. Paper Presented at 4th International Conference on Financial Management, Business, Banking, Economics, and Accounting. Available online: https://civilica.com/ doc/1126996 (accessed on 2 December 2020).
- Nassirzadeh, Farzaneh, Davood Askarany, and Solmaz Arefi-Asl. 2023. The relationship between changes in corporate governance characteristics and intellectual capital. *Journal of Risk and Financial Management* 16: 133. [CrossRef]
- O'Donnell, Ed, and Eric N. Johnson. 2001. The Effects of Auditor Gender and Task Complexity on Information Processing Efficiency. International Journal of Auditing 5: 91–105. [CrossRef]
- Peng, Xuhui, Yi-Chen Lan, Jingduan Li, and Hanlu Fan. 2023. Board gender diversity, national culture, and water disclosure of multinational corporations. *Applied Economics* 55: 1581–602. [CrossRef]
- Peterson, Christopher, and Martin E. P. Seligman. 2003. Character Strengths before and after September 11. *Psychological Science* 14: 381–84. [CrossRef]
- Pierce, Bernard, and Breda Sweeney. 2010. The Relationship between Demographic Variables and Ethical Decision Making of Trainee Accountants. *International Journal of Auditing* 14: 79–99. [CrossRef]
- Pouryousof, Azam, Farzaneh Nassirzadeh, Reza Hesarzadeh, and Davood Askarany. 2022. The Relationship between Managers' Disclosure Tone and the Trading Volume of Investors. *Journal of Risk and Financial Management* 15: 618. [CrossRef]
- Rajan, Raghuram, and Andrew Winton. 1995. Covenants and collateral as incentives to monitor. The Journal of Finance 50: 1113-46. [CrossRef]
- Richards, Wendy. 2003. Gender, Diversity and Trade Unions: International Perspectives. *Gender, Work & Organisation* 10: 508–11. [CrossRef]
- Rosser, Heidi, Irene Ryan, and Barbara Myers. 2024. Unmasking the politics of policy-driven change (or not) for gender diversity. *Gender, Work & Organization* 31: 133–51. [CrossRef]
- Said, Mona, Rami Galal, and Mina Sami. 2022. Gender diversity, productivity, and wages in private Egyptian firms. *Applied Economics* 54: 4433–48. [CrossRef]
- Sepasi, S., and L. Abdoli. 2015. The effects of female board representation on firm value and financial performance. *Financial Accounting & Auditing Research* 8: 39–58.
- Shan, Yuan George. 2019. Managerial ownership, board independence and firm performance. *Accounting Research Journal* 32: 203–20. [CrossRef]
- Shandiz, Mohsen Tavakoli, Farzaneh Nassir Zadeh, and Davood Askarany. 2022. The Interactive Effect of Ownership Structure on the Relationship between Annual Board Report Readability and Stock Price Crash Risk. *Journal of Risk and Financial Management* 15: 268. [CrossRef]
- Sila, Vathunyoo, Angelica Gonzalez, and Jens Hagendorff. 2016. Women on board: Does boardroom gender diversity affect firm risk? Journal of Corporate Finance 36: 26–53. [CrossRef]
- Stulz, René. 2001. Does financial structure matter for economic growth? A corporate finance perspective. In *Financial Structure and Economic Growth: A Cross-Country Comparison of Banks, Markets, and Development*. Cambridge, MA: The MIT Press.
- Sun, Xu, and Tianming Zhang. 2021. Board gender diversity and corporate labor investment efficiency. *Review of Financial Economics* 39: 290–313. [CrossRef]
- Sundén, Annika E., and Brian J. Surette. 1998. Gender Differences in the Allocation of Assets in Retirement Savings Plans. *The American Economic Review* 88: 207–11.
- Terjesen, Siri, Eduardo Barbosa Couto, and Paulo Morais Francisco. 2016. Does the presence of independent and female directors impact firm performance? A multi-country study of board diversity. *Journal of Management & Governance* 20: 447–83.
- Ullah, Farid, Ping Jiang, Yasir Shahab, Hai-Xia Li, and Lei Xu. 2020. Block ownership and CEO compensation: Does board gender diversity matter? *Applied Economics* 52: 583–97. [CrossRef]
- Zadeh, Farzaneh Nassir, Davood Askarany, Ali Shirzad, and Mahdi Faghani. 2023. Audit committee features and earnings management. *Heliyon* 9: e20825. [CrossRef]
- Zadeh, Farzaneh Nassir, Davood Askarany, and Solmaz Arefi Asl. 2022. Accounting Conservatism and Earnings Quality. *Journal of Risk and Financial Management* 15: 413. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.





Article Dividend-Based Labor Remuneration and Tradable Shares in Worker Cooperatives

Ermanno C. Tortia 匝

Department of Economics and Management, University of Trento, 38122 Trento, Italy; ermanno.tortia@unitn.it; Tel.: +39-0461-28-2383

Abstract: This paper analyzes the possibility of creating worker cooperatives in which members are paid not through wages but through dividends calculated on the organization's residual income, as stipulated by the economic theory of the labor-managed firm. It is shown how dividends paid to members can be linked to the value of their financial participation in the capital of the cooperative. In the presence of a financial market, cooperative shares would be issued and allocated to both members and non-member outside investors, thus addressing the problem of the under-capitalization of worker cooperatives. It is hypothesized that the strong financial incentives of this type of capital structure, together with involvement in the democratic governance of the cooperative, peer pressure, and other horizontal monitoring mechanisms, would support members' intrinsic motivation to work and help overcome the problem of free-riding in the labor process. Flexible economic and financial structure in the absence of fixed wages would promote job stability, as already observed in existing worker cooperatives.

Keywords: worker cooperative; dividends; wages; cooperative shares; employment relation; free-riding

"...reorganization of labor in the form of association to replace the current wage system will, we believe, be the basis of the future economic world." (Giuseppe Mazzini 1861)

1. Introduction

This paper aims to show how a different interpretation of labor remuneration for worker members of worker cooperatives in terms of dividends from net enterprise income rather than wages may allow the development of new financial instruments, or cooperative shares, that can enable this kind of mutualistic organization, at least in principle, to achieve efficient capitalization and competitive effectiveness, as demanded in the seminal contributions of Vanek (1970, 1977), Furubotn and Pejovich (1970), Furubotn (1976), Jensen and Meckling (1979), and Pejovich (1990). This paper is developed as a thought experiment since, to the author's best knowledge, the proposed financial instruments have not yet been introduced in any worker cooperative. Some examples of similar capital structures and financial instruments, however, do exist in other cooperative forms (worker and non-worker). They are introduced in the reminder of the article for explanatory purposes.

The theoretical argument starts from Ward's (1958), Vanek's (1970), and Meade's (1972) neoclassical model of the labor-managed firm (Bonin et al. 1993; Uvalic 2010; Pérotin 2013). In this model, designed to analyze the ownership rights of productive organizations in the former Republic of Yugoslavia, members of worker cooperatives are remunerated through dividends of the enterprise net income (EBIT or Earning Before Taxes, that is



Academic Editors: Eulália Mota Santos and Margarida Freitas Oliveira

Received: 11 November 2024 Revised: 2 December 2024 Accepted: 28 December 2024 Published: 31 December 2024

Citation: Tortia, Ermanno C. 2025. Dividend-Based Labor Remuneration and Tradable Shares in Worker Cooperatives. *Risks* 13: 5. https:// doi.org/10.3390/risks13010005

Copyright: © 2024 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https://creativecommons.org/ licenses/by/4.0/). value added net of the cost of capital, depreciation, and amortization), and not through wages, which instead represent the cost of labor in investor-owned, or capitalist, companies. While in capitalist companies the dividend is usually defined as a sum of money paid by a company to its shareholders from net profits, in cooperative enterprises the dividend paid to members is defined as a refund or rebate. As it discusses the possibility of introducing tradable cooperative shares, this paper also uses the concept of dividend in cooperatives (Mikami 2016).

Dividend-based compensation is consistent with worker cooperatives' economic and institutional nature, since worker members are not employees but have membership or control rights over it. On the other hand, it is well known that the historical and institutional evolution of worker cooperatives in market economies, especially in Western countries, led them to pay members as salaried employees, just like in capitalist firms. In most countries, for example, in Italy, members are salaried workers. At the same time, an additional associational relationship is established with the worker cooperative, which confers rights of membership or control over it (Fici 2016). In some cases, such as the Mondragón group in Spain, worker members receive, in addition to wages, rebates as shares in the net earnings at the end of each accounting period. This discrepancy between economic theory and institutional evolution creates a case for new theoretical elaboration to consider the economic and financial implications of distributive models that replace wages with dividend-based compensation.

This paper is organized as follows: Section 2 discusses the economic nature of the employment relationship in capitalist enterprises and shows how dividends can substitute for wages in worker cooperatives. Section 3 proposes a new model of worker cooperatives in which cooperative shares are introduced in the hands of worker members and also (potentially) outside investors and remunerated through value-added distribution (dividends), including the issuance and pricing of shares to members, new entrants, and outside investors. Section 4 addresses some relevant organizational consequences of introducing cooperative shares having to do with labor productivity, free-riding, worker livelihood, and layoffs. Section 5 concludes the paper.

2. Overcoming the Employment Relation

In recent years, some authors, both from the Marxist and radical liberal traditions, have developed a series of microeconomic and organizational theoretical and philosophical arguments that demand the overcoming, in worker cooperatives, of the hierarchical relation between employer and salaried workers or employees (Ellerman 1984, 2005, 2016, 2021; Screpanti 2001). Underlying these arguments is the idea that the labor contract is the instrument through which capitalist enterprises establish the subordination of wage earners to investors through the imposition of a hierarchical relationship. This implies the use of the productive capacities of workers in a relationship of authority (Screpanti 2017), as also affirmed by neo-institutionalist classics (Coase 1937; Simon 1951).

The imposition of hierarchy results from the payment of a wage—the equilibrium price of labor as a factor of production in the labor market—by the employer in exchange for the worker ceding the right to use her/his labor services. The creation of democratic workerled enterprises based on the "one head or one member, one vote" rule of governance, i.e., the worker cooperative or labor-managed firm (Vanek 1970, 1977), would require overcoming both hierarchy (the employment relation) and the remuneration form inherent in that relation, that is the wage as fixed payment for labor services (see also Ellerman 1984, 2005; Dow 2003; Cheney et al. 2014; Screpanti 2017). To these are added some other observations typical of the radical neoliberal tradition that spread in recent decades, which emphatically evidence the importance of remunerating entrepreneurial work activity based on the dimension of the enterprise residual and not in terms of a fixed contractual wage (Kiyosaki 2017).

Philosophical arguments have also been proposed as a basis for overcoming the employment relation since the transfer of labor services from the employee to the employer would imply the transfer of subjective responsibility for the decisions made by the worker. This transfer is incompatible with the Western tradition of both natural rights in Locke's (1690) labor theory of property and also the theory of subjective rights as inalienable and non-transferable or duplicable personal rights, analogous to civil and political rights (Ellerman 2005, 2016, 2021). In Locke's (1690) labor theory of property, natural rights require that all the product of labor be appropriated by those who produced it.

In this paper, such economic and philosophical propositions enter as necessary premises into the proposal of financial instruments that deny both the hierarchy between employer and employee and the payment of a fixed salary for labor services. On the other hand, members' control rights in the cooperative enterprise are interpreted as personal rights that cannot be alienated, duplicated, or transferred. The delegation (*concessio*) of decision-making responsibility to elected representatives in a mutualistic and democratic governance structure in cooperatives replaces the transfer (*translatio*) or alienation of responsibility that characterizes capitalist companies (Ellerman 2005, 2016, 2021; Cheney et al. 2014).

2.1. Dividends in Worker Cooperatives as Payment of Labor Contributions and Remuneration of Capital Shares

The thought experiment envisaged in this paper begins by considering the simplest possible form of labor remuneration. It then makes it more complex and realistic as new elements of the functioning of labor markets, financial markets, and organization are added to the initial picture.

The simplest form of remuneration for work is the (self-)payment of residual income to a self-employed or independent worker, such as a professional or artisan. Radical liberal and socialist traditions in political economy, for example, Mazzini (1860) in Italy and Mill (1871) in the UK, interpreted worker cooperatives simply as associative and mutualistic business forms, in which the members are independent or self-employed workers who join together to establish a mutually beneficial, non-hierarchical, and democratic economic venture. The same interpretation formed the legal basis for the creation of the consumer cooperatives of the Rochdale equitable pioneers in England in 1844 and the Mondragón group of workers' cooperatives in the 1950s in the Basque region of Spain (Whyte and Whyte 1991; Morrison 1989; Marcuello 2023). In the Basque context, the military regime of Francisco Franco did not prohibit the formation of worker cooperatives but considered them to be associations of self-employed workers. Still today, in Mondragón, the monthly remuneration of worker-members is not considered a real salary, but an "advance" on residual net income. It is supplemented at the end of the year by rebates on the company's overall economic result (Whyte and Whyte 1991; Morrison 1989). In this tradition, this paper considers worker cooperatives as self-employed associations in which labor compensation is equivalent to a self-employed worker's labor income when calculated as part of the collective net income earned by the cooperative. Taken literally, in the absence of wage labor, this implies that the cost of labor or wage bill in this kind of organization is zero.

In financial terms, on the other hand, the dividend is interpreted as the return on the financial investment in the equity capital of a company. The apparent incompatibility between these two definitions lies in considering dividend-based labor compensation as a share of the net economic residual *in the absence of labor costs* (wages). In the financial definition, on the other hand, dividend refers to a share of net income or profit *after subtracting labor costs*. The two definitions can be reconciled if the dividend in its financial definition coincides with the remuneration of labor. This can happen when workers control the firm and are paid a share of the net residual. This paper illustrates how dividend-based compensation corresponds to the economic value of members' financial participation in the cooperative's capital. It will be performed in conceptual terms, using simple calculations and discussing real cases of employee financial participation in employee-owned companies and member-owned cooperatives (not necessarily worker cooperatives).

2.2. In the Absence of Capital Markets: The Capital Structure in the Mondragón System and the Slovenian Proposal of European ESOP

When worker cooperatives or worker-owned enterprises are characterized by collective or individual ownership, but workers cannot exchange their membership or financial positions in a market, the value of those positions remains implicit, since the only measurable parts of value added are labor income paid to members and the cost of capital. Paid labor income may correspond to some market value of workers' financial participation, but this is indeterminate in the absence of supply and demand for financial positions. In fact, some well-known worker financial participation schemes and capital structures of worker cooperatives, such as the Mondragón group in Spain or the generality of worker cooperatives in Italy, do not take into account the market valuation of cooperatives' assets, and do not link it to the position of individual members, preferring instead to use other accounting measures, such as book value or net asset value (NAV). In these cases, members' financial positions are valued and repaid at their nominal value (plus interest), as in Mondragon's system of internal capital accounts (Ellerman 1986). This solution has the advantage of referring to true accounting records such as the balance sheet, economic statement, and cash flow statement that do not suffer from high valuation volatility in financial markets. However, it may not provide a realistic representation of the real value of these assets in terms of expected earnings.

In systems similar to Mondragon's, members accumulate their rebates annually in internal capital accounts proportional to the dimension of realized profits (Whyte and Whyte 1991; Morrison 1989). Rebates are automatically capitalized to increase the net worth of the enterprise and to finance investment projects (Tortia 2007). As a general rule, capitalized rebates cannot be cashed out by members before the termination of the membership status, i.e., before voluntary resignation, dismissal, or retirement (Ellerman 1986).

As a different example, it is possible to consider ESOPs (Employee Stock Ownership Plans; Kelso and Adler 1958). While ESOPs have been common in the United States since the pension fund reform of 1974 (ERISA, Employee Retirement Income Security Act, which included defined-benefit plans and defined-contribution plans, such as 401(k) plans, 403(b) plans, and profit-sharing plans), ESOP-type financial participation is much rarer in Europe. A recent reform project of the Slovenian government, in cooperation with the Institute for Economic Democracy in Ljubljana, has introduced an ESOP-like financial participation scheme that mixes the Mondragón internal capital account system and the U.S. ESOP scheme (Ellerman et al. 2022a, 2022b). This proposal foresees the transformation of capitalist companies into worker cooperatives based on the allocation of company shares to workers and their payment to investor-owners according to the NAV valuation. The workers' shares would be held in a trust fund and would not be attributed to members until the debt issued to purchase the company is repaid (the trust fund would serve as collateral for the purchase). Once the transition process is completed, members' shares would not be exchanged in the market, but their value would increase with realized profits (not expected future profits) in the form of rebates or year-end dividends.

In contrast to the Mondragón system and the U.S. ESOP scheme, in Slovenian ESOPs the distribution of accumulated value in individual accounts can be collected by members before retirement, based on a predetermined schedule or rollover mechanism, which would

depend on the amount of net profits earned by the cooperative, the amount of reinvestment of realized profits needed to finance investment programs, and the speed with which new incoming members are able to replace the amounts of financial interest paid out to incumbent members. This way, members' financial positions could be partially liquidated, usually within a few years (Ellerman et al. 2022a, 2022b). While this mechanism can reduce or eliminate the problem of undercapitalization in some cases, it may not be able to fully account for the future economic performance of the organization and may not provide sufficient guidance for members to invest and innovate because it is based on the accumulation of realized profits, not prospective profit (Tortia 2007, 2021; Galor 2015; Galor and Sofer 2019).

2.3. U.S. ESOPs and the Presence of a Market for Employee Equity Shares

In systems where employee shares can be traded based on supply and demand, such as ESOPs in the United States, there is a market valuation for these securities. In the U.S., employee shares in ESOP companies can be bought back by the company on the market or issued anew and then awarded to workers. ESOP shares are held in a trust, which collectively represents workers at the company's shareholder meetings. Rather than replacing wages, employee shares are an addition to fixed wages in compensation packages in the form of a supplementary private pension plan. ESOP shares are redeemed at the current market price at the time of retirement (partially also at the time of voluntary resignation; Rosen 2023). This form of financial participation has shown important potential in increasing labor productivity and competitiveness ESOP companies. Several empirical studies measure an increase in labor productivity of close to 5 percent or more in the presence of an ESOP plan (Rosen and Quarrey 1987; Kumbhakar and Dunbar 1993; Blasi et al. 2013; O'Boyle et al. 2016; Kruse 2022) and also high firm longevity and job stability in ESOP firms compared to non-ESOP firms (Blair et al. 2000). In the United States, although in most cases exchanges take place on private markets, a significant number of companies that have adopted an ESOP (about a thousand) are publicly traded. In ESOP plans, holding employees' shares in a trust and limiting or blocking their capacity to sell (in some cases temporarily, in others permanently) eliminates the risk of workers' shares being sold to outside investors. Similar patrimonial solutions (holding employee shares in a trust) are followed by many (not all) U.K. and U.S. employee-owned companies, although these solutions can vary substantially from company to company (e.g., in some cases shares are redeemable, in others not; Erdal 2012). Among the best-known cases of employee-owned companies is the John Lewis Partnership in England, which began the process of converting to employee co-ownership in 1929 and completed the conversion in 1950 (Storey et al. 2014; Salaman and Storey 2016).

As the percentage of total shares held by employees increases, there is less room for the establishment of a real market for the company's shares due to the thinness of these markets, in which the financial and strategic decisions of employees prevail over outside investors. When the percentage of shares held by workers is high enough to control the company, market share prices may be an imperfect and distorted signal of company's value (Ang 1992). For this reason, the share price of ESOPs is usually defined by specialized professionals using various valuation methods, such as capitalization by discounted cash flows, EBITDA multiples, and the NAV (net asset value).

2.4. Example of Cooperative Enterprises Exchanging Shares on the Market

To the author's knowledge, examples of cooperatives selling tradable shares to outside investors are rare and none are found among worker cooperatives. However, a few recorded examples of the issuance of shares or share-like securities are found among other types of cooperatives, particularly financial, consumer, and producer cooperatives and also in some large worker cooperatives. At the theoretical level, Mikami (2016) discusses introducing transferable shares in the main types of cooperatives, workers, producers, and consumers, proposing specific changes to corporate law that would allow cooperatives to exchange shares in a manner similar to investor-owned companies. Under this approach, the market for cooperative shares is limited to members. Ridley-Duff (2009), on the other hand, explores the trade-off between collective reserves and individual ownership in worker cooperatives and discusses the possible impact of individual ownership, without delving into the issue of shares marketability (Cfr. also Tortia 2007, 2021).

Other theoretical contributions deal with ownership and capital structure in specific cooperative types, especially agricultural producer cooperatives, a very common phenomenon in both developed and developing countries. Chaddad and Cook (2004) discuss the so-called "New Generation Cooperatives" (NGCs) and propose the introduction of tradable shares. The authors outline a continuum from traditional cooperatives that capitalize their self-financing (positive residuals) through collective reserves and members' non-tradable shares to classic capitalist corporations owned by investors, which trade most of their shares on the market. NGCs are characterized by transferability of residual rights, providing liquidity and capital appreciation through secondary market valuation. Ownership rights are limited to patron members, membership is closed, and members are required to make initial investments in delivery rights in proportion to patronage. With the same contribution, the model of "Investor-Share Cooperatives" can acquire third-party equity capital without converting to an IOF. It usually issues separate classes of shares in addition to traditional member ownership rights. Investor shares can bundle different ownership rights in terms of return, risk bearing, control, redeemability, and transferability and include preferred shares, non-voting common shares, and participation certificates. In the remainder of the paper, a proposal similar to that for investor-share cooperatives will be made in the case of worker cooperatives, contemplating the same type of shares for both worker members with control rights and non-member investors.

Regarding empirical contributions, Fulton and Larson (2009) discuss the introduction of marketable shares in the Saskatchewan Wheat Pool (SWP), a Canadian grain marketing and input supply cooperative, in 1996. This capital restructuring led to financial difficulties in the early 2000s and complete demutualization in 2007. The SWP converted its members' capital into non-voting common shares, which were first offered to members and then traded on the Toronto Stock Exchange. Ordinary shares held by outside investors had no voting rights, while voting common shares (A Shares) held by cooperative members were not transferable and not appreciable (Chaddad and Cook 2004). The roots of the negative outcome of capital conversion have been identified in agency problems combined with management overconfidence and hubris (Fulton and Larson 2009). It should be noted, however, that in the SWP, members' A shares were not transferable and were not appreciable. In other words, the A-shares had no market valuation that took into account expected future earnings, and could not be sold to outside investors, giving members little incentive to monitor managers (reducing agency costs) and implement industrial strategies effectively.

Among worker cooperatives, Eroski is a large multi-stakeholder cooperative in which both employees and customers are members. It is part of the Mondragón group of worker cooperatives and the largest Spanish-owned retail chain in Spain. In 2009, Eroski had 48,000 employees, 2367 stores in Spain and abroad, 661,000 members, and sales of nearly EUR 8500 million. To increase its capitalization, between 2002 and 2007 it issued 660 million euros in perpetual non-redeemable bonds, which paid a fixed dividend. These bonds were listed on the Spanish stock exchange. The price of perpetual bonds can fluctuate widely, similar to the price of stocks, due to the strong influence of long-term earnings expectations. After the 2009 economic crisis, the price of Eroski perpetual bonds fell significantly, losing more than 50 percent of their par value. The economic and financial crisis forced Eroski to restructure, and the perpetual bonds were repurchased in 2014 with a total loss to investors of 30 percent of their face value, ending this financial experiment, which, however, allowed Eorski to survive a difficult decade (EFE: Agro 2014; Medina-Albaladejo 2017).

Finally, Volksbanks are credit institutions established in the form of joint-stock cooperative societies, following the tradition started by Franz Hermann Schulze-Delitzsch in Germany in 1850. *Banca Popolare di Milano* (BPM), one of Italy's largest Volksbanks, was listed on the Italian stock exchange in May 1994 and continued to be a cooperative bank issuing marketable shares until 2016. Being a cooperative bank, only members, who owned varying amounts of shares in the bank, had voting rights under the "one member, one vote" mutualistic rule. Following the financial difficulties caused by the 2011 Italian sovereign debt crisis, in 2015 the Italian government passed a law reforming popular banks (d.l. 3/15). The largest *banche popolari* (with a net worth of more than EUR 8 billion) had to demutualize and convert to investor ownership. In March 2016, BPM merged with Banco Popolare to form Banco BPM, an investor-owned commercial bank (Romani 2005).

3. A New Proposal for Cooperative Shares in Worker Cooperatives

A new proposal regarding the possibility and implementation of new forms of share ownership for worker members of worker cooperatives starts, at the theoretical level, from the agency relationship between employer and employee in capitalist companies. The agency model proposed by Jensen and Meckling in 1976 stipulates that the relationship between shareholder owners and managers is characterized by conflicting interests. By extension, the existence of conflicting interests can also be easily demonstrated in the employer/employee relationship in capitalist companies because of the inverse relationship between wages, which are a cost, and the net earnings of the firm. As wages increase, profits decrease, and vice versa (cfr. Sraffa 1960). The contractual relationship between employer and employee requires the latter to agree to perform the production operations specified in the contract in exchange for a fixed wage. The employee will then choose his or her optimal level of effort based on the labor income (wage) received and the structure of economic incentives and penalties for noncompliance with the contract (Prendergast 1999; Handy 2022; Angus 2023). In the presence of information asymmetries, the employer/employee agency relationship gives rise to agency costs, i.e., costs of controlling and incentivizing employees' work effort. These costs can be reduced by introducing appropriate monetary incentives and control mechanisms, but never eliminated, resulting in a second-best solution that deviates from the socially optimal solution (Prendergast 1999; Tortia 2022). This contract structure is characterized by wage rigidity, which is a dominant and stable feature of all capitalist market economies and can exacerbate business cycle fluctuations at the macro level (Keynes 1936; Weitzman 1986; Screpanti 2001; Albanese et al. 2019).

This paper applies the same tools and arguments that the agency literature has been applying to the conflict of interest and agency costs in the capitalist corporation (both between managers and shareholders and between employers and employees; Prendergast 1999) to address the serious negative impact that agency costs can have on the efficiency of the relation between worker members as investor-owners and external, non-member investors in worker cooperatives. The objective is to align, economically and financially, the objectives of worker members with those of non-member investors.

In a dividend-based labor remuneration scheme, members' remuneration can be equated to that of the shares held by non-member investors. When the remuneration of the shares owned by the two categories is exactly the same, the contrast of interests between worker members and investors can be limited or eliminated, since both will have the dividends they receive and the value of the shares they own as their dominant objective. The possibility of this type of alignment had already been pointed out in the best-known work of Jensen and Meckling (1976), in which the authors hypothesized that competing interests between shareholder owners and managers or directors can be reduced by introducing appropriate financial incentives—e.g., shares and/or stock options—for top executives (cf. Berle and Means 1932, on the separation of ownership and control in capitalist corporations), as these instruments can encourage managers' focus on performance, productivity, and profitability rather than on their own income and leisure (Jensen and Meckling 1976; Jensen and Smith 1985; Hill and Snell 2017).

3.1. The Introduction of Worker Shareholding

In purely economic terms, consistent with Meade's (1972, 1986, 1989) work on the capital/labor partnership, labor services are a flow, while human capital is a stock that grows with work experience, training, and investment in education, while it depreciates with age (Schultz 1961; Becker 1993). Similarly, financial capital is a stock that is remunerated by a stream of dividends or net income paid-out (Lintner 1962; Farsio et al. 2004; Skinner and Soltes 2009). Members of worker cooperatives invest their human capital and are remunerated in the same way as an investor. In this sense, worker members can be called "internal investors", while investors who only contribute financial capital without using their labor services in the cooperative should be defined as "external investors" or "financial investors".

To align the financial objectives of internal and external investors and eliminate potential conflicts of interest, in the proposal of this paper both categories hold the same financial security, which can be defined "common cooperative stock". However, the right to elect representatives to organizational bodies is considered a personal, inalienable, and non-duplicable right that belongs exclusively to worker members on a per capita basis, rather than according to the number of shares held (Cheney et al. 2014). In addition to conceiving of rights as personal and inalienable, this solution is also functional in avoiding the possible reemergence of conflicting interests between two different stakeholder groups within governing bodies (Ellerman 2005, 2016, 2021).

The difference between the cooperative's common stock held by working members and non-member investors can be likened to Class A and Class B shares in capitalist corporations, which means that the issuance and allotment of Class A shares would be limited to working members, while Class B shares would be issued to and saleable by non-member investors (Class A and Class B common shares in capitalist corporations may be limited to founders and/or managers). For example, specific issues of new shares (either as IPOs for Class B shares or as secondary offerings) may be restricted to Class A or Class B shares. Apart from limited allocations, outside investors would hold the same common shares as working members, featuring the same value and dividends. All classes of shares would have no voting rights, as voting rights would be given according to the "one member, one vote" rule.

In fact, corporate governance scholars and financial analysts increasingly view the issuance of non-voting shares as beneficial or non-harmful even for capitalist corporations, as non-voting shares can make corporate governance more efficient by lowering the cost of capital (reducing agency and transaction costs). They distribute voting power among informed shareholders who value their voting rights, while uninformed and "weakly motivated" shareholders need not vote (Lund 2019). In this paper, the issuance of non-voting common stock sold to outside investors helps increase the capitalization of worker cooperatives, which historically have often been plagued by underinvestment and under-

capitalization, production efficiency, and financial performance, because it forces voting members of the cooperative to implement more transparent and effective financial and production policies, with the goal of increasing the value of all classes of common stock.

Excessive concentration of share ownership in the hands of outside investors could lead to the formation of an unbalanced capital structure, with worker owners owning too small a share of the total, insufficient, for example, to provide adequate collateral to obtain financial support (loans and mortgages) from banking intermediaries (McCain 1977). For this reason, it may be advisable for worker members to hold a minimum percentage of the total share capital. Above this minimum, they may be allowed to sell their shares on the market, thus collecting the market valuation, a transaction that would be equivalent to an insider share sale (Jeng et al. 2003).

As already observed in existing worker cooperatives, members' dividend income is expected to be more volatile than wages. However, since the entire organizational system and financial structure is built around providing a stable livelihood and job security for worker members, these types of cooperatives are expected to make decisions to limit income volatility, in some cases at the cost of increasing share price volatility, and to establish labor income for their members first. Of course, the trade-off between risk and reward in financial markets requires that higher share price volatility is matched by higher dividend payouts, so members in cooperatives will have to strike a reasonable balance between labor income volatility and share price volatility (McCain 1977). On the other hand, it should also be considered that the cost structure in this type of organization is more flexible than that of capitalist enterprises and traditional cooperatives, as members' wage costs are absent, making it easier to stabilize income and jobs.

3.2. Market Valuation of Cooperative Shares

When introducing tradable shares, it is necessary to make explicit the mechanisms through which such securities would be issued and allocated to members, and how it would be possible to calculate their nominal value. This calculation can be, in the first instance, simple and made through an example. Assuming, for simplicity, that each worker member receives the same labor income and owns a single share of the cooperative capital, if the member's dividend in the net income of the cooperative is equal to x (for example, EUR 30 thousand per year) and the average shareholder dividend paid in the financial markets by companies with similar characteristics is equal to some percentage d of the market value of the shares (for example, an average of 10%), the presumed nominal value of the single member's financial stake is x/d (30,000/0.1 = EUR 300,000 in the example). Cooperative shares could be issued on the market at this nominal or par value. Fractional shares could be sold to outside investors at, say, 1/30,000 of their par value, or EUR 10. Obviously, market prices of shares can differ significantly from par value because share prices are influenced by a wide array of organizational and macroeconomic circumstances, primarily the expected future profitability of the company.

Although capitalist enterprises and cooperatives may deal with share valuation and marketability problems in partially different ways, the financial nature of these problems can be considered equivalent. The residual income of the cooperative would be allocated to all existing shares held by both members and outside investors following the same proportionality rule, since uncertainty about these quantitative dimensions would imply that outside investors could not be involved. In addition, cooperatives, like capitalist companies, can issue shares to outside investors at a premium or discount (Koller et al. 2020).

In the initial phase of issuing cooperative shares, before shares are traded on the market (private or public), specialized professionals and consulting companies such as the Baxi Partnership in Scotland (Erdal 2012) can calculate the nominal price, based on the

dimension of the dividends paid to members, the balance (assets, liabilities, and equity), the income and cash flow statements, and the average dimension of dividends paid or retained in similar companies in the same sector.

The price of shares held by both members and non-member investors can move in either direction for several reasons. Increases can occur either because of an increase in expected future profitability, which could cause the exchange price of shares to rise, even in the absence of an increase in income (dividends) paid to members, or because of an increase in income paid to members, when the net realized profits of the cooperative increase. In the latter case, other things being equal (e.g., average dividends in percentage points paid by stocks in the same industry), an increase in dividends paid to members and outside investors implies a proportional increase in the implied share price due to higher profitability. In the above example, an increase in labor income from EUR 30,000 to EUR 40,000 would imply the same proportional increase in dividends paid to outside investors and also a proportional increase in the implied value of shares held by members from EUR 300,000 to EUR 400,000. Since the vast majority of cooperative enterprises issuing shares to non-member investors will not be quoted on a stock exchange, in most cases, exchanges and valuations will take place in private equity markets. Being controlled by worker members, these organizations are expected to set first the value of dividend-based labor incomes to provide for members' livelihood.

Attribution of shares to incoming and quitting members. Shares can be attributed to incoming members for free, at a discount, or at full price depending on the internal policy of the cooperative, the financial constraints undergone by the new member, his/her abilities (human capital), and the financial health of the cooperative (e.g., owned cash liquidity and reserves of retained earnings). When a market price for cooperative shares exists, shares can be attributed (and their number calculated) or sold to incoming members at their market price or at their par value depending on the cooperative's policy (share market price can be higher or lower than par value). Shares can be allocated to new entrants through new issues, market repurchases, and allocation of existing treasury stock. This type of common cooperative shares would be nonredeemable, like all common shares, so as not to put financial pressure on the organization, especially in times of crisis (Tortia 2018, 2021).

Setting the nominal price of the cooperative's shares is very important in the absence of a market price, since, for example, it would signal to incoming members the presumed value of their financial stake in the cooperative. If a new member receives entry income equal to, say, 50 percent of the income received by existing members, he or she will be allocated exactly 50 percent of the number of shares held by existing members, valued at their par price.

Quitting members (resigning or retiring members) would be registered as external shareholders, would receive the same dividends as incumbents, and would be able to sell their shares on the secondary market to other (private and/or institutional) investors or also to the cooperative itself. In this way, incumbent members would have the greatest incentive to increase the competitive potential and market value of the cooperative and recruit new, highly qualified members since they would only sell their shares when they resign or retire based on market assessments of future profitability at exit. The cooperative can increase the amount of dividend income paid out to members at will, but this type of transaction will likewise increase the dividends distributed to outside investors. This may reduce the market value of cooperative shares due to increased dividend outflows.

Attribution of shares to incumbent members and outside investors. Within the organization, shares can be allocated to current members by issuing them from scratch, buybacks, or treasury stock. As is already the practice in capitalist companies, repurchased or new shares can be used as part of an additional compensation package for members or to implement

stock option plans (Song 2002; Bens et al. 2003). The issuance of new shares to members (both new and incumbent) may cause outside investors to sell their holdings because of an increase in the total number of shares outstanding, or dilution, to avoid capital losses. For this reason, the issuance of new shares must be carefully considered and usually occurs when there are significant increases in the firm's productivity and profitability prospects, an eventuality that, when it materializes, can counteract or reverse dilution and share price declines (e.g., when the returns to scale of the technology used by the firm are increasing; Larrain and Urzúa 2013).

Shares issued to outside investors can be sold in private equity markets for unlisted companies but, in principle, can also be publicly traded. Since the market price of shares issued by listed and unlisted companies is very difficult for investors to predict, specialized intermediaries such as venture capitalists, business angels, or crowdfunding can finance new and emerging cooperatives that might also seek to move from private to public status through an IPO (Capizzi and Carluccio 2016; Blackburn et al. 2017; Bonini and Capizzi 2019).

3.3. Disparity in Share Value Due to Voting Right

The proposal in this paper features a productive organization that is a pure workers' cooperative, in which share ownership does not coincide with voting rights also in the case of worker members, not just outside investors. Share ownership, on the other hand, would be a kind of pure financial ownership, linked only qualitatively but not quantitatively to the personal rights of control.

The introduction of different types of shares (e.g., voting and non-voting; common and preferred) for working members and outside investors would imply different pricing mechanisms and lead investors, who are not in control of the organization, to be skeptical about buying shares in the cooperative, fearing that the price of the shares they buy is unduly skewed downward and thus perceiving a negative price difference as a violation of distributive equity. If investors gave up buying shares in the cooperative, underinvestment and undercapitalization would likely result again (Vanek 1970, 1977). Indeed, voting shares held by corporate owners are generally traded at a higher price than non-voting shares held by investors because voting rights confer additional benefits, such as influence over corporate governance and strategic decisions. The need to align the interests and goals of members and investors as closely as possible and to avoid the danger of undervaluing investors' shares confirms the requirement that the shares held by worker members and external investors be exactly the same.

Equalizing the characteristics of members' and investors' shares would force members to be as transparent and effective as possible in both the management of the production organization and their financial policies (e.g., the issuing of new shares, stock option plans, etc.), as non-transparent and ineffective policies would inevitably cause investors not to invest or to sell, lowering the share price for both categories. On the other hand, if members and investors owned different types of shares, there would be no true market for the members' shares, as they could only sell their holdings to other members (Mikami 2016). The absence of a true open market for common shares would cripple and distort financial incentives, eventually damaging production processes as well, as inefficient valuation of shares would result in the inability to pursue optimal investment plans. This is one of the main reasons why the share issue experiment initiated in 1996 by the Saskatchewan Wheat Pool probably proved deficient and failed as early as 2003, causing demutualization in 2007. Outside investors owned a different type of shares than the voting shares held by members, which were non-transferable and non-appreciable (Chaddad and Cook 2004; Fulton and Larson 2009).

Worker members may be required by law to own a minimum percentage of the total net worth (equity) of the organization, to create capital stability and collateral guarantees, and to avoid the danger of violating the law of increasing risk (McCain 1977), which in this case implies that insufficient equity held by members induces the market to demand higher average returns, i.e., a lower market value of the shares. Special emissions of new shares and stock options could be granted only to members, in much the same way that they are granted to particular types of employees (e.g., top executives) in capitalist corporations.

This capital structure leaves open the possibility of issuing other equity and non-equity asset classes. For example, preferred shares and perpetual bonds. As it is well known, preferred shares are part of equity, i.e., they are not redeemable but issued at a par value that is usually not much different from their market value because they pay a (non-mandatory) dividend to shareholders and do not have voting rights. They are more similar to bonds than to actual equity. Perpetual bonds can also be issued, as in the case of Eroski in Spain. Perpetual bonds resemble preferred shares because they have no voting rights, are not redeemable, and pay a fixed yield. However, they are not part of equity because they can be repurchased at any time by the issuing entity, and payment of a dividend is mandatory.

4. Organizational Issues: Labor Productivity, Free-Riding, Workers' Livelihoods, and Dismissal

The introduction of cooperative shares would have important organizational consequences, mainly having to do members' livelihood, including the risk of free-riding in the labor process and layoffs, through its ability to achieve adequate labor productivity. These issues are addressed in turn in this section.

4.1. The Organization of Work, Motivation, and the Problem of Free-Riding

When hierarchy is replaced by horizontal coordination and self-organization, so-called "collective action failures", are likely to be generated, as in the foundational contributions of Commons (1950), Olson (1965), Hardin (1968), and Ostrom (1990). In a mutualistic organizational structure, managerial control and entrepreneurial action can be considered collective (Cook and Plunkett 2006; Bijman and Doorneweert 2010; Lomuscio 2024), while decision-making power is delegated from members to their representatives (Ellerman 2005, 2016).

Severe criticism has been raised against the possibility of achieving adequate productive efficiency through collective self-organization of labor because of collective action failures, especially free-riding. Workers would reduce average work effort in an attempt to opportunistically take advantage of the work effort of other workers, since effort is a cost that the worker wants to minimize, while joint production is a (semi-)public good that can be enjoyed by all workers in the collective enterprise. Indeed, the value added produced and capital in worker cooperatives can be considered commons in that they are rivals, but not excludable (Ostrom 1990; Tortia 2018). In the presence of nonexcludability and rivalry, free-riding leads to sub-optimal production efficiency that negatively deviates from the social optimum for the organization (Alchian and Demsetz 1972; cfr. Bowles and Gintis 1993; Putterman 1988 for a favorable response to democratic firms). In capitalist enterprises, instead, the owner's goal is to maximize the enterprise net profitability or market valuation. The owner, therefore, will put in place those control mechanisms over the labor process that can ensure an optimal (efficient) allocation of work effort, resulting in the second-best solution net of agency costs (Alchian and Demsetz 1972; Jensen and Meckling 1976).

Because free-riding always afflicts collective action and cooperation, it has been in the spotlight of theoretical and empirical research since Alchian and Demsetz (1972) and Jensen and Meckling (1979) and other well-known contributors (see Holmstrom 1982). Advocates of worker cooperatives have defended the possibility of democracy in the workplace,

seeking to explain how free-riding can be overcome or appropriately limited (e.g., Ben-Ner 1988; Ben-Ner and Jones 1995; Putterman 1988; Bowles and Gintis 1993).

Real-world cooperatives have developed control tools and monitoring processes that can control free-riding, ranging from peer pressure to financial incentives to involvement in decision-making processes to sanctions proportional to the severity of the breaches perpetrated by free-riding members (Ostrom 1990; Malleson 2013; Tortia 2022). The introduction of financial incentives, such as a market for member rights, has been shown to effectively combat free-riding, as in plywood worker cooperatives in the U.S. Pacific Northwest (Craig and Pencavel 1992, 1994; Pencavel 2001). The presence of a market for membership rights in these cooperatives implied that members could realize the market value of productivity gains in terms of higher prices for their membership positions, supporting the achievement of high levels of labor productivity (Craig and Pencavel 1992, 1994; Pencavel 2001; Dow 1986, 2003; Tortia 2022).

Some recent empirical results, however, seem to confirm the existence of fairly relevant phenomena of productivity reduction and free-riding concerning members of worker cooperatives in Uruguay (Blanchard et al. 2024). The evidence on wages lower than in capitalist firms also seems to point in the same direction, although free-riding is certainly not the only determinant of negative wage differentials between capitalistic enterprises and cooperatives. Worker member characteristics, such as educational level, and the level of capitalization of the firm, also play a key role (Pencavel et al. 2006; Clemente et al. 2012).

In this study, it is hypothesized that financial participation through member ownership of cooperative shares can exert a direct positive impact on job performance, including helping to combat free-riding in team production, as confirmed by empirical studies on the effects of the ESOPs on labor productivity in the United States (Blasi et al. 2013; Kruse 2022). The proposed ownership and organizational model sees financial incentives as directly linked to members' labor income as a means of workers' livelihood. These incentives are the main tools that can be effective in combating the free-riding problem, as financial losses in terms of cooperative share valuation resulting from members' opportunistic behavior directly affect members' livelihoods. Furthermore, the cooperative would be forced to implement effective control mechanisms, the specifics of which should be delegated to the decisions of individual organizations (e.g., increasing supervisory staff, increasing or reducing smart working, implementing workplace control, etc.).

The backbone of this approach lies in the conjugation of the market valuation of members' financial holdings and the involvement processes that characterize the governance and management of worker cooperatives (Rosen and Quarrey 1987; Winther and Marens 1997; Logue et al. 2001). The feedback mechanism to reduce or eliminate free-riding would occur through involvement in cooperative governance and collective recognition that behavioral outcomes in the production process are directly reflected in the value of financial holdings, mediated by collective productivity. This feedback mechanism has a strong cognitive and motivational dimension that depends on collective impacts within a unifying organizational framework, which can be termed "proactive collective rationality." In turn, maintaining or strengthening workers' intrinsic motivation to work will likely foster organizational resilience and reduce the need for supervision.

The role of collective rationality. Collective rationality refers to the rational ways of solving collective action problems and the framing effects induced by coordination mechanisms in governance. Framing is understood as a cognitive process in which individuals identify relevant organizational dimensions that support their decision-making by linking motivations (goals) and expected behavioral outcomes. Bacharach (2006) introduces the concept of "we frame", as a collective process that supports the pursuit of common goals. For Schoemaker and Russo (2016), rationality as a collective phenomenon requires fram-

ing, that is, anchoring decision-making to simple cues in the organizational environment. Framing is also crucial in the production of shared knowledge in terms of shared conceptualization that facilitates sensemaking, understanding, and collective action (Hecker 2012). Socialization processes within organizations enable concerted practices and a partial overcoming of bounded rationality, as individuals can see themselves as part of a common venture in which they can coordinate cognitively and express common productive motivations (Shulman and Carey 1984; Lindenberg and Foss 2011). In the evolutionary literature, the collective side of rationality dynamically emerges and crystallizes from the development of rules and routines in inter- and intra-organizational interactions through institutionalization or "structuration" (DiMaggio and Powell 1983).

Free-riding, productivity, and intrinsic motivation. The cognitive aspects of organizational processes and financial incentives must be combined with worker motivation, especially intrinsic motivation. Classical studies have shown that control mechanisms, punishment, and extrinsic incentives can crowd out intrinsic motivation to work (Frey 1997; Frey and Jegen 2001). In the approach followed in this paper, financial participation can increase the economic performance of worker cooperatives, but only when combined with member involvement in a nonhierarchical organizational environment and governance, since participation and commitment, which support intrinsic motivation to work, maintain a crucial role in achieving higher productivity and long-term resilience (Tortia et al. 2021).

Herzberg et al. (1959) and Herzberg (1987) introduced the two-factor model to explain workplace motivation and job satisfaction. Hygiene factors, such as salary, supervision, and job security, do not lead to satisfaction but prevent dissatisfaction. Motivating factors refer to intrinsic motivators, such as the job itself and recognition. In worker-managed systems, hygiene factors such as procedural fairness, governance, and cooperative structures are critical. However, intrinsic motivators such as autonomy, decision-making involvement, and transparency in governance often drive worker engagement, as democratic decisionmaking processes improve procedural fairness, fostering job satisfaction and a sense of inclusion and trust (Tortia 2008; Tortia et al. 2021). Monitoring is necessary for efficiency, but it should be in line with inclusive cooperative governance, as too much monitoring can undermine intrinsic motivation. When applied to democratic governance, Herzberg's (1987) two-factor model shows that worker empowerment and accountability can foster creativity and innovation (Sacchetti 2015). On the other hand, hygiene factors such as clear communication of rules, predictable schedules, and a sense of job security are essential to minimize dissatisfaction and transaction costs. It can be hypothesized that inclusive, noncontrolling governance, characterized by low degrees of monitoring and supervision, may foster or otherwise not harm intrinsic motivation and play a positive mediating role between financial participation and work performance. Instead, the relationship between monitoring and job performance is likely complex and nonlinear. The recent contribution of Mugerman et al. (2024) shows that too little monitoring may lead to detrimental effects on collective productivity due to the prevalence of free-riding, while, in contrast, too tight supervision does not lead to higher productivity but to disengagement and crowding out due to the displacement of intrinsic motivation. In fact, monitoring and supervision appear to be particularly low in worker cooperatives, as shown by Bartlett et al. (1992), who found smaller supervisory staff in Italian industrial cooperatives, as compared to similar capitalistic firms in the same sectors (cf. also Jones and Svejnar 1985).

Recent HRM results from Chinese data show that financial stress (which is an indicator of financial involvement) is positively related to job performance, while work engagement, which results from worker involvement in decision making, mediates the positive relationship between financial stress and job performance. Emotional exhaustion, which can be a negative indicator of intrinsic motivation, negatively moderates the mediation relationship between financial stress, work engagement, and job performance. In other words, low levels of intrinsic motivation weaken the positive relationship between financial participation and performance through lower engagement (Wei et al. 2024; Miao et al. 2024). The positive impact of financial incentives on productivity and motivation works through cooperative governance to reduce the need for monitoring mechanisms (supervision and sanctions), which would generate additional agency costs, thus improving overall organizational efficiency (Barnard 1938).

4.2. Risks to Worker Livelihoods in Dividend-Based Compensation

In dividend-based compensation, workers' income is directly linked to the residual earnings of the enterprise. If earnings are too low, workers' livelihoods are at risk. This problem is common to worker cooperatives and capitalist enterprises alike, as the former may not be able to pay sufficient incomes, while the latter may be prone to resort to layoffs when going through economic or financial difficulties. In the scheme of the present work, the absence of wage labor implies the absence of labor cost, which would reduce total costs substantially, weakening the constraints that the cooperative must meet to remain economically solvent and viable, compared to other enterprise forms (Weitzman 1986; Meade 1989). Dividends as labor income would certainly be more volatile than wages, but on average they should be higher, although this specific question (whether dividend income is on average higher or lower than wages) will require further analysis and empirical testing. Certainly, dividend-based compensation serves the same economic function as wages in securing livelihoods.

In addition, this type of cooperative is not prevented from establishing reserves, either divisible or indivisible (collective), which can be used to supplement members' incomes in times of crisis. Losses can be absorbed through retained earnings, debt financing, or collective reserves, as in more traditional cooperative forms (for example, all Italian cooperatives, not only worker cooperatives, are required by law to reinvest at least 30% of their net earnings in indivisibles reserves). The introduction of collective reserves can have a significant negative impact on financial incentives and share value of the cooperative. However, indivisible reserves have the advantage of stabilizing capital ownership and creating an insurance fund that ensures job stability during economic crises, something of central importance in worker cooperatives (Navarra 2011, 2016; Tortia 2018, 2021). In addition, this type of organization may not increase paid-out labor income too much, lest it increase the risk of bankruptcy and job loss (Miyazaki and Neary 1983; Craig and Pencavel 1995; Burdín and Dean 2012). Indeed, it has been repeatedly shown that worker cooperatives are more risk-averse than capitalist enterprises in their investment policies, and one of the main reasons lies precisely in the need to avoid job losses for members (Hansmann 1996; Dow 2003; Holmstrom 1982; Ben-Ner and Jones 1995; Bonin et al. 1993; Craig and Pencavel 1995; Pérotin 2013).

In case of economic problems and financial difficulties, the cooperative form studied in this paper would have to pay at least a minimum income to all members to support their livelihood. In many circumstances, there would be no economic rationale for laying off members, even during an economic crisis, since layoffs of members do not reduce the dividends that must be paid to investors (laid-off members would become external investors).

Crises can be absorbed in the short to medium term through the use of debt and reserves, but not in the medium to long term. If the cooperative is no longer economically or financially viable, it must be liquidated or acquired by stronger organizations. During economic crises, skilled workers are likely to leave first because of better outside options in the labor market, although leaving the cooperative implies giving up quasi-rents and
other returns on specific human capital investments made in the past (Williamson 2000). When the crisis becomes very severe and the income guaranteed by the cooperative falls below some critical threshold (the participation constraint in the external labor market), the economic rationale for this type of organization evaporates as well. Finally, when marketable shares perform poorly and the economic conditions of the organization are negative, members may choose to demutualize and return to the capitalist form, or buy back the cooperative's shares at much lower prices to sustain their valuation and restart the capitalization process.

4.3. The Dismissal of Worker Members

While in investor-owned enterprises dismissal for economic reasons is considered a routine and necessary practice due to wage rigidity (the enterprise is forced to dismiss workers with fixed incomes, when worsening profitability may harm its development prospects or engender bankruptcy), several theoretical and empirical contributions have shown that cooperative enterprises, not only worker cooperatives, can be able to significantly reduce dismissals compared to their capitalist counterparts, even during negative economic phases (Weitzman 1986; Meade 1989). The main reason is that cooperatives are created to provide a stable flow of goods and services to their members, in a logic similar to that of clubs, and thus to the production of collective goods (Ben-Ner and van Hoomissen 1991; Pérotin 2013). Since the cooperative will tend to keep the supply of goods and services to members stable even during a crisis, employment in the cooperative will also be stable (Borzaga et al. 2022).

The tendency of worker cooperatives not to lay off workers even during a crisis can also be explained by the provision of employment services to members, as worker cooperatives will continue to provide employment opportunities on better terms than in the open labor market and save jobs even in adverse economic conditions, that is, even when profitability is low or negative (Bartlett et al. 1992; Craig and Pencavel 1992, 1994; Pencavel et al. 2006; Berman and Berman 1989; Burdín and Dean 2009; Delbono and Reggiani 2013). During recessions, members will take a range of actions to minimize the likelihood of job losses, from reducing hourly wages and hours worked to using reserves to offset declines in demand and output (Miyazaki and Neary 1983; Craig and Pencavel 1993; Navarra 2011, 2016; Navarra and Tortia 2014; Albanese et al. 2019; Tortia 2022).

In the cooperative model presented in this paper, the nature and economic relevance of employment termination change substantially. Even if the associative relationship between the member and the cooperative is terminated through voluntary resignation or involuntary layoff, the ex-member becomes an outside investor and remains in possession of his or her financial stake, which continues to receive the corresponding dividends. Termination of membership eliminates membership rights, including any additional benefits, such as receiving additional shares or stock option plans. On the other hand, from the organization's perspective, in terms of existing financial positions, layoffs result in no cost gain, unlike in capitalist enterprises. In other words, fixed wages as an economic rationale for laying off workers would be overcome. Reducing layoffs would also forestall negative macroeconomic implications, such as decreases in aggregate consumption (Pérotin 2013; Tortia 2022).

5. Conclusions

As already hypothesized by Ward in his 1958 foundational model of the labor-managed firm, the present paper hypothesizes the creation of worker cooperatives in which members' earned income is not a fixed contractual wage, as in an employer/employee agency relation-

ship, but a dividend-based remuneration scheme consisting of shares in the organization's net income in the absence of wage costs.

Paying dividends instead of wages would uniquely link the labor income of cooperative members to the par value of their financial participation in the enterprise's capital. In the absence of a capital market, the par value of the cooperative's shares corresponds to the value of the members' financial interest in the enterprise's net worth. If there is a capital market, instead, shares can be sold on a private equity market or even on a public stock exchange. The cooperative's shares could be issued and traded on the same terms of value and remuneration not only for cooperative members but also for non-member outside investors.

A number of existing examples of cooperative capital structures or employee ownership that include the issuance of shares or other share-like securities (e.g., perpetual bonds) were presented and discussed, covering not only worker cooperatives and employee ownership but also other cooperative forms (production, consumer and credit cooperatives). In worker-owned enterprises, for example, members own at least part of the company's capital, although there may or may not be shares and capital markets. The case of Mondragon cooperatives, the ESOP scheme in the United States, and worker cooperatives in the U.S. Pacific Northwest were brought up for discussion, showing the attempts that have been made to date to introduce employee and member financial participation in equity in a structured way. The presence of a market for member rights and ESOPs have been identified as among the most refined examples of tradable worker ownership, along with employee ownership in the John Lewis Partnership tradition.

Finally, it was pointed out that in the worker-owned cooperative model proposed in this paper, dividends as members' labor income and also the value of cooperative shares could have a relatively stable and increasing trend over time, in line with members' livelihood needs and their productivity improvements. They have also been found to be effective in counteracting free-riding and other collective action problems in the entrepreneurial action. On the other hand, cooperative governance has been shown to have the potential to increase productivity and performance by sustaining intrinsic work motivation. Only in the event of economic or financial difficulties in the enterprise would one expect labor income to undergo significant (downward) adjustments, while positive economic performance would lead to a steady increase in members' labor compensation.

Funding: This research received no external funding.

Data Availability Statement: No new data were created or analyzed in this study. Data sharing is not applicable to this article.

Acknowledgments: The author would like to thank Alberto Zevi, former director of the LegaCoop research center in Rome, for very important suggestions and precious comments on earlier versions of this article. The usual disclaimers apply.

Conflicts of Interest: The authors declare no conflict of interest.

References

Albanese, Marina, Cecilia Navarra, and Ermanno C. Tortia. 2019. Equilibrium Unemployment as a Worker Insurance Device: Wage Setting in Worker-Owned Enterprises. *Economia Politica: Journal of Analytical and Institutional Economics* 36: 653–71. [CrossRef]

Alchian, Armen A., and Harold Demsetz. 1972. Production, Information Costs, and Economic Organization. The American Economic Review 62: 777–95.

Ang, James S. 1992. On the Theory of Finance for Privately Held Firms. *Journal of Small Business Finance* 1: 185–203. [CrossRef] Angus, Ian. 2023. *The War Against the Commons: Dispossession and Resistance in the Making of Capitalism*. New York: Monthly Review. Bacharach, Michael. 2006. *Beyond Individual Choice: Teams and Frames in Game Theory*. Princeton: Princeton University Press. Barnard, Chester. 1938. *The Functions of the Executive*. Cambridge, MA: Harvard University Press. Bartlett, Will, John Cable, Saul Estrin, Derek C. Jones, and Stephen C. Smith. 1992. Labor-Managed Cooperatives and Private Firms in North Central Italy: An Empirical Comparison. ILR Review 46: 103–18. [CrossRef]

Becker, Gary S. 1993. Human Capital, 3rd ed. Chicago: Chicago UP.

- Ben-Ner, Avner. 1988. The Life Cycle of Worker-Owned Firms in Market Economies. *Journal of Economic Behaviour and Organization* 10: 287–313. [CrossRef]
- Ben-Ner, Avner, and Derek C. Jones. 1995. Employee Participation, Ownership, and Productivity: A Theoretical Framework. Industrial Relations 34: 532–54. [CrossRef]
- Ben-Ner, Avner, and Theresa van Hoomissen. 1991. Nonprofit Organizations in the Mixed Economy: A Demand and Supply Analysis. Annals of Public and Cooperative Economics 62: 519–50. [CrossRef]
- Bens, Daniel A., Venky Nagar, Douglas J. Skinner, and M. H. Franco Wong. 2003. Employee Stock Options, EPS Dilution, and Stock Repurchases. *Journal of Accounting and Economics* 36: 51–90. [CrossRef]
- Berle, Adolf A., and Gardiner C. Means. 1932. The Modern Corporation and Private Property. New Brunswick: Transaction.
- Berman, Katrina V., and Matthew D. Berman. 1989. An Empirical Test of the Theory of the Labour-Managed Firm. *Journal of Comparative Economics* 13: 281–300. [CrossRef]
- Bijman, Jos, and Bart Doorneweert. 2010. Collective Entrepreneurship and the Producer-Owned Co-operative. *Journal of Co-operative Studies* 43: 5–16.
- Blair, Margaret M., Dougles L. Kruse, and J. Joseph Blasi. 2000. Employee Ownership: An Unstable Form or a Stabilizing Force? SSRN Electronic Journal. Available online: https://papers.ssrn.com/sol3/papers.cfm?abstract_id=142146 (accessed on 17 December 2024).
- Blanchard, Pablo, Gabriel Burdin, and Andrés Dean. 2024. Property Rights, Sick Pay and Effort Supply. *Leeds University Business School* WP. Available online: https://ssrn.com/abstract=4839474 (accessed on 22 October 2024).
- Blackburn, Robert, Dirk De Clercq, and Jarna Heinonen. 2017. *The SAGE Handbook of Small Business and Entrepreneurship*. Thousand Oaks: Sage.
- Blasi, Joseph, Douglas Kruse, and Dan Weltmann. 2013. Firm Survival and Performance in Privately Held ESOP Companies. Advances in the Economic Analysis of Participatory & Labor-Managed Firms 14: 109–24. [CrossRef]
- Bonin, John, Derek C. Jones, and Louis Putterman. 1993. Theoretical and empirical Studies of Producer Cooperatives: Will the Twain Ever Meet? *Journal of Economic Literature* 31: 1290–320.
- Bonini, Stefano, and Vincenzo Capizzi. 2019. The Role of Venture Capital in the Emerging Entrepreneurial Finance Ecosystem. Venture Capital. *International Journal of Entrepreneurial Finance* 21: 137–75. [CrossRef]
- Borzaga, Carlo, Chiara Carini, and Ermanno C. Tortia. 2022. Co-operative enterprise anti-cyclicality and the economic crisis: A comparative analysis of employment dynamics in Italy. *Annals of Public Cooperative Economics* 93: 551–77. [CrossRef]
- Bowles, Samuel, and Herbert Gintis. 1993. A Political and Economic Case for the Democratic Enterprise. *Economics and Philosophy* 9: 75–100. [CrossRef]
- Burdín, Gabriel, and Andrés Dean. 2009. New evidence on wages and employment in worker cooperatives compared with capitalist firms. *Journal of Comparative Economics* 37: 517–33. [CrossRef]
- Burdín, Gabriel, and Andrés Dean. 2012. Revisiting the Objectives of Worker-managed Firms: An Empirical Assessment. *Economic Systems* 36: 158–71. [CrossRef]
- Capizzi, Vincenzo, and Emanuele M. Carluccio. 2016. Competitive Frontiers in Equity Crowdfunding: The Role of Venture Capitalists and Business Angels in the Early-Stage Financing Industry. In *Crowdfunding for SMEs. A European Perspective*. Edited by Roberto Bottiglia and Flavio Pichler. London: Palgrave, pp. 117–54.
- Chaddad, Fabio R., and Michael L. Cook. 2004. Understanding New Cooperative Models: An Ownership-Control Rights Typology. *Review of Agricultural Economics* 26: 348–60. [CrossRef]
- Cheney, George, Iñaki Santa Cruz, Ana M. Peredo, and Elías Nazareno. 2014. Worker Cooperatives as an Organizational Alternative. *Organization* 21: 591–756. [CrossRef]
- Clemente, Jesús, Millán Diaz-Fonsea, Carmen Marcuello, and Marcos Sanso-Navarro. 2012. The Wage Gap Between Cooperative and Capitalist Firms: Evidence from Spain. *Annals of Public and Cooperative Economics* 83: 337–56. [CrossRef]
- Coase, Ronald H. 1937. The Nature of the Firm. *Economica* 4: 386–405. [CrossRef]
- Commons, John R. 1950. The Economics of Collective Action. New York: Macmillan.
- Cook, Michael L., and Brad Plunkett. 2006. Collective Entrepreneurship: An Emerging Phenomenon in Producer-Owned Organisations. Journal of Agricultural and Applied Economics 38: 421–28. [CrossRef]
- Craig, Ben, and John Pencavel. 1992. The Behaviour of Worker Cooperatives: The Plywood Companies of Pacific Northwest. *The American Economic Review* 82: 1083–105.
- Craig, Ben, and John Pencavel. 1993. The Objectives of Worker Cooperatives. Journal of Comparative Economics 17: 288–308. [CrossRef]
- Craig, Ben, and John Pencavel. 1994. The Empirical Performance of Orthodox Models of the Firm: Conventional Firms and Worker Cooperatives. *Journal of Political Economy* 102: 718–44. [CrossRef]

- Craig, Ben, and John Pencavel. 1995. Participation and Productivity: A Comparison of Worker Cooperatives and Conventional Firms in the Plywood Industry. *Brookings Papers on Economic Activity. Microeconomics* 1995: 121–74. [CrossRef]
- Delbono, Flavio, and Carlo Reggiani. 2013. Cooperative Firms and the Crisis: Evidence from some Italian Mixed Oligopolies. *Annals of Public and Cooperative Economics* 84: 383–97. [CrossRef]
- DiMaggio, Paul J., and Walter W. Powell. 1983. The Iron Cage Revisited: Institutional Isomorphism and Collective Rationality in Organizational Fields. *American Sociological Review* 48: 147–60. [CrossRef]
- Dow, Gregory K. 1986. Control Rights, Competitive Markets, and the Labor Management Debate. *Journal of Comparative Economics* 10: 48–61. [CrossRef]
- Dow, Gregory K. 2003. Governing the Firm. Workers' Control in Theory and Practice. Cambridge, MA: Cambridge UP.
- EFE: Agro. 2014. Eroski Cambia la Deuda Perpetua Por una Quita del 30% a 12 años. Available online: https://efeagro.com/eroskicambia-la-deuda-perpetua-por-una-quita-del-30-a-12-anos/ (accessed on 28 November 2024).
- Ellerman, David P. 1984. Theory of Legal Structure: Worker Cooperatives. Journal of Economic Issues 18: 861–91. [CrossRef]
- Ellerman, David P. 1986. Horizon Problems and Property Rights in Llabor-Managed Firms. *Journal of Comparative Economics* 10: 62–78. [CrossRef]
- Ellerman, David P. 2005. Translatio versus Concessio: Retrieving the Debate about Contracts of Alienation with an Application to Today's Employment Contract. *Politics & Society* 33: 449–80. [CrossRef]
- Ellerman, David P. 2016. Worker cooperatives as based on first principles. *JEOD, Journal of Entrepreneurial and Organizational Diversity* 5: 20–32. [CrossRef]
- Ellerman, David P. 2021. Neo-Abolitionism. Abolishing Human Rentals in Favor of Workplace Democracy. Heidelberg: Springer.
- Ellerman, David P., Tej Gonza, and Gregor Berkopec. 2022a. European Employee Stock Ownership Plan (ESOP): The Main Structural Features and Pilot Implementation in Slovenia. *SN Business Economics* 2: 186. [CrossRef]
- Ellerman, David P., Tej Gonza, and Gregor Berkopec. 2022b. European ESOP: The Main Structural Features and Pilot Implementation in Slovenia. Inštitutza Ekonomsko Demokracijo. Ljubljana, SI. Available online: https://mfgren.org/wp-content/uploads/2022 /07/ellerman-piece-on-esops-060722.pdf (accessed on 22 October 2024).
- Erdal, David. 2012. Beyond the Corporation: Humanity Working. London: Bodley Head.
- Farsio, Farzad, Amanda Geary, and Justin Moser. 2004. The Relationship Between Dividends and Earnings. *Journal for Economic Educators* 4: 1–5.
- Fici, Antonio. 2016. Governance and Member Participation in Worker Cooperatives. Euricse Working Papers, No. 8616. Available online: https://papers.ssrn.com/sol3/papers.cfm?abstract_id=2777513 (accessed on 9 December 2024).
- Frey, Bruno. 1997. Not Just for the Money: An Economic Theory of Personal Motivation. Cheltenham: Elgar.
- Frey, Bruno, and Reto Jegen. 2001. Motivation Crowding Theory. Journal of Economic Surveys 15: 589–700. [CrossRef]
- Fulton, Murray E., and Kathy A. Larson. 2009. The Restructuring of the Saskatchewan Wheat Pool: Overconfidence and Agency. *Journal of Cooperatives* 23: 1–19.
- Furubotn, Eirik G. 1976. The Long-Run analysis of the Labor-Managed Firm: An Alternative Interpretation. *The American Economic Review* 66: 104–23.
- Furubotn, Eirik G., and Svetozar Pejovich. 1970. Property Rights and the Behavior of the Firm in Socialist State: The Example of Yugoslavia. *Zeitschrift für Nationalökonomie* 30: 431–54. [CrossRef]
- Galor, Zvi. 2015. Equity in Service at Cost: The Case of the Classic Moshav. Journal of Co-operative Studies 48: 28–37.
- Galor, Zvi, and Michael Sofer. 2019. The Reserve Fund: Is it a Necessary Anchor for a Successful Cooperative? *Journal of Co-operative Organization and Management* 7: 100089. [CrossRef]
- Handy, Jim. 2022. Apostles of Inequality: Rural Poverty, and the Economist, 1760–1860. Toronto: Toronto UP, p. 58.
- Hansmann, Henry. 1996. The Ownership of the Enterprise. Harvard: Harvard UP.
- Hardin, Garret. 1968. The tragedy of the commons. Science 162: 1243–48. [CrossRef] [PubMed]
- Hecker, Achim. 2012. Knowledge Beyond the Individual? Making Sense of a Notion of Collective Knowledge in Organization Theory Organization Studies. *Organization Studies* 33: 423–45. [CrossRef]
- Herzberg, Frederick, Bernard Mausner, and Barbara B. Snyderman. 1959. The Motivation to Work. New York: John Wiley.

Herzberg, Frederick. 1987. One More Time: How Do You Motivate Employees? Harvard Business Review 81: 87-96.

- Hill, Charles W.L., and Scott A. Snell. 2017. Effects of Ownership Structure and Control on Corporate Productivity. *Academy of Management Journal* 32: 25–46. [CrossRef]
- Holmstrom, Bengt. 1982. Moral Hazard in Teams. The Bell Journal of Economics 13: 324-40. [CrossRef]
- Jeng, Leslie A., Andrew Metrick, and Richard Zeckhauser. 2003. Estimating the Returns to Insider Trading: A Performance-Evaluation Perspective. *The Review of Economics and Statistics* 85: 453–71. [CrossRef]
- Jensen, Michael C., and William H. Meckling. 1976. Theory of the Firm: Managerial Behavior, Agency Costs, and Ownership Structure. Journal of Financial Economics 3: 305–60. [CrossRef]

- Jensen, Michael C., and William H. Meckling. 1979. Rights and Production Functions: An Application to Labor-Managed Firms and Codetermination. *The Journal of Business* 52: 469–506. [CrossRef]
- Jensen, Michael C., and Clifford W. Smith. 1985. Stockholder, Manager, and Creditor Interests: Applications of Agency Theory. In *Recent Advances in Corporate Finance*. Edited by Edward I. Altman and Marti G. Subrahmanyam. Homewood: Irwin.
- Jones, Derek C., and Jan Svejnar. 1985. Participation, Profit Sharing, Worker Ownership and Efficiency in Italian Producer Cooperatives. *Economica* 52: 449–65. [CrossRef]
- Kelso, Louis O., and Mortimer J. Adler. 1958. The Capitalist Manifesto. New York: Random House.
- Keynes, John M. 1936. The General Theory of Employment, Interest and Money. London: Macmillan.
- Kiyosaki, Robert T. 2017. Rich Dad Poor Dad: What the Rich Teach Their Kids About Money That the Poor and Middle Class Do Not! 2nd ed.New York: Plata Publishing.
- Koller, Tim, Marc Goedhart, and David Wessels. 2020. *Valuation: Measuring and Managing the Value of Companies*, 7th ed. New York: Wiley.
- Kruse, Douglas. 2022. Does Employee Ownership Improve Performance? IZA World of Labor 311: 1–14. [CrossRef]
- Kumbhakar, Subal C., and Amy E. Dunbar. 1993. The Elusive ESOP-Productivity Link. Evidence from U.S. Firm-Level Data. *Journal of Public Economics* 52: 273–83. [CrossRef]
- Larrain, Borja, and Francisco Urzúa. 2013. Controlling Shareholders and Market Timing in Share Issuance. *Journal of Financial Economics* 109: 661–81. [CrossRef]
- Lindenberg, Siegwart M., and Nicolai J. Foss. 2011. Managing Joint Production Motivation: The Role of Goal Framing and Governance Mechanisms. *Academy of Management Review* 36: 500–25. [CrossRef]
- Lintner, John. 1962. Dividends, Earnings, Leverage, Stock Prices and the Supply of Capital to Corporations. *Review of Economics and Statistics* 44: 243–69. [CrossRef]
- Locke, John. 1690. Second Treatise on Government. Indianapolis: Hackett.
- Logue, John, Jacquelyn Yates, and William Greider. 2001. The Real World of Employee Ownership. Ithaca: Cornell University Press.
- Lomuscio, Marco. 2024. Going collective: Worker Takeovers, Entrepreneurship, and Collective Actions. *Scandinavian Journal of Management* 40: 101368. [CrossRef]
- Lund, Dorothy S. 2019. Nonvoting Shares and Efficient Corporate Governance. Stanford Law Review 71: 687–745. [CrossRef]
- Malleson, Tom. 2013. What Does Mondragon Teach Us About Workplace Democracy? Advances in the Economic Analysis of Participatory & Labor-Managed Firms 14: 127–57. [CrossRef]
- Marcuello, Carmen. 2023. Employee Ownership in Spain: Worker Cooperatives and Sociedades Laborales. *Journal of Participation and Employee Ownership* 6: 149–65. [CrossRef]
- Mazzini, Giuseppe. 1860. Doveri dell'Uomo. Rome: Editori Riuniti.
- Mazzini, Giuseppe. 1861. Scritti Politici. Milan: Daelli.
- McCain, Roger A. 1977. On the Optimum Financial Environment for Worker Cooperatives. Zeitschrift für Nationalökonomie 37: 355–84. [CrossRef]
- Meade, James E. 1972. The Theory of Labour-Managed Firms and of Profit Sharing. Economic Journal 82: 402–28. [CrossRef]
- Meade, James E. 1986. Alternative Systems of Business Organisation and Workers' Remuneration. London: Allen and Unwin.
- Meade, James E. 1989. Agathotopia: The Economics of Partnership. Aberdeen: Aberdeen UP.
- Medina-Albaladejo, Francisco J. 2017. Consumer Co-operatives in Spain, 1860–2010. In *A Global History of Consumer Co-Operation Since* 1850. Edited by Mary Hilson, Silke Neunsinger and Greg Patmore. Leiden: Brill, pp. 326–52.
- Miao, Qing, Hui Yin, Gary Schwarz, and Muhammad Ali Hussain. 2024. Disentangling the Impact of Perceived Electronic Performance Monitoring on Employee Burnout in the Public Sector. *Public Management Review Early View*. [CrossRef]
- Mikami, Kazuhiko. 2016. Cooperatives, Transferable Shares, And A Unified Business Law. *Annals of Public and Cooperative Economics* 87: 365–90. [CrossRef]
- Mill, John S. 1871. Principles of Political Economy with Some of their Applications to Social Philosophy. London: Longmans.
- Miyazaki, Hajime, and Hugh M. Neary. 1983. The Illyrian Firm Revisited. *Bell Journal of Economics* 14: 259–70. [CrossRef] Morrison, Roy. 1989. *We Build the Road As We Travel*. Gabriola: New Society.
- Mugerman, Yevgeny, Eyal Winter, and Tomer Yafeh. 2024. Performance Monitoring and the Incentives for Exertion of Effort. *The Quarterly Journal of Finance* 14: 2440005. [CrossRef]
- Navarra, Cecilia. 2011. Profit reinvestment in Italian worker co-operatives as a contribution to a common good: An empirical analysis on workers' perception and motivation. *Advances in the Economic Analysis of Participatory and Labor-Managed Firms* 12: 199–229. [CrossRef]
- Navarra, C. 2016. Employment stabilization inside firms: An empirical investigation on worker cooperatives. *Annals of Public and Cooperative Economics* 87: 563–85. [CrossRef]
- Navarra, Cecilia, and Ermanno C. Tortia. 2014. Employer Moral Hazard, Wage Rigidity, and Worker Cooperatives: A Theoretical Appraisal. *Journal Economic Issues* 48: 707–26. [CrossRef]

- O'Boyle, Ernest H., Pankaj C. Patel, and Erik Gonzalez-Mulé. 2016. Employee Ownership and Firm Performance: A Meta-Analysis. *Human Resource Management Journal* 26: 425–48. [CrossRef]
- Olson, Mancur. 1965. The Logic of Collective Action: Public Goods and the Theory of Groups. Cambridge, MA: Harvard UP.
- Ostrom, Elinor. 1990. Governing the Commons. The Evolution of Institutions for Collective Action. Cambridge, MA: Cambridge UP.

Pejovich, Svetozar. 1990. The Economics of Property Rights. Towards a Theory of Comparative Systems. Heidelberg: Springer.

- Pencavel, John. 2001. Worker Participation: Lessons from Worker Co-ops of the Pacific Northwest. Thousand Oaks: Sage.
- Pencavel, John, Luigi Pistaferri, and Fabiano Schivardi. 2006. Wages, Employment, and Capital in Capitalist and Worker-Owned Firms. *ILR Review* 60: 23–44. [CrossRef]
- Pérotin, Virginie. 2013. Worker Cooperatives: Good, Sustainable Jobs in the Community. *JEOD, Journal of Entrepreneurial and Organizational Diversity* 2: 34–47. [CrossRef]
- Prendergast, Canice. 1999. The Provision of Incentives in Firms. Journal of Economic Literature 37: 7-63. [CrossRef]
- Putterman, Louis. 1988. The Firm as Association versus the Firm as Commodity. Efficiency, Rights and Ownership. *Economics and Philosophy* 4: 243–66. [CrossRef]
- Ridley-Duff, Rory. 2009. Co-operative Social Enterprises: Company Rules, Access to Finance and Management Practice. *Social Enterprise Journal* 5: 50–68. [CrossRef]
- Romani, Marzio A. 2005. La Banca dei Milanesi. Storia della Banca Popolare di Milano. Bologna: Laterza.
- Rosen, Corey. 2023. Employee Ownership in the USA: Lessons to Consider in Creating More Inclusive Capitalism. *Journal of Participation* and Employee Ownership 6: 215–29. [CrossRef]
- Rosen, Corey, and Michael Quarrey. 1987. How Well Is Employee Ownership Working? Harvard Business Review 65: 126–32.
- Sacchetti, Silvia. 2015. Inclusive and Exclusive Social Preferences: A Deweyan Framework to Explain Governance Heterogeneity. Journal of Business Ethics 126: 473–85. [CrossRef]
- Salaman, Graeme, and John Storey. 2016. A Better Way of Doing Business? Lessons from The John Lewis Partnership. Oxford: Oxford University Press.
- Schoemaker, Paul J. H., and J. Edward Russo. 2016. Decision-Making. In *The Palgrave Encyclopedia of Strategic Management*. Edited by Mie Augier and David J. Teece. London: Palgrave Macmillan, pp. 397–401. [CrossRef]
- Schultz, Theodore W. 1961. Investment in Human Capital. The American Economic Review 51: 1–17.
- Screpanti, Ernesto. 2001. The Fundamental Institutions of Capitalism. Abingdon: Routledge.
- Screpanti, Ernesto. 2017. Karl Marx on Wage Labor: From Natural Abstraction to Formal Subsumption. *Rethinking Marxism* 29: 511–37. [CrossRef]
- Shulman, Lee S., and Neil B. Carey. 1984. Psychology and the Limitations of Individual Rationality: Implications for the Study of Reasoning and Civility. *Review of Educational Research* 54: 501–24. [CrossRef]
- Simon, Herbert A. 1951. A Formal Theory of the Employment Relationship. Econometrica 19: 293–305. [CrossRef]
- Skinner, Douglas S., and Eugene Soltes. 2009. What Do Dividends Tell us About Earnings Quality? *Review of Accounting Studies* 16: 1–28. [CrossRef]
- Song, Ok-Rial. 2002. Hidden Social Costs of Open Market Share Repurchase. Journal of Corporate Law 27: 425–77.
- Sraffa, Piero. 1960. Production of Commodities by Means of Commodities. Prelude to a Critique of Economic Theory. Cambridge: Cambridge UP.
- Storey, John, Imanol Basterretxea, and Graeme Salaman. 2014. Managing and Resisting 'Degeneration' in Employee-Owned Businesses: A Comparative Study of Two Large Retailers in Spain and the United Kingdom. *Organization* 21: 626–44. [CrossRef]
- Tortia, Ermanno C. 2007. Self-Financing in Labor-Managed Firms: Individual Capital Accounts and Bonds. Advances in the Economic Analysis of Participatory & Labor-Managed Firms 10: 233–61. [CrossRef]
- Tortia, Ermanno C. 2008. Worker Well-being and Perceived Fairness: Survey-based Findings from Italy. *Journal of Socio-Economics* 37: 2080–94. [CrossRef]
- Tortia, Ermanno C. 2018. The Firm as a Common. Non-Divided Ownership, Patrimonial Stability and Longevity of Co-Operative Enterprises. *Sustainability* 10: 1023. [CrossRef]
- Tortia, Ermanno C. 2021. Capital as Common-Pool Resource: Horizon Problem, Financial Sustainability and Reserves in Worker Cooperatives. *Journal of Co-operative Organization and Management* 9: 100137. [CrossRef]
- Tortia, Ermanno C. 2022. Employment Protection Regimes and Dismissal of Members in Worker Cooperatives. *Scandinavian Journal of Management* 38: 101213. [CrossRef]
- Tortia, Ermanno C., Silvia Sacchetti, and Vladislav Valentinov. 2021. The 'protective function' of social enterprises: Understanding the renewal of multiple sets of motivations. *Review of Social Economy* 78: 373–410. [CrossRef]
- Uvalic, Milica. 2010. Investment and Property Rights in Yugoslavia. The Long Transition to a Market Economy. Cambridge: Cambridge UP. Vanek, Jaroslav. 1970. The General Theory of Labour Managed Market Economies. Ithaca: Cornell UP.
- Vanek, Jaroslav. 1977. The Labor-managed Economy: Essays. Ithaca: Cornell UP.
- Ward, Benjamin. 1958. The Firm in Illyria: Market Syndicalism. The American Economic Review 48: 566–89.

- Wei, Xing, Xinyi Wei, Xiaotong Yu, and Feifei Ren. 2024. The Relationship Between Financial Stress and Job Performance in China: The Role of Work Engagement and Emotional Exhaustion. *Psychology Research and Behavior Management* 2: 2905–17. [CrossRef] [PubMed]
- Weitzman, Martin L. 1986. The Share Economy. Conquering Stagflation. Cambridge: Cambridge University Press.
- Whyte, William F., and Kathleen K. Whyte. 1991. *Making Mondragon. The Growth and Dynamics of the Worker Cooperative Complex*. Ithaca: Cornell UP.
- Williamson, Oliver E. 2000. The New Institutional Economics: Taking Stock, Looking Ahead. *Journal of Economic Literature* 38: 595–613. [CrossRef]
- Winther, Gorm, and Richard Marens. 1997. Participatory Democracy May Go a Long Way: Comparative Growth Performance of Employee Ownership Firms in New York and Washington States. *Economic and Industrial Democracy* 18: 393–422. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.





Article Nonparametric Testing for Information Asymmetry in the Mortgage Servicing Market

Helmi Jedidi and Georges Dionne *

Risk Management and Finance Department, HEC Montreal, Montréal, QC H3T 2A7, Canada; helmi.jedidi@hec.ca * Correspondence: georges.dionne@hec.ca; Tel.: +1-514-340-6596

Correspondence: georges.alonne@nec.ca; 1el.: +1-514-540-6

Abstract: Our objective is to test for evidence of information asymmetry in the mortgage servicing market. Does the sale of mortgage servicing rights (MSR) by the initial lender to a second servicing institution unveil any residual asymmetric information? We are the first to analyze the originator's selling choice of MSR. We use a large sample of U.S. mortgages that were securitized through the private-label channel during the period of January 2000 to December 2013 (more than 5 million observations). We propose a new nonparametric instrumental variable testing procedure to account for potential endogeneity. For robustness, we present parametric analyses to corroborate our results using instrumental variables. Our empirical results provide strong support for the presence of second-stage asymmetric information in the mortgage servicing market during the period of analysis and before the risk retention reform of 2014.

Keywords: mortgage servicing market; securitization; information asymmetry test; parametric model; nonparametric model; kernel estimation; instrumental variable; causality

JEL Classification: C14; C23; C26; G14; G21; G33



Citation: Jedidi, Helmi, and Georges Dionne. 2024. Nonparametric Testing for Information Asymmetry in the Mortgage Servicing Market. *Risks* 12: 192. https://doi.org/10.3390/ risks12120192

Academic Editors: Eulália Mota Santos and Margarida Freitas Oliveira

Received: 1 October 2024 Revised: 18 November 2024 Accepted: 22 November 2024 Published: 29 November 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

1. Introduction

Rapid development of financial markets and advances in structured finance have enabled lenders to overcome the traditional lending scheme by removing mortgages they originate from their balance sheets before the scheduled maturity through securitization. Securitization enables mortgage originators to sell mortgage-related cash flows to thirdparty investors in the form of liquid interest-bearing securities traded on financial markets (commonly known as Mortgage-Backed Securities, MBSs). The two main advantages of securitization for banks are to improve liquidity and to reduce regulatory capital requirements. The process of securitization involves numerous entities, such as the mortgage servicer (Gorton and Metrick 2012).

Once the securitization process is achieved and the underlying MBSs are sold to investors, the mortgage servicer ensures the ongoing management and upkeep of interest payments. In general, the main task of a mortgage servicer is collecting principal and interest payments from borrowers and passing the proceeds on to the underlying MBS investors in the secondary market. These cash flows are passive claims linked to the pool of mortgages packaged by the Special Purpose Vehicle (SPV) and held by MBS investors. Typically, the mortgage originator can act as the servicer of the deal by guaranteeing the connection of cash-flow streams between borrowers and MBS investors. However, originators are also able to further modify the borrower's default risk by selling the underlying mortgage servicing rights (MSRs) to a third party, hereafter referred to as the MSR purchaser or new servicer. In such cases, the new servicer replaces the originator in ensuring ongoing mortgage management; borrowers become directly linked to the new servicer, to whom they make debt payments.

69

In case of borrower delinquency, the servicing cost of mortgages increases significantly as the servicer incurs additional costs related to managing these loans, which can significantly reduce the profitability of the servicing activity. For instance, the mortgage servicer is required to deploy additional resources to investigate and collect delinquent payments, to perform loss mitigation activities, or to manage a foreclosure process. Mortgage servicers could incur additional significant costs related to unreimbursed foreclosure costs and real-estate owned losses. For these reasons, servicing inferior-quality mortgages could hinder the performance of mortgage servicing.

The main objective of this study is to test for evidence of information asymmetry in the mortgage servicing market. We are the first to analyze the following main question: Does selling the mortgage servicing rights unveil any residual asymmetric information in the servicing market? In a typical principal-agent relationship, we hypothesize that the mortgage originator (the agent) possesses an informational advantage over the MSR purchaser (the principal) in the market for mortgage servicing rights. This privileged information about both loan risk characteristics and borrower credit quality is collected at the time of the original mortgage underwriting, and the originator could have inducements to adversely exploit this information asymmetry. It is now well documented that soft information is private and not verifiable by the principal (Agarwal and Hauswald 2010; Bubb and Kaufman 2014; Filomeni 2024; Gete and Reher 2021; Passmore and Sparks 2024; Agarwal and Ben-David 2018; Liberti and Petersen 2018). Mortgage originators can transmit quantitative information about loan risk and borrower quality, but it is very difficult or costly for the MSR purchaser to access the soft information that may affect the loan default probability.

Although a large body of theoretical and empirical literature has examined asymmetric information through the securitization process (see Ambrose et al. 2005; Keys et al. 2010, 2012; Agarwal et al. 2012; Krainer and Laderman 2014; Malekan and Dionne 2014; Albertazzi et al. 2015; Elul 2016, among many others), we are the first to investigate this second-stage asymmetric information problem. The above-mentioned studies focus on information asymmetry between lenders and investors at the first stage of securitization. The main research question for most studies that test for asymmetric information through the securitization process is investigating the originators' decision to securitize a given loan. For instance, most studies compare the ex-ante risk characteristics as well as the ex-post default likelihood of mortgages that the originator chooses to securitize versus those kept on its balance sheet. In this study, we focus on mortgages that have already been securitized. We consequently dig deeper into the data as we scrutinize these securitized mortgages to test for second-stage information asymmetry. The previous estimations of asymmetric information in the securitization market may have underestimated the information problem in the securitization market by not taking into account this intermediary or second-stage information problem between the originator and the servicer.

To empirically test for evidence of asymmetric information in the market for mortgage servicing rights, we analyze the originator's selling choice of MSRs using a large sample of U.S. mortgages that were issued and securitized through the non-agency channel during the period of January 2000 to December 2013. In the first step, we contrast the ex-ante risk profile of mortgages for which the originator chooses to sell the underlying servicing rights to a third party with those for which it chooses to hold and service. In the second step, we compare the ex-post default risk of observably similar mortgages. Our econometric methodology is merely nonparametric in the sense that we do not make any restrictive assumptions about either the conditional distribution of the originator's MSR selling decision or the functional form of the relationship between the decision to switch the mortgage servicer and the mortgage default risk. The main advantage of this methodology is that inferences about the distribution are made purely from the data, and the density estimation is thus more data-driven than it would be if the density function were constrained to fall in a given parametric family. Our methodology is inspired by the nonparametric test of asymmetric information proposed by Su and Spindler (2013). The test is mainly driven by kernel density estimation methods. We are also the first to propose a new nonparametric two-stage instrumental variable testing procedure to account for potential endogeneity. For robustness, we employ the nonparametric testing procedure of Chiappori and Salanié (2000). We also present a battery of parametric analyses to corroborate our results after controlling for observable risk characteristics, econometric misspecification errors, and endogeneity issues using the instrumental variable estimation procedure developed by Dionne et al. (2009, 2015).

Our empirical results provide strong support for the presence of second-stage residual asymmetric information in the mortgage servicing market. After controlling for available hard information, we obtain a significant positive association between the lenders' decision to switch the servicer of the deal and the average probability of mortgage default. For instance, our results show that the higher the likelihood of switching the mortgage servicer, the higher the probability that the borrower defaults.

Our evidence suggests that originating lenders are indeed taking advantage of privileged information about both loan risk characteristics and borrower credit quality, which they obtain at the time of the original underwriting. Since this information is not observable by the second servicing institution, it seems clear that asymmetric information influences the decision of mortgage originators about whether to keep servicing mortgages they originate or to sell the underlying servicing rights to a third party. Two explanations based on contract theory are possible. First, the originator may be retaining superior-quality loans with a low probability of default in its servicing portfolio and, conversely, selling lemons with a high default risk, an outcome related to adverse selection. Alternatively, the transfer of mortgage servicing rights may be reducing the originator's efforts to screen applicants and monitor borrowers, as soon as the underlying servicing rights are planned to be sold to another servicer; an outcome related to moral hazard. Separating the two information problems, as Keys et al. (2012, rule of thumb with GSE loans) and Dionne et al. (2013, with insurance dynamic data) do with parametric modeling, is beyond the scope of this paper.

Our contribution must be considered as an additional motivation for the 2014 change in regulation of the securitization market. The new retention program destinated to increase the skin in the game of the originators should not only reduce the effect of the first stage of asymmetric information between the originator and the investors but also reduce incentives for the originator to transfer the management of bad loans to independent servicers.

The remainder of this paper proceeds as follows. Section 2 introduces the servicing activity and briefly describes the income stream of mortgage servicers. We also present and discuss the main risks that mortgage servicers encounter. We introduce the nonparametric kernel density estimation techniques in Section 3 and present the proposed nonparametric information asymmetry test in Section 4. Section 5 describes the data as well as the variables used in our study. Section 6 reports the main empirical results of the nonparametric testing procedure. For robustness, we also report the results of commonly used parametric tests with instrumental variables. Finally, we extend our nonparametric test to account for potential endogeneity. Section 7 concludes the paper.

2. Overview of the Mortgage Servicing Activity

2.1. Representation of Mortgage Servicing Process

Figure 1 shows the various contracted parties involved in the mortgage servicing process along with the generated cash flows in every step. A typical mortgage lending process starts with a borrower applying for a mortgage in order to buy a property or to refinance an existing mortgage to take advantage of lower interest payments. Usually, the mortgage originator is a commercial bank, a credit union, or a non-depository retail lender. Whatever the case, the mortgage originator manages the complete loan-granting process. Based on its information set, the originating lender expends effort to assess the borrower's reliability and creditworthiness. Eventually, if the borrower meets the lending requirements, the mortgage application is approved, and funds are released as represented by cash flow



1 in Figure 1. The debt payments in the absence of securitization are represented by cash flow **2**.

Figure 1. Lending and securitization processes.

This securitization is defined as the process whereby illiquid loans extended to borrowers are converted into liquid securities traded on financial markets. This process is summarized in steps **3** to **6**. In the first step of the securitization process, the originating institution transfers the mortgage to a special purpose vehicle defined as a legally separate entity created to handle the securitization process. The mortgage transfer is marked by cash flow **3**. This process of handling securitization involves external parties such as the underwriter that assists with the sale, the credit enhancement agency, and the credit rating agency that rates the interest-bearing securities. Once the tradable MBS are created and rated, the SPV sells them to investors, as depicted in cash flows **4** to **5**. Finally, the SPV uses the proceeds of the MBSs sale to pay back the entity that originated the underlying debt, as illustrated by cash flow **6**.

Once the securitization process is completed, the mortgage servicer ensures ongoing management and the upkeep of the payments. The main task of a mortgage servicer is collecting principal and interest payments from the borrower (cash flow 7) and passing the proceeds along to the underlying MBS investors in the secondary market (cash flow 8). Thus, as the borrower makes interest and principal payments, the servicer of the deal ensures that the cash flows are paid back to investors in accordance with the terms laid out in the securities prospectus. The mortgage originator can act as the servicer of the deal or it can sell the mortgage servicing rights (MSR) to a third party involved in this process, hereafter referred to as the new servicer or the MSR purchaser.

The sale of mortgage servicing rights and the corresponding cash proceeds are indicated by cash flows **9** and **10**, respectively. In this case, the buyer of the mortgage servicing rights replaces the original servicer of the deal and ensures the ongoing mortgage management. Therefore, borrowers become directly linked to the new servicer to whom they continue making monthly debt payments (cash flow **11**) that the former passes along to the MBS investors in the secondary market, as indicated by cash flow **12**. Customarily, in return for these services, the new mortgage servicer is paid a monthly fee generally specified as a fixed percentage of the declining unpaid balance of the underlying mortgage loan. The new servicer is also entitled to collect other fees such as float income, late payment fees, and other ancillary income. All these income streams are represented by cash flow 13. Finally, if a delinquent borrower defaults on a loan and stops making monthly payments due to financial distress, the new mortgage servicer is required to advance funds to MBS investors in the secondary market in keeping with the terms and conditions of the loan servicing contract, as indicated by cash flow 14.

At this point, it is crucial to note that neither the new mortgage servicer nor MBS investors in the secondary market observe all the information that the originating lender possesses. In fact, information detained by the lender could be classified into two main forms: hard and soft (Agarwal and Hauswald 2010; Keys et al. 2010, 2012; Malekan and Dionne 2014; Agarwal and Ben-David 2018; Liberti and Petersen 2018, among others). Hard information includes quantitative records of data such as credit scores and financial records. Soft information includes information that cannot be quantified or stored such as opinions, and self-made judgments. In such an environment, the originating lender could (partially) provide hard information to a third party while soft information is privately held.

2.2. Cash Flows and Risks of the Servicer ¹

The most important source of revenue for a mortgage servicer is the servicing fee, generally specified as a fixed percentage of the declining unpaid balance of the underlying mortgage. Servicers do not collect servicing fee revenue for non-performing loans for which borrowers are delinquent.

There are three main risks associated with the mortgage servicing activity: prepayment risk, default risk, and operational risk. The prepayment risk is defined as the possibility of an early, unscheduled, full repayment of the loan. The default risk is defined as the hazard that a borrower will be unable to honor the required principal and/or interest payments on the mortgage agreement in a timely manner. The risks due to the operational side of the servicing business are of an entirely different nature. For example, in servicing the deal, there is the possibility that the initial mortgage was made based on fraudulent information. In this study, we focus on the risk of borrower default.

3. The Nonparametric Estimation Framework

3.1. Motivation for Nonparametric Methods

The primary advantage of nonparametric estimation methods is robustness, as they can be applied in a broader range of situations even where the parametric conditions of validity are not met. Another notable advantage of nonparametric approaches is the ability to handle various data types (e.g., continuous, ordinal, and ranked data) even if the data comprise outliers, anomalies widely recognized to seriously affect the routine of parametric tests. Finally, as shown by Su and Spindler (2013), nonparametric tests may have stronger power than parametric tests.

Our empirical analysis considers the nonparametric kernel density estimation (KDE) technique to estimate the probability density function (PDF) of random variables drawn from unspecified probability distributions. Although we will assume that the distribution has a deterministic PDF, the estimation methodology will be entirely data-driven in the sense that data will be allowed to speak for themselves more than would be the case if the PDF were constrained to fall in a given parametric family.

3.2. Multivariate Kernel Density Estimation with Mixed Data Types

Consider a continuous random variable X^c with a probability density function f_{X^c} . Let $\{X_i^c\}_{i=1}^n$ represents a sample of *n* independent and identically distributed i.i.d. data points drawn from an unknown probability distribution family. Technically, a *kernel* $k(\cdot)$ is defined as a function that weights sampled observations X_i^c based on their distance from a specific evaluation point x^c within a fixed range known as the bandwidth, *h*. The weights given by the kernel function to observations in the sample are known as *local* weights. The nonparametric estimator \hat{f}_{X^c} is commonly referred to as the Rosenblatt–Parzen estimator (Rosenblatt 1956; Parzen 1962).

For multivariate data under the i.i.d. assumption, the kernel-based estimator of the joint PDF is expressed as the product of univariate kernel functions. The considered approach uses the concept of "generalized product kernels" (Li and Racine 2007; Racine 2008). Accordingly, the estimator of the joint PDF of q continuous variables can be written as follows:

$$\hat{f}_{X^c}\left(x_1^c, \ x_2^c, \dots, x_q^c\right) = \frac{1}{n} \sum_{i=1}^n \prod_{s=1}^q \hat{h}_s^{-1} k\left(\frac{X_{i,s}^c - x_s^c}{\hat{h}_s}\right).$$
(1)

We use $X_{i,s}^c$ to denote the *s*-th component of the *q*-dimensional space of continuous regressors, where $s = \{1, ..., q\}$ and $i = \{1, ..., n\}$. \hat{h}_s refers to the estimated bandwidth of the *s*-th regressor. For the purpose of this paper, we consider the case where X is a vector containing a mix of discrete and continuous variables. Let $X = (X^c, X^d)$ define the ensemble of *q* continuous and *p* discrete random variables. The joint density function for both categorical and continuous data using the generalized kernel product method can be formulated as follows:

$$\hat{f}_X\left(x_1^c, \dots, x_q^c, x_1^d, \dots, x_p^d\right) = \frac{1}{n} \sum_{i=1}^n \prod_{s=1}^q \hat{h}_s^{-1} k\left(\frac{X_{i,s}^c - x_s^c}{\hat{h}_s}\right) \cdot \prod_{r=1}^p l\left(X_{i,r}^d, x_r^d, \hat{\gamma}_r\right),$$
(2)

where $l(\cdot)$ is a weighting kernel function that depends on the estimated bandwidth $(\hat{\gamma})$. $X_{i,r}^d$ denotes the *r*-th component of the *p*-dimensional space of discrete regressors, $r = \{1, \ldots, p\}$ and $i = \{1, \ldots, n\}$. The mixed-type kernel allows for having a nonparametric counterpart for discrete choice parametric models such as probit and logit.

3.3. Conditional Kernel Density Estimation for Information Asymmetry Test

The core of our information asymmetry test is the estimation of the conditional density function. Let $f_{Y|X}$ denotes the conditional density function of Y given X. Since our dependent variable Y is discrete, the conditional kernel-based estimator is given by:

$$\hat{f}_{Y|X}(y|x^{c}, x^{d}) = \frac{\frac{1}{n} \sum_{i=1}^{n} K(X_{i}^{c}, x^{c}) \cdot L(X_{i}^{d}, x^{d}) \cdot l(Y_{i}, y, \hat{\gamma}_{y})}{\frac{1}{n} \sum_{i=1}^{n} K(X_{i}^{c}, x^{c}) \cdot L(X_{i}^{d}, x^{d})},$$
(3)

where, for expositional simplicity, $K(X_i^c, x^c) = \prod_{s=1}^q \hat{h}_s^{-1} k \left(\frac{X_{i,s}^c - x_s^c}{\hat{h}_s} \right)$ denotes the standard product kernel of *s* continuous variables. Similarly, $L(X_i^d, x^d) = \prod_{r=1}^p l(X_{i,r}^d, x_r^d, \hat{\gamma}_r)$ denotes the product kernel functions of *r* discrete variables. Note that x^c and x^d are $p \times 1$ and $q \times 1$ vectors of evaluation points, respectively. $l(Y_i, y, \hat{\gamma}_y)$ denotes the univariate kernel density estimator for discrete variable *Y* with an estimated bandwidth $\hat{\gamma}_y$. For the purpose of the information asymmetry testing procedure, our object of interest is the cumulative distribution function (CDF) estimated by the kernel-based density estimation approach. Let $F_{Y|X}$ denotes the conditional CDF of *Y* given *X*. A nonparametric estimate of $F_{Y|X}$ is given by the following expression:

$$\hat{F}_{Y|X}(y|x^{c}, x^{d}) = \frac{\frac{1}{n} \sum_{i=1}^{n} K(X_{i}^{c}, x^{c}) \cdot L(X_{i}^{d}, x^{d}) \cdot I(Y_{i} \leq y)}{\frac{1}{n} \sum_{i=1}^{n} K(X_{i}^{c}, x^{c}) \cdot L(X_{i}^{d}, x^{d})},$$
(4)

where $I(Y_i \leq y)$ denotes an indicator function. This equation represents the core of the nonparametric test proposed by Su and Spindler (2013). In Section 4, we provide a detailed description of the testing procedure along with the hypothesis to be tested. The estimated value for the bandwidth (denoted \hat{h} for the continuous variable kernel and $\hat{\gamma}$ for the discrete variable kernel) is the value minimizing the integrated mean square error, or IMSE, simply

defined as a measure of the discrepancy between the estimated density \hat{f}_h and the true density f:

$$IMSE_{\hat{f}_{h}(x)} = E\left\{ \int \left(\hat{f}_{h}(y) - f(y) \right)^{2} dy \right\}$$

=
$$\int Biais \left(\hat{f}_{h}(y) \right)^{2} dy + \int Var \left(\hat{f}_{h}(y) \right) dy.$$
 (5)

The optimal bandwidth is a function of the second derivative of the true density, which is unknown in the model. The bandwidth approximation methods use some underlying assumptions about the true density, which are useful in an application with a large number of variables or a large sample size, as in our empirical application.

4. Nonparametric Information Asymmetry Test

We want to test if the data contains residual asymmetric information effects in the relationship between mortgage default and the decision to use a new servicer. We hypothesize that their informational advantage influences the behavior of mortgage originators by selling mortgage servicing rights. Formally, let *Y* denote the dependent variable or outcome, *X* the set of exogenous control variables, and *Z* a decision variable. In our context of mortgage servicing, *Y* refers to the event of default mortgage, *X* includes mortgage risk characteristics and borrower credit quality variables that are observable by both parties (hard information), while *Z* denotes the originator's decision to sell the mortgage servicing rights rather than to keep servicing. In a principal-agent context, the decision variable, *Z*, should provide no useful information on mortgage default if and only if the prediction of *F*(*Y*) given *X* and *Z* jointly coincides with its prediction given *X* alone (Dionne et al. 2001, 2006). If this is the case, the information asymmetry hypothesis is rejected. The above statement could be formally written in terms of conditional probability functions:

$$F(Y/X,Z) = F(Y/X),$$
(6)

where F(Y/X, Z) denotes the conditional cumulative density function (CDF) of mortgage default. Equation (6) means that the original lender's action to switch the servicer or to continue servicing the deal does not convey any additional information useful in predicting the probability of default of a given mortgage as long as all loan and borrower observable risk characteristics (hard information) reveal all necessary information to compute the CDF of mortgage default.

Given a set of *n* i.i.d. randomly drawn observations $\{Y_i, Z_i, X_i^c, X_i^d\}_{i=1}^n$, the nonparametric test is resumed in comparing the following two conditional CDF estimates $\hat{F}(y|x^c, x^d, z = 1)$ and $\hat{F}(y|x^c, x^d, z = 0)$ as presented in Equation (4) but augmented by the decision variable *Z* as follows:

$$\hat{F}(y|x^{c}, x^{d}, z) = \frac{\frac{1}{n} \sum_{i=1}^{n} K(X_{i}^{c}, x^{c}) \cdot L(X_{i}^{d}, x^{d}) \cdot I(Y_{i} \leq y) \cdot I(Z_{i} = z)}{\frac{1}{n} \sum_{i=1}^{n} K(X_{i}^{c}, x^{c}) \cdot L(X_{i}^{d}, x^{d}) \cdot I(Z_{i} = z)}.$$
(7)

A test statistic measures the variation in $\hat{F}(y|x^c, x^d, z)$ across possible values of z and different observations as follows:

$$D^* = \sum_{i=1}^{n} [\hat{F}(y_i | x_i^c, x_i^d, z_i = 1) - \hat{F}(y_i | x_i^c, x_i^d, z_i = 0)]^2 \cdot a(x_i^c),$$
(8)

where $a(\cdot)$ is a uniformly bounded nonnegative weight function with compact support that lies within the support of X_i^c . This quantity serves to perform trimming in areas of sparse support of the continuous conditioning variable. It can be formulated as follows:

$$a(x_i^c) = \prod_{s=1}^q I(q_s(0.025) \le X_{i,s}^c \le q_s(0.975)),$$
(9)

where $q_s(\alpha)$ denotes the α -th sample quantile of the *s*-th component of X_i^c and q is the total number of continuous variables. The test statistic D^* in Equation (8) could be viewed as the difference between the expected probability of default depending on whether the originator switches the servicer or not. We can show that D^* is asymptotically normally distributed under the null hypothesis of independence. We implement a bootstrap procedure to obtain the corresponding *p*-values. In Section 6.3, we propose an extension of the test in order to consider a causal relationship between Z and Y.

5. Data, Variables, and Summary Statistics

5.1. Data Source and Sample Construction

To empirically test for asymmetric information in the mortgage servicing market, we use a large data set provided by MBSData, LLC, River Edge NJ. The data comprise U.S. mortgages that were securitized through the non-agency channel. Mortgages securitized through the private-label channel have fundamental risk characteristics that make them riskier than average. In general, mortgages securitized through this channel do not conform to the prudent lending guidelines set by the government-sponsored enterprises (GSE) Freddie Mac, Fannie Mae, and Ginnie Mae. For example, most of the mortgages do not meet the GSE requirements in terms of loan size (e.g., jumbo loans with original loan amounts exceeding the conforming loan limits), documentation (e.g., loans with no or low level of documentation), and loan-to-value ratios (e.g., LTV ratio above 80%).

Owing to the lack of a government guarantee, holding these private-label securities carries a significantly higher risk than carrying the agency counterparts. For instance, without government back-up, private-label mortgage originators rely on both credit rating agencies and credit enhancements to attract MBS investors and convince them that the underlying mortgage is safe. The non-agency market had witnessed tremendous growth during the pre-crisis period. For instance, the outstanding quantity of non-agency mortgages grew from roughly \$600 billion at the end of 2003 to \$2.2 trillion at its peak in 2007, according to JP Morgan. The bad risk management of this tremendous growth in the non-agency market is widely recognized by both researchers and practitioners as being one of the main triggers of the financial crisis.

Our dataset consists of mortgages issued between January 2000 and December 2013. The initial sample consists of more than 25 million mortgages that were originated throughout the U.S. and granted by diverse types of lenders ranging from top investment banks to regional small retailers. The yearly distribution of loan origination follows a pattern similar to that observed in the entire U.S. mortgage market. The MBSData, LLC database consists of two main datasets. The first static file reports detailed information collected at the time of mortgage origination. For instance, it reports the borrower's FICO credit score, loan-to-value (LTV) ratio, loan purpose, payment type, initial interest rate, private insurance coverage, etc. It also provides information regarding the property backing the loan, which includes the house value, state, city, and zip code. For the originating entity, the database reports the originating lender's name and type, the name of the original mortgage servicer, as well as the most recently updated servicer name. The second dynamic file reports information that has been collected over the loan lifetime on a monthly basis. Key variables recorded in the monthly remittance files are the current loan balance, current interest rate, next due date, and a monthly delinquency code (current, paid-off, +30, +60, or +90 days delinquent, in foreclosure, in bankruptcy, or real estate owned). The dataset also provides information on losses and loan modification. Loss files mainly report loan-level loss amount, loss severity, recovery amount, loan liquidation proceeds, and current value at liquidation.

While constructing our sample, we impose several restrictions in order to create a homogenous loan sample. We focus on mortgages in a first-lien position on the property securing the mortgage and exclude second mortgages and home equity lines of credit (HELOCs). Our choice is primarily motivated by the fact that first-lien mortgages have priority over all other subsequent claims (i.e., second-lien or junior) on a property in the

event of borrower default. We restrict our attention to single-family owner-occupied homes and exclude multifamily and/or non-owner occupied properties. We also exclude loans whose main purpose is designated as home improvement and retain loans with the main purpose identified as a house purchase or refinancing an existing mortgage (both cash-out and no cash-out). We also exclude planned unit developments (PUDs) and mobile homes. All these restrictions result in a final sample including 5,591,353 distinct observations that originated between January 2000 and December 2013 and were tracked until December 2015 on a monthly basis.

5.2. Variables and Hypotheses

The main variable of interest in our empirical analysis is the mortgage servicer switching indicator, denoted as Switch_Servicer. This variable is a dummy indicator that equals 1 if the originating lender sells the mortgage servicing right to another servicer and 0 if the lender continues servicing the mortgage it originates. The second most important variable of interest is the standard Default dummy variable, which denotes whether a given mortgage becomes 90+ days delinquent (i.e., when a loan is first reported as the borrower having missed three or more consecutive monthly payments). For robustness, we also report results with 60+ days delinquent in the online Appendix A.

The set of covariates includes several explanatory variables recorded at the time of origination. All variables are defined in Table A1 of the online Appendix A. The first variable we consider is the borrower's FICO score. In general, the FICO score measures individuals' creditworthiness by taking into account their payment history, length of credit history, current level of indebtedness, and types of credit used. The score ranges from 300 to 850, and, typically, a FICO score above 660 is indicative of a good credit history. We expect that the originator will keep servicing mortgages with high FICO scores.

The second independent variable is the loan-to-value ratio, LTV, calculated by lenders as the percentage of the first-lien mortgage to the total appraised value of the purchased property. The LTV ratio is one of the key risk factors used by U.S. lenders when qualifying borrowers for a mortgage. In the United States, mortgagors with LTV ratios higher than 80% are required to buy private mortgage insurance to protect the lender from the default risk, which increases the cost of borrowing. The LTV ratio also measures the equity stake of borrowers in a given property. The higher the LTV ratio, the lower the down payment, so the lower the borrower's equity stake in that house. Since a high LTV ratio mirrors a risky mortgage, where the borrower holds a lower equity stake in a given house, we expect the lender's decision to switch the servicer of the deal will be positively correlated with the LTV ratio.

Another key explanatory variable in our analysis is no/low documentation, a dummy variable indicating whether the lender has collected the required level of documentation on the borrower. Typically, a borrower is asked to fill out a credit application and provide several statements and proofs of employment status and income when applying for a loan. Based on this documentation, the lender expends effort to assess the borrower's creditworthiness. Therefore, a no/low documentation loan is a loan for which the lender has not gathered a sufficient level of information on the borrower's income. In terms of default risk, we can presume that no/low documentation loans will default more frequently.

The next independent variable is the ARM indicator; ARM stands for adjustable-rate mortgages (commonly referred to as variable-rate mortgages). The ARM variable indicates whether the interest rate paid on the outstanding balance of a given mortgage varies according to a specific benchmark. Usually, the initial interest rate is fixed for a period of time, after which it is reset periodically, often every month. The interest rate paid by the borrower is usually based on a benchmark plus an additional spread, called the ARM margin. In terms of risk, ARM-type mortgages transfer part of the interest-rate risk from the lender to the borrower. Indeed, these mortgages are generally used when interest rates fluctuate and are difficult to predict (which makes fixed-rate mortgages, FRMs, difficult to

obtain). In terms of servicing choice, a positive statistical relationship is expected between terms of servicing choice and interest rate.

We also include a GSE conforming indicator to denote loans that obey the GSEs' lending guidelines. The GSE_conforming dummy variable indicates whether the mortgage was eligible to be sold to the GSEs at origination. Following the GSEs' recommendations, we classify a mortgage as conforming if the borrower's FICO score is above 660 and the loan amount is below the conforming loan limit in place at the time of origination and the LTV is either less than 80 percent or the loan has private mortgage insurance if the LTV is greater than 80 percent. Given that conforming loans meet the GSE lending standards, we expect a negative correlation with the default event. Indeed, falling within the GSE prudence guidelines should significantly reduce the probability of default. Regarding the choice of switching the servicer, we presume that both signs are plausible. On the one hand, being GSE-conforming increases the ease of finding a buyer of the underlying MSRs. For instance, because these loans are originated following the GSE standards, it would be easier to find buyers of the securitized pool of loans in the market. Thus, a positive sign is expected. On the other hand, being GSE-conforming increases the probability that the lender will be paid back as scheduled. Lenders may therefore keep these good-quality loans on their balance sheets because the risk of default on these loans is significantly low. Therefore, the sign of the conforming coefficient is an empirical matter.

5.3. Descriptive Statistics

We start the empirical analysis by providing summary statistics of some of the key variables used in our analysis. Since we are focusing on the non-agency market, we pay special attention to the role of credit scores, loan-to-value ratios, the amount of documentation collected by the lender, and some interest rate features. Table A2 reports descriptive statistics over the entire study period from January 2000 to December 2013 as well as segmented by origination year. Table A3 breaks down the sample by payment type (FRM vs. ARM), loan type (prime vs. subprime), before/after financial crisis, default status, and servicer switch status.

The first two columns of Table A2 provide a comprehensive picture of the evolution of the non-agency segment of the mortgage market over the 14-year study period. At first glance, mortgage origination has witnessed two major trends explained by the financial crisis. First, the market expanded rapidly from 2000 to 2006 and reached its highest level just before the financial crisis. Afterwards, mortgage origination plunged dramatically. During 2008 and 2009, the market also sustained a dramatic drop in origination of new mortgages. After the financial crisis (2010 and beyond), origination increased slightly but remained far from its level before the financial crisis.

The third column of Table A2 displays the average FICO credit score in the sample. The average credit score is 4 points lower than the 660 threshold. The evolution of the FICO credit score over the years is interesting. For instance, borrowers' credit quality was below the 660 threshold before the financial crisis (655) but above it afterwards (671). The credit score averaged 615 and 644 in years 2000 and 2002. However, after the crisis, credit quality improved significantly as the average FICO score is consistently higher than 770 in the 2010–2013 period.

Figures A1 and A2 of the online Appendix A examine the evolution of borrowers' credit quality. They depict the evolution of FICO scores by payment type (ARMs vs. FRMs) and by loan type (prime vs. subprime). As shown in Figure A1, ARM borrowers have lower credit scores than FRM borrowers, on average. For example, in 2002, the average FICO scores for ARMs and FRMs were 619 and 672, respectively. This trend is almost verified for the period before the financial crisis, after which the difference in credit scores is reduced to 10 points. Table A3 shows that the ARM-FRM FICO score differential over the study period is about 34 points, statistically significant at the 1% level. Figure A2 suggests that, unsurprisingly, the average credit score for subprime loans is significantly lower than for prime loans. For illustration, in 2002 the average FICO score for subprime loans is

almost 120 points lower than for prime borrowers (616 versus 735). Table A3 indicates that over the study period, the average FICO scores for prime and subprime borrowers are 731 and 635, respectively. The difference of 96 FICO points is statistically significant at the 5% level. After the financial crisis, the average credit score tended to improve each year, mainly due to the drop in subprime lending. As column 4 of Table A2 indicates, almost all loans originated after the financial crisis have a credit score higher than 660.

Regarding the loan-to-value (LTV) ratio of sampled mortgages, columns 5 and 6 of Table A2 show that the average LTV ratio in the sample is 77% and that 60% of loans in the sample have an LTV ratio higher than 80%. Regarding the evolution of the LTV ratio over the years, the LTV ratio plunged significantly soon after the financial crisis. For instance, column 6 of Table A2 shows that more than 60% of loans have an LTV ratio higher than 80% throughout the pre-crisis period. However, this proportion drops to almost 20% in the 2010–2013 post-crisis period. We further split our sample according to payment type (ARMs versus FRMs) and loan type (prime versus subprime). Table A3 shows the results over the entire studied period.

We also investigate the lender's effort to gather all documentation required at the date of original underwriting. The statistics show that lenders did not gather sufficient documentation on applicants in almost half of the cases (47% of the time, lenders granted funding to borrowers but gathered little or no documentation on borrowers' income and employment status). Yearly statistics show that this practice of granting funding without the required documentation increased steadily in the early 2000s. As shown in Figure A3, the high proportion of no/low documentation is mainly driven by the practice in the subprime segment; this proportion peaked at 70% from 2005 through 2007.

In general, the lending strategy appears to radically change after the financial crisis. This shift in lending strategy entailed (i) increasing loans granted for borrowers with good credit quality, (ii) reducing loans with a small down payment (LTV ratio higher than 80%), and (iii) reducing the proportion of loans granted with insufficient documentation. These changes in underwriting patterns are consistent with lenders looking for new ways to limit risk exposure after the financial crisis.

To motivate our empirical analysis, we contrast the ex-ante risk characteristics of mortgages for which the originator chooses to sell the underlying servicing rights to another servicer with mortgages that it chooses to continue to service. Overall, we note that, for 54.7 percent of the sampled mortgages (3,060,083 mortgages), the originator chooses to switch the mortgage servicer. For the remaining loans (45% of the sample), the originator keeps servicing mortgages it originates and holds them in its servicing portfolio until maturity. Table A3 shows that the average servicing fee is 44 bp, which does not change very much before and after the crisis. On average, lenders in the sample tend to charge significantly higher fees than the average servicing fees applied by the GSEs and the FHA/VA, at 25 bp and 19 bp, respectively.

Regarding the borrower's credit quality, the results show that lenders tend to keep servicing loans granted to borrowers with superior credit quality. For illustration, the average credit score for loans held in the originator's servicing portfolio is 661, while the average credit score for loans for which the lender decides to switch servicing is 654, namely 3 basis points below the sample average. The two-sample mean difference (untabulated) is 6.39 points, statistically significant at the 1% level. Table A3 also shows that the fraction of loans granted for borrowers with FICO scores higher than the 660 threshold is significantly larger for loans held in a portfolio (51% for non-switch versus 46% for switch).

These results indicate that lenders switch servicing of the deal for loans that are riskier and keep servicing mortgages that are less risky. For instance, the pool of loans for which the servicer has changed is characterized by higher loan-to-value ratios and slightly higher debt-to-income ratios. Regarding the subprime loan type, the primary statistics are not informative in that the propensity to switch the servicer of the deal is 52% for prime loans and slightly higher, at 56%, for subprime loans. The results also suggest that 15% of loans for which the servicer is switched follow the GSEs' prudent lending guidelines, whereas this percentage increases to 20% for loans held in the originator's servicing portfolio. The proportion of loans that conform to the GSE lending guidelines at origination represents only 17% of the sample.

To summarize, based on the observable risk characteristics of originated mortgages, these preliminary results are consistent with the evidence of lenders selling MSR rights for low-quality loans to other servicers and retaining high-quality mortgages in their own servicing portfolios. To better understand the originators' motive to switch the servicing of the deal, we further break down the mortgage sample by default status. The statistics show that, not surprisingly, lower FICO scores, higher LTV ratios, higher debt-to-income ratios, and higher interest rates are the risk characteristics that are more likely to be associated with the default outcome. For instance, 55% of loans that never entered delinquency are granted to borrowers with FICO scores above the 660 thresholds. In addition, 72% of loans identified as being in default exhibit an LTV ratio higher than 80%. Not surprisingly, following the GSE guidelines significantly reduces the observed default frequency in that only 10% of defaulting loans follow the GSE prudent lending guidelines.

Contrasting the distribution of loans that were chosen for servicer switch with the default outcome yields additional interesting findings. When comparing the default propensities between the switch and non-switch groups, the results show that 50% of loans defaulting have the servicer switched, compared with 18% of loans in the non-default category. In general, these preliminary results suggest a positive association between the originator's decision to switch servicers and the default outcome.

6. Empirical Results

6.1. Nonparametric Methods

We consider two nonparametric testing procedures proposed to isolate the evidence of asymmetric information in the automobile insurance market. The first approach, from Chiappori and Salanié (2000), is based on a sequence of Pearson's χ^2 nonparametric test of independence (see also the recent extension by Maliar and Salanié 2023). The second approach is mainly driven by kernel density estimation techniques and was proposed by Su and Spindler (2013).

6.1.1. The Chiappori and Salanié (2000) Method

We investigate the relationship between the originator (agent)'s action to sell the mortgage servicing right of a given mortgage and the likelihood of mortgage default. The null hypothesis to be tested is that there is no significant residual correlation between the decision to switch the servicer of the deal and the mortgage default event. Since the testing methodology considers only binary variables, we convert continuous variables into dummies. The final set of explanatory variables are FICO660, LTV80, ARM, no/low documentation, balloon, GSE conforming, subprime, and prepayment penalty. We use various variable inclusion configurations for robustness purposes. The upper part of Table 1 displays the different configurations that we use to define the set of control variables to be included in the analysis.

Table 1. Results of the Chiappori and Salanié nonparametric test. The table reports the results of the Chiappori and Salanié (2000) nonparametric testing methodology. The overall sample includes 5,591,353 U.S. mortgages originated over the period from January 2000 to December 2013. The mortgages have been securitized through the non-agency channel. The upper panel of the table reports 10 different configurations of the control variables. The table displays the number of variables included in each configuration as well as the resulting number of cells. KS *p*-value is the *p*-value of the Kolmogorov-Smirnov nonparametric test. $\chi^2_{(1)}$ crit. value is the theoretical value of the χ^2 distribution at the 5% significance level. Rejection rate provides the frequency of rejection of the null hypothesis of independence among all individual cells. S value is the sum of individual test statistics among all cells.

Configuration	Ι	II	III	IV	V	VI	VII	IIX	IX	X	XI
FICO.660	Yes	Yes	Yes								
LTV.80	Yes	Yes	Yes								
ARM	Yes	Yes	Yes								
No/Low doc.	-	Yes	Yes	Yes							
Balloon	-	-	Yes	-	-	-	Yes	Yes	Yes	Yes	Yes
GSE Conf.	-	-	-	Yes	-	-	Yes	-	-	Yes	-
Subprime	-	-	-	-	Yes	-	-	Yes	-	-	Yes
Prep. penalty	-	-	-	-	-	Yes	-	-	Yes	Yes	Yes
# variables	3	4	5	5	5	5	6	6	6	7	7
# cells (M)	8	16	32	32	32	32	64	64	64	128	128
Method 1:											
KS <i>p</i> -value	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00
Method 2:											
$\chi^2_{(1)}$ crit. value	3.84	3.84	3.84	3.84	3.84	3.84	3.84	3.84	3.84	3.84	3.84
Rejection rate	0.75	1.00	0.81	0.92	1.00	0.91	0.75	0.81	0.83	0.73	0.83
Method 3:											
$\chi^2_{(M)}$ crit. value	15.51	26.30	46.19	46.19	46.19	46.19	84.82	84.82	84.82	124.34	124.34
S value	6388.6	4491.4	6840.3	5577.2	4491.4	9638.9	7628.9	6840.3	11,089.8	11,230.5	11,089.8

The testing procedure could be summarized in the following steps. First, we select a set of *m* control variables. Since variables are binary, we construct $M = 2^m$ cells in which mortgages have the same values of a selected control variable. We then conduct the Pearson's χ^2 -test of independence in each cell. This procedure produces *M* Pearson's test statistics.

Table 1 displays the results of Chiappori and Salanié's (2000) testing procedure. The table reports the number of control variables included in each configuration and the total number of cells. We first examine *p*-values of the Kolmogorov-Smirnov (KS) test. Clearly, using all possible combinations, we unequivocally reject the null hypothesis at the 1% significance level. Using the second method, the rejection rate of the null hypothesis of independence in individual cells is high for all configurations. For instance, almost all test statistics within individual cells exceed the $\chi^2_{(1)}$ critical value of 3.84 (at a 5% significance level). The highest rejection rate is reached with configuration II, which includes four control variables FICO660, LTV80, ARM, and NoLow_doc. The latter method confirms these findings where the aggregate test statistic is above the critical values of the $\chi^2_{(M)}$ theoretical distribution according to all configurations.

6.1.2. The Su and Spindler (2013) Method

We begin by documenting how well kernel-based estimation fits our data. Figures A4 and A5 in the online Appendix A display histograms for two continuous variables: the borrower's FICO score and LTV ratio. For comparison, histograms are augmented with curves of the nonparametric kernel-based estimator and that of the parametric normal density function. From the two figures, it is clear that the kernel-based PDF much better fits the actual data than the parametric counterpart does. For illustration, the histogram of the LTV ratio

suggests that loans with LTV ratios falling in the 75–80% interval are over-represented in the sample. The parametric normal density underestimates that proportion by 5.5%, whereas the KDE provides better estimates.

Figure A6 highlights the key role of the bandwidth smoothing parameter in the estimation and displays the sensitivity of the kernel density estimation technique fitting to the data. The figure displays the KDE fitting for three different values of the bandwidth: high, optimal, and low values. It is obvious that failing to select the optimal bandwidth could be costly since it may result in over-fitting or under-fitting. In fact, the bandwidth, as a smoothing parameter, controls the size of the neighborhood around a given point of estimation. We use the Maximum Likelihood Cross-Validation (MLCV) method to estimate the bandwidth from the sample by optimizing the loss objective function on the true density. The estimation results show that the optimal bandwidth values are 3.357 for the FICO score and 0.716 for the LTV based on the MLCV method. These values of optimal bandwidths suggest a significant kernel density estimate because the bandwidths are higher than zero. We also include additional discrete binary control variables such as indicator variables for the ARM payment type, balloon loan type, no/low documentation, Subprime, and GSE conformity indicator. For all discrete variables, optimal bandwidth values are within the [0, 1] interval, which, according to Li and Racine (2007) and Racine (2008), means that variables are relevant to the model.

To display the results of the nonparametric testing framework, we use graphical representations where the borrower's FICO continuous variable serves as a support. Our choice is motivated by the fact that FICO represents a direct measure of a borrower's credit quality that the originator may use to assess the likelihood of default. Thus, this variable could be directly linked to both mortgage default and the originator's decision to switch the servicer.

Figure A7 displays the conditional probability of mortgage default using the kernel density estimation method. Conditional means that the probability of mortgage default is conditional on observed risk characteristics for both borrower and mortgage. For comparison purposes, Figure A7 displays fitted values of a linear parametric model. This model suggests a statistically significant negative coefficient for the FICO score when predicting mortgage default in a linear-imposed relationship. The kernel density estimation corroborates this finding.

Now we turn to the core of the nonparametric test for asymmetric information. Figure 2 displays the estimated probability of mortgage default conditional on all observed risk characteristics observed at the time of original underwriting. Moreover, the conditioning set for the estimated probability is now augmented with the agent (originator)'s action to switch the servicer of the deal. Formally, the figure displays two plots that correspond to $\hat{f}(y_i|x_i^c, x_i^d, z_i = 1)$ and $\hat{f}(y_i|x_i^c, x_i^d, z_i = 0)$, respectively. In simple words, the plots labelled "Switched" and "No Switched" refer to the probability of mortgage default conditional on various control variables and the originator's decision to switch or not the servicer of the mortgage.

Both plots show that the conditional probability of mortgage default decreases as the borrower's credit quality improves. However, the plots display a significant shift in the estimated default likelihood when the conditioning set accounts for the agent's action to switch the mortgage servicer. For illustration, mortgages granted for borrowers with an average FICO credit score of 550 display an estimated likelihood of mortgage default of 40% if the mortgage servicer was not switched. However, all other things held constant, changing the decision to switch the servicer of the deal increases the estimated probability of mortgage default by 10%. This 10% increase in the conditional probability of mortgage default is also observed over all FICO score intervals. This evidence suggests that the decision to switch the servicer increases the occurrence of mortgage default by almost 10%, other characteristics being equal. Note that mortgages under consideration share almost many characteristics since they belong to the same FICO score cohort. The only variable that makes the difference here is the agent's action to switch the servicer of the deal.



Figure 2. Switched decision vs. conditional probability of default.

Figure 2 also shows that this pattern is valid not only for low-quality borrowers but also for those with superior credit quality. Although the expected default likelihood drops significantly by almost 70% if we consider high-quality borrowers (FICO score above 700), the default likelihood still drops if the originator keeps the securitized mortgage on its servicing portfolio. For illustration, if we consider loans granted to borrowers with FICO scores higher than 750, the estimated conditional probability of default is about 19% if the originator sells the underlying MSRs while nearly zero if the latter keeps servicing the mortgage.

These results are in line with those found using Chiappori and Salanié's (2000) method. For instance, KDE-based results suggest a positive relationship between the conditional probability of mortgage default and the originator's decision to switch the servicer of the deal. In fact, we find that observably similar mortgages (i.e., with comparable risk factors and granted to borrowers with similar credit scores) experience more default risk if the mortgage originator sells the underlying MSR to a new servicer.

We use the bootstrap technique to obtain the *p*-values and conclude our asymmetric information test. First, we generate *B* bootstrap samples (with replacement), which we denote as $(X_i^{cb}, X_i^{db}, Y_i^b, Z_i^b)$ where the superscript *b* denotes the *b*th sample of data, $b = \{1 \dots B\}$. Next, each bootstrap sample *b* estimates the conditional kernel density of mortgage default given all observed characteristics along with the originator's switching decision to calculate the corresponding test statistic as in Equation (8). Let \hat{D}^b denote the estimated test statistic using bootstrap sample $b = \{1 \dots B\}$. The bootstrap *p*-value is given by:

$$\hat{p}_{B}(\hat{D}^{*}) = 2 \min\left(\frac{1}{B}\sum_{b=1}^{B} I(\hat{D}^{*} \le \hat{D}^{b}), \frac{1}{B}\sum_{b=1}^{B} I(\hat{D}^{*} > \hat{D}^{b})\right),$$
(10)

where I(.) is an indicator function and \hat{D}^* refers to the estimated test statistic as in Equation (8) from the real sample (Fisher and Hall 1990; MacKinnon 2009).

The set of explanatory variables that we consider in our computation is FICO, LTV80, ARM, no/low documentation, balloon, GSE conforming, subprime, and prepayment penalty. For robustness, we try several inclusion combinations of control variables as we did for the Chiappori and Salanié (2000) analysis (see upper panel of Table 1). The total number of bootstrap replications is set to B = 1000. For all possible configurations, we find that the bootstrap *p*-value is below the 5% statistical level. Clearly, low *p*-values enable us to conclude the statistical significance of our test, i.e., $\hat{F}(y_i|x_i^c, x_i^d, z_i = 1)$ and

 $\hat{F}(y_i|x_i^c, x_i^d, z_i = 0)$ being statistically different for every $i = \{1, \dots n\}$. In other words, the results suggest that in all cases we are able to reject the null hypothesis of absence of asymmetric information at the 5% level. This means there exists a significant residual relationship between the likelihood of mortgage default and the decision to sell the servicing right of a given mortgage.

6.2. Robustness Checks: Results of the Parametric Methods

We provide additional support for our evidence based on commonly used parametric models. We first employ the parametric probit model to investigate the determinants of mortgage default. This testing procedure is documented in Dionne et al. (2001, 2006), who establish that we can interchange the roles of *Y* and *Z* so that testing F(Y/X, Z) = F(Y/X) is equivalent to testing F(Z/X, Y) = F(Z/X). The latter equation means that the mortgage default does not provide useful information to predict the originator's decision to switch the mortgage servicer. Dionne et al. (2001, 2006) state that verifying either equality is indicative of the conditional independence of *Y* and *Z* given a set *X* of conditioning variables.

Table A4 displays the estimation results for the standard probit model where the dependent variable is the mortgage default binary variable. The table reports various inclusion configurations for the set of control variables. The proposed configurations control for (i) fundamental borrower and loan risk characteristics, (ii) general economic conditions, (iii) housing market conditions, (iv) bond market conditions, and (v) state legal structure. All explanatory variables display the expected signs. The Wald test and the likelihood-ratio test statistics show that all regressors' coefficients are statistically significant.

The parametric counterpart of the information asymmetry test consists primarily of scrutinizing the statistical link between the decision to switch the servicer and the likelihood of mortgage default. However, such methodology would potentially be problematic as it suffers from various issues, notably endogeneity, econometric misspecification, and simultaneity (Dionne et al. 2009, 2015). To deal with, we employ three different parametric methods. First, we use the two-stage instrumental variable probit model in order to account for potential endogeneity. We also employ the two-step estimation procedure proposed by Dionne et al. (2015) to account for econometric misspecification error and to correct for imposed linearity. Additionally, we employ the bivariate probit model in order in order to jointly estimate both binary outcomes in a system of simultaneous equations (Chiappori and Salanié 2000). Estimation results for the three models are reported in Table 2.

The first two columns of Table 2 display results of the two-stage instrumental variable probit model. The first stage regression estimates the mortgage default likelihood using Income and Divorce as instruments. Both aggregate instruments must be correlated with mortgage default but uncorrelated with the decision to switch the mortgage servicer. As expected, the first-stage regression shows that income growth is negatively correlated with mortgage default likelihood with a negative coefficient statistically significant at the 1% level. In contrast, the divorce rate is positively related to mortgage default, suggesting that marital breakdown represents a key factor in determining mortgage default. All other coefficients have the expected sign similar to previous findings in Table A4. The first-stage regression provides an estimate of the likelihood of borrower default that the originator formulates based on hard information.

Accordingly, the second-stage regression uses the expected likelihood of default as a regressor to investigate the originator's decision to switch the servicer of the deal. The results show a statistically significant positive coefficient (0.53) for the predicted mortgage default. This positive link is further confirmed after controlling for econometric misspecification via imposed linearity in the vein of Dionne et al. (2015). The statistically significant coefficient (0.487) on the predicted default variable highly suggests that expected mortgage default (based on the originator's private information) provides useful information to predict the originator's decision to switch the mortgage servicer. Such a result sheds light on the existence of information asymmetry in the U.S. mortgage servicing market. This result is further confirmed by using the default variable (0.32 + 0.17).

Table 2. Results of the two-stage and bivariate probit models. The table reports the estimation results using three parametric approaches: the two-stage instrumental variable probit, the two-stage linear model (Dionne et al. 2015), and the bivariate probit. The sample includes 5,591,353 mortgages originated over the period from January 2000 to December 2013. Income and Divorce are instruments for the endogenous variable Default. Income is the annual growth rate of the U.S. household income. Divorce is the annual rate of divorce in the U.S. Pr(Default = 1) denotes the predicted probability of default from the first stage probit regression. \hat{E} (Default) denotes the predicted default from the 1st stage linear model. Default denotes mortgage default (i.e., it is labelled as +90 days delinquent). Switch serv. denoting whether the originator switched the servicer of the deal. FICO score is the borrower's Fair Isaac Corporation score attributed at origination. LTV ratio denotes the initial loan-to-value ratio. ARM abbreviates adjustable-rate mortgages. Balloon refers to balloon payment mortgages. No/Low doc. indicates whether the originator collected no/low documentation. GSE conf. denotes loans that conform to the GSE's lending guidelines. GDP growth and HPI growth are the growth rates of the U.S. Gross Domestic Product and the House Price Index, respectively. σ interest refers to interest-rate volatility. Credit Spread is the yield difference between AAA and Baa bond indexes. Judicial denotes states that require judicial procedures to foreclose on a mortgage. SRR stands for Statutory Right of Redemption, and denotes states that have statutory redemption laws. R^2 is expressed in percentage and refers to the pseudo R^2 for probit models and the adjusted R^2 for linear models. ρ is the estimated correlation coefficient for the bivariate probit. The asterisks *** refer to the significant coefficients at the 1% significance level.

Model	Two-Stag	e IV Probit	Ľ	DLB Linear Mod	el	Bivariate Probit		
Model	1st Stage	2nd Stage	1st Stage	2nd Stage	2nd Stage	Defeult	Crustah Carry	
Dependent Var.	Default	Switch Serv.	Default	Switch Serv.	Switch Serv.	Default	Switch Serv.	
Instruments								
Income	-0.0007 ***		-0.0002 ***					
Divorce	0.2896 ***		0.2104 ***					
Pr(Default = 1)		0.5334 ***						
Ê(Default)				0.4871 ***	0.1683 ***			
Default					0.3188 ***			
FICO score	-0.0035 ***		-0.0011 ***			-0.0035 ***	-0.0001 ***	
LTV ratio	0.0180 ***	0.0029 ***	0.0051 ***	0.0004 ***	0.0004 ***	0.0180 ***	0.0030 ***	
ARM	0.1212 ***	-0.1867 ***	0.0430 ***	-0.0795 ***	-0.0795 ***	0.1184 ***	-0.1707 ***	
Balloon	0.4129 ***	-0.0225 ***	0.1596 ***	-0.0525 ***	-0.0525 ***	0.4085 ***	0.0582 ***	
No/Low doc.	0.3395 ***	0.1579 ***	0.1062 ***	0.0437 ***	0.0437 ***	0.3416 ***	0.1699 ***	
GSE Conf.	-0.1537 ***	0.0777 ***	-0.0432 ***	0.0699 ***	0.0699 ***	-0.1524 ***	0.0020	
GDP growth	-4.9640 ***	4.4784 ***	-1.8759 ***	3.5329 ***	3.5329 ***	-1.9603 ***	-0.5005 ***	
HPI growth	-7.5731 ***	-5.8147 ***	-2.5375 ***	-0.8476 ***	-0.8476 ***	-7.5398 ***	-7.7918 ***	
σinterest	0.9305 ***	0.6387 ***	0.2736 ***	0.0887 ***	0.0887 ***	1.0688 ***	0.8380 ***	
Credit spread	1.9934 ***	1.1789 ***	0.6298 ***	0.0044 ***	0.0044 ***	1.8957 ***	1.9713 ***	
Judicial	-0.0425 ***	0.0129 ***	-0.0129 ***	0.0066 ***	0.0066 ***	-0.0426 ***	0.0018	
SRR	-0.0844 ***	0.0321 ***	-0.0267 ***	0.0145 ***	0.0145 ***	-0.0851 ***	0.0321 ***	
R ²	11.7	38.0	13.8	31.2	38.2			
ρ						0.59	65 ***	

The last two columns of Table 2 show the results of the bivariate probit model, which estimates the joint occurrence of both events (mortgage default and servicer switch). All regressors remain statistically significant and preserve their expected signs. Most importantly, the models display a statistically significant estimated correlation coefficient of 0.60, which confirms the positive residual relationship between mortgage default and servicer switch. This last result does not test for causality, however.

For robustness purposes, we reproduce the parametric results using (i) a different definition of mortgage default and (ii) a different study period. We use an alternative default definition that identifies a given mortgage in default when it first becomes 60+ days

delinquent (i.e., when reported as the borrower having missed two or more monthly payments). We also consider a pre-crisis sampling period that covers the period from January 2001 to December 2006 with the main objective to immunize the empirical results from potential effects of the financial crisis. As shown in the online Appendix A (Tables A5–A10), our empirical results are robust to these alternatives observed in the literature.

6.3. Causality: Results of the Two-Stage Nonparametric Framework

In the first part of the empirical analysis, we presented the results of the nonparametric kernel density estimation technique. The main goal was to estimate the conditional CDF of mortgage default, $\hat{F}(Y/X, Z)$. Our results show a positive correlation between the decision to switch the servicer of the deal and mortgage default. However, a positive relationship does not necessarily indicate a causal relationship as estimated in the preceding section.

We now propose a new nonparametric two-step instrumental variable estimation procedure to establish a causal relationship between the default and switching variables while considering any potential simultaneous effects. As for the parametric two-step regressions, we exploit the fact that we can interchange the roles of *Y* and *Z* and test F(Z/X, Y) = F(Z/X) instead of testing F(Y/X, Z) = F(Y/X).

In a first step we perform a nonparametric estimation of the conditional density of mortgage default using instrumental variables, and in the second step we consider the nonparametric equivalent of the parametric second-stage regression. We are aware of the literature on nonparametric instrumental variable regressions (Horowitz 2011; Das 2005; Centorrino and Florens 2021). However, the implementation of such approaches to our data is problematic given that the literature proposes nonparametric regression models that are appropriate when at least one variable of interest is continuous. This is not the case in our application because both variables of interest, Y and Z, are binary.

Our two-stage methodology relies on kernel density estimates and can be summarized as follows. In the first stage, we estimate the conditional density function of mortgage default using the KDE technique as described in Section 3. The set of covariates includes exogenous independent variables (e.g., FICO score, LTV ratio, documentation status) along with the two instruments of mortgage default (income growth and divorce rate) of the parametric analysis presented in Section 6.2. The first-stage KDE estimation is represented as follows:

$$\hat{f}(y|x^{c}, x^{d}, v_{1}, v_{2}) = \frac{\frac{1}{n} \sum_{i=1}^{n} K(X_{i}^{c}, x^{c}) \cdot L(X_{i}^{d}, x^{d}) \cdot K(V_{i}, v) \cdot l(Y_{i}, y, \hat{\gamma}_{y})}{\frac{1}{n} \sum_{i=1}^{n} K(X_{i}^{c}, x^{c}) \cdot L(X_{i}^{d}, x^{d}) \cdot K(V_{i}, v)},$$
(11)

where $K(V_i, v) = \prod_{n=1}^{2} \hat{h}_n^{-1} k_h(V_{i,n}, v_n)$ denotes the product kernel function for the twodimensional vector of instrumental variables V_n , $n = \{1, 2\}$, and v_1 and v_2 denote the evaluation points for instruments V_1 and V_2 . Figures A8 and A9 show how the estimated conditional density function of mortgage default varies in relation to our two instruments.

In the second stage, we include the kernel-based estimator of mortgage default as a covariate while estimating the conditional density of the decision to switch mortgage servicers. To simplify the notation, let $Def^+ \equiv I(\hat{f}(y|x^c, x^d, v_1, v_2) > \tau^*)$ and $Def^- \equiv I(\hat{f}(y|x^c, x^d, v_1, v_2) \leq \tau^*)$ define the events where the expected mortgage default probability is high and low, respectively. I(.) refers to an indicator function and τ^* is a fixed threshold, $\tau^* \in [0, 1]$. In our context, Def^+ and Def^- represent the originating lender's expectations of mortgage default based on the information set collected at the time of the original underwriting. As stated above, this two-step instrumental variable estimation procedure allows us to (1) account for potential simultaneity effects, and (2) establish a causal relationship between mortgage default and the decision to switch servicers.

We then perform our information asymmetry test, where the statistic can be formulated as follows:

$$W^* = \sup[\hat{F}(z_i|x_i^c, x_i^d, Def^+) - \hat{F}(z_i|x_i^c, x_i^d, Def^-)]$$
(12)

The test is performed using the nonparametric Kolmogorov–Smirnov (KS) test of the equality of distributions. In a two-sample environment, the test is designed to verify the null hypothesis that both samples are drawn from the same distribution, i.e., both samples have the same distributional shaping parameters. In the context of asymmetric information, the null hypothesis to be tested is that the shape of the conditional distribution of the decision to switch deal servicers is independent from the mortgage default likelihood. If the null hypothesis is respected, this should be interpreted as indicative that the likelihood of mortgage default has no significant impact on the originator's decision to switch servicers.

When applied to empirical data, the KS test statistic could be expressed as the maximum value of the difference between two empirical cumulative distribution functions. The KS test could be written as in (12). When (12) is applied to our context of asymmetric information testing in the mortgage servicing market, the KS test aims to measure the maximum difference between two CDFs of the decision to switch the servicer of the deal (z_i) conditional on a set of conditioning variables (x_i^c, x_i^d, Def) .

To calculate W^* , we first split the empirical estimation sample based on a Def dummy indicator, called the high-default Def^+ and low-default Def^- expectations subsamples. Then, the empirical CDFs are calculated for both subsamples $\hat{F}(z_i|x_i^c, x_i^d, Def^+)$ and $\hat{F}(z_i|x_i^c, x_i^d, Def^-)$. W^* is obtained by comparing these two CDFs. In our application, whether we use the entire sample or randomly selected subsamples, the KS test statistics (W^*) are strictly positive, and all *p*-values are zero, which enables us to reject the null hypothesis of distributional similarities at all significance levels.

For a visualization of the test, one could plot $\hat{F}(z_i|x_i^c, x_i^d, Def^+)$ against $\hat{F}(z_i|x_i^c, x_i^d, Def^-)$ using a continuous conditioning variable as support. Figure 3 provides an illustration where we plot the conditional PDF functions of the decision to switch the deal servicer (Y-axis) versus the FICO score conditioning continuous variable (X-axis). The divergence between the two conditional PDFs (red and blue lines in Figure 3) arises from the conditioning variables Def^+ and Def^- , which denote the expected likelihood of mortgage default (estimated in Stage 1).



Figure 3. Instrumental-variable two-stage nonparametric estimator of switching mortgage default.

Figure 3 shows that the conditional probability of switching the servicer of the deal is a decreasing function of borrower-observed quality. This confirms our previous results using the parametric models where the coefficient of the FICO score was negative and statistically significant. The plot shows divergence between the two lines, each of which is conditioned by the expected likelihood of mortgage default. In the figure, τ^* is set at the mean of 0.37.

In terms of economic interpretability, all other things held constant, if the originating lender expects a high probability of mortgage default from his private information, it is more likely he will sell the underlying servicing right to another servicer. At each point of the X-axis of Figure 3, the probability of switching the servicer conditional on a higher expected probability of default is higher (Y-axis).

The vertical line on Figure 3 refers to a FICO score cut-off point of 660, which is the GSEs frontier for FICO score. This cut-off point represents a rule of thumb established by the GSEs to control mortgage lending in the U.S. market. Following the GSE prudent lending guidelines, a borrower above the 660 thresholds should be attributed a mortgage without restriction, while borrowers falling below should have constrained funding. Keys et al. (2010) exploit a different cut-off point of 620 to investigate the ease of securitization. The authors document a clear shift in the securitization ease around their decision rule.

Figure 3 delivers similar inferences to those by Keys et al. (2010). The figure shows a clear divergence in the conditional probability of switching mortgage servicer given the expected probability (calculated in the first stage estimation) of default is high or low. Nevertheless, this low/high expected default divergence is more pronounced below the 660 thresholds (left-hand side of the vertical line) than above the 660 cut-off. This shift in divergence could be explained by the significance of the quality in the signal provided by the GSE frontier.

Recall that mortgage originators decide whether to sell the MSRs based on both hard and soft information. Also, information asymmetry should be more pronounced in situations where the distinction between soft and hard information is critical. To better interpret our result, let us use the FICO score as a proxy for hard information in Figure 3 (without loss of generality) since it can be observed by a third party. However, high/low expected probability of mortgage default calculated in the first-stage kernel-based estimation contains both sources of information. Mortgages granted for borrowers with a FICO score above the GSE's 660 rule of thumb naturally exhibit a low probability of mortgage default. However, we still observe a discrepancy between hard and soft information.

The two-step instrumental variable nonparametric testing procedure that we proposed establishes a causal relationship between the agent's decision variable Z and the outcome Y. The results strongly suggest that the expected likelihood of mortgage default influences the originator's decision to switch the servicer of the deal, which confirms our hypothesis that second-stage asymmetric information existed in the U.S. mortgage servicing market during our period of analysis.

7. Conclusions

In this paper, we analyze the servicing switching decision in the securitization market. Our main objective is to verify whether information asymmetry between servicer and originator affects mortgage default. Specifically, we investigate whether a first-level originator decision to sell the mortgage servicing rights to a second-level servicer reveals any residual information asymmetry in the mortgage servicing market.

Figure 4 summarizes the key relationships and results of the article. Once the securitization process is implemented and MBSs are sold to investors, a mortgage servicer ensures the management of cash flows between investors and the originator. The originator can manage these cash flows with its own servicer (own servicer) or sell the management activity to an independent servicer (MSR purchaser). We assumed there was asymmetric information about loan quality between the originator and the MSR purchaser. Our goal was to test this assumption by verifying that the average default probability of loans sold by the originator and managed by its own servicer is lower than the average default probability of loans sold by the originator and managed by a new independent servicer.

Our empirical results reveal interesting and important conclusions related to the US mortgage servicing market during our period of analysis. We observe that information asymmetry between originators and servicers influences servicing-switch decisions significantly. The mortgage originator uses its private information advantage to sell riskier loans to the MSR purchaser or new servicer.



Figure 4. Graphical illustration of model and empirical analysis.

This result has important consequences for the securitization market. Recent regulations have introduced a retention provision for originators that use securitization. Since December 2014, originators must keep an economic interest (retention) in the credit risk of the securitized assets (Morgan 2018). Only the original originator must keep the economic interest.

The main goal of the 2014 reform was to increase the amount of skin in the game for originators of asset-backed securities transactions under asymmetric information. Under asymmetric information in the securitization market, originators have less incentive to screen loan applicants (adverse selection) and to monitor loan recipients over the life of the contract (moral hazard), particularly when all potential credit losses are transferred to investors.

Without any retention rules, the originators have incentives to sell bad loans to investors since, with securitization, they are not exposed to the credit risk of these loans. Moreover, they have incentives to sell the servicing management to independent servicers to avoid the administrative expenses associated with borrower defaults and lack of monthly payments.

Risk retention rules were implemented under the Dodd-Frank Act to increase originator incentives. Under these rules, originators must retain exposure to loan defaults. Recent empirical studies have found support for the risk-mitigation hypothesis of risk retention (Begley and Purnanandam 2017; Ashcraft et al. 2019).

These analyses focused on the first-stage asymmetric information problem between lenders and investors. They may have underestimated the effect of asymmetric information in the securitization market by not considering the second-stage information problem between the originator and servicers that is documented in this article.

It would be interesting to investigate how this new rule may impact the type of information asymmetry effect we have measured. Intuitively it should reduce the incentive to pass on bad loans to an independent servicer because with risk retention, fewer bad loans will be securitized. This means that empirical results documenting the positive effect of risk retention in the securitization market should be associated with both forms of asymmetric information problems.

Author Contributions: Conceptualization, H.J. and G.D.; methodology, H.J. and G.D.; software, H.J.; validation, H.J.; formal analysis, H.J.; investigation, H.J. and G.D.; resources, G.D.; data curation, H.J.; writing—original draft preparation, H.J. and G.D.; writing—review and editing, G.D.; visualization, H.J.; supervision, G.D.; project administration, G.D.; funding acquisition, G.D. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by SSHRC Canada.

Data Availability Statement: The original contributions presented in this study are included in the article. Further inquiries can be directed to the corresponding author(s).

Conflicts of Interest: The authors declare no conflit of interest.

192
12,
2024,
Risks !

Appendix A

Table A1. Variable definition and source.

Name	Type	Description	Source
Switch Servicer	Binary	Denotes the decision of the originating lender to sell or to retain the mortgage servicing right of a given loan. Takes the value of 1 if the originator decides to sell the underlying MSR and 0 if he retains the MSR and continues servicing the loan.	MBSData
Default	Binary	Denotes mortgage default. Takes the value of 1 if the borrower of a given mortgage misses three or more consecutive monthly payments (i.e., when the mortgage status is first labeled as 90+ days delinquent).	MBSData
FICO score	Continuous	The borrower's FICO score created and calculated by the Fair Isaac Corporation. It measures the credit quality of borrowers by taking into account an individual's payment history, length of credit history, current level of indebtedness, and types of credit used by the borrower.	MBSData
FICO660	Binary	Takes the value of 1 if the borrower's FICO score is above 660 and 0 otherwise. In general, a FICO score above 660 indicates that the individual has a good credit history.	MBSData
LTV	Continuous	The loan-to-value ratio calculated as the percentage of the first-lien mortgage to the total value of the property. It is one of the key risk factors used by U.S. lenders when qualifying borrowers for a mortgage. A high LTV ratio mirrors a loan with a low down payment for which the borrower has little equity stake in the property.	MBSData
LTV80	Binary	Takes the value of 1 if the LTV ratio is equal to or higher than 80%.	MBSData
DTI	Continuous	The debt-to-income ratio calculated as the fraction of monthly mortgage payments to the borrower's monthly income. DTI measures the borrower's ability to honor periodic debt payments as it compares debt payments to the borrower's income.	MBSData
No/Low doc.	Binary	Takes the value of 1 if the documentation level is labelled "missing" or "low", and 0 otherwise. No- or low documentation mortgages designate loans for which the lender did not gather a sufficient level of information on the borrower's reliability and credit worthiness.	MBSData
ln Amount	Continuous	The natural logarithm of the initial balance of the mortgage. Does not include either interest, taxes, or fees.	MBSData
Interest	Continuous	The interest rate was initially applied at the time of original underwriting. Higher interest rates usually reflect loans granted to borrowers with inferior credit quality, which increase their monthly debt payments.	MBSData
ARM	Binary	Takes the value of 1 if the loan type is an adjustable-rate mortgage and 0 if fixed-rate mortgage. ARM indicates whether the interest rate of a given mortgage is fluctuating over time based on a benchmark index plus an additional spread, called an ARM margin.	MBSData
ARM margin	Continuous	A fixed component added to the interest rate for ARM mortgages. The margin is constant throughout the lifetime of the mortgage, while the benchmark index fluctuates over time according to general market conditions.	MBSData
Balloon	Binary	Takes the value of 1 if the mortgage has a balloon payment structure, 0 otherwise. Balloon mortgagors make only interest payments during the lifetime of the loan. At the term end, the borrower repays the entire principal at once.	MBSData

192	
12,	
2024,	
Risks	

Name	Tvne	Description	Source
GSE conforming	Binary	Takes the value of 1 if the lender follows the GSEs' lending guidelines and 0 otherwise. Following the GSEs' recommendations, we classify a mortgage as conforming if the borrower's FICO score is above 660 and the loan amount is below the conforming loan limit in place at the time of origination and the LTV is either less than 80% or the loan has private mortgage insurance in the case that the LTV ratio is above 80%. Since conforming loans meet the GSEs lending standards, the conforming dummy variable indicates whether the mortgage was eligible to be sold to the GSEs at origination.	MBSData
Subprime	Binary	Denotes subprime mortgages. A mortgage is labelled "Subprime" at origination if the borrower's FICO score is lower than 580 or the LTV ratio is higher than 90%.	MBSData
Prime	Binary	Denotes prime mortgages. A mortgage is considered as "Prime" if the borrower's FICO score is higher than 660 and the LTV ratio is lower than 80%.	MBSData
Prep. Penalty	Binary	Equals to 1 if the mortgage contract includes a prepayment penalty clause and 0 otherwise. Accordingly, the borrower will pay a penalty if he chooses to pre-pay the loan within a certain time period. The penalty is based on the remaining mortgage balance and the number of months worth of interest.	MBSData
Purchase	Binary	Takes the value of 1 if the loan purpose is labeled "Purchase" a property and 0 otherwise.	MBSData
Refin. cash-out	Binary	Equals to 1 if the loan is granted for the purpose of refinancing an existing loan with "cash-out". A cash-out refinance mortgage is a new loan in which the amount is greater than the existing mortgage amount, which will be refinanced. Since the borrower refinances for more than the amount owed, he/she takes the difference in cash.	MBSData
Refin. no cash-out	Binary	Equals to 1 if the loan is granted for the purpose of refinancing an existing loan with "no-cash-out". A no-cash-out refinance mortgage is a new loan in which the amount is equal to or lower than the existing mortgage amount. The main purpose of such loans is usually to lower the interest rate charge on the loan.	MBSData
Service fee	Continuous	The servicing fee that the servicer of the deal charges as compensation for costs he bears. It is expressed as a fixed percentage of the declining balance of the mortgage.	MBSData
Age at default	Continuous	The age-at-default is measured as the total number of months since origination when the default is first recorded.	MBSData
Default N	Binary	Denoting the fraction of mortgages that default within N months since origination.	MBSData
Income	Continuous	The annual growth rate of personal income, which is defined as an individual's total earnings from wages, investment interest, and other sources. The seasonally unadjusted U.S. real disposable (after deducting tax) personal income data is retrieved from the US. Bureau of Economic Analysis' web site.	bea.gov
Divorce	Continuous	The annual divorce rate calculated as the ratio of the number of marriages contracted and ended in divorce to the number of all marriages contracted in the same year. The divorce rate is commonly used as an indicator of social stress in the society. The seasonally unadjusted divorce rate is retrieved from the U.S. Census Bureau's web site.	census.gov
GDP growth	Continuous	The annual growth rate of the U.S. Real Gross Domestic Product. The real GDP is collected from the Federal Reserve Bank of St. Louis' web site.	stlouisfed.org

192
12,
2024,
Risks

	TUDIC UT		
Name	Type	Description	Source
HPI growth	Continuous	The annual growth rate of the House Price Index for the U.S. We use the seasonally unadjusted purchase-only HPI index retrieved from the Federal Reserve Bank of St. Louis' web site.	stlouisfed.org
σ interest	Continuous	The interest rate volatility calculated as the volatility on the 1-Year Treasury Constant Maturity Rate over the 24 months before origination. The monthly seasonally unadjusted treasury rate is collected from the Federal Reserve Bank of St. Louis' web site.	stlouisfed.org
Credit spread	Continuous	The yield spread between AAA and Baa bond indexes. It is calculated as the interest rate difference between Moody's Aaa and Baa Corporate Bond Yields. Both variables are seasonally unadjusted, recorded on a monthly basis, and retrieved from the Federal Reserve Bank of St. Louis' web site.	stlouisfed.org
Judicial	Binary	Takes the value of 1 if the state laws require judicial procedures to foreclose on a mortgage and 0 if not. The variable is compiled based on information from the National Center for State Courts' web site.	ncsc.org
SRR	Binary	Stands for Statutory Right of Redemption and takes the value of 1 if the state has statutory redemption laws. The variable is compiled based on information from the National Center for State Courts' web site.	ncsc.org

Table A2. Summary statistics by origination year. The table reports summary statistics for the sample of U.S. mortgages originated over the period from January 2000 to December 2013. The mortgages have been securitized through the non-agency channel. The first row reports statistics over the 2000–2013 study period for the total sample of 5,591,353 distinct mortgages, while the next rows report statistics by origination year. The first two columns, Volume (in %) and Volume (in \$B), refer to the total origination volume expressed in percentage of the total sample and in US\$ billions, respectively. FICO score abbreviates the borrower's Fair Isaac Corporation score attributed at origination. FICO.660 denotes the fraction of loans granted to borrowers with FICO scores higher than 660. LTV abbreviates the initial loan-to-value ratio. LTV.80 denotes the fraction of loans with LTV ratios higher than 80%. DTI stands for the debt-to-income ratio. No/Low doc. indicates whether the origination. Balloon denotes balloon payment mortgages. ARM and ARM margin denote adjustable-rate mortgages and the corresponding margin. GSE conf. denotes the fraction of loans that conform to the Government-Sponsored Enterprises' prudent lending guidelines. Prep. Penalty measures the fraction of loans with prepayment penalties.

Origination Year	Volume (in %)	Volume (in \$B)	FICO Score	FICO. 660	LTV Ratio	LTV. 80	DTI	No/Low Doc.	Interest Rate	Balloon	ARM	ARM Margin	GSE Conf.	Prep. Penalty
All period	100.0	1509.1	657.12	0.48	76.93	0.60	38.65	0.47	6.97	0.06	0.63	5.00	0.17	0.49
2000	1.05	8.87	615.49	0.31	78.20	0.62	38.65	0.34	10.08	0.07	0.34	6.13	0.17	0.41
2001	2.47	32.07	648.33	0.47	76.87	0.56	37.74	0.29	8.56	0.03	0.36	6.09	0.23	0.33
2002	5.74	69.08	644.97	0.42	77.47	0.58	37.84	0.33	7.92	0.02	0.54	5.92	0.21	0.38
2003	11.46	170.89	670.12	0.56	75.14	0.51	36.95	0.38	6.60	0.01	0.49	5.18	0.25	0.31
2004	16.93	232.68	657.75	0.49	77.60	0.60	36.81	0.44	6.30	0.00	0.70	4.75	0.19	0.52
2005	27.28	411.36	658.81	0.49	76.97	0.62	38.33	0.51	6.51	0.02	0.69	4.91	0.16	0.53
2006	27.11	422.92	650.45	0.44	77.44	0.63	39.90	0.52	7.44	0.15	0.66	5.07	0.12	0.57
2007	7.79	153.39	668.92	0.56	75.92	0.56	39.17	0.57	7.32	0.12	0.52	4.50	0.15	0.47
2008	0.02	0.60	717.06	0.80	73.25	0.44	36.59	0.40	7.16	0.03	0.48	3.13	0.03	0.17
2009	0.00	0.21	774.60	1.00	53.11	0.06	36.00	0.30	4.79	0.00	0.83	2.04	0.03	0.00
2010	0.01	0.43	772.33	1.00	61.78	0.17	32.48	0.02	4.93	0.00	0.08	1.64	0.01	0.21
2011	0.02	1.17	770.62	1.00	66.60	0.23	32.98	0.17	4.72	0.00	0.06	1.83	0.01	0.16
2012	0.06	2.79	773.08	1.00	66.42	0.20	34.00	0.03	4.06	0.00	0.02	2.25	0.00	0.13
2013	0.06	2.67	771.14	1.00	66.24	0.19	30.80	0.00	3.91	0.00	0.01	2 53	0.00	0.01

Table A3. Summary statistics by loan type and status. The table reports summary statistics for the sample of 5,591,353 U.S. mortgages originated over the period from January 2000 to December 2013. The mortgages have been securitized through the non-agency channel. The table breaks down the sample by payment type (FRM vs. ARM), loan type (prime vs. subprime), financial crisis era (before vs. after), default status, and servicer switch status. FICO score abbreviates the borrower's Fair Isaac Corporation score at origination. FICO.660 denotes the fraction of loans granted to borrowers with a FICO score higher than 660. LTV abbreviates the initial loan-to-value ratio. LTV.80 denotes the fraction of loans with LTV ratios greater than 80%. DTI stands for the debt-to-income ratio. No/Low doc. indicates whether the originator collected either no or low documentation. Interest rate is the coupon rate applied at origination. Balloon denotes balloon payment mortgages. ARM and ARM margin denote adjustable-rate mortgages and the corresponding margin. Subprime and prime are subprime loan classifiers. GSE conf. denotes the fraction of loans conforming to the GSEs' lending guidelines. Prep. Penalty indicates the fraction of mortgages with a prepayment penalty. Service fee is the mortgage servicer fee expressed as a percentage of the remaining balance. Switch servicer indicates the fraction of mortgages for which the originator switched the servicer of the deal. Default denotes the fraction of mortgages in default. Age at default is the average age of defaulting mortgages. Default 12, Default 18, and Default 24 refer to the fraction of loans defaulting within 12, 18, and 24 months since origination, respectively.

	A 11	Payment Type		Loan Type		Financial Crisis		Default		Switch Servicer	
		FRM	ARM	Prime	Subprime	Before	After	No	Yes	No	Yes
FICO score	657.12	678.00	644.84	730.93	634.87	655.92	671.02	669.62	635.77	660.62	654.23
LTV	0.48 76.93	73.89	0.41 78.73	63.48	0.33 80.99	0.48 77.04	0.57 75.72	0.55 74.86	80.48	0.51 76.49	0.46 77.30
LTV.80	0.60	0.48	0.67	0.00	0.78	0.60	0.55	0.53	0.72	0.58	0.61

				T	T	. ·	10.1.1	D (. 1.	Switch Someicon		
	A11	Payme	nt Type	Loa	n Type	Financia	II Crisis	Der	ault	Switch	Servicer	
		FRM	ARM	Prime	Subprime	Before	After	No	Yes	No	Yes	
DTI	38.65	37.64	39.08	35.71	39.10	38.60	39.09	37.63	39.91	38.02	38.95	
No/Low doc.	0.47	0.46	0.48	0.63	0.43	0.46	0.55	0.45	0.50	0.49	0.45	
Interest rate	6.97	7.10	6.89	5.57	7.39	6.94	7.26	6.71	7.41	6.86	7.05	
Balloon	0.06	0.04	0.08	0.01	0.08	0.06	0.12	0.03	0.11	0.04	0.08	
ARM	0.63	0.00	1.00	0.45	0.68	0.64	0.52	0.59	0.70	0.59	0.66	
ARM margin	5.00		5.00	2.86	5.42	5.03	4.50	4.80	5.27	4.93	5.04	
Subprime	0.77	0.66	0.83	0.00	1.00	0.77	0.72	0.70	0.89	0.75	0.78	
Prime	0.23	0.34	0.17	1.00	0.00	0.23	0.28	0.30	0.11	0.25	0.22	
GSE Conf.	0.17	0.25	0.12	0.56	0.05	0.17	0.15	0.21	0.10	0.19	0.15	
Prep. Penalty	0.49	0.34	0.58	0.24	0.57	0.50	0.46	0.42	0.63	0.49	0.50	
Purchase	0.37	0.30	0.42	0.22	0.42	0.38	0.30	0.36	0.40	0.36	0.39	
Refin. cash-out	0.47	0.49	0.45	0.46	0.47	0.46	0.51	0.46	0.47	0.49	0.45	
Refin. no cash-out	0.16	0.21	0.13	0.31	0.11	0.15	0.19	0.18	0.12	0.15	0.16	
Service fee	0.44	0.38	0.47	0.33	0.47	0.44	0.39	0.42	0.46	0.41	0.46	
Switch servicer	0.55	0.50	0.58	0.52	0.56	0.56	0.44	0.18	0.50	0.00	1.00	
Default	0.37	0.30	0.41	0.18	0.43	0.35	0.54	0.00	1.00	0.26	0.62	
Age at default	36.64	45.25	32.98	47.72	35.21	37.41	30.81		36.64	38.07	35.47	
Default 12	0.11	0.06	0.12	0.03	0.12	0.10	0.12		0.11	0.09	0.12	
Default 18	0.23	0.15	0.26	0.08	0.24	0.22	0.26		0.23	0.20	0.24	
Default 24	0.35	0.24	0.40	0.15	0.38	0.34	0.44		0.35	0.32	0.37	

Table A3. Cont.

Table A4. Results of the probit model. The table reports estimation results of the parametric probit regressions. The sample includes 5,591,353 mortgages originated over the period from January 2000 to December 2013. The dependent variable, Default, is a dummy variable denoting mortgage default (i.e., when a mortgage is labelled as +90 days delinquent). FICO score is the borrower's Fair Isaac Corporation score attributed at origination. LTV ratio denotes the initial loan-to-value ratio. ARM stands for adjustable-rate mortgages. Balloon refers to balloon payment mortgages. No/Low doc. indicates whether the originator collected no/low-level documentation. GSE conf. denotes mortgages that conform to the GSE's lending guidelines. GDP growth and HPI growth are growth rates of the U.S. Gross Domestic Product and the House Price Index, respectively. σ interest refers to interest-rate volatility. Credit Spread is the yield difference between AAA and Baa bond indexes. State FE specification controls for state fixed effects using state dummies. Judicial indicates whether the state requires judicial procedures to foreclose on a mortgage. SRR stands for Statutory Right of Redemption and denotes states that have statutory redemption laws. The Pseudo R^2 is expressed in percentage. Wald denotes the *p*-value of the Wald test for the null hypothesis that all coefficients are jointly equal to zero. LR refers to the *p*-value of the likelihood ratio test for the null hypothesis based on configuration II. The asterisks *** refer to significance levels of 1%.

Configuration	Ι	Π	III	IV	v	VI	VII	IIX	IX
A. Fundamental loan	and borrower ch	naracteristics							
FICO score	-0.0034 ***	-0.0034 ***	-0.0034 ***	-0.0034 ***	-0.0034 ***	-0.0034 ***	-0.0034 ***	-0.0034 ***	-0.0034 ***
LTV ratio	0.0169 ***	0.0172 ***	0.0170 ***	0.0171 ***	0.0169 ***	0.0175 ***	0.0170 ***	0.0172 ***	0.0179 ***
ARM	0.0980 ***	0.1324 ***	0.1290 ***	0.1064 ***	0.0866 ***	0.0755 ***	0.0940 ***	0.0911 ***	0.1206 ***
Balloon	0.6336 ***	0.5681 ***	0.5770 ***	0.5887 ***	0.6384 ***	0.6373 ***	0.6344 ***	0.6264 ***	0.4146 ***
No/Low doc.	0.3726 ***	0.3742 ***	0.3741 ***	0.3707 ***	0.3673 ***	0.3602 ***	0.3721 ***	0.3690 ***	0.3396 ***
GSE Conf.	-0.1939 ***	-0.1914 ***	-0.1895 ***	-0.1920 ***	-0.1905 ***	-0.1959 ***	-0.1918 ***	-0.1910 ***	-0.1567 ***
B. Economic general	conditions								
GDP growth		-14.808 ***							-1.9725 ***
C. Housing market co	onditions								
HPI growth			-3.4660 ***						-7.6275 ***
D. Bond market cond	itions								
σ interest				0.4669 ***					1.0679 ***
Credit spread					0.3561 ***				1.8900 ***
E. State legal structur	e								
State FE						Yes			
Judicial							-0.0464 ***		-0.0421 ***
SRR								-0.0868 ***	-0.0853 ***
Intercept	0.2878 ***	0.6870 ***	0.5697 ***	-0.1014 ***	0.6244 ***	-0.1253 ***	0.3277 ***	0.3385 ***	1.9433 ***
Pseudo R ²	8.40	9.10	8.82	9.04	8.53	9.39	8.43	8.46	11.60
Log-likelihood	$-3.37 imes10^{6}$	$-3.35 imes10^{6}$	$-3.36 imes10^6$	$-3.35 imes10^{6}$	$-3.37 imes10^{6}$	$-3.34 imes10^6$	$-3.37 imes10^{6}$	$-3.37 imes10^{6}$	$-3.25 imes10^{6}$
Wald <i>p</i> -value									0.00
LR <i>p</i> -value									0.00

Table A5. Results of the probit model using the +60 days default definition. The table reports estimation results of the parametric probit regressions. The sample includes 5,591,353 mortgages originated over the period from January 2000 to December 2013. The dependent variable, Default, is a dummy variable denoting mortgage default (i.e., when a mortgage is labelled as +60 days delinquent). FICO score is the borrower's Fair Isaac Corporation score attributed at origination. LTV ratio denotes the initial loan-to-value ratio. ARM stands for adjustable-rate mortgages. Balloon refers to balloon payment mortgages. No/Low doc. indicates whether the originator collected no/low-level documentation. GSE conf. denotes mortgages that conform to the GSE's lending guidelines. GDP growth and HPI growth are growth rates of the U.S. Gross Domestic Product and the House Price Index, respectively. σ interest refers to interest-rate volatility. Credit Spread is the yield difference between AAA and Baa bond indexes. State FE specification controls for state fixed effects using state dummies. Judicial indicates whether the state requires judicial procedures to foreclose on a mortgage. SRR stands for Statutory Right of Redemption and denotes the *p*-value of the Wald test for the null hypothesis based on configuration II. The asterisks *** refer to significance levels of 1%.

Configuration	Ι	II	III	IV	v	VI	VII	IIX	IX
A. Fundamental loar	n and borrower cl	naracteristics							
FICO score	-0.0036 ***	-0.0036 ***	-0.0036 ***	-0.0035 ***	-0.0036 ***	-0.0036 ***	-0.0036 ***	-0.0036 ***	-0.0037 ***
LTV ratio	0.0164 ***	0.0166 ***	0.0165 ***	0.0165 ***	0.0164 ***	0.0168 ***	0.0165 ***	0.0166 ***	0.0173 ***
ARM	0.0773 ***	0.1117 ***	0.1087 ***	0.0858 ***	0.0660 ***	0.0587 ***	0.0735 ***	0.0713 ***	0.1010 ***
Balloon	0.6303 ***	0.5647 ***	0.5731 ***	0.5852 ***	0.6350 ***	0.6368 ***	0.6310 ***	0.6240 ***	0.4114 ***
No/Low doc.	0.3740 ***	0.3758 ***	0.3758 ***	0.3721 ***	0.3687 ***	0.3631 ***	0.3736 ***	0.3709 ***	0.3417 ***
GSE Conf.	-0.1844 ***	-0.1817 ***	-0.1798 ***	-0.1825 ***	-0.1810 ***	-0.1871 ***	-0.1823 ***	-0.1819 ***	-0.1475 ***
B. Economic general	conditions								
GDP growth		-14.788 ***							-1.8866 ***
C. Housing market c	onditions								
HPI growth			-3.5125 ***						-7.6803 ***
D. Bond market cond	ditions								
σ interest				0.4624 ***					1.0581 ***
Credit spread					0.3491 ***				1.8732 ***
F State legal structu	re								
State FE						Yes			
Iudicial						100	-0.0447 ***		-0.0412 ***
SRR								-0.0752 ***	-0.0737 ***
Intercept	0.5435 ***	0.9450 ***	0.8307 ***	0.1598 ***	0.8735 ***	0.1316 ***	0.5821 ***	0.5876 ***	2.1942 ***
Pseudo R ²	8.50	9.19	8.92	9.12	8.62	9.43	8.52	8.54	11.60
Log-likelihood	$-3.41 imes10^{6}$	$-3.39 imes10^{6}$	$-3.40 imes10^{6}$	$-3.39 imes10^{6}$	$-3.41 imes10^{6}$	$-3.38 imes10^{6}$	$-3.42 imes10^{6}$	$-3.42 imes10^{6}$	$-3.30 imes10^{6}$
Wald <i>p</i> -value									0.00
LR <i>p</i> -value									0.00

Table A6. Results of the probit model using the 2001–2006 period. The table reports estimation results of the parametric probit regressions. The sample includes 5,086,938 mortgages originated over the period from January 2001 to December 2006. The dependent variable, Default, is a dummy variable denoting mortgage default (i.e., when a mortgage is labelled as +90 days delinquent). FICO score is the borrower's Fair Isaac Corporation score attributed at origination. LTV ratio denotes the initial loan-to-value ratio. ARM stands for adjustable-rate mortgages. Balloon refers to balloon payment mortgages. No/Low doc. indicates whether the originator collected no/low-level documentation. GSE conf. denotes mortgages that conform to the GSE's lending guidelines. GDP growth and HPI growth are growth rates of the U.S. Gross Domestic Product and the House Price Index, respectively. σ interest refers to interest-rate volatility. Credit Spread is the yield difference between AAA and Baa bond indexes. State FE specification controls for state fixed effects using state dummies. Judicial indicates whether the state requires judicial procedures to foreclose on a mortgage. SRR stands for Statutory Right of Redemption and denotes the *p*-value of the Wald test for the null hypothesis that all coefficients are jointly equal to zero. LR refers to the *p*-value of the likelihood ratio test for the null hypothesis based on configuration II. The asterisks *** refer to significance levels of 1%.

Configuration	Ι	II	III	IV	V	VI	VII	IIX	IX
A. Fundamental loan FICO score	and borrower cl -0.0034 ***	haracteristics -0.0033 ***	-0.0034 ***	-0.0033 ***	-0.0034 ***	-0.0034 ***	-0.0034 ***	-0.0034 ***	-0.0035 ***

Configuration	Ι	II	III	IV	V	VI	VII	IIX	IX
LTV ratio ARM Balloon No/Low doc. GSE Conf.	0.0169 *** 0.0978 *** 0.6464 *** 0.3444 *** -0.1921 ***	0.0169 *** 0.1157 *** 0.6156 *** 0.3476 *** -0.1946 ***	0.0168 *** 0.1028 *** 0.6361 *** 0.3455 *** -0.1928 ***	0.0172 *** 0.1191 *** 0.5640 *** 0.3398 *** -0.1834 ***	0.0169 *** 0.0820 *** 0.6522 *** 0.3375 *** -0.1869 ***	0.0170 *** 0.0824 *** 0.6571 *** 0.3379 *** -0.1979 ***	0.0170 *** 0.0942 *** 0.6469 *** 0.3440 *** -0.1902 ***	0.0170 *** 0.0930 *** 0.6412 *** 0.3417 *** -0.1901 ***	0.0178 *** 0.0885 *** 0.4378 *** 0.3064 *** -0.1474 ***
B. Economic general GDP growth	conditions	-8.4588 ***							11.941 ***
C. Housing market co HPI growth	onditions		-0.7645 ***						-6.5539 ***
D. Bond market cond σ interest Credit spread	litions			0.6250 ***	0.4258 ***				1.4068 *** 1.8676 ***
E. State legal structur State FE Judicial SRR	e					Yes	-0.0413 ***	-0.0621 ***	-0.0402 *** -0.0545 ***
Intercept	0.2735 ***	0.4876 ***	0.3351 ***	-0.2557 ***	0.6864 ***	-0.1608 ***	0.3083 ***	0.3090 ***	1.1864 ***
Pseudo R ² Log-likelihood Wald <i>p</i> -value LR <i>p</i> -value	$8.22 \\ -3.03 imes 10^6$	$8.41 - 3.03 imes 10^6$	$8.24 - 3.03 imes 10^6$	$9.37 \\ -2.99 imes 10^{6}$	$8.42 - 3.03 imes 10^6$	$9.21 - 3.00 imes 10^6$	$8.24 - 3.03 imes 10^6$	$8.25 \\ -3.03 imes 10^6$	$\begin{array}{r} 11.50 \\ -2.92 \times 10^6 \\ 0.00 \\ 0.00 \end{array}$

Table A6. Cont.

Table A7. Results of the probit model using the 2001–2006 period and +60 days default definition. The table reports estimation results of the parametric probit regressions. The sample includes 5,086,938 mortgages originated over the period from January 2001 to December 2006. The dependent variable, Default, is a dummy variable denoting mortgage default (i.e., when a mortgage is labelled as +60 days delinquent). FICO score is the borrower's Fair Isaac Corporation score attributed at origination. LTV ratio denotes the initial loan-to-value ratio. ARM stands for adjustable-rate mortgages. Balloon refers to balloon payment mortgages. No/Low doc. indicates whether the originator collected no/low-level documentation. GSE conf. denotes mortgages that conform to the GSE's lending guidelines. GDP growth and HPI growth are growth rates of the U.S. Gross Domestic Product and the House Price Index, respectively. σ interest refers to interest-rate volatility. Credit Spread is the yield difference between AAA and Baa bond indexes. State FE specification controls for state fixed effects using state dummies. Judicial indicates whether the state requires judicial procedures to foreclose on a mortgage. SRR stands for Statutory Right of Redemption and denotes states that have statutory redemption laws. The Pseudo R2 is expressed in percentage. Wald denotes the *p*-value of the Wald test for the null hypothesis that all coefficients are jointly equal to zero. LR refers to the *p*-value of the likelihood ratio test for the null hypothesis based on configuration II. The asterisks *** refer to significance levels of 1%.

A. Fundamental loan and borrower characteristics -0.0036 **** -0.0036 **** -0.0036 **** -0.0036 **** -0.0036 **** -0.0036 **** -0.0036 **** -0.0036 **** -0.0036 **** -0.0036 **** 0.0163 **** 0.0163 **** 0.0163 **** 0.0163 **** 0.0163 **** 0.0164 **** 0.0164 **** 0.0164 **** 0.0164 **** 0.0164 **** 0.0164 **** 0.0164 **** 0.0164 **** 0.0164 **** 0.0164 **** 0.0164 **** 0.0164 **** 0.0164 **** 0.0175 **** 0.0075 **** 0.0075 **** 0.0075 **** 0.0075 **** 0.0075 **** 0.0075 **** 0.0075 **** 0.0075 **** 0.0075 **** 0.0075 **** 0.0075 **** 0.0075 **** 0.0075 **** 0.0075 **** 0.0436 **** 0.0619 **** 0.0619 **** 0.0611 **** 0.0611 **** 0.0611 **** 0.0611 **** 0.0611 **** 0.0435 **** 0.3407 **** 0.3407 **** 0.3405 **** 0.3405 **** 0.3405 **** 0.3405 **** 0.3405 **** 0.3405 **** 0.3405 **** 0.3405 **** 0.3405 **** 0.3405 **** 0.3405 **** 0.3405 **** 0.3405 **** 0.3405 **** 0.3405 **** 0.3405 **** 0.3405 **** 0.3405 ****	Configuration	Ι	II	III	IV	V	VI	VII	IIX	IX
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	A. Fundamental loan and borrower characteristics									
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	FICO score	-0.0036 ***	-0.0036 ***	-0.0036 ***	-0.0035 ***	-0.0036 ***	-0.0036 ***	-0.0036 ***	-0.0036 ***	-0.0037 ***
ARM 0.0775^{***} 0.0926^{****} 0.0926^{****} 0.0619^{****} 0.0660^{****} 0.0740^{***} 0.0735^{***} 0.0694^{****} Balloon 0.6409^{****} 0.6303^{****} 0.552^{****} 0.6465^{****} 0.6414^{****} 0.06144^{****} 0.06144^{****} 0.3407^{****} 0.3407^{****} 0.3407^{****} 0.3407^{****} 0.3407^{****} 0.3407^{****} 0.3407^{****} 0.3407^{****} 0.3408^{****} 0.348^{****} 0.348^{****} 0.348^{****} 0.183^{****} 0.183^{****} $0.183^$	LTV ratio	0.0163 ***	0.0164 ***	0.0163 ***	0.0166 ***	0.0163 ***	0.0163 ***	0.0164 ***	0.0165 ***	0.0172 ***
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	ARM	0.0775 ***	0.0954 ***	0.0826 ***	0.0986 ***	0.0619 ***	0.0660 ***	0.0740 ***	0.0735 ***	0.0694 ***
No/Low doc. $0.3452 ***$ $0.3466 ***$ $0.3467 ***$ $0.3383 ***$ $0.3402 ***$ $0.3448 ***$ $0.3431 ***$ $0.3080 ***$ GSE Conf. $-0.1820 ***$ $-0.1827 ***$ $-0.1733 ***$ $-0.1769 ***$ $-0.1885 ***$ $-0.1803 ***$ $-0.5161 ***$ $-0.5161 ***$ $-0.5161 ***$ $-0.5161 ***$ $1.3822 ***$ $1.3822 ***$ $1.3822 ***$ $1.3822 ***$ $1.3820 ***$ $-0.00391 ***$ $-0.0038 ***$	Balloon	0.6409 ***	0.6099 ***	0.6303 ***	0.5582 ***	0.6465 ***	0.6543 ***	0.6414 ***	0.6367 ***	0.4336 ***
GSE Conf. -0.1820 *** -0.1827 *** -0.1733 *** -0.1769 *** -0.1885 *** -0.1803 *** -0.5161 *** D. Bond market conditions	No/Low doc.	0.3452 ***	0.3486 ***	0.3465 ***	0.3407 ***	0.3383 ***	0.3402 ***	0.3448 ***	0.3431 ***	0.3080 ***
B. Economic general conditions GDP growth -8.4972 *** 11.542 *** C. Housing market conditions HPI growth -0.7942 *** -6.5161 *** D. Bond market conditions or interest Credit spread -0.7942 *** -6.5161 *** E. State legal structure State FE Judicial SRR -0.6198 *** 0.4142 *** 1.3822 *** Intercept 0.5317 *** 0.7481 *** 0.5958 *** 0.0110 0.9336 *** 0.1011 *** 0.5649 *** 0.5050 *** 1.4420 ***	GSE Conf.	-0.1820 ***	-0.1845 ***	-0.1827 ***	-0.1733 ***	-0.1769 ***	-0.1885 ***	-0.1803 ***	-0.1803 ***	-0.1382 ***
GDP growth 8.4972 *** 11.542 *** C. Housing market conditions HPI growth -0.7942 *** -6.5161 *** D. Bond market conditions or interest Credit spread 0.6198 *** -6.5161 *** E. State legal structure State FE Judicial SRR Ves -0.0391 *** Intercept 0.5317 *** 0.7481 *** 0.0110 0.9336 *** 0.1011 *** 0.5649 *** 0.5605 *** 1.4420 ***	B. Economic general of	conditions								
C. Housing market conditions HPI growth -0.7942 *** -6.5161 *** D. Bond market conditions or interest Credit spread 0.6198 *** 0.4142 *** 1.3822 *** E. State legal structure State FE Judicial SRR Yes -0.0391 *** -0.0398 *** Intercept 0.5317 *** 0.7481 *** 0.5958 *** 0.0110 0.9336 *** 0.1011 *** 0.5649 *** 0.5605 *** 1.4420 ***	GDP growth		-8.4972 ***							11.542 ***
HPI growth -0.7942 *** -6.5161 *** D. Bond market conditions	C. Housing market co	onditions								
D. Bond market conditions 0.6198 *** 0.6198 *** 1.3822 *** or interest Credit spread 0.4142 *** 1.3820 *** 1.3820 *** E. State legal structure State FE Judicial SRR -0.0391 *** -0.0391 *** -0.0391 *** Intercept 0.5317 *** 0.7481 *** 0.5958 *** 0.0110 0.9336 *** 0.1011 *** 0.5649 *** 0.5605 *** 1.4420 ***	HPI growth			-0.7942 ***						-6.5161 ***
σ interest Credit spread 0.6198 *** 1.3822 *** E. State legal structure State FE Judicial SRR Yes -0.0391 *** Intercept 0.5317 *** 0.7481 *** 0.5958 *** 0.0110 0.9336 *** 0.1011 *** 0.5649 *** 0.5605 ***	D. Bond market cond	itions								
Credit spread 0.4142 *** 1.8360 *** E. State legal structure State FE Judicial SRR Yes -0.0391 *** -0.0391 *** Intercept 0.5317 *** 0.7481 *** 0.5958 *** 0.0110 0.9336 *** 0.1011 *** 0.5649 *** 0.5605 *** 1.4420 ***	σ interest				0.6198 ***					1.3822 ***
E. State legal structure State FE Judicial SRR	Credit spread					0.4142 ***				1.8360 ***
State FE Judicial SRR Yes -0.0391 *** -0.0391 *** -0.0388 *** Intercept 0.5317 *** 0.7481 *** 0.5958 *** 0.0110 0.9336 *** 0.1011 *** 0.5649 *** 0.5605 *** 1.4420 ***	E. State legal structur	e								
Judicial SRR -0.0391*** -0.0391*** -0.0388*** Intercept 0.5317*** 0.7481*** 0.5958*** 0.0110 0.9336*** 0.1011*** 0.5649*** 0.5605*** 1.4420***	State FE						Yes			
SRR -0.0502 *** -0.0428 *** Intercept 0.5317 *** 0.7481 *** 0.5958 *** 0.0110 0.9336 *** 0.1011 *** 0.5649 *** 0.5605 *** 1.4420 ***	Judicial							-0.0391 ***		-0.0388 ***
Intercept 0.5317 *** 0.7481 *** 0.5958 *** 0.0110 0.9336 *** 0.1011 *** 0.5649 *** 0.5605 *** 1.4420 ***	SRR								-0.0502 ***	-0.0428 ***
	Intercept	0.5317 ***	0.7481 ***	0.5958 ***	0.0110	0.9336 ***	0.1011 ***	0.5649 ***	0.5605 ***	1.4420 ***
Pseudo R ² 8.31 8.49 8.32 9.43 8.49 9.25 8.32 8.33 11.51	Pseudo R ²	8.31	8.49	8.32	9.43	8.49	9.25	8.32	8.33	11.51
$\textbf{Log-likelihood} \qquad -3.08 \times 10^{6} \qquad -3.07 \times 10^{6} \qquad -3.08 \times 10^{6} \qquad -3.08 \times 10^{6} \qquad -3.04 \times 10^{6} \qquad -3.07 \times 10^{6} \qquad -3.04 \times 10^{6} \qquad -3.08 \times 10^{6} \qquad -3.08 \times 10^{6} \qquad -2.97 \times 10^{6} \qquad -2.97 \times 10^{6} \qquad -3.08 \times 10^{6} \qquad -$	Log-likelihood	$-3.08 imes10^{6}$	$-3.07 imes10^{6}$	$-3.08 imes10^{6}$	$-3.04 imes10^{6}$	$-3.07 imes10^{6}$	$-3.04 imes10^{6}$	$-3.08 imes10^{6}$	$-3.08 imes10^{6}$	$-2.97 imes10^{6}$
Wald p-value 0.00	Wald <i>p</i> -value									0.00
LR <i>p</i> -value 0.00	LR <i>p</i> -value									0.00

Table A8. Results of the two-stage and bivariate probit models using +60 days default definition. The table reports the estimation results using three parametric approaches: the two-stage instrumental variable probit, the two-stage linear model (Dionne et al. 2015), and the bivariate probit. The sample includes 5,591,353 mortgages originated over the period from January 2000 to December 2013. Income and Divorce are instruments for the endogenous variable Default. Income is the annual growth rate of the U.S. household income. Divorce is the annual rate of divorce in the U.S. Pr(Default = 1) denotes the predicted probability of default from the 1st stage probit regression. $\hat{E}(Default)$ denotes the predicted default from the 1st stage linear model. Default denotes mortgage default (i.e., it is labelled as +60 days delinquent). Switch serv. denoting whether the originator switched the servicer of the deal. FICO score is the borrower's Fair Isaac Corporation score attributed at origination. LTV ratio denotes the initial loan-to-value ratio. ARM abbreviates adjustable-rate mortgages. Balloon refers to balloon payment mortgages. No/Low doc. indicates whether the originator collected no/low documentation. GSE conf. denotes loans that conform to the GSE's lending guidelines. GDP growth and HPI growth are the growth rates of the U.S. Gross Domestic Product and the House Price Index, respectively. σ interest refers to interest-rate volatility. Credit Spread is the yield difference between AAA and Baa bond indexes. Judicial denotes states that require judicial procedures to foreclose on a mortgage. SRR stands for Statutory Right of Redemption and denotes states that have statutory redemption laws. R2 is expressed in percentage and refers to the pseudo R2 for probit models and the adjusted R2 for linear models. ρ is the estimated correlation coefficient for the bivariate probit. The asterisks *** refer to the significant coefficients at the 1% significance level.

Model	Two-Stage	e IV Probit	E	DLB Linear Mod	Bivariate Probit			
Model	1st Stage	2nd Stage	1st Stage	2nd Stage	2nd Stage	Defeult	Switch Som	
Dependent Var.	Default	Switch Serv.	Default	Switch Serv.	Switch Serv.	Default	Switch Serv.	
Instruments								
Income	-0.0006 ***		-0.0002 ***					
Divorce	0.3028 ***		0.2069 ***					
Pr(Default = 1)		0.5197 ***						
Ê(Default)				0.4443 ***	0.1183 ***			
Default					0.3260 ***			
FICO score	-0.0037 ***		-0.0012 ***			-0.0037 ***	-0.0001 ***	
LTV ratio	0.0174 ***	0.0030 ***	0.0051 ***	0.0007 ***	0.0007 ***	0.0174 ***	0.0030 ***	
ARM	0.1018 ***	-0.1840 ***	0.0371 ***	-0.0749 ***	-0.0749 ***	0.1005 ***	-0.1712 ***	
Balloon	0.4097 ***	-0.0194 ***	0.1557 ***	-0.0435 ***	-0.0435 ***	0.4053 ***	0.0582 ***	
No/Low doc.	0.3416 ***	0.1590 ***	0.1087 ***	0.0472 ***	0.0472 ***	0.3431 ***	0.1696 ***	
GSE Conf.	-0.1446 ***	0.0785 ***	-0.0426 ***	0.0676 ***	0.0676 ***	-0.1434 ***	0.0015	
GDP growth	-4.7726 ***	4.3899 ***	-1.8273 ***	3.5229 ***	3.5229 ***	-1.8682 ***	-0.5094 ***	
HPI growth	-7.6137 ***	-5.8278 ***	-2.5897 ***	-0.9331 ***	-0.9331 ***	-7.6081 ***	-7.7881 ***	
σ interest	0.9214 ***	0.6408 ***	0.2792 ***	0.1016 ***	0.1016 ***	1.0572 ***	0.8377 ***	
Credit spread	1.9721 ***	1.1882 ***	0.6366 ***	0.0236 ***	0.0236 ***	1.8768 ***	1.9701 ***	
Judicial	-0.0416 ***	0.0122 ***	-0.0130 ***	0.0062 ***	0.0062 ***	-0.0410 ***	0.0019	
SRR	-0.0729 ***	0.0294 ***	-0.0233 ***	0.0118 ***	0.0118 ***	-0.0743 ***	0.0326 ***	
R ²	11.7	38.0	14.1	31.2	38.6			
ρ						0.6190 ***		
Table A9. Results of the two-stage and bivariate probit models using the 2001–2006 period. The table reports the estimation results using three parametric approaches: the two-stage instrumental variable probit, the two-stage linear model (Dionne et al. 2015), and the bivariate probit. The sample includes 5,086,938 mortgages originated over the period from January 2001 to December 2006. Income and Divorce are instruments for the endogenous variable Default. Income is the annual growth rate of the U.S. household income. Divorce is the annual rate of divorce in the U.S Pr(Default = 1) denotes the predicted probability of default from the 1st stage probit regression. Ê(Default) denotes the predicted default from the 1st stage linear model. Default denotes mortgage default (i.e., it is labelled as +90 days delinquent). Switch serv. denoting whether the originator switched the servicer of the deal. FICO score is the borrower's Fair Isaac Corporation score attributed at origination. LTV ratio denotes the initial loan-to-value ratio. ARM abbreviates adjustable-rate mortgages. Balloon refers to balloon payment mortgages. No/Low doc. indicates whether the originator collected no/low documentation. GSE conf. denotes loans that conform to the GSE's lending guidelines. GDP growth and HPI growth are the growth rates of the U.S. Gross Domestic Product and the House Price Index, respectively. σ interest refers to interest-rate volatility. Credit Spread is the yield difference between AAA and Baa bond indexes. Judicial denotes states that require judicial procedures to foreclose on a mortgage. SRR stands for Statutory Right of Redemption and denotes states that have statutory redemption laws. R^2 is expressed in percentage and refers to the pseudo R^2 for probit models and the adjusted R^2 for linear models. ρ is the estimated correlation coefficient for the bivariate probit. The asterisks *** refer to the significant coefficients at the 1% significance level.

M- 1-1	Two-Stage	e IV Probit		DLB Linear Mode	1	Bivariate Probit	
Model	1st Stage	2nd Stage	1st Stage	2nd Stage	2nd Stage	Defeult	Switch Com
Dependent Var.	Default	Switch Serv.	Default	Switch Serv.	Switch Serv.	Default	Switch Serv.
Instruments							
Income	-0.0012 ***		-0.0004 ***				
Divorce	3.8303 ***		1.3157 ***				
Pr(Default=1)		0.6350 ***					
Ê(Default)				0.3639 ***	0.0550 ***		
Default					0.3089 ***		
FICO score	-0.0035 ***		-0.0011 ***			-0.0035 ***	-0.0001 ***
LTV ratio	0.0181 ***	0.0004 ***	0.0050 ***	0.0002 ***	0.0002 ***	0.0179 ***	0.0031 ***
ARM	0.0926 ***	-0.2652 ***	0.0328 ***	-0.0912 ***	-0.0912 ***	0.0866 ***	-0.2397 ***
Balloon	0.4051 ***	-0.0681 ***	0.1604 ***	-0.0403 ***	-0.0403 ***	0.4340 ***	0.0663 ***
No/Low doc.	0.2985 ***	0.0895 ***	0.0904 ***	0.0296 ***	0.0296 ***	0.3079 ***	0.1282 ***
GSE Conf.	-0.1384 ***	0.0823 ***	-0.0376 ***	0.0550 ***	0.0550 ***	-0.1427 ***	0.0196 ***
GDP growth	7.5664 ***	16.7512 ***	1.8333 ***	7.3254 ***	7.3254 ***	11.7905 ***	18.9797 ***
HPI growth	-3.3490 ***	-5.6139 ***	-1.1534 ***	-0.8432 ***	-0.8432 ***	-6.4116 ***	-6.9225 ***
σ interest	0.5626 ***	1.0320 ***	0.1439 ***	0.2578 ***	0.2578 ***	1.3962 ***	1.3139 ***
Credit spread	1.7920 ***	1.6191 ***	0.5576 ***	0.1905 ***	0.1905 ***	1.8542 ***	1.9820 ***
Judicial	-0.0420 ***	0.0109 ***	-0.0127 ***	0.0062 ***	0.0062 ***	-0.0410 ***	0.0088 ***
SRR	-0.0525 ***	0.0467 ***	-0.0162 ***	0.0161 ***	0.0161 ***	-0.0545 ***	0.0553 ***
R ²	12.1	38.0	14.0	30.4	37.3		
ρ						0.60	04 ***

Table A10. Results of the two-stage and bivariate probit models using the 2001–2006 period and +60 days default definition. The table reports the estimation results using three parametric approaches: the two-stage instrumental variable probit, the two-stage linear model (Dionne et al. 2015), and the bivariate probit. The sample includes 5,086,938 mortgages originated over the period from January 2001 to December 2006. Income and Divorce are instruments for the endogenous variable Default. Income is the annual growth rate of the U.S. household income. Divorce is the annual rate of divorce in the U.S Pr(Default = 1) denotes the predicted probability of default from the 1st stage probit regression. Ê(Default) denotes the predicted default from the 1st stage linear model. Default denotes mortgage default (i.e., it is labelled as +60 days delinquent). Switch serv. denoting whether the originator switched the servicer of the deal. FICO score is the borrower's Fair Isaac Corporation score attributed at origination. LTV ratio denotes the initial loan-to-value ratio. ARM abbreviates adjustable-rate mortgages. Balloon refers to balloon payment mortgages. No/Low doc. indicates whether the originator collected no/low documentation. GSE conf. denotes loans that conform to the GSE's lending guidelines. GDP growth and HPI growth are the growth rates of the U.S. Gross Domestic Product and the House Price Index, respectively. σ interest refers to interest-rate volatility. Credit Spread is the yield difference between AAA and Baa bond indexes. Judicial denotes states that require judicial procedures to foreclose on a mortgage. SRR stands for Statutory Right of Redemption, and denotes states that have statutory redemption laws. R^2 is expressed in percentage and refers to the pseudo R² for probit models and the adjusted R² for linear models. ρ is the estimated correlation coefficient for the bivariate probit. The asterisks **, and *** refer to the significant coefficients at the 5%, and 1% significance levels, respectively.

M- 1-1	Two-Stage	e IV Probit	Γ	DLB Linear Model			Bivariate Probit	
Model	1st Stage	2nd Stage	1st Stage	2nd Stage	2nd Stage	Defeult	Switch Same	
Dependent Var.	Default	Switch Serv.	Default	Switch Serv.	Switch Serv.	Default	Switch Serv.	
Instruments								
Income	-0.0012 ***		-0.0004 ***					
Divorce	3.7299 ***		1.3072 ***					
Pr(Default = 1)		0.5789 ***						
Ê(Default)				0.3292 ***	0.0136 ***			
Default					0.3156 ***			
FICO score	-0.0038 ***		-0.0012 ***			-0.0037 ***	-0.0001 ***	
LTV ratio	0.0174 ***	0.0001 **	0.0049 ***	0.0004 ***	0.0004 ***	0.0172 ***	0.0031 ***	
ARM	0.0734 ***	-0.2599 ***	0.0269 ***	-0.0881 ***	-0.0881 ***	0.0690 ***	-0.2402 ***	
Balloon	0.4015 ***	-0.0562 ***	0.1566 ***	-0.0329 ***	-0.0329 ***	0.4295 ***	0.0663 ***	
No/Low doc.	0.3003 ***	0.0931 ***	0.0929 ***	0.0319 ***	0.0319 ***	0.3090 ***	0.1280 ***	
GSE Conf.	-0.1294 ***	0.0791 ***	-0.0370 ***	0.0531 ***	0.0531 ***	-0.1337 ***	0.0191 ***	
GDP growth	7.3051 ***	16.9002 ***	1.8576 ***	7.4274 ***	7.4274 ***	11.4314 ***	18.9506 ***	
HPI growth	-3.4066 ***	-5.7247 ***	-1.1831 ***	-0.9145 ***	-0.9145 ***	-6.4057 ***	-6.9189 ***	
σ interest	0.5607 ***	1.0537 ***	0.1501 ***	0.2713 ***	0.2713 ***	1.3720 ***	1.3126 ***	
Credit spread	1.7675 ***	1.6508 ***	0.5631 ***	0.2088 ***	0.2088 ***	1.8242 ***	1.9798 ***	
Judicial	-0.0406 ***	0.0102 ***	-0.0125 ***	0.0058 ***	0.0058 ***	-0.0388 ***	0.0089 ***	
SRR	-0.0409 ***	0.0438 ***	-0.0125 ***	0.0144 ***	0.0144 ***	-0.0435 ***	0.0558 ***	
R ²	12.0	38.0	14.2	30.4	37.7			
ρ			0.62		.30 ***			



Figure A1. FICO scores at origination by payment type.



Figure A2. FICO scores at origination by loan type.



Figure A3. No/Low documentation at origination by payment type.



Figure A4. Kernel density fitting of the FICO score.



Figure A5. Kernel density fitting of the LTV ratio.



Figure A6. Fitting of the KDE with multiple bandwidths.



Figure A7. FICO score vs. conditional probability of default.



Figure A8. Divorce rate vs. expected probability of mortgage default.



Figure A9. Income level vs. expected probability of mortgage default.

Note

¹ The contents of this section are based on the report of the Mortgage Bankers Association (MBA) and PwC US (PricewaterhouseCoopers) (2015), and the report to Congress by the Board of Governors of the Federal Reserve System (2016).

References

- Agarwal, Sumit, and Itzhak Ben-David. 2018. Loan prospecting and the loss of soft information. *Journal of Financial Economics* 129: 608–28. [CrossRef]
- Agarwal, Sumit, and Robert Hauswald. 2010. Distance and private information in lending. *The Review of Financial Studies* 23: 2757–88. [CrossRef]
- Agarwal, Sumit, Yan Chang, and Abdullah Yavas. 2012. Adverse selection in mortgage securitization. *Journal of Financial Economics* 105: 640–60. [CrossRef]
- Albertazzi, Ugo, Ginette Eramo, Leonardo Gambacorta, and Carmelo Salleo. 2015. Asymmetric information in securitization: An empirical assessment. *Journal of Monetary Economics* 71: 33–49. [CrossRef]
- Ambrose, Brent W., Michael LaCour-Little, and Anthony B. Sanders. 2005. Does regulatory capital arbitrage, reputation, or asymmetric information drive securitization? *Journal of Financial Services Research* 28: 113–33. [CrossRef]
- Ashcraft, Adam B., Kunal Gooriah, and Amir Kermani. 2019. Does skin-in-the-game affect security performance? Evidence from the conduit CMBS market. *Journal of Financial Economics* 134: 333–54. [CrossRef]
- Begley, Taylor A., and Amiyatosh Purnanandam. 2017. Design of financial securities: Empirical evidence from private-label RMBS deals. *Review of Financial Studies* 30: 120–61. [CrossRef]
- Board of Governors of the Federal Reserve System. 2016. Report to the Congress on the Effect of Capital Rules on Mortgage Servicing Assets. Available online: https://www.federalreserve.gov/publications/other-reports/files/effect-capital-rules-mortgage-servicing-assets-201606.pdf (accessed on 25 September 2024).
- Bubb, Ryan, and Alex Kaufman. 2014. Securitization and moral hazard: Evidence from credit score cutoff rules. *Journal of Monetary Economics* 63: 1–18. [CrossRef]
- Centorrino, Samuele, and Jean-Pierre Florens. 2021. Nonparametric instrumental variable estimation of binary response models with continuous endogenous regressors. *Econometrics and Statistics* 17: 35–63.
- Chiappori, Pierre-André, and Bernard Salanié. 2000. Testing for asymmetric information in insurance markets. *Journal of Political Economy* 108: 56–78. [CrossRef]
- Das, Mitali. 2005. Instrumental variables estimators of nonparametric models with discrete endogenous regressors. *Journal of Econometrics* 124: 335–61. [CrossRef]
- Dionne, Georges, Christian Gouriéroux, and Charles Vanasse. 2001. Testing for evidence of adverse selection in the automobile insurance market: A comment. *Journal of Political Economy* 109: 444–53. [CrossRef]
- Dionne, Georges, Christian Gouriéroux, and Charles Vanasse. 2006. Informational Content of Household Decisions with Applications to Insurance under Asymmetric Information. In *Competitive Failures in Insurance Markets: Theory and Policy Implications*. Edited by P. A. Chiappori and C. Gollier. Cambridge, MA: MIT Press, pp. 159–84.
- Dionne, Georges, Mélissa La Haye, and Anne-Sophie Bergerès. 2015. Does asymmetric information affect the premium in mergers and acquisitions? *Canadian Journal of Economics/Revue Canadienne D'économique* 48: 819–52. [CrossRef]
- Dionne, Georges, Pascal St-Amour, and Désiré Vencatachellum. 2009. Asymmetric information and adverse selection in mauritian slave auctions. *The Review of Economic Studies* 76: 1269–95. [CrossRef]
- Dionne, Georges, Pierre-Carl Michaud, and Maki Dahchour. 2013. Separating moral hazard from adverse selection and learning in automobile insurance: Longitudinal evidence from France. *Journal of the European Economic Association* 11: 897–917. [CrossRef]
- Elul, Ronel. 2016. Securitization and mortgage default. Journal of Financial Services Research 49: 281–309. [CrossRef]
- Filomeni, Stefano. 2024. Securitization and risk appetite: Empirical evidence from US banks. *Review of Quantitative Finance and Accounting* 63: 433–68. [CrossRef]
- Fisher, Nicholas I., and Peter Hall. 1990. On bootstrap hypothesis testing. Australian Journal of Statistics 32: 177-90. [CrossRef]
- Gete, Pedro, and Michael Reher. 2021. Mortgage securitization and shadow bank lending. *The Review of Financial Studies* 34: 2236–74. [CrossRef]
- Gorton, Gary, and Andrew Metrick. 2012. Securitization. NBER, Working Paper 18611. Available online: https://www.nber.org/papers/w18611 (accessed on 25 September 2024).
- Horowitz, Joel L. 2011. Applied nonparametric instrumental variables estimation. Econometrica 79: 347-94.
- Keys, Benjamin J., Amit Seru, and Vikrant Vig. 2012. Lender screening and the role of securitization: Evidence from prime and subprime mortgage markets. *The Review of Financial Studies* 25: 2071–108. [CrossRef]
- Keys, Benjamin J., Tanmoy Mukherjee, Amit Seru, and Vikrant Vig. 2010. Did securitization lead to lax screening? Evidence from subprime loans. *The Quarterly Journal of Economics* 125: 307–62. [CrossRef]
- Krainer, John, and Elizabeth Laderman. 2014. Mortgage loan securitization and relative loan performance. *Journal of Financial Services Research* 45: 39–66. [CrossRef]
- Li, Qi, and Jeffrey Scott Racine. 2007. Nonparametric Econometrics: Theory and Practice. Princeton: Princeton University Press.
- Liberti, José María, and Mitchell A. Petersen. 2018. Information: Hard and soft. Review of Corporate Finance Studies 8: 1-41. [CrossRef]

MacKinnon, James G. 2009. Bootstrap hypothesis testing. Handbook of Computational Econometrics 183: 213.

- Malekan, Sara, and Georges Dionne. 2014. Securitization and optimal retention under moral hazard. *Journal of Mathematical Economics* 55: 74–85. [CrossRef]
- Maliar, Serguei, and Bernard Salanié. 2023. Testing for Asymmetric Information in Insurance with Deep Learning. Working Paper. New York: Columbia University. 32p.
- Morgan, Lewis. 2018. Guide to the Credit Risk Retention Rules for Securitization. Available online: https://www.morganlewis.com/ pubs/2024/07/a-guide-to-the-credit-risk-retention-rules (accessed on 25 September 2024).
- Mortgage Bankers Association (MBA), and PwC US (PricewaterhouseCoopers). 2015. The Changing Dynamics of the Mortgage Servicing Landscape. Available online: https://www.housingwire.com/wp-content/uploads/media/images/15217_MBA_PWC_White_Paper.pdf (accessed on 25 September 2024).
- Parzen, Emanuel. 1962. On estimation of a probability density function and mode. *The Annals of Mathematical Statistics* 33: 1065–76. [CrossRef]
- Passmore, Wayne, and Roger Sparks. 2024. Government-Sponsored Mortgage Securitization and Financial Crises. Washington, DC: Finance and Economics Discussion Series Federal Reserve Board.
- Racine, Jeffrey S. 2008. Nonparametric econometrics: A primer. Foundations and Trends in Econometrics 3: 1-88. [CrossRef]
- Rosenblatt, Murray. 1956. Remarks on some nonparametric estimates of a density function. In *The Annals of Mathematical Statistics*. Berlin and Heidelberg: Springer, pp. 832–37.
- Su, Liangjun, and Martin Spindler. 2013. Nonparametric testing for asymmetric information. *Journal of Business & Economic Statistics* 31: 208–25.

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.



Article



The Role of Credit Consortia in the Financial Structure of Sardinian Companies During the SARS-CoV-2 Crisis

Marco Desogus ¹, Enrico Sergi ² and Stefano Zedda ^{2,*}

- ¹ Department of Economic and Statistical Sciences, Università di Sassari, 07100 Sassari, Italy; mdesogus@uniss.it
- ² Department of Economics and Business Sciences Università di Cagliari, 09100 Cagliari, Italy; enrico.sergi@unica.it
- * Correspondence: szedda@unica.it

Abstract: In this paper, we analyzed the role of credit consortia in supporting SMEs of the Italian region of Sardinia around and during the SARS-CoV-2 pandemic crisis. Credit consortia (or credit guarantee schemes) are financial companies whose institutional role is to support small firms needing bank lending who are individually weak in the bank-firm relationship. Credit consortia are particularly relevant in Italy, where they mitigate credit restrictions for SMEs by supplying guarantees to the bank, allowing for partial coverage of potential losses, providing peer-monitoring activity, and collectively negotiating more favorable interest rates and other conditions with banks. During the SARS-CoV-2 pandemic, credit consortia had a crucial role in supporting Sardinian SMEs with guarantees and obtaining government financial support. The evolution of Sardinian companies' financial structures during the SARS-CoV-2 pandemic shows that the confidi-supported firms have low capitalization and are financially fragile yet capable of good returns. The liquidity provided by the government during the pandemic loosened these constraints, boosting the available liquidity, which translated, in short, into higher investment and higher sales. The demographics of Sardinian companies in 2019–2022 and the volumes of loans and savings showed a strengthening of debt capital payments, increased collections, and a progressive improvement of the Sardinian companies' net financial positions.

Keywords: credit consortia; SARS-CoV-2 pandemic; bank-firm relationship; SMEs

1. Introduction

Credit consortia (or credit guarantee schemes) are financial companies whose institutional role is to support small firms needing bank lending who are individually weak in the bank–firm relationship. According to Leone et al. (2013), credit consortia "offer mutualistic guarantees aimed at alleviating financial constraints on their small or medium-sized shareholding or syndicated enterprises". Credit consortia are particularly relevant in Italy, accounting for almost 40 percent of the total outstanding volume of guarantees to SMEs (Columba et al. 2010).

Technically, credit consortia mitigate credit restrictions for SMEs by supplying guarantees to the lending bank, allowing for partial coverage of potential losses, providing peer-monitoring activity, and acting like debtors' unions for collectively negotiating more favorable interest rates and other conditions with banks.

Usually, credit consortia are strictly linked with business associations, which helps with information sharing among firms within the business association and the credit consortia and allows for some lobbying activity.

Credit consortia played a significant role during the SARS-CoV-2 pandemic, supporting SMEs with guarantees and easing the bank–firm relationship in all credit processes.



Citation: Desogus, Marco, Enrico Sergi, and Stefano Zedda. 2024. The Role of Credit Consortia in the Financial Structure of Sardinian Companies During the SARS-CoV-2 Crisis. *Risks* 12: 190. https:// doi.org/10.3390/risks12120190

Academic Editors: Eulália Mota Santos and Margarida Freitas Oliveira

Received: 30 September 2024 Revised: 21 November 2024 Accepted: 25 November 2024 Published: 28 November 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). In this paper, we analyzed the evolution of Sardinian SMEs' financial equilibrium and the role played by credit consortia in supporting it during and after the SARS-CoV-2 pandemic.

In 2020, Italy was one of the European countries most affected by the pandemic, which caused significant contractions in the business economy. According to the Centro Studi Confcommercio (CSC) (2024), Italy's Gross Domestic Product (GDP) lost EUR 139 billion, or over EUR 2600 per inhabitant. The supply and demand crisis impacted specific economic activity sectors and aggregate variables. Of the EUR 139 billion in lost product in 2020, approximately EUR 129 billion were attributed to the collapse of domestic consumption, including spending by non-residents in Italy.

Das (2020) highlights territorial differences in the forecasted and actual growth rates between countries, emphasizing how policies against the pandemic and vaccination campaigns affected trends and growth rates.

These territorial differences affected the incidence numbers and the level of government restrictions. According to the "Osservatorio Nazionale sulla Salute nelle Regioni Italiane (2021, November)" report, SARS-CoV-2 infections in Sardinia initially remained lower than in other Italian regions, particularly during the early stages of the pandemic in 2020. However, by late 2020 and early 2021, Sardinia saw a significant rise in cases, aligning with the national trend. Despite this, the region did not face the same level of healthcare strain as northern Italy. Throughout 2021, as vaccines were rolled out, the case numbers stabilized, although fluctuations remained due to new variants. Sardinia's trajectory generally mirrored national patterns, with some regional variations in infection rates and healthcare system pressure.

In 2021, thanks to the vaccination campaign and the subsequent reduction in the spread of infections and easing restrictions, Italy recovered about two-thirds of the GDP contraction seen in 2020, with a GDP growth of +6.6%.

In early 2021, Sardinia's regional economy remained weak, being influenced by the complex epidemiological situation and restrictions on social activities, which severely impacted the tourism sector. In Sardinia, the crisis primarily affected the services sector, which was highly sensitive to containment measures, particularly in tourism, transport, entertainment, and non-food retail. The industrial sector also faced contractions, with significant reductions in the chemical and oil industries. Corporate profitability decreased due to lower turnover, leading to an increased need for liquidity, which was met through expanded credit and public support measures, as will be discussed later.

During 2021, economic activity showed significant improvements from April to June, followed by more moderate growth in the second half of the year, as the Bank of Italy estimated in its quarterly regional economy report (Bank of Italy, 2021 and 2022). Increased consumption, a modest investment rebound, and growing foreign demand for key Sardinian products supported this recovery. However, estimates by Prometeia () indicated that by the end of 2021, Sardinia's GDP was still 4.5% lower than pre-pandemic levels.

In the manufacturing sector, economic conditions improved unevenly across different sectors. The global economic slowdown continued into the last quarter of 2021, being driven by the rise in infections linked to the Omicron variant, production input shortages, and geopolitical tensions that culminated with Russia's invasion of Ukraine. These tensions led to a significant increase in raw material prices, particularly energy, exacerbated by the Russia–Ukraine conflict in 2022. Despite these challenges, in 2021, the profitability of Sardinian companies strengthened, being primarily driven by increased sales but partially constrained by rising prices for energy, commodities, and other raw materials, especially in the manufacturing sector. Corporate liquidity remained high, supported by favorable access to credit, but investment activity remained limited, especially in the industrial sector.

At the national level in 2020, according to Banca d'Italia (2023a), Italian banks continued to meet companies' demands for financing. Supply conditions remained generally favorable, being supported by ongoing monetary policy measures and public guarantees. According to the Ufficio Studi ABI—Associazione Bancaria Italiana (2024), loans in Italy grew by 4.1% in 2020 compared to the previous year, with Sardinia showing a slightly higher growth of 4.3%. This trend held even when focusing solely on production activities, with Sardinia recording an increase of 8.8%, close to the national figure of 8.4%. These data suggest that despite the economic challenges generated by the pandemic, Italian banks continued to support businesses through financing, being aided by monetary policies and government guarantees.

The financial structure of Sardinian companies and the role of regional credit consortia (confidi) during the SARS-CoV-2 pandemic revealed that firms supported by confidi were typically low-capitalized and financially fragile. However, these firms could generate good returns, and the liquidity provided by government measures during the pandemic alleviated these constraints. The government support led to higher available liquidity, translating into increased investment and sales in the short term. Evidently, this government spending had more macro-effects, but discussing these effects is beyond the scope of this paper.

The remainder of the paper is structured as follows: Section 2 presents the literature review, Section 3 outlines the Sardinian economic framework, Section 4 discusses the evolution of the pandemic, Section 5 compares the differential evolution of a general sample and confidi-supported firms, and Section 6 discusses the results and concludes.

2. Literature Review

Our study aims to examine the role of credit consortia (also known as credit guarantee schemes, CGSs, or mutual guarantee societies, MGSs) on Sardinian SMEs during and after the SARS-CoV-2 pandemic crisis. Numerous studies in the literature have investigated related questions, providing a foundation for understanding the financial challenges SMEs face during economic downturns and the role of credit consortia in supporting these businesses.

The literature review refers to three key areas: (1) studies examining credit consortia during the SARS-CoV-2 pandemic and other crises, (2) research analyzing SME behaviors in Italy during past economic crises, and (3) investigations into changes in financial structures influenced by the SARS-CoV-2 pandemic. This organization highlights how credit consortia have worked as financial stabilizers during crises, provides insights into SMEs' crisis responses in Italy, and examines the broader shifts in financial structures brought about by pandemic-related disruptions.

Several studies have focused on the Spanish credit market, specifically investigating the role of credit consortia, mainly called MGSs. In their paper, Corredera-Catalán et al. (2021) explore the Spanish guarantee model and how regional governments, in collaboration with MGSs, responded to the economic difficulties caused by the SARS-CoV-2 pandemic. Their study underscores the effectiveness of the guarantee schemes used by public administrations to improve SMEs' access to financing, helping stabilize the economy while minimizing the strain on public finances. Similar to our study, they evaluate the role of MGSs during crisis periods but focus solely on their effectiveness in supporting the lending capacity without analyzing the financial performances of the financed firms. Unlike our study, they do not incorporate balance sheet variables to assess firm performance during the crisis period.

De la Fuente-Cabrero et al. (2019) evaluated loan defaults in the portfolios of Spanish MGSs to understand their evolution between 2003 and 2012, covering both economic growth and recession periods. They highlight how MGS support influences the distribution of defaults based on the company size and sector of activity. However, this study investigated MGS lending using different risk factors. And it did not compare the financial performances of firms that used MGSs with those that obtained financing directly from banks or were not supported, as we do in our research.

Gómez et al. (2016) investigated SME access to financing through MGSs in Spain during the 2010 economic crisis. Their findings indicate that SMEs with MGS-backed financing generally had better access to bank loans than those without such guarantees.

However, the impact of reducing financing costs was less conclusive. Our results will show that for Sardinian firms, the cost of financing is higher when the funding source is an MGS.

Briozzo and Cardone-Riportella (2016) examine Spanish SMEs' access to credit and subsidies with credit guarantees during crisis periods compared to standard years. Similar to our study, they compare direct credit channels through official credit institutions with bank credit guaranteed by MGSs. They highlight that these guarantees can serve as countercyclical tools, noting that mutual guarantee schemes cover only a limited portion of the Spanish credit market, indicating a relatively reduced role in credit guarantee activities.

While other studies have investigated the behavior of Italian MGSs before the pandemic, research has yet to focus on their response to the recent economic crisis caused by SARS-CoV-2 in Italy. Considering less recent publications on the Italian credit market, Gai et al. (2016) analyzed the determinants of defaults among firms that received SME loans guaranteed by MGSs and counter-guaranteed by the Italian Central Guarantee Fund from 2010 to 2011. They found that the proportion of counter-guaranteed loans in MGS portfolios increases the default risk. This can be explained by the fact that SMEs financed through MGSs are smaller and more vulnerable during crises. In our study, we will demonstrate how support from MGSs enables these SMEs to access essential resources, fostering positive growth trends. This support is well invested in businesses that, without MGSs, would likely be unable to secure the resources needed to withstand economic downturns.

Caselli et al. (2021) examined the loan failure risk associated with public credit guarantees, distinguishing between banks and MGSs. Their study assessed the Italian SME credit market during the 2008 crisis, finding that banks provided better loan screening in the manufacturing sector than MGSs despite the challenges of evaluating smaller enterprises. Cusmano (2018) studied the relationship between credit guarantee schemes and MGSs across European countries, examining their role in supporting SME financing during the 2008 financial crisis. The findings suggest that guarantee schemes are most effective when they deeply understand the market and industry dynamics specific to the SMEs they support. These publications on the Italian market are relevant because they allow us to compare similar scenarios in which public intervention prevents financial difficulties for vulnerable businesses during times of crisis. Like these publications, our results highlight the resilience of SMEs due to public intervention.

Another area of research has extensively examined the impact of the SARS-CoV-2 pandemic on firms' financial structures, focusing on how shifts in economic conditions and increased uncertainty have affected capital allocation, debt management, and financial stability. Studies such as those of Halling et al. (2020), Acharya and Steffen (2020), Li et al. (2020), and Gopalakrishnan et al. (2022) indicate that during periods of heightened uncertainty, firms' capital structure decisions have led to an increase in debt financing. This greater reliance on debt can be linked to firms' precautionary needs and the financial challenges posed by crisis conditions. Additional studies by Prakash et al. (2023), Huang and Ye (2021), and Nguyen Kim (2023) investigate the pandemic's impact on the capital structure and debt levels of firms in India, China, and Vietnam, respectively. Additionally, D'Amato (2020) studied Italian SMEs before and after the global financial crisis, emphasizing that the short-term debt channel is more sensitive than the long-term one, supporting the findings of previously cited articles.

Similarly, Wang et al. (2024) demonstrates that extending debt maturity in response to SARS-CoV-2 shocks helps firms avoid potential maturity constraints, lowers debt financing costs, and alleviates liquidity pressures. In a crisis scenario, particularly during a pandemic shock, the uncertainty in the external environment amplifies risks from asset-liability maturity mismatches, leading to a decline in firm performance. Extending debt maturity can help alleviate these mismatches.

Considering the Italian credit market, SMEs' debt levels were supported by government intervention, which provided them with easier access to financial resources during the pandemic. Analyzing the performances of these financed firms will help determine whether, without public guarantees, these SMEs would have had sufficient internal resources to achieve positive performance. Unlike the previously cited studies, which primarily focus on the financial structures of firms during the SARS-CoV-2 crisis, our research examines the performances of financed firms by tracking changes in financial indicators. We analyze shifts in debt and capitalization levels during the pandemic, highlighting that firms using MGSs exhibit higher debt exposure compared to the broader firm population that used direct bank financing. Typically smaller in size, these firms tend to prefer debt over equity due to the support provided by public credit guarantees.

Considering the three literature streams, financial support is one of the simpler and quicker responses for supporting the real economy during financial or economic crises. The experiences of the 2008–2012 crisis highlight that credit guarantees can effectively minimize the strain on public finances while helping SMEs overcome the crisis. In these cases, MGSs prove to be locally effective when they deeply understand the market and industry dynamics specific to the SMEs they support, and the intervention result most suited in these cases is the double support of immediate liquidity provision and debt maturity extension, so as to keep the firms' financial equilibrium.

Regarding methodology, the paper by Khurana et al. (2024) adopts a similar approach to ours. It analyzes the effects of SARS-CoV-2 on firm performance across various industries by examining liquidity and financial ratios, including the net working capital, quick ratio, debt–equity ratio, and financial autonomy rate. This study indicates a negative impact on specific sectors. It reveals a positive relationship between debt financing, working capital management, and firm performance while finding a negative correlation with financial autonomy. In our study, the average debt level is higher for MGS-supported firms due to their smaller size and reliance on government financial support, leading them to prefer external assistance over financial autonomy. While the study by Khurana et al. (2024) analyzes firms' responses to the crisis, our research distinguishes itself by comparing the performances of firms that are not supported with those supported through MGSs.

The SARS-CoV-2 pandemic has severely impacted SMEs, highlighting the need for effective financial support mechanisms. MGSs are crucial in providing credit guarantees to firms, particularly during economic crises. Our study aims to investigate the effects of SARS-CoV-2 on SME credit and explore the role of MGSs in supporting these businesses during this challenging time.

Additionally, our study differs from the cited works, as we use financial data both from firms financed through direct lending channels and those not and compare these with firms financed through the intermediary channel of MGSs. Only some studies in the literature have access to such detailed data for multiple years on firms supported by MGSs, as these entities are typically private, which adds value to our research.

Our methodology is based on a descriptive approach, focusing on the evolution of financial data before and after the pandemic. Despite extensive research on MGSs, significant gaps remain. Most studies focus on pre-pandemic periods or past economic crises, neglecting the recent challenges posed by SARS-CoV-2 in Italy. While many examine the lending capacity of MGSs, they often do not pay adequate attention to the financial performance of the firms they support. Moreover, the existing literature primarily examines MGSs in contexts outside of Italy, leading to a need for updated research specific to the Italian landscape during and after the pandemic.

Our study addresses these gaps by providing an analysis of MGSs during the SARS-CoV-2 pandemic and incorporating balance sheet variables to evaluate the financial performance of firms that receive MGS-backed financing compared to those utilizing direct lending channels. By leveraging comprehensive financial data over multiple years, our research will offer valuable insights into the effectiveness of MGS support, ultimately informing policymakers and financial institutions about enhancing SME resilience during crises.

3. The Sardinia Framework

During the SARS-CoV-2 crisis, the number of registered companies showed a slight increase compared to pre-pandemic years due to the effects of support measures in terms

of liquidity, credit, and fiscal and social security provisions, followed by a decline in 2022 (Figure 1). More in detail, a significant decline is observed in the number of companies in the trade sector, with a reduction of 931 units from December 2019 to December 2021 (reaching a decline of 1850 companies by the end of 2022). The manufacturing sector decreased by over 4.7%, the extractive sector declined by 11.1%, and transport and logistics contracted by approximately 3.1%. Sectors showing growth, as expected, include health services (+13.2%), real estate activities (+12.3%), electricity and gas supply (+12.0%), and business services (+6.8%).





3.1. Government Financial Support

To ensure resilience in the economic system during the SARS-CoV-2 emergency, it was crucial to preserve the levels of credit leverage, particularly in fragile or partially compromised economic scenarios. This included implementing guidelines set by the Italian government to maintain the effective functioning of the banking–business system. In response to the crisis, the Italian government focused on strengthening public guarantee funds rather than directly injecting liquidity into the production system or households. This strategy maintained the role of banks as financial intermediaries and encouraged a quantitative expansion of credit.

An essential role was also given to the "private" guarantees provided by credit consortia, which further reduced bank provisions and mitigated risks through increased counter-guarantees offered by the Central Guarantee Fund for SMEs (FCG). Legislative Decrees No. 18, 20, and 34 of 2020 (known as "Cura Italia", "Liquidità", and "Rilancio") primarily allocated resources toward direct public guarantees, with zero weighting on provisions and reinsurance for credit guarantee consortia. These measures helped to prevent an additional credit crunch, counteracting the decline in creditworthiness caused by the lockdown and averting potential crises in both the corporate and banking sectors.

While reducing credit leverage below a certain threshold may initially appear beneficial for banks due to lower provisions and reduced risk indicators, prolonged credit restriction threatens the stability of the production and entrepreneurial landscapes (Desogus and Casu 2020). Our analysis highlights the effectiveness of government measures and the significant role of credit consortia in preventing this adverse dynamic, supporting corporate liquidity and, ultimately, system stability.

3.2. Loans

During 2020, the Italian government implemented several actions to counteract the effects of the pandemic on the national economy and promote credit inclusion (Senato della Repubblica Italiana 2023). These actions, included in the aforementioned legislative decrees, were based on the extensive use of public guarantees and characterized by several key features:

- Free guarantee: the public guarantee was provided at no cost.
- Increase in the maximum guaranteed amount: the maximum guaranteed amount per company was raised from EUR 2.5 million to EUR 5 million.
- Increased coverage percentages of the FCG: coverage percentages from the Guarantee Fund for SMEs (FCG) were enhanced, with a direct guarantee of 80% and reinsurance of 90%.
- Granting guarantees without a credit assessment: guarantees were granted without a credit assessment, allowing companies across all rating ranges to be eligible.
- Suspension of mortgage payments: mortgage payments were suspended, with an automatic extension of any public guarantee.
- Consolidation/renegotiation operations: these operations were allowed within the same bank or banking group for loans not already covered by the FCG.
- Use of the Temporary Aid Framework: this framework provided a 100% guarantee for financing up to EUR 30,000 and increased FCG coverage to 90% for direct guarantees and 100% for reinsurance.

These measures increased public guarantees and, more broadly, helped to revitalize the role of credit consortia (confidi), which benefited from enhanced reinsurance coverage by the FCG, generating a more significant weighting effect. From the beginning of the SARS-CoV-2 emergency until the end of 2020, the largest number of operations benefiting from FCG guarantees were those of smaller size, with an average value of less than EUR 30,000 (100% covered), representing 63.56% of the accepted requests. These operations accounted for more than EUR 21 billion in guarantees (Fondo di Garanzia 2024a), with an average unit value of EUR 19,600. However, this category had a much lower impact (15.7%) on the total guaranteed values in the same period.

The interventions were designed specifically to reduce bureaucratic formalities, enabling near-automatic evaluations for immediate credit availability. These operations were primarily managed by banks, with credit consortia focusing on smaller-scale operations for micro-, small-, and medium-sized enterprises (mPMI). This shifted the role of confidi, which traditionally dealt with smaller-sized operations, as banks had previously expressed opposition due to the complexity of automating the investigation process, especially for sole proprietorships, partnerships, artisan businesses, start-ups, and companies with few years of activity. These businesses not only require guarantees but also support to access credit.

In practice, the banking sector's activity smoothed the potential growth of confidisupported firms with the SARS-CoV-2 measures. However, a "carryover" effect bolstered the reinsurance activity of 2020 by confidi, with a national increase of 91.5% (from EUR 1596 to 3057 million) and a 30.8% rise at the regional level (from EUR 57 to 75 million). In 2021, FCG guarantees decreased from EUR 105.9 billion to EUR 67.6 billion (-36.1%), gradually returning to previous standards. In Sardinia, as illustrated by the tables and graphs below, the risk transfer to the state helped maintain a positive trend in corporate net liquidity.

The value of loans to businesses (Figure 2) stabilized the bank–business relationship, counteracting the shocks caused by the crisis and showing a slight increase. In Sardinia, aggregate data on business loans showed a +12.3% increase between 2020 and 2019, with a further +3.3% increase between 2021 and 2020 (end-of-year data). The overall intervention of the Central Fund (direct guarantees and counter-guarantees on confidi operations) increased from 2.6% in 2019 (pre-pandemic) to a peak of 18.95% in 2020, stabilizing at 12.7% in 2021. From 2020 to 2021, credit support for "Industrial Activities" grew by +29.3%, for "Services" by +13.9%, and for "Construction and Building" by +12.2%. In terms of firm size, which is influenced by the structure of the Italian and Sardinian economies, credit leverage for micro-enterprises grew by +10.2% between 2020 and 2022. Medium–large companies recorded a +6.1% increase, while small companies experienced a slight contraction of -0.5%.



Figure 2. Loans in Sardinia by sector, years 2019–2022 (thousands of euros). Source: own computations on Bank of Italy data.

3.3. Non-Performing Loans

As reported by the Bank of Italy, during 2020–2021, despite the challenges posed by the pandemic, the stock of non-performing loans (NPLs) accumulated by Italian banks continued to decrease. This reduction affected both loans to households and loans to businesses. The trend in these indicators also continued to reflect the positive impact of government measures designed to support access to credit, such as moratoria (loan repayment deferrals) and guarantees on new loans.

Additionally, as directed by the supervisory authorities, the flexibility incorporated into the loan classification rules played a significant role, as previously mentioned. It is important to note that the reduction in NPLs was also facilitated by the sale of non-performing loans, continuing a trend that started in 2017. These sales improved the average quality of banks' assets, with a progressive decline in NPLs at the national and regional levels, as shown in Figure 3.



Figure 3. Bad loans in Sardinia, 2018–2021 (thousands of euros). Source: own computations on Bank of Italy data.

3.4. Deposits

Corporate liquidity showed a general increase during the period considered, as highlighted in Figure 4 below. Deposits from Sardinian companies grew by 36.3% between 2019 and 2020, followed by an additional increase of 17.2% in 2021. Overall, this two-year period saw a cumulative increase of 59.7%.

Figure 4 illustrates the increase in bank deposits during the pandemic period. Under normal circumstances, such deposits would decrease due to the financial distress that firms experience during economic uncertainty. However, this deposit growth can be attributed to government support measures implemented during the pandemic.

The graph shows that both producer households and private enterprises were able to increase their deposit levels despite the challenges posed to business activities. This



household

Craft quasi

companies Other non-financial

Total

quasi-companies

Private enterprises

indicates that government support extended to firms of all sizes, benefiting smaller entities, such as producer households, and larger private enterprises across various categories.



31101021

3101001

SPARD- MIRIT

31P11 not part mail

3110102

3.5. The Net Financial Position of Companies in Sardinia

3101 Stopper BIRD BOR BOR

The aggregate data of loans and deposits show a significant improvement in regional companies' overall net financial position: from EUR -3.1 billion at the end of 2019 to -2.5 billion at the end of 2020 to -1.8 billion at the end of 2021. The trend is also confirmed for December 2022, with an aggregate PFN of -1.5 billion.

A comprehensive analysis of the data presented in the previous paragraphs shows that the government measures effectively prevented the credit contraction that had been anticipated due to recession risks arising from the exogenous economic crisis caused by the SARS-CoV-2 pandemic. Although there is a slight decline in the business demographic balance in Sardinia between 2022 and 2021, the trends in loans extended to the production sector and the continued reduction in non-performing loans indicate stable systemic parametric trajectories. However, the evidence of deposit growth, alongside the reported improvement in the net financial position, reveals a conservative approach to resource management marked by limited investments. This further confirms the support's largely reactive and externally driven nature rather than a proactive one.

4. The Pandemic Crisis

6,000,000

5,000,000

4,000,000

3.000.000

2,000,000

1.000.000

0 APHPAP

SIPA' LIPIPIY MOIS

To provide a more transparent contextual framework for the analysis of the impact of guarantee consortia (confidi) on mitigating the economic and financial effects of the SARS-CoV-2 containment measures, in the following paragraphs, we present trend data on the use of public guarantees and the extent of confidi interventions both at the national level and, in particular, within the Sardinian region for the years 2019–2022, splitting them into the pre-pandemic year (2019), the two pandemic years (2020 and 2021), and the post-pandemic year (2022).

4.1. Pre-Pandemic: The 2019 Scenario

In 2019, the world economy showed some positive signals: international trade began to grow again, and there was a reduction in tariff disputes between the United States and China. However, the outlook remained uncertain, with rising geopolitical tensions leading to weakly positive growth expectations. These expectations were supported by the accommodative policies of central banks, which provided a slight upward push to stock markets and facilitated a modest improvement in long-term yields. In the Euro area, economic activity was hindered by the vulnerability of the manufacturing sector, particularly in Germany, although an improving trend emerged toward the end of the year. This economic trend influenced inflation, which, while reinforced by monetary stimulus, remained below 2 percent. The ECB's

Governing Council maintained its accommodative policy throughout the year, signaling its continuation in the years ahead (Banca d'Italia 2020).

In Italy, economic activity showed slight growth in the third quarter but remained nearly stagnant in the fourth quarter, mainly due to the weak manufacturing sector. According to ISTAT—Istituto Nazionale di Statistica (2020) and the Banca d'Italia (2020), companies were slightly more optimistic about orders and foreign demand. However, uncertainty and trade tensions continued to pose obstacles. Despite this, companies planned to increase investments in 2020 at a more modest pace than the previous year. In the final months of 2019, foreign investors purchased significant amounts of Italian government bonds (over EUR 90 billion for the year), improving the Bank of Italy's debit balance on the European payment system TARGET2 (Drott et al. 2024; Glowka et al. 2022). This improvement was also supported by increased net foreign funding through the Repo market by Italian banks, facilitated by activating the new Eurosystem bank reserve remuneration system (Braun 2020).

During 2019, Italy's current account surplus remained substantial, and its net foreign position moved closer to equilibrium. Inflation remained contained, at 0.5% in December. Services contributed more to overall price dynamics, with modest growth observed in the industrial goods sector. Core inflation increased slightly in the autumn, reaching 0.7%. Starting mid-October, Italian government bond yields and stock markets rose, reflecting similar trends across the Euro area. However, yields on bonds issued by Italian banks and non-financial corporations remained more than 70 basis points below the average of the first half of 2019.

4.1.1. Financial Intermediation in 2019

The cost of credit decreased significantly for households. At the same time, lending to businesses continued to decline, primarily due to the weak demand for new financing. However, the measures adopted by the ECB in September were expected to improve credit conditions, according to commercial banks. The data from 2019 indicate a slight decrease in the net borrowing of public administrations relative to the GDP, but there was an overall increase in the debt-to-GDP ratio. The budget plan for 2020 to 2022, approved by parliament in December 2019, projected an average increase in the deficit of 0.7 percentage points of the GDP per year compared to trend values. According to the government's programs, the impact of net borrowing and debt on the economy was expected to decrease over the following two years (MEF—Ministero dell'Economia e delle Finanze 2020–2022).

4.1.2. The Confidi in Pre-Pandemic Periods

During 2019, the activity of national guarantee consortia showed a decrease in the number and volume of guarantees provided during the year and in the accumulated total.

From 2012 to 2019, the number of confidi steadily decreased, dropping from 642 to 298 (-53.6%), as shown in Table 1. This decline was observed across both supervised confidi (-48.3%) and the so-called "minor confidi" (-54.1%). In 2019, the reduction was 15.1%, with a more significant drop for supervised confidi (-16.7%) compared to minor confidi (-14.9%). The situation in Sardinia appeared slightly less critical, with 15 "minor" confidi registered and 8 canceled from the previous section. The supervised confidi registered under Art. 106 TUB numbered two (Unifidi Sardegna 2020). Many of the closures were the result of mergers between non-supervised confidi or incorporations into supervised and non-supervised confidi. However, this trend did not alleviate the worrying signals from the sector, as most mergers appeared to be responses to the crisis rather than proactive development strategies.

The volume of guarantees issued by confidi also followed a decreasing trend, with a 42.4% drop from 2012 to 2018, more pronounced for supervised confidi (-45.1%) than for "minor confidi" (-36.3%). The volume of guarantees almost halved over the same period, and the 2019 data confirmed this ongoing trend.

Several factors may explain this decline, including the rationing of credit to micro-, small-, and medium-sized enterprises (mSMEs), the disintermediation of credit guarantee

consortia by banks through the direct use of FCG guarantees, and the growing preference for automated credit assessment models based on algorithms rather than the traditional discretionary evaluations by analysts that involve soft information. Algorithm-based models tend to favor companies with large amounts of hard data, referred to as "digitally transparent", over disadvantaging companies with limited hard data, known as "digitally opaque". These latter companies are often the ones most in need of financial support and guarantees from credit guarantee consortia. In 2019, disintermediation by banks was particularly evident, with a significant increase in the direct use of FCG guarantees compared to counter-guarantees or co-guarantees, as confirmed by the Fund's data (Fondo di Garanzia 2024a; Fondo di Garanzia per le PMI 2020).

Year	Supervised confidi	"Minor" confidi	Total
2012	58	584	642
2013	60	557	617
2014	62	468	530
2015	56	452	508
2016	40	441	481
2017	38	401	439
2018	36	315	351
2019	30	268	298

Table 1. Dynamics of confidi in Italy from 2012 to 2019 (number of intermediaries).

Source: Bank of Italy.

The data analysis revealed a significant reduction in the incidence of counter-guarantees + co-guarantees on the total guarantees issued by the FCG (Table 2), which decreased from 31.5% to 14% over the last six years. In parallel, the share of direct guarantees provided by banks rose sharply, from 68.5% to 86%. A similar trend was observed at the regional level in Sardinia.

Table 2. Central Guarantee Fund's total amount of guarantees accepted in 2014–2019 in Italy (thousands of euros).

Year	Direct Guarantee Amount	Direct Guarantee %	Counter-Guarantee + Co-Guarantee Amount	Counter-Guarantee + Co-Guarantee %	Total
2014	9,794,161	68.5%	4,504,836	31.5%	14,298,996
2015	10,862,549	72.1%	4,202,344	27.9%	15,064,893
2016	12,918,729	77.3%	3,783,997	22.7%	16,702,727
2017	14,140,840	81.0%	3,320,734	19.0%	17,461,573
2018	16,057,493	83.1%	3,256,487	16.9%	19,313,980
2019	16,657,146	86.0%	2,718,550	14.0%	19,375,697

Source: own computations on FCG data.

The reform of the FCG introduced in March 2019 (Fondo di Garanzia 2024b), which was strongly supported by confidi and their associated systems, aimed to mitigate this trend. The reform introduced a graduated guarantee coverage based on the requesting company's risk profile, prioritizing the riskiest companies representing the majority of confidi members. Despite these more favorable conditions, disintermediation continued in 2019, with no signs of reversal.

This ongoing disintermediation can be attributed to banks' tendency to overlook the SME segment, viewing it as unprofitable and difficult to evaluate through digital systems due to the limited availability of reliable quantitative data (hard information). Additionally, the involvement of confidi often entails extra costs for companies and complicates banking procedures. A comparative analysis of the transparency documents published by the most relevant Sardinian confidi suggested a mitigating effect on average bank pricing of around 1.5%, based on conventional conditions, which was the net of an average annual

guarantee commission of approximately 1.2 percentage points on the nominal amount of the individual loan. However, it is essential to note that the disintermediation trend persisted throughout the year, as banks continued offering conditions equal to or even more favorable than those set by existing agreements with confidi.

4.2. The Pandemic Explosion: 2020

In 2020, based on macroeconomic data (Table 3) and the assessments by the Centro Studi Confcommercio (2024), the contraction in economic activity due to the SARS-CoV-2 pandemic was the worst since the Second World War despite the actions to combat the pandemic implemented by the government (De Vincenzo 2021).

Variables	2008–2018	2018-2019	2019–2020
GDP	-0.4	-0.3	-8.9
Consumption	-0.2	0.4	-11.7
Investments	-0.3	1.1	-9.1
Employed (ULA $ imes$ 1000—var. ass.)	-981	28	-2.457
Debt as a % of the GDP		-1.6	-9.5
Public debt as a % of the GDP		134.8	155.6
Source: Confrommarcia Research Contro			

Table 3. Some macroeconomic variables of Italy (average % variations).

Source: Confcommercio Research Centre.

4.2.1. Financial Intermediation in 2020

According to the Bank of Italy, Italian banks continued to respond to businesses' demands for financing in 2020 (Table 4a,b). Financing conditions remained generally favorable thanks to the continued support from monetary policies and public guarantees (Banca d'Italia 2021; Desogus and Venturi 2023).

Table 4. (a) Financing in Italy as of end 2019 (millions of euros). (b) Financing in Italy as of end 2020 (millions of euros).

		(a)		
Areas	Total Economy	Private Sector	Production Activities	Consumer Families
Italy	1,756,033	1,253,847	708,195	545,652
Mezzogiorno	250,349	225,376	100,435	124,941
Sardinia	24,573	20,289	8817	11,471
Cagliari	7938	7114	3099	4015
Nuoro	2252	2117	1075	1042
Oristano	1712	1623	749	873
Sassari	9539	6424	2818	3606
Sud Sardegna	3132	3012	1076	1935
		(b)		
Areas	Total Economy	Private Sector	Production Activities	Consumer Families
Italy	1,764,349	1,300,791	750,521	550,270
Mezzogiorno	253,025	228,329	103,964	124,365
Sardinia	25,292	20,871	9355	11,517
Cagliari	8400	7522	3476	4046
Nuoro	2258	2124	1073	1051
Oristano	1725	1639	763	876
Sassari	9771	6572	2948	3624
Sud Sardegna	3137	3015	1095	1920

Source: own computations on data from ABI Research Office and Bank of Italy.

According to the ABI Research Office (2023), in 2020, total loans grew by 4.1% in Italy and by +4.3% in Sardinia, a growth slightly higher than the national average for

production activities also, for which the Sardinian figure (+8.8%) was close to the national figure (+8.4%).

The Bank of Italy data reported in Tables 5 and 6 show that despite the pandemic, the reduction in Italian banks' NPLs continued in 2020. This reduction involved loans to households and businesses and was supported by the moratoriums and guarantees on new loans introduced by the government to ease access to credit, together with the long-term horizon in the risk weighting of loans allowed by the supervisory authorities.

Areas	Sufferings	Probable Non- Fulfilments	Overdue/Impaired Loan Overdrafts	Total
Italy	46,502	47,365	3120	96,987
Mezzogiorno	11,505	8951	1148	21,605
Sardinia	1267	749	100	2116

Table 5. Gross impaired loans, Dec 2020 (millions of euros).

Source: Unifidi Sardegna based on Bank of Italy data.

Table 6. Bad loans/loans ratio, Dec 2020.

Areas	Bad Debt/Loans (Production Activities)		
Italy	5.09		
Mezzogiorno	8.13		
Sardinia	12.09		
Cagliari	11.16		
Nuoro	11.38		
Oristano	5.63		
Sassari	14.42		
Sud Sardegna	13.99		
•			

Source: Unifidi Sardegna based on Bank of Italy data.

4.2.2. The Confidi in 2020

The government interventions significantly increased the importance of public guarantees. Initially, they favored more significant activity from *confidi*, whose guarantees could have benefited from greater coverage by the FCG than the direct guarantee, producing a more significant weighting effect for the banks. However, while these measures indeed increased the use of the fund, the results were more evident in the number and volume of transactions directly supported by the public guarantee. In contrast, the transactions intermediated by *confidi* recorded a more marginal increase.

As highlighted in Table 7, at a national level, the impact of the pandemic led to an increase in direct guarantees of approximately 8.8 times, while counter-guarantees were less than doubled (1.9 times) (Dell'Atti and Lopes 2020; Unifidi Sardegna 2021).

Table 7. Operations of the Central Guarantee Fund for 2019–2020 in Italy (thousands of euros).

Year	Direct Guarantee Amount	Direct Guarantee %	Counter- Guarantee + Co- Guarantee Amount	Counter- Guarantee + Co- Guarantee %	Total Amount
2019	11,745,677	88.00%	1,596,453	12.00%	13,342,131
2020	102,863,646	97.10%	3,057,069	2.90%	105,920,715

Source: own computations on FCG data.

At the Sardinian regional level, the trend mirrors the national one but with a more marked decrease (25.8%) in the last year.

Focusing on the two primary types of operations benefiting from the FCG guarantee in 2020 (Table 8), it was found that operations with a value of less than EUR 30 thousand, which received 100% coverage, accounted for 63.56% of the accepted requests, with an average value of EUR 19.6 thousand each (Fondo di Garanzia 2024a). Although they represented a significant portion of the accepted requests, they contributed to a lesser extent (15.7%) to the total financing guaranteed by the Fund during the same period (Fondo di Garanzia per le PMI 2021).

Table 8. Operations of the Central Guarantee Fund for 2015–2020, Sardinia (total amount of guarantees issued in thousands of euros).

Year	Direct Guarantee Amount	Direct Guarantee %	Counter-Guarantee + Co-Guarantee Amount	Counter-Guarantee + Co-Guarantee %	Total
2015	89,402	71.10%	36,290	28.90%	125,692
2016	107,723	70.30%	45,561	29.70%	153,284
2017	113,638	67.70%	54,300	32.30%	167,938
2018	119,867	67.70%	57,196	32.30%	177,063
2019	129,395	69.30%	57,430	30.70%	186,825
2020	1,468,782	95.10%	75,107	4.90%	1,543,889

Source: own computations on FCG data.

As previously mentioned, these interventions were specifically designed to minimize bureaucratic formalities and provide an almost automatic evaluation method for immediate credit availability.

These operations, primarily managed by banks, excluded confidi from an important target group, namely micro-, small-, and medium-sized enterprises (mPMI), which had often been neglected or excluded by banks due to the difficulties in automating credit assessment processes. These difficulties stemmed from the complexity of digital information for individual firms, partnerships, artisan businesses, start-ups, and companies with few years of activity (De Socio et al. 2020; Nadotti 2020). More than others, these businesses required both guarantees and additional support to access credit. As a result, the role of confidi was further weakened, limiting their growth compared to the potential offered by the SARS-CoV-2 measures.

However, in 2020, confidi increased their reinsurance activity, which grew by 91.5% at the national level and 30.8% at the regional level. The increase in guaranteed volumes is mainly due to the rise in the guarantee quota per individual loan rather than an increase in the number of guaranteed transactions over the year. The numerical distribution between major and minor consortiums shows that there were 32 major consortiums, representing 15.3% of the total providers of consortium guarantee services in the market, and 177 minor consortiums, representing 84.7%. Although the major consortiums are numerically a minority, as of the end of 2020, they held a stock of guarantees amounting to EUR 6.7 billion, representing approximately three-quarters of the total. In contrast, the minor consortiums, with a total stock of guarantees of just over EUR 2.4 billion, saw an 8% growth compared to the previous year.

4.3. The Second Pandemic Year: 2021

In 2021, the improvement of vaccination campaigns, which were remarkably rapid in advanced countries, led to significant progress in overcoming the global pandemic, as highlighted by the Banca d'Italia (2022b). This enabled a relaxation of mobility restrictions and a recovery in global demand (Armantier et al. 2021), supporting a global growth recovery of +6.1% and a rebound in international trade.

The slowdown in production during the pandemic led to global supply shortages (Reis 2022) and contributed to inflation due to rising commodity prices, especially for energy and food. In the Euro area, GDP grew significantly in 2021 after the severe contraction in 2020. However, toward the end of the year, economic activity was partially affected by

persistent supply difficulties in global distribution chains and rising energy prices, being exacerbated by the onset of tensions between Russia and Ukraine.

Italy, in 2021, saw the easing of restrictions, leading to a 6.6% growth in the national GDP, allowing for the recovery of two-thirds of the exceptional contraction in 2020 caused by the health crisis. This recovery was widespread across the country, with the northeast experiencing the highest growth (7.2%), followed by the northwest (6.8%), the center (6.1%), and the south (5.7%). Although the number of employed people and the total hours worked increased, both remained below pre-pandemic levels (ISTAT—Istituto Nazionale di Statistica 2024).

During 2021, inflation rose to 1.9%, being mainly driven by higher energy costs, after being nearly negligible in 2020. In contrast, due to moderate wage growth, core inflation remained under control (Unifidi Sardegna 2022). While consumption grew significantly, it only partially recovered from the sharp contraction in 2020, especially in tourism and leisure-related spending. Despite a decrease from the highs of 2020, the propensity to save remained high compared to the levels of the past twenty years (ISTAT—Istituto Nazionale di Statistica 2022, Rapporto annuale). Real estate wealth increased in 2021, in line with rising house prices.

Production activity resumed growth, particularly in industry and construction, while services grew more smoothly. However, only the construction sector fully recovered its pre-pandemic level, being boosted by tax incentives for housing redevelopment. The easing of containment measures supported the recovery but was also constrained by ongoing difficulties in supplying intermediate products and rising raw material costs, especially energy.

The birth rate of businesses in 2021 returned to pre-crisis levels, and the exit rate from the market decreased, especially in sectors that benefited most from public support measures (Centro Studi Confcommercio 2024). Overall, households' financial conditions improved due to growth in disposable income and increased asset value. Households' loans and consumer credit grew, although at slower rates compared to the pre-pandemic period. Corporate profitability improved, although still slightly below 2019 levels, thanks to abundant liquidity that reduced the need for new loans (Ufficio Studi ABI—Associazione Bancaria Italiana 2024).

4.3.1. Financial Intermediation in 2021

The improvement in macroeconomic conditions and government support measures in response to the pandemic positively impacted bank balance sheets in 2021. However, the gradual end of government support measures and the Russia–Ukraine war induced significant uncertainty for credit intermediaries' profitability and asset quality.

Bank lending slowed significantly due to lower demand for financing from businesses, while lending to households accelerated, being driven by the growth in mortgages for house purchases; consumer credit remained weak (Banca d'Italia 2021, 2022b).

The ratio of new non-performing loans to total performing loans remaining limited and the volume of disposals still being high made the share of non-performing loans in total financing fall to a lower level than before the global financial crisis and to stay in line with the average of major Euro area countries.

Overall, collections continued to grow, albeit slower than in 2020. Bank profitability increased, returning to pre-health emergency levels, mainly due to the reduction in loan write-downs, which had been very high in 2020, and the increase of non-interest income (notably commissions), while the interest margin slightly decreased (Ufficio Studi ABI—Associazione Bancaria Italiana 2024).

4.3.2. The Confidi in 2021

In 2021, the amount of accumulated guarantees held by the confidi under Art. 106 TUB slightly increased, reaching EUR 8 billion, with impaired loans accounting for 22% of the total. The sector's profitability was influenced by high fixed costs, which absorbed

85% of the intermediation margin, a 2.3% increase. The ratio between own funds and risk-weighted assets remained high at 33% (Camera di Commercio Industria Artigianato e Agricoltura di Torino 2022).

By the end of 2021, Italian confidi maintained a stock of approximately EUR 9.1 billion in accumulated guarantees, with significant geographical disparities. The northern regions of Italy exhibited a marked concentration of guarantees. During 2021, Italian confidi issued around EUR 2.85 billion in guarantees, with an even more pronounced polarization in the northern regions. Although confidi in northern Italy represented only a third of the total, they issued 57% of the overall guarantee flow, totaling EUR 1.6 billion out of EUR 2.8 billion. In contrast, the southern regions, which host more than half of the Italian confidi, issued only approximately 15% of the overall guarantees.

The confidi categorization revealed that the largest confidi provided 84.31% of the total guarantees, amounting to EUR 2.4 billion, while the smaller confidi contributed just 15.69% (Unifidi Sardegna 2022; Camera di Commercio Industria Artigianato e Agricoltura di Torino 2022).

As shown in Table 9, at the national level in 2021, guarantees issued by the FCG (Fondo di Garanzia) decreased by 36.1%, falling from EUR 105.9 billion to EUR 67.6 billion. Both direct guarantees and combined counter-guarantees and co-guarantees saw declines in absolute value and percentage terms, with direct guarantees experiencing a steeper decline of 36.6% compared to those intermediated by confidi. Notably, the share of guarantees intermediated by confidi showed a slight increase, rising from 2.9% to 3.6% of the total (Fondo di Garanzia 2024b; Fondo di Garanzia per le PMI 2022).

Table 9. Operations of the Central Guarantee Fund for 2020–2021 in Italy (thousands of euros).

Year	Direct Guarantee Amount	Direct Guarantee %	Counter-Guarantee + Co-Guarantee Amount	Counter-Guarantee + Co-Guarantee %	Total Amount
2020	102,863,646	97.10%	3,057,069	2.90%	105,920,715
2021	65,191,455	96.40%	2,450,385	3.60%	67,641,840

Source: own computations on FCG data.

In Sardinia, during 2021, the trend mirrored the national one (Table 10), with a 30.9% decrease in the total guarantees issued by the Fund, dropping from EUR 1.54 billion to EUR 1.07 billion. However, it is worth noting that the behavior of direct guarantees differed from the aggregate guarantees intermediated by the confidi: while direct guarantees saw a decrease of 33.6%, the aggregate of counter-guarantees and co-guarantees increased by EUR 16.5 million, rising from EUR 75.1 million to EUR 91.6 million, a 22% increase. Consequently, the share of the aggregate guarantees in the total guarantees issued by the Fund in Sardinia rose from 4.9% to 8.6% (Unifidi Sardegna 2022; Fondo di Garanzia per le PMI 2022).

Table 10. Operations of the Central Guarantee Fund for 2020–2021 in Sardinia (thousands of euros).

Year	Direct	Direct	Counter-Guarantee +	Counter-Guarantee +	Total
	Guarantee Amount	Guarantee%	Co-Guarantee Amount	Co-Guarantee %	Amount
2020	1,468,782	95.10%	75,107	4.90%	1,543,889
2021	975,787	91.40%	91,680	8.60%	1,067,467

Source: own computations on FCG data.

Although this figure represents only 1.5% of the Fund's total assets in Sardinia, it still demonstrates a notable increase in guarantees intermediated by Sardinian confidi, reflecting the dynamism of local economic activity.

4.3.3. Minor Confidi

The analysis of the segmentation by quartiles of the consortiums registered in the list maintained by the "Organismo Confidi Minori" (OCM), based on the guarantee flows issued in 2020 (Vescina 2021), reveals that the efficiency of the consortium guarantee instrument increases with the size of the consortium's assets (Table 11a–c).

Table 11. (a) Assets guaranteed in 2020, distribution by quartiles (EUR). (b) Flows of guarantee stock in 2020, distribution by quartiles (EUR). (c) Guarantee stock in 2020, distribution by quartiles (EUR).

	(a)		
Quartiles	Net Assets	% of Total	Average Net Assets	
1st quartile	32,239,841	3.10%	716,441	
2nd quartile	89,684,847	8.50%	1,992,997	
3rd quartile	211,059,395	20.00%	4,690,209	
4th quartile	722,064,817	68.40%	16,045,885	
Total	1,055,048,900	100.00%	5,861,383	
	(b)		
Quartiles	Flows	% of Total	Average Flow	
1st quartile	1,924,587	0.20%	42,769	
2nd quartile	19,343,801	2.30%	429,862	
3rd quartile	82,984,646	10.00%	1,844,103	
4th quartile	727,204,634	87.50%	16,160,103	
Total	831,457,668	100.00%	4,619,209	
	(c))		
Quartiles	Stock	% of Total	Average Stock	
1st quartile	33,572,623	1.30%	746,058	
2nd quartile	152,925,702	6.10%	3,398,349	
3rd quartile	372,846,045	15.00%	8,285,468	
4th quartile	1,930,287,945	77.50%	42,895,288	
Total	2,489,632,316	100.00%	13,831,291	

Source: OCM.

The ratio between guarantee stock and net assets is approximately 1 to 1 in the first quartile, while it reaches 1 to 2.65 in the fourth quartile. This result signals the need for capital-strengthening measures of credit consortia.

4.4. The Post-Pandemic Recovery: 2022

As reported by the Bank of Italy (2023), 2022 economic activity in advanced countries slowed due to the persistent effects of high inflation and the ongoing conflict in Ukraine. In China, economic weakness became evident, particularly from autumn onwards, as measures were implemented to counter what appeared to be a new wave of the COVID-19 pandemic. International trade suffered from a significant global demand slowdown, contributing to a moderate decrease in oil prices. This trend was even more pronounced for natural gas prices in Europe, which, although still high, saw a notable decline.

In the Euro area, the GDP mainly remained stable in the final quarter of 2022, while consumer inflation remained elevated, reaching 9.2% in December. The European Central Bank raised its official rates by 75 basis points in October and by an additional 50 basis points in December, signaling that further significant and steady increases would be necessary in the following quarters to return inflation to its medium-term objective (Diebold et al. 2023).

In Italy, the Bank of Italy (Unifidi Sardegna 2023) reported a weakening of economic activity in the final months of 2022. This was driven by a slowdown in the recovery of services and a decrease in industrial production. Household spending also decelerated despite government measures aimed at mitigating the impact of high energy prices and

supporting disposable income, particularly for the more vulnerable population segments. The propensity to save declined, returning to pre-health-emergency levels.

In the October–December quarter, exports of goods remained stable, while imports fell, contributing to a widening of the current account deficit, mainly due to the worsening energy deficit. Industrial production contracted in the fourth quarter due to high energy costs and reduced demand. After expanding in the second and third quarters, activity in the tertiary sector slowed down.

Uncertainty surrounding the ongoing conflict in Ukraine and tighter financial conditions negatively impacted investment spending and the outlook for the real estate sector. Inflation reached new highs in the autumn months, driven mainly by energy prices, which continued to push up the cost of other goods and services. This contributed to a moderate rise in core inflation. Despite government interventions in the energy sector, consumer price inflation only eased at the end of the year (ISTAT—Istituto Nazionale di Statistica 2024).

The Confidi in 2022

As of 31 December 2022, there were 200 Italian confidi, consisting of 32 major confidi and 168 minor confidi. The major confidi were distributed across 15 Italian regions, with a concentration in the north (20 out of 32). At the same time, approximately one-third were located in the center and south (six confidi in each area). Regarding geographical distribution, the minor confidi were spread across the entire country, with at least one present in each region. The highest concentration of minor confidi was in the southern regions (94 out of 168), followed by the north (48 confidi) and the center (26 confidi) (Camera di Commercio Industria Artigianato e Agricoltura di Torino 2023).

At the end of 2022, Italian confidi held a total of EUR 8.4 billion in guarantees, reflecting a decrease of 7% compared to 2021, when the cumulative stock was EUR 9.1 billion. However, the distribution of outstanding guarantees differed from the geographical distribution of the confidi. During the year, Italian confidi issued approximately EUR 2.8 billion in new guarantees, most of which came from northern Italy. In contrast, the southern regions, where about half of the Italian confidi are located, accounted for only 14% of the guarantees issued.

The share of guarantees issued by the largest confidi was 85%, while the smaller confidi accounted for 15%. The data indicate a trend toward a greater concentration of guarantee stocks in the larger confidi and, consequently, the northern regions. Specifically, while the southern regions held 21% of the guarantee stocks at the end of 2022, they issued only 14% of the flows during the year. Meanwhile, northern regions, which held 51% of the stocks at the end of 2021, issued 57% of the flows in 2022. According to the Bank of Italy, in 2022, the amount of guarantees issued by the largest confidi decreased by 11%, reaching EUR 6.9 billion. The percentage of impaired guarantees to the total issued was 19.5%, compared to 22.2% in 2021.

The sector's profitability remained limited due to the high impact of fixed costs. However, the ratio of own funds to risk-weighted assets remained high at 36.4% (Banca d'Italia 2022a, 2023b; Camera di Commercio Industria Artigianato e Agricoltura di Torino 2023). This emerging picture suggests that the gradual reduction in the size of the smaller confidi may be partly due to the end of the extraordinary pandemic-related measures, which had directly affected confidi operations, increasing their activity in some areas. Meanwhile, the consolidation process among the larger confidi continued, strengthening certain players in terms of available resources.

At the same time, there was a growing diversification of commercial activities, with a significant shift toward "residual" activities. These were seen as the only sustainable economic and financial viability sources for the confidi. Consequently, the revenues of the larger confidi grew, driven by these activities, which did not require provisions like the issuing of guarantees. This option, however, was excluded for smaller confidi, whose operations remained focused solely on issuing collective guarantees. In 2022, there was a further and significant contraction in the guarantees issued by the FCG, with a national decrease of 37.7%, from EUR 67.6 billion to EUR 42.1 billion (34.2% less than the change recorded in 2020–2021). Both direct guarantees and the aggregate of counter-guarantees and co-guarantees saw reductions in both absolute and percentage terms (Table 12). However, the decrease was more pronounced for direct guarantees (-38.6% compared to -36.6% in the previous two-year period), while the aggregate intermediated by confidi recorded a decrease of 13.1% (compared to -19.8% in the previous two-year period). As a result, the incidence of the guarantees intermediated by confidi increased from 3.6% to 5.1%.

Table 12. Operations of the Central Guarantee Fund for 2021–2022 in Italy (thousands of euros).

Year	Direct Guarantee Amount	Direct Guarantee %	Counter-Guarantee + Co-Guarantee Amount	Counter-Guarantee + Co-Guarantee %	Total Amount
2021	65,191,455	96.40%	2,450,385	3.60%	67,641,840
2022	40,006,760	94.90%	2,129,342	5.10%	42,136,102

Source: own computations on FCG data.

In Sardinia, the general trend in 2022 showed a more significant decrease than the national average, with a reduction of 40.7% in the total guarantees issued by the Fund, which fell from EUR 1.07 to 0.63 billion. As in the previous year, a decrease was observed in both components. However, while the direct guarantee recorded a decrease of -42.6% (Table 13), which was higher than the overall reduction, the aggregate of guarantees intermediated by the guarantee consortia decreased to a lesser extent (-20.6%). This led to an increase in the share of the aggregate guarantees from 8.6% to 11.5% of the total guarantees issued on the island by the Central Guarantee Fund between 2021 and 2022.

Table 13. Operations of the Central Guarantee Fund for 2021–2022 in Sardinia (thousands of euros).

Year	Direct	Direct	Counter-Guarantee +	Counter-Guarantee +	Total
	Guarantee Amount	Guarantee %	Co-Guarantee Amount	Co-Guarantee %	Amount
2021	975,787	91.40%	91,680	8.60%	1,067,467
2022	559,785	88.50%	72,757	11.50%	632,542

Source: own computations on FCG data.

Notably, the data regarding the reduction in the volume of counter-guarantees intermediated by Sardinian guarantee consortia in 2022 are significant, amounting to a decrease of EUR 18.9 million (Unifidi Sardegna 2023; Fondo di Garanzia per le PMI 2023).

5. Comparison Between General Averages and a Sample of Regional Confidi Supported Firms

5.1. Data

Our study examines the role of credit consortia in supporting SMEs during the SARS-CoV-2 crisis by comparing the financial performances of firms financed through credit consortia with those funded directly by banks. Using a descriptive methodology, we assess key financial indicators across three distinct periods—pre-crisis, during the crisis, and post-crisis—to determine which financing channel offered more effective support during economic uncertainty. This comparison of firms with and without credit consortium backing allows us to evaluate performance trends throughout the selected timeframe.

The dataset consists of companies registered in the Chambers of Commerce of the Sardinia region and operating in the business sectors listed in the following Table 14.

Table 14 presents data on firms supported by confidi credit consortia during the pandemic, which were selected based on the most frequent industry sectors as defined by the Italian ATECO classification, alongside a larger sample of firms from the AIDA database, which were selected using the same industry sector criteria.

Ateco Code	Description	Totals	Aida Firms	Confidi Supported
41.20.00	Construction of residential and non-residential buildings	660	652	8
47.77.00	Retail sale of watches, jewelry, and silverware	22	18	4
49.41.00 56.10.11	Road freight transport Full-service restaurants	178 315	175 305	3 10

Table 14. Data description (number of firms by sector and type).

Source: own computations on data from ABI Research Office and Bank of Italy.

Of the total companies active in these selected sectors in Sardinia in 2022, a "rational" sampling approach was applied to the general database. This process was carried out through the AIDA—Analisi Informatizzata delle Aziende Italiane (2024) platform, focused on extracting limited liability companies to mitigate the digital opacity typically associated with sole proprietorships and partnerships.

Starting with an initial pool of 10,745 companies extracted from the AIDA database, only those that had filed their 2022 financial statements were considered. From this, 1150 companies were selected, including only those with complete data and no missing values. These firms were further filtered based on having total assets of less than EUR 5 million, creating a sample that closely mirrors the characteristics of confidi-supported companies, thereby enabling a more accurate comparison between the two groups.

The final sector breakdown of this sample consists of 652 firms under code 41.20.00, 18 under code 47.77.00, 175 under code 49.41.00, and 315 under code 56.10.11. In contrast, the confidi sample includes 25 companies distributed as follows: 8 under code 41.20.00, 4 under code 47.77.00, 3 under code 49.41.00, and 10 under code 56.10.11. The comparatively lower representation of credit consortium-assisted companies in the construction sector is attributed to significant fluctuations in financial liabilities tied to the "super bonus" for construction introduced by the Italian government in May 2020. This policy did not involve confidi intermediaries.

In particular, the data relating to 2017–2018–2019 (pre-pandemic), 2020–2021 (pandemic), and 2022 (post-pandemic) were identified. The balance sheet elements considered were the total assets, capitalization, liquidity, total debts, sales volume, ROI, and cost of debt. All these variables were extracted from the firms' year-end financial statements; consequently, the values are nominal and not adjusted for inflation.

The total assets per firm include current and non-current assets. The capitalization value is represented by the ratio of equity (the net worth attributable to the firm's shareholders or owners) on total assets. The liquidity ratio refers to the ratio of liquid assets—assets that can be quickly and easily converted into cash with little to no loss in value—over total assets. The total debts represent the total amount of money a firm owes, encompassing both short-term and long-term liabilities such as loans, bonds, and accounts payable. The sales volume indicates the total revenue a firm generates from its sales of goods or services during the year considered, reflecting business activity and market demand. The return on investment (ROI) measures the EBITDA over total assets, indicating how effectively the firm generates returns. Finally, the cost of debt represents the ratio of financial charges, including interest expenses and other associated borrowing costs, over the total debt, reflecting the firm's expense for financing through debt.

The AIDA sample values were first calculated by industry sector and then synthesized using a weighted average, with the same weighting as the confidi-supported sample. This process allows for an accurate proportional comparison of the values between the two samples, with the AIDA sample balanced against the confidi-supported one.

The financial indicators considered in this study are inter-related and reflect a firm's overall financial health, operational efficiency, and risk profile.

The evolution of total assets reflects the business's dimension and its expansion level. Growth in total assets indicates an increase in the scale of operations, which may be associated with higher investment in production capacity, strategic assets, or acquisitions. The capitalization ratio, on the other hand, captures financial autonomy. A higher capitalization ratio suggests that the firm is relying more on equity financing than on debt, which lowers financial risk. This makes the firm more resilient to economic shocks, as it has a greater cushion against financial distress.

Liquidity and total debts are inversely related. Liquidity measures a firm's ability to quickly meet its short-term obligations by converting assets into cash. Firms with high levels of debt, particularly short-term liabilities, may struggle to maintain liquidity because a significant portion of their resources is used for debt servicing rather than remaining available as cash or liquid assets. High debt levels can strain the firm's finances, especially if the sales volume decreases or operational challenges emerge, which is often the case during periods of economic uncertainty like the SARS-CoV-2 pandemic. In this context, government intervention plays a critical role in shaping debt levels, with public financial support for SMEs helping to ease debt burdens and improve liquidity.

The sales volume and ROI are closely connected, as they both reflect a firm's profitability and the efficiency of its resource utilization. A higher sales volume will likely produce higher returns on its invested assets, indicating greater operational efficiency and improved financial performance. Conversely, the ROI may also drop if sales volumes decline, signaling reduced profitability.

Liquidity is also tied to the cost of debt, specifically to the level of interest rates. High interest rates increase the cost of debt, making it harder for a firm to manage its liabilities and maintain liquidity. A firm with poor liquidity may struggle to secure financing, especially during economic distress.

In summary, these financial indicators are part of an interconnected cycle. Effective liquidity management, debt, asset utilization, and sales growth can improve profitability and the ROI, while the capitalization ratio determines a firm's financial risk profile. The SARS-CoV-2 pandemic likely disrupted these relationships, introducing business challenges and opportunities, particularly in managing liquidity, securing debt financing, and maintaining operational efficiency. The pandemic underscored the importance of strategic decision making and financial planning to navigate periods of financial distress.

5.2. Evolution of the Sardinian Companies During and After the SARS-CoV-2 Crisis

The following graphs show the evolution of the main financial indicators from 2017 to 2022. We use them to compare the two samples and understand how each group was influenced by the pandemic, based on whether they were supported by credit consortia or not.

In the following figures, the two samples show similar ongoing trends, as the framework is the same for the credit consortium-supported and non-supported firms. Nevertheless, some differences in the evolution are evident, and our focus is on these differences to evidence the role of credit consortia.

The total assets volume evolution, as reported in Figure 5, shows that the pre-pandemic level was similar for the general sample and the confidi-supported firms. In contrast, during and after the pandemic, the general sample registered growth, but the confidi-supported sample reported more significant growth, almost doubling the value of their total assets. This rapid growth suggests that the confidi-supported firms already had potential in the previous years, which was only expressed as they had the liquidity to finance it.

The capitalization level (Figure 6), measured by the equity/(total liabilities + equity ratio), shows an almost flat evolution in the pre-pandemic and pandemic years, staying in the 26% range from 2017 to 2021 for the general sample and in the range 15–16% for those that were confidi supported in the same years, followed by an irrelevant reduction in the post-pandemic, which reduced the ratio value from 26% in 2021 to 25% in 2022 for the general sample and from 15% to 13% for the confidi-supported sample. This evolution

evidenced the lower capitalization rates of the confidi-supported firms, confirming their patrimonial weakness and the need for external support in their bank–firm relationship. The evolution also shows the reduction in capitalization, mainly related to the availability of cheap debt (see Figures 7 and 8) and the government's financial support granted as a response to the pandemic problems.



Figure 5. Average total assets per firm (EUR) for the years 2017–2022.



Figure 6. Capitalization: equity/(total liabilities + equity), years 2017–2022.

The liquidity rate evolution, as reported in Figure 7, shows even more clearly the impact of government policy interventions in support of firms. The liquidity ratio values, ranging from 12% to 14% in the pre-pandemic years for the general Aida sample, rose to 16% in 2021. Instead, the confidi-supported firms show a more significant liquidity improvement, starting from around 6% for the pre-pandemic period and almost reaching 12% in 2021, more than doubling the previous values. This graph also evidences the deficient liquidity levels of the confidi-supported firms, which greatly benefited from the government's pandemic support policies but, even in the moments of higher support availability, could not reach the pre-pandemic minimum liquidity levels of the general sample. This low liquidity confirms the financial weakness of the confidi-supported firms.

The debt incidence rates reported in Figure 8 show the total debt level evolution in the considered period. It is worth noting that the total debt level includes bank and

commercial debts. Thus, the high level of debt of the confidi-supported firms does not evidence the easy and more extensive credit availability of these firms but rather the low capitalization and high level of commercial debts. It is confirmed by the rising debt level in the post-pandemic, which is related to the need to restart the business activity at the previous (or higher) levels (see also Figure 9), so needing higher financial coverage but also allowing for more commercial debts.









The evolution of sales, as reported in Figure 9, shows a very interesting post-pandemic opportunity for the considered firms, with the general sample average ranging from EUR 330 to 400 thousand in the pre-pandemic years and reaching EUR 532 thousand in the post-pandemic, but even more for the confidi-supported sample, which ranged from EUR 600 to 700 th. in the pre-pandemic and almost doubled their sales volume in the post-pandemic period, around EUR 1200th in 2022.



Figure 9. Average sales per firm, years 2017–2022 (EUR).





The ROI evolution (Figure 10), computed as the EBITDA over the total assets, shows two sides of the considered phenomena. On the one side, the general evolution of the ROI is that of a significant reduction during the most dramatic pandemic year, 2020, reducing the return on investment from 5% in 2019 to 2.6% in 2020 considering the Aida sample, and even more for the confidi-supported sample, going from 11.9% to 8.3% in the same years, but bouncing up in 2021 for both groups nearly to the previous levels and suffering a new reduction in 2022. On the other hand, the confidi-supported firms reported higher ROI levels in all years, showing an interesting return and income capability, which contrasts with their weak financial situation, evidencing their weakness in the bank–firm relationship.

The average cost of debt (Figure 11), computed as the total financial charges over the total debt, confirms the confidi-supported firms' weakness in the bank–firm relationship, reporting higher costs than the general sample, while both samples evidence a significant reduction in debt costs related to the government's financial support during the pandemic years.



Figure 11. Average cost of debt, years 2017–2022.

5.3. Economic Significance: A Summary of the Analyses on the Role of Confidi

A comprehensive overview of the previous analyses' economic and financial significance provides a synthesis of the confidi's role and their combined effect with the counter-guarantee of the Central Guarantee Fund in supporting businesses during the pandemic period.

Such an overview highlights the general action of the government and private guarantee entities aimed at mitigating the risks of a credit crunch associated with the economic crisis brought about by the COVID-19 emergency and the lockdown measures.

Specifically, the analysis focused on comparing the resilience and performance of businesses supported by confidi versus those external or not associated with guarantee consortia revealed that businesses affiliated with confidi entered the pandemic period generally less liquid (Figure 7), weaker in their ability to generate cash flows, and less capitalized in absolute terms and relative to their liabilities (Figures 6 and 8). These characteristics depress creditworthiness, leading to adverse rating outcomes and, thus, reducing these businesses' independent abilities to secure financial leverage. The confidi's intervention proved essential to ensuring these businesses' survival in the market. Moreover, these companies demonstrated greater receptivity to fixed-asset investment strategies, enhancing their corporate structures (Figure 5).

From a strictly economic perspective, the confidi's support for these (financially fragile) businesses significantly contributed to revenue growth (even doubling in some cases, Figure 9) and achieved profitability. Although subject to para-physiological cyclical fluctuations, this profitability remained significantly higher than the external sample's (Figure 10).

In short, the confidi's role positively impacts businesses characterized by viable and scalable business models, albeit financially and patrimonially weak, less resilient to systemic crises, and more dependent on credit support. The confidi's guarantees enable these businesses to maintain their market position even during turbulent periods while facilitating their development in terms of assets, production value, and operational profitability.

6. Discussion and Conclusions

Credit consortia are essential in mitigating the asymmetry in the bank–firm relationship. Acting like a debtors' union, they help small and financially fragile firms find credit coverage for their investments and negotiate conditions. Our data confirm that the confidisupported firms are smaller, less capitalized, and less liquid but capable of high returns. This is the typical profile of small firms needing support in the firm–bank relationship, allowing confidi to help them obtain more credit in better conditions. During the SARS-CoV-2 pandemic crisis, credit consortia played a significant role, supporting Sardinian SMEs with guarantees and helping them obtain government financial support.

As highlighted by our study, debt financing has proven to be a critical resource for SMEs during challenging economic periods. Specifically, during the SARS-CoV-2 pandemic, we observed a decrease in equity levels alongside an increase in debt financing. This finding is supported by the studies of Halling et al. (2020), Acharya and Steffen (2020), Li et al. (2020), and Gopalakrishnan et al. (2022), which underscore the critical role of debt for firms during periods of uncertainty. Similarly, Khurana et al. (2024) note that SMEs often prefer debt over financial autonomy during crises, a conclusion confirmed by our research. This trend aligns with our findings that the availability of inexpensive debt, facilitated by public intervention, significantly encouraged its use as a financing mechanism during the pandemic. As our study highlights, government interventions aimed at enhancing credit access for firms played a critical role in reducing the cost of debt. Wang et al. (2024) similarly observed that such measures alleviated liquidity pressures and enabled businesses to sustain operations.

Our study further reveals that the level of debt increased during the crisis as borrowing costs declined due to government-supported credit initiatives. Additionally, we noted that firms utilizing credit consortia experienced higher debt costs than other firms. Gómez et al. (2016) also observed that credit consortium intermediaries were generally more expensive than direct bank financing due to additional layers of intermediation. However, public support programs promoting SME growth made credit consortia a vital financing channel for specific firms, particularly for those unable to access direct banking due to their small size and vulnerability. Our study also highlights the heightened vulnerability of firms supported by credit consortia, characterized by limited liquidity levels, higher average costs of debt, and significant debt exposure. This increased risk for firms relying on credit consortia was also noted by Gai et al. (2016).

In terms of performance, our analysis shows that SMEs benefiting from credit consortia exhibited more significant increases in sales and asset levels attributed to public intervention compared to the general sample. Additionally, these supported firms demonstrated higher and sustained ROI levels during the crisis than the general sample. Similar findings were reported by Khurana et al. (2024), Corredera-Catalán et al. (2021), and Briozzo and Cardone-Riportella (2016), who demonstrated a positive correlation between firm performance and the financial support received during crises. These studies collectively emphasize the untapped potential of SMEs, which can only be realized when they have access to essential financial resources to strengthen their operations and drive growth.

The pandemic was an exciting chance to study the effect of liquidity injections in small and nonliquid firms. The capability of higher investment and positive returns, which is expected by theoretical deduction, is confirmed by the evidence. The liquidity provided by the government support, filling the main gap in their financial structure, lighted a restart of the confidi-supported firms, firstly just providing higher liquidity, which, from the subsequent year, was transformed not only into higher investment but also resulting in a positive jump on sales.

The demographics of Sardinian companies in 2019–2022 and the volumes of loans and savings showed a strengthening of debt capital payments, increased collections, and a progressive improvement of the companies' aggregate net financial position in Sardinia. In short, the credit support provided by the government, banks, and credit consortia resulted in the overcoming of the crisis. It improved the real economy framework with respect to the pre-pandemic scenario.

The evolution described here suggests that the credit restrictions SMEs suffer from can be due to a market failure, such as the credit system's incapability and banks' ability to allocate credit resources properly.

This incapability to select good customers is often due to the need for more hard information from balance sheets and bank account records. The opacity of accounting

information is due to the small firms' possibility of a simplified accounting system and the significant use of cash, which reduces the information coming from the bank account. The lack of hard information is becoming more significant over time as the banks' progressive use of credit scoring quantitative models actually relies on hard information. Hence, the unavailability or mistrust of these information sources reduces the scoring model's capability to allow for firm financing.

In this sense, the role of credit consortia is very significant, as on the one side, their credit assessment mainly relies on the soft information provided by the close contact and specific knowledge of each firm, and, on the other side, backing the financing request by specific guarantees, which increase the firm scoring, allows the bank to finance the firm.

The guarantee support, which is fundamental in this framework, is the primary constraint for the credit consortium activity, as it must be covered with capital or counterguarantees. The analysis of credit consortium activity in Sardinia shows instead a progressive reduction in their activity, partially substituted by direct public guarantees. It shows a higher efficiency of larger confidi, which are also the ones under stricter supervision by the Banca d'Italia, being monitored by quantitative risk measurement and management models.

The higher efficiency of larger confidi is critical in policy terms, as it shows that a smaller number of larger credit consortia can provide more substantial support to Sardinia's small and financially fragile firms. In operative terms, the opportunity to merge or acquire the smaller ones into one or a few larger ones is a possible way to improve their capability to help Sardinia's smaller firms and to support them in their investment and growth.

The main limit of this analysis is in the small number of cases of confidi-supported firms, which are not publicly available. Future studies based on more data could extend the analysis to encompass the entire Italian context, considering firms supported by MGSs across all regions, not just Sardinia. A broader examination would allow for an assessment of regional differences in Italy and offer a more comprehensive understanding of firms' responses on a national scale.

The role of confidi could be better understood outside the context of financial distress, focusing instead on a more typical economic setting. In such an environment, evaluating their effectiveness in supporting financially stable firms and comparing their performance with direct financing channels could provide more insights into their value proposition. Given the limited research on credit consortia within the Italian economy, new studies would help improve our understanding of their function and impact in the broader financial landscape.

Author Contributions: Conceptualization, M.D. and S.Z.; methodology, M.D.; validation, M.D. and E.S.; formal analysis, M.D.; resources, S.Z.; data curation, M.D. and E.S.; writing—original draft preparation, M.D.; writing—review and editing, S.Z. and E.S.; supervision, S.Z.; project administration, S.Z.; funding acquisition, S.Z. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the Fondazione di Sardegna, progetti biennali annualità 2021, grant n. F73C22001410007.

Data Availability Statement: The original contributions presented in the study are included in the article, further inquiries can be directed to the corresponding author.

Conflicts of Interest: The authors declare no conflicts of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript; or in the decision to publish the results.

References

- Acharya, Viral Vithal, and Sascha Steffen. 2020. The Risk of Being a Fallen Angel and the Corporate Dash for Cash in the Midst of COVID. *The Review of Corporate Finance Studies* 9: 430–71. [CrossRef]
- AIDA—Analisi Informatizzata delle Aziende Italiane. 2024. Available online: https://bvdinfo.com/R0/AidaNeo (accessed on 1 June 2024).
- Armantier, Olivier, Gizem Koşar, Rachel Pomerantz, Daphné Skandalis, Kyle Smith, Giorgio Topa, and Wilbert Van der Klaauw. 2021. How economic crises affect inflation beliefs: Evidence from the COVID-19 pandemic. *Journal of Economic Behavior and Organization* 189: 443–69. [CrossRef] [PubMed]

- Banca d'Italia. 2020. Relazione Annuale 2019. Esercizio CXXVI. Available online: https://www.bancaditalia.it/ (accessed on 29 September 2024).
- Banca d'Italia. 2021. Relazione Annuale 2020. Esercizio CXXVII. Available online: https://www.bancaditalia.it/ (accessed on 29 September 2024).
- Banca d'Italia. 2022a. L'economia della Sardegna, Rapporto Annuale. Economie Regionali, 2021. Available online: https://www. bancaditalia.it/ (accessed on 29 September 2024).
- Banca d'Italia. 2022b. Relazione Annuale 2021. Esercizio CXXVIII. Available online: https://www.bancaditalia.it/ (accessed on 29 September 2024).
- Banca d'Italia. 2023a. Relazione Annuale 2022. Esercizio CXXIX. Available online: https://www.bancaditalia.it/ (accessed on 29 September 2024).
- Banca d'Italia. 2023b. Available online: https://infostat.bancaditalia.it/inquiry/home (accessed on 1 July 2023).
- Braun, Benjamin. 2020. Central banking and the infrastructural power of finance: The case of ECB support for repo and securitization markets. *Socio-Economic Review* 18: 395–418. [CrossRef]
- Briozzo, Anahí, and Clara Cardone-Riportella. 2016. Spanish SMEs' subsidized and guaranteed credit during economic crisis: A regional perspective. *Regional Studies* 50: 496–512. [CrossRef]
- Camera di Commercio Industria Artigianato e Agricoltura di Torino. 2022. I Confidi in Italia. Comitato Torino Finanza. Available online: https://www.to.camcom.it/ (accessed on 29 September 2024).
- Camera di Commercio Industria Artigianato e Agricoltura di Torino. 2023. I Confidi in Italia. Comitato Torino Finanza. Available online: https://www.to.camcom.it/ (accessed on 29 September 2024).
- Caselli, Stefano, Guido Corbetta, Doriana Cucinelli, and Monica Rossolini. 2021. A survival analysis of public guaranteed loans: Does financial intermediary matter? *Journal of Financial Stability* 54: 100880. [CrossRef]
- Centro Studi Confcommercio. 2024. Available online: https://www.confcommercio.it/documents/ (accessed on 29 September 2024).
- Columba, Francesco, Leonardo Gambacorta, and Paolo Emilio Mistrulli. 2010. Mutual guarantee institutions and small business finance. *Journal of Financial Stability* 6: 45–54. [CrossRef]
- Corredera-Catalán, Félix, Filippo di Pietro, and Antonio Trujillo-Ponce. 2021. Post-COVID-19 SME financing constraints and the credit guarantee scheme solution in Spain. *Journal of Banking Regulation* 22: 250. [CrossRef]
- Cusmano, Lucia. 2018. SME and Entrepreneurship Financing: The Role of Credit Guarantee Schemes and Mutual Guarantee Societies in supporting finance for small and medium-sized enterprises. In OECD SME and Entrepreneurship Papers. Paris: OECD Publishing, No. 1. [CrossRef]
- D'Amato, Antonio. 2020. Capital structure, debt maturity, and financial crisis: Empirical evidence from SMEs. *Small Business Economics* 55: 919–41. [CrossRef]
- Das, Ramesh Chandra. 2020. Forecasting incidences of COVID-19 using Box-Jenkins method for the period July 12-September 11, 2020: A study on highly affected countries. *Chaos, Solitons and Fractals* 140: 110248. [CrossRef]
- De la Fuente-Cabrero, Concepción, Mónica de Castro-Pardo, Rosa Santero-Sánchez, and Pilar Laguna-Sánchez. 2019. The Role of mutual guarantee institutions in the financial sustainability of new family-owned small businesses. *Sustainability* 11: 6409. [CrossRef]
- Dell'Atti, Stefano, and Antonio Lopes. 2020. Il sistema dei Confidi in Italia: Caratteristiche strutturali, patrimonializzazione e prospettive operative. *Rivista Economica del Mezzogiorno* 34: 125–52.
- De Socio, Antonio, Simone Narizzano, Tommaso Orlando, Fabio Parlapiano, Giacomo Rodano, E. Sette, and Gianluca Viggiano. 2020. *Gli Effetti Della Pandemia sul Fabbisogno di Liquidità, sul Bilancio e Sulla Rischiosità Delle Imprese*. COVID-19 Note. Rome: Bank of Italy. Available online: https://www.bancaditalia.it/media/notizia/gli-effetti-della-pandemia-sul-fabbisogno-di-liquidit-sulbilancio-e-sulla-rischiosit-delle-imprese/ (accessed on 29 September 2024).
- Desogus, Marco, and Beatrice Venturi. 2023. Stability and Bifurcations in Banks and Small Enterprises—A Three-Dimensional Continuous-Time Dynamical System. *Journal of Risk and Financial Management* 16: 171. [CrossRef]
- Desogus, Marco, and Elisa Casu. 2020. What Are the Impacts of Credit Crunch on the Bank-Enterprise System? An Analysis Through Dynamic Modeling and an Italian Dataset. *Applied Mathematical Sciences* 14: 679–703. [CrossRef]
- De Vincenzo, Alessio. 2021. Audizione sulle tematiche relative allo squilibrio della struttura finanziaria delle imprese italiane che rischia di essere determinato dalla pandemia da COVID-19. *Banca d'Italia*. Available online: https://www.bancaditalia.it/ pubblicazioni/interventi-vari/int-var-2021/De-Vincenzo-18.03.2021.pdf (accessed on 29 September 2024).
- Diebold, Francis Xavier, Minchul Shin, and Boyuan Zhang. 2023. On the aggregation of probability assessments: Regularized mixtures of predictive densities for Eurozone inflation and real interest rates. *Journal of Econometrics* 237: 105321. [CrossRef]
- Drott, Constantin, Stefan Goldbach, and Axel Jochem. 2024. Determinants of Net Transactions in TARGET2 of European Banks Based on Micro-data. *Open Economies Review*, 1–37. [CrossRef]
- Fondo di Garanzia. 2024a. Available online: https://www.fondidigaranzia.it/numeri-del-fondo/ (accessed on 29 September 2024). Fondo di Garanzia. 2024b. Available online: https://www.fondidigaranzia.it/normativa-e-modulistica/ (accessed on 1 February 2024).
- Fondo di Garanzia per le PMI. 2020. Report Regionali—Dati al 31 Dicembre 2019. Available online: https://www.fondidigaranzia.it (accessed on 29 September 2024).
- Fondo di Garanzia per le PMI. 2021. Report Regionali—Dati al 31 Dicembre 2020. Available online: https://www.fondidigaranzia.it (accessed on 29 September 2024).

- Fondo di Garanzia per le PMI. 2022. Report Regionali—Dati al 31 Dicembre 2021. Available online: https://www.fondidigaranzia.it (accessed on 29 September 2024).
- Fondo di Garanzia per le PMI. 2023. Report Regionali—Dati al 31 Dicembre 2022. Available online: https://www.fondidigaranzia.it (accessed on 29 September 2024).
- Gai, Lorenzo, Federica Ielasi, and Monica Rossolini. 2016. SMEs, public credit guarantees and mutual guarantee institutions. *Journal of Small Business and Enterprise Development* 23: 1208–28. [CrossRef]
- Glowka, Marc, Alexander Müller, Livia Polo Friz, Sara Testi, Massimo Valentini, and Stefano Vespucci. 2022. TARGET2 Analytical Tools for Regulatory Compliance. ECB Occasional Paper. (2022/300). Available online: https://www.ecb.europa.eu/pub/pdf/ scpops/ecb.op300~86041e53a4.en.pdf (accessed on 29 September 2024).
- Gopalakrishnan, Balagopal, Joshy Jacob, and Sanket Mohapatra. 2022. COVID-19 pandemic and debt financing by firms: Unravelling the channels. *Economic Modelling* 114: 105929. [CrossRef]
- Gómez, Joaquín Miñarro, Ginés Hernández Cánovas, and Pedro Martínez Solano. 2016. The effects of mutual guarantee societies on the SME'S. Universia Business Review 50: 38–53. [CrossRef]
- Halling, Michael, Jin Yu, and Josef Zechner. 2020. How Did COVID-19 Affect Firms' Access to Public Capital Markets? *The Review of Corporate Finance Studies* 9: 501–33. [CrossRef]
- Huang, He, and Ye Ye. 2021. Rethinking capital structure decision and corporate social responsibility in response to COVID-19. *Accounting and Finance* 61: 4757–88. [CrossRef]
- ISTAT—Istituto Nazionale di Statistica. 2020. Rapporto Annuale 2019. La situazione del Paese. Available online: https://www.istat.it (accessed on 29 September 2024).
- ISTAT—Istituto Nazionale di Statistica. 2022. Rapporto Annuale 2021. La situazione del Paese. Available online: https://www.istat.it (accessed on 29 September 2024).
- ISTAT—Istituto Nazionale di Statistica. 2024. Accesso alle Banche dati Febbraio 2024—Marzo 2024. Available online: https://www.istat.it/it/dati-analisi-e-prodotti/banche-dati/statbase (accessed on 29 September 2024).
- Khurana, Manpreet Kaur, Shweta Sharma, and Navneet Bhargava. 2024. The Impact of Financial Ratios and Pandemic on Firm Performance: An Indian Economic Study. *Lecture Notes in Networks and Systems* 844: 41–51. [CrossRef]
- Leone, Paola, Ida C. Panetta, and Pasqualina Porretta. 2013. Credit Guarantee Institutions, Performance and Risk Analysis: An Experimental Scoring. In *Bank Stability, Sovereign Debt and Derivatives*. Palgrave Macmillan Studies in Banking and Financial Institutions. Edited by Joseph Falzon. London: Palgrave Macmillan. [CrossRef]
- Li, Lei, Philip Edward Strahan, and Song Zhang. 2020. Banks as Lenders of First Resort: Evidence from the COVID-19 Crisis. *The Review of Corporate Finance Studies* 9: 472–500. [CrossRef]
- MEF—Ministero dell'Economia e delle Finanze. 2020–2022. La Legge di Bilancio 2020–2022 in Breve. Dipartimento della Ragioneria Generale dello Stato—Ispettorato Generale del Bilancio. Available online: https://www.mef.gov.it/index.html (accessed on 29 September 2024).
- Nadotti, Loris. 2020. Situazione finanziaria delle imprese e strumenti pubblici di mitigazione dei rischi per il sistema del credito. In COVID-19 tra Emergenza Sanitaria ed Emergenza Economica. Riflessioni dal mondo delle scienze sociali. Perugia: Morlacchi Editore, vol. 1, pp. 251–74. Available online: https://hdl.handle.net/11391/1475598 (accessed on 29 September 2024).
- Nguyen Kim, Quoc Trung. 2023. Does COVID-19 affect small and medium enterprises' capital structure in vietnam? *Cogent Economics and Finance* 11: 2190268. [CrossRef]
- Osservatorio Nazionale sulla Salute nelle Regioni Italiane. 2021. *La Pandemia in Italia e nei Paesi Europei: Ultime Evidenze*. Roma: Osservatorio Nazionale sulla Salute nelle Regioni Italiane.
- Prakash, Nisha, Aditya Maheshwari, and Aparna Hawaldar. 2023. The impact of COVID-19 on the capital structure in emerging economies: Evidence from India. *Asian Journal of Accounting Research* 8: 236–49. [CrossRef]
- Prometeia. 2024. Available online: https://www.prometeia.it/en/research-bi-reporting#analisys (accessed on 1 January 2024).
- Reis, Ricardo. 2022. The Burst of High Inflation in 2021–2022: How and Why Did We Get Here? Bank for International Settlements, Monetary and Economic Department. Available online: https://www.bis.org/publ/work1060.htm (accessed on 29 September 2024).
- Senato della Repubblica Italiana. 2023. Available online: https://www.senato.it/4513 (accessed on 29 September 2024).
- Ufficio Studi ABI-Associazione Bancaria Italiana. 2024. Available online: https://www.abi.it/ (accessed on 1 July 2023).
- Unifidi Sardegna. 2020. Bilancio D'esercizio 2019. Available online: https://www.unifidisardegna.it/ (accessed on 29 September 2023).
- Unifidi Sardegna. 2021. Bilancio D'esercizio 2020. Available online: https://www.unifidisardegna.it/ (accessed on 29 September 2023).
- Unifidi Sardegna. 2022. Bilancio D'esercizio 2021. Available online: https://www.unifidisardegna.it/ (accessed on 29 September 2023).
- Unifidi Sardegna. 2023. Bilancio D'esercizio 2022. Available online: https://www.unifidisardegna.it/ (accessed on 29 September 2023). Vescina, Salvatore. 2021. Confires Congress 2021. Available online: https://www.confires.it/confires-2021/ (accessed on 29 September 2024).
- Wang, Zhenshan, Dandan Feng, and Chengcheng Li. 2024. Pandemic shock, debt maturity structure and corporate performance.

Accounting and Finance, 1–33. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.




Article The Role of Personal Remittances in Economic Development: A Comparative Analysis with Foreign Direct Investment in Lebanon

Samar F. Abou Ltaif ¹,*¹, Simona Mihai-Yiannaki ¹ and Alkis Thrassou ²

- ¹ Department of Accounting, Economics and Finance, School of Business Administration, European University Cyprus, Nicosia 2404, Cyprus; s.mihai@euc.ac.cy
- ² Department of Management, School of Business, University of Nicosia, 46 Makedonitissas Avenue, Nicosia 2417, Cyprus; thrassou.a@unic.ac.cy
- * Correspondence: sf191834@students.euc.ac.cy; Tel.: +357-96171801008

Abstract: Understanding the role of personal remittances in economic development is crucial, particularly for countries like Lebanon, where these inflows play a significant role in economic stability. This study investigates the impact of personal remittances on Lebanon's economic development over the period from 2002 to 2022, employing a mixed-methods approach that combines quantitative regression analyses and qualitative data from surveys. The research finds that personal remittances have a more substantial effect on Lebanon's GDP compared to foreign direct investment (FDI), with positive correlations observed between remittances and key economic indicators such as GDP, public debt, and unemployment rates. Additionally, qualitative findings reveal that remittances are vital for addressing basic living expenses, education, and healthcare needs, illustrating their multifaceted influence on household well-being. This study contributes to the existing literature by providing a nuanced understanding of how remittances impact economic development in Lebanon and highlights the need for policy interventions aimed at enhancing financial literacy and promoting productive investments. The findings offer valuable implications for policymakers and stakeholders, suggesting that improving the management and utilization of remittances could significantly bolster Lebanon's economic resilience and growth prospects.

Keywords: personal remittances; Lebanon; FDI; sustainability; development economics

1. Introduction—Research Context and Aim

Worldwide, the volume of personal remittances has been expanding. Personal remittances increased from \$420.07 billion in 2010 to \$658.06 billion in 2020 (World Bank Group 2022a). In 2023, remittance flows to low- and middle-income nations amounted to approximately \$669 billion (World Bank Group 2023). This surge was primarily driven by stable labor markets in developed economies and GCC nations, enabling migrants to continue sending money back home. The regions witnessing growth in remittance inflows included Latin America and the Caribbean (8%), South Asia (7.2%), East Asia and the Pacific (3%), and Sub-Saharan Africa (1.9%). Conversely, the Middle East and North Africa experienced a 5.3% decrease, mainly attributed to a sharp decline in Egypt. The United States maintained its position as the largest source of remittances (World Bank Group 2023).

This increase in remittances of formal transfers against a decrease in financial inflows after the COVID-19 pandemic (El Hamma 2019) in developing countries, especially labor-exporting countries, has gained the attention of researchers considering the impact of foreign currency inflows in those countries. Furthermore, the volume of money sent through informal channels is estimated to be as high as 10% to 50% of the personal remittances received through formal transfers (El Hamma 2019), making personal inflows (formal plus informal) even higher than disclosed inflows.



Citation: Abou Ltaif, Samar F., Simona Mihai-Yiannaki, and Alkis Thrassou. 2024. The Role of Personal Remittances in Economic Development: A Comparative Analysis with Foreign Direct Investment in Lebanon. *Risks* 12: 176. https://doi.org/10.3390/risks12110176

Academic Editors: Eulália Mota Santos and Margarida Freitas Oliveira

Received: 2 September 2024 Revised: 27 October 2024 Accepted: 31 October 2024 Published: 7 November 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). In 2020, Lebanon's remittances constituted 38% of its GDP, making it the second highest globally. By 2023, remittances made up 28% of Lebanon's GDP, highlighting their essential role in mitigating current account and fiscal deficits (World Bank Group 2022b). Meanwhile, even though the amount of inflow of those personal remittances remains constant and high, the economic indicators in Lebanon are deteriorating (refer to Figure 1).



Figure 1. Lebanon economic indicators (2002–2023): A timeline of turning points and influencing factors, adapted by the authors (World Bank Group 2023; Mrad 2024).

Over the last two decades, Lebanon's economic journey has been punctuated by pivotal moments. The early 2000s saw a post–Civil War recovery, but the 2006 Israel–Hezbollah conflict slowed economic growth. During 2007–2008, the country benefited from an economic boom, although, in general, the world faced a global financial crisis. Lebanon exhibited resilience in the face of the 2008 global financial crisis, leveraging its robust banking sector and prudent financial measures. Despite global economic turmoil, Lebanon's GDP increased, and key indicators such as the fixed exchange rate, money supply, and foreign interest rates remained stable (BDL 2023).

The Syrian Civil War in 2011 and protests in 2019 heightened regional tensions and destabilized Lebanon's economy (Rijnoveanu and DemiR 2013; Abdo et al. 2020). By 2020, Lebanon faced a severe economic crisis, worsened by currency devaluation, banking sector collapse, and the Beirut Port explosion. Economic and political uncertainties persist as of 2024.

1.1. Previous Research and Findings on Remittances

Research has extensively examined the role of personal remittances in economic development. Studies emphasize that remittances provide crucial foreign currency, fostering consumption, reducing poverty, and stabilizing economies, particularly in labor-exporting countries like Lebanon (El Hamma 2019). However, the negative implications have also been highlighted, with some arguing that heavy reliance on remittances discourages local investment and can lead to complacency in addressing structural economic issues (Rijnoveanu and DemiR 2013). In Lebanon, remittances have proven resilient in times of crisis, supporting the economy through events such as the 2008 global financial crisis and the 2020 economic collapse. However, despite their importance, limited research compares the impacts of personal remittances with other financial inflows like foreign direct investment (FDI), which also play a crucial role in the economy.

1.2. Identifying and Addressing the Research Gap

While remittances are widely acknowledged as vital to Lebanon's economy, there is a significant gap in understanding how they compare to other financial flows, such as foreign direct investment (FDI), in terms of their contributions to economic stability and growth. Moreover, the current literature often overlooks the strategic potential of remittances in fostering long-term development. Specifically, few studies analyze the strategic decisions that could help Lebanon harness remittances more effectively to promote sustainable economic growth, particularly in comparison to FDI. Understanding this gap is critical as Lebanon seeks solutions to its ongoing economic crisis.

Filling this gap is crucial for both theory and practice. By comparing the contributions of personal remittances and FDI, this study will provide insights into how Lebanon can leverage these financial flows for sustainable development. The findings will contribute to broader discussions on remittances, helping policymakers in Lebanon and similar labor-exporting countries better understand how to strategically maximize remittances' impact on long-term growth. Thus, addressing this gap will advance both academic research and practical policy recommendations, offering potential strategies to mitigate Lebanon's economic challenges.

1.3. Proposed Solution and Research Approach

This paper aims to explore the impact of personal remittances on Lebanon's economic development using a mixed-methods approach. It compares the significance of personal remittances to foreign direct investment (FDI) and identifies strategic decisions to leverage remittances for sustainable economic growth.

Using regression analysis, a questionnaire, and qualitative data, the research will provide empirical insights into how remittances and FDI contribute to Lebanon's economic stability.

Based on that, the research questions are as follows:

RQ1—How had personal remittances affected economic factors in Lebanon amid several shocks and events?

RQ2—What is the comparative significance of personal remittances (PR) versus foreign direct investment (FDI) in driving economic growth and development, particularly in the context of Lebanon's economy?

RQ3—What strategic decisions may be considered to support economic development, specifically by benefiting from the potential of personal remittances?

The paper starts with an introduction, highlighting the global rise in personal remittances and providing the context, aim, and contribution of the research. Subsequently, the literature review explores extant research, focusing on both the positive and the negative impacts, particularly in the context of Lebanon. The methodology section next explains how the study was conducted. Regression analyses of the empirical results then present the findings, and thereafter, insights are reached that are aligned with social capital theory. Thereafter, discussion interprets the results, leading to surmised conclusions to end industry recommendations and policy implications. Figure 2, below, schematically presents the comprehensive research framework.



Figure 2. Comprehensive research framework, developed by the authors.

2. Literature Review

2.1. Theoretical Foundations: An Integrated Theory Approach

To conduct this study, a mixed research approach was employed, integrating insights from development economics and social capital theory to comprehensively analyze the impact of personal remittances on Lebanon's economy.

- Development economics: Development economics focuses on improving economic, social, and political well-being in developing countries (Hayami and Godo 2005). Personal remittances' impact on Lebanon's developmental trajectory is analyzed, considering dimensions such as economic growth, income distribution, and poverty alleviation (UNDP 2023).
- 2. Social capital theory: Social capital theory provides insights into the societal effects of personal remittances, emphasizing the value embedded in social relationships and networks within communities (Palloni et al. 2001). This framework allows for an examination of how remittances influence wealth distribution and shape societal dynamics. Informed by social capital theory, studies explore the impact of remittance-related social networks on communities, and they emphasize the role of an efficient institutional environment in fostering positive outcomes (Kadozi 2019; UNCTAD 2013).
- 3. Sustainability: Sustainability considerations underscore the need to ensure long-term economic stability and development in Lebanon. While personal remittances contribute to short-term stability, sustainable development goals emphasize economic growth while protecting the environment and promoting social inclusivity (Aniche 2020). Diversification of revenue sources is essential for Lebanon's long-term sustainability (Chang and Lebdioui 2020).
- 4. Inequality: Inequality considerations suggest that personal remittances may alleviate poverty and reduce income inequality in recipient countries (Adenutsi 2011). However, remittances may also widen inequality gaps if not distributed equitably or if they primarily benefit certain segments of the population (Islam and Azad 2023).
- 5. Dutch disease phenomenon: The Dutch disease phenomenon is a crucial aspect to consider in highlighting the negative consequences of remittances. Scholars argue

that remittances may negatively impact productivity and growth, as funds are often directed toward consumption dominated by foreign goods rather than productive investments (Perez-Saiz et al. 2019). This phenomenon suggests that a surge in remittance inflows may lead to a real appreciation of the exchange rate, causing a shift away from tradable to non-tradable sectors (Fullenkamp et al. 2008b).

To provide a comprehensive theoretical framework, this study links key theories that explain the impact of personal remittances on economic development. Development economics emphasizes how remittances can stimulate growth by improving household consumption, education, and health, contributing to overall economic stability. Social capital theory further elaborates on the social dynamics of remittances, as they strengthen relationships between the diaspora and local communities, enhancing social cohesion and economic participation. At the same time, the Dutch disease phenomenon warns of the potential drawbacks of large remittance inflows, such as currency appreciation and reduced competitiveness in export sectors. These economic effects also intersect with concerns of inequality, where remittances may both reduce poverty and simultaneously widen income gaps if unequally distributed. Together, these theories highlight the multifaceted nature of remittances: while they offer vital economic support, especially in crises, their long-term impacts require careful management to promote sustainability and equitable growth.

The relevance of these theoretical frameworks to the context of Lebanon and personal remittances underscores the need for a nuanced understanding of the dynamics at play. Incorporating sustainability and inequality broadens the scope of the literature review, providing a comprehensive understanding of the multifaceted impacts of personal remittances on Lebanon's economy.

2.2. Positive and Negative Effects of Personal Remittances

Meyer and Shera (2015) provided a comprehensive overview of the impact of personal remittances on macroeconomic development. Their study highlighted the positive effects of remittances, such as increasing investment and productivity and improving market conditions to alleviate poverty. However, other research suggested potential negative consequences of remittances, such as moral hazard, reduced work effort, and inflation. El Hamma (2019) examined the impact of global economic changes and the COVID-19 pandemic on remittance flows, which significantly implicate macroeconomic stability. Jongwanich and Kohpaiboon (2019) examined this relationship in developing countries across Asia and the Pacific using data from 1993 to 2013. They discovered a significant negative impact when the ratio of remittances to GDP exceeded 10 percent, while the impact was insignificantly negative when the ratio remained below 10 percent. Furthermore, the Dutch disease phenomenon, as discussed by El Hamma (2019), suggests that spending might shift toward consumption rather than productive sectors, hindering growth. The study critically assessed this paradox, indicating that while remittances can boost consumption, they might also divert resources away from more productive uses.

The positive effects of personal remittances include their role in fostering economic growth through consumption, savings, and investment (Perez-Saiz et al. 2019). Studies in countries like Morocco, India, Pakistan, and Mediterranean nations have shown favorable impacts on investment acceleration (Sahoo 2006). Scholars like Zghidi et al. (2018) and the World Bank emphasized the positive contribution of remittances to developing countries' balance of payments, influencing economic growth directly through savings and investment in human and physical capital and indirectly through consumption. Research by Giuliano and Ruiz-Arranz (2009) highlighted the developmental effects of remittances on consumption multipliers, financial institution development, and foreign exchange utilization.

Conversely, Meyer and Shera (2015) cautioned against potential negative consequences of remittances, including real exchange rate appreciation and resource allocation shifts from tradable to non-tradable sectors. Rodrik's (2008) study suggested that real exchange rate overvaluation can hinder long-term economic growth, especially in developing countries with weak institutions. Sobiech (2019) found partial or negative impacts of remittances on growth in various countries, and Chowdhury et al. (2023) discovered a negative correlation between remittances and per capita income growth. Remittances may indirectly affect the real exchange rate, leading to the Dutch disease phenomenon and negatively impacting economic growth in countries heavily reliant on remittances (Acosta et al. 2009).

Overall, the literature indicates a mixed correlation between remittances and macroeconomic growth. El Hamma (2019) suggested that achieving a positive impact from remittances requires financial development and an efficient institutional environment.

3. Methodology

This study combines quantitative methods to investigate the impact of personal remittances and FDI on economic development in Lebanon. By employing a comprehensive approach, the research integrates both secondary and primary data sources to provide a robust analysis of the influence of this type of financial flow on GDP growth. Additionally, data collection via questionnaires offers contextual understanding (Maxwell 2012) and captures the nuanced perspectives of the Lebanese population regarding the impact of remittances.

Based on the literature review and theoretical approaches in this study, the hypotheses tested are as follows:

Hypothesis (H1). The relationship between foreign direct investment (FDI) inflows and GDP growth in Lebanon is insignificant or weakly positive. This expectation is supported by literature suggesting that, in economies facing political instability and financial crises, such as Lebanon's post-2019 environment, FDI may have limited influence on economic growth due to lack of investor confidence, governance challenges, and declining business opportunities (Kadozi 2019).

Hypothesis (H2). Personal remittances have a positive and significant impact on GDP growth in Lebanon, suggesting their substantial impact on the country's economic development. This is justified by extensive literature indicating that remittances, especially in developing economies like Lebanon, serve as a vital source of income, boost household consumption, and support economic resilience in times of financial instability (UNDP 2023; Fullenkamp et al. 2008a). Given the surge in remittances post-2019, they are expected to significantly contribute to GDP growth.

These hypotheses were subject to empirical testing using a regression analysis. The outcomes offered valuable insights into the dynamics of economic development driven by remittances and FDI in Lebanon.

The general form of the regression equation, applicable to each of the analyses conducted, is as follows:

$$Y = \beta 0 + \beta 1 \times X + \varepsilon$$

where

- Y represents the dependent variable (e.g., foreign direct investment, public debt, unemployment rate, GDP, or inflation rate)
- X represents the independent variable (personal remittances and FDI in this case)
- $\beta 0$ is the intercept term, representing the value of *Y* when *X* is zero
- β 1 is the coefficient, indicating the change in *Y* for a one-unit change in *X*
- *ε* represents the error term, capturing the difference between the observed and predicted values of *Y*.

3.1. Secondary Data Collection and Regression Analysis

In our research, we employed secondary data collection methods to tap into existing knowledge from academic papers, government reports, and statistical data, enhancing our understanding of the research topic. Academic papers offered scholarly perspectives, government reports provided insights into societal issues, and statistical data allowed for a quantitative analysis of various aspects.

Additionally, statistical data from reputable sources such as the International Monetary Fund (IMF), the World Bank, and Lebanon's Central Bank facilitated a quantitative analysis of various economic indicators. We specifically utilized yearly data spanning from 2002 to 2021. This time frame allowed for an in-depth examination of long-term trends related to personal remittances, foreign direct investment (FDI), and GDP growth in Lebanon.

The use of SPSS ensured rigorous data analysis, facilitating efficient management, organization, and visualization of the data. This secondary data were critical in testing the hypotheses regarding the relationships between personal remittances, FDI, and GDP growth in Lebanon. We considered the assumptions and limitations of statistical techniques, seeking expert guidance for accurate interpretation.

3.2. Primary Data Collection: Questionnaires

In tandem with our literature review and quantitative analysis, which utilized regression models to examine various variables, we conducted a questionnaire-based survey in December 2023, collecting 157 responses from Lebanese residents.

This methodological choice was driven by the need to delve deeper into the profound impact of personal remittances on a substantial segment of the Lebanese populace. The sampling method was non-probability, given the specific focus on individuals likely affected by remittances. The survey was conducted online due to logistical advantages, such as broader reach across different regions of Lebanon and abroad, as well as the ability to gather responses in a cost-effective and timely manner. Online surveys also offer flexibility, allowing participants to respond at their convenience, thus increasing response rates.

The decision to limit the sample size to 157 was purposeful, statistically reflecting Lebanon's modest population and allowing for a more detailed analysis of the open-ended responses. By integrating both quantitative and qualitative insights, the survey provided a nuanced understanding of personal remittances' role in the economy, complementing our regression analysis.

By incorporating firsthand perspectives through the survey, we aimed to capture the nuanced and pervasive influence of personal remittances, enriching our quantitative analysis with additional contextual data. The integration of quantitative insights from the survey supplements our regression analysis findings (Wang and Ji 2010) and amplifies the significance of personal remittances in shaping Lebanon's economic terrain. This process aligns with the principles of methodological triangulation (Creswell and Clark 2017). This approach ensures a holistic understanding of the subject matter, reinforcing the robustness and validity of our research outcomes.

The integration of these methods contributed to the quality and validity of our research, yielding a comprehensive understanding of the subject matter and supporting the recommendations and contribution of the research. Figure 3 summarizes the methodology framework followed.



Figure 3. Methodology framework, adapted by the authors, based on Creswell (2013).

4. Empirical Analysis Results

4.1. Lebanon-Specific Aspects

Lebanon is currently facing a critical financial situation, particularly in its banking sector, where the ongoing crisis has led to a severe lack of trust in financial institutions. One potential solution in such crises is a bank bailout, as explored by Mihai Yiannaki (2011) in her study on lessons from recent bank bailouts. Bailouts could offer opportunities for financial recovery and strategic growth, but in Lebanon's case, the lack of public trust in banks and the broader financial system might render this approach unfeasible. Instead, innovative strategies must be explored, as emphasized in the work of Vrontis et al. (2023), which highlights the importance of sustainable and innovative business practices.

International remittances have always been a supporting factor for the financial system and economy of Lebanon. The remittances considered in our case, the Lebanese economy, are transferred by the Lebanese diaspora to their home country. Although there is an existing potential loss represented by brain drain, as well as by skilled employees' migration, we will focus in this study on the foreign currency that those labor forces provide Lebanon with.

The remittances received in Lebanon are high compared to other markets. For example, in 2016, as shown in Figure 4, Lebanon received \$7.61 billion in personal remittances, while the Russian Federation received \$6.69 billion, and the United States received \$6.33 billion

(World Bank Group 2022a). That is despite the gap in GDP between these countries and the USA being the highest. In the same year, Lebanon's GDP stood at \$51.15 billion, in comparison to Russia's \$1.277 trillion and the United States' significantly larger \$18.7 trillion (World Bank Group 2024).



Figure 4. Personal remittances received in Lebanon, the Russian Federation, and the United States in 2016 and 2023. (World Bank Group 2024, adapted by the authors).

Following the financial and liquidity crisis in 2019, accompanied by a surge in unemployment and poverty rates, there has been a heightened reliance on international inflows in Lebanon. In the aftermath of the 2019 crisis, it is estimated that the proportion of Lebanese families depending on remittances has surged, reaching approximately 15% to 30% (Shehadi 2022). A recent study conducted by Mercy Corps Lebanon (2022) further highlighted that 32% of Lebanese individuals now rely on personal remittances to meet their essential expenses. According to World Bank Open Data (World Bank Group 2024), in 2023, Lebanon's remittances continued to be significant at \$6.37 billion, reflecting its ongoing reliance on remittances compared to Russia and the United States, which saw increases to \$6.80 billion and \$7.22 billion, respectively. Our survey is elaborated on in Section 5 and supports this estimation.

Recent research provides additional insights into the broader economic implications of remittances in Lebanon. Kassab (2024) highlighted the inflationary pressures caused by remittance inflows, finding that remittance growth shocks led to significant short-run increases in inflation. This is particularly relevant to Lebanon's economy, where sudden inflows contribute to domestic liquidity but may exacerbate inflationary tendencies. Dauod and Ramirez (2024) explored remittances' positive effects on industrial production in Lebanon, using oil prices as an instrument due to the large share of remittances from Gulf States. Their study found that remittance-driven growth in industrial output is often underestimated, reinforcing the importance of these inflows to the productive sectors of the economy.

Additionally, Ayhan (2021) examined the role of remittances in the banking sector, finding that they positively affected return on equity (ROE) for Lebanese banks, indicating a strong relationship between remittances, banking profitability, and financial sector health. On a regional level, Kratou and Khlass (2022) examined how remittances reduce income inequality in the MENA region by targeting low-income households, which is particularly relevant in Lebanon, where remittances play a key role in poverty alleviation. These studies underline the multifaceted impact of remittances on Lebanon's economy beyond their role as a source of foreign currency.

The average cash inflow from remittances for the decade was about \$7.15 billion. In 2022, Lebanon was ranked third in the MENA region after receiving an estimated \$6.8 billion in remittances. This comes after Egypt, with a record \$32.3 billion, and Morocco, registering \$11.4 billion (Aoun 2022). As Figure 5 represents, personal remittances stand higher than FDI, which almost disappeared after the crisis of 2019 (Choueiri et al. 2023).



Figure 5. Personal remittances and FDI to Lebanon in billions of USD (World Bank Group 2022a).

Furthermore, Figure 5 illustrates the disparity between personal remittances and FDI, alongside the trade imbalance over the years:

- Personal remittances (PR) vs. FDI: The consistent trend of personal remittances surpassing FDI underscores the significant role of remittances in Lebanon's economy (Shahzad et al. 2014). Despite fluctuations, the overall dominance of PR implies a substantial inflow of funds from diaspora compared to direct investment. This raises questions about Lebanon's attractiveness to foreign investors and the factors driving the preference for remittances over FDI.
- Trade imbalance: The persistent negative values representing trade imbalances indicate Lebanon's reliance on imports, surpassing its export capacity (Azzi 2023). The fluctuating levels of trade deficits over time reflect dynamic economic conditions and external factors influencing trade dynamics. This imbalance poses challenges to Lebanon's economic sustainability and underscores the need for strategies to address trade disparities and promote export-driven growth.
- Impact as sources of foreign currency and economic implications: Personal remittances, crucial for Lebanon's financial stability and household support, dominate as stable sources of foreign currency (UNDP 2023). In contrast, FDI represents longer-term commitments subject to economic conditions. Heavy reliance on remittances to offset deficits raises sustainability concerns (Fullenkamp et al. 2008a). While providing short-term relief, remittances' long-term impact on development requires careful consideration. Lebanon must diversify foreign currencies sources and address structural issues for sustainable growth and resilience to shocks.

4.2. Regression Analysis

The first group of regression analyses was conducted on various dependent variables, each linked to personal remittances. Table 1 summarizes the contributions, implications, and further interpretations of each analysis.

Analysis	FDI (Foreign Direct Investment)	Public Debt	Unemployment Rate	GDP (Gross Domestic Product)	Inflation Rate
Contribution	A strong positive relationship exists between personal remittances and FDI, supported by a high R-squared value (0.9289) and low <i>p</i> -value (5.63×10^{12}).	There is a significant positive correlation between personal remittances and public debt, evidenced by the high R-squared value (0.8962) and low <i>p</i> -value (1.75×10^{10}) .	The regression reveals a strong positive association between personal remittances and unemployment (R-squared = 0.9607 , <i>p</i> -value = 1.18×1011).	A substantial positive relationship between personal remittances and GDP is evident, supported by a high R-squared value (0.9517) and low p -value (1.71 × 10 ¹³).	The analysis indicates a weak positive correlation between personal remittances and the inflation rate (R-squared = 0.1291 , <i>p</i> -value = 0.1106).
Implication	Higher personal remittances correlate with increased FDI, suggesting remittance- receiving countries may attract more foreign investment.	Increased personal remittances coincide with higher public debt, possibly indicating elevated government spending or borrowing influenced by remittance inflows.	Higher personal remittances are linked to increased unemployment rates, suggesting remittances may not effectively contribute to job creation or economic development.	Countries receiving higher personal remittances experience significant GDP growth, likely due to stimulation of consumption, investment, or other factors.	While there is a modest increase in the inflation rate with higher remittances, the result lacks statistical significance, suggesting a tenuous relationship between the two variables.

Table 1. The impact of personal remittances on economic indicators.

Based on the analysis in Table 1, we can interpret the observed relationships between personal remittances and key economic indicators.

Foreign Direct Investment (FDI): Although there is a strong positive correlation between personal remittances and FDI, further research is necessary to establish causality. Policymakers should investigate whether remittances directly attract FDI or if other factors contribute. Additionally, strategies should be explored to harness remittances for sustainable FDI and economic growth.

Public Debt: While the correlation suggests a relationship between remittances and public debt, it is crucial to assess the underlying factors driving this trend. Policymakers must consider whether increased public debt reflects productive investment or unsustainable fiscal practices. Strategies to manage public finances effectively and ensure transparency in government spending are essential for mitigating risks associated with rising debt levels.

Unemployment Rate: The observed correlation between remittances and unemployment underscores the complexity of labor market dynamics. Policymakers should investigate structural factors contributing to unemployment and explore targeted interventions to enhance job creation and workforce development. Additionally, assessing the quality of employment opportunities and addressing mismatches between labor supply and demand is critical for leveraging remittances to reduce unemployment effectively.

GDP (Gross Domestic Product): While remittances contribute positively to GDP growth, policymakers should consider the sustainability and inclusivity of economic expansion. Strategies to diversify the economy, promote innovation, and enhance productivity are essential for achieving resilient and inclusive growth beyond remittance-driven gains. Moreover, addressing structural constraints and investing in human capital development can amplify the long-term impact of remittances on economic prosperity.

Inflation Rate: The weak correlation between remittances and inflation underscores the importance of considering broader macroeconomic factors influencing price stability. Policymakers should monitor inflationary pressures and implement appropriate monetary policies to ensure price stability while harnessing the benefits of remittance inflows. Additionally, exploring the impact of remittances on income distribution and purchasing power can provide deeper insights into their implications for inflation dynamics.

In summary, these analyses provide valuable insights into the multifaceted impacts of personal remittances on various economic indicators, informing policymakers and researchers about the dynamics of these relationships. The implications highlight both positive and potentially challenging aspects of the influence of remittances on a country's economic landscape.

Thus, considering Table 2, a one-unit increase in personal remittances is associated with a substantial increase of 1.3718 units in FDI. For every one-unit increase in personal remittances, public debt is expected to increase by 0.4232 units. A one-unit increase in personal remittances corresponds to a significant 23.3070-unit increase in unemployment.An increase of one unit in personal remittances is linked to a 5.8345-unit increase in GDP. While a one-unit increase in personal remittances is associated with a 2.2184-unit increase in the inflation rate, the relationship lacks strong statistical support.

Dependent Variable (PR as Independent Variable)	FDI	Public Debt	Unemployment	GDP	Inflation Rate
R-squared	0.9289	0.8962	0.9607	0.9517	0.1291
Significance (<i>p</i> -value)	$5.63 imes 10^{12}$	$1.75 imes 10^{10}$	$1.18 imes10^{11}$	$1.71 imes 10^{13}$	0.1106
Coefficient (PR)	1.3718	0.4232	23.307	5.8345	2.2184

Table 2. Summary of results of regression analysis.

Note: Personal remittances (PR) = independent variable. FDI, public debt, unemployment, GDP, and inflation rate = dependent variables.

The second concept of regression analysis was comparing the impact of Personal Remittances and FDI on GDP in Lebanon:

The comparison between the regression results for FDI and GDP, and personal remittances (PR) and GDP reveals stark differences in their impact on Lebanon's economic output:

Table 3 summarizes the regression statistics for both personal remittances to GDP and foreign direct investment to GDP, highlighting their respective coefficients, standard errors, t statistics, *p*-values, and confidence intervals.

Table 3. Comparing regression analysis.

	Personal Remittances (PR) to GDP	Foreign Direct Investment (FDI) to GDP
Multiple R	0.976	0.12
R-squared	0.952	0.014
Adjusted R-squared	0.899	-0.04
Coefficient	5.834	1.534
Standard error	0.302	2.996
t stat	19.347	0.512
<i>p</i> -value	$5.82 imes10^{14}$	0.615
Lower 95%	5.203	-4.761
Upper 95%	6.466	7.828

Note: Data sources: IMF, World Bank, Lebanese Central Bank.

The comparison between the regression results for FDI and GDP, as well as personal remittances and GDP, reveals stark differences in their impact on Lebanon's economic output. Regarding PR to GDP, the regression analysis demonstrates an exceptionally strong positive correlation, with a multiple R value of 0.976, indicating a robust relationship. The R-squared value of 0.952 suggests that approximately 95.2% of the variation in Lebanon's

GDP can be explained by variations in personal remittances, further supported by the adjusted R-squared value of 0.899, indicating statistical significance. The coefficient for personal remittances, standing at 5.834, signifies that for every unit increase in personal remittances, Lebanon's GDP increases by approximately 5.834 units. Conversely, the regression analysis for FDI and GDP reveals a weak positive correlation, with a multiple R value of 0.120. The R-squared value of 0.014 indicates that only about 1.4% of the variation in Lebanon's GDP can be explained by variations in FDI, and the adjusted R-squares value of -0.040 suggests a poor fit of the model to the data. The coefficient for FDI is 1.534, but it lacks statistical significance, with a *p*-value of 0.615. In summary, while personal remittances significantly impact Lebanon's GDP, FDI does not appear to be a significant driver of economic growth in the country. This underscores the necessity of understanding the varied sources of economic influence and implementing targeted policies to capitalize on the benefits of remittances for sustainable development in Lebanon.

In summary, while personal remittances have a substantial and statistically significant impact on Lebanon's GDP, FDI does not appear to be a significant driver of economic growth in the country. This highlights the importance of understanding the diverse sources of economic influence and the need for targeted policies to harness the benefits of remittances for sustainable development in Lebanon.

4.3. Hierarchical Multiple Regression Models

In order to strengthen the reliability of the regression analysis and the robustness of the pairwise estimates, a hierarchy of several multiple regression models was conducted. This approach involved the progressive addition of control variables to the baseline model, which initially focused on the relationship between foreign direct investment (FDI) and gross domestic product (GDP). The aim was to demonstrate the robustness of the regression estimates by exploring alternative models that control for other economic variables.

Hierarchical Regression Analysis Procedure:

Baseline Model 1: FDI to GDP

- Description: This baseline model examines the relationship between foreign direct investment (FDI) and gross domestic product (GDP). It seeks to establish whether FDI has a significant impact on economic growth in Lebanon when considered in isolation.
- Findings: The results show that FDI has a weak and statistically insignificant relationship with GDP. This indicates that FDI is not a major driver of economic growth in Lebanon, as expected by traditional economic theory. The low R-squared value (0.014) means that only 1.4% of the variation in GDP can be explained by FDI alone.

Model 2: Control Variables Added

- Description: In the second model, additional control variables—such as public debt, unemployment rate, and inflation rate—are introduced to account for other factors that may affect GDP. The aim is to isolate the specific effect of FDI on GDP while controlling for these economic variables.
- Findings: With the inclusion of control variables, the model fit improves significantly (R-squared increases to 0.819), explaining approximately 81.9% of the variation in GDP. However, even with the control variables, FDI remains a weak predictor of economic growth, reinforcing the findings from Model 1.

Model 3: Personal Remittances (PR) Added

- Description: This model adds personal remittances (PR) to the analysis as a new independent variable, given the strong correlation observed between PR and GDP in earlier pairwise analyses. The objective is to compare the relative contributions of FDI and PR to Lebanon's GDP.
- Findings: The inclusion of PR drastically improves the model, with R-squared rising to 0.899, meaning that 89.9% of the variation in GDP can now be explained by the model. The results show that PR has a much stronger and statistically significant impact on

GDP compared to FDI. This suggests that PR plays a more prominent role in driving Lebanon's economic growth.

Model 4: PR and FDI with Control Variables Removed

- Description: In this final model, control variables are removed to focus solely on the relationship between PR, FDI, and GDP. This step helps further evaluate the individual contributions of PR and FDI without the influence of other economic variables.
- Findings: The model fit decreases compared to Model 3 (R-squared drops to 0.616), but it still indicates that PR is a dominant predictor of GDP. The exclusion of control variables reinforces that FDI's impact on GDP remains minimal, while PR continues to explain a significant portion of the variation in GDP.

As summarized in Table 4, the hierarchical multiple regression results confirmed that personal remittances exhibited a much stronger and statistically significant impact on GDP than FDI, supporting earlier findings. These findings are crucial for policymakers aiming to harness remittances effectively for sustainable development, while also reconsidering the limited impact of FDI on Lebanon's economic trajectory.

Model	Variable	Coefficient (B)	Std. Error	t-Value	<i>p</i> -Value	R	R Square	Adjusted R Square
1. FDI	FDI	1.534	2.996	0.512	0.615	0.12	0.014	-0.04
2. Control Variables Added	FDI	0.819	0.77	1.063	0.262	0.905	0.819	0.77
	Public Debt	-0.547	0.092	-5.951	< 0.001 ***			
	Unemployment Rate	6.855	1.295	5.29	< 0.001 ***			
	Inflation Rate	-0.226	0.051	-4.431	< 0.001 ***			
3. PR Added	FDI	-2.633	2.086	-1.263	0.207	0.948	0.899	0.863
	Public Debt	-0.309	0.1	-3.09	0.002 **			
	Unemployment Rate	4.302	1.256	3.429	< 0.001 ***			
	Inflation Rate	-0.22	0.039	-5.641	< 0.001 ***			
	Personal Remittances (PR)	8.197	1.588	5.159	<0.001 ***			
4. PR and FDI with Control Variables Removed	FDI	-2.633	2.086	-1.263	0.207	0.785	0.616	0.571
	Personal Remittances (PR)	8.197	1.588	5.159	<0.001 ***			

Table 4. Model summary for hierarchical multiple regression.

Notes: Significance levels: **: Significant at the 5 percent level; ***: Significant at the 1 percent level.

4.4. Questionnaire Results

Our investigation into the impact of personal remittances in Lebanon through the questionnaire has unveiled critical insights that align seamlessly with social capital theory. Rooted in Lebanon's intricate social fabric and close-knit communities, our findings provide a nuanced perspective on how personal remittances contribute to or disrupt social capital within the country. The unique societal structure of Lebanon, marked by diverse religious and cultural groups, underscores the paramount importance of social relationships and networks.

The questionnaire, with a focus on the role and impact of remittances, served as a lens through which we explored the intricate interplay between financial flows and social capital. Notably, a substantial portion of respondents (37.6%) receive remittances (Figure 6),



and about 60% of those receive transfers from family members or friends working abroad (Figure 7).

Figure 6. Percentage of receivers of personal remittances in Lebanon, based on survey responses (Question concerning if respondents, or their immediate family, sends or receive PR to/from Lebanon).



Figure 7. Primary source of remittances received, based on survey responses (If you or your immeditate family receive remittance in Lebanon, please select the primary source(s) of these remittances. (You can select multiple options if applicable)).

Furthermore, the majority (about 52.4%) receive less than \$1000 per month, and the respondents were asked to select the primary purpose or purposes for which these remittances are used (Figure 8).



Figure 8. Primary purpose for remittances, based on survey responses (Purposes of remittances utilization among respondents).

The highest percentage of respondents (54.8%) reported that remittances primarily cover basic living expenses, indicating the crucial role of these funds in meeting day-to-day needs such as housing and food. Additionally, significant percentages allocate remittances to education expenses (26.1%) and healthcare (24.2%), reflecting priorities toward human capital development and well-being. The lower percentages directed toward investments or business ventures (5.1%) and savings for the future (8.9%) suggest a predominant focus on immediate needs rather than long-term financial planning or wealth accumulation.

The data related to inequality and sustainability of remittances in Lebanon suggests the following insights.

4.4.1. Inequality Impact

Perceptions regarding the impact of remittances on income inequality among respondents provide diverse insights into Lebanon's socioeconomic dynamics. Based on our survey, approximately 30.6% view remittances as a force for reducing income inequality, aligning with literature suggesting their positive contribution to poverty alleviation (Maimbo and Ratha 2005). Conversely, 16.6% believe remittances have no substantial effect on income inequality, reflecting a nuanced understanding of economic disparities. Concerns expressed by 18.5% about exacerbating income inequality highlight apprehensions about unequal access to remittance funds (Adenutsi 2011). Additionally, 34.4% expressed uncertainty, underscoring the need for further research and policy interventions. Engaging with existing literature can guide comprehensive evaluations of remittances' role in income distribution dynamics in Lebanon.

4.4.2. Sustainability

The analysis reveals diverse perceptions of remittance sustainability among respondents (Aniche 2020). While 25.5% expressed confidence in long-term sustainability, 42% harbored concerns about its continuity, reflecting apprehensions about potential disruptions. An additional 32.5% remain uncertain, highlighting the complexity of remittance dynamics. These findings align with literature discussing the resilience of remittance-based support systems amid economic challenges. However, they also underscore vulnerabilities to external shocks and the need for comprehensive research and policy initiatives to navigate the complexities of remittance flows and their impact (United Nations Department of Economic and Social Affairs 2020) on Lebanon's sustainable development.

4.4.3. Impact on Well-Being, Financial Stability, Development Economics

The analysis reveals varying perceptions of remittance impact on well-being and financial stability among respondents, reflecting the nuanced nature of these financial inflows (Arapi-Gjini 2022). While 50.3% acknowledged a positive or significant impact, 10.2% perceived no significant effect, and 2% noted negative outcomes. The sizable portion (38.2%) that claimed not to receive remittances highlights economic disparities. These findings underscore the need for tailored approaches to leverage remittances effectively and address diverse household needs, aligning with literature emphasizing context-specific strategies for sustainable development (United Nations Department of Economic and Social Affairs 2020).

In summary, the data reflect diverse perspectives on the impact of remittances on income inequality, with a range of opinions regarding whether remittances contribute to reducing or exacerbating disparities. Additionally, there are varying viewpoints on the sustainability of remittances, indicating a mix of optimism, concerns, and uncertainties among the respondents.

5. Discussion

The regression results (Table 1) imply that the analysis of the relationship between personal remittances and key economic indicators in Lebanon reveals nuanced dynamics with significant implications for economic policy and development strategies. While the positive correlation between remittances and FDI suggests potential benefits for attracting foreign investment, it necessitates further investigation into the underlying mechanisms driving this relationship. The association between remittances and increased public debt underscores the importance of fiscal prudence and sustainable debt management to mitigate risks to long-term economic stability. Unexpectedly, the positive link between remittances and unemployment challenges conventional assumptions, highlighting the need for targeted interventions to translate remittance inflows into sustainable job creation and economic opportunities. Despite the substantial contribution of remittances to GDP growth, concerns persist regarding the sustainability and inclusivity of economic expansion, necessitating efforts to address structural constraints and promote broad-based development. Lastly, while the modest correlation between remittances and inflation warrants monitoring, the lack of statistical significance underscores the complexity of inflation dynamics and the need for cautious interpretation in policymaking. In essence, the findings underscore the intricate interplay between remittances and economic variables, emphasizing the importance of evidence-based policymaking and proactive economic management to harness the potential of remittances for inclusive and sustainable development in Lebanon.

In the context of social capital theory, the data provide insights into how personal remittances influence social relationships and networks within the unique societal structure of Lebanon. The prevalence of remittances, with 37.6% of respondents receiving them, reflects the interconnectedness of families and friends working abroad. Most of these remittances come from family members or friends (60%), emphasizing the importance of close-knit relationships. This aligns with social capital theory, which highlights the value embedded in such social networks.

The breakdown of remittance purposes further illustrates how these financial flows contribute to societal dynamics. The primary use for covering basic living expenses (54.8%) indicates a direct impact on the day-to-day well-being of recipients, reinforcing social ties as a means of support. Education and healthcare expenses also feature prominently, emphasizing the role of remittances in facilitating access to essential services within the community.

However, the data also reveal variations in the use of remittances, with smaller percentages allocated to investments or business ventures, savings for the future, and other purposes. This diversity suggests that while remittances strengthen social ties, the economic choices made by recipients can vary, influencing the distribution of wealth within the community. Upon examining the data considering the Dutch disease phenomenon, notable patterns emerge:

- Consumption focus: The majority (54.8%) said that directing remittances toward basic living expenses raised concerns about excessive consumption, aligning with the Dutch disease phenomenon. This pattern may divert funds from productive sectors necessary for sustained economic growth.
- Limited investment allocation: A small percentage (5.1%) said they allocated remittance to investments or business ventures, suggesting challenges in fostering entrepreneurship and productive capital formation. This limitation may hinder the development of sectors crucial for economic resilience.
- Sustainability concerns: A significant proportion (42%) expressed concerns about remittance sustainability, underscoring worries about the economy's overreliance on external inflows. This sentiment echoes Dutch disease caution against unsustainable dependence on remittance-driven support.

Moving to the data on inequality and sustainability, the insights align with the broader economic implications discussed regarding the Dutch disease phenomenon and social capital theory:

- Inequality impact: The data reflect respondents' diverse opinions on whether remittances contribute to reducing or exacerbating income inequality. While a significant portion (30.6%) believed in the reduction of income inequality, concerns were raised by 18.5% about the potential exacerbation of such disparities. This mirrors the nuanced nature of social relationships and economic choices within the community.
- Sustainability: Opinions on the sustainability of remittances are varied. While 25.5% believed in the long-term sustainability, a substantial percentage (42%) expressed concerns about continuity. This dichotomy echoes the potential drawbacks, such as the Dutch disease phenomenon, where the sustainability of remittance-driven support might face challenges.
- Impact on well-being and financial stability: The positive impact perceived by 50.3% aligns with the idea that, for a significant portion, remittances contribute positively to economic conditions. However, the diversity of opinions, including concerns about sustainability and uncertainty, underscores the complexity of assessing the overall effects of remittances on well-being and financial stability.

In conclusion, the data provide a rich tapestry of insights into the interplay between remittances, social capital, economic choices, and potential challenges. The findings align with the theoretical frameworks discussed and confirm hypotheses 1 and 2, emphasizing the need for a nuanced understanding of the societal and economic dynamics influenced by personal remittances in Lebanon.

The study findings suggest a potential need for careful consideration and policy planning to ensure that remittance inflows contribute to a balanced and sustainable economic development rather than inadvertently leading to challenges associated with the Dutch disease.

6. Conclusions and Recommendations

6.1. Theoretical Conclusions and Implications

This study highlights the complex landscape of Lebanon's economy, which is significantly influenced by personal remittances amid financial instability and political unrest. Remittances, a crucial source of foreign currency, have historically bolstered household incomes and consumption, showcasing their strong correlation with GDP. However, concerns about their long-term sustainability and their impact on productive sectors underline the challenges of economic resilience in Lebanon.

The observed disparity between the substantial impact of personal remittances and the relatively insignificant influence of FDI on GDP underscores Lebanon's struggle to attract sustainable investments. This disparity points to the need for a diversified economic strategy that does not solely depend on remittances.

6.2. Industry and Policy Implications

To mitigate the risks associated with remittance dependency, it is essential to prioritize economic diversification, institutional reforms, and social inclusion. By leveraging remittances alongside targeted measures, Lebanon can navigate its economic challenges more effectively. Given the limited government capacity and the complexities of political instability, our strategic recommendations focus on engaging non-governmental actors like NGOs, private sector entities, and community-based organizations. These stakeholders are often more agile and responsive, fostering collaborative partnerships and promoting a sustainable approach to leveraging remittances for economic development.

Our proposed strategic plan encompasses actions like community needs assessment, partnership building, capacity program design, financial inclusion, entrepreneurship facilitation, and social enterprise promotion. These measures aim to empower local communities and enhance project sustainability without over-relying on government support.

6.3. Recommendations

Based on the results of this study, a strategic plan outlining the steps needed to implement the recommendations for leveraging personal remittances for economic development without relying heavily on direct government support is suggested (Table 5). The focus is on personal remittances, considering the significant correlation of remittances on economic variables and on FDI itself, while the impact of FDI on GDP is insignificant at such a low rate as it is now, and as per the study conducted.

Table 5. Strategic plan for leveraging personal remittances for economic development.

Steps	Recommended Actions			
1. Assess community needs	Conduct comprehensive assessments to understand socioeconomic challenges, opportunities, and aspirations. Evaluate existing levels of financial literacy, entrepreneurial skills, access to services, and community assets.			
2. Build partnerships	Forge strategic partnerships with local NGOs, microfinance institutions, academia, the private sector, and the diaspora. Collaborate to co-create and implement development programs, leveraging collective expertise and resources.			
3. Design capacity programs	Develop tailored capacity-building programs focused on financial literacy, entrepreneurship, and digital skills. Customize learning materials and methodologies to cater to the specific needs and preferences of target groups.			
4. Implement financial inclusion	Pilot innovative initiatives like mobile banking, digital payments, and community-based savings groups. Introduce user-friendly financial products and microcredit schemes designed for remittance beneficiaries.			
5. Establish investment funds	Create social investment funds, impact platforms, or community development funds to mobilize remittances. Develop transparent governance structures and investment criteria to ensure accountability and sustainability.			
6. Facilitate entrepreneurship	Launch entrepreneurship programs, business incubators, and startup accelerators to provide mentorship and funding. Foster innovation through hackathons, pitch competitions, and networking events connecting entrepreneurs.			

Steps	Recommended Actions			
7. Promote social enterprises	Encourage the formation of social enterprises, cooperatives, and community-owned businesses using remittance funds. Provide capacity-building support and technical assistance to help establish sustainable business models.			
8. Monitor and evaluate	Establish robust monitoring and evaluation systems to track progress, outcomes, and impact of development initiatives. Collect quantitative and qualitative data for adaptive management and program refinement based on stakeholder feedback.			
9. Advocate policy reforms	Lobby for policy reforms, regulatory incentives, and supportive environments to facilitate remittance flows. Engage policymakers, regulators, and development agencies in dialogues addressing barriers to financial inclusion.			
10. Scale successful models	Identify scalable solutions demonstrating impact and sustainability in leveraging remittances for development. Document case studies and success stories to inspire replication and adaptation of effective interventions globally.			

Table 5. Cont.

6.4. Limitations and Future Research Avenues

While this study provides valuable insights, it also has limitations. The small sample size, while reflective of Lebanon's population, might not capture the full spectrum of remittance impacts. Future research should explore larger samples or longitudinal studies to understand the evolving role of remittances and FDI in Lebanon's economy.

Moreover, this study primarily relied on pairwise regression results without incorporating control variables. This methodological choice, while simplifying the analysis, may compromise the reliability of the estimates. Future research should consider employing multiple regression models that include relevant control variables, enabling a more comprehensive understanding of the factors influencing the relationships examined.

Additionally, while our strategic plan for leveraging personal remittances for economic development provides a framework for future initiatives, further investigations into the specific barriers to attracting sustainable FDI are warranted. Identifying and addressing these barriers will empower researchers and policymakers to develop targeted strategies aimed at enhancing investment opportunities in Lebanon.

Furthermore, future research is encouraged to expand the survey questions to assess respondents' willingness to invest in Lebanon, particularly in initiatives aimed at supporting small and medium-sized enterprises (SMEs) or launching new businesses. Additionally, employing regression analysis on the survey data could yield valuable insights into the intricate dynamics of remittances and foreign direct investment, enriching our understanding of their impact on Lebanon's economic landscape.

In conclusion, this paper contributes to the literature by proposing an innovative strategic plan for leveraging personal remittances in economic development. Our comprehensive approach, involving diverse stakeholders and robust monitoring mechanisms, emphasizes collaboration and sustainability. By focusing on community empowerment and economic resilience, this study provides a fresh perspective on utilizing remittance flows to foster sustainable development. As highlighted by Bhatia et al. (2022), crises open doors for innovation, necessitating the formation of cross-functional teams to achieve transformative and revolutionary organizational change. This study emphasizes that amidst Lebanon's current economic challenges, such strategies can pave the way for meaningful transformation and growth.

Author Contributions: Conceptualization, S.F.A.L.; methodology, S.F.A.L.; software, S.F.A.L.; validation, S.F.A.L., S.M.-Y. and A.T.; formal analysis, S.F.A.L.; investigation, S.F.A.L.; resources, S.F.A.L.; data curation, S.F.A.L.; writing—original draft preparation, S.F.A.L.; writing—review and editing, S.F.A.L., S.M.-Y. and A.T.; visualization, S.F.A.L., S.M.-Y. and A.T.; project administration, S.F.A.L. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: This paper is based on original research conducted by the author in accordance with ethical principles and standards set by the journal. The author has no conflicts of interest to declare, and the accuracy of the data and unbiased analysis and interpretation are affirmed.

Data Availability Statement: All data analyzed during this study are included in the article.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Abdo, Nabil, Dana Abed, Bachir Ayoub, and Nizar Aouad. 2020. *The IMF and Lebanon: The Long Road Ahead–An Assessment of How Lebanon's Economy May Be Stabilized While Battling a Triple Crisis and Recovering from a Deadly Blast*. Briefing Paper. Oxford: Oxfam. [CrossRef]
- Acosta, Pablo A., Emmanuel K. K. Lartey, and Frederico S. Mandelman. 2009. Remittances and the Dutch Disease. *Journal of International Economics* 79: 102–16. [CrossRef]
- Adenutsi, Deodat E. 2011. Do Remittances Alleviate Poverty and Income Inequality in Poor Countries? Empirical Evidence from Sub-Saharan Africa. Munich Personal RePEc Archive. Available online: https://mpra.ub.uni-muenchen.de/id/eprint/37130 (accessed on 7 February 2024).
- Aniche, Ernest T. 2020. Migration and Sustainable Development: Challenges and Opportunities. In Migration Conundrums, Regional Integration and Development: Africa-Europe Relations in a Changing Global Order. Edited by Inocent Moyo, Christopher Changwe Nshimbi and Jussi P. Laine. Singapore: Springer, pp. 37–61. [CrossRef]
- Aoun, Stephanie. 2022. MENA Remittances in 2022. Blom Investbank. Available online: https://blog.blominvestbank.com/45721/ mena-remittances-in-2022/ (accessed on 13 January 2023).
- Arapi-Gjini, Arjola. 2022. *Migration, Remittances and Well-Being in Kosovo*. Research Report No. 98, Studies on the Agricultural and Food Sector in Transition Economies. Halle: Institute of Agricultural Development in Transition Economies.
- Ayhan, Fatih. 2021. The Determinants of Commercial Banks Profitability and Importance of Remittance in Lebanese Banks: Return of Equity and Remittance in Lebanon. *International Journal of Contemporary Economics and Administrative Sciences* 11: 117–33. [CrossRef]
- Azzi, Aline. 2023. What Drove the Two-Digit Trade Deficit in Lebanon in 2022? *Blom Invest Bank*. January 19. Available online: https://blog.blominvestbank.com/46001/what-drove-the-two-digit-trade-deficit-in-lebanon-in-2022/ (accessed on 18 February 2023).
- BDL–Banque du Liban. 2023. Economic and Financial Data. Available online: https://www.bdl.gov.lb/economicandfinancialdatasub. php?docId=17&code=1&filecode=100 (accessed on 13 June 2023).
- Bhatia, Asha, Komal Jain, Ravi Teja Boddapati, and Prashant Goyal. 2022. Transforming India's Logistic Industry During COVID-19 Crisis. In *Business Under Crisis Volume I: Contextual Transformations*. Edited by Demetris Vrontis, Alkis Thrassou, Yaakov Weber, S. M. Riad Shams, Evangelos Tsoukatos and Leonidas Efthymiou. Cham: Palgrave Macmillan, pp. 247–68. [CrossRef]
- Chang, Ha-Joon, and Amir Lebdioui. 2020. From Fiscal Stabilization to Economic Diversification: A Developmental Approach to Managing Resource Revenues. WIDER Working Paper No. 2020/108. Helsinki: UNU-WIDER. [CrossRef]
- Choueiri, Fadlo I., Jad Abi Haidar, and Marc Mourkarzel. 2023. Net FDI Inflows to Lebanon at \$359 Million in 2022. Credit Libanais Economic Research. Available online: https://economics.creditlibanais.com/Article/211681 (accessed on 25 January 2024).
- Chowdhury, Emon K., Bablu K. Dhar, and Md. Abu Issa Gazi. 2023. Impact of Remittance on Economic Progress: Evidence from Low-Income Asian Frontier Countries. *Journal of the Knowledge Economy* 14: 382–407. [CrossRef]
- Creswell, John W. 2013. Research Design: Qualitative, Quantitative, and Mixed Methods Approaches. Thousand Oaks: Sage Publications.
- Creswell, John W., and Vicki L. Plano Clark. 2017. Designing and Conducting Mixed Methods Research. Thousand Oaks: SAGE Publications.
- Dauod, Suria, and Carlos D. Ramirez. 2024. Do Remittances Affect Output? Evidence from Lebanon. *Applied Economics Letters* 31: 774–78. [CrossRef]
- El Hamma, Imad. 2019. Migrant Remittances and Economic Growth: The Role of Financial Development and Institutional Quality. In *Economie & Statistique*; No. 503–504; Paris: Institut National de la Institut National de la Statistique et des Études Économiques (INSEE), pp. 123–42.
- Fullenkamp, Connel, Thomas F. Cosimano, Michael T. Gapen, Ralph Chami, Peter J. Montiel, and Adolfo Barajas. 2008a. V Macroeconomic Implications of Remittances: Theory. In *Macroeconomic Consequences of Remittances*. Washington, DC: International Monetary Fund. [CrossRef]
- Fullenkamp, Connel, Thomas F. Cosimano, Michael T. Gapen, Ralph Chami, Peter J. Montiel, and Adolfo Barajas. 2008b. VII An Empirical Investigation of the Macroeconomic Effects of Remittances. In *Macroeconomic Consequences of Remittances*. Washington, DC: International Monetary Fund. [CrossRef]
- Giuliano, Paola, and Marta Ruiz-Arranz. 2009. Remittances, Financial Development, and Growth. *Journal of Development Economics* 90: 144–52. [CrossRef]

- Hayami, Yujiro, and Yoshihisa Godo. 2005. *Development Economics: From the Poverty to the Wealth of Nations*, 3rd ed. Oxford: Oxford University Press.
- Islam, Md. Saiful, and Abdul Kalam Azad. 2023. The Impact of Personal Remittance and RMG Export Income on Income Inequality in Bangladesh: Evidence from an ARDL Approach. *Review of Economics and Political Science* 9: 134–50. [CrossRef]

Jongwanich, Juthathip, and Archanun Kohpaiboon. 2019. Workers' Remittances, Capital Inflows, and Economic Growth in Developing Asia and the Pacific. *Asian Economic Journal* 33: 39–65. [CrossRef]

Kadozi, Edward. 2019. Remittance Inflows and Economic Growth in Rwanda. Research in Globalization 1: 100005. [CrossRef]

- Kassab, Sara. 2024. The Impact of Remittance Inflows on Inflation: The Case of Lebanon. *Review of Middle East Economics and Finance* 20: 113–27. [CrossRef]
- Kratou, Hajer, and Najeh Khlass. 2022. Remittances, Income Inequality, and Brain Drain: An Empirical Investigation for the MENA Region. In Key Challenges and Policy Reforms in the MENA Region: An Economic Perspective. Edited by Mohamed Sami Ben Ali. Cham: Springer International Publishing, pp. 85–101. [CrossRef]

Maimbo, Samuel M., and Dilip Ratha. 2005. *Remittances: Development Impact and Future Prospects*. Chicago: World Bank Publications. Maxwell, Joseph A. 2012. *Qualitative Research Design: An Interactive Approach*. Thousand Oaks: SAGE Publications.

- Mercy Corps Lebanon. 2022. Understanding the Limitations of Remittances as an Informal Social Safety Net in Lebanon. November 24. Available online: https://mercycorps.org.lb/understanding-the-limitations-of-remittancesas-an-informal-social-safety-net-inlebanon/ (accessed on 1 March 2023).
- Meyer, Dietmar, and Adela Shera. 2015. The Impact of Remittances on Economic Growth: An Econometric Model. *EconomiA* 18: 147–55. [CrossRef]

Mihai Yiannaki, S. 2011. Bank bailouts: Lessons to learn when patience is a virtue. EuroMed Journal of Business 6: 192–205. [CrossRef]

- Mrad, Helmi. 2024. Lebanon's Trade Deficit Down by 10.1% YOY to \$13.99B by December 2023. Blom Invest Bank. May 14. Available online: https://blog.blominvestbank.com/49578/lebanons-trade-deficit-down-by-10-1-yoy-to-13-99b-by-december-2023/ (accessed on 14 August 2024).
- Palloni, Alberto, Douglas S. Massey, Miguel Ceballos, Kristin Espinosa, and Michael Spittel. 2001. Social Capital and International Migration: A Test Using Information on Family Networks. *American Journal of Sociology* 106: 1262–98. [CrossRef]
- Perez-Saiz, Hector, Jemma Dridi, Tunc Gursoy, and Mounir Bari. 2019. The Impact of Remittances on Economic Activity: The Importance of Sectoral Linkages. IMF Working Papers. Washington, DC: International Monetary Fund, p. 175. [CrossRef]
- Rijnoveanu, Carmen, and Sertif DemiR. 2013. Suriye Krizinin Küresel ve Bölgesel Politik Dinamikler Üzerine Etkisi. Ege Universitesi Turk Dunyasi Incelemeleri Dergisi XIII: 55–77. [CrossRef]

Rodrik, Dani. 2008. The Real Exchange Rate and Economic Growth. Brookings Papers on Economic Activity 2008: 365–412. [CrossRef]

- Sahoo, Pravakar. 2006. Foreign Direct Investment in South Asia: Policy, Trends, Impact and Determinants. Discussion Paper. Mandaluyong: ADB–Asian Development Bank.
- Shahzad, Sayed J. H., Mobeen U. Rehman, Faiza Abbasi, and Muhammad Zakaria. 2014. Relationship Between Remittance, Export, Foreign Direct Investment and Growth: A Panel Cointegration and Causal Analysis in South Asia. *Munich Personal RePEc Archive*. November 29. Available online: https://mpra.ub.uni-muenchen.de/60290/ (accessed on 19 February 2024).
- Shehadi, Sebastian. 2022. Lebanon Is World's Most Remittance-Dependent Nation. *Investment Monitor*. December 14. Available online: https://www.investmentmonitor.ai/features/lebanon-remittance-dependent-country-world/ (accessed on 15 January 2023).
- Sobiech, Izabela. 2019. Remittances, Finance and Growth: Does Financial Development Foster the Impact of Remittances on Economic Growth? *World Development* 113: 44–59. [CrossRef]
- UNCTAD—United Nations Conference on Trade and Development. 2013. *Maximising the Development Impact of Remittances*. Geneva: UNCTAD.
- UNDP—Unitied Nations Development Program. 2023. The Increasing Role and Importance of Remittances in Lebanon. UNDP.org. Available online: https://www.undp.org/arab-states/publications/increasing-role-and-importance-remittances-lebanon (accessed on 7 February 2024).
- United Nations Department of Economic and Social Affairs. 2020. World Social Report 2020: Inequality in a Rapidly Changing World. New York: United Nations.
- Vrontis, Demetris, Alkis Thrassou, Naziyet Uzunboylu, and Leonidas Efthymiou. 2023. An Overview of Business for Sustainability: Strategic Avenues and Managerial Approaches. In Business for Sustainability: Strategic Avenues and Managerial Approaches, Palgrave Macmillan. Edited by Demetris Vrontis, Alkis Thrassou, Leonidas Efthymiou, Yaakov Weber, S. M. Riad and E. Tsoukatos. Cham: Springer Nature.
- Wang, Ting, and Ping Ji. 2010. Understanding Customer Needs through Quantitative Analysis of Kano's Model. International Journal of Quality & Reliability Management 27: 173–84. [CrossRef]
- World Bank Group. 2022a. Personal Remittances, Paid (Current US\$)-Lebanon. Available online: https://data.worldbank.org/ indicator/BM.TRF.PWKR.CD.DT?locations=LB (accessed on 2 November 2022).
- World Bank Group. 2022b. Personal Remittances, Received (% of GDP). World Bank.org. Available online: https://data.worldbank. org/indicator/BX.TRF.PWKR.DT.GD.ZS?most_recent_value_desc=false&locations=LB (accessed on 13 January 2023).
- World Bank Group. 2023. Remittance Flows Continue to Grow in 2023 Albeit at Slower Pace. Press Release, World Bank.org. Available online: https://www.worldbank.org/en/news/press-release/2023/12/18/remittance-flows-grow-2023-slower-pacemigration-development-brief (accessed on 14 August 2024).

World Bank Group. 2024. World Bank Open Data. World Bank.org. Available online: https://data.worldbank.org (accessed on 14 August 2024).

Zghidi, Nahed, Imen M. Sghaier, and Zouheir Abida. 2018. Remittances, Institutions, and Economic Growth in North African Countries. Journal of the Knowledge Economy 9: 804–21. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.





Effective Machine Learning Techniques for Dealing with Poor Credit Data

Dumisani Selby Nkambule¹, Bhekisipho Twala^{2,*} and Jan Harm Christiaan Pretorius¹

- 1 Postgraduate School of Engineering Management, University of Johannesburg, Auckland Park,
- Johannesburg 2001, South Africa; selbynkambule@gmail.com (D.S.N.); jhcpretorius@uj.ac.za (J.H.C.P.)
- ² Office of the DVC (Digital Transformation), Tshwane University of Technology, Staatsatillerie Rd, Pretoria West Pretoria, Pretoria 0183, South Africa
- * Correspondence: bhekisiphotwala@gmail.com or bhekisiphotwalal@gmail.com; Tel.: +27-12-382-5426

Abstract: Credit risk is a crucial component of daily financial services operations; it measures the likelihood that a borrower will default on a loan, incurring an economic loss. By analysing historical data for assessment of the creditworthiness of a borrower, lenders can reduce credit risk. Data are vital at the core of the credit decision-making processes. Decision-making depends heavily on accurate, complete data, and failure to harness high-quality data would impact credit lenders when assessing the loan applicants' risk profiles. In this paper, an empirical comparison of the robustness of seven machine learning algorithms to credit risk, namely support vector machines (SVMs), naïve base, decision trees (DT), random forest (RF), gradient boosting (GB), K-nearest neighbour (K-NN), and logistic regression (LR), is carried out using the Lending Club credit data from Kaggle. This task uses seven performance measures, including the F1 Score (recall, accuracy, and precision), ROC-AUC, and HL and MCC metrics. Then, the harnessing of generative adversarial networks (GANs) simulation to enhance the robustness of the single machine learning classifiers for predicting credit risk is proposed. The results show that when GANs imputation is incorporated, the decision tree is the best-performing classifier with an accuracy rate of 93.01%, followed by random forest (92.92%), gradient boosting (92.33%), support vector machine (90.83%), logistic regression (90.76%), and naïve Bayes (89.29%), respectively. The classifier is the worst-performing method with a k-NN (88.68%) accuracy rate. Subsequently, when GANs are optimised, the accuracy rate of the naïve Bayes classifier improves significantly to (90%) accuracy rate. Additionally, the average error rate for these classifiers is over 9%, which implies that the estimates are not far from the actual values. In summary, most individual classifiers are more robust to missing data when GANs are used as an imputation technique. The differences in performance of all seven machine learning algorithms are significant at the 95% level.

Keywords: machine learning; credit risk modelling; missing data; generative adversarial networks

1. Introduction

Additional risks emerge in the credit life cycle because of the spike in activities from the generation of credit to the awarding of loans. These difficulties have prompted the incorporation of technologies that can create reliable credit risk models and handle the underlying data quality problems. While data-driven tools significantly rely on the data to establish links between the inputs and outputs for an empirical framework, data quality is still crucial at every level of the model generation process. Biasness will result from failing to recognise and remove the noise from the data.

Many studies have investigated the effects of poor credit risk management procedures and excessive levels of credit risk in the banking system after the onset of the global financial crisis that caused economic downturns from 2007 to 2009. Banking organisations have now created sophisticated methods for measuring and controlling credit risk across various product lines. The advantages of credit risk management through the Internal Risk-Based



Citation: Nkambule, Dumisani Selby, Bhekisipho Twala, and Jan Harm Christiaan Pretorius. 2024. Effective Machine Learning Techniques for Dealing with Poor Credit Data. *Risks* 12: 172. https://doi.org/10.3390/ risks12110172

Academic Editors: Montserrat Guillén and Shengkun Xie

Received: 13 August 2024 Revised: 1 October 2024 Accepted: 16 October 2024 Published: 30 October 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). (IRB) approach, where banks use internal data to estimate risk components to quantify exposure to credit loss, have been noted by the Basel Committee (Häger and Andersen 2010).

Due to the missing data utilised to create these models, data quality becomes a fundamental difficulty in the new method. To make strategic decisions that will lead to business success, data are crucial to organisational operations, according to (Madnick and Lee 2009). Given the evolution of big data and its impact on business productivity, data scarcity or abnormalities in data management systems will have a significant adverse effect. According to (Haug et al. 2011), numerous studies have found that inadequate data continues to be an issue in most businesses despite adequate data within the organisation.

Unfortunately, the study on credit risk modelling, which addresses poor data problems, is limited, and reliable imputation strategies are used to reduce the missingness of the data using modern AI approaches. This study will eventually determine if credit risk models reduce the underlying financial loss in banks, which serves as a foundation for predicted credit loss.

By learning from data, artificial intelligence enables machines to replicate human intelligence. These programs will produce biased findings if their underlying data does not fully represent the population, eventually influencing their decision-making. This brings us to the research's main goal: address the significance of data quality, find ways to deal with missing data methodically, and propose effective machine-learning techniques for poor credit data.

The remainder of the paper is outlined as follows: Section 2 presents a literature review of the methods used in the paper. Section 3 related work for poor data phenomena, imputation strategies, mechanisms, and common statistical and machine learning approaches leveraged to tackle the problem; Section 4 presents the experimental set-up for the proposed use of the generative adversarial network strategy and the experimental results. We close with remarks and conclusions in Section 5.

2. Background Review

"Credit" is a derivative of the Greek word "Credere", which means reliable. According to the Greek interpretation, the fundamental principle of credit is the mutual trust between the creditor, which can be either an individual or an entity, and the debtors (the credit applicants). Credit risk is one of the most significant risks associated with banks' operations. Credit risk modelling aims to segregate the customers, identify borrowers likely to default, and eventually calculate the expected credit loss of the loan portfolio. It also assigns customers the proper cost based on their credit score, for instance, charging high rates to high-risk clients and cheap rates to low-risk clients.

As the granting of credit is essential to carrying out normal banking operations, it is safe to argue that this is where the majority of the risks the banks are exposed to originate (Andersen et al. 2001). To adequately manage credit risk exposures, banks need to understand credit risk, according to (Kim and Wu 2008). Furthermore, (Ljubić et al. 2015) assert that this calls for a system of credit risk management that offers accurate and adequate credit risk rating and exact segmentations of credits among priorities that should be closely overseen. A good credit risk policy should lead to the adoption of a system that encourages controlled risk, resulting in the development of a strong credit rating.

2.1. Machine Learning Classification Techniques

2.1.1. Support Vector Machines

Machine learning describes a system that can automatically learn information from experience and other sources. Various machine learning algorithms have been used to solve diverse classification issues. The approach taken to the learning process in each method differs significantly from one another (Danso et al. 2014). As opposed to modelling continuous valued functions, classification predicts categorical labels. In contrast to classification, which focuses on extending known structures to new data, clustering is the problem of identifying groups and structures in the data that are in some other way without using known structures (David and Balakrishnan 2010).

One of the supervised machine learning approaches found in the literature is called support vector machine (SVM), a classification tool. SVM is based on the structural risk minimisation principle and was first introduced by Vapnik in 1999 to address classification issues. SVM's operating principles can be characterised as a hybrid of linear and non-linear. For regression, classification, and other learning tasks the kernel mapping technique can be used with separate and combined data, and it performs well across various learning tasks. Data storage kernels, (Cömert and Kocamaz 2017), are used to determine how similar or dissimilar data objects are.

In the analytical study of the service quality of Indian railways under soft computing, (Majumder et al. 2024), used support vector machines, extra trees, and multinomial naive Bayes as the three machine learning classification methods they utilised. They conducted a comparative analysis based on seven performance metrics and predicted the overall train rating. According to their findings, the support vector machine is the best estimator out of the three classifiers due to its improved capacity for making predictions on the train and railway data.

2.1.2. Logistic Regression

A form of regression model called a logistic regression is used to forecast the outcome of a categorical dependent variable using one or more predictor variables. Several axes can be related to a dichotomous dependent variable, such as D, using the mathematical modelling technique known as logistic regression. When the illness measure is dichotomous, logistic regression is the most frequent modelling technique used to analyse epidemiologic data. However, other modelling approaches are also viable. In other words, logistic regression computes probability scores for the dependent variable. It uses them to calculate the association between a continuous independent variable (or numerous continuous independent variables) and a categorical dependent variable (Chaudhary et al. 2013).

Like ordinary least squares (OLS) regression, logistic regression is a prediction method. The prediction outcome in logistic regression, however, is dichotomous. Logistic regression is one of the most used tools for applied statistics and discrete data analysis. Since linear interpolation is used in logistic regression (Osisanwo et al. 2017). The approach of choice for credit risk modelling over time has been logistic regression. For instance, (Kutty 1990) provided a logistic regression model for estimating the likelihood that debt owed by developing nations will default. The analysis included the debts of 79 nations over 19 years. The model anticipated the country's debt default two years in advance for Mexico, Brazil, and Argentina.

To forecast the likelihood of default based on financial data, (Hol et al. 2002) used a logistic regression model. Current estimates of the PD of US banks were made (Gurný and Gurný 2013) utilising various statistical techniques, including linear discriminate analysis, probit model, and logistic regression (LDA). For a model estimation in his study, the author examined a sample of 298 American commercial banks gathered from 2007 to 2010 during the financial crisis. Logit and probit models were also subjected to stepwise selection. Despite the probit model having one more indicator, the logit model and the probit model exhibited remarkably similar explanatory powers based on the fit in the training data (96.30% for the logit model and 95.85% for the probit model in terms of pseudo-R-square) (Zhang 2015).

2.1.3. Naïve Bayes

The Naive Bayes variation, known as Gaussian naïve Bayes, assumes that each class is normally distributed and supports continuous values. Based on Bayes' Theorem, Gaussian naïve Bayes strongly assumes that predictors should be independent of one another. For instance, whether we should grant a loan depends on the applicant's income, age, history of prior loans, geography, and transaction history. It is implausible that data points in a real-world setting won't interact with one another, but unexpectedly, Gaussian naïve Bayes works well in that circumstance.

Despite oversimplified presumptions, it frequently performs better in various challenging real-world scenarios. The naïve Bayes theorem has the significant benefit of requiring less training data to estimate the parameters (Alam and Pachauri 2017). In agriculture, medicine, and biometrics, NB has been applied and proven successful. Nevertheless, nothing is performed with credit scoring to improve banks' and other financial institutions' assessments of a customer's creditworthiness. Finding the likelihood of a label given some observable features is our goal in Bayesian classification. Gaussian, multinomial, and Bernoulli are the three categories of Naive Bayes models.

2.1.4. Decision Trees

Decision trees (DTs) are supervised algorithms that repeatedly divide the data into subsets according to its properties until a stopping requirement is satisfied; a tree-like structure results from this iterative partitioning. DTs are white boxes because it is simple to follow the path from the root node to each leaf node in the tree and deduce the classification rules they learned. Even with the massive quantities of data, DTs are quite effective. This is owing to the algorithm's partitioning nature, which works on ever-smaller portions of the dataset. It typically only works with straightforward attribute–value data that is simple to alter.

$$\inf(S) = -\sum_{i=1}^{n} \frac{|t_i|}{|t|} \times \log_2\left(\frac{|t_i|}{|t|}\right) \tag{1}$$

Multistage decision-making can be approached in several ways, including the decision tree classifier (DTC). The capacity of DTCs to break down a complex decision-making process into a set of more straightforward options is its most crucial attribute since it results in a solution that is frequently simpler to understand (David and Balakrishnan 2010).

$$(x, Y) = (x1, x2, x3, \dots, xk, Y)$$
 (2)

Many conventional applications in numerous domains have successfully used decision trees. Although it can be said that DT is an older strategy, it has proven effective. For instance, DT has recently been used as a machine-learning technique to create automatic classification models for data related to pancreatic cancer. By classifying cases and arranging them based on feature values, DT-based algorithms "learn" from training examples. Each node in a DT stands for a characteristic of an instance that needs to be classified, and each branch stands for a value that the node might consider when planning. The image in the following figure shows how DT operates in the feature space (Danso et al. 2014).

Due to the replication problem, decision trees can represent some concepts substantially more sophisticatedly. One approach is to create intricate characteristics at nodes using an algorithm to prevent replication. The FICUS building technique was first introduced by (Markovitch and Rosenstein 2002). It employs the typical input of supervised learning and a feature representation specification to create a set of generated features. While FI-CUS shares certain similarities with other feature-creation algorithms, its key advantages are its flexibility and generality. FICUS was made to generate features from any feature representation specification that complies with its general-purpose language.

2.1.5. Random Forest

A random forest is a group of unpruned classification or regression trees inferred from bootstrap samples of the training data using random feature selection. Classifier trees don't require exploratory variables to be multicollinear or to have a functional form. Since they use non-parametric techniques, no distributional presumptions are necessary. Decision trees can be constructed using a variety of algorithms. Classification trees, first famous by Breiman et al. (1984). These are the most common solutions for binary problems.

$$\hat{f}_{\text{bag}}(x) = \frac{1}{B} \sum_{b=1}^{B} \hat{f}^{b}(x)$$
 (3)

Splitting rules, which we may use to divide the variable space into more manageable chunks, are the foundation of classification trees. Classification trees have the advantage of being easily understood and easy to understand. Overfitting is a problem that arises when developing an entire tree. Due to its intricacy, the tree's predictive capability may not be good. The pruning strategy, which is outlined, for instance, is the typical approach to address this problem; (Ortl 2016).

The ensemble's predictions are combined (by majority vote for classification or by average for regression) to produce the prediction. Random forest typically shows a significant performance gain compared to single tree classifiers like CART and C4.5 (Yadav and Tiwari 2015). In ensemble learning, RF is a classifier that improves generalisation and classification accuracy for big databases by growing over many classification trees. To use numerous prediction models, RF combines a set of base classifiers that operate independently of one another.

2.1.6. Gradient Boosting

The ensemble model, known as gradient boosting (GB) or stochastic gradient boosting, consists of several elementary decision trees. By merging the predictions of various base models and iteratively reducing the error term, ensemble models seek to increase accuracy. After the initial base model (tree) is set up, each succeeding base model is fitted to the residuals of the previous model to minimise the error term and avoid errors in the current ensemble. Since it is a homogeneous ensemble classifier, more randomness is introduced through bootstrap sampling (Andrić et al. 2019).

Gradient boosting results from Leo Breiman's discovery that boosting may be seen as an optimisation technique for an appropriate cost function. Later, Jerome H. Friedman created explicit regression gradient boosting techniques in parallel with Llew Mason's more comprehensive functional gradient boosting viewpoint (Mason et al. 2000).

The two later studies provide an abstract interpretation of boosting techniques as iterative functional gradient descent algorithms. This refers to algorithms that select a function (weak hypothesis) that points toward the negative gradient iteratively to optimise a cost function across function space. In many fields of machine learning and statistics outside of regression and classification, boosting techniques have been developed due to this functional gradient interpretation of boosting.

2.2. Poor Data Overview

In reality, the data's qualities frequently vary across its many dimensions. The default coverage, for instance, may be less extensive in the early period of the data than in the latter time. Also, smaller firms' financial statements may not be as accurate or consistent compared to larger organisations. For example, sample selection biases, data collection difficulties, and default definition variations are just a few of the data factors that (Dwyer 2007) offer an overview of when validating private enterprises' default risk models.

Most analysts know that these data issues exist in their validation and development samples. Still, they cannot quantify their exact scope, according to (Tang et al. 2011). A standard statistical issue in many application areas is missing values in datasets. Internal records may be lacking in credit analysis for various reasons, including improper registration, clients who refuse to respond to enquiries, and database or recording system errors. One possible approach is to drop the missing values from the original dataset, as performed by (Adams et al. 2002) and (Berger et al. 2005), or perform a preprocessing to replace the missing values, as performed by Banasik et al. (2003) and Paleologo et al. (2010). These procedures are missing data imputations (Louzada et al. 2016).

2.2.1. Listwise Deletion Approach

Listwise deletion, also known as complete-case analysis, is a generally used approach in data analysis. It involves removing any case with one or more missing values from the dataset. The simplest option for dealing with missing data is to use deletion techniques, in which the dataset's rows or columns having missing values are just eliminated. Even though deletion techniques are simple to comprehend and apply, they could yield biased results, particularly when handling missing data that adheres to a unique missingness mechanism. Randomly deleting data might result in the loss of essential information and deliver biases into the analysis that comes afterwards (Zhou et al. 2024).

If a single value is absent from a record, the entire record is removed from the analysis, known as listwise deletion. However, if there are a lot of incomplete records, this strategy could be wasteful and drastically impair statistical power. This happens when the likelihood of missing data is the same across all cases and the process is unrelated to the data (MCAR). As such, there won't be any bias additions to the dataset. Instead, there will only be a loss of data and diminished statistical power (Woods et al. 2021).

A complete-case analysis will provide unbiased estimates in this situation, but standard errors and confidence intervals will reflect the smaller data group containing complete records. Removing any subject with an incomplete set of measurements is comparable to having a lower sampling fraction and a smaller sample size and does not introduce bias. Thus, listwise deletion decreases efficiency even when appropriate, sometimes leading to a dramatically smaller sample size.

2.2.2. Imputation Techniques

Various imputation strategies might be used to keep records of incomplete data in the analysis. Maximum likelihood and multiple imputation techniques can generate accurate estimates for MCAR and MAR without sacrificing statistical power. Observations that are missing are replaced by values that are in some way projected, frequently from a model, using imputation techniques. In a single imputation, the missing observation may be replaced with the sample mean or median, with a predicted value of the variable (for example, from a regression model, bootstrap, or a random dataset from multiple imputation), or with the value from a study patient who matches the patient with the missing data on a set of chosen covariates.

2.2.3. Single Imputation (Median/Mean) and Mode Imputation

Machine learning algorithms can learn from incomplete data with the support of missing value imputation. The three basic missing value imputation strategies are mean, median, and mode. The mean of a set is its average value, the median is its middle value when the numbers are arranged by size, and the mode is the value that appears most frequently in two or more sets. They are only conjectures; it is reasonable to assume that the missing values are most likely very close to the value of the mean or the median of the distribution since these represent the most frequent/average observation when data are missing completely at random. The missing observations most likely look like the majority of the observations in the variable.

2.2.4. Multiple Imputation

Various imputation strategies that accommodate records with missing value data have been used to deal with the incomplete data problem. Maximum likelihood and multiple imputation techniques can generate accurate estimates for MCAR and MAR without sacrificing statistical power. Observations that are missing are replaced by values that are in some way projected, frequently from a model, using imputation techniques. In a single imputation, the missing observation may be replaced with the sample mean or median, with a predicted value of the variable (for example, from a regression model, bootstrap, or a random dataset from multiple imputation), or with the value from a study patient who matches the patient with the missing data on a set of chosen covariates.

2.2.5. K-Nearest Neighbour

A supervised technique called *k*-nearest neighbours (KNNs) categorises new cases based on a similarity metric after storing all the existing data (e.g., distance functions). KNN has been employed in statistical estimation and pattern identification based on their nearest neighbours and an odd number. Its resemblance is quantified by a distance function, like the Euclidean one (Payrovnaziri et al. 2020). K is typically treated as a bizarre number exclusively when making decisions. If K = 1, it will be automatically placed in the class of its closest neighbour. The class with the most votes from its closest neighbour is assigned if K is an odd number; (Reddy and Babu 2018).

dist
$$(X_1, X_2) = \sqrt{\sum_{i=1}^{n} (x_{1i} - x_{2i})^2},$$
 (4)

where $X_1 = (x_{11}, x_{12}, \dots, x_{1n})$ and $X_2 = (x_{21}, x_{22}, \dots, x_{2n})$.

The two types of nearest neighbour algorithms are structure-based KNN and structureless KNN. The structure-based approach deals with the data's fundamental structure, with fewer mechanisms connected to training data samples. The data are divided into training and sample points using a structure-less technique. The distance between each training and sample point is determined, and the training point with the least distance is referred to as the nearest neighbour. The KNN technique's effectiveness for big training datasets and robustness against noisy training sets are among its primary features.

Rather than using all available instances in the data, the KNN imputation algorithm uses only similar cases with incomplete patterns. Given an incomplete pattern x, this method selects the K closest cases that are not missing values in the attributes to be imputed (i.e., features with missing values in x), such that they minimise some distance measures.

KNN's benefits include its simplicity, transparency, robustness against noisy training data, ease of understanding, and ease of implementation, while its drawbacks include its computational complexity, memory constraints, poor runtime performance for large training sets, and the potential for issues caused by irrelevant attributes (Soofi and Awan 2017).

The goal is to commit the training set to memory and then predict the label of any new instance based on the labels of its close training set neighbours. This method's justification is predicated on the idea that the characteristics utilised to characterise the domain points are pertinent to their labelling in a way that makes nearby points likely to share the same label (Shalev-Shwartz and Ben-David 2014).

2.2.6. Generative Adversarial Networks (GANs)

A technique called generative adversarial networks (GANs) imputation uses GANs to fill in the missing values in a dataset. A generator network plus a discriminator network makes up GANs, a subtype of deep neural network. The discriminator network attempts to differentiate between the generated samples and the genuine data, while the generator network creates samples similar to the input data. The GAN network is trained to develop data samples identical to the actual data using the available data, including the features and the target variable.

The generator network is taught to produce samples that the discriminator network cannot differentiate from actual data. The GAN network can fill in the missing values once trained by providing the available data along with the missing values and utilising the generator to create the imputed data.

The goal of generative modelling, an unsupervised learning task in machine learning, is to automatically identify and learn the patterns in input data so that the model can produce new examples that could have been reasonably drawn from the original dataset. Figure 1 below shows the GAN architecture for image complexion, where G denotes a generative network trained to produce fake images and the discriminator that learns to categorise them. The discriminator learns to distinguish the counterfeit images made by the generator from the actual images from the dataset.



Figure 1. Generative adversarial networks architecture.

It takes two networks to train a GAN: a discriminator network and a generator network. Real data from a dataset is used in the process, along with fake data continuously produced by the generator throughout training. The discriminator receives instructions similarly to any other deep neural network-defined classifier. Data from the training set is presented to the discriminator. The discriminator has been trained to classify the data into the "real" class, and the training process also uses fake data. First, a random vector z is sampled from a previous distribution over the model's latent variables to create the bogus data.

Consequently, the generator generates a sample x = G(z). The function G serves as a neural network model of the random, unstructured z vector meant to be statistically undetectable from the training data. The discriminator subsequently categorises this fake material. Training the discriminator assigns this data to the "fake" class. The generator can be trained using the discriminator's output derivatives concerning its input due to the backpropagation process.

The generator is trained to fool the discriminator into identifying its input with the "real" class. Except for using data from a parameter distribution for the "fake" class, which is updated dynamically as the generator learns, the discriminator's training process resembles that of any other binary classifier. The generator is rewarded by generating outcomes that fool its opponent rather than being given explicit output targets, and its learning process is somewhat different; (Goodfellow et al. 2020).

Motivated by the generative adversarial networks in the previous studies, this paper aims to simulate GAN architecture in the incomplete case context to stimulate the classification accuracy of machine learning models for credit risk modelling. The credit data are mainly classified under a missing at-random mechanism, and the analysis will simulate the GANs architecture to build a robust strategy to improve the data quality.

3. Related Work

First, the nature of the missingness in this paper must be explained to align the methods and assumptions that will handle the dataset used in the experiment. There are three missing data categories: MAR, MCAR, and NMCAR. The data variable is missing completely at random (MCAR) if the probability that the feature is missing is independent of the feature's value and any other features' values. This is often the best-case missingness scenario, with no relation to observed or unobserved values and data equally likely to be missing.

In the medical field, (Jerez et al. 2010) applied machine learning techniques, such as the multi-layer perceptron (MLP), self-organising maps (SOM), and k-nearest neighbour (KNN), to data collected through the "El lamo-I" project. The results were compared to those obtained from the listwise deletion (LD) imputation method. The prediction of patient outcomes was improved by imputation techniques based on machine learning algorithms over imputation statistical techniques.

Templ et al. (2011) mentioned that the imputation method can be used for multiple imputation, producing more than one option for a missing cell if it can handle the inherent

randomness in the data. The use of multiple imputations to reflect sampling variability should only be performed after carefully examining the distributional assumptions and underlying models.

In datasets gathered through the delivery of psychological and educational assessments, missing data are a prevalent issue. It is well known that missing data can cause significant problems like skewed parameter estimates and inflated standard errors. An empirical comparison of the effectiveness of the missing data imputation techniques IRT model-based imputation (MBI), expectation–maximization (EM), multiple imputation (MI), and regression imputation (RI) was conducted in this study. Results demonstrated that MBI performed better than other imputation approaches, particularly with larger sample sizes, in retrieving item difficulty and the mean of the ability characteristics. However, when recovering item discrimination parameters, MI delivered the most significant results (Kalkan et al. 2018).

In practical machine learning tasks, incompleteness is one of the primary data quality concerns. Numerous studies have been carried out to address this problem. Although there is little research on symbolic regression with missing data, most concentrate on the classification task. In this study, a brand-new imputation technique for symbolic regression using unfinished data is presented. The method seeks to enhance the effectiveness and efficiency of symbolic regression using imputed missing data.

Genetic programming (GP) and weighted k-nearest neighbours (KNN.) are the foundations of this approach. To forecast the missing values of partial features, it builds GP-based models utilising other already available features. Such models are built using weighted KNN to choose the instances. The experimental results on actual datasets demonstrate that the suggested method exceeds a number of state-of-the-art methods in terms of imputation accuracy, symbolic regression performance, and imputation time (Al-Helali et al. 2021).

Banks and other lending institutions can build credit risk control models for lending businesses by harnessing machine learning algorithms. The result helps to mitigate the negative aspects of conventional evaluation techniques, such as low efficiency and an overreliance on subjective matter assessment. Nonetheless, data with missing credit features will always be encountered during the practical evaluation of the process. For those machine learning algorithms to be trained appropriately, the missing attributes must be filled in, especially when applying the algorithms to small banks that have little credit data. In this study, we introduced an autoencoder-based approach that can recover the missing data items in the features by leveraging the correlation between the data (Yan 2023).

Thin-file borrowers are consumers whose creditworthiness assessment is uncertain because they do not have a credit history. Many researchers have utilised borrowers' social interactions as an alternative data source to address missing credit information. Traditionally, manual feature engineering has been used to exploit social networking data; however, in recent times, graph neural networks have emerged as a promising alternative. Muñoz-Cancino et al. (2023) introduced an information-processing framework to improve credit scoring models using several methods of graph-based learning: feature engineering graph embeddings.

In several areas of medical research, missing data frequently occurs, especially in questionnaires. In addition to the widely used complete case analysis, this article aims to describe and compare six conceptually distinct multiple imputation methods. It also examines whether the methodology for handling missing data may affect the clinical conclusions drawn from a regression model when the data are categorical.

The missing handling methods selected significantly influence the clinical interpretation of the supplementary statistical analysis. The decision to impute missing data and the imputation method can affect the clinical conclusions made from a regression model and should, therefore, be given adequate consideration (Stavseth et al. 2019).

According to (Wu et al. 2021), they have to categorise a sizable amount of data in the study of mining software repositories to build a predictive model. How accurate the labels are will significantly impact a model's performance. However, the effect of incorrectly labelled occurrences on a predictive model has only been the subject of a few small studies. The case study was on the prediction of the security bug report (SBR) to close the gap. They discovered that five publicly accessible datasets for SBR prediction contain many occurrences that have been incorrectly classified, which has negatively impacted the accuracy of SBR prediction models used in recent studies.

Although the concept of a decision tree is not recent, decision tree algorithms have been gaining popularity with the growth of machine learning. This technique uses mathematical formulas like the Gini index to find an attribute of the data and a threshold value of that attribute to make splits of the input space (Patil et al. 2016; Amaro 2020). (Namvar et al. 2018) presented an empirical comparison of various combinations of classifiers and resampling methods within a novel risk assessment methodology that integrates unbalanced data to solve these problems. The credit projections from each combination are assessed using a G-mean measure to prevent bias toward the majority class, which has not been considered in previous studies. As a result of their findings, combining random forest and undersampling may be a valuable method for determining the credit risk of loan applications in social lending markets.

(Sharma et al. 2022), in the study titled "A Study on Decision-Making of the Indian Railways Reservation System during COVID-19" and under machine learning application, used a random forest classifier and an extra trees classifier. They also examined the classifiers' predictive power to the cross-validation score and six performance metrics, which include accuracy, precision, recall, F1-score, Hamming loss, and Matthew's correlation coefficient. Their findings indicated no differences in the confusion matrices or values of any performance measures between the two classifiers. However, ETC performs better than RFC in terms of cross-validation score when measured using 10-fold stratified cross-validation.

By adapting the popular generative adversarial networks (GANs) paradigm, we provide a novel approach to imputing missing data. Because of this, they refer to our technique as generative adversarial imputation networks (GAIN). After observing a portion of an actual data vector, the generator (G) imputes the missing data based on the observed data and returns a completed vector. Next, given a finished vector, the discriminator (D) tries to identify which components were imputed and which were genuinely observed. We give D some extra data in the form of a hint vector to make sure D makes G learn the intended distribution. D uses the partial information about the original sample's missingness that the hint provides to concentrate on the imputation quality of specific components. With this edge, G is guaranteed to learn how to generate based on the actual data distribution. We conducted tests of our method on multiple datasets and observed that GAIN performed much better than the state-of-the-art imputation methods (Yoon et al. 2018).

The possibility of missing credit risk data might significantly diminish the assessment model's effectiveness. Therefore, building a data imputation technique is highly beneficial for accurate missing data prediction. Due to the complicated random missing patterns and high missing rate of credit risk assessment datasets, creating an efficient imputation model is typically exceedingly tricky. Multiple generative adversarial imputation networks (MGAINs), a novel imputation technique, is proposed in this research, and this was the first time missing credit risk assessment data were imputed using GANs (Zhao et al. 2022).

The studies above provide various methods for coping with incomplete data in credit risk and other industries. The studies presented one solution in a particular area instead of many solutions to different issues in the credit risk environment. Although robust methods were identified and acknowledged from the studies above, a gap exists in integrating effective strategies to handle poor data extensively. Given the current fourth industrial revolution age and the complexity of its systems and technology component, the paper aims to illustrate the integration of various techniques holistically to underline the necessity to remediate faulty data from end to end but not from its totality.

4. Experiments

4.1. Experimental Set-Up

This section will propose using generative adversarial network unsupervised learning to deal with poor credit data. The aim is to test the robustness of machine learning classification algorithms, namely SVM, naïve base, decision trees, random forest, gradient boosting, K-NN, and logistic regression, when GANs are harnessed as an imputation strategy for the incomplete case scenario. In addition, the empirical comparison of the classification algorithms when the complete case approach, single imputation (mean/median), mode, multiple imputation, and K-NN machine learning are employed is also explored to determine the effectiveness of the machine learning strategies when handling poor data. Furthermore, the complete case approach involves the instance where records with missing values are eliminated from the dataset. While the complete case approach is more effective for the MCAR missingness mechanism, depending on how outcomes and missingness are related, the validity of this strategy may hold in scenarios where data are missing at random.

4.1.1. Data Source

The loan application data are sourced from the Lending Club website in Kaggle. Lending Club is a US peer-to-peer lending company headquartered in San Francisco, California. It was the first peer-to-peer lender to register its securities with the Securities and Exchange Commission (SEC) and offer loan trading on a secondary market. The Lending Club is the world's largest peer-to-peer lending platform.

In terms of missing values, there are a total of 74 identified variables with missing factor ratios, and it is composed of 57 features of numeric data types and 17 categorical data types. The missingness rate percentage ranges from (0.01%) to (100%). Out of the 57 numeric variables, only 16 features have a missingness rate percentage below (50%), 18 features have a missingness rate between +50% and 90% and the remaining 23 features have above a 90% missingness rate.

4.1.2. Data Processing

We applied Z-Score with the threshold method and isolation forest approaches as outlier detection methods to identify potential outliers for the numeric features. The results of the two outlier detectors were compared to see which method was more stable when categorising the outlier instances. The identified outlier's values were treated using median imputation, Winsorize to with (5% and 10%), and different transformations (logarithms, square root, Yeo, and quantiles) were tested. Additionally, this transformation was experimented with to deal with highly skewed data. This attempt was intended to complement the Z-score approach to enhance the technique's performance. Finally, the outliers that impacted the model performance results were discarded.

The missing rate percentage assessed was observed since we are dealing with an incomplete case scenario. The features with over 50% missingness rates were all dropped. Features with missing values were treated using mean/median, mode, multiple imputation, K-NN, and GANs imputation techniques. One-hot encoding was applied to convert categorical variables to numerical data format, and the data were normalised using the min–max scaler.

Correlation metrics for feature relationships were conducted to understand the variables' relationships better. A SelectKBest approach based on a scoring function was used to select the best features and leveraged to train the classifiers. The model performance results were compared with the case where variable selection was not used. The outcome shows almost identical results.

4.1.3. Missing Data

The credit data are mainly classified under a missing at-random mechanism, and the analysis will simulate the GANs architecture to build a robust strategy to solve the data

quality issue. Nicoletti and Peracchi (2006) summarise missing at random (MAR) for the default loan data and state the missing values depend on the observed data, can be fully described by other variables in the dataset, and are not dependent on the missing data.

This assumption underlies most imputation procedures. For example, even though respondents at the lower and upper end of the income distribution are less likely to provide survey responses than those in the middle, these missing data points are related to demographics and other socioeconomic variables, which can be observed in the data (Pedersen et al. 2017).

4.1.4. Data Partition

According to (Siddiqi 2006), there are various ways to split the development (sample on which the scorecard is developed) and validation ("hold") datasets. Typically, 70% to 80% of the sample is used to create each scorecard; the remaining 20% to 30% is set aside and then used to test or validate the scorecard independently. Where sample sizes are small, the scorecard can be developed using 100% of the sample and validated using several randomly selected samples of 50% to 80% each. In this paper, we propose to use a 70% split of the data from the population sample since the primary purpose is to assess the robustness of the classification algorithm to manage poor data.

4.1.5. Model Evaluation

According to (Castermans et al. 2010), the primary aim of back-testing PD discrimination is to verify whether the model still correctly distinguishes or separates between defaulters and non-defaulters or provides a correct ordinal ranking of default risk such that defaults are assigned low ratings and non-defaulters high ratings. Consequently, discriminatory power aims to discriminate the difference between good default events and good non-default events. Therefore, the model can make wrong forecasts for the event.

With that setting in mind, (Wójcicka 2012) stated that in times of economic crises, banks (financial institutions) first and foremost need to minimise their losses by limiting the probability of default of the companies they finance. Of course, on the other hand, they would also want to optimise the profit that comes from funding "good" companies, and lowering the number of "good" companies by classifying too many of them into the group of "bad" firms will result in decreasing their income.

There are two scenarios in which default models can produce misleading results. First, if the model shows a low risk when the risk is high, it points to a type 1 error. This error issues customers with no chance of defaulting with high credit quality. Type 2 error is when the model indicates a high risk when, in fact, the risk is low. The illustration explains the Table 1 below scenarios further on the contingency table.

		ESTIN	1ATED
		NO DEFAULT	DEFAULT
OBSERVED	NON-DEFAULT	True	(1-Sensitivity) = TYPE II Error
OBSERVED	DEFAULT	(1-Specificity) = TYPE I Error	Hit

Table 1. The classification report (Type I and Type II errors).

4.1.6. Model Performance Analysis

The KS statistic and Gini coefficient are the two most frequently used metrics in an industry context, and the Basel Committee recommends that the Gini coefficient or accuracy ratio (AR) be used in banks to measure models of discriminants. The KS statistic measures the maximum vertical separation (deviations) between two cumulative distributions (good and bad) in scorecard modelling.

To know how to order the attributes from best to worst risk, one must know whether the variable is positively or negatively correlated with the outcome variable. If there is a positive correlation, the higher the variable values, the higher their levels of risk, and vice versa (Bjornsdottir et al. 2009).

(Guo et al. 2017) state that receiver operating characteristic (ROC) curves display the discrimination potential of fitted logistic models by evaluating the tradeoffs between actual positive rate (sensitivity) and false positive rate (1—specificity). If you have a cutoff value, you can classify the subjects as positive or negative according to their predicted probabilities. Hence, a 2×2 classification table can be constructed to show the relationship between the expected and actual outcomes. Sensitivity is the fraction of positive subjects predicted as positive, and specificity is the fraction of negative subjects correctly predicted as unfavourable.

A hamming loss (HL) is the ratio of incorrect labels to total labels. In multiclass classification, the hamming distance between y_true and y_pred is used to compute hamming loss. Consequently, HL considers the prediction and missing errors normalised across the whole number of samples and classes.

The Matthews coefficient is a machine learning performance statistic for binary classifiers, commonly known as the Matthews correlation coefficient (MCC). It evaluates the relationship between the actual and expected binary outcomes, considering each of the confusion matrix's four components.

The randomised search from the sklearn Python library was leveraged to select the best set of hyperparameters so that the classifiers could achieve improved model classification accuracy. The following are the hyperparameters per classifier. The random state for all the classifiers was set to 42. The following are the hyperparameters selected using the randomised search criteria for the seven machine learning classifiers.

- Support vector machines: C = 1.3, gamma = 0.7, kernel = 'linear';
- Random forest: n_estimators = 50, min_samples_split = 5, min_samples_leaf = 2, max_features = 'sqrt', max_depth = None, bootstrap = False;
- Logistic regression: C = 0.2, penalty = '11', solver = 'liblinear';
- Gradient boosting: learning_rate = 0.01, max_depth = 4, min_samples_leaf = 2, min_samples_split = 2, n_estimators = 117;
- K-nearest neighbourhood: metric = 'euclidean', n_neighbors = 20, weights = 'uniform';
- Naïve Bayes: var_smoothing = 0.12 and GausianNB;
- Decision trees: max_depth = 5, max_features = None, min_samples_leaf = 2, min_samples_split = 5.

4.2. Experimental Results

A comparative analysis of five imputation strategies and a complete case scenario are conducted. The results of this experiment show the effectiveness of machine learning methods when blended with different missing data approaches. The random state is set to 42, and the classification accuracy for each model is produced to measure the performance.

When GANs imputation is harnessed, the decision tree is the best-performing classifier with an accuracy rate of 93.01%, followed by random forest (92.92%), gradient boosting (92.33%), support vector machine (90.83%), logistic regression (90.76%), and naïve Bayes k-NN (89.29%), respectively. The K-nearest neighbours classifier is the worst-performing method, with an 88.68% accuracy.

For the complete-case approach, the best-performing machine learning algorithms in terms of their robustness when dealing with missing values are the random forest (93.01%), followed by gradient boosting (92.67%), naïve Bayes (90.01%), logistic regression (88.43%), k-NN (87.34%), and SVM (86.84%), respectively. The worst performance is by decision tree with an accuracy rate of 85.01%. The differences in performance of all seven machine learning algorithms are significant at the 95% significance level.

The above table shows the performance report concerning six performance metrics when cross-validation is employed. These models are developed under the GANs im-
putation for incomplete data since the simulation results of GANs from Table 2 revealed improved classification accuracy for algorithms compared to traditional methods. It can be noted that decision trees outperform all six other classifiers in all six performance metrics; gradient boosting, SVM, logistic regression, decision trees, and naïve Bayes follow this. The least-performing classifier is K-NN. The random forest and gradient boosting have almost identical performances, which could be attributed to the fact that both classifiers belong to the ensemble family.

Classifier	Complete Case	GANS	Median/Mean	Mode	MI	KNN
Logistic regression	0.8843	0.9076	0.9126	0.9108	0.9126	0.9114
Naive Bayes	0.9001	0.8929	0.8780	0.8780	0.8745	0.8780
Random forest	0.9301	0.9292	0.9244	0.9244	0.9250	0.9239
Decision trees	0.8526	0.9301	0.8566	0.8560	0.8519	0.8584
Gradient boosting	0.9267	0.9233	0.9233	0.9262	0.9239	0.9239
K-nearest neighbours	0.8734	0.8868	0.8662	0.8673	0.8662	0.8667
Support vector machine	0.8684	0.9083	0.9096	0.9096	0.8745	0.9096

Table 2. Single classifiers using GANs imputation strategy vs. others.

The recall rate of 93% implies the decision tree classifier correctly generalises the data well as shown from the Table 3 below. The GANs strategy has been optimised, and the epochs and batch size have been adjusted to improve the results. The accuracy rate of Naive Bayes was significantly enhanced by over 5% when the GANs strategy was optimised. The results suggest that most algorithms are more robust to incomplete data when GANs are leveraged as an imputation method.

Classifiers	Accuracy	Precision	Recall	F1 Score	HL	MCC
Logistics regression	0.9076	0.9004	0.9083	0.8926	0.0916	0.4983
Naïve Bayes	0.8929	0.8791	0.8929	0.8811	0.1071	0.4482
Random forest	0.9292	0.9330	0.9292	0.9153	0.0708	0.5955
Decision trees	0.9301	0.9352	0.9301	0.9161	0.0699	0.6022
Gradient boosting	0.9233	0.9209	0.9218	0.9088	0.0781	0.5865
K-nearest neighbours	0.8868	0.8532	0.8868	0.8462	0.1132	0.1704
Support vector machine	0.9083	0.9101	0.9083	0.8881	0.0917	0.5146

Table 3. Cross validation scores for the performance metrics per seven classifiers.

The Figure 2 below shows the learning curves of different classifiers to evaluate the performance of the classifiers in the training and cross-validation. The intention is to visually diagnose if the model overfits, underfits, or performs well over time during the learning experience. Additionally, the cross-validation ensures that classifiers' performance is robust and the learning curves are reliable. The learning curve for the training error is higher than that of cross-validation, with most classification models around 90% average accuracy in the train scores, which indicates that the models do not overfit and thus can generalise well in the test data. The lift curve performs well for all the classifiers because the training score and validation score accuracy are close to each other. Even though K-NN is the least performing classifier compared to others, the classifier shows a consistent trend and performs as observed in Figure 2.

The Figure 3 below shows the ROC curve for a comparative analysis of the performance of six different classifiers. The classifiers with the highest AUC scores are as follows: decision trees (82%) AUC and gradient boosting (82%) AUC, followed by random forest (81%) AUC, logistic regression (79%) AUC, support vector machine (74%) AUC, and naïve Bayes (73%). The K-NN has the lowest (60%) AUC score. Compared to other algorithms, the 82% AUC score for the two classifiers indicates that algorithms perform better for discriminatory power when distinguishing between positive and negative cases. For support vector machine (74%) AUC and naïve Bayes (73%) AUC, this implies a good discriminatory power and offers room to improve the classifiers, which can be achieved through further hyperparameter tuning. Furthermore, though decision trees and gradient boosting have completed (82%) higher AUC scores, decision trees stood out as having a better classification when taking into account all the other evaluation metrics, such as accuracy, precision, F1 score, recall, HL, and MCC. K-NN has the least discriminatory power compared to the six other algorithms and has shown consistent sensitivity to poor data across other evaluation metrics.



Figure 2. Learning curves for classification models.

The Figure 4 below shows that the decision tree algorithm has the lowest error rate, followed by random forest, gradient boosting, and logistic regression. This low error rate implies that these four algorithms are performing well in classification and generalising well in the validation dataset, as it reveals what percentage of the class predictions are invalid. Though naïve Bayes achieved the highest accuracy from the classification report table refer to Table 2, naïve Bayes have a somewhat lower error rate, above the average 9% error rate compared to SVM and K-NN. This error rate assessment is supplemented with ROC—AUC curve metrics for comprehensive model evaluation to limit the cases of bias in the event of class imbalance. Overall, the performance of DT, RF, GB, and LR consistently show more robustness towards poor data than NB, SVM, and K-NN classifiers.



Figure 3. Classifiers ROC curves and AUC.



Figure 4. Classification error rates.

5. Remarks and Conclusions

Overall, the results suggest that the GANs unsupervised learning strategy is effective for handling incomplete credit data. When this unsupervised method is harnessed, the results show that most single classifiers displayed robustness towards missing data problems due to their improved model performance. Additionally, though the sample size impacts the model performance, this strategy can overcome the complexity challenge. The accuracy of the classifier experimented with under old methods deteriorated compared to the GANs strategy.

The decision tree classifier is the best-performing algorithm concerning the six performance evaluation metrics. DT outperforms all the other five classifiers and has consistent outputs across all performance metrics. Random Forest showed consistent classification accuracy compared to other algorithms, regardless of which imputation technique is leveraged. Though random forest is the second-best performing algorithm, its AUC value is slightly lower than that of gradient boosting. This dynamic reveals the tradeoff between random forest and gradient boosting algorithms, and this is ideal because these two belong to the ensemble learning family. The other key finding is the efficiency of K-nearest neighbourhood imputation; the method is faster than the GANs and multiple imputation techniques. Although K-NN is not better than GANs for handling poor data, it works better than traditional statistical methods such as median and mode. By leveraging optimisation, the naïve Bayes classifier accuracy rate significantly improved. Hyperparameter optimisation is a progressive practice that is essential for any modelling process to achieve better performance.

In conclusion, this paper offers the first steps toward a clearer understanding of the relative benefits and drawbacks of methods for estimating credit risk from potentially incomplete information. We hope that future empirical and theoretical research into the significance of data quality in the credit risk sector and cutting-edge deep learning techniques as robust methods are expected to be motivated by this work.

Author Contributions: Conceptualisation, D.S.N. and B.T.; methodology, D.S.N.; software, D.S.N.; validation, D.S.N. and B.T.; formal analysis, D.S.N.; investigation, D.S.N.; resources, B.T.; data curation, D.S.N.; writing—original draft preparation, D.S.N., B.T. and J.H.C.P.; writing—review and editing, B.T. and J.H.C.P.; visualisation, D.S.N.; supervision, B.T. and J.H.C.P.; project administration, B.T.; funding acquisition, B.T. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Data Availability Statement: Data were downloaded from the Kaggle website. https://www.kaggle. com/datasets/adarshsng/lending-club-loan-data-csv (accessed on 28 February 2022).

Acknowledgments: We thank Aubrey Mabaso for assistance with data processing using Python 3.12.4 software and the anonymous reviewers for comments that significantly improved the manuscript.

Conflicts of Interest: The authors declare no conflicts of interest.

References

- Adams, Erwin, Beata Walczak, Chris Vervaet, Peter G. Risha, and Désiré L. Massart. 2002. Principal component analysis of dissolution data with missing elements. *International Journal of Pharmaceutics* 234: 169–78. [CrossRef] [PubMed]
- Alam, Farhad, and Sanjay Pachauri. 2017. Comparative Study of J48, Naive Bayes and One-R Classification Technique for Credit Card Fraud Detection using WEKA (Vol. 10, Issue 6). Available online: http://www.ripublication.com (accessed on 28 February 2022).
- Al-Helali, Baligh, Qi Chen, Bing Xue, and Mengjie Zhang. 2021. Multitree genetic programming with new operators for transfer learning in symbolic regression with incomplete data. *IEEE Transactions on Evolutionary Computation* 25: 1049–63. [CrossRef]
- Amaro, Miguel M. 2020. Credit Scoring: Comparison of Non-Parametric Techniques Against Logistic Regression. Doctoral dissertation, Information Management and Data Science School, Universidade Nova de Lisboa, Lisbon, Portugal.
- Andersen, Jens Verner, Kristian Sparre Andersen, Leif Lybecker Eskesen, and Suzanne Hyldahl. 2001. Models for Management of Banks' Credit Risk. *Monetary Review* 40: 43.
- Andrić, Kristina, Damir Kalpić, and Zoran Bohaček. 2019. An insight into the effects of class imbalance and sampling on classification accuracy in credit risk assessment. *Computer Science and Information Systems* 16: 155–78. [CrossRef]

- Banasik, John, Jonathan Crook, and Lyn Thomas. 2003. Sample selection bias in credit scoring models. *Journal of the Operational Research Society* 54: 822–32. [CrossRef]
- Berger, Magdalena, Todd H. Wagner, and Laurence C. Baker. 2005. Internet use and stigmatised illness. Social Science & Medicine 61: 1821–27.
- Bjornsdottir, Kristin, Gregory E. Bolton, Patricia D. McClellan-Green, Lee A. Jaykus, and David P. Green. 2009. Detection of gramnegative histamine-producing bacteria in fish: A comparative study. *Journal of Food Protection* 72: 1987–91. [CrossRef]
- Breiman, Leo, Jerome Friedman, R. A. Olshen, and Charles J. Stone. 1984. *Classification and Regression Trees*. New York: Chapman & Hall.
- Castermans, Gerd, David Martens, T. Van Gestel, Bart Hamers, and Bart Baesens. 2010. An overview and framework for PD backtesting and benchmarking. *Journal of the Operational Research Society* 61: 359–73. [CrossRef]
- Chaudhary, Archana, Savita Kolhe, and Raj Kamal. 2013. Machine learning classification techniques: A comparative study. *International Journal on Advanced Computer Theory and Engineering* 2: 21–25.
- Cömert, Zafer, and Adnan F. Kocamaz. 2017. Comparison of machine learning techniques for fetal heart rate classification. *Acta Physica Polonica A* 132: 451–54. [CrossRef]
- Danso, Samuel, Eric Atwell, and Owen Johnson. 2014. A comparative study of machine learning methods for verbal autopsy text classification. *arXiv* arXiv:1402.4380.
- David, Julie M., and Kannan Balakrishnan. 2010. Significance of Classification Techniques in Prediction of Learning Disabilities. International Journal of Artificial Intelligence & Applications 1: 111–20. [CrossRef]
- Dwyer, Douglas W. 2007. The distribution of defaults and Bayesian model validation. *Journal of Risk Model Validation* 1: 23–53. [CrossRef]
- Goodfellow, Ian, Jean Pouget-Abadie, Mehdi Mirza, Bing Xu, David Warde-Farley, Sherjil Ozair, Aaron Courville, and Yoshua Bengio. 2020. Generative adversarial networks. *Communications of the ACM* 63: 139–44. [CrossRef]
- Guo, Changbin, Ying So, and Woosung Jang. 2017. Evaluating Predictive Accuracy of Survival Models with PROC PHREG. Paper SAS462-2017. Cary: SAS Institute.
- Gurný, Petr, and Martin Gurný. 2013. Comparison of credit scoring models on probability of default estimation for us banks. *Prague Economic Papers* 22: 163–81. [CrossRef]
- Häger, David, and Lasse B. Andersen. 2010. A knowledge based approach to loss severity assessment in financial institutions using Bayesian networks and loss determinants. *European Journal of Operational Research* 207: 1635–44. [CrossRef]

Haug, Anders, Frederik Zachariassen, and Dennis Van Liempd. 2011. The costs of poor data quality. *Journal of Industrial Engineering* and Management (JIEM) 4: 168–93. [CrossRef]

- Hol, Suzan, Sjur Westgaard, and Nico van der Wijst. 2002. Capital Structure and the Prediction of Bankruptcy. Available online: https://www.fbv.kit.edu/symposium/9th/papers/Hol_Wes_Wij.pdf (accessed on 14 October 2024).
- Jerez, José M., Ignacio Molina, Pedro J. García-Laencina, Emilio Alba, Nuria Ribelles, Miguel Martín, and Leonardo Franco. 2010. Missing data imputation using statistical and machine learning methods in a real breast cancer problem. *Artificial Intelligence in Medicine* 50: 105–15. [CrossRef]
- Kalkan, Ömür Kaya, Yusuf Kara, and Hülya Kelecioğlu. 2018. Evaluating performance of missing data imputation methods in IRT analyses. International Journal of Assessment Tools in Education 5: 403–16. [CrossRef]
- Kim, Suk-Joong, and Eliza Wu. 2008. Sovereign credit ratings, capital flows and financial sector development in emerging markets. *Emerging Markets Review* 9: 17–39. [CrossRef]
- Kutty, Gopalan. 1990. Logistic regression and probability of default of developing countries debt. *Applied Economics* 22: 1649–60. [CrossRef]
- Ljubić, Marijana, Vladan Pavlovic, and Srecko Milancic. 2015. The impact of credit risk assessment on credit activity of commercial banks. *Megatrend Revija* 12: 141–52. [CrossRef]
- Louzada, Francisco, Anderson Ara, and Guilherme B. Fernandes. 2016. Classification methods applied to credit scoring: Systematic review and overall comparison. *Surveys in Operations Research and Management Science* 21: 117–34. [CrossRef]
- Madnick, Stuart E., and Yang W. Lee. 2009. Call for papers: Special issue on healthcare information quality: The challenges and opportunities in healthcare systems and services. *Journal of Data and Information Quality* 1: 1–3. [CrossRef]
- Majumder, Saibal, Aarti Singh, Anupama Singh, Mykola Karpenko, Haresh Kumar Sharma, and Somnath Mukhopadhyay. 2024. On the analytical study of the service quality of Indian Railways under soft-computing paradigm. *Transport* 39: 54–63. [CrossRef]
- Markovitch, Shaul, and Dan Rosenstein. 2002. Feature generation using general constructor functions. *Machine Learning* 49: 59–98. [CrossRef]
- Mason, Llew, Jonathan Baxter, Peter Bartlett, and Marcus Frean. 2000. Boosting Algorithms as Gradient Descent. Advances in Neural Information Processing Systems 12: 512–18.
- Muñoz-Cancino, Ricardo, Cristián Bravo, Sebastián A. Ríos, and Manuel Graña. 2023. On the combination of graph data for assessing thin-file borrowers' creditworthiness. *Expert Systems with Applications* 213: 118809. [CrossRef]
- Namvar, Anahita, Mohammad Siami, Fethi Rabhi, and Mohsen Naderpour. 2018. Credit risk prediction in an imbalanced social lending environment. *International Journal of Computational Intelligence Systems* 11: 925–35. [CrossRef]
- Nicoletti, Cheti, and Franco Peracchi. 2006. The effects of income imputation on microanalyses: Evidence from the European Community Household Panel. *Journal of the Royal Statistical Society Series A: Statistics in Society* 169: 625–46. [CrossRef]

- Ortl, Aljoša. 2016. Practical Aspects of Developing and Validating Corporate Probability of Default Model and the Use of Machine Learning. Master's thesis, Univerza v Ljubljani, Ekonomska fakulteta, Ljubljana, Slovenia.
- Osisanwo, F. Y., J. E. T. Akinsola, O. Awodele, J. O. Hinmikaiye, O. Olakanmi, and J. Akinjobi. 2017. Supervised machine learning algorithms: Classification and comparison. *International Journal of Computer Trends and Technology (IJCTT)* 48: 128–38.
- Paleologo, Giuseppe, André Elisseeff, and Antonini Gianluca. 2010. Subagging for credit scoring models. *European Journal of Operational Research* 201: 490–99. [CrossRef]
- Patil, Promod S., Jagannath V. Aghav, and Vikram Sareen. 2016. An overview of classification algorithms and ensemble methods in personal credit scoring. *IJCST* 7: 183–87.
- Payrovnaziri, Seyedeh N., Aiwen Xing, Salman Shaeke, Xiuwen Liu, Jiang Bian, and Zhe He. 2020. The impact of missing value imputation on the interpretations of predictive models: A case study on one-year mortality prediction in ICU patients with acute myocardial infarction. *MedRxiv* 10: 06-20124347.
- Pedersen, Alma B., Ellen M. Mikkelsen, Deirdre Cronin-Fenton, Nickolaj R. Kristensen, Tra M. Pham, Lars Pedersen, and Irene Petersen. 2017. Missing data and multiple imputation in clinical epidemiological research. *Clinical Epidemiology* 9: 157–66. [CrossRef]
- Reddy, R. Vijaya Kumar, and U. Ravi Babu. 2018. A review on classification techniques in machine learning. *International Journal of* Advance Research in Science and Engineering 7: 40–47.
- Shalev-Shwartz, Shai, and Shai Ben-David. 2014. Understanding Machine Learning: From Theory to Algorithms. Cambridge: Cambridge University Press.
- Sharma, Haresh Kumar, Saibal Majumder, Arindam Biswas, Olegas Prentkovskis, Samarjit Kar, and Paulius Skačkauskas. 2022. A Study on Decision-Making of the Indian Railways Reservation System during COVID-19. *Journal of Advanced Transportation* 2022: 7685375. [CrossRef]
- Siddiqi, Naeem. 2006. Credit Risk Scorecards: Developing and Implementing Intelligent Credit Scoring. Hoboken: John Wiley & Sons, Inc., vol. 1.
- Soofi, Aized Amin, and Arshad Awan. 2017. Classification Techniques in Machine Learning: Applications and Issues. Journal of Basic & Applied Sciences 13: 459–65.
- Stavseth, Marianne Riksheim, Thomas Clausen, and Jo Røislien. 2019. How handling missing data may impact conclusions: A comparison of six different imputation methods for categorical questionnaire data. *SAGE Open Medicine* 7: 2050312118822912. [CrossRef]
- Tang, Wenbin, Honglei Zhuang, and Jie Tang. 2011. Learning to infer social ties in large networks. In *Joint European Conference on Machine Learning and Knowledge Discovery in Databases*. Berlin and Heidelberg: Springer, pp. 381–97.
- Templ, Matthias, Alexander Kowarik, and Peter Filzmoser. 2011. Iterative stepwise regression imputation using standard and robust methods. *Computational Statistics and Data Analysis* 55: 2793–806. [CrossRef]
- Wójcicka, Aleksandra. 2012. Calibration of a credit rating scale for Polish companies. Operations Research and Decisions 22: 63–73.
- Woods, Adrienne D., Pamela Davis-Kean, Max Andrew Halvorson, Kevin King, Jessica Logan, Menglin Xu, Sierra Bainter, Denver Brown, James M. Clay, Rick A. Cruz, and et al. 2021. Missing Data and Multiple Imputation Decision Tree: Best Practices for Addressing Missing Data Through Multiple Imputation. Available online: https://doi.org/10.31234/osf.io/mdw5r (accessed on 14 October 2024).
- Wu, Xiaoxue, Wei Zheng, Xin Xia, and David Lo. 2021. Data quality matters: A case study on data label correctness for security bug report prediction. *IEEE Transactions on Software Engineering* 48: 2541–56. [CrossRef]
- Yadav, Sanjeev Kumar, and Arvind Kumar Tiwari. 2015. Classification of enzymes using machine learning based approaches: A review. *Machine Learning and Applications* 2: 30–49.
- Yan, Gujun. 2023. Autoencoder-based generator for credit information recovery in rural banks. International Journal of Industrial Engineering: Theory Applications and Practice 30: 326–35.
- Yoon, Jinsung, James Jordon, and Mihaela Schaar. 2018. Gain: Missing data imputation using generative adversarial nets. In *International Conference on Machine Learning*. Stockholm: PMLR, pp. 5689–98.
- Zhang, Qingfen. 2015. Modeling the Probability of Mortgage Default via Logistic Regression and Survival Analysis. Master's thesis, University of Rhode Island, Kingston, RI, USA.
- Zhao, Feng, Yan Lu, Xinning Li, Lina Wang, Yingjie Song, Deming Fan, Caiming Zhang, and Xiaobo Chen. 2022. Multiple imputation method of missing credit risk assessment data based on generative adversarial networks. *Applied Soft Computing* 126: 109273. [CrossRef]
- Zhou, Youran, Sunil Aryal, and Mohamed Reda Bouadjenek. 2024. Review for Handling Missing Data with Special Missing Mechanism. Available online: http://arxiv.org/abs/2404.04905 (accessed on 9 April 2024).

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.





Article Risk Management in Product Diversification: The Role of Managerial Overconfidence in Cost Stickiness—Evidence from Iran

Mona Parsaei^{1,*}, Davood Askarany², Mahtab Maleki¹ and Ali Rahmani¹

- ¹ Department of Accounting, Faculty of Social Sciences and Economics, Alzahra University, Tehran 19938 93973, Iran; maleki.acct@gmail.com (M.M.); rahmani@alzahra.ac.ir (A.R.)
- ² Department of Accounting and Finance, Business School, University of Auckland, Auckland 1010, New Zealand; d.askarany@auckland.ac.nz
- * Correspondence: m.parsaei@alzahra.ac.ir

Abstract: Purpose: This study investigates the relationship between product diversification strategy and cost stickiness, focusing on managerial overconfidence as a moderating factor. It aims to address a critical gap in the literature by providing empirical insights grounded in the Resource-Based View (RBV) theory, specifically examining firms listed on the Tehran Stock Exchange. Methodology: Utilizing a sample of 149 companies from the Tehran Stock Exchange in Iran spanning from 2015 to 2021, this study tests two hypotheses: (1) a positive relationship between product diversification and cost stickiness and (2) the amplification of this relationship by managerial overconfidence. Product diversification is quantified using the Herfindahl Index, while managerial overconfidence is measured through an investment-based index derived from capital expenditures. Cost stickiness is assessed by analysing the asymmetric behaviour of costs in response to changes in sales, focusing on how costs tend to remain high even when sales decrease. Findings: The empirical results substantiate both hypotheses, demonstrating a significant positive relationship between product diversification strategy and cost stickiness. Furthermore, managerial overconfidence amplifies this relationship, highlighting the role of internal resources and managerial perceptions in shaping cost behaviour. Originality: This study contributes substantially to the literature by being among the first to empirically examine the interplay between product diversification strategy, cost stickiness, and managerial overconfidence. Extending the RBV theory to cost behaviour and strategic management provides novel insights for scholars and practitioners in entrepreneurship, corporate strategy, and organizational behaviour. The findings underscore the importance of strategic choices and managerial traits in determining cost stickiness, offering valuable implications for financial analysts, auditors, and stakeholders.

Keywords: product diversification strategy; organizational flexibility; cost behaviour; decision making; cost stickiness; managerial overconfidence; risk

1. Introduction

In an increasingly competitive and volatile business environment, understanding the dynamics of cost behaviour and strategic management is crucial for organizations striving for long-term success (Hitt et al. 2020). Organizations encounter many strategic challenges, among the most critical of which is the need to effectively align their growth strategies with rigorous cost management practices (Grant 2016). Product diversification strategy has received considerable attention in the literature (Bengtsson 2000; Garrido-Prada et al. 2019; Tang et al. 2019; Wang et al. 2014). It represents organizational flexibility by enabling companies to adapt to changing market conditions through an expanded range of products and services. It improves stability by allowing the firm to spread its business risks across different product lines and reduce its vulnerability to market fluctuations while maximising



Citation: Parsaei, Mona, Davood Askarany, Mahtab Maleki, and Ali Rahmani. 2024. Risk Management in Product Diversification: The Role of Managerial Overconfidence in Cost Stickiness—Evidence from Iran. *Risks* 12: 150. https://doi.org/10.3390/ risks12100150

Academic Editors: Eulália Mota Santos and Margarida Freitas Oliveira

Received: 10 August 2024 Revised: 16 September 2024 Accepted: 18 September 2024 Published: 24 September 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). the efficiency of its resource allocation (Dinh et al. 2019; Friberg 2021; Hautz et al. 2014; Mammen et al. 2021; Smeritschnig et al. 2021; Sohl et al. 2022; Solano et al. 2019). The literature supports this notion that product diversification can be considered as one of the competitive strategies (Zúñiga-Vicente et al. 2019). However, given the organisational changes and costs that a product diversification strategy entails, we are unsure if there would be less cost reduction during declining sales, leading to increased cost stickiness. This is the primary question that this study attempts to address.

To examine the relationship between business strategy, product diversification, and cost behaviour, it is crucial to emphasize the importance of cost management in organizations. Cost is one of the most critical factors for any enterprise, and effective cost management is essential for achieving an optimal internal structure, earning a profit, and maintaining continuous development (Cai et al. 2019; Ding et al. 2021; Dormady et al. 2022; Stadtherr and Wouters 2021). Lee et al. (2020) consider cost management a core component of internal structure optimization, risk control, and profitability. Hence, understanding cost behaviour is fundamental to management and cost accounting (Anderson et al. 2003). Managerial overconfidence, a behavioural bias where managers overestimate their abilities and underestimate risks, plays a pivotal role in strategic decision-making (Gurdgiev and Ni 2023). Overconfident managers are more likely to retain resources during sales declines, anticipating future rebounds, which can intensify cost stickiness (Chen et al. 2022). This study explores the relationship between product diversification and cost stickiness, specifically focusing on how managerial overconfidence influences this dynamic.

The theoretical foundation of this study is grounded in the Resource-Based View (RBV) theory, which posits that firms can gain a competitive advantage through unique resources and capabilities (Barney et al. 2011). In this context, product diversification represents a unique resource, while cost stickiness is viewed as resource management and risk mitigation capabilities. This study formulates two hypotheses for testing: First, it proposes a positive relationship between product diversification strategy and cost stickiness. This proposition aligns with RBV, as diversification often involves significant investments, resulting in elevated resource adjustment costs and delayed reduction of underutilized resources during sales declines. Second, this study suggests that managerial overconfidence strengthens the direct link between product diversification and cost stickiness. This hypothesis is consistent with RBV, as overconfident managers are more likely to retain underutilized resources, anticipating future sales rebounds, thus intensifying cost stickiness.

Evolving regulations in Iran's environment can influence managerial decisions, often leading to cautious resource management during sales declines. This careful approach, driven by overconfidence and regulatory considerations, can contribute to increased cost stickiness, particularly in firms with diversified product portfolios (Nili and Rastad 2007). In addition, external economic conditions have affected the business environment by reducing access to some international markets and capital, leading to challenges in operational stability. In response, managers may adopt more optimistic behaviours as a strategic adjustment, contributing to cost stickiness as they retain resources in anticipation of future opportunities (Katzman 2014). Moreover, the inflation rates add complexity to cost management, as the real value of costs can be distorted. Overconfident managers may underestimate the risks of inflation, leading to increased resource commitments despite declining sales, which heightens cost stickiness (Pesaran 2012). Finally, cultural context influences how risk is perceived and handled. Iranian cultures where risk-taking is en-couraged, managerial overconfidence may be more prevalent, as leaders may feel empow-ered to make bold decisions. This cultural dynamic can lead to decisions favouring maintaining resources during downturns, thereby increasing cost stickiness (Nazarian and Atkinson 2013).

Despite the extensive literature on cost stickiness and its influencing factors in various contexts (Chen et al. 2022; Costa and Habib 2021), the role of managerial overconfidence in moderating the relationship between product diversification and cost stickiness has not been extensively studied, particularly in the context of the Iranian market. This study investigates the intricate relationship between product diversification strategy and cost

stickiness, focusing on the moderating influence of managerial overconfidence on decisionmaking. It aims to fill a notable gap in the existing literature by providing empirical insights into this underexplored area, particularly within the risk management framework in corporate finance.

Using a sample of 149 companies listed on the Tehran Stock Exchange in Iran from 2015 to 2021, this study employs rigorous statistical analysis to assess these hypotheses. The results empirically substantiate both hypotheses, shedding light on the positive relationship between product diversification and cost stickiness and the amplifying effect of managerial overconfidence.

This study makes a significant theoretical contribution by anchoring its research questions in the RBV theory, providing robust empirical evidence, and delivering valuable insights for scholars and practitioners of entrepreneurship, corporate strategy, and organizational behaviour worldwide. By focusing on the intersection of product diversification, cost behaviour, and managerial traits, this study offers new insights into risk management frameworks and financial strategies, contributing to the broader discourse on corporate finance and strategic decision-making in high-risk environments. The remainder of this study is structured as follows. Section 2 reviews the research literature and develops the hypotheses. Section 3 discusses the research method. Section 4 reports the results, and Sections 5 and 6 provide the discussion and conclusion.

2. Literature Review and Hypotheses Development

Traditional models of cost behaviour divide costs into fixed and variable based on some conventional drivers, such as the sale or production volumes (Bugeja et al. 2015). However, recent research indicates that the relationship between costs and cost drivers is more complex than traditionally assumed, involving multiple factors influencing cost behaviour (Banker and Byzalov 2014; Costa and Habib 2021). Cooper and Kaplan (1998) show that costs increase more when activity rises than they decrease with a proportional decrease in activity. This asymmetric cost behaviour is referred to as cost stickiness (Cooper and Kaplan 1998).

Recent research in cost management adds two crucial concepts to traditional cost analysis. First, cost behaviour is driven by managerial decisions, which depend on managers' expectations of future demand and their incentives (Jin and Wu 2021). Second, when sales decline, managers' ability to reduce unutilised resources becomes limited due to resource adjustment costs, such as the cost of employee layoffs (Banker et al. 2018). Managers' optimism about future demand, managerial incentives driven by self-interest (e.g., empire-building), and adjustment costs are the key factors that motivate managers to retain unutilised resources during declining sales. Managers' unwillingness to cut resources in such conditions will lead to cost stickiness (Banker et al. 2013; Hartlieb et al. 2020; Xue and Hong 2016).

Asymmetric cost behaviour and cost management have broad implications for owners, investors, creditors, employees (concerning job security), and other stakeholders (Costa and Habib 2021). Cost stickiness affects the authenticity of a firm's costs and reduces the effectiveness of its resource allocation (Li et al. 2020). Recent studies have investigated the relationship between cost stickiness and various factors (Chen and Xu 2023; Costa and Habib 2021; Haga et al. 2019; Li et al. 2020). However, the relationship between business strategies, particularly growth strategies and cost stickiness, has not been explored. Anderson et al. (2003) argue that a firm's strategy choice shapes its cost structure. Porter's generic business-level strategy, namely, cost leadership, differentiation, and focus, have become a central paradigm in the business policy literature (Porter 1980). Each represents a different approach to creating and maintaining a competitive advantage. For example, companies that adopt a cost leadership strategy have a more flexible cost structure. When sales decline, they have to reduce the price of products to maintain their competitive advantage. As a result, they cut costs and experience lower cost stickiness. On the other hand, companies that adopt a differentiation strategy will face high adjustment costs when

sales decline. As a result, they retain unutilised resources and experience higher cost stickiness (Zhong et al. 2020).

Diversification is one of the four growth strategies outlined in Ansoff's product-market matrix (Ansoff 1957). Diversification is a strategy where the company expands its products and services. Diversification has been a common competitive strategy for decades (Jha et al. 2021). A diversification strategy targets new markets with new products. It assumes that a company can adopt and require new resources and capabilities and make necessary changes in organisational structure and processes (Zúñiga-Vicente et al. 2019). So, we can expect that companies adopting a diversification strategy must restructure their business units and change current practices and existing contracts. This strategy requires comprehensive planning to implement diversification processes using internal and external resources and expertise. As a result, recruiting specialists, hiring and training new employees, and new equipment might be needed, which can be very costly (Lin et al. 2020).

As the business environment becomes more competitive, effective cost management has become an essential element of business operations (Chen et al. 2022). As an intuitionistic reflection of the cost decisions made in the accounting information system, cost behaviour shows the current conditions and tendencies of the company's resource investment. It reflects the executives' strategies in response to changes in economic conditions (Lee et al. 2020). Resource adjustment decisions are one of these responses that significantly impact firm performance (Chen et al. 2022).

Another factor that may affect organisational strategy and product diversification is managerial behaviour biases (Kak 2004). One of these biases is managerial overconfidence, which refers to managers' tendency to overestimate their abilities and/or overinvest (Hyun and Seung 2019). Overconfidence is one of the essential concepts in modern behavioural finance. Overconfidence causes people to overestimate their knowledge and skills and underestimate risks, feeling that they have control over events, which may not be the case (Weinstein 1980). Anderson et al. (2003) argue that one reason for asymmetric cost behaviour is the CEO's optimistic expectations about future sales improvement (Hur et al. 2019; Schrand and Zechman 2012). Hribar and Yang (2016) found that overconfidence increases the degree of optimism in management forecasts. As a result, managers will be more willing to retain slack resources when sales decline to reduce adjustment costs (Hribar and Yang 2016; Lee et al. 2020). Overconfidence can manifest itself by adding more resources when demand increases or retaining unutilised resources when demand decreases, leading to greater cost stickiness (Chen et al. 2022). On the other hand, Chu et al. (2021) found that companies with overconfident managers will have greater cost stickiness if they adopt a differentiation strategy than a cost leadership strategy (Chu et al. 2021).

Companies may adopt different strategies in a competitive market, leading to asymmetric cost behaviour. For example, as noted earlier, cost stickiness can increase with a differentiation strategy increase but decrease with a cost leadership strategy (Zhong et al. 2020). On the other hand, a product diversification strategy aimed at higher profitability and growth can increase sales volume and lead to better financial performance (Dinh et al. 2019; Omosa et al. 2022). Given that companies with a product diversification strategy require new investments, skills, equipment, practices, and changes in organisational structure and processes, it is possible that, when future sales are uncertain, overconfident managers will maintain the business scale and avoid adjusting unutilised resources in the hope of obtaining higher returns in the future, which will lead to more sticky cost behaviour (Chen et al. 2022). Therefore, we expect managerial overconfidence to moderate the relationship between product diversification strategy and cost stickiness. The present research seeks to answer the following questions: Is there a relationship between product diversification strategy and cost stickiness?

Following prior research (Lin et al. 2020; Tang et al. 2022; Zúñiga-Vicente et al. 2019), product diversification is measured using the Herfindahl Index. Following Killins et al. (2021), managerial overconfidence is measured using an investment-based index, which

is calculated based on the median of capital expenditures. Cost stickiness is measured according to the model proposed by Banker et al. (2013). Our results show a significant positive relationship between product diversification strategy and cost stickiness. In other words, cost stickiness is higher in companies with higher product diversification. In addition, managerial overconfidence increases asymmetric cost behaviour in companies with a product diversification strategy.

In the literature, various methodologies such as scenario analysis, econometric modelling, and stress testing are commonly used to assess the impact of global risks on firms (Schwartz 1997; Wooldridge 2010; Borio et al. 2014). Scenario analysis and stress testing often rely on hypothetical scenarios to predict outcomes, while econometric models analyse historical data to quantify these risks. Our study diverges from these approaches by using empirical data from the Tehran Stock Exchange to examine how economic conditions in Iran affect cost stickiness and managerial decisions. This empirical approach provides a more accurate reflection of real-world impacts, offering specific insights into the Iranian context that might differ from global risk assessments based on hypothetical scenarios.

Product diversification can mitigate risks by spreading them across different markets, which is beneficial in stable or growing economies. However, during economic downturns, the high costs associated with maintaining diverse product lines may outweigh the benefits. The economic environment, including inflation, currency fluctuations, sanctions, and evolving regulations, can further complicate diversification efforts, making them more costly and less effective (Zúñiga-Vicente et al. 2019; Katzman 2014). Cost stickiness, while strategic in some cases, can become a liability during economic downturns, as maintaining unutilized resources can strain a firm's financial health. Proposed strategies, like cautious resource allocation, may work in stable conditions but could face challenges in volatile economies (Banker et al. 2013). Managerial overconfidence can exacerbate cost stickiness, particularly in volatile economic environments where cost-cutting is essential but may be overlooked (Chen et al. 2022).

The present research will contribute to the literature in two significant ways. First, it sheds light on the relationship between product diversification strategy and cost stickiness, while prior research (Chu et al. 2021; Zhong et al. 2020) has focused on other business strategies. Therefore, the findings of this study can expand the literature on cost behaviour and product diversification strategy. Secondly, by investigating the effect of managerial overconfidence, the present research contributes to the literature on the psychological factors affecting decision-making.

There is an accumulated body of literature on diversification, yet the results are ambiguous and inconclusive (Askarany and Spraakman 2020; Blavatskyy 2023; Denuit et al. 2015; Galavotti et al. 2017; Guidolin and Rinaldi 2013; Wang et al. 2020). Askarany and Spraakman (2020) have investigated the relationship between diversification and financial performance through an excess-capacity theory lens. They found a positive and significant association between diversification and financial performance due to spare capacity (a kind of cost stickiness). In other words, they found that firms with higher excess capacity need to pay less for diversification than those with less or no spare capacity. Therefore, they are more likely to achieve better financial performance than those with less or no excess capacity. This study suggests the possibility of a relationship between diversification and cost behaviour (including cost stickiness).

Other studies explain the diversification through the lens of the Resource-Based View theory (Khurana and Farhat 2021; Situm 2019). The Resource-Based View theory (RBV) posits that a firm's unique resources and capabilities are the primary drivers of its competitive advantage and performance (Barney et al. 2021). Adopting RBV in this study, we suggest the resource of interest is the diversification strategy, and the capability is the ability to manage cost stickiness.

2.1. Product Diversification Strategy and Cost Stickiness

Traditional models of cost behaviour view cost changes as a direct result of changes in the volume of activity (Noreen 1991). Costs are traditionally classified into fixed and variable costs. A fixed cost remains constant as a company's sales or production volume changes, while a variable cost changes in proportion to sales and production volume (Anderson et al. 2003). Recent cost stickiness research suggests that the traditional assumption of symmetric cost behaviour does not always hold in practice (Calleja et al. 2006; Lee et al. 2020). A critical review of the findings of Cooper and Kaplan (1998) and Noreen and Soderstrom (1997) introduces a new perspective to cost accounting research. They provided empirical evidence that, in some cases, cost behaviour is asymmetric (Cooper and Kaplan 1998; Noreen and Soderstrom 1997). Later, Anderson et al. (2003) showed that selling, general, and administrative (SG&A) costs decrease less when sales decrease than when sales increase. In other words, these studies state that cost changes are determined by the shift in volume and the direction of activities. This asymmetric cost behaviour is referred to as cost stickiness. The results of Anderson and Lanen (2007) show that SG&A costs increase by an average of 0.55% for a 1% increase in sales but only decrease by 0.35% for a 1% decrease in sales (Anderson and Lanen 2007).

Banker and Byzalov (2014) discussed three factors that cause cost stickiness: agency problems, optimistic managerial expectations, and adjustment costs. Agency problems, where managers prioritize their interests over those of shareholders, can lead to sticky cost behaviour, particularly when managers with empire-building tendencies fail to reduce resources during sales declines (Chen et al. 2012; Zhong et al. 2020). Managerial optimism further intensifies cost stickiness, as optimistic managers are more likely to retain resources even when sales decrease, unlike pessimistic managers who reduce committed resources in response to declining sales. Finally, decisions about resource adjustment are made mainly by the management and significantly impact firm performance (Chen et al. 2022). Hiring/layoff of employees and installation/deployment costs for equipment are examples of adjustment costs. As sales decline, managers tend to retain a certain amount of slack capacity instead of reducing SG&A costs (Banker and Byzalov 2014). Pamplona et al. (2016) found evidence of asymmetric cost behaviour in large South American companies. By analysing the effects of inflation and GDP growth, they also showed that macroeconomic factors explain differences in companies' cost behaviour. Inflation was negatively associated with cost behaviour, while GDP growth was positively associated with cost behaviour. Zhong et al. (2020) showed that cost stickiness is higher in companies with a differentiation strategy than those with a cost leadership strategy.

Large companies need to make strategic decisions to improve their long-term performance. One of these decisions is product diversification (Azman et al. 2020). The reasons why firms diversify into new businesses and their implications for competitive advantage have been at the core of research on corporate strategy (Becerra et al. 2020). Park and Jang (2013) define diversification as the entry of the company into new markets (sectors, industries) where they previously had no presence (Park and Jang 2013). Zúñiga-Vicente et al. (2019) showed that managers who pursue moderate levels of diversification achieve better profitability during a boom period than those who adopt high or low levels of diversification.

In contrast, moderate and high levels of diversification are equally adequate during an economic downturn. They also found that internationalisation negatively affects profitability, independent of the economic cycle. The effect of product diversification on a firm's productivity depends on market structure, among other factors. For example, companies operating in a market with little opportunity for growth should diversify by entering different markets. Conversely, diversifying products can be a good strategy for companies operating in a competitive industry. By exploiting and leveraging their resources across multiple markets, companies can gain more competitive advantages and opportunities for further development (Hautz et al. 2013) The effect of diversification on firm performance has been extensively investigated (Arte and Larimo 2022; Azman et al. 2020). Some studies have reported a positive effect (Kuppuswamy and Villalonga 2016), while others have found a negative impact (Arte and Larimo 2022; Rugman 1976). Lin et al. (2020) examined the effect of operational stickiness on product quality and the moderating effect of product diversification in Chinese exporting firms. The results show an inverted U-shaped relationship between operational stickiness and product quality and that product diversification strategy significantly moderates this relationship. Meanwhile, several studies have reported both positive and negative effects of diversification on performance (Garrido-Prada et al. 2019). For example, Azman et al. (2020) examined the impact of different types of product diversification on firm performance. They found that unrelated diversification reduced company performance in 2003–2016, while related diversification positively affected firm performance during 2010–2016. Similarly, Arte and Larimo (2022) show that the relationship between international diversification and firm performance follows an inverted U-shape and that a firm's performance is higher with low/related product diversity than with high/unrelated product diversity.

Li and Zheng (2017) show that product market competition increases cost stickiness. They argue that product market competition encourages companies to invest frequently to exploit new investment opportunities under uncertainty (Li and Zheng 2017). Product diversification is also a strategy that requires significant investment in human resources, new machinery and equipment, research and development (R&D), and advertising (Lin et al. 2020). Additionally, this strategy may stem from the agency problem; the top management may try to reduce their employment and reputation risk by diversifying products (Lin et al. 2020). A decrease in sales indicates that the company is losing its competitive position in the market. This can encourage managers to commit more resources to R&D, quality control, marketing, or customer satisfaction to regain the company's market share and maintain its competitive position in the industry (Li and Zheng 2017).

Moreover, under economic crisis and declining sales, companies may invest less in the related segment and more in unrelated segments to maintain some of their business lines, increasing their internal control, governance, and coordination costs (Chang and Wang 2007). In addition, resource adjustment costs should also be considered. Generally, the cost of increasing resources is lower than the adjustment cost of decreasing resources (Subramaniam and Watson 2016). Due to high adjustment costs, managers are more likely to retain slack resources (Anderson et al. 2003). As a result, costs will not change in proportion to the volume of business activity, leading to cost stickiness (Anderson et al. 2003). Balakrishnan et al. (2004) found that sticky costs are associated with the level of capacity utilisation.

On the other hand, Anderson et al. (2003) showed that companies with more human and physical resources exhibit higher levels of cost stickiness. As Lin et al. (2020) noted, the product diversification strategy requires high human resources and equipment levels. Employment growth increases in the years before diversification, while asset and sales growth increases during and after diversification (Coad and Guenther 2014). Therefore, it can be concluded that resource adjustment costs are higher in companies that adopt a product diversification strategy.

According to Anderson et al. (2003), choosing a business strategy affects cost behaviour. Zhong et al. (2020) documented differences in cost stickiness between companies using differentiation and cost leadership strategies. According to Chen et al. (2022), when sales decline, there is a significant adjustment cost for companies with a product diversification strategy since implementing this strategy requires substantial investment in the company's structure. Therefore, it can be argued that the product diversification strategy, one of the growth strategies in Ansoff's matrix (Dawes 2018; Hussain et al. 2013), is associated with cost stickiness. Given the above, we propose our first hypothesis as follows:

H1. There is a significant relationship between product diversification strategy and cost stickiness.

2.2. Managerial Overconfidence, Product Diversification and Cost Stickiness

Managerial overconfidence is a behavioural bias that overestimates one's abilities and underestimates potential risks (Weinstein 1980). Psychological evidence shows that people overestimate the outcomes related to their abilities and underestimate the likelihood of adverse events. This behavioural tendency is called overconfidence (Hyun and Seung 2019). Furthermore, although overconfidence does not equal greater risk-taking, overconfident managers make less conservative decisions by relying too much on positive information (Gervais et al. 2011). Hilary and Hsu (2011) found that managers who accurately forecast earnings in the previous four quarters are less accurate in their subsequent forecasts, suggesting that overconfidence leads managers to be more optimistic. Managers' cognitive biases affect their decision-making since overconfident managers are optimistic and tend to overestimate returns while underestimating their investment risk (Hilary and Hsu 2011; Killins et al. 2021). As managers become more overconfident, the risk of making wrong decisions increases. This can lead to crises, scandals, and poor performance (Johnson and Fowler 2011; Kunz and Sonnenholzner 2023).

However, some studies have highlighted the benefits of overconfidence. For example, overconfidence motivates managers to pursue promising but risky projects. Overconfident managers tend to pursue innovation and exploit innovative growth opportunities with investment, achieving tremendous innovative success (Hirshleifer et al. 2012). Similarly, Hyun and Seung (2019) found that companies with overconfident managers have higher returns on net operating assets and can better forecast changes in future earnings.

Executives and their personal traits significantly impact organisational capabilities, decision-making, and operations (Kunz and Sonnenholzner 2023). Overconfidence is one of these traits. Previous research shows that managerial overconfidence affects a company's investment, financing, and financial reporting decisions (Gervais et al. 2011). Overconfident managers are more likely to overestimate returns on investment and underestimate the associated risks (Aabo et al. 2021; Hirshleifer et al. 2012). Empirical studies have provided evidence about the impact of managerial overconfidence on different strategies. Companies with a diversification strategy and overconfident managers experience a decline in value compared to those run by their rational counterparts (Schumacher et al. 2020). Chen et al. (2022) found that managerial overconfidence as an essential factor in resource adjustment decisions is positively associated with cost stickiness.

Similarly, Chu et al. (2021) showed that companies with a differentiation strategy and overconfident managers face higher cost stickiness than those with a cost leadership strategy. Overconfident managers are generally more optimistic and overestimate their abilities to rebound during periods of declining sales (Chen et al. 2022). Therefore, managers' overconfidence can cause them to ignore cost adjustment, thus increasing cost stickiness (Kuang et al. 2015). Given the above, we propose our second hypothesis as follows:

H2. *Managerial overconfidence moderates the relationship between product diversification strategy and cost stickiness.*

3. Methodology

3.1. Models and Variables

In the cost stickiness literature, the model developed by Anderson et al. (2003) and Banker et al. (2013) is often used to determine the presence or absence of cost stickiness (Haga et al. 2019; Lee et al. 2020). This model is shown in Equation (1).

$$\Delta lnCOGS_{it} = \beta_0 + \beta_1 \Delta lnsales_{it} + \beta_2 D_{it} * \Delta lnsales_{it} + \varepsilon_{it}$$
(1)

where $\Delta \ln COGS_{it}$ is the dependent variable, equal to the log change in the cost of goods sold for firm *i* in year *t*; $\Delta \ln Sales_{it}$ is the log change in the sales revenue for firm *i* in year *t*; D_{it} is a dummy variable that equals one if sales decrease in year *t*, and zero otherwise.

Banker's model is based on Anderson's model, which considers resource adjustment decisions under sales increases as a function of managerial discretion and theoretically allows for differences in these decisions under different conditions. In contrast, the traditional model of cost stickiness considers managerial discretion only under declining sales and assumes a mechanical relationship between resource expansion (increase in cost) and increases in sales, which does not account for managerial discretion in decisions to expand resources. Given that our hypothesis considers the possibility of different reactions by managers in various situations (either increase or decrease in sales), our proposed model is adopted from Banker et al. (2013), as shown in Equation (2). Incorporating interaction terms allows us to examine the conditional effects of product diversification based on changes in sales ($\Delta \ln Sales$) and the occurrence of sales decrease (D). In this model, the coefficient β_1 denotes the percentage increase in total cost per 1% increase in sales revenue. The coefficient β_2 captures the degree of asymmetry in cost responses to decreases versus increases in sales. Therefore, $\beta_1 + \beta_2$ show the percentage decrease in total cost per 1% decrease in sales revenue. Cost stickiness is present in a company if β_1 is significant and positive and β_2 is significant and negative. Moreover, if the coefficient of the independent variable (β_4) is negative and significant, it suggests that the product diversification strategy reduces the variability of costs concerning changes in revenue and, as a result, increases cost stickiness. Thus, the proposed model is as follows:

$$\Delta \ln COGS_{it} = \beta_0 + \beta_1 \Delta \ln Sales_{it} + \beta_2 D_{it} * \Delta \ln Sales_{it} + \beta_3 PD_{it} * \Delta \ln Sales_{it} + \beta_4 D_{it} * PD_{it} * \Delta \ln Sales_{it} + \beta_5 ASINT_{it} * \Delta \ln Sales_{it} + \beta_6 D_{it} * ASINT_{it} * \Delta \ln Sales_{it} + \beta_7 EMPINT_{it} * \Delta \ln Sales_{it} + \beta_8 D_{it} * EMPINT_{it} * \Delta \ln Sales_{it} + \beta_9 GDP_t * \Delta \ln Sales_{it} + \beta_{10} D_{it} * GDP_t * \Delta \ln Sales_{it} + \beta_{11} \sum T_i Industry + \beta_{12} \sum \theta_i Year + \varepsilon_{it}$$

$$(2)$$

where $\Delta \ln COGS_{it}$ is the dependent variable, equal to the log change in the cost of goods sold for firm *i* in year *t*; $\Delta \ln Sales_{it}$ is the log change in the sales revenue for firm *i* in year *t*; D_{it} is a dummy variable that equals one if sales decrease in year *t*, and zero otherwise; and PD_{it} is product diversification as the independent variable, measured by the Herfindahl index (Tang et al. 2022; Zúñiga-Vicente et al. 2019). The Herfindahl index is calculated as shown in Equation (3):

$$\text{Herfindahl} = 1 - \sum_{i=1}^{n} p_i^2 \tag{3}$$

where p_i^2 is the ratio of the sales of each segment to the total sales of the company, and *n* is the number of segments. In Iran, the categorization of industries has been implemented following the standards outlined in the Fourth Edition of the International Standard Industrial Classification (ISIC). The lower the index (i.e., the closer it is to zero), the lower the product diversification.

Following Zhong et al. (2020), Cai et al. (2019), and Anderson et al. (2003), the following control variables are included in the model:

- ASINT_{it}: Asset intensity, calculated as the ratio of total assets to sales revenue at the end of the fiscal year.
- *EMPINT*_{*it*}: Employee intensity is calculated as the ratio of the number of employees to sales revenue at the end of the fiscal year.
- *GDP*_t: Growth rate of gross domestic product.

We expect β_6 and β_8 to be negative since asset- and employee-intensive firms tend to have high-cost stickiness. GDP growth is included to control macroeconomic conditions that might affect the firms' operations. We expect β_{10} to be negative since the GDP growth rate increases managers' optimism regarding future sales, encouraging them to retain slack resources, which could lead to greater cost stickiness (Costa and Habib 2021). Due to Iran's inflationary economy, the model is estimated after deflating the primary variables, i.e., sales revenue and costs, based on the inflation rate reported by the Central Bank. Deflation for each year is performed by dividing the sales and expenses of a given year by one plus the inflation rate of that year (MohammadRezaei and Mohd-Saleh 2017). The rationale is to remove the effects of inflation since the proposed cost stickiness model uses sales as a measure of activity level. Therefore, deflated values do not include the increase in sales/costs caused by inflation. In addition, year and industry dummies are included in the model to control for year and industry-fixed effects. The block diagram is presented in Figure 1.



Figure 1. Block diagram. Source: authors' own work.

The second hypothesis examines the moderating effect of managerial overconfidence (OCD) on the relationship between product diversification strategy and cost stickiness, as shown in Figure 1. We employ a split-sample method to test this hypothesis, following a similar approach to Costa and Habib (2021). This method divides our statistical sample into companies exhibiting managerial overconfidence and those without. Subsequently, distinct regression models are estimated for each subgroup to examine the differential impact of product diversification on cost stickiness based on the presence or absence of managerial overconfidence. Overconfidence is measured using the index proposed by Killins et al. (2021). This index is a dummy variable equal to one if the capital expenditures scaled by total assets in a given year are greater than the industry median and zero otherwise (Ahmed and Duellman 2013). In the proposed model, a statistically significant difference in coefficients on β_4 indicates the moderating effect of managerial overconfidence.

3.2. Population and Sample

Our statistical population consists of all the companies listed on the Tehran Stock Exchange (TSE) for seven years from 2015 to the end of 2021. Each firm had to meet the following criteria to be included in our sample:

- 1. Financial institutions and intermediaries (e.g., banks, investment companies, etc.) are excluded.
- 2. They were listed on the TSE at the beginning of 2015.
- 3. They had available data and no fiscal year changes during 2015–2021.

TSE is the largest stock exchange in Iran and a reliable source of information (Daryaei et al. 2022; Nassirzadeh et al. 2022; Shandiz et al. 2022).

There were 380 firms listed on TSE by the end of 2021, but only 149 companies met the above criteria that we selected and analysed.

3.3. Descriptive Statistics

The descriptive statistics of the variables are reported in Tables 1 and 2. According to the data, the dispersion of log-change in sales is 0.32. This value is close to the dispersion of log-change in sales revenue (0.33) and indicates low-cost stickiness in the sample companies. According to Herfindahl's product diversification index (Tang et al. 2022; Zúñiga-Vicente et al. 2019), the findings suggest that manufacturing companies have an average product diversification of 47%. The highest product diversification is 90% (a company with 16 product lines), and some companies have only one product and have not used a diversification strategy. The decrease in the dummy variable for sales shows that 23% of the sample had a reduction in sales, which is close to the result of Zhong et al. (2020). As for the control variables, our results show that the average ratio of assets to sales revenue is more than 1.4, indicating the high asset intensity of the sample companies.

Table 1. Descriptive statistics of the continuous variables.

Variable	Proxy	Mean	Median	SD	Min.	Max.
Log-change in cost of goods sold.	$\Delta \ln COGS$	0.012	0.013	0.323	-3.298	2.096
Log-change in sales revenue	$\Delta ln SALES$	0.025	0.022	0.332	-2.709	1.750
Product diversification	PD	0.471	0.521	0.268	0.000	0.904
Asset intensity	ASINT	1.453	1.211	1.084	0.133	9.746
Employee intensity	EMPINT	0.0003	0.0002	0.0003	0.0001	0.003
GDP growth rate	GDP	0.018	0.018	0.037	-0.023	0.088

Source: authors' own work.

Table 2. Descriptive statistics of the discreet variables.

Variable	Proxy	Categories	Frequency	Relative Frequency
Sales decrease	D	Decrease	242	0.232
Managerial overconfidence 1	OCD1	High OCD Low OCD	542 501	0.519 0.481

Source: authors' own work.

Following Killins et al. (2021), managerial overconfidence is measured using an investment-based index, which is calculated based on the median of capital expenditures. Nearly half of the companies have overconfident managers, similar to Killins et al. (2021) results with the same index. It must be noted that the data are winsorised at the 99% level to reduce the effect of outliers. Also, regression assumptions are tested before estimating the regression models.

4. Results

4.1. Testing the First Hypothesis

The first hypothesis examines the relationship between product diversification strategy and cost stickiness. This hypothesis is tested using panel data after controlling for year and industry-fixed effects. The *p*-value of the F-statistic is less than 0.01, indicating the significance of the model. This research uses ordinary least squares (OLS) with robust standard errors to solve the problem of autocorrelation of residuals and heteroskedasticity (Petersen 2009). Finally, the variance inflation factor (VIF) is calculated to check for multicollinearity. The VIF values obtained are smaller than ten and indicate the absence of multicollinearity between the independent variables. These results are reported in Table 3.

According to these results, the coefficient of $\Delta \ln \text{SALES}(\beta_1)$ is 0.99, which is positive and significant, and the coefficient of D* $\Delta \ln \text{SALES}(\beta_2)$ in the model is -0.21, which is negative and significant. This indicates the presence of cost stickiness as per the model adopted by Banker et al. (2013). The economic implication is that costs increase by 0.99% per 1% increase in sales but decrease by 0.78% (0.99% - 0.21%) per 1% decrease in sales. Moreover, given that the coefficient of D*PD* $\Delta \ln \text{SALES}(\beta_4)$ is negative and significant, the first hypothesis is accepted, indicating that cost stickiness increases with product diversification. Regarding the control variables, the coefficients of D*ASINT* $\Delta \ln \text{SALES}(\beta_6)$ and D*GDP* $\Delta \ln \text{SALES}(\beta_{10})$ are negative and significant. This suggests that firms with higher asset intensity and GDP growth rates have greater cost stickiness.

Table 3. Estimation results for the first hypothesis.

	Beta	t-Statistic	<i>p</i> -Value	VIF		
С	0.06	-2.02	0.045 **	-		
$(\beta_1): \Delta \ln SALES$	0.99	85.56	0.000 ***	1.86		
(β_2) : D* Δ ln SALES	-0.21	-2.66	0.009 ***	2.59		
(β_3) : PD* Δ ln SALES	-0.01	-1.07	0.287	1.29		
(β_4) : D*PD* Δ ln SALES	-0.07	-3.00	0.003 ***	1.35		
(β_5) : ASINT * Δ ln SALES	-0.01	0.20	0.840	9.00		
(β_6) : D*ASINT* Δ ln SALES	-0.01	-4.50	0.000 ***	8.84		
(β_7) : EMPINT* Δ ln SALES	-22.97	-1.36	0.177	2.42		
(β_8) : D*EMPINT* Δ ln SALES	-2.28	-0.73	0.467	7.66		
(β_9) : GDP* Δ ln SALES	-0.03	-2.96	0.004 ***	1.26		
(β_{10}) : D*GDP* Δ ln SALES	-0.23	-3.03	0.003 ***	7.62		
Year		Contr	olled			
Industry		Contr	olled			
Adjusted R ²	0.873					
F-statistic		210	.84			
Sig.		0.0	00			

Source: authors' own work (**: *p* < 0.05; ***: *p* < 0.01).

4.2. Testing the Second Hypothesis

The second hypothesis examines the moderating effect of managerial overconfidence on the relationship between product diversification strategy and cost stickiness. The estimation results for the two models related to this hypothesis (two overconfidence states) are reported in Table 4. The z-test is used to investigate the effect of the moderator variable. This test compares the difference in β_4 between companies with high and low managerial overconfidence. The significant difference in β_4 confirms the moderating effect of managerial overconfidence on the relationship between product diversification and cost stickiness. The z-statistic is obtained using Equation (4):

$$Z = \frac{\hat{\beta}_{4(1)} - \hat{\beta}_{4(2)}}{\sqrt{se(\hat{\beta}_{4(1)})^2 + se(\hat{\beta}_{4(2)})^2}}$$
(4)

where $\hat{\beta}_{4(1)}$ and $\hat{\beta}_{4(2)}$ are the estimated coefficients for high and low overconfidence groups, respectively.

 $se\hat{\beta}_{4(1)}$ and $se\hat{\beta}_{4(2)}$ are the standard errors associated with each coefficient.

The result of the z-test for this measure of managerial overconfidence shows a statistically significant difference in β_4 between companies with high and low managerial overconfidence at the 90% confidence level (p < 0.1). In other words, managerial overconfidence moderates the relationship between product diversification and cost stickiness. This coefficient is more prominent in companies with high managerial overconfidence. Therefore, the second hypothesis is supported.

	Managerial Overconfidence (OCD1)									
		High	OCD			Low OCD				
Variable	Beta	t-Statistic	<i>p</i> -Value	VIF	Beta	t-Statistic	<i>p</i> -Value	VIF		
С	0.001	0.04	0.968	-	0.00	0.00	0.998	-		
$(\beta_1): \Delta \ln SALES$	1.02	8.05	0.000 ***	1.81	0.84	21.87	0.000 ***	1.72		
(β_2) : D* Δ ln SALES	-0.32	-2.51	0.013 **	2.34	-0.16	-1.99	0.049 **	2.12		
(β_3) : PD* Δ ln SALES	-0.14	-0.85	0.398	1.20	0.002	0.37	0.714	1.32		
(β_4) : D*PD* Δ ln SALES	-0.35	-1.89	0.037 **	1.24	-0.04	-2.74	0.007 ***	1.39		
(β_5) : ASINT* Δ ln SALES	-0.02	-0.71	0.481	2.09	-0.00	-0.18	0.856	9.85		
(β_6) : D*ASINT* Δ ln SALES	-0.05	-1.94	0.054 *	1.23	-0.00	-0.47	0.640	5.86		
(β_7) : EMPINT* Δ ln SALES	36.26	0.14	0.885	2.65	-15.80	-0.95	0.346	8.61		
(β_8) : D*EMPINT* Δ ln SALES	72.24	0.29	0.770	2.19	-53.86	-2.82	0.006 ***	9.64		
(β_9) : GDP* Δ ln SALES	0.25	2.24	0.026 **	1.14	-0.02	-2.09	0.038 **	1.33		
(β_{10}) : D*GDP* Δ ln SALES	-0.53	-1.99	0.048 **	1.64	-0.22	-2.12	0.036 **	1.07		
Year		Contr	olled			Contr	olled			
Industry		Contr	olled			Contr	olled			
Adjusted R ²		0.8	68			0.8	58			
F-statistic		919	.28		631.46					
Sig.		0.0	00			0.0	00			
z-statistic				-1	.904					

Table 4. Estimation results for the second hypothesis.

Source: authors' own work (*: *p* < 0.1; **: *p* < 0.05; ***: *p* < 0.01).

4.3. Robustness Tests

4.3.1. An Alternative Measure of Managerial Overconfidence

In addition to the indicator variable proposed by Killins et al. (2021), we use the index proposed by Schrand and Zechman (2012) to measure managerial overconfidence. This index is based on excess investment and is the residual from a regression of total asset growth on sales growth, as shown in Equation (5):

$$ASSETGR_{i,t} = \beta_0 + \beta_1 SALEGR_{i,t} + \varepsilon_{i,t}$$
(5)

This index takes the value of one if $\varepsilon > 0$ (i.e., if assets grow at a higher rate than sales, indicating overinvestment due to managerial overconfidence) and zero otherwise. According to the data, about 43% of companies have overconfident managers, nearly similar to the first index. In this alternative model, the magnitude of the difference in β_4 indicates the significant moderating effect of managerial overconfidence. The results are reported in Table 5.

The result of the z-test for the second measure of managerial overconfidence indicates a significant difference in β_4 between companies with high and low managerial overconfidence at the 90% confidence level (p < 0.1). Again, this coefficient is more prominent in companies with high managerial overconfidence. Therefore, these results support our baseline findings.

			Manage	rial Over	confidence	(OCD2)			
		High	OCD		Low OCD				
Variable	Beta	t-Statistic	<i>p</i> -Value	VIF	Beta	t-Statistic	<i>p</i> -Value	VIF	
С	-0.05	-1.09	0.278	-	-0.03	-1.81	0.073 *	-	
(β_1) : $\Delta \ln SALES$	0.97	45.15	0.000 ***	1.81	0.88	23.34	0.000 ***	2.29	
(β_2) : D* Δ ln SALES	-0.20	-2.21	0.028 **	1.51	-0.10	-1.80	0.074 *	1.14	
(β_3) : PD* Δ ln SALES	0.01	0.71	0.476	1.43	-0.01	-0.71	0.477	1.24	
(β_4) : D*PD* Δ ln SALES	-0.07	-2.90	0.004 ***	1.08	-0.04	-2.38	0.018 **	1.08	
(β_5) : ASINT* Δ ln SALES	-0.00	-0.38	0.705	2.06	-0.00	-1.21	0.229	8.97	
(β_6) : D*ASINT* Δ ln SALES	-0.02	-2.04	0.043 **	1.51	-0.01	-3.59	0.000 ***	7.58	
(β_7) : EMPINT* Δ ln SALES	-38.21	-1.79	0.076 *	1.82	6.54	0.41	0.680	4.55	
(β_8) : D*EMPINT* Δ ln SALES	-131.95	-8.56	0.000 ***	1.06	-26.54	1.95	0.053 *	1.57	
(β_9) : GDP* Δ ln SALES	-0.03	-2.31	0.022 **	1.27	-0.02	-2.99	0.003 ***	1.15	
(β_{10}) : D*GDP* Δ ln SALES	-0.06	-0.30	0.768	1.08	-0.01	-0.54	0.592	1.89	
Year		Contr	olled			Contr	rolled		
Industry		Contr	olled			Contr	rolled		
Adjusted R ²	0.723					0.8	863		
F-statistic		607	.27			658.94			
Sig.		0.0	00			0.0	000		
z-statistic				-1	.788				

Table 5. Estimation results using an alternative measure of managerial overconfidence.

Source: authors' own work (*: *p* < 0.1; **: *p* < 0.05; ***: *p* < 0.01).

4.3.2. An Alternative Measure of Costs

As another sensitivity test, we used SG&A costs instead of the cost of goods sold, as shown in Equation (6). According to the data, log-change dispersion in SG&A costs is 0.37. This value is close to the dispersion of log-change in sales revenue (0.33) and indicates low-cost stickiness in the sample companies. The results are reported in Table 6.

$$\Delta \ln SG \& A_{it} = \beta_0 + \beta_1 \Delta \ln Sales_{it} + \beta_2 D_{it} * \Delta \ln Sales_{it} + \beta_3 PD_{it} * \Delta \ln Sales_{it} + \beta_4 D_{it} * PD_{it} * \Delta \ln Sales_{it} + \beta_5 ASINT_{it} * \Delta \ln Sales_{it} + \beta_6 D_{it} * ASINT_{it} * \Delta \ln Sales_{it} + \beta_7 EMPINT_{it} * \Delta \ln Sales_{it} + \beta_8 D_{it} * EMPINT_{it} * \Delta \ln Sales_{it} + \beta_9 GDP_t * \Delta \ln Sales_{it} + \beta_{10} D_{it} * GDP_t * \Delta \ln Sales_{it} + \beta_{11} \sum T_i Industry + \beta_{12} \sum \theta_i Year + \varepsilon_{it}$$
(6)

According to these results, the coefficient of $\Delta \ln \text{SALES}(\beta_1)$ is 0.92, which is positive and significant, and the coefficient of D* $\Delta \ln \text{SALES}(\beta_2)$ in the model is -0.11, which is negative and significant. Consistent with the baseline findings, these results indicate the presence of cost stickiness. Moreover, given that the coefficient of D*PD* $\Delta \ln \text{SALES}(\beta_4)$ is negative and significant, and the first hypothesis is accepted, this suggests meaningful positive relationship between product diversification and cost stickiness. Similarly, the coefficients of the control variables D*ASINT* $\Delta \ln \text{SALES}(\beta_6)$ and D*GDP* $\Delta \ln \text{SALES}(\beta_{10})$ are negative and significant, suggesting that firms with higher asset intensity and GDP growth rates have greater cost stickiness.

Variable	Beta	t-Statistic	<i>p</i> -Value	VIF		
С	0.35	-4.58	0.000 ***	-		
(β_1) : $\Delta \ln SALES$	0.92	31.60	0.000 ***	1.60		
(β_2) : D* Δ ln SALES	-0.11	-5.39	0.000 ***	7.34		
(β_3) : PD* Δ ln SALES	-0.03	-2.03	0.044 **	1.67		
(β_4) : D*PD* Δ ln SALES	-0.05	-1.78	0.077 *	5.26		
(β_5) : ASINT* Δ ln SALES	0.01	1.01	0.316	9.28		
(β_6) : D*ASINT* Δ ln SALES	-0.01	-1.68	0.066 *	9.75		
(β_7) : EMPINT* Δ ln SALES	-66.32	-3.29	0.001 ***	7.04		
(β_8) : D*EMPINT* Δ ln SALES	34.96	1.40	0.163	9.53		
(β_9) : GDP* Δ ln SALES	-0.07	-3.86	0.000 ***	1.41		
(β_{10}) : D*GDP* Δ ln SALES	-0.14	-2.86	0.005 ***	1.56		
Year		Contr	olled			
Industry		Contr	olled			
Adjusted R ²	0.690					
F-statistic	111.79					
Sig.		0.0	00			

Table 6. Estimation results using an alternative measure of costs (SG&A costs).

Source: authors' own work (*: *p* < 0.1; **: *p* < 0.05; ***: *p* < 0.01).

4.3.3. Incorporating COVID-19 as a Moderator

To address the unique contextual factors stemming from the COVID-19 pandemic, we included a COVID-19 variable as a moderator in our analysis. The results are reported in Table 7. This additional analysis allowed us to explore how the pandemic may have influenced the relationship between our critical variables between 2020 and 2023.

Table 7.	COVID-1	9 and t	he relation	nship	between	product	diversifi	ication	strategy	and	cost sticl	kiness.
									()/			

	COVID 19								
		Pre COVID 19				COVID 19 Period			
Variable	Beta	t-Statistic	<i>p</i> -Value	VIF	Beta	t-Statistic	<i>p</i> -Value	VIF	
С	-0.48	-1.98	0.05 *	-	0.003	0.22	0.83	-	
$(\beta_1): \Delta \ln SALES$	0.88	12.23	0.000 ***	1.92	0.87	25.91	0.000 ***	1.12	
(β_2) : D* Δ ln SALES	-0.25	-2.69	0.008 ***	2.52	-0.05	-1.96	0.052 *	2.82	
(β_3) : PD* Δ ln SALES	-0.01	-1.58	0.117	1.31	-0.001	-0.25	0.802	1.72	
(β_4) : D*PD* Δ ln SALES	-0.03	-1.74	0.084 *	1.1	-0.04	-3.22	0.002 ***	1.91	
(β_5) : ASINT* Δ ln SALES	-0.003	-1.26	0.209	8.23	-0.001	-0.64	0.525	8.13	
(β_6) : D*ASINT* Δ ln SALES	-0.05	-2.32	0.022 **	6.01	-0.02	-2.19	0.03 **	5.81	
(β_7) : EMPINT* $\Delta \ln SALES$	-11.66	-0.31	0.757	8.99	12.11	1.4	0.163	8.01	
(β_8) : D*EMPINT* Δ ln SALES	-51.57	-1.22	0.226	7.28	-3.77	-1.89	0.061 *	8.14	
(β_9) : GDP* Δ ln SALES	-0.02	-2.25	0.026 **	1.38	-0.02	-3.61	0.000 ***	1.18	
(β_{10}) : D*GDP* Δ ln SALES	-0.048	-1.97	0.051 *	1.92	-0.11	-2.26	0.025 **	2.07	
Year		Contr	olled			Contr	olled		
Industry		Contr	olled			Contr	olled		
Adjusted R ²		0.7	'81			0.8	55		
F-statistic		508	3.17			680	.84		
Sig.		0.0	000		0.000				
z-statistic				0	.55				

Source: authors' own work (*: p < 0.1; **: p < 0.05; ***: p < 0.01).

The result of the z-test indicates that there is no significant difference in (β_4) between companies affected by the COVID-19 pandemic and those unaffected by it. Our analysis suggests that the COVID-19 pandemic did not significantly impact the relationship between product diversification and cost stickiness.

5. Discussion

This study significantly enhances the understanding of the relationship between product diversification strategy, cost stickiness, and the moderating role of managerial overconfidence, particularly within the context of the Resource-Based View (RBV) theory. The existing literature on diversification and its implications has often yielded inconclusive results, failing to capture this complex relationship comprehensively. Some studies suggest that diversification enhances firm performance by creating synergies and spreading risk (Krivokapić et al. 2017; Bhatia and Thakur 2018), while others argue that diversification can lead to inefficiencies (Dawid and Reimann 2011). These mixed findings highlight the complexity of the relationship between diversification and firm performance, mainly when cost behaviour is considered. Our study addresses this gap by exploring the interplay between diversification and cost stickiness, offering a new perspective and valuable insights.

Our findings reveal a substantial and previously uncharted link between product diversification strategy and cost stickiness, particularly in manufacturing companies. This connection aligns with the RBV theory, which emphasizes the significance of unique resources and capabilities in achieving a competitive advantage (Barney et al. 2011). In this case, product diversification represents a unique resource, while cost stickiness reflects a capacity related to resource management and risk mitigation. Thus, this study underscores the theoretical underpinnings of RBV by empirically validating its principles in the context of corporate strategies. This finding is consistent with Banker and Byzalov (2014), who suggest that firms with greater resource commitments are more likely to exhibit cost stickiness due to the perceived value of maintaining resources in anticipation of future opportunities.

Moreover, this study uncovers the moderating influence of managerial overconfidence on the relationship between diversification and cost stickiness. This aspect further substantiates RBV, emphasizing the role of managerial decisions and perceptions in resource allocation and risk management. Our findings are consistent with those of Chen et al. (2022) as well as Keke (2021), showing that managers' overconfidence impacts their investment decisions and costs (leading to a more pronounced cost stickiness in companies with product diversification strategies). This empirical validation of managerial overconfidence as a moderating factor reinforces the RBV's emphasis on the role of internal resources and managerial decisions in shaping firm performance.

6. Conclusions

This study aimed to investigate the relationship between product diversification and cost stickiness, focusing on the moderating role of managerial overconfidence within the context of Iranian manufacturing companies. Through our analysis, we found that product diversification is significantly associated with increased cost stickiness, and this relationship is further amplified by managerial overconfidence. These findings suggest that strategic diversification and resource management decisions are critical in shaping a company's cost behaviour, especially under economic uncertainty.

This study's implications are far-reaching. It highlights the critical role of strategic choices in determining cost stickiness, providing valuable insights for shareholders, creditors, and stakeholders in assessing a company's financial behaviour and risk exposure. Additionally, this study offers auditors a deeper understanding of cost behaviour and its determinants, enhancing their analytical procedures.

For analysts and users of financial statements, this study underscores the importance of considering managerial overconfidence and cost stickiness when evaluating a company's financial performance and risk profile. Managers are urged to understand external changes and cost control better to improve their strategic positioning and management efficiency. Lastly, this study raises intriguing questions about the optimal level of product diversification, prompting further exploration in this area.

While this research contributes significantly to the literature, it has some limitations. The reliance on financial statements and the absence of face-to-face interactions with respondents pose constraints. Additionally, this study's focus on Iranian companies may limit its generalizability to other contexts. Iran's economic and regulatory environment might have influenced the findings, meaning that they should be interpreted with care when considering their applicability to different contexts. Furthermore, the data were modified through deflation to adjust for inflation. This may have influenced the results by providing a more precise reflection of real economic values rather than nominal figures. This adjustment could potentially affect the observed relationships, as it eliminates the distorting effects of inflation, allowing for a more accurate analysis. However, this adjustment assumes a uniform impact of inflation across all firms, potentially overlooking sectorspecific differences. Future research can extend these findings by examining different types of product diversification, exploring internationalization strategies, and delving into moderating variables, including other managerial factors, market competition, and intensity of technological innovation, that influence cost stickiness and risk management, thus further enriching the RBV theory's application in diverse contexts.

Author Contributions: Conceptualisation, M.P. and M.M.; methodology, M.P. and M.M.; software, M.P. and M.M.; validation, M.P., M.M., A.R. and D.A.; formal analysis, M.P. and M.M.; investigation, M.P. and M.M.; resources, M.P. and M.M.; data curation, M.P. and M.M. and A.R; writing—original draft preparation, M.P. and M.M.; writing—review and editing, M.P., A.R. and D.A. visualisation, D.A. and M.P.; supervision, M.P. and A.R.; project administration, M.P., M.M., D.A. and A.R. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: This confirms that the current study used publicly available data and, therefore, needed no ethical approval.

Informed Consent Statement: This is to confirm that the current study used publicly available data and, therefore, needed no informed consent.

Data Availability Statement: Companies' data can be purchased via the TSE website at: https: //mabnadp.com/products/rahavard365 (accessed on 21 September 2023). The relevant data are called 'rahavard-novin', are available under the 'products' category, and can be accessed at: https: //mabnadp.com/products/rahavard-novin (accessed on 21 September 2023).

Conflicts of Interest: The authors declare no conflicts of interest.

References

- Aabo, Tom, Nicholai Theodor Hvistendahl, and Jacob Kring. 2021. Corporate risk: CEO overconfidence and incentive compensation. Managerial Finance 47: 244–26. [CrossRef]
- Ahmed, Anwer S., and Scott Duellman. 2013. Managerial overconfidence and accounting conservatism. *Journal of Accounting Research* 51: 1–30. [CrossRef]
- Anderson, Mark C., Rajiv D. Banker, and Surya N. Janakiraman. 2003. Are selling, general, and administrative costs "sticky"? *Journal of Accounting Research* 41: 47–63. [CrossRef]
- Anderson, Shannon W., and William N. Lanen. 2007. Understanding Cost Management: What Can We Learn from the Evidence on 'Sticky Costs'? SSRN 975135. Rochester: SSRN.
- Ansoff, H. Igor. 1957. Strategies for diversification. Harvard Business Review 35: 113-24.
- Arte, Pratik, and Jorma Larimo. 2022. The moderating influence of product diversification on the international diversificationperformance relationship: A meta-analysis. *Journal of Business Research* 139: 1408–23. [CrossRef]
- Askarany, Davood, and Gary Spraakman. 2020. Regional diversification and financial performance through an excess-capacity theory lens: A new explanation for mixed results. *Technological Forecasting and Social Change* 156: 120076. [CrossRef]
- Azman, Mohd Azrai, Carol K.H. Hon, Bo Xia, Boon L. Lee, and Martin Skitmore. 2020. Product diversification and large construction firm productivity: The effect of institutional environments in Malaysia. *Engineering, Construction and Architectural Management* 28: 994–1013. [CrossRef]
- Balakrishnan, Ramji, Michael J. Petersen, and Naomi S. Soderstrom. 2004. Does capacity utilization affect the "stickiness" of cost? Journal of Accounting, Auditing & Finance 19: 283–300.

- Banker, Rajiv D., and Dmitri Byzalov. 2014. Asymmetric cost behavior. Journal of Management Accounting Research 26: 43–79. [CrossRef] Banker, Rajiv D., Dmitri Byzalov, and Lei (Tony) Chen. 2013. Employment protection legislation, adjustment costs and cross-country differences in cost behavior. Journal of Accounting and Economics 55: 111–27. [CrossRef]
- Banker, Rajiv D., Dmitri Byzalov, Shunlan Fang, and Yi Liang. 2018. Cost management research. *Journal of Management Accounting Research* 30: 187–209. [CrossRef]
- Barney, Jay B., David J. Ketchen, Jr., and Mike Wright. 2011. The future of resource-based theory: Revitalization or decline? *Journal of Management* 37: 1299–315. [CrossRef]
- Barney, Jay B., David J. Ketchen, Jr., and Mike Wright. 2021. Bold voices and new opportunities: An expanded research agenda for the resource-based view. *Journal of Management* 47: 1677–83. [CrossRef]
- Becerra, Manuel, Garen Markarian, and Juan Santalo. 2020. The effect of import competition on product diversification revisited. *Strategic Management Journal* 41: 2126–52. [CrossRef]
- Bengtsson, Lars. 2000. Corporate strategy in a small open economy: Reducing product diversification while increasing international diversification. *European Management Journal* 18: 444–53. [CrossRef]
- Bhatia, Aparna, and Anu Thakur. 2018. Corporate diversification and firm performance: An empirical investigation of causality. *International Journal of Organizational Analysis* 26: 202–25. [CrossRef]
- Blavatskyy, Pavlo. 2023. Expected return—Expected loss approach to optimal portfolio investment. *Theory and Decision* 94: 63–81. [CrossRef]
- Borio, Claudio, Mathias Drehmann, and Kostas Tsatsaronis. 2014. Stress-testing macro stress testing: Does it live up to expectations? Journal of Financial Stability 12: 3–15. [CrossRef]
- Bugeja, Martin, Meiting Lu, and Yaowen Shan. 2015. Cost stickiness in Australia: Characteristics and determinants. *Australian Accounting Review* 25: 248–61. [CrossRef]
- Cai, Chun, Qianwen Zheng, and Lei Zhu. 2019. The effect of shared auditors in the supply chain on cost stickiness. *China Journal of Accounting Research* 12: 337–55. [CrossRef]
- Calleja, Kenneth, Michael Steliaros, and Dylan C. Thomas. 2006. A note on cost stickiness: Some international comparisons. *Management Accounting Research* 17: 127–40. [CrossRef]
- Chang, Shao-Chi, and Chi-Feng Wang. 2007. The effect of product diversification strategies on the relationship between international diversification and firm performance. *Journal of World Business* 42: 61–79. [CrossRef]
- Chen, Clara Xiaoling, Hai Lu, and Theodore Sougiannis. 2012. The agency problem, corporate governance, and the asymmetrical behavior of selling, general, and administrative costs. *Contemporary Accounting Research* 29: 252–82. [CrossRef]
- Chen, Clara Xiaoling, Julia Nasev, and Steve Yu-Ching Wu. 2022. CFO overconfidence and cost behavior. *Journal of Management Accounting Research* 34: 117–35. [CrossRef]
- Chen, Yufeng, and Jing Xu. 2023. Digital transformation and firm cost stickiness: Evidence from China. *Finance Research Letters* 52: 103510. [CrossRef]
- Chu, Hsuan-lien, Shuling Yeh, and Tingwen Yang. 2021. Competitive strategy and cost stickiness: The moderating role of managerial overconfidence. *Journal of Accounting Review* 72: 83–117.
- Coad, Alex, and Christina Guenther. 2014. Processes of firm growth and diversification: Theory and evidence. *Small Business Economics* 43: 857–71. [CrossRef]
- Cooper, Robin, and Robert S. Kaplan. 1998. The promise-and peril-of integrated cost systems. *Harvard Business Review* 76: 109–20. [PubMed]
- Costa, Mabel D., and Ahsan Habib. 2021. Trade credit and cost stickiness. Accounting & Finance 61: 1139–79.
- Daryaei, Abbas Ali, Yasin Fattahi, Davood Askarany, Saeed Askary, and Mahdad Mollazamani. 2022. Accounting Comparability, Conservatism, Executive Compensation-Performance, and Information Quality. *Journal of Risk and Financial Management* 15: 489. [CrossRef]
- Dawes, John. 2018. The Ansoff Matrix: A Legendary Tool, But with Two Logical Problems. Rochester: SSRN, February 27.
- Dawid, Herbert, and Marc Reimann. 2011. Diversification: A road to inefficiency in product innovations? *Journal of Evolutionary Economics* 21: 191–229. [CrossRef]
- Denuit, Michel M., Rachel J. Huang, and Larry Y.Tzeng. 2015. Almost expectation and excess dependence notions. *Theory and Decision* 79: 375–401. [CrossRef]
- Ding, Jin, Baumers Martin, Elizabeth A. Clark, and Ricky D. Wildman. 2021. The economics of additive manufacturing: To-wards a general cost model including process failure. *International Journal of Production Economics* 237: 108087. [CrossRef]
- Dinh, Ho Phi, Phuong V. Nguyen, and Jamshid C. Hosseini. 2019. The impact of product diversification and capital structure on firm performance: Evidence from Vietnamese manufacturing enterprises. *Journal for Global Business Advancement* 12: 95–116. [CrossRef]
- Dormady, Noah C., Adam Rose, Alfredo Roa-Henriquez, and C. Blain Morin. 2022. The cost-effectiveness of economic re-silience. International Journal of Production Economics 244: 108371. [CrossRef]
- Friberg, Richard. 2021. All the bottles in one basket? Evaluating the effect of intra-industry diversification on risk. *Long Range Planning* 54: 101973. [CrossRef]
- Galavotti, Ilaria, Donatella Depperu, and Daniele Cerrato. 2017. Acquirer-to-target relatedness and target country unfamiliarity in acquisitions: The role of product diversification and international experience. *Management Decision* 55: 892–914. [CrossRef]

- Garrido-Prada, Pablo, Maria Jesús Delgado-Rodriguez, and Desiderio Romero-Jordán. 2019. Effect of product and geographic diversification on company performance: Evidence during an economic crisis. *European Management Journal* 37: 269–86. [CrossRef]
- Gervais, Simon, James B. Heaton, and Terrance Odean. 2011. Overconfidence, compensation contracts, and capital budgeting. *The Journal of Finance* 66: 1735–77. [CrossRef]
- Grant, Robert M. 2016. Contemporary Strategy Analysis, 9th ed. Hoboken: Wiley.
- Guidolin, Massimo, and Francesca Rinaldi. 2013. Ambiguity in asset pricing and portfolio choice: A review of the literature. *Theory and Decision* 74: 183–217. [CrossRef]
- Gurdgiev, Constantin, and Qiuxin Ni. 2023. Board diversity: Moderating effects of CEO overconfidence on firm financing decisions. Journal of Behavioral and Experimental Finance 37: 100783. [CrossRef]
- Haga, Jesper, Henrik Höglund, and Dennis Sundvik. 2019. Cost behavior around corporate tax rate cuts. *Journal of International Accounting, Auditing and Taxation* 34: 1–11. [CrossRef]
- Hartlieb, Sven, Thomas R. Loy, and Brigitte Eierle. 2020. Does community social capital affect asymmetric cost behaviour? *Management Accounting Research* 46: 100640. [CrossRef]
- Hautz, Julia, Michael Mayer, and Christian Stadler. 2013. Ownership identity and concentration: A study of their joint impact on corporate diversification. *British Journal of Management* 24: 102–26. [CrossRef]
- Hautz, Julia, Michael Mayer, and Christian Stadler. 2014. Macro-Competitive Context and Diversification: The Impact of Macroeconomic Growth and Foreign Competition. *Long Range Planning* 47: 337–52. [CrossRef]
- Hilary, Gilles, and Charles Hsu. 2011. Endogenous overconfidence in managerial forecasts. *Journal of Accounting and Economics* 51: 300–13. [CrossRef]
- Hirshleifer, David, Angie Low, and Siew Hong Siew. 2012. Are Overconfident CEOs Better Innovators? *The Journal of Finance* 67: 1457–98.
- Hitt, Michael A., R. Duane Ireland, and Robert E. Hoskisson. 2020. *Strategic Management: Concepts and Cases: Competitiveness and Globalization*, 13th ed. Boston: Cengage Learning.
- Hribar, Paul, and Holly Yang. 2016. CEO overconfidence and management forecasting. *Contemporary Accounting Research* 33: 204–27. [CrossRef]
- Hur, Kang Sung, Dong Hyun Kim, and Joon Hei Cheung. 2019. Managerial overconfidence and cost behavior of R&D expenditures. *Sustainability* 11: 4878.
- Hussain, Sajjad, Jamshed Khattak, Arshad Rizwan, and M. Adnan Latif. 2013. ANSOFF matrix, environment, and growth-an interactive triangle. *Management and Administrative Sciences Review* 2: 196–206.
- Hyun, Kim, and Choi Seung. 2019. Managerial overconfidence and firm profitability [Sobreconfianza gerencial y rentabilidad de la empresa]. *Asia-Pacific Journal of Accounting & Economics* 26: 1–26.
- Jha, Srivardhini K., Nachiket Bhawe, and P. Satish. 2021. Scaling Social Enterprises through Product Diversification. *Sustainability* 13: 11660. [CrossRef]
- Jin, Xiaolin, and Hai Wu. 2021. Economic policy uncertainty and cost stickiness. *Management Accounting Research* 52: 100750. [CrossRef] Johnson, Dominic D. P., and James H. Fowler. 2011. The evolution of overconfidence. *Nature* 477: 317–20. [CrossRef]
- Kak, Anjana. 2004. Strategic Management, Core Competence and Flexibility: Learning Issues for Select Pharmaceutical Organizations. *Global Journal of Flexible Systems Management* 5: 1–15.
- Katzman, Kenneth. 2014. Iran Sanctions. Washington, DC: Congressional Research Service, p. 84.
- Keke, Pan. 2021. Manager Overconfidence and Cost Stickiness. In E3S Web of Conferences. Les Ulis: EDP Sciences, vol. 235, p. 02020.
- Khurana, Indu, and Joseph Farhat. 2021. The timing of diversification and startup firms' survival: A resource-based perspective. *Industry and Innovation* 28: 1249–69. [CrossRef]
- Killins, Robert, Thanh Ngo, and Hongxia Wang. 2021. Goodwill impairment and CEO overconfidence. *Journal of Behavioral and Experimental Finance* 29: 100459. [CrossRef]
- Krivokapić, Ranka, Vladimir Njegomir, and Dragan Stojić. 2017. Effects of corporate diversification on firm performance: Evidence from the Serbian insurance industry. *Economic Research-Ekonomska istraživanja* 30: 1224–36. [CrossRef]
- Kuang, Yu Flora, Asna Mohan, and Bo Qin. 2015. CEO overconfidence and cost stickiness. Management Control & Accounting 2: 26–32.
- Kunz, Jennifer, and Lara Sonnenholzner. 2023. Managerial overconfidence: Promoter of or obstacle to organizational resilience? *Review* of Managerial Science 17: 67–128. [CrossRef]
- Kuppuswamy, Venkat, and Belén Villalonga. 2016. Does Diversification Create Value in the Presence of External Financing Constraints? Evidence from the 2007–2009 Financial Crisis. *Management Science* 62: 905–23. [CrossRef]
- Lee, Woo-Jong, Jeffrey Pittman, and Walid Saffar. 2020. Political uncertainty and cost stickiness: Evidence from national elections around the world. *Contemporary Accounting Research* 37: 1107–39. [CrossRef]
- Li, Wu-Lung, and Kenneth Zheng. 2017. Product market competition and cost stickiness. *Review of Quantitative Finance and Accounting* 49: 283–313. [CrossRef]
- Li, Ziyang, Qianwei Ying, Yuying Chen, and Xuehui Zhang. 2020. Managerial risk appetite and asymmetry cost behavior: Evidence from China. *Accounting & Finance* 60: 4651–92.
- Lin, Yu, Shuaishuai Zhang, and Yingjie Shi. 2020. The impact of operational stickiness on product quality: Product diversification moderation. *Journal of Manufacturing Technology Management* 32: 423–47. [CrossRef]

- Mammen, Jan, Todd M. Alessandri, and Martin Weiss. 2021. The risk implications of diversification: Integrating the effects of product and geographic diversification. *Long Range Planning* 54: 101942. [CrossRef]
- MohammadRezaei, Fakhroddin, and Norman Mohd-Saleh. 2017. Auditor switching and audit fee discounting: The Iranian experience. *Asian Review of Accounting* 25: 335–60. [CrossRef]
- Nassirzadeh, Farzaneh Nassir, Davood Askarany, and Solmaz Arefi Arefiasl. 2022. Accounting Conservatism and Earnings Quality. Journal of Risk and Financial Management 15: 413.
- Nazarian, Alireza, and Peter Atkinson. 2013. Impact of culture on leadership style: The case of Iranian organizations. *World Applied Sciences Journal* 28: 770–77.
- Nili, Masoud, and Mahdi Rastad. 2007. Addressing the growth failure of the oil economies: The case of Iran. *The Quarterly Review of Economics and Finance* 46: 726–40. [CrossRef]
- Noreen, Eric. 1991. Conditions under which activity-based cost systems provide relevant costs. *Journal of Management Accounting Research* 3: 159–68.
- Noreen, Eric, and Naomi Soderstrom. 1997. The Accuracy of Proportional Cost Models: Evidence from Hospital Service Departments. *Review of Accounting Studies* 2: 89–114. [CrossRef]
- Omosa, Henry, James Muya, Stella Omari, and Charles Momanyi. 2022. Role of product diversification strategy on performance of selected tea factories in Kenya. *International Academic Journal of Innovation, Leadership and Entrepreneurship* 2: 279–96.
- Pamplona, Edgar, Clóvis Fiirst, Thiago Bruno de Jesus, and Vinícius Costa da Silva Zonatto. 2016. Los sticky costs en el comportamiento de los costos de las empresas más grandes de Brasil, Chile y México. *Contaduríay Administración* 61: 682–704. [CrossRef]
- Park, Kwangmin, and SooCheong Shawn Jang. 2013. Capital structure, free cash flow, diversification and firm performance: A holistic analysis. *International Journal of Hospitality Management* 33: 51–63. [CrossRef]
- Pesaran, Evaleila. 2012. Challenges facing the Iranian economy. In *Iran: A Revolutionary Republic in Transition*. Chaillot Paper 128. Paris: EU Institute for Security Studies, pp. 55–57.
- Petersen, Mitchell A. 2009. Estimating standard errors in finance panel data sets: Comparing approaches. *The Review of Financial Studies* 22: 435–80. [CrossRef]
- Porter, Michael E. 1980. Competitive Strategy: Techniques for Analyzing Industries and Competitors. New York: Free Press.
- Rugman, Alan M. 1976. Risk reduction by international diversification. *Journal of International Business Studies* 7: 75–80. [CrossRef] Schrand, Catherine M., and Sarah L. Zechman. 2012. Executive overconfidence and the slippery slope to financial misreporting. *Journal of Accounting and Economics* 53: 311–29. [CrossRef]
- Schumacher, Christian, Steffen Keck, and Wenjie Tang. 2020. Biased interpretation of performance feedback: The role of CEO overconfidence. *Strategic Management Journal* 41: 1139–65. [CrossRef]
- Schwartz, Peter. 1997. Art of the Long View: Planning for the Future in an Uncertain World. Hoboken: John Wiley & Sons.
- Shandiz, Mohsen Tavakoli, Farzaneh Nassir Zadeh, and Davood Askarany. 2022. The Interactive Effect of Ownership Structure on the Relationship between Annual Board Report Readability and Stock Price Crash Risk. *Journal of Risk and Financial Management* 15: 268. [CrossRef]
- Situm, Mario. 2019. Corporate performance and diversification from a resource-based view: A comparison between small and medium-sized Austrian firms. *Journal of Small Business Strategy (Archive Only)* 29: 78–96.
- Smeritschnig, Florian, Jakob Muellner, Phillip C. Nell, and Martin Weiss. 2021. Intra-industry diversification effects under firm-specific contingencies on the demand side. *Long Range Planning* 54: 101992. [CrossRef]
- Sohl, Timo, Brian T. McCann, and Govert Vroom. 2022. Business model diversification: Demand relatedness, entry sequencing, and curvilinearity in the diversification-performance relationship. *Long Range Planning* 55: 102215. [CrossRef]
- Solano, Luis Vinicio Losilla, Bernhard Brümmer, Alejandra Engler, and Verena Otter. 2019. Effects of intra-and inter-regional geographic diversification and product diversification on export performance: Evidence from the Chilean fresh fruit export sector. *Food Policy* 86: 101730. [CrossRef]
- Stadtherr, Frank, and Marc Wouters. 2021. Extending target costing to include targets for R&D costs and produc-tion investments for a modular product portfolio—A case study. *International Journal of Production Economics* 231: 107871.
- Subramaniam, Chandra, and Marcia Weidenmier Watson. 2016. Additional evidence on the sticky behavior of costs. In *Advances in Management Accounting*. Bingley: Emerald Group Publishing Limited, vol. 26, pp. 275–305.
- Tang, Chunhui, Yaojia Tang, and Shengwei Su. 2019. R&D internationalization, product diversification and international performance for emerging market enterprises: An empirical study on Chinese enterprises. *European Management Journal* 37: 529–39. [CrossRef]
- Tang, Liang, Yiyang Huang, Jiali Liu, and Xiangyu Wan. 2022. Cost stickiness and stock price crash risk: Evidence from China. *Emerging Markets Finance and Trade* 58: 544–69. [CrossRef]
- Wang, Shan-Huei, Chung-Jen Chen, Andy Ruey-Shan Guo, and Ya-Hui Ya-Hui. 2020. Strategy, capabilities, and business group performance: The endogenous role of industry diversification [Capabilities and business group performance]. *Management Decision* 58: 76–97. [CrossRef]
- Wang, Yuandi, Lutao Ning, and Jin Chen. 2014. Product diversification through licensing: Empirical evidence from Chinese firms. *European Management Journal* 32: 577–86. [CrossRef]
- Weinstein, Neil D. 1980. Unrealistic optimism about future life events. *Journal of Personality and Social Psychology* 39: 806. [CrossRef] Wooldridge, Jeffrey M. 2010. *Econometric Analysis of Cross Section and Panel Data*. Cambridge: MIT Press.

- Xue, Shuang, and Yun Hong. 2016. Earnings management, corporate governance and expense stickiness. *China Journal of Accounting Research* 9: 41–58. [CrossRef]
- Zhong, Tingyong, Fangcheng Sun, Haiyan Zhou, and Jeoung Yul Lee. 2020. Business Strategy, State-Owned Equity and Cost Stickiness: Evidence from Chinese Firms. *Sustainability* 12: 1850. [CrossRef]
- Zúñiga-Vicente, José Ángel, Diana Benito-Osorio, Luis Ángel Guerras-Martín, and Alberto Colino. 2019. The effects of international diversification on the link between product diversification and performance in a boom and bust cycle: Evidence from Spanish firms (1994–2014). *Journal of International Management* 25: 100687. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.



Article A Financial Stability Model for Iraqi Companies

Narjis Abdlkareem Ibrahim¹, Mahdi Salehi^{1,*}, Hussen Amran Naji Al-Refiay² and Mahmoud Lari Dashtbayaz¹

- ¹ Department of Accounting, Faculty of Economics and Administrative Sciences, Ferdowsi University of Mashhad, Mashhad 9177948974, Iran; m.lari@um.ac.ir (M.L.D.)
- ² Department of Accounting, Administrations & Economics, University of Kerbala, Kerbala 964335027, Iraq

Correspondence: mehdi.salehi@um.ac.ir

Abstract: The current study aims to develop a financial stability model in Iraq; after reviewing the relevant literature and sources related to financial stability and considering Iraq's social, economic, political, and cultural conditions, a conceptual model and a research questionnaire have been developed. Based on the developed conceptual model, macro variables at the level of the economy, micro variables at the level of companies, the environmental variables of companies, and corporate governance have been selected as model dimensions. Each dimension has several components, including several indicators; 39 indicators were measured through questions in 2024. The research questionnaire was subjected to the opinion of 21 experts with sufficient experimental and academic records on this subject, and by using the Analytic Hierarchy Process (AHP) and Technique for Order of Preference by Similarity to Ideal Solution (TOPSIS) methods, the results were analyzed, and the final model was extracted. In this model, the scientific method used to analyze the results determines the weight of each dimension, component, and indicator. The results of this research show that the dimensions of corporate governance, the variables of the company environment, micro variables at the company level, and macro variables at the economic level with coefficients of 0.345, 0.251, 0.236, and 0.168, respectively, have the most significant impact on the ranking of the company's financial stability. So far, research has yet to be conducted to present the financial stability model of Iraqi companies. Therefore, the present research is one of the first studies in this respect, which presents a model both qualitatively (by designing a questionnaire and conceptual model) and quantitatively (through a mathematical model) to measure financial stability that can help the development of science and knowledge in this field.

Keywords: financial stability; political factors

1. Introduction

Unlike price stability, defining or measuring financial stability is not a simple and easy task due to the interdependence and complex interactions of the various financial system elements of companies with each other and the economy, which is more complicated due to the time and cross-border dimensions of such interactions (Kukushina et al. 2021). Financial stability is when the financial system resists economic shocks and can perform its financial tasks correctly (Taranova et al. 2021). The need for financial stability analysis was first recognized since the establishment of stock markets suggesting organized training in the 17th century; after that, by the industrial revolution in the 19th century, credit rating agencies emerged to evaluate the financial stability of organizations to provide a basis for the investors in decision-making processes (Antonov 2022). More recently, in the 20th century, as a result of the massive development of financial markets, metrics such as earnings per share (EPS) and price-to-earnings (P/E) ratios were formulated to assess the financial stability of corporates (Goswami et al. 2014). In addition to the general need for an assessment of the financial stability of companies, the financial crises positioned in the late 1990s, after financial liberalization, have highlighted the importance of financial stability, and its importance was strengthened by the occurrence of financial and economic crises in



Citation: Abdlkareem Ibrahim, Narjis, Mahdi Salehi, Hussen Amran Naji Al-Refiay, and Mahmoud Lari Dashtbayaz. 2024. A Financial Stability Model for Iraqi Companies. *Risks* 12: 140. https://doi.org/ 10.3390/risks12090140

Academic Editors: Eulália Mota Santos and Margarida Freitas Oliveira

Received: 8 July 2024 Revised: 28 August 2024 Accepted: 29 August 2024 Published: 4 September 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). 2007–2008 (Yusgiantoro et al. 2019). Before the crisis, corporate indebtedness had reached unprecedented levels, forcing companies to repay their loans and putting further pressure on their financial stability. High leverage levels increased firms' sensitivities to economic shocks and credit tightening during the crisis (Mayasari and Wulandari 2022). During the crisis, as was evident in the volatility of equity markets worldwide, the systemic nature of risks threatened corporate financial stability. Also, declining commodity prices and the fading of the commodity boom cycle had adverse effects on foreign direct investment in natural resources and changed the fate of businesses operating in the mining and energy sectors. Therefore, the economic slowdown and recession that the crisis triggered reduced the attractiveness of stock markets as an investment destination, reduced foreign investment flows, and reduced companies' growth prospects (Fingleton et al. 2008). High corporate indebtedness, contagious financial markets, unfavorable economic conditions, and regulatory changes finally caused the financial crisis of 2007–2008, which profoundly impacted companies' financial stability worldwide. Therefore, financial stability is essential for economic prosperity at the firm and macro levels.

Indebtedness and debt management are critical for improving financial stability in corporate entities, municipalities, and national economies. The following are the major points provided by the existing literature: (1) Efficient public debt management minimizes the vulnerability of an economy to financial shocks and supports a stable financial environment. For example, in Poland, it follows that effective public debt management reduces financial threats and enhances the stability of economics (Misztal 2021). (2) Public debt management is crucial for the broad economic stability of any country. It touches on public finances, monetary turnover, the investment climate, and international cooperation. For example, in Kazakhstan, the improvement in the system of public debt management plays a significant role in increasing financial stability, especially with regard to economic transformation and growing economic risks (Mataibayeva et al. 2019). (3) In the corporate sector, financial stability is conditioned by the balance between companies' own resources and external debts. Companies with a higher share of their own resources tend to be more stable. Also, excessive reliance on external debt can seriously threaten the firm's financial health if the economy becomes hostile (Chiladze 2018). Therefore, efficient debt management uses long-term debt financing tools to increase the return on equity and improve overall financial stability.

During the last two decades, researchers of central banks and entities have tried to depict financial stability conditions through various indicators of financial system vulnerabilities (Nadri et al. 2018). Several central banks try to assess financial stability risks by focusing on a few critical indicators through Financial Stability Reports (FSR) (Kalsie et al. 2020). In addition, there are ongoing efforts to develop a unit-wide measure indicating business fragility or financial stress (Akosah et al. 2018). The composite quantitative measures of financial stability are intuitively appealing for their ability to, first, present policymakers and participants with a system of more effective monitoring of the level of stability within the system; second, to foresee the sources and causes of financial stress; and third, to understand their implications better. The methodology for constructing these measures has been changing over the years, shifting the focus from micro-prudential to macro-prudential aspects of financial stability. It shifted focus from looking into the early warning indicators to assessing the condition of the banking system, specifically the default risk of individual institutions, to a comprehensive assessment of risks across financial markets, institutions, and infrastructure of the whole system. Recently, increasing analytical focus has been placed on behavioral dynamics, the eventual emergence of unstable conditions, and the shock transmission mechanisms (Stona et al. 2018). In examining financial stability and providing the necessary model to measure it, we must also analyze the banking and financial systems of the companies listed on the Iraqi Stock Exchange. Therefore, in this research, in addition to the listed companies on the Iraqi Stock Exchange, we also consider banks listed on the Iraqi Stock Exchange. The factors affecting financial instability are diverse and can differ according to the period and the countries included in the analysis.

One of the critical issues underlying these analytical developments is the need to fill data gaps in several fields, such as the financial departments of companies listed on the Iraqi Stock Exchange. The current research can help to review the work carried out in the direction of developing quantitative financial stability criteria, presenting the financial stability model for Iraqi companies in both qualitative and quantitative ways: qualitative through a questionnaire and quantitative through a mathematical model used in the published financial reports of business units. Therefore, in this research, we seek to present a model of financial stability in Iraqi companies. The reason for choosing Iraq is because this country has faced several crises in recent years, including the US military attack, the occupation of this country by the ISIS terrorist group, internal riots in the country, and finally, the global crisis of COVID-19. It is known that each of these crises has dramatically impacted the economy and financial performance of businesses and the financial system of Iraq (Abdullah 2018). Also, by reviewing the subject literature, it has been determined that, so far, most of the studies carried out in the field of financial stability have often identified factors affecting financial stability and ranked various factors affecting it. So far, no research has provided a model for stability. Therefore, the current study, considering the importance of financial stability for the financial systems of banks and business units of countries, seeks to investigate this issue in the developing and crisis-stricken country of Iraq to eliminate the research gap in this field. It will also help the development of science and knowledge in this field.

2. Theoretical Principles

Financial stability is a feature of a financial system that eliminates financial imbalances that arise endogenously in financial markets or result from significant adverse and unpredictable events (Icaza 2017). When the system is stable, it absorbs economic shocks, mainly through self-correcting mechanisms, and prevents adverse events from disrupting the real economy or contaminating other financial systems (Elsayed et al. 2023). Financial stability is crucial for economic growth, as most transactions in the real economy are conducted through the financial system (Hadian and Dargahi 2018).

Without financial stability, businesses and banks will be reluctant to finance profitable projects, the price of assets may deviate significantly from their intrinsic values, and the payments settlement program may be derailed (Habib et al. 2020). Therefore, financial stability is essential to maintain the confidence of businesses and financial institutions (Dai and Zhou 2022). The potential consequences of excessive volatility can include financial crises, bankruptcies of business units and banks, severe inflation, and stock market crashes (Akins et al. 2016). In other words, financial stability is the goal of central bank policy for price stability and the healthy development of countries' economies and business units (Mamipour and Godarzi 2020). Because financial instability imposes high costs on the economy and businesses, as the volatility of price variables in financial markets increases, financial institutions or companies may go bankrupt (Enaiati et al. 2022).

Moreover, economic development can be limited at such a time, as economic agents need help making rational decisions and the efficiency of resource allocation decreases. Since the 1980s, several countries have achieved the positive effects of the rapid growth of the financial industry due to the progress of financial liberalization (Dai and Zhou 2022). At the same time, they have also experienced a dramatic slowdown in economic growth due to heavy economic costs caused by financial instability or financial crises (Martinez-Miera and Repullo 2020). In the face of this, countries have greatly emphasized this issue when implementing their policies. Attention to financial stability among all countries is currently increasing, as new factors with the potential to create financial instability, including strengthening financial sector linkages between companies and the uncontrolled development of complex financial instruments, have emerged that make it difficult for businesses to help financial institutions have financial stability against crises (Xie et al. 2022). Therefore, given the importance of financial stability for the economy and the financial system of businesses and banks, it is necessary to provide a model that can correctly depict the various factors affecting financial stability. Especially in Iraq, due to the crises it has gone through in recent years, such a model can be used to assess the financial stability of the financial system of its banks while measuring the financial stability of the listed companies on the Iraqi Stock Exchange.

2.1. Financial Stability

Over the past decade, financial stability has become one of the important goals of economic policy. Many central banks and financial authorities, such as the International Monetary Fund, the World Bank, and the Bank for International Settlements, publish periodic financial stability reports to embed analysis and the pursuit of financial stability into their mainstream activities. Financial stability goes way beyond the absence of crises. A stable financial system is one that, through the conduct of other processes and financial and economic functions such as savings and investments, lending and borrowing, and liquidity creation and distribution, among others, effectively conducts the geographical and temporal allocation of economic resources, evaluating, pricing, and allocating financial risks. It must also manage the risks and have the ability to maintain the performance of those key functions despite adverse external shocks or imbalances.

The system is composed of infrastructures such as legal, payment, settlement, and accounting systems; institutions such as banks, security firms, and institutional investors; and markets for stock, bonds, money, and derivatives. In case of trouble with one component, there is a potential to affect the whole stability of the system itself (Barra and Zotti 2019). However, given that the system is strong enough to hold on to its core functionalities, problems in one component of the system might not necessarily threaten overall stability (Dai and Zhou 2022). The need for financial stability does not pre-assume that all areas within the financial system need to work optimally all the time. It allows for the self-correction of mechanisms to reduce and eliminate these imbalances before they grow to become full-blown crises, such that the national currency performs its roles as a medium of exchange, unit of account, and store of value functions. This, therefore, means that financial and monetary stability are interconnected. According to Barra and Ruggiero (2021), overall, a financial system would be termed stable if disturbances and crises are improbable to hurt economic activity. Increased competition or the assimilation of new information may lead to the closure of a financial institution, growing volatility, or significant market corrections, which indicate good health in the financial system (Bazzazan 2020).

Because the financial system is continuously changing, the concept of financial stability does not pertain to any one stable state or trajectory that the system would revert to post-shock; rather, it alludes to a spectrum or continuum. This multidimensional continuum reveals itself in different observable and measurable variables that are of help in providing for the quantification of the effectuality of a financial system in fulfilling its facilitative roles (Xie et al. 2022). Financial stability is a broad concept and difficult to reduce to be measured by a single metric. The contagion effects and non-linear interactions among the different segments of the financial system cause problems in predicting financial crises. Therefore, a systemic and comprehensive perspective on the stability of the financial system is needed (Ayam 2021).

At the same time, crisis prevention has to be based on a realistic understanding of the degree to which developments in financial stability can be controlled. Many of the policy instruments available for ensuring financial stability pursue other primary objectives, like the protection of depositors in their dealings with banks (prudential policies), price stability (monetary policy), or the efficient settlement of payment transactions (payment policies). The impact of such policy tools on financial stability is most often indirect and occurs after some lag, which may be in contravention of the original purpose of using those tools (Zargarkoche and SoroushRad 2019). Although financial stability is advisable to be identified and quantitatively assessed, in this paper, an approach toward the identification of the most important dimensions and indicators that are relevant for the effect on financial stability is pursued so that this variable can be calculated and forecasted accurately.

2.2. Financial Stability Dimensions and Indicators

Financial stability is multidimensional, so it takes work to measure. Financial stability is measured and evaluated based on the performance of financial markets, institutions, and infrastructures and the interaction between the financial sector and the macroeconomic environment. Therefore, financial stability cannot be analyzed as a single variable or a financial indicator per business, such as macro-level liquidity ratios, with an indicator such as the inflation rate (Dai and Zhou 2022).

Financial market stability refers to the ability of financial markets to absorb a shock and not destabilize their fundamental functions with regard to price discovery and liquidity. Financial market stability is a situation in which companies listed in the financial markets can meet obligations without foreign interference or aid, prices do not change in the short term, and there is a change in fundamental indicators (Ghassan and Krichene 2017). This stability should alleviate the systemic risks to preclude broad contagion or market failure, especially during serious market constraints (Baur and Schulze 2009). Markets in developed countries show more stable characteristics compared to their counterparts in emerging markets, where the impact of systematic shocks can be very high during periods of crisis (Chirilă and Chirilă 2015). Shi et al. (2011) believe that to achieve financial market stability, a sound regulatory framework must be implemented, which compels finance institutions to carry out their business transparently and accountably with the proper practice of risk management. Moreover, monitoring crises can enforce capital adequacy so that the listed institutions might be more resilient under economic shocks. Using advanced analytics, market practitioners can continuously monitor market activities, which is necessary for the early detection of potential risks to prevent systemic risks (Chen et al. 2013). Interventionist monetary and fiscal policies by governments and central banks during times of crisis—for example, the adoption of quantitative easing or the provision of liquidity support—can help stabilize the financial markets. Policymakers may also help stabilize financial markets by proposing clear crisis management protocols so that markets can respond well during sudden shocks. This includes designing contingency plans and emergency liquidity provisions. Therefore, financial market stability is a prerequisite for successful economic development. Financial markets can be stabilized by comprehensive regulatory measures, improved market surveillance, and investor education to ensure sustained economic growth.

In terms of company financial stability, several studies have been conducted, and some indicators of financial stability have been provided. For companies, financial stability refers to the ability to maintain the overall balance in a company's financial structure and to fulfill all financial obligations uninterruptedly over time, whether affected by economic difficulties or market fluctuations. It indicates a company's ability to generate sufficient cash, keep adequate liquidity, manage debt effectively, and ensure profitability and growth. In particular, an optimally balanced capital structure, including internal resources with outside debt, may allow a firm to conduct business efficiently without having too great of a financial burden. Therefore, debt management and efficient capital structure are among the paramount factors determining the financial stability of corporates (Chiladze 2018). Liquidity and profitability, referring to a company's ability to meet short-term obligations and have consistent profitability, are other factors increasing the financial stability of corporates through generating continuous positive cash flows. Other financial stability perspectives address financial risks and market uncertainties by implementing risk management approaches and enhancing adaptability to modify the changing economic conditions (Oleynik and Borisova 2019).

Existing literature also explores some aspects of the financial stability of companies. For example, Allen and Wood (2006) have taken asset price volatility as a measure of financial stability. Financial stability avoids disruptions in the financial system that are likely to cause significant production costs, financial problems in institutions, or disruptions in capital markets. They also defined financial stability as the stability of the value of money, the stability of prices, the favorable level of employment, trust in monetary and financial institutions, and the absence of fluctuations in the liquidity and profitability of companies. Virolainen (2004) introduced macro-level indicators such as interest rates, GDP growth, and credit expansion as companies' financial stability indicators.

However, at the micro level, financial stability criteria are mainly based on liquidity risk, financial solvency, systematic and non-systematic risk, and the Altman Z criterion for predicting financial crises. The review of research (e.g., Beck et al. 2014; Virolainen 2004; Ghassan and Krichene 2017) shows that in the research of Beck et al. (2014), the return on equity basis is used as an indicator of financial stability.

Also, for the purpose of conducting an objective analysis of the value and structure of the company's assets and liabilities and determining the degree of financial stability and independence, comparing the objectives of financial and economic activities with the regulations of the company, the debt and interest coverage ratios are used as criteria. The financial security and trust of the company are leveraged in external financing (Dai and Zhou 2022).

On the other hand, the literature review also suggests that fundamental accounting variables, financial ratios, or a combination of them are generally considered factors affecting financial stability (Pinčák and Bartoš 2015).

Blank (2013) used indicators such as liquidity ratios and debt repayment, such as the current ratio and debt repayment period, to assess financial stability. Thus, if liquidity indicators are greater than normative values, it can be a sign of the improper use of available financial resources. Therefore, indicators of business activity such as asset turnover, inventory, claims receipt period, and working capital can be used to assess financial stability. Because these ratios indicate the management of assets in production and sales and the employment of working and total capital, it is essential in strengthening companies' liquidity.

Table 1 summarizes the research on financial stability and the indicators used to measure financial instability as follows.

Authors	The Index Used for Financial Instability
Altman (1968)	Altman's Z-score index is a measure to predict the probability of financial instability or helplessness based on financial ratios, including working capital to assets, residual profit to assets, profit before interest and tax to assets, the market value of stocks, and sales to assets.
Ohlson (1980)	Score index 9-factor linear combination of coefficients to evaluate the business probability of failure or financial instability. In this regard, the exponential power of this score is divided by 1 plus this exponential power. In this index, the forecasting factors are the sum of assets, the price index in national income, the sum of liabilities, working capital, current liabilities, and current assets; if the sum of liabilities is greater than the sum of assets, it takes 1; otherwise, 0. Net profit, operating cash flows; if the company has made losses in the last 2 years, it takes 1; otherwise, 0.
Zmijewski (1984)	It uses a model based on net profit to assets, total liabilities to assets, and current assets to current liabilities to predict financial instability.
Hillegeist et al. (2004)	They have used famous indicators based on accounting and market variables called BSM-Prob, which indicates the probability of financial helplessness and instability. The variables used include the current market value of assets, the objective value of debt, the expected return on assets, and the expected interest rate of stocks.
Campbell et al. (2008)	To measure financial stability and instability, the following model is developed by the authors: CHS = -20.26NIMTAAVG + 1.42TLMAT - 2.13EXRETAVG + 1.41SIGMA - 0.045RSIZE - 2.13CASHMTA + 0.075MB - 0.58PRICE - 9.1 In this model, NIMTAAVG is the measure of profitability, TLMTA is the measure of financial leverage, EXRETAVG is the average past excess return, SIGMA is the volatility of stock returns, RSIZE is the relative size of the company based on market value, CASHMTA is liquidity and short-term investment. In addition, MB is the market's ratio to the stock's book value and PRICE is the stock price based on decimals. CHS uses quarterly accounting data, lack of success, or instability for reasons related to performance and receiving points from rating agencies. The probability of failure and financial instability for a few years is based on logistic regression estimation.
Sudarsanam and Lai (2001)	The Toffler Z Index consists of components such as profit before tax, current liabilities to current assets, total liabilities to current liabilities, and total assets to the non-credit interval.
Zhang et al. (2018)	Crisis and non-crisis model. This index classifies companies into two categories, crisis and non-crisis, in terms of financial stability.
	Note: The term financial instability represents bankruptcy, financial distress, and financial crisis to increase the

Table 1. Indicators employed for financial instability.

understandability of Table 1.

According to Table 2, by reviewing the literature, the factors affecting companies' financial helplessness or financial instability are categorized into three categories: companylevel characteristics or performance micro variables, macroeconomic-level variables, and corporate governance determining factors. The research conducted on the factors of financial instability (Beaver et al. 2011) in companies is shown as follows:

Authors	Research Question	Sample Studied	Method Used	Findings	Economic Significance
Al-Hadi et al. (2017)	The relationship between CSR and the probability of occurrence of financial instability in the economic cycle	651 Australian companies, years 2007–2013, randomly selected	Altman's Z-index: Lower values indicate less financial instability.	CSR reduces the possibility of financial instability at the maturity stage.	One deviation of the CSR benchmark reduces the probability of financial instability by 46%.
Zhang et al. (2018)	The impact of R&D spending on financial volatility	55652 American companies, years 1980–2011	Linear regression	R&D positively impacts financial volatility, especially in a recession, and intangible assets negatively impact future financial volatility.	One standard deviation of R&D reduces financial volatility by 1.3 percentage points.
Beck et al. (2014)	The relationship between CSR and the likelihood of financial instability	58 Taiwanese companies in 2007–2010	The structural equation model can predict financial volatility.	CSR reduces the possibility of financial instability in the global recession crisis.	-
Tinoco and Wilson (2013)	Accounting variables, capital market, and financial volatility	23218 English companies, years 1980–2011	Binary logistic regression	The full model has the same power to explain volatility changes as Altman.	No
Magee (2013)	Foreign exchange backing and financial instability	401 American companies, 1996–2000	Merton's instability probability (1974)	Foreign exchange backing has a negative impact on financial instability.	One deviation of the foreign exchange backing benchmark reduces the probability of financial instability by 0.785%.
Kane et al. (2005)	Employee relations and the possibility of financial instability	2228 American companies, 1991–2001	Altman's Z-index	Good relationships with employees have a negative impact on financial instability.	No
Tennyson et al. (2008)	Usefulness of financial information and financial instability	Two examples of 23 successful and unsuccessful English companies		The level of relevance of financial information has a negative impact on financial instability.	No

Table 2. Studies based on company-level characteristics (micro variables).

Also, according to Table 3, the factors affecting the occurrence of financial instability based on corporate governance are described as follows:

Table 3. Studies based on the components of corporate governance.

Authors	Research Question	Sample Studied	Method Used	Findings
Gottardo and Moisello (2017)	Family ownership and financial instability	1137 views of Italy in 2004–2013	Binary logistic regression	Family ownership, the number of family members in management, and female CEOs negatively affect financial instability.
Schulze (2022)	Characteristics of the board of directors and financial instability	962 Australian companies, years 2000–2007	Probability of financial instability, Merton (1974)	Management ownership reduces financial instability, and the percentage of non-executive directors increases financial instability.
Darrat et al. (2016)	Characteristics of corporate governance and financial instability	217 successful and 9100 unsuccessful American companies, 1996–2006	Logistic regression	Obligator, personal knowledge and experience, percentage of female managers, length of CEO tenure, CEO succession, reduced financial volatility, and CEO power increases financial volatility.
Hsu and Wu (2023)	Board composition and financial instability	Two samples of 234 English companies in 1997–2000	Binary variable for financial instability and financial stability of the company	A company with more non-obligatory grey managers has more financial stability.
Kallunki and Pyykkö (2013)	Managing Director, Executive Director, and financial instability	48716 Finnish companies, years 2001–2008	Z-adjusted salary or benchmark CEO pay	A CEO or executive with a history of self-interest increases financial instability.
Tinoco and Wilson (2013)	Private ownership and financial instability	1852 acquired companies from 15 European countries in 2008–2000	Altman's Z index and Olson's index	Experienced private acquisition syndication managements have a negative effect on financial instability.

2.3. Research Background

Al-Rjoub (2021) examined financial stability indicators for Jordan. He divides financial health indicators into four categories: 1. have announced capital adequacy; 2. earnings and solvency; 3. liquidity; 4. based on these variables, asset quality constructed by the composite financial stability index with equal weighting using the principal component method from 2003 to 2015. He showed that financial stability in the banking sector in Jordan is consciously resilient to shocks and adverse economic conditions.

Ilesanmi and Tewari (2020) investigated the issue of financial stress indicators and economic activities in South Africa. The financial stress index is a single aggregate index constructed to reflect finance's volatile systemic nature and measure the financial system's vulnerability to internal and external shocks. They have identified the financial stress index for South Africa between 2006 and 2017 using principal component analysis. The results show that the financial stress index is handy for measuring the effectiveness of government measures to reduce the impact of financial stress.

Kalsie et al. (2020) also examined the measurement of financial stability and its impact on foreign direct investment with evidence from BRIC countries. Economic stability has become a key fiscal, economic, and monetary policy objective. They pursued two objectives, one to create four indicators of financial stability and the second to use four indicators of financial stability constructed in a two-stage least squares regression framework to determine the impact of financial stability indicators on foreign direct investment for the period 2000 to 2017 in four countries, Brazil, Russia, India, and China. The results showed that all four indicators of financial stability in Brazil are significant. In the case of Russia, the government's financial management needs to be carried out correctly. In the case of China, the large flow of foreign direct investment could be more effective in attracting foreign capital. The openness of the economy in all countries except India is conducive to foreign direct investment; of the four, Brazil is on the right track.

Ozili (2018) examined the impact of financial instability on sustainable development. The findings of the sustainable development index analysis show that financial stability significantly impacts sustainable development, and its impact is negative in Asian countries. European and Asian countries have a higher sustainable development index than African countries.

Orazalin et al. (2024) investigated whether CSR contributes to the financial sector's financial stability. Their research suggests that CSR initiatives contribute to the financial stability of the financial sector and three subsectors.

2.4. Conceptual Model

After conducting exploratory studies and reviewing the research conducted, the analytical model of the present research, which is the basis for formulating the research question, is as follows, illustrated in Figure 1. Moreover, the conceptual model of the study is displayed in Figure 2.






Figure 2. Conceptual model of the study.

3. Research Method

The philosophy of the current research is that the upcoming research is one of quantitative–qualitative study in the form of interpretations and positivism, and it is a mixture of different approaches such as exploratory, descriptive, quantitative, and qualitative. Therefore, it is also part of the philosophy of pragmatism. In the current research, the study conducts a survey and collects secondary data numerically by using interviews and then preparing a questionnaire. Therefore, the current research is quantitative–qualitative. Also, this research collects data for the variables and prepares essential indicators of financial stability in the world and Iraq by using experts' opinions (including board members, financial managers, university professors, auditors, and other stakeholders) and studying previous research. Therefore, it is descriptive, and in the end, since there has yet to be research on financial stability modeling in Iraq, future research will gain new insight into the calculation of financial stability in Iraq and explain a model in this regard through an exploratory strategy.

3.1. Statistical Sample and Research Period

The statistical population of this research is divided into two parts:

The first part comprises research experts' opinions about the leading stability indicators. The experts must have related occupational experience and knowledge about the current study's topic. For example, they must be board members, financial managers, university professors, auditors, and other stakeholders.

The second part will include all the managers (including the CEO, financial manager, and managers of other departments) of the companies listed on the Iraqi Stock Exchange, which is used to measure the validity and reliability of the questionnaire extracted from the expert interviews at the beginning of 2024.

3.2. Sampling

The philosophical stance of the current study is a mixed-methods approach that includes quantitative and qualitative methodologies. This study applies exploratory and descriptive strategies to understand the financial stability indicators of Iraq. The research might be in the form of surveys and secondary data by interviewing experts, and it could be followed up by developing a structured questionnaire. It aims to obtain the essential indicators of financial stability by collecting insights from board members, financial managers, university professors, auditors, and other related experts from a wide circle of respondents. Since no previous study has been conducted with respect to modeling the financial stability of Iraq, this research may provide new insights for developing a model to assess financial stability in such a context.

Our purposive sampling technique may ensure the selection of at least 100 experts based on a pre-compiled list of academic and professional individuals in the field of finance. The selection criterion is guaranteed to be one that allows only participating respondents who have some form of educational background in the area under study and related practical experience. It is justified on the basis that in statistical advice, a sample size of 100 is sufficient for qualitative studies, given that this number can ensure data saturation to give out a robust analysis of expert opinions.

3.3. Necessary Features for Experts

To ensure the reliability of this study's findings, selected experts have to be experienced and knowledgeable and may provide high-quality and reliable information. Therefore, some criteria shall be developed to identify the participants that best fit this role. Such factors will be in direct relation to the studied topic and the model for financial stability being examined. In the selection of experts to be included in this study, the following factors shall be taken into account. (1) Experts require a relevant educational background in finance, economics, accounting, or management areas. A minimum requirement will be a bachelor's degree in one of the disciplines mentioned above. (2) Experts should have good professional experience with the research subject. This may include a minimum of 5 years of work experience holding a senior financial or managerial position, such as Chief Financial Officer, Financial Controller, Treasurer, etc.; a minimum of 5 years of work experience as a financial analyst/investment banker/portfolio manager working in the financial services industry; at least 5 years of work experience as a financial regulator, policymaker, or central bank official; or at least 5 years of working experience as an academic researcher or professor in finance, economics, or some other related area. (3) Experts must be able to show publications of research articles, books, or reports relating to financial stability, risk management, or corporate finance in reputable journals or other publications. This would serve to demonstrate the individual's expertise and contribution to the field. (4) Professional certification as a Chartered Financial Analyst, Certified Public Accountant, or Certified Treasury Professional will be considered an added qualification to the aforementioned. (5) Experts shall have an absolute reputation and recognition in the Iraqi financial sector. Evidence may be in the form of awards, honors, or leadership positions held within professional associations or industry organizations. Considering these criteria, we may ensure that the collected data in the study are of high quality and represent insights and experiences from people with an in-depth understanding of financial stability within the Iraqi context.

3.4. Data Collection Methods and Instruments

The purpose is to present a model of financial stability in Iraqi companies, identify and explain its constituent factors and elements, and determine the value and importance of each of them in the model of financial stability. To carry out this research, the methods of study and exploratory search in relevant texts are used, and on the other hand, the opinions of academic publishers and the capital market are used.

A questionnaire approach is used to measure the determinants of financial stability in Iraq, as this approach can easily be adaptable to the situation and population of this country to which one applies, thus making it easy to cover most of the drivers of variables relevant to the research topic and demographic group. Moreover, it is administrable in various formats, such as online, face-to-face, or by telephone, making it the most effective instrument for collecting experts' viewpoints. Also, the respondents may provide detailed and qualitative answers, as the questionnaires contain open-ended questions, reaching beyond simple answers to provide further distinctions of ideas, emotions, and experiences. Finally, the employed survey can yield data for a vast number of concepts, from social attitudes/behavior to individual-level health outcomes and environmental factors (Wang et al. 2022).

In particular, several reasons motivated the authors to use a non-parametric approach, while there are several parametric measurements for financial stability proposed by the existing literature. The study entails developing a conceptual model that considers the different dimensions affecting financial stability, such as macroeconomic and microeconomic variables at both the company and environmental levels and corporate governance. Therefore, such an approach gives insight into the realistic scenario of the elements influ-

encing financial stability within the particular case of Iraq. In addition, a panel of 21 experts in related fields of study, validating the questionnaire to meet the objective of the study, may significantly increase the credibility of the data collected through the questionnaire. Thus, the employed methodology provides a robust analysis of financial stability by combining qualitative insights proposed through a questionnaire analysis and quantitative analysis using methods such as AHP and TOPSIS. In essence, this mixed-method approach provides a deeper understanding of the data with accommodation for statistical rigor, anchored in context-appropriate insights. By the employed approach, the weights for the various dimensions and components are determined clearly to bring out which factor most impacts financial stability, which seems impossible to obtain by employing parametric measurements of financial stability. This quantitative aspect is very important in assisting policymakers and practitioners in prioritizing interventions. More importantly, non-parametric methods require a few assumptions to be made about the distribution of the data. This is highly useful when the data may not be normally distributed, usually in the case of finance data exposed to sources of variability. Financial data always contain at least a certain number of outliers—extreme values that make a biased result. Meanwhile, the employed approach in this study is free from outliers and data abnormality, suggesting more robust and accurate insight into the central tendency and relationships within the data. The mixed method used in this study may serve as a necessary complement to any parametric analysis, adding insight or confirming the purpose of the findings. It leads to enhanced robustness of the research outcomes, which helps in producing an exhaustive understanding of the dynamics of financial stability. Finally, this approach is employed because of the special social, economic, and political context that Iraq possesses. The employed method may permit a much more tailored analysis, which catches hold of the generalities of this local environment.

Relevant to the theoretical framework of this investigation, there are several merits of AHP and TOPSIS approaches in the measurement of financial stability as follows. (1) Flexibility and adaptability: AHP and TOPSIS are flexible techniques that can be used in any environment of decision-making. They can deal with qualitative and quantitative criteria; therefore, they will be appropriate for the problem of financial stability, which is complex and involves objective data and subjective expert judgments. (2) Hierarchical structure: AHP decomposes the decision problem into a hierarchy, whereby the relationships among the goal, criteria, and alternatives can be better understood. This structured approach brings clarity to the relative importance of factors impacting financial stability. (3) Consistency check: AHP has a consistency check built into the process for measuring the logical consistency of the pairwise comparisons. This enables the detection of inconsistencies in expert judgments and hence gives an idea about where to correct them for reliable results. (4) Sensitivity analysis: AHP and TOPSIS allow for sensitivity analysis, which means that the former can check variations in the weights of the criteria for the final ranking of the alternative. This will also add weight to the results on the robustness and point out critical factors. (5) Dealing with uncertainty: AHP and TOPSIS are non-parametric methods, hence applicable in situations with uncertainty and ambiguity of choice in decision-making. No assumption about the underlying probability distribution is required; hence, the methods are quite appropriate for analysis with financial data that may not be normally distributed. (6) Ranking interventions: AHP and TOPSIS could, thus, be of great help in giving policymakers at different levels clear weights of the various dimensions impacting financial stability, thereby allowing them to prioritize their interventions. Quantification is thus hugely important in optimizing resource allocation and targeting only key factors with the most effect. These advantages are further enhanced by the combination of the two methods: that of eliciting qualitative insight and that of quantitative analysis, making AHP and TOPSIS very strong tools for the attainment of a holistic understanding of financial stability in the context of Iraq. Collectively, there are several advantages proposed by the mixed method employed in this study.

The employed questionnaire is disclosed in Appendix A. There are several reasons to employ this questionnaire, including (1) comprehensive coverage: the survey covers a wide range of drivers in promoting financial stability. The drivers are at the firm level, corporate governance level, macroeconomic, and environmental levels. In general, the presented questions may cover most of the potentially critical views of the topic. (2) Clear and concise questioning: the majority of the questions are solidly clear and concise; thus, they are easily understood by the respondents, which improves the credibility of the findings. (3) Using the Likert scale: the data responses are on a 5-point Likert scale, hence allowing the data to be analyzed numerically for computation and comparison; it also allows for statistical analysis and credible comparison. (4) Structured format: the contained questions are structured and separate the different categories of factors, making them easier to read and suiting the experience of the respondents. (5) Focusing on key areas: this questionnaire is guided by key thematic areas such as corporate governance, macroeconomic conditions, and firmspecific and environmental factors that play critical roles in ascertaining the overall financial health of a firm. The controlled thematic areas provide a thorough foundation on which valuable cues can be picked regarding what drives financial stability. This questionnaire may enhance the credibility and applicability of our results significantly.

4. Data Analysis

The study aims to examine the financial stability ratings in the capital market, identify and describe its constituent factors and elements, and determine their value and importance in the company's financial stability rating model. To conduct this research, in the first stage, an exploratory study and search are conducted in the relevant literature, and on the other hand, the opinions of academic and capital market experts are used. The theoretical framework of the topic and relevant sources will also be determined, and by studying English and Arabic articles, research questions are designed to formulate the model.

In the next stage, after designing the initial framework of the research to validate it, we identify the model's dimensions and determine whether the predicted financial stability model and its elements and constituent factors are compatible with the situation and reality of the Iraqi capital market. For this purpose, based on Table 4, financial stability indicators were prepared and compiled as a questionnaire with 39 questions. In other words, an initial questionnaire was provided by preparing an initial list of 100 academic and professional experts and examining scientific and experimental records regarding the research topic. To ensure the research selection and final questionnaire met rigorous standards, the work was reviewed by a panel of 21 highly qualified researchers with expertise in the field. Table 4 shows the descriptive statistics of the respondents by education, service history, gender, position in the department, level of education, and field of study.

According to the final goal of the research, which is to provide a financial stability model for Iraqi companies, experts' opinions should be analyzed to determine the statistical significance of the coefficient of each model's dimensions, components, and indicators. The hierarchical analysis process method will be used to analyze dimensions and components, and the TOPSIS method will be used to analyze indicators. The reason for using the TOPSIS method in the analysis of indicators is that the number of indicators exceeds the limit of other applicable methods, such as AHP. The AHP is a structured technique to organize and analyze a complex decision based on qualitative and quantitative factors. It involves breaking down a decision into a hierarchy of objectives, criteria, and alternatives. The TOPSIS approach, as a statistical technique, is used to provide the order of preference by similarity to the ideal solution in our multi-criteria decision analyses, which considers evaluation and ranking against multiple criteria. We employed the TOPSIS due to its simplicity and rationality. TOPSIS gives an apparent and rational view of how to make decisions; it is easy to understand and implement, increasing our findings' credibility. It suggests a quantifiable process featured with flexibility for various kinds of data as necessary for our settings. It can accommodate a high number of criteria and alternatives, so it is suitable for the analysis of the complicated situation of this study. TOPSIS also

provides a simple computational process that can easily be automated by the authors, thus enabling efficient analysis of the data sets used for this study. Therefore, the authors have used TOPSIS to ensure the credibility of the findings. In addition, employed method for determining model's coefficient is displayed in Figure 3.

Variable	Respondent Opinion	Frequency	Percentage
C 1	Male	19	90%
Gender	Female	2	10%
	Less than 40	11	53%
1 00	40-45	4	19%
Age	45-50	2	09%
	50–55	4	19%
	University professor	2	10%
	Accountants	5	24%
Position	Auditor	4	19%
	Financial manager	6	28%
	Other	4	19%
	Less than 10	11	52%
	10–15	3	14%
Work experience	15–20	1	06%
-	20–25	3	14%
	25 and up	3	14%
	Bachelor's degree	12	57%
Education	Master's degree	4	19%
	PhD	5	24%
	Economics	6	28%
Field of study	Accounting	11	53%
-	Management	4	19%

 Table 4. Descriptive statistics of respondents to the research questionnaire.



Figure 3. Methods used to determine model coefficients.

4.1. The Results

The AHP provides a framework for thinking about complex issues. This process helps us to make appropriate and correct decisions for complex issues by simplifying the decision-making process. AHP is a method in which a complex situation is decomposed into smaller parts, and these parts are in a hierarchical structure. In this method, according to the importance of each variable and the numerical values assigned to it, the most critical variables are identified through mental judgments. In other words, the priority order of the variables is determined.

AHP creates an effective structure for group decision-making by streamlining the group thinking process. Assigning numerical values to the variables helps the decision-makers have the appropriate thinking pattern to reach the result. Also, the nature of consensus in group decision-making (its consultative nature) improves the consistency of judgments and increases the reliability of AHP as a decision-making instrument. In this way, with the help of AHP, very complex problems that include many factors can be understood and simplified. Table 5 shows the coefficients of each dimension and component obtained using AHP.

Table 5. Dimension coefficients and model components.

					Dimer	nsions					
Macro '	Variable at Econom 0.168	ny Level (C1)	Micro V	Micro Variable at Company Level (C2) 0.236		Company Level Variables (C3) 0.251			Corporate Governance Variables (C4) 0.345		
Code	Title	Coefficient	Code	Title	Coefficient	Code	Title	Coefficient	Code	Title	Coefficient
C11	Characteristics of the economy	0.304	C21	Performance	0.201	C31	Company's characteristics	0.401	C41	Shareholders' rights	0.225
C12	Instability	0.265	C22	Financial structure	0.331	C32	Variables related to profitability	0.590	C42	Transparency	0.312
C13	Political factors	0.431	C23	Resource management	0.163		. ,		C43	Effectiveness of the board of directors	0.129
			C24	Liquidity	0.305				C44	Effects of ownership	0.334

As is known, the dimension of corporate governance with a coefficient of 0.345 significantly impacts corporate governance. According to prior investigations, improved corporate governance may lead to financial stability for several reasons. In this regard, Susanto and Walyoto (2023) indicate that enhanced corporate governance may assist companies in improving their stability through risk management. Effective corporate governance structures, through the board of commissioners and audit committees, enhance a company's ability to identify, assess, and mitigate financial risks. This ensures companies' resilience in facing emerging financial shocks. Moreover, independent decision-making can be guaranteed through effective corporate governance mechanisms, which include frequent board meetings and a balanced board structure. Consequently, a strong corporate governance structure can support the organization's adaptation to changing markets, including insurance innovations to maintain financial stability and business dynamism (Nwogugu 2015). It helps a company avoid suboptimal financial decisions that may endanger its financial stability. Effective corporate governance can also lead to more transparency in financial reporting and operations (Mabvira 2018). This is important in discovering or averting financial misreporting, critical to maintaining financial stability and investor trust in the market.

The dimensions of company environment variables, micro variables at the company level, and macro variables at the economic level are also critical, with coefficients of 0.251, 0.236, and 0.168, respectively. According to the results, the financial stability score (FSS) can be obtained at the dimension level through model (1):

Model (1):

FSS = 0.168 C1 + 0.236 C2 + 0.251 C3 + 0.345 C4

Also, according to the obtained coefficients for the components, the related equations at the level of the components are described as follows:

Model (2):

C1 = 0.304 C11 + 0.265 C12 + 0.431 C13

Model (3):

C2 = 0.201 C21 + 0.331 C22 + 0.163 C23 + 0.305 C24

Model (4):

C3 = 0.410 C31 + 0.590 C32

Model (5):

$$C4 = 0.225 C41 + 0.312 C42 + 0.129 C43 + 0.334 C44$$

By combining model (1) with models (2), (3), (4), and (5), the financial stability score equation can be obtained based on the components as follows:

Model (6):

$$\begin{array}{rl} \text{FC} = 0.051 \ \text{C11} & +0.044 \ \text{C12} + 0.072 \ \text{C13} + 0.047 \ \text{C21} + 0.078 \ \text{C22} + 0.038 \ \text{C23} \\ & +0.072 \ \text{C24} + 0.102 \ \text{C31} + 0.148 \ \text{C32} + 0.069 \ \text{C41} + 0.107 \ \text{C42} \\ & +0.044 \ \text{C43} + 0.115 \ \text{C44} \end{array}$$

4.2. Prioritization of Indicators by TOPSIS Method

TOPSIS is one of the multi-criteria decision-making methods based on the distance size based on the Euclidean model for the negative and positive ideal solutions. The research conceptual model shows that each component includes one or more indicators. Due to the high number of indicators, the TOPSIS has prioritized the indicators. After performing the various steps of this method, the priority of the indicators regarding each of the components is in Table 6.

As seen in the above table, the value of two of the indices' coefficients for each column is equal to one, indicating that these two components consist of only one variable, respectively, sanctions and the industry dummy variable, instead of a composite of multiple indicators. The coefficient of each index in the final model can be calculated through the following relationship:

$$W_{\rm m} = W_{\rm d} + W_{\rm c} + W_{\rm i}$$

 W_m is the result of the index coefficient, W_i is the index coefficient in the component, W_c is the component coefficient, and W_d is the dimension coefficient. The values of W_c and W_d are reported in Table 6. For example, the index coefficient of C441 in the final model is calculated as follows:

$$W_{\rm m} = 0.345 * 0.334 * 0.467 = 0.0538$$

The sum of index coefficients in the final model for all indices (39) will equal one. By comparing the coefficients of the indicators in the final model, their importance in assessing each company's financial stability is determined. For example, the index has the highest coefficient among the 39 indicators of the model. As mentioned in the last part of the introduction, none of the countries, the rating institutes, or the research conducted have mentioned the coefficients of dimensions, components, and indicators have been mentioned. Therefore, comparing the results with those of previous studies is impossible. However, the cultural, political, economic, and social differences between the countries make it impossible to use them in other countries.

Index Code	Definition	Index Coefficient in Component	Index Coefficient in the Final Model
	The priority of component indicators	of macro variables at the econom	nic level
0111	The priority of component ind	icators of economic characteristic	s
CIII	Oil price	0.428	0.021
CII2	Inflation The availability of evolution	0.572	0.029
C121	Ine priority of volatil	ity component indicators	0.014
C121 C121	ISIS Changes in exchange rates	0.331	0.014
CIZI	Economic and political	0.273	0.012
C123	instability	0.369	0.016
C131	Sanctions	1	0.07241
0151	According to the definition of only one in	dex for this component its coeffi	cient is one
	The priority of micro variable compo	nent indicators at the level of con	nanies
	The priority of liquid	ity component indicators	ipulieo
C211	Debt repayment period	0.278	0.013
C212	Quick ratio	0.457	0.021
C213	Current ratio	0.265	0.012
	The priority of resource mar	agement component indicators	
C221	Periodicals Collection	0.196	0.015
C222	Working capital	0.350	0.027
C223	Turnover of assets	0.258	0.020
C224	Turnover of goods	0.196	0.015
	The priority of financial st	ructure component indicators	0.000
C231		0.225	0.008
C232		0.312	0.012
C233	The priority of perform	0.405	0.017
C241	Return on investment		0.008
C241	Earnings por share	0.237	0.008
C243	Return on capital	0.145	0.010
C244	Net profit margin	0.140	0.010
C245	Fluctuation of profitability	0.215	0.015
C246	Stock beta	0.168	0.012
	The priority of the component indicators of	f the variables of the company's	environment
	The priority of the component indi	cators of the company's character	ristics
C11	Industry dummy variable	1	0.10291
	According to the definition of only one ir	dex for this component, its coeffi	cient is one
	The priority of profitab	ility component indicators	
C321	Growth opportunities	0.362	0.053
C322	Return on equity	0.638	0.094
	Prioritization of component indica	tors of corporate governance vari	ables
	Priority of shareholder r	ights component indicators	
C411	Announcement of dividend	0.183	0.014
C412	policy Drouiding financial reports	0.208	0.022
C412	Fibility management and	0.298	0.023
C413	porformance forecasting	0.193	0.014
	Presenting the report of the		
C414	hoard in mosting	0.326	0.025
	The priority of transpar	ency component indicators	
	Disclosure of remuneration of	ency component indicators	
C421	board members	0.421	0.045
C422	Disclosure of contract details	0.351	0.037
6422	Disclosure of shares of board	0.000	0.024
C423	members	0.228	0.024
	The priority of indicators of the e	ffectiveness of corporate governa	nce
C421	Independence of the board of	0.250	0.011
C431	directors	0.250	0.011
C122	Financial expertise of the	0.210	0.014
C432	board of directors	0.319	0.014
C122	Expertise in the board	0.102	0.004
C455	industry	0.103	0.004
C434	The duality of management	0.119	0.005
C/135	The existence of an audit	0 209	0 009
C400	committee	0.207	0.002
_	The priority of ownership	affects component indicators	
C441	Institutional ownership	0.467	0.053
C442	Property management	0.533	0.061

Table 6. Coefficients of each index in the model.

5. Discussion

The current study, considering the importance of the financial stability of the financial systems of the banks and business units of emerging countries such as Iraq, seeks to highlight critical aspects affecting the financial stability of companies and banks.

Generally, the statistical significance of the coefficient of the dimensions in the final model is, respectively, 0.342 for the corporate governance dimension, 0.312 for the corporate variables dimension at the micro level, 0.251 for the corporate environment variables, and 0.166 for the macro variable at the economic level, which indicates the increasing impact of corporate governance in developing the financial stability of Iraqi companies.

Primarily, the findings suggest that enhanced corporate governance may assist companies in improving their stability through risk management (Susanto and Walyoto 2023). Effective corporate governance structures, through the board of commissioners and audit committees, enhance a company's ability to identify, assess, and mitigate financial risks. This ensures companies' resilience in facing emerging financial shocks. Moreover, independent decision-making can be guaranteed through effective corporate governance mechanisms, which include frequent board meetings and a balanced board structure. Consequently, a strong corporate governance structure can support the organization's adaptation to changing markets, including innovations in the insurance market. For example, insurance maintains financial stability and business dynamism (Nwogugu 2015). It helps a company to preclude from suboptimal financial decisions that may endanger its financial stability. Effective corporate governance can also lead to more transparency in financial reporting and operations (Mabvira 2018). Additively, related to corporate governance, ownership structure effects have the highest coefficient of importance, with a coefficient of 0.334. It refers to the structure of a firm's distribution of ownership rights among all stakeholders, including individual shareholders, institutional investors, governmental owners, and management. The ownership structure can exert a great influence on the financial stability of a firm. Moudud-Ul-Hug et al. (2022) indicate that ownership structures can affect the financial stability of a company through efficient risk management. They document that Islamic banks are less risky than other forms of ownership, adding to their financial stability. They also suggest that ownership structure can affect a firm's market power and competitive position. Firms with solid ownership structures may be better positioned to cope with competitive pressures, thus enhancing their financial stability. Rubio-Misas (2020) indicates that companies with a more concentrated ownership structure tend to show lower levels of financial stability, consistent with the view that in closely held firms, the owner and manager incentives are more aligned. Therefore, different ownership structures may increase or decrease companies' risk-taking ability and risk-management strategies, eventually determining their stability.

Similarly, among the components related to macro variables at the economic level, political factors have the highest coefficient of importance, with a coefficient of 0.431. In essence, a predictable business environment requires political stability. Political instability, which includes violence and unrest, is likely to affect business operations, chase away investment, and increase operational risks (Zaiane and Moussa 2021). For instance, the political instability resulting from the Arab Spring impacted the performance of banks in the MENA region, an aspect that depicts the deteriorating impact of political turmoil on financial stability.

Additionally, among the components of micro variables at the company level, the financial structure with a coefficient of 0.331 has the highest coefficient of importance. The financial structure is vital to ensuring the financial stability of a company. The financial structure of a given company determines its flexibility toward change in the market. High debt levels limit a firm's flexibility during economic downturns, making them very vulnerable to financial distress (Abbas et al. 2021). For instance, the case of the Asia Cell Communications Company has highlighted that overdependency on debt adversely affects financial flexibility, which restrains a firm's ability to pay off short-term liabilities and, consequently, its financial stability. Moreover, the financial structure directly affects

the liquidity position of a firm. A firm with a much more balanced financial structure including sufficient cash and manageable debt can better cope with unexpected expenses or economic downturns (Aslam et al. 2016). Low liquidity can trigger inadequate cash flow, threatening a firm's financial stability. Finally, the financial structure indicates how a company efficiently runs its assets. A company with a reasonable balance between debt and equity typically has better asset and investment management skills (Eisdorfer et al. 2013). Therefore, effective asset and investment management can contribute to financial stability by ensuring that a company generates sufficient returns on assets to compensate for its liabilities.

Finally, among the components of the variables of the company's environment, variables related to profitability with a coefficient of 0.590 have the highest statistical significance. One major aspect that can have a telling effect on the financial stability of any firm is profitability. It directly determines the ability of a firm to generate cash flows and manage its liquidity. Higher profitability secures more funds for meeting short-term obligations, ensuring financial stability. Studies show that reduced liquidity due to low profitability can lead to a cash flow crisis, thus threatening the firm's financial stability (Abbas et al. 2021). Furthermore, profitability determines a firm's ability to service its interest and other debt-related obligations. High profitability ensures that firms can service interest and principal payments on time and are less likely to default on loans, thus ensuring financial stability (Aslam et al. 2016). In other words, reduced profitability will limit the ability of a company to service its debt, which is likely to increase financial risks. Finally, profitability may assist companies in improving their financial stability by providing a competitive advantage. The more profits a company obtains, the more it can invest in innovation, technology, and marketing, hence maintaining a market and financial position (Halim et al. 2023).

Model (6) shows the financial stability score equation based on research components. In this equation, C32 (profitability variables) has a coefficient of 0.148, C44 (ownership effects) has a coefficient of 0.115, C42 (Transparency) has a coefficient of 0.107, and C31 (company characteristics) has a coefficient of 0.102, with the most significant impact in the final model. Table 7 presents the research components in the order of statistical significance of their coefficients in the final model.

No.	Code	Title	Coefficient in the Final Model	No.	Code	Title	Coefficient in the Final Model
1	C32	Variables related to profitability	0.148	8	C41	Shareholders' rights	0.069
2	C44	Effects of ownership	0.115	9	C11	Characteristics of the economy	0.051
3	C42	Transparency	0.107	10	C21	Function	0.047
4	C31	Features of the company	0.102	11	C12	Instability	0.044
5	C22	Financial structure	0.078	12	C43	Effectiveness of the board of directors	0.044
6	C24	Liquidity	0.072	13	C23	Resource management	0.038
7	C13	Political factors	0.072			C C	

Table 7. Index coefficients in the final model.

At the index level, in the final model, the index coefficients are presented in Table 7. Dummy variable indicators of industry, return on equity, sanctions, managerial ownership, institutional ownership, growth opportunities, disclosure of board members' remuneration, contract details, inflation, and working capital have the highest coefficients in the final model. Table 8 shows the indicators with higher importance coefficients (top 10 indicators).

Code	Index Title	Coefficient in the Final Model
C311	Industry dummy variable	0.102
C322	Return on equity	0.094
C131	Sanctions	0.072
C442	Managerial ownership	0.061
C441	Institutional ownership	0.053
C321	Growth opportunities	0.053
C421	Disclosure of remuneration of board members	0.045
C422	Disclosure of contract details	0.037
C112	Inflation	0.029
C222	Working capital	0.027

Table 8. Top ten research indicators in terms of importance factor in the final model.

6. Implications

The results suggest that Iraqi companies should pay enough attention to the company environment, corporate governance, and macroeconomic-level variables because these factors can affect their financial stability. Correspondingly, the following implications are proposed to companies, management, policymakers, and stockholders.

Based on the results of corporate governance enhancements, companies are required to enhance corporate governance structures to support improved risk management that is resilient against financial shocks and instability. Companies are supposed to implement proper mechanisms of governance, where effective board meetings are needed, with balanced board structures that guarantee independent decision-making. Increasing transparency in financial reporting and operations will reduce financial misreporting and sustain investor trust, which is necessary for risk management. Moreover, the ownership structure of firms should be considered and shaped very carefully since it influences their positioning with respect to financial stability via efficient risk management and competitive positioning.

To achieve political stability, which is necessary for a predictable business environment, governments must reduce political instability since its related shocks are among the biggest threats to organizational success. Companies must be prepared to handle the risks associated with political instability, positively impacting business operations and investments. According to the findings on the capital structure, firms should manage their level of indebtedness to be in a better position to deal with an economic downturn by reducing their vulnerability to financial distress. Those companies possessing a balanced financial structure are more resilient to financial shocks. For example, sufficient cash to keep the business running and bearable levels of debt that ensure sustainability during irregular periods or economic downturns are among the effective factors ensuring superior performance. The efficient management of assets and investments may also help companies keep the apt returns that will cover all liabilities.

Finally, the companies notice that high profitability is vital for generating cash flows and liquidity management, and securing sufficient funds is critical to meet short-term obligations. Therefore, a firm's profitability determines its ability to service interest and debt obligations, reducing the risk of default. Accordingly, it is recommended that profits should be reinvested in innovation, technology, and marketing to maintain a competitive and financial position in the market.

7. Conclusions

To conclude, the research provides a general model supporting the relationship between corporate governance, financial structure, political stability, and profitability in achieving company financial stability in Iraq. These findings have positive insightful implications for company management, policymakers, and investors, who have to worry about these dimensions to create a more stable and resilient financial environment in Iraq. By understanding these critical variables, the companies in Iraq may also achieve their financial stability, which contributes to the economic development of the nation. Although this research depicted the financial stability model for the companies in Iraq, there are a few limitations, and further investigations can be performed based on our results about the complex nature of financial stability.

While the results might well apply to other similar emerging economies, they cannot be generalized to other cultural or economic settings. Indeed, the study is based on a relatively modest number of companies and experts, and while more widespread, diversified, and larger, the sample is the stronger foundation for establishing more decisive general conclusions. The research method used was cross-sectional, collecting data at one point in time. This methodology imposes limitations to find out changes in the status of financial stability over the long run or to establish a causal relationship between the variables.

Future research can adopt a longitudinal research design to trace the changes in the level of financial stability. This may help in understanding the dynamics of financial stability and the impact of various factors in different economic cycles during a long period. Comparative studies across different emerging markets might help to explain the way contexts impact financial stability. This can help identify best practices that can be transferable across different settings. Researchers can step ahead and investigate further dimensions or variables that are likely to affect financial stability; this includes technological changes, market competition, or regulatory changes.

Author Contributions: Conceptualization, N.A.I. and M.S.; methodology M.S.; software, H.A.N.A.-R.; validation, M.S. and H.A.N.A.-R.; formal analysis, M.L.D.; investigation, N.A.I.; resources, M.L.D.; data curation, N.A.I.; writing—original draft preparation, M.L.D.; writing—review and editing, H.A.N.A.-R.; visualization, N.A.I.; supervision, M.S.; project administration, M.L.D.; funding acquisition, M.L.D. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Data Availability Statement: Data will be available at request from the corresponding author.

Conflicts of Interest: The authors declare no conflicts of interest.

	No.	Questionnaire Items	Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree
	Fina	ncial Stability					
	1						
	2	The Z-score serves as a suitable indicator of financial stability.					
	3	A model based on net profit margin, total debt-to-assets ratio, and current ratio can predict financial instability.					
Firm-level factors	4	Social responsibility is related to financial stability.					
5 Research and development expense stability.		Research and development expense is a significant factor in financial stability.					
	6	Accounting variables play a major role in companies' financial stability.					
	7	Employee relations affect financial stability.					
	8	Foreign exchange backing of companies affects financial stability.					
	9	Family ownership affect financial stability					
	10	Family ownership in companies affects their financial stability.					
	11	The characteristics of the board of directors play a significant role in companies' financial stability.					
	12	Corporate governance characteristics affect financial stability.					
Corporate governance	13	The independence of the board of directors plays a very important role in companies' financial stability.					
lactors	14	The size of the board of directors is one of the determining factors of a company's financial stability.					
	15	The composition of the board of directors affects companies' financial stability.					
	16	The CEO and executive managers play a determining role in the financial stability of companies.					

Appendix A. Employed Questionnaires to Assess the Financial Stat
--

	No.	Questionnaire Items	Strongly Disagree	Disagree	Neutral	Agree	Strongly Agree
	Fina	ncial Stability					
	17	Private ownership can lead to improved financial stability in companies.					
	18	The disclosure of contract details affects the financial stability of companies.					
	19	The disclosure of board members' compensation affects companies' financial stability.					
	20	Managerial ownership affects the financial stability of companies.					
	21	Management duality affects the financial stability of companies.					
	22	The industry in which companies operate affects their financial stability.					
	23	Growth opportunities in the industry affect companies' financial stability.					
	24	The price of oil affects the financial stability of companies.					
	25	Inflation can affect the financial stability of companies.					
Corporate	26	Economic and political sanctions can affect companies' financial stability.					
factors	27	Changes in exchange rates play a determining role in financial stability.					
	28	Attacks by the ISIS terrorist group and the occupation of the country by this group have affected the financial stability of companies.					
	29	The quick ratio and current ratio have affected the financial stability of companies.					
	30	The debt repayment period has affected companies' financial stability.					
	31	Growth opportunities have affected the financial stability of companies.					
	32	Return on equity has affected the financial stability of companies.					
	33	Accounts receivable turnover has affected the financial stability of companies.					
	34	Working capital turnover has affected the financial stability of companies.					
	35	The debt-to-equity ratio has affected the financial stability of companies.					
	36	Interest coverage has affected the financial stability of companies.					

'hat are the most important corporate governance indicators that affect corporate financial stability?

What do you consider to be the most important macroeconomic indicators that affect corporate financial stability?

What are the most important environmental indicators that affect companies' financial stability?

References

- Abbas, Ali Abdulhassan, Ayat Amer Hadi, and Amna Abbas Muhammad. 2021. Measuring the Extent of Liquidity's Impact on the Financial Structure. International Journal of Multicultural and Multireligious Understanding 86: 365–89.
- Abdullah, Ranj Tahir. 2018. The Impact of Financial Crisis and ISIS on the Performance of Telecommunication Companies in Iraq. Sulaymaniyah: University of Sulaimani.
- Akins, Brian, Lynn Li, Jeffrey Ng, and Tjomme O. Rusticus. 2016. Bank competition and financial stability: Evidence from the financial crisis. Journal of Financial and Quantitative Analysis 51: 1-28. [CrossRef]
- Akosah, Nana, Francis Loloh, Natalia Lawson, and Claudia Kumah. 2018. Measuring Financial Stability in Ghana: A New Index-Based Approach. MPRA Paper, Ghana. Available online: https://mpra.ub.uni-muenchen.de/86634/ (accessed on 12 May 2018).
- Al-Hadi, Ahmed, Mostafa Monzur Hasan, Grantley Taylor, Mahmud Hossain, and Grant Richardson. 2017. Market risk disclosures and investment efficiency: International evidence from the Gulf Cooperation Council financial firms. Journal of International Financial Management and Accounting 28: 349–93. [CrossRef]
- Allen, William, and Geoffrey Wood. 2006. Defining and achieving financial stability. Journal of Financial Stability 2: 152–72. [CrossRef] Al-Rjoub, Samer. 2021. A financial stability index for Jordan. Journal of Central Banking Theory and Practice 10: 157–78. [CrossRef]
- Altman, Edward I. 1968. Financial ratios, discriminant analysis and the prediction of corporate bankruptcy. The Journal of Finance 23: 589–609. [CrossRef]
- Antonov, Ivan. 2022. Efficiency of financial management of energy companies as a component of ensuring the financial stability of the industry. Russian Journal of Resources, Conservation and Recycling 9: 1–7. [CrossRef]
- Aslam, Ejaz, Farrukh Ijaz, and Anam Iqba. 2016. Does working capital and financial structure impact profitability of Islamic and conventional banks differently? Islamic Banking and Finance Review 301: 50-67. [CrossRef]
- Ayam, Justice Ray Achoanya. 2021. Higher education financial sustainability in Ghana: A study of the perceived influential factors. Ghana Journal of Education: Issues and Practice (GJE) 7: 162–98.
- Barra, Cristian, and Nazzareno Ruggiero. 2021. The role of nonlinearity on the financial development-economic performance nexus: An econometric application to Italian banks. Empirical Economics 60: 2293–322. [CrossRef]
- Barra, Cristian, and Roberto Zotti. 2019. Bank performance, financial stability and market concentration: Evidence from cooperative and non-cooperative banks. Annals of Public and Cooperative Economics 90: 103-39. [CrossRef]

- Baur, Dirk, and Niels Schulze. 2009. Financial market stability—A test. *Journal of International Financial Markets Institutions and Money* 193: 506–19. [CrossRef]
- Bazzazan, Fatemeh. 2020. The effects of parametric reforms on retirees' welfare and financial sustainability of the Social Security Organization pension system. Social Welfare Quarterly 20: 237–69.
- Beaver, William H., Maria Correia, and Maureen F. McNichols. 2011. Financial Statement Analysis and the Prediction of Financial Distress. *Foundations and Trends*[®] in Accounting 5: 99–173. [CrossRef]
- Beck, Thorsten, Hans Degryse, and Christiane Kneer. 2014. Is more finance better? Disentangling intermediation and size effects of financial systems. *Journal of Financial Stability* 10: 50–64. [CrossRef]
- Blank, Iven Aktus. 2013. Upravlenie Finansovoj Bezopasnost'ju Predprijatija. Working Paper. Iowa City: Игорь Бланк (Igor Blank).

Campbell, John Y., Jens Hilscher, and Jan Szilagyi. 2008. In search of distress risk. The Journal of Finance 63: 2899–939. [CrossRef]

- Chen, Ju, He Chang, and Yang Xiao. 2013. The Study on Financial Market Stability Based on Quantile Regression Technique. *Finance* 3: 59–68. Available online: https://hanspub.org/DownLoad/Page_DownLoad?FileName=FIN20130400000_35430647.pdf (accessed on 6 July 2024). [CrossRef]
- Chiladze, Izolda. 2018. Theoretical and practical aspects of profitability factorial analysis. *Science and Studies of Accounting and Finance Problems and Perspectives* 12: 12–19. [CrossRef]
- Chirilă, Viorica, and Ciprian Chirilă. 2015. Financial market stability: A quantile regression approach. *Procedia Economics and Finance* 20: 125–30. [CrossRef]
- Dai, Zhiyi, and Zhengming Zhou. 2022. Research and forecast analysis of financial stability for policy uncertainty. *Computational Intelligence and Neuroscience* 2022: 8799247. [CrossRef]
- Darrat, Ali, Stephen Gray, Jung Chul Park, and Yanhui Wu. 2016. Corporate governance and bankruptcy risk. *Journal of Accounting Auditing and Finance* 31: 163–202. [CrossRef]
- Eisdorfer, Assaf, Carmelo Giaccotto, and Reilly White. 2013. Capital structure executive compensation and investment efficiency. Journal of Banking and Finance 372: 549–62. [CrossRef]
- Elsayed, Ahmed, Nader Naifar, and Samia Nasreen. 2023. Financial stability and monetary policy reaction: Evidence from the GCC countries. *The Quarterly Review of Economics and Finance* 87: 396–405. [CrossRef]
- Enaiati, Anvar, Gholamreza Kordestani, and Ataollah Mohammadi Molgharni. 2022. Factors affecting financial stability in the social security organization. *Governmental Accounting* 9: 341–72.
- Fingleton, Bernard, Paul Charles Cheshire, Harry Garretsen, Danilo Camargo Igliori, Julie Le Gallo, John Stuart Landreth McCombie, Philip McCann, Basil J. Moore, Vassilis Monastiriotis, Mark Roberts, and et al. 2008. Editorial. *Spatial Economic Analysis* 3: 269–73. [CrossRef]
- Ghassan, Hassan, and Noureddine Krichene. 2017. Financial Stability of Conventional and Islamic Banks: A Survey. Munich: University Library of Munich.
- Goswami, Shubham, Bibhas Chandra, and Vineet Chouhan. 2014. Predicting financial stability of select BSE companies revisiting Altman Z score. *International Letters of Social and Humanistic Sciences* 26: 92–105.
- Gottardo, Pietro, and Anna Maria Moisello. 2017. Family firms, risk-taking and financial distress. *Problems and Perspectives in Management* 15: 168–77. [CrossRef]
- Habib, Ahsan, Mabel D' Costa, Hedy Jiaying Huang, Bhuiyan Md. Borhan Uddin, and Sun Li. 2020. Determinants and consequences of financial distress: Review of the empirical literature. *Accounting and Finance* 60: 1023–75. [CrossRef]
- Hadian, Mehdi, and Hassan Dargahi. 2018. The Role of Macroeconomic Policies in the Financial Stability of Iran's Economy. *Iranian Journal of Economic Research* 22: 45–82.
- Halim, Md. Abdul, Syed Moudud-Ul-Huq, Farid Ahammad Sobhani, Ziaul Karim, and Zinnatun Nesa. 2023. The nexus of banks' competition ownership structure and economic growth on credit risk and financial stability. *Economies* 118: 203. [CrossRef]
- Hillegeist, Stephen A., Elizabeth K. Keating, Donald P. Cram, and Kyle G. Lundstedt. 2004. Assessing the probability of bankruptcy. *Review of Accounting Studies* 9: 5–34. [CrossRef]
- Hsu, Vernon, and Jing Wu. 2023. Inventory as a Financial Instrument: Evidence from China's Metal Industries. *Management Science* 70: 3381–4165. [CrossRef]
- Icaza, Victor Echevarria. 2017. The literature on the interaction of fiscal risk and financial stability—A survey. *Cuadernos de Economía* 40: 177–90. [CrossRef]
- Ilesanmi, Kehinde Damilola, and Devi Datt Tewari. 2020. Financial stress index and economic activity in South Africa: New evidence. *Economies* 8: 110. [CrossRef]
- Kallunki, Juha-Pekka, and Elina Pyykkö. 2013. Do defaulting CEOs and directors increase the likelihood of financial distress in the firm? *Review of Accounting Studies* 18: 228–60. [CrossRef]
- Kalsie, Anjala, Jyoti Dhamija, and Ashima Arora. 2020. The Measure of Financial Stability and its Impact on Foreign Direct Investment: Evidence from BRIC Nations. *International Journal of Accounting and Finance Review* 5: 80–93. [CrossRef]
- Kane, Michael J., David Z. Hambrick, and Andrew RA Conway. 2005. Working memory capacity and fluid intelligence are strongly related constructs: Comment on Ackerman, Beier, and Boyle (2005). *Psychological Bulletin* 31: 66–71. [CrossRef]
- Kukushina, O.S., N.A. Kanishcheva, M.V. Kivarina, and E.A. Okomina. 2021. Modeling the Financial Stability of Enterprises in the Third Sphere of the Agro-Industrial Complex of the Region. *IOP Conference Series: Earth and Environmental Science* 852: 012057. [CrossRef]

- Mabvira, Lazarus Tapuwa. 2018. Essays on the Interplay between Bank Competition Corporate Governance Financial Stability and Misreporting in the Context of the Global Financial Crisis. Ph.D. thesis, University of Sussex, Brighton, UK. Available online: https://hdl.handle.net/10779/uos.23460965.v1 (accessed on 9 August 2023).
- Magee, Shane. 2013. The effect of foreign currency hedging on the probability of financial distress. *Accounting and Finance* 53: 1107–27. [CrossRef]
- Mamipour, Siab, and Farzaneh Godarzi. 2020. Investigating of Fiscal Policy Sustainability in Iran, by Markov Switching Unit Root Test. Journal of Economic Research (Tahghighat-E-Eghtesadi) 55: 437–62.
- Martinez-Miera, David, and Rafael Repullo. 2020. Interest Rates, Market Power, and Financial Stability. *Finance and Economics* 26: 4706–22.
- Mataibayeva, Gaukhar, Serik Makysh, Nurilya Kuchukova, Saule Zhalbinova, and Aigerim Zhussupova. 2019. Conceptual approaches to the public debt management and its impact on financial stability. *Entrepreneurship and Sustainability* 72: 1496–513. [CrossRef] [PubMed]
- Mayasari, Mayasari, and Niken Wulandari. 2022. Pengaruh financial stability, efectivitas internal control, dan auditor change fraud triangle financial statement fraud. *Jurnal Manajemen Dan Bisnis* 22: 128–42. [CrossRef]
- Misztal, Piotr. 2021. Public Debt Management and The Country's Financial Stability. Studia Humana 10: 10–18. [CrossRef]
- Moudud-Ul-Huq, Syed, Tanmay Biswas, Md Abdul Halim, Miroslav Mateev, Imran Yousaf, and Mohammad Abedin. 2022. The effects of bank competition financial stability and ownership structure: Evidence from the Middle East and North African MENA countries. *International Journal of Islamic and Middle Eastern Finance and Management* 154: 717–38. [CrossRef]
- Nadri, Kamran, Sajad Ebrahimi, and Abbas Fadaie. 2018. An Investigation of Co-Movement of Financial Stability Index with Macro-Prudential Indicator through Wavelet Analysis. *Journal of Money and Economy* 13: 125–51.
- Nwogugu, Michael. 2015. Corporate Governance Financial Stability and Evolving Insurtech: The Case of Insurance Australia Group 2011–15. Working Paper. Available online: https://papers.ssrn.com/sol3/papers.cfm (accessed on 20 June 2023).
- Ohlson, James A. 1980. Financial ratios and the probabilistic prediction of bankruptcy. *Journal of Accounting Research* 1: 109–31. [CrossRef]
- Oleynik, Olga, and Irina Borisova. 2019. The Comprehensive Assessment of the Strategy of Achieving Financial Sustainability in Engineering Companies. *Regionalnaya Ekonomika Yug Rossii* 7: 165–72. [CrossRef]
- Orazalin, Nurlan, Cemil Kuzey, Ali Uyar, and Abdullah S. Karaman. 2024. Does CSR contribute to the financial sector's financial stability? The moderating role of a sustainability committee. *Journal of Applied Accounting Research* 25: 105–25. [CrossRef]
- Ozili, Peterson. 2018. Banking stability determinants in Africa. International Journal of Managerial Finance 14: 462–83. [CrossRef]
- Pinčák, Richard, and Erik Bartoš. 2015. With string model to time series forecasting. *Physica A: Statistical Mechanics and Its Applications* 436: 135–46. [CrossRef]
- Rubio-Misas, María. 2020. Ownership structure and financial stability: Evidence from Takaful and conventional insurance firms. *Pacific-Basin Finance Journal* 62: 101355. [CrossRef]
- Schulze, Tatjana. 2022. Monetary Policy, Financial Stability, and Default in Advanced and Emerging Marketing Economies. Doctoral dissertation, University of Oxford, Oxford, UK.
- Shi, Jin-feng, wei-qi Liu, and wei Yang. 2011. Test for financial market stability based on quantile regression method. *Chinese Journal of Management Science* 192: 24–29.
- Stona, Filipe, Igor AC Morais, and Divanildo Triches. 2018. Economic dynamics during periods of financial stress: Evidence from Brazil. *International Review of Economics and Finance* 55: 130–44. [CrossRef]
- Sudarsanam, Sudi, and Jim Lai. 2001. Corporate financial distress and turnaround strategies: An empirical analysis. *British Journal of Management* 12: 183–99. [CrossRef]
- Susanto, Joko, and Sri Walyoto. 2023. The Effect of Corporate Governance on Financial Stability Sharia Bank in Indonesia. *Journal of Management and Islamic Finance* 31: 132–43. [CrossRef]
- Taranova, Irina V., Irina M. Podkolzina, Fatima M. Uzdenova, Oksana S. Dubskaya, and Alla V. Temirkanova. 2021. Methodology for assessing bankruptcy risks and financial sustainability management in regional agricultural organizations. In *The Challenge of Sustainability in Agricultural Systems*. Cham: Springer International Publishing, vol. 2, pp. 239–45. [CrossRef]
- Tennyson, Ben, Robert Ingram, and Michael Dugan. 2008. Assessing the Information Content of Narrative Disclosures in Explaining Bankruptcy. Journal of Business Finance and Accounting 17: 391–410. [CrossRef]
- Tinoco, Mario Hernandez, and Nick Wilson. 2013. Financial distress and bankruptcy prediction among listed companies using accounting, market and macroeconomic variables. *International Review of Financial Analysis* 30: 394–419. [CrossRef]
- Virolainen, Kimmo. 2004. Macro stress testing with a macroeconomic credit risk model for Finland. Bank of Finland Research Discussion Paper 1: 1–48.
- Wang, Jiaming, Andrey P. Jivkov, Dirk L. Engelberg, and Qingming Li. 2022. Image-Based vs. Parametric Modelling of Concrete Meso-Structures. *Materials* 15: 704. [CrossRef] [PubMed] [PubMed Central]
- Xie, Zhikang, Xinglin Liu, Hina Najam, Qinghua Fu, Jawad Abbas, Ubaldo Comite, Laura Mariana Cismas, and Andra Miculescu. 2022. Achieving financial sustainability through revenue diversification: A green pathway for financial institutions in Asia. Sustainability 14: 3512. [CrossRef]
- Yusgiantoro, Inka, Wahyoe Soedarmono, and Amine Tarazi. 2019. Bank consolidation and financial stability in Indonesia. *International Economics* 159: 94–104. [CrossRef]

Zaiane, Salma, and Fatma Ben Moussa. 2021. What drives banking profitability during financial crisis and political turmoil? Evidence from the MENA region. *Global Journal of Emerging Market Economies* 133: 380–407. [CrossRef]

Zargarkoche, Narges, and Saba SoroushRad. 2019. Investigating the dimensions of financial sustainability in the public sector. *Accounting and Budgeting of the Public Sector* 1: 24–36.

Zhang, Honghui, Hedy Jiaying Huang, and Ahsan Habib. 2018. The effect of tournament incentives on financial restatements: Evidence from China. *The International Journal of Accounting* 53: 118–35. [CrossRef]

Zmijewski, Mark E. 1984. Methodological issues related to the estimation of financial distress prediction models. *Journal of Accounting Research* 22: 59–82. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.





Mapping the Landscape of Key Performance and Key Risk Indicators in Business: A Comprehensive Bibliometric Analysis

Ștefan Ionescu¹, Gabriel Dumitrescu², Corina Ioanăș² and Camelia Delcea^{1,*}

- ¹ Department of Economic Informatics and Cybernetics, Bucharest University of Economic Studies, 010552 Bucharest, Romania
- ² Department of Accounting and Audit, Bucharest University of Economic Studies, 010552 Bucharest, Romania
 - Correspondence: camelia.delcea@csie.ase.ro

Abstract: Our study investigates the relevance and application of key performance indicators (KPIs) and key risk indicators (KRIs) in business management from 1992 to 2023 through a comprehensive bibliometric analysis performed in RStudio using the Bibliometrix platform and in VOSviewer. Utilizing data from the Web of Science database, we identify trends, key themes, and influential research in this domain, observing an annual growth rate of 17.76%. Our analyses include the top 10 most globally cited documents, word clouds based on authors' keywords and Keywords Plus, clustering by coupling, co-occurrence networks, and factorial analysis. Our findings reveal a significant increase in research interest post-2004, with sustainability and corporate social responsibility emerging as central themes. We confirm positive correlations between KPIs, improved organizational performance, and effective risk management via KRIs. This research underscores the importance of international collaboration and diverse thematic exploration in advancing the field.

Keywords: key risk indicators; key performance indicators; performance management



Citation: Ionescu, Ștefan, Gabriel Dumitrescu, Corina Ioanăș, and Camelia Delcea. 2024. Mapping the Landscape of Key Performance and Key Risk Indicators in Business: A Comprehensive Bibliometric Analysis. *Risks* 12: 125. https://doi.org/ 10.3390/risks12080125

Academic Editor: Chengguo Weng

Received: 22 June 2024 Revised: 31 July 2024 Accepted: 2 August 2024 Published: 6 August 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

1. Introduction

The ability to successfully manage risks and monitor performance is essential for organizational success in today's dynamic business climate. The usage of key performance indicators (KPIs) and key risk indicators (KRIs) has grown in importance as enterprises attempt sustainable growth while navigating through uncertainty (Frederico et al. 2020). These indicators support strategic choices that lead to long-term success in addition to assisting with performance evaluation and threat identification.

Performance management is a crucial concept for any economic system, whether we are discussing the microeconomic, mesoeconomic, or macroeconomic levels. While its core function is to measure and manage the performance of an entity such as a business, organization, or industry, the most challenging aspect of performance management is defining it from the perspective of each entity. In other words, the way we measure performance can vary depending on the goals we set and the strategic, tactical, or operational decisions we make. In fact, we can say that performance management aims to build and develop an organization capable of achieving its goals, making good choices, ensuring that the various objectives of the company align with performance indicators.

The recent literature indicates a lack of comprehensive studies exploring the interdependent relationships between KPIs and KRIs in the context of performance and risk management. This study aims to fill this gap by providing a detailed bibliometric analysis of the existing research.

Performance management can also be described and understood from the perspective of three essential characteristics or components. The first person to describe performance management was Anthony (1965), where in his study, he defined management control as "the process by which managers ensure that resources are obtained and used efficiently

and effectively in the accomplishment of the organization's objectives". This definition practically affirms that performance management is a process, not an isolated act, explicitly referring to the concept of objectives, highlighting the behavioral aspect of performance management. Armstrong (2006), also states that performance management is a systematic process for improving organizational performance by developing the performance of individuals and teams.

Thus, performance management can be seen as a process, a loop that involves iterative learning based on setting core objectives. This approach is similar to a cybernetics approach to business processes, according to Nica and Ionescu (2021). Another perception of performance management is that it can only be understood and applied within an enterprise that has a defined purpose and clear objectives. Another perspective relies on applying incentives to increase the motivation of managers and employees. Mainly, performance management control systems have a monitoring dimension that should implement performance measures based on learning processes or cybernetics control loops and the implementation of strategic objectives and an incentive system.

This iterative approach is essential for developing an organization capable of achieving its objectives and adapting to changes in the business environment. In this context, the role of KPIs and KRIs becomes crucial.

KPIs are essential tools for measuring an organization's efficiency and effectiveness in achieving its strategic goals. These indicators provide a clear picture of organizational performance and help identify areas needing improvement. For example, KPIs may include metrics such as sales revenue (Van De Ven et al. 2023), customer satisfaction levels (Setijono and Dahlgaard 2007), or product delivery times (Munmun et al. 2023). By monitoring these indicators, organizations can make informed decisions and implement strategies to enhance performance.

On the other hand, KRIs are used to identify and monitor risks that could negatively impact organizational performance. These indicators provide early warning signals, allowing for organizations to take preventive measures to manage risks before they become major issues. Risk can be encountered in any type of business and in any type of environment—e.g., affected or not by different economic, social, or pandemic crises (Santos et al. 2022; Tavares et al. 2023, 2024). For instance, KRIs may include the incidence of cybersecurity breaches (Melançon et al. 2021; Giudici et al. 2024) or rate of attrition of high-value customers (Salamah et al. 2022). Constant monitoring of these risks helps organizations prioritize resources and implement effective mitigation measures.

KRIs and KPIs should not be confused, as both are essential tools for evaluating the performance of the economic system. To successfully use them in business management and decision-making, it is crucial to understand the differences and similarities between KPIs and KRIs. For example, the customer retention rate can be considered a performance indicator (KPIs) aimed at evaluating customer satisfaction and measuring success in maintaining customer loyalty (Valenzuela-Fernández et al. 2016). From a KRI perspective, the rate of losing high-value customers can serve as a risk indicator for identifying and managing the risk of losing valuable customers.

The use of KPIs and KRIs can significantly improve the strategic decision-making process within organizations. By continuously monitoring performance and identifying risks early, managers can make informed decisions that reduce uncertainty and optimize resources to achieve organizational objectives. Additionally, implementing a robust system of KPIs and KRIs allows for organizations to optimize resource utilization and improve operational efficiency. This can lead to reduced operational costs and increased profitability while ensuring a rapid response to potential threats. Continuous monitoring of KPIs and KRIs enables organizations to become more agile and capable of quickly adapting to changes in the business environment. This ability to promptly respond to new challenges and opportunities is essential for maintaining competitiveness in the market.

In this context, integrating KPIs and KRIs is essential for ensuring long-term organizational success. These indicators not only support performance evaluation and risk identification but also facilitate strategic choices that lead to sustainable growth. This research fills a gap in the specialized literature by providing a comprehensive and systematic analysis of key performance and risk indicators in the business context. The use of a bibliometric approach is justified by the need to understand the current research landscape and identify future directions. Given the importance of KPIs and KRIs in performance and risk management, it is essential to understand how these tools are utilized and what their impact on organizations is. Previous studies have highlighted various aspects of these indicators, but there are still gaps in our knowledge. For example, it is not clear whether the use of KPIs is consistently correlated with the improvement in organizational performance or how the implementation of KRIs contributes to reducing operational risks.

To develop our research, we propose the following hypotheses:

H₁*. The specialized literature frequently highlights a correlation between the use of KPIs and the improvement in organizational performance.*

H₂*. Existing studies suggest that the implementation of KRIs is associated with a reduction in operational risks in companies.*

H₃*. International collaboration in KPIs and KRIs research is correlated with the production of more influential and highly cited studies.*

 H_4 . The specialized literature indicates that the effective management of a variety of KPIs and KRIs is essential for organizational success and value growth.

To answer this question, we conducted a bibliometric analysis to identify existing studies on business model KPIs and KRIs. The bibliometric approach allows for the analysis of a large volume of specialized literature, identification of research trends, determination of the influence of specific works and researchers, and highlighting of international collaborations. By using bibliometric techniques, we can provide a clear and detailed picture of the evolution of research and identify gaps and future opportunities. Hypothesis H₁ explores the impact of KPIs on the overall performance of companies, evaluating whether their use leads to measurable increases in efficiency and effectiveness. The second research hypothesis, H₂, analyzes whether the use of KRIs helps companies anticipate and manage risks more efficiently, thereby preventing major issues. Hypothesis H₃ examines the impact of international collaboration on the quality and influence of research, suggesting that global partnerships can lead to more robust and recognized results. The final proposed hypothesis, H₄, aims to identify the types of KPIs and KRIs most frequently mentioned in the literature and to evaluate their impact on organizational performance and value.

Thus, the primary aim of this study is to map the research landscape of KPIs and KRIs in the business context through a bibliometric analysis. This study aims to identify and evaluate the correlations between the use of KPIs and the improvement of organizational performance, the impact of implementing KRIs on reducing operational risks, the influence of international collaboration on research quality, and the diversity in KPIs and KRIs frequently mentioned in the academic literature, as well as their effective management for organizational success and value growth.

This study primarily focuses on performance and risk indicators at the micro- and mesoeconomic levels. We concentrated on these economic levels, considering that the concept of business is central to the selection of keywords used for database extraction. Thus, our analysis aims to enhance the understanding of how KPIs and KRIs are applied at the organizational and industry levels, without directly focusing on macroeconomic indicators. This approach allows for us to provide more precise and relevant recommendations for performance and risk management in specific business contexts.

2. Literature Review

Organizational success and strategic decision-making in the context of the dynamic and evolving business world make the measurement and analysis of economic system performances a critical component influencing these objectives (Asiaei and Bontis 2019; Tasheva and Nielsen 2022).

The study of KPIs and KRIs has gained significant attention in the field of business management and performance analysis. The main objectives of performance management systems are the identification and monitoring of KPIs (Cruz Villazón et al. 2020). These indicators are critical for organizations to monitor and manage their performance and risk effectively. KPIs are metrics that help organizations understand how well they are achieving their strategic and operational goals, while KRIs provide early warning signals about potential risks that could impact these objectives.

Research in this domain has evolved over the years, with a growing body of literature exploring various aspects of KPIs and KRIs. Early studies focused on defining and categorizing these indicators, establishing their importance in performance management frameworks. As the field increased in maturity, researchers began to investigate the application of KPIs and KRIs in different industries and contexts, examining their impact on organizational performance and risk management.

Recent advancements in technology and data analytics have further expanded the scope of KPI and KRI research. The integration of big data, artificial intelligence, and machine learning techniques has enabled more sophisticated and real-time monitoring of performance and risk. This has led to a more nuanced understanding of how organizations can leverage these indicators to drive decision-making and strategic planning.

Burlea-Schiopoiu and Ferhati (2020) aim to identify specific KPIs for the healthcare system to help propose recommendations for managers and employees in this sector. These recommendations will contribute to the evaluation and monitoring of critical factors that can influence the performance of this sector in Algeria. Khalifa and Khalid (2015), in their study, classify key performance indicators into three levels: operational, tactical, and strategic. Another study conducted by (Braglia et al. 2022) proposes investigating how to design and implement KPIs that can be used to monitor the evolution of core processes that will benefit from Industry 4.0 technologies. Additionally, Gackowiec et al. (2020) consider it essential to identify the most relevant KPIs for the analyzed processes within an organization. Their study emphasizes efficiency indicators adjusted to the needs of the mining industry specific to Industry 4.0.

In the study conducted by (Tseng et al. 2011), the authors develop an integrated approach that includes key risk factors along with management strategy risk. The aim is to facilitate the periodic evaluation of these risk factors and their involvement in the dynamics of business-to-business international internet banking.

Dahal et al. (2024) conducted a bibliometric analysis to address and respond to the gaps observed in the characteristics proposed by performance management systems. Based on their analysis, they concluded that (i) there is a notable increase in the collaborative nature of KPI research; (ii) there is a significant shift towards integrating sustainability and technological advancements into performance parameters; organizations can better understand performance by combining multiple performance perspectives such as internal regulatory frameworks, social effects, and customer satisfaction.

Another study conducted by (Van De Ven et al. 2023) aimed to systematically review the specialized literature to analyze and consolidate the current state of KPIs used in business models. The main findings focused on synthesizing business model-specific KPIs into a catalog structured around the dimensions of the business model. Additionally, Li et al. (2023) reviewed energy flexibility KPIs and open datasets that can be used for testing these KPIs. They concluded that incorporating occupant comfort and the acceptance of reduced comfort in the development of KPIs can provide valuable insights and improve the performance of B2G services. Regarding the importance of KRIs, Niță et al. (2023) analyzed the impact of preventive measures against cyber fraud, focusing on the implementation of risk indicators and the use of protection mechanisms based on artificial intelligence. They addressed economic and legal perspectives and proposed a method for evaluating the compliance of artificial intelligence tools for cyber fraud prevention and monitoring, and adjusting them based on the evolution of key risk indicators over time. Additionally, F. Liu (2023) presents in his study that the increasing uncertainty in the operations and development of many enterprises due to the new global economic recession leads to the emergence of new risks. Enterprises urgently need more effective methods and approaches for risk management to improve the sensitivity in identifying and responding to risks. Establishing a risk early warning mechanism, especially the construction of a key risk indicator system, can become one of the effective methods to help enterprises respond rapidly to risks.

Lihachev (2024), through his study, emphasizes that the selection of a system of indicators can aid in evaluating the economic efficiency of production industries and in choosing the most efficient investment projects.

The literature review highlights the importance of performance and risk management systems, particularly using KPIs and KRIs, in the context of organizational success and strategic decision-making. The analyzed studies emphasize that the identification and monitoring of these indicators are essential for the effective evaluation and management of organizational performance and risks.

3. Results

3.1. Preliminary Data Analysis

In this section, we provide an initial analysis of the most significant data collected from our bibliometric study on KPIs and KRIs in business. This analysis offers insights into the research landscape, highlighting the active and extensive community of researchers involved in this field. We explore the dynamics and impact of the research, underscoring a clear trend towards collaboration and internationalization. These findings offer a comprehensive view of the current state and future directions of research in KPIs and KRIs, emphasizing the importance of global cooperation in advancing knowledge and practices.

Table 1 provides an overview of the data analyzed in our bibliometric study. Thus, our analysis covers the years 1992–2023, reflecting over three decades of research. The total number of sources used, including journals, books, and other documents, is 760. The total number of documents analyzed is 1395. Additionally, we observe that the annual average growth rate of the number of documents is 17.67% and the average age of the documents is 7.41 years. Furthermore, each document has an average of 18.16 citations, and the total number of references cited in all documents is 58,511. These data indicate growing interest and robust academic production in the analyzed field, with a significant growth rate and a considerable number of citations, suggesting the relevance and influence of research on KPIs and KRIs in business.

Table 1. Main information.

Main Information about Data	Results WoS
Timespan	1992:2023
Sources (journals, books, etc.)	760
Documents	1395
Annual growth rate (%)	17.67
Document average age	7.41
Average citations per doc	18.16
References	58,511

Table 2 provides key information about the content of the documents analyzed in our bibliometric study. We observe that the total number of Keywords Plus used in the analyzed articles is 1789. These keywords are generated by databases to enhance searches and better reflect the content of the articles. Additionally, the total number of author's keywords is 4317. These keywords reflect the main themes and subjects addressed in the articles, offering insight into the points of interest in KPI and KRI research. The large number of Keywords Plus and author's keywords indicates significant thematic diversity in the studied literature. This suggests a comprehensive approach to topics related to KPIs and KRIs, highlighting the complexity and multidimensionality of research in this field.

Table 2. Document contents.

Main Information about Documents	Results WoS
Keywords Plus	1789
Author's keywords	4317

Important details regarding the writers of the examined articles are shown in Table 3, as can be seen. There are 4094 authors in all who have contributed to the materials under analysis. This figure indicates the size of the community of active researchers in the subject of business KPIs and KRIs. Only 179 authors can claim to have authored articles. This implies that while there are a lot of collaborations in this sector, there is also a lot of interest in solitary study. The total number of authors as well as the number of single authors in the papers signal a vibrant and varied research community. The existence of both individual and collaborative efforts highlights the complexity and diversity of techniques in the study of KPIs and KRIs and advises a balance between cooperative and independent research.

Table 3. Authors.

Main Information about Authors	Results WoS	
Authors	4094	
Authors of single-authored docs	179	

The data from Table 4 reflect a balance between individual and collaborative research, with a strong emphasis on international collaborations. There are 182 single-authored documents, indicating that while collaboration is prevalent, individual research holds significant importance in the field. On average, each document has 3.15 co-authors, highlighting a high level of collaboration among researchers. Additionally, 23.87% of the documents involve international collaborations. This high percentage underscores the importance of international cooperation in KPI and KRI research. This suggests that the field benefits from a global network of researchers working together to advance knowledge and practices.

Table 4. Authors collaboration.

Main Information about Authors	Results WoS
Single-authored docs	182
Co-authors per doc	3.15
International co-authorships (%)	23.87

Thus, regarding the preliminary analysis of the most important data, we observe that research on KPIs and KRIs in business is characterized by a vast and active community of researchers. All this information provides a complex picture of the dynamics and impact of research in the analyzed field, highlighting a clear trend towards collaboration and internationalization.

3.2. Research Output and Impact Analysis

In this section, we analyze key visualizations to better understand the research landscape for KPIs and KRIs in business. Specifically, we examine the annual scientific production, the average citation per year, and a three-field plot. These analyses provide insights into the evolution, impact, and interconnectedness of research in this domain.

The scientific output for the considered time-period is highlighted with the help of Figure 1. With regard to the number of articles that have been published annually, a slight fluctuation can be observed between 1992 and 2003, but this period contains only a few articles, ranging from 1 in 1992 to 8 in 2002. However, starting from 2004 onwards, significant interest in the subject's matter took place, making the number of published articles range between 9 in 2004 and 159 in 2023, leading the trend to gradually increase over the years, most notably for the recent ones, registering a growth rate higher than the one in the late 2000s. This exponential growth for the recent years reflects that the researchers are more engaged and interested in the subject.



Figure 1. Annual scientific production.

It can be observed in Figure 2 that, throughout the considered period, there was very high fluctuation in the average citations, ranging from 0.1 to 3.9. From 1992 to 1998, the fluctuation was relatively small. Yet, 1999 registered a first peak in the trend with 3.5 citations, as companies began to think about risk indicators, considering the fact that specialists in the area of economics predicted the recession that was to come in the next year. However, in the following years, from 2000 to 2006, it is evident that interest in key performance indicators and key risk indicators decreased, compared to 1999, but still with fluctuations in this timeframe. Despite that, the year 2007 registered the trend's second and highest peak, with 3.9 citations, as everyone was expecting the next year's global economic crisis. For the remaining period, between 2008 and 2023, the number of citations still fluctuated, particularly for year 2011, when the global stock market crashed, and the year 2019, considering the start of the COVID-19 pandemic. Considering the dynamic of the average citations over the years, it is notable that the researchers' interest in the analyzed field is strongly linked with expected major world events that may impact the global economy.



Figure 2. Average citations per year.

In Figure 3, we created a three-field plot describing the keywords, countries and Keywords Plus. The plot's first column shows how the considered keywords are linked to the countries, allowing for us to examine if a balanced distribution exists or if there are certain countries that dominate in a specific field. The second column orients our focus to the countries that are involved in the research, providing information about how each of them is related to the used keywords and Keywords Plus. The third and last column showcases, similarly to the first one, the association between Keywords Plus and the countries of origin, offering insights on how they are distributed.



Figure 3. Three-field plot.

3.3. Author's Analysis

According to Figure 4, it is certain that Resinas M. is the most relevant author from the top 20, having the most contributions, with six articles, closely followed by Del-Rio-Ortega A. and Mate A., both of them having published five scientific papers. Among other authors that stand out, with four publications each, there are Loizia P., Ruiz-Cortes A., Singh S., Skouloudis A., Tadic D., Voukkali I., and Zorpas AA.



Figure 4. The most relevant 20 authors according to the total number of scientific publications.

With regard to the publication date of the scientific papers considered for the analyzed topic, in which only the first 20 most productive authors are taken into consideration, there is no doubt that there is a notable emerging trend, according to Figure 5. More explicitly, this trend is described by the fact that most of the articles were published from the year 2013 onwards.





Figure 6 illustrates the distribution of authors' productivity based on Lotka's Law, which states that the number of authors publishing *n* papers is inversely proportional to n^2 . We observe that 94% of the authors have written only one paper (3857 authors), indicating that the majority of contributors are occasional authors. Then, the vast majority of authors contribute infrequently, writing only one paper. A very small proportion of authors produce multiple papers, reflecting higher effort, specialization, or sustained research activity in the field. This distribution supports Lotka's Law, demonstrating that prolific authors are rare, while most authors have limited publications.



Figure 6. Author productivity through Lotka's Law.

3.4. Affiliations and Funding Agencies

In this subsection, we analyze the top 20 most relevant affiliations of authors. This analysis puts an emphasis on the institutions and countries that contribute the most to research on KPIs and KRIs. It can help recognize centers of excellence in this field. Additionally, the analysis can show how collaborations are distributed among different countries and institutions, suggesting trends in international collaboration and academic partnerships. Furthermore, identifying institutions that frequently publish in this field can help in assessing the impact and relevance of their research, providing a broader context for the quality and influence of these works. Funding agencies and academic institutions can use these data to direct resources to the areas and institutions with the highest activity and research potential. We also conduct an analysis from a funding perspective.

Table 5 highlights the most relevant affiliations of authors, showing the number of published articles and the countries of origin. We observe that the most prolific contributor is the Ministry of Education and Science of Ukraine with 63 articles, indicating substantial academic interest in Ukraine. Additionally, with 25 articles, the Indian Institute of Technology System (IIT System) ranks second, reflecting significant involvement in KPI and KRI research. Another important contributor is the University of Belgrade in Serbia with 22 articles, demonstrating a strong academic base in Serbia. Furthermore, the Bucharest University of Economic Studies in Romania, with 17 articles, indicates increased interest and notable academic activity in Romania. Other institutions, including universities from Croatia, Spain, Slovakia, Italy, the USA, Australia, Greece, and others, also contribute to this field, underscoring the global and collaborative nature of research on KPIs and KRIs.

Figure 7 highlights the top 20 funding agencies that support research on KPIs and KRIs, showing the number of funded projects. We observe that with 43 funded projects, the European Union is the main supporter of research, reflecting a strong commitment to advancing knowledge in this field. Additionally, the Spanish government has funded 16 projects, indicating significant involvement in KPI and KRI research. The Ministry of Science and Technology Taiwan has supported 11 projects, underscoring the importance of research in Taiwan. Other funding agencies, including those from Australia, Canada, Korea, and others, underscore the global nature of financial support for research. This suggests a vast international commitment to advancing knowledge and practices in the field of KPIs and KRIs.

Affiliation	No. of Articles	Country
Ministry of Education and Science of Ukraine	63	Ukraine
Indian Institute of Technology System (IIT System)	25	India
University of Belgrade	22	Republic of Serbia
Bucharest University of Economic Studies	17	Romania
National Institute of Technology (NIT System)	15	India
Egyptian Knowledge Bank	15	Egypt
University of Zagreb	14	Croatia
University of Sevilla	13	Spain
Technical University Kosice	13	Slovakia
Indian Institute of Technology (IIT)—Delhi	12	India
Sapienza University Rome	12	Italy
State University System of Florida	12	United States of America
Melbourne Genomics Health Alliance	11	Australia
University of Aegean	11	Greece
Indian Institute of Management (IIM System)	10	India
Loughborough University	10	United Kingdom
Delft University of Technology	9	Netherlands
Monash University	9	Australia
Sakarya University	9	Turkey
Universidade Federal De Santa Catarina (UFSC)	9	Brazil







3.5. Countries Analysis

The first 20 countries with the most remarkable corresponding authors can be seen in Figure 8, which showcases the global distribution with regards to the scientific papers that have been written for the analyzed matter. Based on the below figure, it can be seen that the United States of America is the country with the most corresponding authors, having 112 articles, whereas China and United Kingdom follow closely, with 96 and 93 scientific papers each. This suggests that the aforementioned countries are able to produce a very high level of research output, emphasizing the strong engagement in the academic activities that are linked to the analyzed subject. Yet, countries such as India, Australia, Spain, and Italy are also among the top contributors, considering their number of articles, along with Germany, Russia, and Ukraine. It is worth mentioning that for Canada, even if the number of scientific papers is relatively low when compared to other countries, the rate between main corresponding publications and the total number of articles is rather high, implying that the country is more focused on research that is impactful. The distribution of the corresponding author countries emphasizes the global nature and importance of academic



collaboration, highlighting the fact that these are the main catalysts related to the overall expertise in the subject's matter.



With the help of a visualization map covering all the world's countries, according to Figure 9, we can easily see the states that have had the highest scientific contribution. In addition to this, the colors follow a certain pattern, ranging from dark blue to gray, basically underscoring the scientific production of each country. Due to a darker color used, the countries with the most contributions can be observed rather easily—countries such as the United States of America, with 241 articles; China, having 203 scientific papers; the United Kingdom, with 189 articles; and Italy and Spain, with a combined 156 publications. Yet, there are also countries that had a low contribution, implicitly a lighter blue or gray color on the visual map, for instance, Argentina, with two scientific papers, or Peru, with three articles.



Figure 9. Corresponding authors' countries.

By examining the global trend with regard to the scientific research, it is visible in Figure 10 that the United States of America and China are the countries with the most citations, being cited 3331 and 3149 times, respectively.



Figure 10. Top 20 most cited countries.

3.6. Sources Analysis

The conducted analysis on the top 20 journals most relevant to the subject matter illustrates the diversification and importance of research among various journals, as can be observed in Figure 11.



Figure 11. Top 20 most relevant sources.

The most significant one is *Sustainability*, with 67 articles, highlighting the significance of KPIs and KRIs in businesses, organizations, and companies with regard to the overall economic state of the countries. Among other noteworthy journals, we find the *International Journal of Productivity and Performance Management*, with 20 scientific papers; *Journal of Cleaner Production*, taking into account 16 articles; *Benchmarking: An International Journal*, with 15 articles; *Financial and Credit Activity: Problems of Theory and Practice* and the *International Journal of Production Economics*, each of them having 12 publications; and the *Business Process Management Journal* and *Expert Systems with Applications*, each accounting for 11 articles. Therefore, this journal variation indicates that the analyzed topic is a comprehensive one, covering not only the impact that the studied indicators may have on the general economic context within the countries, but also the performance of companies themselves, thus opening the door to more strategic decision-making for implementation within organizations, but only after the business itself chooses the right KPIs and KRIs as measurement units for its performance.

3.7. Document Analysis

In this section, we delve into various analytical methods to understand the research landscape regarding key performance indicators (KPIs) and key risk indicators (KRIs). We present the top 10 most globally cited documents, providing insights into the foundational works in this field. Additionally, we utilize word clouds based on authors' keywords and Keywords Plus to identify the most frequent terms. Our analysis extends to clustering by coupling, co-occurrence networks, and factorial analysis, revealing the structural relationships and thematic clusters within the research. These comprehensive analyses offer a multifaceted view of the key trends and priorities in the literature on KPIs and KRIs.

Table 6 lists the top 10 most globally cited documents. The analysis of these highly cited documents highlights significant contributions to the specialized literature on KPIs and KRIs. The first document, published by Sun et al. (2007) in the *Academy of Management Journal*, has the highest total number of citations, indicating major influence. Hanna et al. (2011) in *Business Horizons* presents the highest annual and normalized citations, reflecting a continuous and robust impact. Overall, the analyzed articles come from renowned journals and cover diverse themes, ranging from financial performance and sustainability to supply chain management and corporate governance. This diversity underscores the relevance and complexity of research on KPIs and KRIs, as well as their importance for various aspects of organizational management. We continue with a qualitative analysis of these articles to observe how performance and risk metrics and indicators are utilized.

First Author, Year, Journal, References	Total Citations	Total Citation (TC) per Year	Normalized TC
Sun LY., 2007, <i>Academy of Management Journal,</i> (Sun et al. 2007)	956	53.11	13.42
Hanna R., 2011, Business Horizons, (Hanna et al. 2011)	783	55.93	21.67
Milne M.J., 2013, <i>Journal of Business Ethics,</i> (Milne and Gray 2013)	608	50.67	23.22
Ceccagnoli M., 2012, <i>MIS Quarterly</i> , (Ceccagnoli et al. 2012)	445	34.23	10.88
López M.V., 2007, <i>Journal of Business Ethics,</i> (López et al. 2007)	440	24.22	6.18
Lee A.H.I., 2008, <i>Expert Systems with Applications</i> , (Lee et al. 2008)	366	21.53	11.32
Hermann B.G., 2007, <i>Journal of Cleaner Production,</i> (Hermann et al. 2007)	237	13.17	3.33
Franking-Johnson E., 2016, <i>Journal of Cleaner Production</i> , (Franklin-Johnson et al. 2016)	225	25.00	13.27
Varsei M., 2014, Supply Chain Management, (Varsei et al. 2014)	220	20.00	7.57
Turner R., 2012, <i>Project Management Journal,</i> (Turner and Zolin 2012)	219	16.85	5.35

Table 6. Top 10 most cited documents globally.

With regard to the article published by Sun et al. (2007), service-oriented organizational citizenship behavior was studied in order to reveal how much it can impact the linkage between performant human resource policies and KPIs, the latter taking into account only turnover and productivity. The authors conducted their analysis on hotels in the People's Republic of China.

For the second research article (Hanna et al. 2011), the authors gave insights about the importance of setting and the right KPIs and how to measure these with regard to the online social media world, for platforms such as YouTube, Twitter, or Facebook, in order for them to reach their target audience—that of young consumers. More than this, the researchers treated this matter by approaching the social media context as a mix of both traditional and digital media, suggesting some best practices for these aforementioned online social platforms to increase the potential range of strategic marketing decisions that can be made with the help of KPIs.

From a critical point of view, Milne and Gray (2013) emphasized in their research the fact that an analysis of life-supporting ecological systems should also be considered when practices of sustainability reporting are being tackled. Therefore, the authors consider that the corporate or organizational level of sustainability may also be described by the level of environmental, social, and economic performance indicators. Yet, nowadays, from their perspective, even if there are diverse benchmarking approaches that should reinforce the sustainability of a business, it seems that these are not enough to also cover the ecological part that seems to be somehow disregarded in this process, thus arguing about the fact that the benchmarking tools should give more credit to the ecological area.

Ceccagnoli et al. (2012) studied how partnerships between enterprise software companies and relatively small companies that act as software vendors (ISVs) impact the latter, with the help of KPIs proving that these kinds of partnerships are beneficial for ISVs. The software vendors' sales rose considerably, and the partnership offers they received from software enterprises did as well.

López et al. (2007) analyzed how corporate social responsibility (CSR) accounting practices impact business performance, if adopted. The researchers did this by comparing the differences between the KPIs of European organizations that had already adopted some of the CSR practices to those who did not. The authors proved that there was a slight negative impact on the KPIs of companies that adopted CSR practices in comparison to businesses that had not implemented any CSR norms.

Lee et al. (2008) highlighted how important information technology (IT) is in the context of any business, underscoring the way that IT departments can contribute to the strategic goals of organizations. The scientists highlighted the fact that these kinds of technical departments cannot apply their performance measurement from a monetary unit perspective, but rather from four major ones: financial, internal business process, customer, and growth. Therefore, the authors implied that each perspective should have its own set of KPIs, as all these perspectives are different from a performance point of view. Furthermore, the results that the scientists obtained after constructing a model in which any of the abovementioned perspectives was considered separately in order to rightly evaluate the overall performance of the IT departments proved to offer very valuable insights within the department. Therefore, a detailed productivity tree of the department via performance evaluation should provide enough data for the business to consider the right strategy for further development of the department.

By their study, Hermann et al. (2007) provided a tool that can be used by businesses to find out essential information on the environmental impact they have. The authors mentioned that parts of diverse tools, such as environmental performance indicators, multi-criteria analysis, and life cycle assessment were integrated with their solution and environmental performance indicators.

The study published by Franklin-Johnson et al. (2016) proposes a new indicator in order to assess the environmental performance with regard to the circular economy, mentioning that the current methodology is quite cumbersome. This new suggested indicator considers the earned recycled lifetime, initial lifetime, and earned refurbished lifetime, and its management is crucial for the decision-making process, but also for making the right KPI assessment with regard to the circular economy, as per the authors' mentioning. The researchers also state that the indicator should play a major role on a managerial level as well, not only on an organizational one, thus helping in the measurement that a business decision has against the longevity of materials.

The article published by Varsei et al. (2014) proposes a framework that can be used by the focal companies to assist them in developing supply chains that are sustainable. Therefore, the authors' framework excels in detecting and assessing a specter of various key performance, economic, and environmental indicators, thus helping the companies achieve a sustainable supply chain. Certainly, this is of utmost need nowadays, as the researchers believe that there is an increasingly tendency for these focal companies to incorporate the assessment and advancement of sustainability with regard to supply chains in the supply chain's management, pointing out that the companies may otherwise miss out on an overall better estimation of the supply chain's sustainability.

The aim that Turner and Zolin (2012) had in their paper was to propose a set of critical KPIs in order to help managers who handle big projects give a better prognosis about stakeholder perceptions of success while projects are still ongoing. The researchers mentioned with certainty that some of these projects may be very long-living within an organization, basically increasing the chance that the project's stakeholders' definitions of success change over time. Thus, the authors suggested a model that should achieve stakeholder success at any point in the future. The advised model extends the traditional triple constraint that is described by scope, cost, and time, each of them with their own specific KPIs, to also include two KPIs related to the projects' success and seven KPIs linked to the stakeholders' satisfaction about the projects. Therefore, the usage of this tool by project managers should provide them with better control over the projects they lead.

Figure 12 presents the word cloud analysis created based on the authors' keywords, revealing frequently used terms in the specialized literature on KPIs and KRIs, providing valuable insights into the dominant themes and their alignment with our research hypotheses. The central term "key performance indicators" reflects the focus on KPIs as essential tools for measuring and improving organizational performance. Additionally, the appearance of the keyword "sustainability" highlights the importance of sustainability in organizational performance and the alignment of KPIs with sustainable development goals. The keywords "performance", "performance measurement", and "performance indicators" reflect major interest in measuring and evaluating performance in various organizational contexts. Furthermore, "business performance", "business intelligence", and "benchmarking" emphasize the link between KPIs and business performance, the use of business intelligence, and benchmarking practices for performance evaluation.



Figure 12. Word cloud based on authors' keywords.

Regarding the authors' keywords relevant to KRIs, we observe "risk management" and "risk". These terms highlight the importance of identifying and managing risks within organizations. Risk management is essential for protecting and optimizing business operations and emphasizing the role of KRIs in anticipating and mitigating risks. Additionally, the appearance of the keyword "supply chain management" may indicate that managing risks in supply chains is a major concern. KRIs are used to identify risks in the supply chain and implement mitigation measures to ensure continuity and efficiency in operations. Additionally, the concept of "financial performance" is highly relevant to KRIs because financial risks are a crucial part of KRI assessment. These terms underscore the central role of KRIs in identifying, monitoring, and managing risks from various aspects of business, from supply chains to financial performance and corporate governance. These perspectives provide a deeper understanding of how KRIs are integrated into risk management strategies and highlight their relevance in ensuring organizational success.

The word cloud in Figure 13, created based on Keywords Plus, provides valuable insights into the dominant themes and their frequency in the literature related to KPIs and KRIs in business. The most frequent term is "management", indicating that management is a central topic in discussions about KPIs and KRIs. Additionally, key concepts such as "impact" and "performance" highlight the importance of evaluating effects and performance in the context of using KPIs and KRIs. Keywords like "model" and "framework" reflect the extensive use of models and conceptual frameworks in analyzing and applying KPIs and KRIs. More than this, we observe a high frequency of keywords such as "innovation" and "strategy". This demonstrates the connection between innovation, strategy, and organizational performance. Significant concern is also highlighted for "sustainability" and "corporate social responsibility".



Figure 13. Word cloud based on Keywords Plus.

Additionally, other keywords found in this word cloud include "decision-making", "quality", "satisfaction", "risk", and "balanced scorecard". The keyword "decision-making" is highly relevant to our analysis, as the ability to make informed decisions is essential for organizational success, emphasizing the importance of KPIs and KRIs in the decision-making process. KPIs and KRIs provide data and information that support strategic and tactical decision-making. The keyword "quality" highlights that it is a vital aspect of organizational performance, and the use of KPIs is essential for monitoring and improving the quality of products and services. The keyword "satisfaction" indicates the importance of customer and employee satisfaction in evaluating organizational success through KPIs. Satisfaction KPIs help measure satisfaction levels and identify areas needing improvement. The concept of "risk" is crucial in KRI analysis, underscoring the importance of identifying and managing risks to protect and optimize business operations. The appearance of this keyword in the word cloud reflects the focus on risk evaluation and reduction using specific indicators. Finally, "balanced scorecard" is also a keyword observed in our word cloud. The

balanced scorecard is a frequently used strategic methodology to align business activities with the organization's vision and strategy, and its integration into KPI research indicates a structured and holistic approach to performance management.

The clustering by coupling graph in Figure 14 provides a visualization of clusters based on centrality and impact. On the vertical axis, we have impact, and on the horizontal axis, we have centrality. Each cluster is colored differently and represented by a specific set of dominant keywords. Table 7 presents the values of centrality and impact for each label and formed cluster.



Figure 14. Clusterization by document coupling (coupling by references, global impact, cluster labels, and author keywords).

Table 7. Clusters based on coupling map analysis.

Label	Cluster	Freq	Centrality	Impact
corporate social responsibility—conf 70% sustainability—conf 26.7% financial performance—conf 60%	1—red	21	0.307	4.290
performance—conf 30% performance measurement—conf 21.4% performance management—conf 21.4%	2—blue	33	0.311	2.725
organizational performance—conf 40% risk—conf 40% business performance—conf 18.2%	3—green	14	0.373	2.201
key performance indicators—conf 26.9% performance measurement—conf 25% supply chain management—conf 41.7%	4—purple	44	0.439	3.305
balanced scorecard—conf 69% key performance indicators—conf 53.8% performance measurement—conf 35.7%	5—orange	91	0.475	2.575
business performance—conf 36.4% competitive advantage—conf 66.7% dynamic capabilities—conf 100%	6—brown	21	0.357	2.519
balanced scorecard—conf 17.2% performance measurement—conf 14.3% benchmarking—conf 42.9%	7—pink	26	0.308	2.044

Thus, Cluster 1 (red) focuses on corporate social responsibility and sustainability, with very high impact and moderate centrality, suggesting these themes are essential for KPI and KRI research. Cluster 2 (blue), with a centrality of 0.311 and an impact of 2.725, indicates moderate values, reflecting the importance of performance evaluation and management in the analyzed research. Cluster 3 (green) combines organizational performance with

risk and business performance, having relatively high centrality and moderate impact. Cluster 4 (purple) has a high centrality level of 0.439 and high impact, indicating the importance of KPIs and supply chain management. Cluster 5 (orange) has the highest centrality, but a more moderate impact compared to Cluster 4, highlighting the strategic use of these tools. Regarding Cluster 6 (brown), this cluster reflects the importance of business performance and competitive advantage, with moderate centrality and impact. Finally, Cluster 7 (pink) focuses on performance evaluation and benchmarking, with moderate impact and centrality.

Thus, the information in Figure 14 and Table 7 shows how various themes are grouped in KPI and KRI research. The most central and impactful themes are KPIs, balanced scorecard, and supply chain management, highlighting their importance in the organizational context. The clusters illustrate the distribution and relevance of different subjects in the specialized literature, providing a clear picture of research priorities.

The co-occurrence network created in Figure 15 is based on the authors' keywords reveals the structure and relationships among central themes in KPI and KRI research. Each node represents a keyword, and the clusters indicate thematic groups. The values of centrality and impact provide additional information about the importance and interconnectedness of these terms. We observe that seven clusters were formed, which can be interpreted as follows:

- Cluster 1 (red): Key concepts include "performance measurement", "balanced scorecard", "benchmarking", "performance management", and "supply chain management". This cluster has medium centrality and high impact and focuses on performance measurement and management methods, including balanced scorecard and supply chain management, emphasizing the importance of these tools in evaluating organizational performance.
- Cluster 2 (blue): Includes terms such as "process mining" and "machine learning". This cluster, with low centrality and moderate impact, reflects the use of advanced technologies such as process mining and machine learning in performance analysis.
- Cluster 3 (green): Key concepts such as "key performance indicators", "performance", "business performance", and "business intelligence" are identified in this cluster, which has high centrality and impact. The cluster has a clear focus on KPIs and business intelligence, highlighting the role of KPIs in improving business performance and decision-making.
- Cluster 4 (purple): Terms such as "sustainability", "performance indicators", "corporate social responsibility", and "innovation" are present in this cluster. It is centered on sustainability and corporate social responsibility, indicating concern for the social and environmental impact of organizations, having medium centrality but high impact.
- Cluster 5 (orange): With medium centrality and low impact, this classification includes terms such as "big data" and "data mining", reflecting their use in performance analysis and risk management.
- Cluster 6 (brown): Includes terms such as "COVID-19" and "risk". It has both low centrality and impact, highlighting the recent concern for risk management in the context of the COVID-19 pandemic (Munmun et al. 2023).
- Cluster 7 (pink): Includes the concept of "business process management" and has medium centrality and moderate impact. The cluster emphasizes the importance of managing business processes for improving organizational performance.

The co-occurrence network shows how various themes are interconnected in KPI and KRI research. The most central and impactful themes include performance measurement, balanced scorecard, KPIs, and sustainability, highlighting the importance of these aspects in the organizational context. The clusters show the distribution and relevance of different subjects in the specialized literature, providing a clear picture of research priorities.



Figure 15. Co-occurrence network by authors' keywords.

The factorial analysis in Figure 16 shows how various themes are interconnected and grouped in the research on KPIs and KRIs. Additionally, the factorial analysis of the authors' keywords highlights the structure and relationships among the different research themes regarding KPIs and KRIs. In this case, using the multiple correspondence analysis (MCA) method, five clusters were formed as follows:

- Cluster 1 (red) includes keywords such as "key performance indicators", "performance", "performance measurement", "business process management", "business performance", "firm performance", "business intelligence", etc. This cluster comprises essential terms for performance measurement and management. The positioning of the terms in Dim1 and Dim2 shows that they are closely related to organizational performance evaluation and the use of management tools.
- Cluster 2 (blue) groups keywords such as "financial performance", "sustainability", and "performance indicator". This cluster is focused on sustainability and financial performance, highlighting the importance of integrating sustainable development goals into organizational performance.
- Cluster 3 (green) groups keywords like "benchmarking", "supply chain management", "environmental management", etc. This cluster concentrates on supply chain management and benchmarking, indicating performance comparison practices and efficient supply chain management.
- Cluster 4 (purple) includes keywords such as "corporate social responsibility", "competitiveness", and "corporate governance". This cluster focuses on corporate social responsibility and corporate governance, reflecting concerns about social impact and business competitiveness.
- Cluster 5 (orange) groups key terms such as "innovation", "management", "productivity", and "entrepreneurship", highlighting innovation, management, and productivity, and emphasizing the importance of innovation and entrepreneurship in organizational performance.

Thus, we observe that the most prominent themes include performance measurement, sustainability, supply chain management, and corporate social responsibility, underscoring the importance of these aspects in the organizational context.


Figure 16. Factorial analysis.

Figure 17 is a map created in VOSviewer that highlights the most relevant key concepts.



🔥 VOSviewer



VOSviewer identified 999 relevant key concepts and automatically calculated a relevance score for each term, reflecting its importance in the context of the scientific articles analyzed in our study. Of all the identified terms, the top 60% of most relevant terms were selected based on their calculated relevance scores. This selection was made to focus the analysis on the terms that are most significant and contribute most meaningfully to the understanding of the research domain. Thus, the created map visualizes the structure and relationships among the essential terms, highlighting crucial themes, topics, and connections within the analyzed articles.

The keyword map from Figure 17 generated 12 clusters as follows:

- Cluster 1: Technological Integration and AI in business processes—This cluster, with 154 keywords, includes keywords such as "accuracy", "artificial intelligence", "business process management", "customer service", "digitalization", "enterprise performance", "IoT", "machine learning", and "predictive model". These terms highlight the increasing integration of advanced technologies and analytics in enhancing business processes and performance. This cluster underlines the role of digital transformation and smart technologies in optimizing KPIs, suggesting a trend towards more data-driven and technology-enabled business strategies.
- Cluster 2: Business Environment and Competitive Strategies—Cluster 2, with 92 keywords, features terms like "agricultural enterprise", "auditor", "business incubator", "competitive environment", "digital transformation", "economic development", and "strategic management". This cluster indicates the importance of external economic factors and competitive dynamics in shaping business strategies and performance metrics. It underscores the necessity for businesses to adapt to economic changes and competitive pressures through strategic management and innovation.
- Cluster 3: Industrial and Educational Performance—This cluster includes 73 keywords such as "construction industry", "economic performance", "education", and "human factor". The focus here is on the performance metrics specific to different industries and sectors. The inclusion of terms related to education and human factors highlights the role of human capital and industry-specific knowledge in driving performance.
- Cluster 4: Business Intelligence and Strategic Planning—Cluster 4, with 65 keywords, contains terms like "agility", "balanced scorecard", "business intelligence", "cloud computing", and "strategic planning". This cluster points to the strategic tools and methodologies used in measuring and enhancing business performance. It highlights the use of balanced scorecards and business intelligence as essential tools in strategic planning and decision-making.
- Cluster 5: Environmental Performance and Sustainability—This cluster, containing 50 keywords, includes terms such as "circular economy", "environmental performance", "energy efficiency", and "social impact". It underscores the growing underline on sustainability and social responsibility in business performance metrics. This cluster reflects how environmental and social factors are becoming integral to the evaluation of business performance.
- Cluster 6: Business success and quality management—With 38 keywords, this cluster features terms like "DEA (Data Envelopment Analysis)", "business success", "Confirmatory Factor Analysis", and "quality management system". It indicates a focus on quality management and success metrics in various industries, particularly in the hotel industry. This cluster points to the use of sophisticated analytical methods to assess and improve business success and quality.
- Cluster 7: Financial Health and Risk Management—Cluster 7 includes 33 keywords such as "accounting", "capital", "financial distress", "financial ratio", "ICT investment", and "intellectual capital". This cluster highlights the financial metrics and risk factors that are critical to business performance. It emphasizes the importance of financial health, investment in information and communication technologies, and intellectual capital in achieving business success.
- Cluster 8: Corporate Responsibility and Reporting—Featuring terms like "annual report", "content analysis", "corporate social responsibility", "regulator", and "sustainability performance", this cluster underscores the importance of corporate governance, regulatory compliance, and social responsibility in business performance. It highlights the role of transparency and accountability in achieving sustainable business success.
- Cluster 9: Innovation and Training—This cluster includes terms such as "conceptual framework", "innovation performance", "sales performance", "training", and "skill". It underlines the role of innovation, skills development, and training in driving busi-

ness performance. This cluster suggests that continuous improvement and innovation are key to maintaining a competitive advantage.

- Cluster 10: Risk Indicators and Economic Crises—With significant terms like "business risk", "credit risk", "economic crisis", "financial risk", "firm performance", "operational risk", "risk management", and "risk indicator", this cluster highlights the critical importance of identifying, assessing, and managing risks in business. It underscores the various types of risks that can impact firm performance and the necessity of robust risk management strategies.
- Cluster 11: Value Creation and Stability—This smaller cluster, with 15 keywords, includes terms like "car company", "further development", "value creation", and "stability". It focuses on the industrial sector and the factors contributing to business stability and value creation.
- Cluster 12: Market and Economic Conditions—Containing six keywords, namely, "climate risk", "economic condition", "market value", "nation", "price", and "regression analysis", this cluster emphasizes the broader economic conditions and market factors that influence business performance and risk.

In order to validate the clusters formed and the trends resulting from the previous analyses, Table 8 was created, in which several works that support the analytical trends were extracted and reviewed. These references support the clusters identified in our bibliometric analysis, providing a comprehensive understanding of the key performance and risk indicators in business contexts.

Cluster	Keywords	Empirical Findings	References
Business process and technological integration	Accuracy, artificial intelligence (AI), business process management, machine learning (ML)	AI and ML improve business process efficiency and decision-making accuracy.	(Davenport and Rajeev 2018; Chen et al. 2022; Jawad and Balázs 2024)
Business environment and competitive strategies	Auditor, competitive environment, digital transformation, strategic management	Strategic management and digital transformation are important for competitiveness in various environments.	(Teece 2018; Hess et al. 2020)
Industrial and educational performance	Construction industry, economic performance, education	Improved construction processes and educational advancements enhance economic performance.	(Marginson 2016)
Business intelligence and strategic planning	Balanced scorecard, business intelligence, strategy map	Balanced scorecard and BI systems are vital for strategic planning and performance measurement.	(Kaplan and Norton 1996; Wixom and Watson 2010)
Environmental performance and sustainability	Circular economy, environmental performance, social impact	Circular economy principles and sustainability initiatives significantly improve environmental performance.	(Geissdoerfer et al. 2017; Marrucci et al. 2024)
Business success and quality management	Business success, confirmatory factor analysis, hotel industry	Quality management systems and confirmatory factor analysis are crucial for achieving business success in the hospitality sector.	(Ennis and Harrington 1999; Tambare et al. 2021)
Financial health and risk management	Accounting, financial distress, risk management	Financial ratios and risk management frameworks help mitigate financial distress and manage business risks.	(Brühl 2023; Cernisevs et al. 2023)

Table 8. Empirical analysis for key findings on KPIs and KRIs.

Cluster	Keywords	Empirical Findings	References
Corporate responsibility and reporting	Annual report, corporate social responsibility, sustainability performance	Corporate social responsibility (CSR) practices and integrated reporting enhance sustainability performance and stakeholder trust.	(Eccles et al. 2014; McCullough and Trail 2023)
Innovation and training	Conceptual framework, innovation performance, training	Innovation performance and training programs are key for maintaining competitive advantage and adapting to market changes.	(Damanpour and Aravind 2012; Filho et al. 2023)
Risk indicators and economic crises	Business risk, economic crisis, risk indicator	Identifying and managing risk indicators are critical for navigating economic crises and ensuring firm	(Elyasiani and Jia 2019; Deverell and Ganic 2024)
Value creation and stability	Value creation, stability, further development	Creating value through strategic resource management ensures long-term business stability.	(Kavadis et al. 2024; Qiao et al. 2024)
Market and economic conditions	Climate risk, economic condition, market value	Managing climate risk and understanding economic conditions are essential for maintaining market value and sustainability.	(Pindyck 2013; Silva et al. 2024)

Table 8. Cont.

4. Discussion

In this section, a discussion regarding the research hypothesis is provided, along with a brief comparison in terms of scientific production, authors' keywords, and most cited papers of a dataset extracted from the Scopus database using the same extraction keywords.

4.1. Discussion on Research Hypotheses

This section focuses on discussing the research hypotheses proposed in our study on KPIs and KRIs in business. Through bibliometric analysis and various visualization methods, we gained valuable insights into the research landscape and the ways these concepts are applied and studied in the literature. We analyze each hypothesis in turn to highlight the main findings and their implications for organizational management and business policies. KRIs play an important role in risk management by providing early warning signs of potential threats. These indicators help organizations anticipate and mitigate risks before they escalate, ensuring the continuity and stability of operations. For instance, KRIs can be used to monitor financial health, supply chain vulnerabilities, and compliance risks, allowing for proactive risk management strategies. In the financial sector, the use of KRIs such as liquidity ratios and market volatility indicators has enabled firms to maintain financial stability during economic downturns. Similarly, in the manufacturing industry, monitoring KRIs related to supplier reliability and production downtime has helped companies mitigate supply chain disruptions. Integrating KPIs and KRIs into a unified performance and risk management system allows for organizations to achieve a comprehensive view of their operations. While KPIs focus on tracking and improving performance metrics such as sales growth and operational efficiency, KRIs provide insights into potential risks that could hinder these performance goals. For example, an organization might use KPIs to measure customer satisfaction and delivery times, while simultaneously monitoring KRIs related to cybersecurity threats and regulatory compliance to ensure uninterrupted service and adherence to legal standards.

Regarding hypothesis H₁, which proposed the investigation of a positive correlation between the use of KPIs and the improvement of organizational performance, several results from our bibliometric analysis validated this correlation. The exponential growth in scientific production from 2004 to 2023 indicates increased interest in using KPIs, suggesting a positive correlation with organizational performance. Additionally, the analysis of highly cited documents and keywords shows continuous interest in performance measurement and management (e.g., "key performance indicators", "performance", "business performance"). This indicates that the use of KPIs is crucial for improving organizational performance, supporting hypothesis H₁. Furthermore, the word clouds highlight key terms such as "performance", "efficiency", and "impact", supporting H₁ and suggesting that the use of KPIs is essential for improving organizational performance. Essential concepts such as "key performance indicators", "performance", "performance", and "business performance" further support the hypothesis that the use of KPIs is positively correlated with improving organizational performance.

Hypothesis H₂ proposed that the implementation of KRIs significantly contributes to reducing operational risks. This hypothesis is also validated, with the observation that KRIs are often documented and used in a broad sense as part of risk management. The fluctuation in citations during economic crises (2007, 2011, and 2019) indicates increased interest in KRIs for anticipating and managing risks, validating H₂. Additionally, terms such as "risk management" and "risk", identified in the co-occurrence analysis, factorial analysis, and word clouds, highlight the importance of identifying and managing risks within organizations. The cluster including these terms emphasizes the role of KRIs in anticipating and mitigating risks, supporting hypothesis H₂.

Hypothesis H₃ posited that international collaboration in KPI and KRI research is correlated with producing more influential and highly cited studies. This hypothesis is validated by our results. The analysis of affiliations and funding agencies shows extensive international collaboration, with institutions from various countries significantly contributing to research. For example, the Ministry of Education and Science of Ukraine and the Indian Institute of Technology reflect strong international collaboration. This indicates that global partnerships can lead to more robust and recognized results, supporting H₃. The analysis of international collaboration networks shows strong collaboration, with the USA, China, and the UK being research leaders, reflecting the influence of global collaboration on research quality. Word clouds highlight terms such as innovation, big data, and integration, suggesting the importance of collaboration and integration in research, supporting hypothesis H₃ regarding international collaboration and its impact.

Finally, hypothesis H₄ suggests that the diversity in KPIs and KRIs and their effective management contribute to organizational success. The tree field plot and cluster analysis highlights the diversity in research themes and the use of KPIs and KRIs in various contexts, underscoring their importance for organizational success. Additionally, the analysis of keywords, clustering, and co-occurrence networks shows the thematic diversity in research related to KPIs and KRIs (e.g., "sustainability", "corporate social responsibility", "supply chain management", "circular economy"). These themes are essential for organizational success, supporting hypothesis H₄.

In order to emphasize the previous conclusions, an empirical analysis was also carried out based on analysis developed in VOSviewer that highlighted 12 main clusters of key concepts related to KPIs and KRIs. Each cluster is supported by empirical references that demonstrate the validity and relevance of our conclusions. For example, Cluster 1, which includes terms such as artificial intelligence and business process management, is supported by studies showing how these technologies improve the efficiency and accuracy of decisions (Davenport and Rajeev 2018; Chen et al. 2022; Jawad and Balázs 2024). Similarly, Cluster 5, which focuses on environmental performance and the circular economy, is supported by work highlighting the positive impact of sustainable practices on organizational performance (Geissdoerfer et al. 2017; Marrucci et al. 2024).

Our detailed analysis supports all four hypotheses, highlighting the importance of KPIs and KRIs in improving organizational performance, reducing risks, fostering international collaboration, and thematic diversity. These conclusions underscore the relevance

and complexity of research in KPIs and KRIs, providing a solid foundation for understanding how these tools are used to ensure organizational success and value.

From our perspective, policy implications could include recommendations for establishing common standards and practices in the use of KPIs and KRIs across different industries. Additionally, our study emphasizes the importance of integrating these tools into organizational strategies to support informed decision-making and efficient risk management. The findings can guide the development of public policies by highlighting the benefits of using KPIs and KRIs in the contexts of sustainability, social responsibility, and operational efficiency. Moreover, this study can influence future research directions by encouraging further exploration of the interdependencies between these indicators in various industrial and cultural contexts.

4.2. Brief Comparison with Scopus Extracted Dataset

Using the methodology described in Section 5, we extracted a dataset from the Scopus database to provide a brief comparison between it and the previously obtained dataset. The purpose of this comparison is to determine whether relevant information might differ when using the WoS database compared to Scopus in this specific context.

First, in terms of extracted papers, as can be observed from Table 9, there is a difference of 559 papers. It shall be noted that for the papers' extraction from Scopus database we have used the same keywords as in the case of the papers extracted using WoS database. The "*" placed at the end of the keyword indicates that any ending is possible for the search keyword—e.g., the use of the "risk_indicator*" keyword will search for both "risk indicator" and "risk indicators" keywords. Also, it shall be noted that "_" used in the search keyword will return only the papers in which both words between which the "_" is used are subsequent in the text—e.g., not separated by another word or one/more phrase/s.

Table 9. Comparison results for datasets extracted from WoS and Scopus.

Exploration Steps	Questions on Web of Science/Scopus	Description	Query	Query Number	Count WoS	Count Scopus
1	Title	Contains specific keywords related to risk indicators or risk indicators in business context	(((TI = (risk_indicator*)) OR TI = (perfor- mance_indicator*))) AND TI = (business)	#1	64	94
2	Abstract	Contains specific keywords related to risk indicators or risk indicators in business context	(((AB = (risk_indicator*)) OR AB = (perfor- mance_indicator*))) AND AB = (business)	#2	2457	4053
3	Keywords	Contains specific keywords related to risk indicators or risk indicators in business context	(((AK = (risk_indicator*)) OR AK = (perfor- mance_indicator*))) AND AK = (business)	#3	231	1073
4	Title/abstract/ keywords	Contains specific keywords related to risk indicators or risk indicators in business context	#1 OR #2 OR #3	#4	2602	4375
5	Language	Contains only documents written in English	(#4) AND LA = (English)	#5	2473	4224
6	Document type	Limited to articles	(#5) AND DT = (Article)	#6	1489	2086
7	Year published	Excludes 2024	(#6) NOT PY = (2024)	#7	1395	1954

As the quality of a research article highly depends on missing information, we performed a data quality check for the dataset extracted from the Scopus database and compared it with the one provided in Section 5 for the dataset extracted from WoS. The results of the data quality check in the case of the two datasets are presented in Table 10. In Table 10, the categories in which the extracted information is excellent are marked in green, the categories in which the information is good are in light green, the categories considered acceptable from the point of view of extracted information are in yellow, poor information is marked in yellow, and categories with completely missing information are marked in red. It should be noted that the data quality check was performed using Biblioshiny software 4.3.0.

		WoS 1	WoS Extracted Dataset			Scopus Extracted Dataset	
Metadata	Description	Missing Counts	Missing (%)	Status	Missing Counts	Missing (%)	Status
AU	Author	0	0.00%	Excellent	0	0.00%	Excellent
DT	Document type	0	0.00%	Excellent	0	0.00%	Excellent
SO	Journal	0	0.00%	Excellent	0	0.00%	Excellent
LA	Language	0	0.00%	Excellent	0	0.00%	Excellent
PY	Publication year	0	0.00%	Excellent	0	0.00%	Excellent
SC	Science categories	0	0.00%	Excellent	1929	100.00%	Completely missing
TI	Title	0	0.00%	Excellent	0	0.00%	Excellent
TC	Total citation	0	0.00%	Excellent	0	0.00%	Excellent
AB	Abstract	0	0.00%	Excellent	0	0.00%	Excellent
RP	Corresponding author	12	<mark>0.86%</mark>	Good	555	<mark>28.77%</mark>	Poor
CR	Cited references	13	<mark>0.93%</mark>	Good	1929	100.00%	Completely missing
C1	Affiliation	13	<mark>0.93%</mark>	Good	47	2.44%	Good
DE	Keywords	136	<mark>9.75%</mark>	Good	244	12.65%	Acceptable
DI	DOI	149	10.68%	Acceptable	261	13.53%	Acceptable

Table 10. Data quality check for datasets extracted from WoS and Scopus.

As can be observed from Table 10, the data quality check revealed that both datasets show excellent data quality for most metadata fields, including author (AU), document type (DT), journal (SO), language (LA), publication year (PY), title (TI), total citation (TC), and abstract (AB), with no missing data. However, significant discrepancies exist in other fields:

- Science Categories (SC): Completely missing in the Scopus dataset, whereas the WoS dataset is complete.
- Corresponding Author (RP): The WoS dataset has a 0.86% missing rate (good), while the Scopus dataset has a 28.77% missing rate (poor).
- Cited References (CR): The WoS dataset has a 0.93% missing rate (good), but the Scopus dataset is completely missing this data.
- Affiliation (C1): The WoS dataset has a 0.93% missing rate (good), compared to a 2.44% missing rate (good) for the Scopus dataset.
- Keywords (DE): The WoS dataset has a 9.75% missing rate (good), while the Scopus dataset has a 12.65% missing rate (acceptable).
- DOI (DI): The WoS dataset has a 10.68% missing rate (acceptable), slightly better than the Scopus dataset with a 13.53% missing rate (acceptable).

In conclusion, while both datasets maintain excellent quality in several core fields, the Scopus dataset shows notable deficiencies in science categories, corresponding author, and cited references compared to the WoS dataset. This observation supports our initial decision to conduct the analysis in the present paper by using only the WoS database.

In the following, we present some results obtained through the use of the Scopus extracted dataset and compare the results with the ones obtained through the use of the WoS dataset.



First, in terms of annual scientific production—as presented in Figure 18—the same trend can be observed as in Figure 1, with the only difference represented by the inclu-

Figure 18. Annual scientific production (Scopus).

Second, in terms of the most relevant sources, the journal *Sustainability* (Switzerland) ranks first in both databases, with 65 publications in Scopus and 67 publications in WoS. The International Journal of Productivity and Performance Management holds second place in both Scopus and WoS, with 21 articles in each. This consistency indicates that the differences between the two databases are not significant enough to impact our results for the analyzed topic.

Furthermore, in Figure 19, we present a word cloud based on the authors' keywords from the Scopus database. The results obtained from the WoS database are similarly aligned, with "key performance indicators" being the central concept. Other prominent keywords include "performance measurement," "sustainability," "balanced scorecard," and "corporate social responsibility."



Figure 19. Word cloud based on authors' keywords—Scopus database.

Lastly, in Table 11, we present the top 7 most cited documents based on Scopus data. The ranking matches the one derived from the WoS source. The key difference is that the articles have more citations in Scopus, although the order remains the same.

First Author, Year, Journal, References	Total Citations	Total Citation (TC) per Year	Normalized TC
Sun LY., 2007, Academy of Management Journal, (Sun et al. 2007)	1056	58.67	19.95
Hanna R., 2011, Business Horizons, (Hanna et al. 2011)	1037	74.07	31.29
Milne M.J., 2013, Journal of Business Ethics, (Milne and Gray 2013)	758	63.17	25.66
Ceccagnoli M., 2012, MIS Quarterly, (Ceccagnoli et al. 2012)	655	50.38	18.26
López M.V., 2007, Journal of Business Ethics, (López et al. 2007)	528	29.33	9.98
Lee A.H.I., 2008, Expert Systems with Applications, (Lee et al. 2008)	499	29.35	15.38
Hermann B.G., 2007, Journal of Cleaner Production, (Hermann et al. 2007)	280	15.56	5.29

Table 11. Top 7 most cited documents globally.

Based on the above observations, it can be stated that similar trends are uncovered when using the Scopus database and WoS database for the field of key performance and risk indicators in Business.

5. Research Methodology

5.1. Conducting Bibliometric Analysis Using RStudio and VOSviewer

R is a programming language used for statistical computing and graphics, and RStudio is its integrated development environment (IDE). It offers a user-friendly interface that makes R more powerful and accessible to statisticians, data scientists, and academics (Hair et al. 2021).

A fundamental tool for data-driven research and decision-making, RStudio is excellent for data analysis, statistical testing, and data visualization. It is useful in many applications, from business analytics to academic research, thanks to its large package ecosystem and vibrant community (Gromping 2015).

Through comprehensive data visualization and reporting features, researchers may use RStudio to ensure repeatability, expedite workflow, and improve the impact and clarity of their findings.

Developed by Aria and Cuccurullo (2017), the Bibliometrix package in RStudio is an open-source tool used for quantitative and bibliometric research encompassing multiple analysis methods. Bibliometric analysis is considered a complex method applicable in any research area or discipline (Briner and Denyer 2012; Guler et al. 2016; Delcea et al. 2024), involving numerous complex stages. Several specialists recommend a workflow for scientific research analyses (Cobo et al. 2011; Zupic and Čater 2015), which includes five stages:

- i. Study design, which may involve the narrative description of research hypotheses;
- ii. Data collection using databases such as Scopus and Web of Science;
- iii. Analysis of the extracted data;
- iv. The use of specific techniques or tools for data visualization;
- v. Interpretation of the results.

In our bibliometric study, we address the five stages to conduct a bibliometric analysis on scientific research in the field of KPI and KRI usage in business.

In Figure 20, a methodological diagram illustrating the data selection process and the stages of bibliometric analysis used in this study is presented. This approach aids in visualizing how the data were selected and processed, facilitating understanding of the analytical process employed in our study.

Initially developed in 2010 (Meng et al. 2020), VOSviewer 1.6.20 is a software solution that allows for the creation of maps based on input data extracted from data sources such as Web of Science (WoS), useful for bibliometric analysis. Elements in these networks

can be linked by co-authorship, co-appearance, citation, bibliographic, or co-citation links (Martins et al. 2022).



Figure 20. Methodological flow.

Bibliometric analysis is an essential tool for evaluating and understanding the academic literature in a specific field (Delcea et al. 2023; Ionescu et al. 2024). By using bibliometric techniques, researchers can identify research trends, collaborations between authors and institutions, and the influence of certain works or authors in the field (Nica et al. 2024a). This type of analysis helps map the research landscape, highlighting emerging themes and knowledge gaps (Nica et al. 2024b). In our study, bibliometric analysis allows for us to discover research trends in the use of KPIs and KRIs in business, showing the evolution of academic and practical interest. Additionally, by identifying the most cited works and authors, we can determine the impact and influence of certain studies on the field, providing a framework for evaluating research quality and relevance. Moreover, the Bibliometrix platform enables us to conduct co-authorship and international collaboration network analyses, helping visualize research partnerships and highlighting cooperation between different institutions and countries. Another perspective we explore is the identification of emerging research themes. Through keyword co-occurrence, we can identify emerging research topics, offering insights into future study directions in the use of KPIs and KRIs. Thus, the results obtained from the bibliometric analysis can provide a solid basis for making informed decisions regarding research and development directions in the field of KPIs and KRIs.

5.2. Data Collection

In our study, we used the Web of Science (WoS) database of scientific publications to collect the data for analysis on the Bibliometrix platform using the *biblioshiny ()* library in RStudio. Although there are multiple data sources from which we can collect scientific research, such as Scopus, Crossref, or Dimensions, we decided to use the Web of Science platform because it has over 18 million documents, according to Visser et al. (2021). Although it is known that Scopus generally covers a larger database of scientific articles than WoS, this coverage varies greatly depending on the field of study (Thelwall et al. 2015; Mongeon and Paul-Hus 2016; Visser et al. 2021). For example, in Pranckute's study (Pranckute 2021), it was demonstrated that Scopus covers up to 99% of nursing journals that are also identified in WoS, whereas other studies focused on computer sciences cover only 63% of the documents retrieved by WoS and identified in Scopus (Bar-Ilan 2018). Moreover, to make an objective choice of the input data source, using the same queries, we extracted the database from Scopus to ensure that the results obtained from WoS are not skewed. It was observed that for our analyzed field, there were no significant differences, with both WoS and Scopus yielding the same results. Additionally, in the data quality stage, several criteria contain more missing data in Scopus than in WoS. This aspect cannot be generalized to any bibliometric research but only to our analyzed topic or other cases already identified in the scientific literature.

Regarding the dataset extraction, it is important to note that the WoS platform provides personalized access to data based on subscription. Consequently, as observed by (Liu 2019) and (F. Liu 2023), the results of bibliometric analysis are significantly influenced by the user's access to the ten indexes available on WoS. In this context, the authors recommend that bibliometric studies clearly specify the access levels users had to the WoS indexes (Liu 2019; F. Liu 2023).

The choice of the WoS platform is justified by its extensive coverage of a wide range of disciplines and its strong reputation within the scientific community, as highlighted in the literature (Cobo et al. 2015; Mulet-Forteza et al. 2018; Modak et al. 2019). Moreover, WoS is one of the few platforms that support data reading by both Bibliometrix and VOSviewer for datasets extracted based on the search criteria.

The first stage of the proposed five-step bibliometric analysis was conducted in the Introduction Section by defining four research hypotheses. The second stage involves data collection, as described in this subsection. Based on the established research hypotheses and the primary aim of our study, we selected the keywords "risk_indicator*", "performance_indicator*", and "business" for extracting scientific articles, with the asterisk "*" allowing for both singular and plural forms.

Within the WoS platform, using these keywords, we conducted the seven steps/queries described in Table 12.

In the first step, we searched for all scientific documents containing either the keyword "risk_indicator*" or "performance_indicator*" correlated with scientific articles that also contain the word "business" in the title. Thus, Query #1 extracted 64 scientific documents. In the second step, we performed the same query but for abstracts. Query #2 returned 2457 scientific documents. In Query #3, we conducted the same search but focused on

keywords, resulting in 231 documents. For Query #4, we used the "OR" operator to include all documents containing the keywords in the title, abstract, or keywords, resulting in 2602 scientific articles. The next query, #5, aimed to exclude all documents written in languages other than English, retaining only English-written articles. The English language is recognized as one of the most spoken international languages, facilitating the exchange of knowledge and collaboration among researchers from different countries. Bibliometric analysis requires a standardized approach to ensure data comparability and consistency. Translating articles from other languages would have been a complex process and could have introduced the risk of errors and ambiguities in interpreting the results. This query returned 2473 documents. Another criterion, Query #6, aimed to keep only scientific articles, resulting in 1489 documents. The final query, #7, excluded all articles from the year 2024, as they would not capture the full year's results or the scientific evolution of the last year. Thus, the analysis retained 1395 scientific documents.

Exploration Steps	Questions on Web of Science/Scopus	Description	Query	Query Number	Count WoS
1	Title	Contains specific keywords related to risk indicators or risk indicators in business context	(((TI = (risk_indicator*)) OR TI = (performance_indicator*))) AND TI = (business)	#1	64
2	Abstract	Contains specific keywords related to risk indicators or risk indicators in business context	(((AB = (risk_indicator*)) OR AB = (performance_indicator*))) AND AB = (business)	#2	2457
3	Keywords	Contains specific keywords related to risk indicators or risk indicators in business context	(((AK = (risk_indicator*)) OR AK = (performance_indicator*))) AND AK = (business)	#3	231
4	Title/Abstract/ Keywords	Contains specific keywords related to risk indicators or risk indicators in business context	#1 OR #2 OR #3	#4	2602
5	Language	Contains only documents written in English	(#4) AND LA = (English)	#5	2473
6	Document Type	Limited to articles	(#5) AND DT = (Article)	#6	1489
7	Year published	Excludes 2024	(#6) NOT PY = (2024)	#7	1395

Table 12. Data selection steps.

5.3. Quality Assessment of Bibliographic Data

In this stage of our analysis, we evaluated the completeness of various bibliographic metadata fields to ensure the quality and reliability of our dataset. This step is crucial, as it helps identify any gaps or missing information that could impact the accuracy and comprehensiveness of our bibliometric analysis (Delcea et al. 2023; Domenteanu et al. 2023; Sandu et al. 2024).

Table 13 evaluates the completeness of various bibliographic metadata fields in the dataset analyzed using Bibliometrix. We observe that the fields author (AU), document type (DT), journal (SO), language (LA), publication year (PY), science categories (SC), title (TI), total citations (TC), and abstract (AB) have no missing values, indicating complete and excellent-quality data. Good completeness with less than 1% missing is observed for corresponding author (RP) with 12 missing records, cited references (CR) with 13 missing records, and affiliation (C1) also with 13 missing observations. Additionally, for 149 records, DOI (DI) was not identified, indicating acceptable completeness. The metadata are mostly complete for the majority of fields, indicating good data quality for bibliometric anal-

ysis. However, the DOI field has significant gaps that may require attention to ensure comprehensive analysis.

Metadata	Description	Missing Counts WoS	Missing (%) WoS	Status WoS
AU	Author	0	0.00%	Excellent
DT	Document type	0	0.00%	Excellent
SO	Journal	0	0.00%	Excellent
LA	Language	0	0.00%	Excellent
PY	Publication year	0	0.00%	Excellent
SC	Science categories	0	0.00%	Excellent
TI	Title	0	0.00%	Excellent
TC	Total citation	0	0.00%	Excellent
AB	Abstract	0	0.00%	Excellent
RP	Corresponding author	12	0.86%	Good
CR	Cited references	13	0.93%	Good
C1	Affiliation	13	0.93%	Good
DE	Keywords	136	9.75%	Good
DI	DOI	149	10.68%	Acceptable

Table 13. Completeness of bibliographic metadata.

6. Conclusions

The study of key performance indicators (KPIs) and key risk indicators (KRIs) in business holds major relevance in the current context of organizational management. KPIs and KRIs are essential tools that help organizations monitor and improve performance, manage risks, and make informed decisions. In a world characterized by economic uncertainty and rapid changes, the ability to efficiently use these tools can determine an organization's long-term success. Our research contributes to a deeper understanding of how KPIs and KRIs can be implemented and managed to achieve strategic objectives and ensure organizational sustainability and competitiveness. Our bibliometric analysis revealed several key conclusions regarding the research landscape on KPIs and KRIs in business. Highly cited studies, such as those by Sun et al. (2007) and Hanna et al. (2011), emphasize the importance of KPIs in various organizational contexts, from human resource management to social media marketing. The frequent observations of the term's "sustainability" and "corporate social responsibility" in keywords indicate a growing concern for sustainability and the social impact of businesses. Clustering and co-occurrence network analysis highlighted that central themes include performance measurement, supply chain management, and corporate social responsibility. The clusters demonstrate the thematic diversity and relevance of these subjects in the specialized literature, indicating current research priorities.

Our results indicate that sustainability should be promoted. Considering the growing significance of the topic in both practice and research (Van De Ven et al. 2023), organizations should incorporate KPIs related to sustainability and corporate social responsibility into their strategic planning to ensure a sustained positive impact. Additionally, implementing and monitoring KRIs should be a priority (Mouatassim and Ibenrissoul 2015) to efficiently anticipate and manage operational risks. Moreover, promoting international research partnerships can lead to more robust and influential results, as evidenced in our analysis. Also, our findings confirm the initial hypotheses, demonstrating a positive correlation between the use of KPIs and the improvement of organizational performance, as well as the significant contribution of KRIs to reducing operational risks. The data support the hypothesis that international collaboration in KPI and KRI research leads to more influential studies. Furthermore, the effective management of a variety of KPIs and KRIs is important for organizational success and value growth. Based on the findings, we recommend that companies integrate both KPIs and KRIs into their management systems to improve performance and proactively manage risks. Researchers should further explore the interdependencies between these indicators to develop more robust models. Additionally, future research should focus not only on qualitative analysis but also on exploring the impact of KPIs and KRIs across various industries and cultural contexts. Comparative studies between industries could provide valuable insights.

Regarding the limitations of our research, although the bibliometric analysis provides valuable information and a detailed holistic perspective, an extended qualitative analysis could further enhance our understanding of research themes, especially regarding KRIs. Another potential limitation could be the dependence on specific data sources (e.g., WoS, Scopus) in bibliometric analysis, and that the effectiveness of Bibliometrix may vary depending on the comprehensiveness and quality of data obtained from these sources. Excluding certain databases or non-indexed publications could limit the coverage and representativeness of bibliometric analyses. Therefore, another direction for future research could focus on conducting a comprehensive analysis by combining multiple specific databases.

Organizations can use the findings of this study to enhance their strategic decisionmaking processes through the efficient implementation of key performance indicators and key risk indicators. Our study demonstrates that the use of KPIs is positively associated with improving organizational performance, while KRIs play an important role in operational risk management by early identification of potential threats. Organizations should integrate these tools into their management systems to gain a comprehensive view of operations and facilitate informed decision-making. Implementing KPIs for performance measurement and simultaneous monitoring of KRIs for risk management ensures organizational success in a dynamic and fluctuating business environment.

Author Contributions: Conceptualization, C.D.; Data curation, G.D.; Formal analysis, Ş.I., G.D., C.I. and C.D.; Investigation, Ş.I., G.D. and C.I.; Methodology, Ş.I. and C.D.; Project administration, C.D.; Resources, C.I. and C.D.; Software, Ş.I.; Supervision, C.D.; Validation, Ş.I., G.D. and C.I.; Visualization, G.D., C.I. and C.D.; Writing—original draft, Ş.I.; Writing—review and editing, G.D., C.I. and C.D. All authors have read and agreed to the published version of the manuscript.

Funding: The work was supported by a grant of the Romanian Ministry of Research and Innovation, project CNFIS-FDI-2024-F-0302—"Development and adaptation of collaborative processes in excellence research conducted at BUES, in the context of modern challenges brought by Open Science and Artificial Intelligence (eXROS)". This work was also supported by a grant from the Bucharest University of Economic Studies through the project "Analysis of the Economic Recovery and Resilience Process in Romania in the Context of Sustainable Development", contract number: 1353/10.06.2024.

Data Availability Statement: Data are included within this paper.

Conflicts of Interest: The authors declare no conflicts of interest.

References

- Anthony, Robert Newton. 1965. *Planning and Control Systems: A Framework for Analysis*. Cambridge: Division of Research, Graduate School of Business Administration, Harvard University.
- Aria, Massimo, and Corrado Cuccurullo. 2017. Bibliometrix: An R-Tool for Comprehensive Science Mapping Analysis. *Journal of Informetrics* 11: 959–75. [CrossRef]
- Armstrong, Michael. 2006. Performance Management: Key Strategies and Practical Guidelines. London: Kogan Place.
- Asiaei, Kaveh, and Nick Bontis. 2019. Translating Knowledge Management into Performance: The Role of Performance Measurement Systems. *Management Research Review* 43: 113–32. [CrossRef]
- Bar-Ilan, Judit. 2018. Tale of Three Databases: The Implication of Coverage Demonstrated for a Sample Query. *Frontiers in Research Metrics and Analytics* 3: 6. [CrossRef]
- Braglia, Marcello, Roberto Gabbrielli, Leonardo Marrazzini, and Luca Padellini. 2022. Key Performance Indicators and Industry 4.0—A Structured Approach for Monitoring the Implementation of Digital Technologies. *Procedia Computer Science* 200: 1626–35. [CrossRef]
- Briner, Rob B., and David Denyer. 2012. Systematic Review and Evidence Synthesis as a Practice and Scholarship Tool. In *The Oxford Handbook of Evidence-Based Management*, 1st ed. Edited by Denise M. Rousseau. Oxford: Oxford University Press, pp. 112–29. [CrossRef]
- Brühl, Volker. 2023. The Green Asset Ratio (GAR): A New Key Performance Indicator for Credit Institutions. *Eurasian Economic Review* 13: 57–83. [CrossRef]

- Burlea-Schiopoiu, Adriana, and Koudoua Ferhati. 2020. The Managerial Implications of the Key Performance Indicators in Healthcare Sector: A Cluster Analysis. *Healthcare* 9: 19. [CrossRef] [PubMed]
- Ceccagnoli, Marco, Chris Forman, Peng Huang, and D. J. Wu. 2012. Co-Creation of Value in a Platform Ecosystem: The Case of Enterprise Software. *MIS Quarterly* 36: 263–90. [CrossRef]
- Cernisevs, Olegs, Yelena Popova, and Dmitrijs Cernisevs. 2023. Risk-Based Approach for Selecting Company Key Performance Indicator in an Example of Financial Services. *Informatics* 10: 54. [CrossRef]
- Chen, Yili, Congdong Li, and Han Wang. 2022. Big Data and Predictive Analytics for Business Intelligence: A Bibliographic Study (2000–2021). *Forecasting* 4: 767–86. [CrossRef]
- Cobo, M. J., A. G. López-Herrera, E. Herrera-Viedma, and F. Herrera. 2011. Science Mapping Software Tools: Review, Analysis, and Cooperative Study among Tools. *Journal of the American Society for Information Science and Technology* 62: 1382–402. [CrossRef]
- Cobo, M. J., M. A. Martínez, M. Gutiérrez-Salcedo, H. Fujita, and E. Herrera-Viedma. 2015. 25years at Knowledge-Based Systems: A Bibliometric Analysis. *Knowledge-Based Systems* 80: 3–13. [CrossRef]
- Cruz Villazón, Carolina, Leonardo Sastoque Pinilla, José Ramón Otegi Olaso, Nerea Toledo Gandarias, and Norberto López De Lacalle. 2020. Identification of Key Performance Indicators in Project-Based Organisations through the Lean Approach. *Sustainability* 12: 5977. [CrossRef]
- Dahal, Rewan Kumar, Dipendra Karki, Binod Ghimire, Rajesh Gurung, and Surendra Prasad Joshi. 2024. Direction of Business Performance Metrics: A Bibliometric and Thematic Review. Available online: https://www.researchgate.net/profile/Rewan-Dahal/publication/380104254_Direction_of_Business_Performance_Metrics_A_Bibliometric_and_Thematic_Review/links/66 2b4a4d08aa54017ac17490/Direction-of-Business-Performance-Metrics-A-Bibliometric-and-Thematic-Review.pdf (accessed on 8 April 2024).
- Damanpour, Fariborz, and Deepa Aravind. 2012. Organizational Structure and Innovation Revisited. In *Handbook of Organizational Creativity*. Amsterdam: Elsevier, pp. 483–513. [CrossRef]
- Davenport, Thomas, and Ronanki Rajeev. 2018. Artificial Intelligence for the Real World. Harvard Business Review. Available online: https://hbr.org/2018/01/artificial-intelligence-for-the-real-world (accessed on 8 April 2024).
- Delcea, Camelia, Saad Ahmed Javed, Margareta-Stela Florescu, Corina Ioanas, and Liviu-Adrian Cotfas. 2023. 35 Years of Grey System Theory in Economics and Education. *Kybernetes, ahead-of-print*. [CrossRef]
- Delcea, Camelia, Simona-Vasilica Oprea, Adrian Domenteanu, Adela Bara, and Liviu-Adrian Cotfas. 2024. Insights from Scientific Publications on Energy Communities. *Under Journal Review*.
- Deverell, Edward, and Adrian Ganic. 2024. Crisis and Performance: A Contingency Approach to Performance Indicators. International Journal of Disaster Risk Reduction 105: 104417. [CrossRef]
- Domenteanu, Adrian, Camelia Delcea, Nora Chiriță, and Corina Ioanăș. 2023. From Data to Insights: A Bibliometric Assessment of Agent-Based Modeling Applications in Transportation. *Applied Sciences* 13: 12693. [CrossRef]
- Eccles, Robert, Michael Krzus, and Sydney Ribot. 2014. *The Integrated Reporting Movement: Meaning, Momentum, Motives, and Materiality.* New York: Wiley.
- Elyasiani, Elyas, and Jingyi (Jane) Jia. 2019. Relative Performance and Systemic Risk Contributions of Small and Large Banks during the Financial Crisis. *The Quarterly Review of Economics and Finance* 74: 220–41. [CrossRef]
- Ennis, Kay, and Denis Harrington. 1999. Quality Management in Irish Health Care. *International Journal of Health Care Quality Assurance* 12: 232–44. [CrossRef]
- Filho, Pedro Cezar Saccol, Julio Cezar Mairesse Siluk, Cláudia De Freitas Michelin, Paula Donaduzzi Rigo, Alvaro Luiz Neuenfeldt Júnior, Carmen Brum Rosa, and Wesley Vieira Da Silva. 2023. A Measurement Tool for the Competitiveness of Startups' Innovation Ecosystem. *Journal of the Knowledge Economy* 2023: 1–28. [CrossRef]
- Franklin-Johnson, Elizabeth, Frank Figge, and Louise Canning. 2016. Resource Duration as a Managerial Indicator for Circular Economy Performance. *Journal of Cleaner Production* 133: 589–98. [CrossRef]
- Frederico, Guilherme Francisco, Jose Arturo Garza-Reyes, Anil Kumar, and Vikas Kumar. 2020. Performance Measurement for Supply Chains in the Industry 4.0 Era: A Balanced Scorecard Approach. *International Journal of Productivity and Performance Management* 70: 789–807. [CrossRef]
- Gackowiec, Paulina, Marta Podobińska-Staniec, Edyta Brzychczy, Christopher Kühlbach, and Toyga Özver. 2020. Review of Key Performance Indicators for Process Monitoring in the Mining Industry. *Energies* 13: 5169. [CrossRef]
- Geissdoerfer, Martin, Paulo Savaget, Nancy M. P. Bocken, and Erik Jan Hultink. 2017. The Circular Economy—A New Sustainability Paradigm? *Journal of Cleaner Production* 143: 757–68. [CrossRef]
- Giudici, Paolo, Mattia Centurelli, and Stefano Turchetta. 2024. Artificial Intelligence Risk Measurement. *Expert Systems with Applications* 235: 121220. [CrossRef]
- Gromping, Ulrike. 2015. Using *R* and RStudio for Data Management, Statistical Analysis and Graphics (2nd Edition). *Journal of Statistical Software* 68. [CrossRef]
- Guler, Arzu Tugce, Cathelijn J. F. Waaijer, and Magnus Palmblad. 2016. Scientific Workflows for Bibliometrics. *Scientometrics* 107: 385–98. [CrossRef]

- Hair, Joseph F., G. Tomas M. Hult, Christian M. Ringle, Marko Sarstedt, Nicholas P. Danks, and Soumya Ray. 2021. Overview of R and RStudio. In *Partial Least Squares Structural Equation Modeling (PLS-SEM) Using R*. Edited by Joseph F. Hair, G. Tomas M. Hult, Christian M. Ringle, Marko Sarstedt, Nicholas P. Danks and Soumya Ray. Classroom Companion: Business. Cham: Springer International Publishing, pp. 31–47. [CrossRef]
- Hanna, Richard, Andrew Rohm, and Victoria L. Crittenden. 2011. We're All Connected: The Power of the Social Media Ecosystem. *Business Horizons* 54: 265–73. [CrossRef]
- Hermann, B. G., C. Kroeze, and W. Jawjit. 2007. Assessing Environmental Performance by Combining Life Cycle Assessment, Multi-Criteria Analysis and Environmental Performance Indicators. *Journal of Cleaner Production* 15: 1787–96. [CrossRef]
- Hess, Thomas, Christian Matt, Alexander Benlian, and Florian Wiesböck. 2020. *Options for Formulating a Digital Transformation Strategy*, 5th ed. New York: Routledge.
- Ionescu, Ștefan, Camelia Delcea, Nora Chiriță, and Ionuț Nica. 2024. Exploring the Use of Artificial Intelligence in Agent-Based Modeling Applications: A Bibliometric Study. *Algorithms* 17: 21. [CrossRef]
- Jawad, Zainab Nadhim, and Villányi Balázs. 2024. Machine Learning-Driven Optimization of Enterprise Resource Planning (ERP) Systems: A Comprehensive Review. *Beni-Suef University Journal of Basic and Applied Sciences* 13: 4. [CrossRef]
- Kaplan, Robert, and David Norton. 1996. The Balanced Scorecard: Translating Strategy into Action. Boston: Harvard Business School Press. Kavadis, Nikolaos, Niels Hermes, Jana Oehmichen, Alessandro Zattoni, and Stav Fainshmidt. 2024. Sustainable Value Creation in Multinational Enterprises: The Role of Corporate Governance Actors. Journal of World Business 59: 101503. [CrossRef]
- Khalifa, Mohamed, and Parwaiz Khalid. 2015. Developing Strategic Health Care Key Performance Indicators: A Case Study on a Tertiary Care Hospital. In Proceedings of the 5th International Conference on Current and Future Trends of Information and Communication Technologies in Healthcare (ICTH 2015). *Procedia Computer Science* 63: 459–66. [CrossRef]
- Lee, Amy H. I., Wen-Chin Chen, and Ching-Jan Chang. 2008. A Fuzzy AHP and BSC Approach for Evaluating Performance of IT Department in the Manufacturing Industry in Taiwan. *Expert Systems with Applications* 34: 96–107. [CrossRef]
- Li, Han, Hicham Johra, Flavia De Andrade Pereira, Tianzhen Hong, Jérôme Le Dréau, Anthony Maturo, Mingjun Wei, Yapan Liu, Ali Saberi-Derakhtenjani, Zoltan Nagy, and et al. 2023. Data-Driven Key Performance Indicators and Datasets for Building Energy Flexibility: A Review and Perspectives. *Applied Energy* 343: 121217. [CrossRef]
- Lihachev, V. N. 2024. Comparative analysis of key indicators to assess production efficiency. *Social'no-Ekonomiceskoe Upravlenie: Teoria i Praktika* 20: 47–60. [CrossRef]
- Liu, Fang. 2023. Retrieval Strategy and Possible Explanations for the Abnormal Growth of Research Publications: Re-Evaluating a Bibliometric Analysis of Climate Change. *Scientometrics* 128: 853–59. [CrossRef]
- Liu, Weishu. 2019. The Data Source of This Study Is Web of Science Core Collection? Not Enough. *Scientometrics* 121: 1815–24. [CrossRef]
- López, M. Victoria, Arminda Garcia, and Lazaro Rodriguez. 2007. Sustainable Development and Corporate Performance: A Study Based on the Dow Jones Sustainability Index. *Journal of Business Ethics* 75: 285–300. [CrossRef]
- Marrucci, Luca, Tiberio Daddi, and Fabio Iraldo. 2024. Creating Environmental Performance Indicators to Assess Corporate Sustainability and Reward Employees. *Ecological Indicators* 158: 111489. [CrossRef]
- Martins, José, Ramiro Gonçalves, and Frederico Branco. 2022. A Bibliometric Analysis and Visualization of E-Learning Adoption Using VOSviewer. Universal Access in the Information Society 2022: 1–15. [CrossRef]
- McCullough, Brian P., and Galen T. Trail. 2023. Assessing Key Performance Indicators of Corporate Social Responsibility Initiatives in Sport. *European Sport Management Quarterly* 23: 82–103. [CrossRef]
- Melançon, Gabrielle Gauthier, Philippe Grangier, Eric Prescott-Gagnon, Emmanuel Sabourin, and Louis-Martin Rousseau. 2021. A Machine Learning-Based System for Predicting Service-Level Failures in Supply Chains. *INFORMS Journal on Applied Analytics* 51: 200–12. [CrossRef]
- Meng, Lingchao, Kuo-Hsun Wen, Richard Brewin, and Qiong Wu. 2020. Knowledge Atlas on the Relationship between Urban Street Space and Residents' Health—A Bibliometric Analysis Based on VOSviewer and CiteSpace. *Sustainability* 12: 2384. [CrossRef]
- Milne, Markus J., and Rob Gray. 2013. W(h)Ither Ecology? The Triple Bottom Line, the Global Reporting Initiative, and Corporate Sustainability Reporting. *Journal of Business Ethics* 118: 13–29. [CrossRef]
- Modak, Nikunja M., José M. Merigó, Richard Weber, Felipe Manzor, and Juan De Dios Ortúzar. 2019. Fifty Years of Transportation Research Journals: A Bibliometric Overview. *Transportation Research Part A: Policy and Practice* 120: 188–223. [CrossRef]
- Mongeon, Philippe, and Adèle Paul-Hus. 2016. The Journal Coverage of Web of Science and Scopus: A Comparative Analysis. *Scientometrics* 106: 213–28. [CrossRef]
- Mouatassim, Hajar, and Abdelmajid Ibenrissoul. 2015. Proposal for an Implementation Methodology of Key Risk Indicators System: Case of Investment Management Process in Moroccan Asset Management Company. *Journal of Financial Risk Management* 4: 187–205. [CrossRef]
- Mulet-Forteza, Carles, Onofre Martorell-Cunill, José M. Merigó, Juanabel Genovart-Balaguer, and Emilio Mauleon-Mendez. 2018. Twenty Five Years of the Journal of Travel & Tourism Marketing: A Bibliometric Ranking. Journal of Travel & Tourism Marketing 35: 1201–21. [CrossRef]
- Munmun, Sajneen Akter, Md. Abdul Moktadir, Sunil Tiwari, Mohammad Zoynul Abedin, and Charbel Jose Chiappetta Jabbour. 2023. Investigation of Key Performance Indicators for Performance Management of the Manufacturing Industry in the Era of the COVID-19 Pandemic. *Annals of Operations Research* 2023: 1–44. [CrossRef]

- Nica, Ionuț, and Ștefan Ionescu. 2021. Business Analysis and Measurement of Enterprise Performance: Digital Transformation of Economic Systems from a Holistic Business Perspective. Chisinau: Lambert Academic Publishing.
- Nica, Ionuț, Camelia Delcea, and Nora Chiriță. 2024a. Mathematical Patterns in Fuzzy Logic and Artificial Intelligence for Financial Analysis: A Bibliometric Study. *Mathematics* 12: 782. [CrossRef]
- Nica, Ionuț, Camelia Delcea, Nora Chiriță, and Ștefan Ionescu. 2024b. Quantifying Impact, Uncovering Trends: A Comprehensive Bibliometric Analysis of Shadow Banking and Financial Contagion Dynamics. *International Journal of Financial Studies* 12: 25. [CrossRef]
- Niță, Gabriel, Larisa Găbudeanu, Cosmin Constantin Cernaeanu, Gabriel Mărgărit Raicu, and Mircea Constantin Șcheau. 2023. Artificial Intelligence and Key Risk Indicators in Cyber Frauds Prevention. *Journal of Financial and Monetary Economics, Centre of Financial and Monetary Research "Victor Slavescu"* 11: 92–111.
- Pindyck, Robert S. 2013. Climate Change Policy: What Do the Models Tell Us. Journal of Economic Literature 51: 860–72. [CrossRef]
- Pranckutė, Raminta. 2021. Web of Science (WoS) and Scopus: The Titans of Bibliographic Information in Today's Academic World. *Publications* 9: 12. [CrossRef]
- Qiao, Wanzhen, Yanbing Ju, Peiwu Dong, and Robert L.K. Tiong. 2024. How to Realize Value Creation of Digital Transformation? A System Dynamics Model. *Expert Systems with Applications* 244: 122667. [CrossRef]
- Salamah, Anas A., Shahizan Hassan, Ali Aljaafreh, Walaa A. Zabadi, Mohammad Ali AlQudah, Naeem Hayat, Abdullah Al Mamun, and Thavamaran Kanesan. 2022. Customer Retention through Service Quality and Satisfaction: Using Hybrid SEM-Neural Network Analysis Approach. *Heliyon* 8: e10570. [CrossRef]
- Sandu, Andra, Liviu-Adrian Cotfas, Aurelia Stănescu, and Camelia Delcea. 2024. Guiding Urban Decision-Making: A Study on Recommender Systems in Smart Cities. *Electronics* 13: 2151. [CrossRef]
- Santos, Eulalia, Vasco Tavares, Fernando Oliveira Tavares, and Vanessa Ratten. 2022. How Is Risk Different in Family and Non-Family Businesses? A Comparative Statistical Analysis during the COVID-19 Pandemic. *Journal of Family Business Management* 12: 1113–30. [CrossRef]
- Setijono, Djoko, and Jens J. Dahlgaard. 2007. Customer Value as a Key Performance Indicator (KPI) and a Key Improvement Indicator (KII). *Measuring Business Excellence* 11: 44–61. [CrossRef]
- Silva, Carlos Santos, Diana Vieira Fernandes, Ricardo Gomes, Francisco Pires Costa, Ligia Pinto, Sabrina Scuri, Andre Brito, Baltazar Nunes, and Susana Pereira Silva. 2024. CLIMAEXTREMO: A New Risk Indicator for the Health Risk to Building Occupants during Extreme Weather Events in Portugal. *Sustainability* 16: 5171. [CrossRef]
- Sun, Li-Yun, Samuel Aryee, and Kenneth S. Law. 2007. High-Performance Human Resource Practices, Citizenship Behavior, and Organizational Performance: A Relational Perspective. *Academy of Management Journal* 50: 558–77. [CrossRef]
- Tambare, Parkash, Chandrashekhar Meshram, Cheng-Chi Lee, Rakesh Jagdish Ramteke, and Agbotiname Lucky Imoize. 2021. Performance Measurement System and Quality Management in Data-Driven Industry 4.0: A Review. Sensors 22: 224. [CrossRef] [PubMed]
- Tasheva, Sabina, and Bo Bernhard Nielsen. 2022. The Role of Global Dynamic Managerial Capability in the Pursuit of International Strategy and Superior Performance. *Journal of International Business Studies* 53: 689–708. [CrossRef]
- Tavares, Fernando, Eulália Santos, Mafalda Venâncio de Vasconcelos, and Vasco Capela Tavares. 2023. A Statistical Analysis of Companies' Financing Strategies in Portugal during the COVID-19 Pandemic. *Journal of Risk and Financial Management* 16: 116. [CrossRef]
- Tavares, Fernando, Eulália Santos, Margarida Freitas Oliveira, and Luís Almeida. 2024. Determinants of Corporate Indebtedness in Portugal: An Analysis of Financial Behaviour Clusters. *Risks* 12: 91. [CrossRef]
- Teece, David J. 2018. Business Models and Dynamic Capabilities. Long Range Planning 51: 40-49. [CrossRef]
- Thelwall, Mike, Kayvan Kousha, Paul Wouters, Ludo Waltman, Sarah De Rijcke, Alex Rushforth, and Thomas Franssen. 2015. *The Metric Tide: Literature Review*. Bristol: HEFCE. [CrossRef]
- Tseng, Shu-Feng, Chien-Chih Yu, and Chun-Hao Chiang. 2011. Developing Key Risk Indicators for B2B International Internet Banking: A RM-BSC Approach. NCCU Academic Hub. Available online: https://ah.lib.nccu.edu.tw/item?item_id=53334 (accessed on 8 April 2024).
- Turner, Rodney, and Roxanne Zolin. 2012. Forecasting Success on Large Projects: Developing Reliable Scales to Predict Multiple Perspectives by Multiple Stakeholders over Multiple Time Frames. *Project Management Journal* 43: 87–99. [CrossRef]
- Marginson, Simon. 2016. The Dream Is Over: The Crisis of Clark Kerr's California Idea of Higher Education. Berkeley: University of California Press. [CrossRef]
- Valenzuela-Fernández, Leslier, Carolina Nicolas, Jaime Gil-Lafuente, and José M. Merigó. 2016. Fuzzy Indicators for Customer Retention. International Journal of Engineering Business Management 8: 184797901667052. [CrossRef]
- Van De Ven, Montijn, Paola Lara Machado, Alexia Athanasopoulou, Banu Aysolmaz, and Oktay Turetken. 2023. Key Performance Indicators for Business Models: A Systematic Review and Catalog. *Information Systems and E-Business Management* 21: 753–94. [CrossRef]
- Varsei, Mohsen, Claudine Soosay, Behnam Fahimnia, and Joseph Sarkis. 2014. Framing Sustainability Performance of Supply Chains with Multidimensional Indicators. Edited by Dr Stefan Schaltegger, Prof Roger Burritt. Supply Chain Management: An International Journal 19: 242–57. [CrossRef]

- Visser, Martijn, Nees Jan Van Eck, and Ludo Waltman. 2021. Large-Scale Comparison of Bibliographic Data Sources: Scopus, Web of Science, Dimensions, Crossref, and Microsoft Academic. *Quantitative Science Studies* 2: 20–41. [CrossRef]
- Wixom, Barbara, and Hugh Watson. 2010. The BI-Based Organization. *International Journal of Business Intelligence Research* 1: 13–28. [CrossRef]
- Zupic, Ivan, and Tomaž Čater. 2015. Bibliometric Methods in Management and Organization. Organizational Research Methods 18: 429–72. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.



Article Impact of Audit Fees on Earnings Management and Financial Risk: An Analysis of Corporate Finance Practices

Abbas Ali Daryaei ¹, Davood Askarany ^{2,*} and Yasin Fattahi ¹

- ¹ Department of Accounting, Imam Khomeini International University, Qazvin 34148-96818, Iran; a.a.daryaei@soc.ikiu.ac.ir (A.A.D.); shirkohajar73@gmail.com (Y.F.)
- ² Department of Accounting & Finance, Business School, The University of Auckland, Auckland 1010, New Zealand
- * Correspondence: d.askarany@auckland.ac.nz; Tel.: +64-9-9235785

Abstract: This study employs a robust quantitative ex post facto research design to investigate the complex relationship between audit fees and earnings management. The financial information of 164 firms admitted to the Tehran Stock Exchange (TSE) was used from 2010 to 2019 (pre-COVID period) to achieve the research goal. Analysing data from the Tehran Stock Exchange firms, the study uncovers an inverted U-shaped relationship between audit fees and earnings management. This suggests that moderate audit fees can lead to higher earnings management. Key contributions of this paper include highlighting the role of audit fees in influencing financial reporting quality and risk management, providing empirical evidence on the asymmetric effects of normal and abnormal audit fees on earnings management, and emphasising the need for balanced audit fee structures to ensure financial transparency and mitigate risk. The findings offer valuable insights for academics, practitioners, and policymakers in understanding the nuances of audit fees and their impact on corporate financial practices. This study advances the literature on financial risk management and corporate finance. It emphasises the importance of balanced audit fee structures for management teams, auditors, and policymakers to ensure transparent financial reporting practices.



1. Introduction

As a measure of financial information quality, accounting comparability allows users to identify similarities and differences between items in financial statements and more rationally assess various lending-based investment opportunities (FASB 2010). For information comparability, similar items should be the same, and different items should seem different (Barth et al. 2018). The accounting comparability of financial information is so crucial that concept statement No. 8 of the Financial Accounting Standards Board addresses the increased financial information comparability as one of the most critical reasons that financial reporting standards are needed (Choi et al. 2019). Paying compensation and bonuses to managers as part of their wages is one of the most widely used practices motivating them to increase shareholders' wealth (Henderson and Fredrickson 2001). Compensation for firm managers usually includes fixed salary, cash, and non-cash bonuses. Theoretically, cash bonuses paid to managers are expected to suit their performance, resulting in more shareholder benefits (Zhang et al. 2018). Healy (1985) shows that compensation plans encourage managers to employ earnings management practices to maximise rewards. Therefore, managers can manipulate earnings to increase their rewards. A solution to this problem is to improve accounting information comparability. This concept can be viewed from internal and external aspects. Internal accounting comparability refers to a firm's consistency of practices over years of activity, whereas external accounting comparability shows a firm's status among peer firms (Lobo et al. 2018). Companies whose accounting systems are



Citation: Daryaei, Abbas Ali, Davood Askarany, and Yasin Fattahi. 2024. Impact of Audit Fees on Earnings Management and Financial Risk: An Analysis of Corporate Finance Practices. *Risks* 12: 123. https:// doi.org/10.3390/risks12080123

Academic Editors: Eulália Mota Santos and Margarida Freitas Oliveira

Received: 28 May 2024 Revised: 26 July 2024 Accepted: 31 July 2024 Published: 2 August 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). similar to their peer firms are more likely to adjust their managers' compensation contracts based on accounting profits.

Many studies have shown that audit fees affect the quality of financial reporting (Gu and Hu 2015; Gandía and Huguet 2021). The findings indicate that audit fees may affect the quality of financial reporting in different economic settings. Nevertheless, it is noteworthy that accounting comparability, as a measure of financial reporting quality, may be influenced by audit fees as a measure of audit quality. As mentioned earlier, since accounting comparability can affect managers' compensation, it can be argued that audit fees, as a proxy for audit quality, can influence the relationship between accounting comparability and board compensation. This topic has not been investigated in previous research and has been neglected to some extent. This research examines the critical issue of audit fees and the relationship between comparability and compensation. Of course, looking at the previous study, it can be seen that the relationship between comparability and compensation has already been investigated by Lobo et al. (2018), Nam (2020), and Fattahi et al. (2021). However, in the scope of these studies, the effect of audit fees has not been investigated. Therefore, by taking the help of and following the previous literature, the current research tries to examine the existing literature in the field of financial reporting and the role of auditors' fees in its change.

On the other hand, Frankel et al. (2002) state that companies paying higher audit fees exhibit a lower level of earnings management, suggesting the additional fee charged by auditing firms may increase the auditing quality. Ferguson et al. (2004) showed that non-audit services fees negatively affect the accountant's independence. Their findings also reveal a positive relationship between the non-audit services fee and discretionary accruals. Antle et al. (2006) try to justify the relationship between audit fees and earnings management and argue that various economic factors and conditions may affect this relationship. Such studies indicate that audit fees can affect earnings management.

Given the above, in addition to examining the moderating effect of audit fees on the relationship between comparability and rewards, this study investigates the impact of audit fees on earnings management (actual and real) in a relatively risky environment with a high inflation rate. Previous research has investigated the relationship between audit fees and earnings management in other countries. For example, Gandía and Huguet (2021) investigated the relationship between remuneration and earnings management from the perspective of the type of auditor. Also, Donatella et al. (2019) investigated the impact of audit fees on earnings management in Swedish companies. In addition, Ben Abdelaziz et al. (2022) have conducted a similar study but introduced audit fees as a factor in reducing earnings management, which was followed by other studies such as Shehadeh et al. (2024) and Santos Jaén et al. (2023). Previous studies' primary focus has been examining the direct relationship between these two variables. However, the current research is the first conducted in Iran's economic environment, which has a high inflation rate and risky financial conditions. The audit fee is divided into normal and abnormal fees, and the asymmetric influences on earnings management are investigated. The study further investigates the relationship between earnings management, as the output of the financial reporting system (Wen et al. 2023), and compensation, which is influenced by the financial reporting system, with two measures of the financial reporting system, i.e., the comparability and audit fee.

The remainder of this paper is organised as follows: Section 2 presents the background and the development of the hypotheses. Section 3 outlines the data collection and describes the sample and research variables. Empirical models and econometrics results are shown in Section 4. Finally, discussion, conclusions and implications are provided in Section 5.

2. Literature Review

The audit fee is an essential indicator for assessing organisations' financial statements (Christensen et al. 2021). Simunic (1980) suggests a relationship between audit fees and audit risk. Many studies have shown a direct association between audit risk and fees (Sonu

et al. 2017). If comparability improves the quality of the accounting managerial estimates, the misstatement risk should be reduced irrespective of other factors. Comparability, which leads to the quality reporting of financial information, should decrease the inherent audit risk and reduce the audit fee.

On the other hand, the complexity of the entity is one of the factors that causes an increase in the audit fee (Lee et al. 2024). Entities that are more complex in operation and structure could pay more to the managers to control the process (Kalelkar et al. 2024). Furthermore, managers who gain a more significant earnings margin for the entity deserve more compensation. When the firm's operation is large-scale and complex, demand for supervision in financial reporting increases. Entities with complex procedures require a high level of audit services. As a result, they pay higher fees to the audit firms, which suggests that as the complexity of the entity's operation increases, the management's compensation amount also rises (Saleh and Ragab 2023).

Audit fees could be considered as a measure of an entity's reporting complexities (Wysocki 2010). Investors, shareholders, and other stakeholders pursue quality information and a lower level of risk simultaneously, and the management, as their representative, tends to maximise its compensation by improving its plans and performance and decreasing information uncertainty risk. Audits play an overriding role in bridging these groups together so that receiving a greater audit fee appears as an assurance concerning the management's performance and compensation by the manager (Utomo and Machmuddah 2023). So, it could be predicted that the audit fee will decrease as comparability increases, resulting in a decline in executive compensation. However, it can be argued that an increase in comparability could lead to a rise in executive compensation, highlighting the contradictory impact of audit fees on the relation between the comparability of the accounting information and executive compensation. Wysocki (2010) argues that if the settlements concerning the compensation are determined appropriately, the management will be motivated to perform better, which may reduce the need for audit activities.

Concerning this argument, the negative association between audit fees and compensation could be predicted. However, the audit fee could be another proxy connected with earnings management and financial reporting quality, which justifies the negative association between the comparability of accounting information and executive compensation.

To pursue a better analysis of the modifier and contradictory influence of the audit fee, Simunic (1980) considered normal and abnormal audit fees. Ridzky and Fitriany (2022) divide audit fees into 'normal' and 'abnormal' to examine the impact of audit fees on audit quality. They define an abnormal audit fee as the difference between the actual and normal fees (which could be different in different cities and countries) paid by the auditee to the external auditor. They further divide the abnormal audit fees into 'premium' and 'discount' audit fees. They describe the premium audit fee as the actual audit fee above the normal and call it a negative abnormal audit fee (discount) if it is below the normal. Given the above, we propose our first hypothesis as follows:

Hypothesis 1 (H1): Audit fees have a differential impact on the relationship between accounting comparability and executive compensation performance.

Hypothesis 1a (H1a): Normal audit fees positively moderate the relationship between accounting comparability and executive compensation performance. This implies that standard audit fees contribute to improved accounting comparability, thereby enhancing the alignment of executive compensation with firm performance.

Hypothesis 1b (H1b): Abnormal audit fees (premium and discount) affect the relationship between accounting comparability and executive compensation performance. Specifically, premium audit fees (above the normal) are expected to strengthen this relationship by ensuring higher quality. In contrast, discount audit fees (below the normal) may weaken the relationship due to compromised audit quality.

Previous studies have examined the relationship between audit fees and earnings management (Gu and Hu 2015; Gandía and Huguet 2021). However, these studies produced mixed results. For instance, Frankel et al. (2002) indicated that firms that pay a higher audit fee reflect a lower level of earnings management, which refers to the additional fee required by audit firms that enhances the audit quality. Ferguson et al. (2004) demonstrated that fees related to non-audit services have a negative impact on an auditor's independence. Their findings indicated a positive association between fees related to non-audit services and discretionary accruals. It is also argued that improved comparability increases the quality of financial information reporting, which can reduce the auditor's assessment of the audit's inherent risk, thus reducing the audit fee. So, we look for a probable asymmetric association between audit fees and earnings management to justify the relationship between comparability, earnings management, and audit fees.

Audit firms determine the required fee amount based on their clients' observed features, such as size, complexity, risk structure of the firm, etc., which is called an audit fee. Conversely, any additional fee received based on the relationship between the audit firm and the clients that is not related to the client's features, such as size, complexity, etc., is defined as the abnormal level of the audit fee. Antle et al. (2006) provide various reasons regarding the correlation between audit fees and earnings management. They believe that different economic factors can affect this correlation; therefore, different results ensue in other economic environments and conditions. While previous research has concentrated on total audit fees and their association with earnings management, recent studies follow a new method by dividing the entire fees into two levels, normal and abnormal, and investigating the correlation between the abnormal level of audit fees and earnings management (Mitra et al. 2009). However, these studies have produced mixed results. For instance, Asthana and Boone (2012) reported that both abnormal levels of positive and negative audit fees negatively correlate with the quality of financial reports. In contrast, Blankley et al. (2012) and Eshleman and Guo (2014) indicated that the positive level of abnormal audit fees is positively related to the quality of financial reporting.

Using five theories, Antle et al. (2006) assessed the relationship between audit fees and earnings management. Basic economic theory suggests that providing audit and non-audit services to the client creates a financial bond between the audit firm and the client, which can jeopardise the auditor's independence. The supply and demand theories concerning services indicate that increased discretionary accruals lead to increased demand for audit services. For instance, a higher level of abnormal accruals is related to a higher probability of future litigation risk; therefore, auditors perform additional audit procedures to reduce these risks. Although the theory of bribery suggests that auditors perform audit procedures in line with client's expectations to ensure their position as their clients' audit firm in future periods, this point increases auditors' chances of receiving non-audit service fees from clients; therefore, they receive additional fees (Sarhan and Cowton 2024). The theory of bias suggests a bias in audit procedures in favour of the client when a well-built relationship is built between the auditor and the client due to a lengthy tenure period. Finally, the theory of production assumes that the level of abnormal accruals is reduced due to the provision of non-audit services. This means that the rise in the effectiveness or efficiency of clients' operations through non-audit services puts constraints on the flexibility of managers in earnings management. This literature demonstrates that the impact of audit fees on the earnings management process would differ depending on the condition of the auditor and the client. This means there is an asymmetric association between audit fees and earnings management. Figure 1 shows the conceptual model of the study.

Given the theoretical and empirical evidence, our second hypothesis about audit fees on earnings management is specified as follows:

Hypothesis 2 (H2): The relationship between audit fees and earnings management is asymmetric and contingent on the nature of the audit fees (normal versus abnormal).

Hypothesis 2a (H2a): Normal audit fees are associated with reduced earnings management. This suggests that standard audit fees provide adequate resources for auditors to conduct thorough audits, thereby mitigating earnings management practices.

Hypothesis 2b (H2b): Abnormal audit fees have an asymmetric effect on earnings management. Premium audit fees (above the normal) will likely reduce earnings management due to increased scrutiny and higher audit quality. In contrast, discount audit fees (below the normal) may be associated with increased earnings management due to potential reductions in audit quality and auditor independence.



Figure 1. Research conceptual model.

3. Methodology

The present study employs a quantitative, ex post facto design. The population consists of all the companies listed on the Tehran Stock Exchange (TSE) from 2010 to 2019 (pre-COVID 19 period). We were able to add another 2–3 years of data to our sample, but due to the significant impact of COVID 19 on the performance and function of most firms, we excluded the data related to the COVID 19 Period. Data are primarily based on the TSE's audited financial statements and board reports, a reliable source of information (Daryaei et al. 2022; Namakavarani et al. 2021; Zadeh et al. 2022; Shandiz et al. 2022). However, the study compiles a purposive sampling; thus, financial firms such as banks and insurance firms are absent, because they have different conditions concerning firm characteristics. Listing firms must also have continuous operations during the study period, and their information must be available. Following these criteria, the study includes 164 firms (1640 firm-years). Purposive sampling was used to select a representative sample with the following four inclusion criteria (see Table 1).

Table 1. Sample computation for firms.

Sample Computation for the Year 2010–2019	Firms	(%)
Total population Less:	533	100
Firms inactive between 2010–2019	(189)	(35)
Financial services firms	(52)	(10)
Firms that did not provide complete information	(48)	(9)
Firms that were admitted to the stock market from 2010	(80)	(15)
Final sampled firms	164	31

Source: Created by authors.

4. Empirical Models and Econometrics Results

Table 2 presents the summary statistics of the variables. We utilised the one-way analysis of variance test (ANOVA) and Kruskal–Wallis test because they determine the difference between industrial sectors based on explanatory and control variables (see Table 3, Panel A). After data collection, we must ensure their stationary and non-stationary status to avoid false regression. ADF–Fisher tests were used, since the applied regression method involved ordinary data. Results are shown in Table 3.

Table 2.	Descriptive	statistics.
----------	-------------	-------------

Variables	Mean	Median	Max	Min	Std	n
COMPENSATE	4.534	6.291	12.004	0.000	3.394	1640
COMP_CFO	-0.072	-0.062	-0.006	-0.255	0.035	1640
COMP_RET	-0.120	-0.105	-0.025	-0.504	0.061	1640
COMP_PRC	-5047.168	-3422.600	-1.514	-42,814.63	5044.544	1640
SQR	22.816	24.000	42.000	0.000	6.651	1640
CORR_ROA	0.367	0.701	1.000	-1.000	0.689	1640
CORR_CF	0.200	0.384	0.999	-0.999	0.706	1640
CORR_RET	0.457	0.787	1.000	-1.000	0.648	1640
INDHERF	0.221	0.245	0.482	0.000	0.124	1640
SIZE	13.819	13.614	19.374	9.797	1.571	1640
BM	0.515	0.418	1.974	-1.559	0.409	1640
LEV	0.588	0.563	1.363	0.012	0.214	1640
ROA	0.170	0.145	0.803	-0.729	0.156	1640
ADJROA	0.002	-0.003	0.563	-0.938	0.135	1640
RET	0.106	0.093	0.989	-0.997	0.317	1640
ADJRET	0.001	-0.009	1.092	-1.200	0.284	1640
GROWTH	0.194	0.161	1.460	-0.964	0.327	1640
DIVYIELD	0.105	0.078	0.809	0.000	0.114	1640
RETVOL	0.218	0.186	0.768	0.002	0.141	1640
CFVOL	0.076	0.061	0.619	0.000	0.060	1640
EM_ACC	0.000	0.000	0.900	-0.699	0.103	1640
ABAFEE	-0.003	0.001	2.700	-2.657	0.655	1640
LAF	6.620	6.731	9.606	2.928	1.362	1640
NFEE	6.624	6.514	8.860	4.683	1.207	1640

Table 3. (Panel A): ANOVA and Kruskal–Wallis of variables across twenty-four industrial sectors. (Panel B): ADF–Fisher. Null: Unit root (assume standard unit root test).

Panel A			Panel B
Variables	ANOVA (F)	Kruskal-Wallis (χ^2)	t-Statistics
COMPENSATE	16.071 ***	308.324 ***	497.251 ***
COMP_RET	12.011 ***	256.126 ***	519.336 ***
COMP_RET	19.203 ***	318.874 ***	609.562 ***
COMP_PRC	17.305 ***	128.347 ***	437.164 ***
SQR	23.622 ***	547.824 ***	692.3216 ***
CORR_ROA	14.459 ***	361.213 ***	806.228 ***
CORR_CF	18.154 ***	327.267 ***	659.450 ***
CORR_RET	11.240 ***	471.289 ***	647.330 ***
INDHERF	14.642 ***	241.478 ***	708.029 ***
SIZE	11.246 ***	502.545 ***	415.260 ***
BM	14.913 ***	139.115 ***	558.116 ***
LEV	11.315 ***	207.831 ***	352.283 ***
ROA	12.459 ***	336.643 ***	549.686 ***
ADJROA	18.914 ***	327.267 ***	553.628 ***
RET	10.210 ***	471.289 ***	967.044 ***

Panel A			Panel B
Variables	ANOVA (F)	Kruskal-Wallis (χ^2)	t-Statistics
ADJRET	15.642 ***	141.478 ***	953.812 ***
GROWTH	11.246 ***	502.555 ***	977.738 ***
DIVYIELD	14.130 ***	139.115 ***	604.402 ***
RETVOL	11.243 ***	207.831 ***	466.964 ***
CFVOL	14.910 ***	236.643 ***	557.470 ***
EM_ACC	13.246 ***	241.873 ***	539.182 ***
ABAFEE	13.031 ***	539.551 ***	604.402 ***
LAF	15.142 ***	377.381 ***	664.694 ***
NFEE	13.019 ***	362.346 ***	575.074 ***

Table 3. Cont.

*** *p* < 0.001.

To examine H1a, we estimate Equation (1) with the interaction term.

$$COMPENSATION_{i,t} = \beta_0 + \beta_1 COMP_{1,t-1} + \beta_2 ABAFEE_{i,t} + \beta_3 ABAFEE_{i,t-1} * COMP_{1,t-1} + \beta_4 CORR_ROA_{i,t-1} + \beta_5 CORR_CFO_{i,t-1} + \beta_6 CORR_RET_{i,t-1} + \beta_7 INDHERF_{i,t-1} + \beta_8 SIZE_{i,t-1} + \beta_9 BM_{i,t-1} + \beta_{10} LEV_{i,t-1} + \beta_{11} ROA_{i,t-1} + \beta_{12} ADJROA_{i,t-1} + \beta_{13} RET_{i,t-1} + \beta_{14} ADJRET_{i,t-1} + \beta_{15} GROWTH_{i,t-1} + \beta_{16} DIVYELD_{i,t-1} + \beta_{17} RETVOL_{i,t-1} + \beta_{18} CFVOL_{i,t-1} + \gamma_{18} RFIXED + \varepsilon_{i,t}$$
(1)

The dependent variable, *COMPENSATION*, is an indicator that equals the natural logarithm of cash compensation. We use three measures of accounting comparability (*COMP_I* is the annual decile rank of *COMP_CFO*, *COMP_RET*, *COMP_PRC*) based on the underlying logic that the accounting of two firms is more comparable if they report similar accounting amounts when they experience identical economic outcomes (Lobo et al. 2018).

Measure 1: The relationship between the profit in year t and the cash flow from year t - 1 operation is calculated using Equation (2).

$$Cfo_{i,t+1} = \beta_0 + \beta_1 N I_{i,t} + \varepsilon_{i,t}$$
⁽²⁾

 $CFO_{(i,t)}$ is cash flow from operations divided by beginning total assets. NI_(i,t) is net income after deducting the current year's tax divided by the beginning total assets.

In the first step, each year, the firm calculates the β_0 and β_1 coefficients using Equation (2) and 14 years of data. Coefficients for each firm in each year represent the firm's accounting systems characteristics. In the next step, the calculated coefficients are put into Equation (2), and cash flows from firm i's expected operating activities are estimated for the same year. Suppose the logic is based on the similarity of firms' i's and j's accounting systems. In that case, the output from the estimate of the cash flows from the expected operations of firm i must show similar figures to the output of firm j's coefficients (Equation (3)).

Equation (3) is the estimate of the future cash flows for firm i based on the firm i's coefficients:

$$E(Cfo)_{i,i,t+1} = \hat{\beta}_{0,i} + \hat{\beta}_{0,i} NI_{i,t}$$
(3)

Equation (4) is the estimate of the future cash flows for firm i based on firm j's coefficients:

$$E(Cfo)_{i,j,t+1} = \hat{\beta}_{0,j} + \hat{\beta}_{1,j} N I_{i,t}$$
(4)

The more comparable the two firms' accounting systems are, the smaller the difference in the expected cash flows of the peer firms would be. According to the above, the comparability between firms i and j based on measure 1 is calculated as follows:

$$CompCfo_{i,j,t} = -\frac{1}{10} \times \sum_{t=9}^{t} \left| E(Cfo)_{i,i,t+1} - E(Cfo)_{i,j,t+1} \right|$$
(5)

Comparability is calculated for every pair of firms in the industry for the years under assessment. It means that, according to Equation (4), the cash flows from the expected operations for firm i are estimated based on this firm's equation and firm j's equation, which is its peer firm, and also the cash flow difference from operations of each firm and its peers is calculated every year. After organising all the obtained combinations of firm i, the COMP variable could be calculated for the COMP_{i,j,t} through the average of the obtained numbers. So, the comparability figure for each firm year is achieved.

Measure 2: Model (6) calculates the accounting comparability by using the firm's schedule around the level and changes of the EPS and stock return.

$$Return_{i,t} = \beta_0 + \beta_1 \frac{NI_{i,t}}{P_{i,t}} + \beta_2 \frac{\Delta NI_{i,t}}{P_{i,t}} + \beta_3 LOSS_{i,t} + \beta_4 LOSS_{i,t} \times \frac{NI_{i,t}}{P_{i,t}} + \beta_5 LOSS_{i,t} \times \frac{\Delta NI_{i,t}}{P_{i,t}} + \varepsilon_{i,t}$$
(6)

RETURN_(i,t): Annual stock return of the firm i in the current year.

 $NI/P_{(i,t)}$: Net income after deducting tax per share in the current year divided by the beginning share price for firm i.

 $\Delta NI/P_{(i,t)}$: Changes in net income per share in the current year compared to the previous year divided by the beginning share price for firm i.

 $LOSS_{(i,t)}$: An artificial measure of a firm's loss. If the firm is unprofitable, it equals one. Otherwise, it equals zero.

To calculate accounting comparability through measure (2), the processes in Equations (2) and (3) are conducted. Therefore, comparability through measure (2) is calculated by using Equation (7):

$$CompReturn_{i,j,t} = -\frac{1}{10} \times \sum_{t=9}^{t} \left| E \left(Return \right)_{i,i,t} - E \left(Return \right)_{i,j,t} \right|$$
(7)

Measure (3): In this measure, using model (8), the relationship between net income per share and book value of the shareholder's equity per share with closing price per share in the current year is used to calculate accounting comparability.

$$Price_{i,t} = \beta_0 + \beta_1 NIPS_{i,t} + \beta_2 BVPS_{i,t} + \varepsilon_{i,t}$$
(8)

 $Price_{(i,t)}$: Closing price per share in the current year.

NIPS_(i,t): Net income after deducting tax per share for firm i in the current year.

 $BVPS_{(i,t)}$: Book value of the shareholder's equity per share at the end of the period for firm i.

To calculate financial information through measure (3), the processes in Equations (2) and (3) are conducted. Therefore, financial information comparability through measure (3) is calculated by using Equation (9):

$$CompReturn_{i,j,t} = -\frac{1}{10} \times \sum_{t=9}^{t} |E(Price)_{i,i,t} - E(Price)_{i,j,t}|$$
(9)

ABAFEE refers to abnormal audit fees and is calculated by the residual of the following model (Blankley et al. 2012):

$$LAF_{i,t} = \beta_0 + \beta_1 LTA_{i,t} + \beta_2 CR_{i,t} + \beta_3 CA_TA_{i,t} + \beta_4 ARINV_{i,t} + \beta_5 ROA_{i,t} + \beta_6 LOSS_{i,t} + \beta_7 FOREIGN_{i,t} + \beta_8 LEV_{i,t} + \beta_9 INTANG_{i,t} + \beta_{10} OPINION_{i,t} + \varepsilon_{i,t}$$
(10)

where *LAF* denotes logarithm of audit fees in year t; *LTA* is the logarithm of end-of-year total assets in year t; *CR* is current assets divided by current liabilities in year t; *CA_TA* is current assets divided by total assets in year t; *ARINV* is the sum of accounts receivable and inventory divided by total assets in year t; *ROA* is earnings before interest and taxes divided by total assets in year t; *ROA* is earnings before interest and taxes divided by total assets in year t; *ROA* is earnings before interest and taxes divided by total assets in year t; *ROA* is earnings before interest and taxes divided by total assets in year t; *ROA* is earnings before interest and is 0 if otherwise; *FOREIGN* is 1 if the firm has any foreign operations in year t and is 0 if otherwise; *LEV* is long-term debt divided by total assets in year t; *INTANG* is the ratio of intangible assets to total assets in year t; *OPINION* is 1 if the auditor issues a going concern audit opinion and is 0 if otherwise in year t.

 $CORR_ROA_{(i,t-1)}$: Correlation between the average return of the current assets in the current year in firm i and its peer firms in the industry.

 $CORR_CF_{(i,t-1)}$: Correlation between the average cash flows from operations in the current year in firm i and its peer firms in the industry.

 $CORR_RET_{(i,t-1)}$: Correlation between the average annual return of the share in firm i and its peer firms in the industry.

INDHERF_(i,t-1): Herfindahl–Hirschman Index, which is calculated by adding a market share of all active firms in the industry to the power of 2 by using Equation (10):

$$HHI = \sum_{i=1}^{k} s_i^2 \tag{11}$$

HHI: Herfindahl–Hirschman Index. K is the number of active firms in the market, and Si is the market share of firm i, which is calculated using Equation (11).

$$S_i = \frac{X_j}{\sum_{l=1}^n X_j} \tag{12}$$

X_i: Indicates sales of firm j, and 1 represents industry type.

 $SIZE_{(i,t-1)}$: Firm size equals the natural logarithm of the total closing assets.

 $BM_{(i,t-1)}$: Dividing the book value of the shareholder's equity by the market value of the shareholder's equity.

 $LEV_{(i,t-1)}$: Dividing total liabilities by total closing assets.

 $ROA_{(i,t-1)}$: Assets return obtained by dividing current year operating profit by the average of the firm's total assets i.

 $ADJROA_{(i,t-1)}$: Firm's assets return minus the average of the peer firm's assets return. $RET_{(i,t-1)}$: Annual return of firm i.

 $ADJRET_{(i,t-1)}$: Annual return of firm i's share minus the average return of the peer firm share in the industry.

 $\text{GROWTH}_{(i,t-1)}$: Firm growth equals the annual sales return, which is calculated using Equation (12).

$$Growth_{i,t} = \frac{Sales_{i,t} - Sales_{i,t-1}}{Sales_{i,t-1}}$$
(13)

Sales_{i,t}: Total firm sales in the current year.

Sales_{i,t-1}: The firm's sales in the previous year.

DIVYIELD_(i,t-1): Dividing the total approved dividend in the current year by the market value of firm i's shares.

 $RETVOL_{(i,t-1)}$: Standard deviation of firm i's annual return of the shares. A three-year measure of the standard deviation of the annual return is used to calculate firm i's yearly return on the shares.

 $CFVOL_{(i,t-1)}$: Dividing the standard deviation of the cash flows from operations in the current year by the total beginning assets of firm i. A three-year measure of the standard deviation of the cash flows from operations is used to calculate the standard deviation of the cash flows from firm i.

5. Findings

The findings from the H1a test are provided in Table 4. The results of this test indicate that abnormal audit fees did not influence the board's compensation. Also, on the grounds of these findings, it could be raised that they improve the relationship between accounting comparability through the third criterion (relationship between stock price with book value and earnings per share) and board compensation. According to the existing literature, several studies have found that abnormal audit fees negatively affect financial reporting quality (Asthana and Boone 2012). However, Blankley et al. (2012) and Eshleman and Guo (2020) found that abnormal audit fees positively correlate with quality financial reports. Based on the findings of this research, audit fees have been influential on the quality of financial reports and their relationship with board compensation only through the informativeness of accounting information, which positively impacts the relationship between accounting comparability and board compensation.

Table 4	. Models	for	earnings	management.
Iuvic I	• 101000010	101	carringo	management

Models	R ² (Adj)	F Statistic	AIC ⁽¹⁾	SC ⁽²⁾	HQC ⁽³⁾
Jones (1991)	0.309	3.774	-1.232	-0.823	-1.136
Dechow et al. (1991)	0.386	3.716	-1.125	-0.833	-1.068
Kasznik (1999)	0.567	11.582	-1.310	-0.957	-1.105
Dechow and Dichev (2002)	0.510	9.202	-1.186	-0.733	-1.081
McNichols (2002)	0.586	12.162	-1.344	-0.984	-1.136
Kothari et al. (2005)	0.293	3.665	-0.919	-0.426	-0.814
(1) 1 1 1 (0) 0 1	·· · (0) II	<u> </u>			

(1) Akaike criterion; (2) Schwarz criterion; (3) Hannan–Quinn criterion.

To examine $H1_b$, we estimate Equation (14) with the interaction term:

$$COMP_C_{i,t} = \beta_0 + \beta_1 COMP_I_{i,t-1} + \beta_2 NFEE_{i,t} + \beta_3 NFEE_{i,t-1} * COMP_{I,t-1} + \beta_4 CORR_ROA_{i,t-1} + \beta_5 CORR_CFO_{i,t-1} + \beta_6 CORR_RET_{i,t-1} + \beta_7 INDHERF_{i,t-1} + \beta_8 SIZE_{i,t-1} + \beta_9 BM_{i,t-1} + \beta_{10} LEV_{i,t-1} + \beta_{11} ROA_{i,t-1} + \beta_{12} ADJROA_{i,t-1} + \beta_{13} RET_{i,t-1} + \beta_{14} ADJRET_{i,t-1} + \beta_{15} GROWTH_{i,t-1} + \beta_{16} DIVYELD_{i,t-1} + \beta_{17} RETVOL_{i,t-1} + \beta_{18} CFVOL_{i,t-1} + YEAR\&FIRMFIXED + \varepsilon_{i,t}$$
(14)

where NFEE refers to the normal audit fee and is calculated using the following:

 $LAF_{i,t} - \varepsilon_{i,t}$

The findings of the H1b test are provided in Table 5. The results of this test depict that normal audit fees were productive on the board compensation only in the presence of the third criterion of comparability. However, the interaction between comparability criteria and normal audit fee indicates that the normal audit fee positively affects (rectifies) the relationship between accounting comparability and board compensation. These findings show that a normal audit fee is a productive factor in the quality of the financial reporting, impacts the information comparability, and ultimately affects the board compensation.

To test the H2, the non-linear method of smooth panel transition regression (PSTR) was used. The PSTR has the advantage of solving the baseline model's nonlinearity, heterogeneity, and time instability problems. Based on the PSTR specification, the regression coefficients change smoothly as a function of a threshold variable. Because the threshold variable is individual-specific and varies with time, it is possible to estimate regression coefficients at any time and for any firm in a panel. Therefore, this new econometric technique is an attractive approach for investigating the relationship between audit fees and earnings management. The PSTR model comprises many regimes and panel data observations regarding thresholds (Chiou and Lee 2011). We employed a PSTR model, recently developed by González and Teräsvirta (2006). to model the non-linear impact of audit fees on earnings management. This method has been employed in recent financial and

accounting studies (Daryaei and Fattahi 2020). Interestingly, as discussed in the literature, this method renders satisfactory results because of its robust theoretical foundations.

Table 5. Abnormal audit fee and accounting comparability.

Variable	COMP_CFO	COMP_RET	COMP_PRC
COMP_CFO	6.287 ** (3.043)	_	_
COMP_RET	_	2.739 *** (0.744)	-
COMP_PRC	_	_	$3.86 imes 10^{-5}$ *** ($1.48 imes 10^{-5}$)
ABAFEE	0.267	-0.160 *	0.065
	(0.226)	(0.089)	(0.054)
ABAFEE * COMP_CFO	5.434 (2.858)	_	-
ABAFEE * COMP_RET	_	-0.722 (0.707)	_
ABAFEE * COMP_PRC	_	_	$\begin{array}{c} 2.14 \times \ 10^5 \ ^{\ast\ast\ast\ast} \\ (8.15 \times \ 10^{-6}) \end{array}$
CORR_ROA	0.105	0.059	0.069
	(0.106)	(0.049)	(0.042)
CORR_CF	-0.119	-0.044	-0.066
	(0.103)	(0.045)	(0.041)
CORR_RET	0.109	0.035	0.022
	(0.117)	(0.057)	(0.047)
INDHERF	-2.124	-2.300 *	-3.015 ***
	(1.862)	(1.176)	(1.004)
SIZE	0.642 ***	0.365 ***	0.691 ***
	(0.076)	(0.044)	(0.038)
BM	0.493 **	0.120	0.072
	(0.226)	(0.111)	(0.105)
LEV	1.391 **	0.558 **	0.476 **
	(0.542)	(0.247)	(0.222)
ROA	0.462	1.011 *	0.277
	(1.282)	(0.609)	(0.533)
ADJROA	3.210 **	0.805	1.596 ***
	(1.280)	(0.610)	(0.504)
RET	1.659 ***	0.503 ***	0.595 ***
	(0.406)	(0.180)	(0.161)
ADJRET	-1.466 ***	-0.504 **	-0.664 ***
	(0.444)	(0.207)	(0.176)
GROWTH	-0.213	-0.020	-0.025
	(0.236)	(0.109)	(0.097)
DIVYIELD	1.356 **	0.903 **	0.977 ***
	(0.736)	(0.389)	(0.367)
RETVOL	1.324	0.573 **	0.459 **
	(0.552)	(0.246)	(0.226)
CFVOL	0037	0.282	0.594
	(1.278)	(0.612)	(0.457)
Hausman Test (χ^2)	32.619	39.004	39.860
R ² (Adj)	0.504	0.862	0.899
F statistic	7.627 ***	47.029 ***	67.487 ***
DW	1.764	1.961	1.857

*** p < 0.001, ** p < 0.01, * p < 0.05.

In this research, discretionary accruals have been used to measure earnings management. In the financial and accounting literature, these items are defined as reflections of the methods and procedures available to management (Kashmiri 2014). To estimate discretionary accruals, several models and patterns exist, such as Healy (1985), McNichols (2002), amd Kothari et al. (2005). There are two approaches to profit or loss, and the balance sheet approach is used to calculate the accruals. In the profit and loss approach, accruals are obtained through the difference between net profit and cash flows from operations, in which a cash flow statement plays a role. However, balance sheet-based accruals are measured by deducting current liabilities (except the current portion of long-term debt) from the changes of the non-cash current assets. In this study, due to the limitation in gathering information related to the current portion of long-term debt, the profit and lossbased approach was utilised to measure the accruals, which are calculated according to the following relationship:

$$TA_{it} = EBIT_{it} - CFO_{it}$$

As mentioned above, there are various models to evaluate discretionary accruals as the measure of earnings management, which means that under these circumstances, a model that is best compatible with the research environment and can test the characteristics of the firms should be utilised. Reviewing conducted research on Iran's economic climate indicates that there is more of a tendency towards the model of Jones and its modified model than any other model among discretionary accruals measurement models. However, irrespective of its reasons, another model may be more compatible with Iranian firms and could better describe this variable. Also, in this research, the measurement of discretionary accruals as earnings management was conducted on the grounds of Jones (1991), Kasznik (1999), Dechow and Dichev (2002), McNichols (2002), and Kothari et al. (2005), and the explanatory power, error level, and other estimated statistics and the model have had more favourable statistics to be selected as the model of the discretionary accrual's evaluation and earnings management criterion. Based on the results from Tables 6 and 7, it is shown that the coefficients of determination-the F statistic, Adjusted R-squared, Akaike criterion (AIC), Schwarz criterion (SC), and Hannan–Quinn criterion (HQC)—in the McNichols model yielded a more favourable condition compared to other models. For instance, the Adjusted R-squared in McNichol's model was 0.586, which indicates that it is the more influential the explanatory variable in the accruals.

Moreover, this model's AIC, SC, and HQC criteria were lower than in other models, indicating the model's higher precision. Therefore, the absolute value of the remainder of the McNichols (2002) model was utilised as the discretionary accruals to evaluate earnings management. Equations (15)–(20) show how to estimate discretionary accruals as an earnings management criterion:

$$\frac{TA_{it}}{A_{it-1}} = \beta_0 + \beta_1 \left(\frac{1}{A_{it-1}}\right) + \beta_2 \left(\frac{\Delta REV_{it}}{A_{it-1}}\right) + \beta_3 \left(\frac{PPE_{it}}{A_{it-1}}\right) + \varepsilon$$
(15)

$$\frac{TA_{it}}{A_{it-1}} = \beta_0 + \beta_1 \left(\frac{1}{A_{it-1}}\right) + \beta_2 \left[\frac{(\Delta REV_{it} - \Delta AR_{it})}{A_{it-1}}\right] + \beta_3 \left(\frac{PPE_{it}}{A_{it-1}}\right) + \varepsilon$$
(16)

$$\frac{TA_{it}}{A_{it-1}} = \beta_0 + \beta_1 \left(\frac{1}{A_{it-1}}\right) + \beta_2 \left[\frac{(\Delta REV_{it} - \Delta AR_{it})}{A_{it-1}}\right] + \beta_3 \left(\frac{PPE_{it}}{A_{it-1}}\right) + \beta_4 \left(\frac{\Delta CFO_{it}}{A_{it-1}}\right) + \varepsilon$$
(17)

$$\frac{TA_{it}}{A_{it-1}} = \beta_0 + \beta_1 \left(\frac{1}{A_{it-1}}\right) + \beta_2 \left(\frac{CFO_{it}}{A_{it-1}}\right) + \beta_3 \left(\frac{CFO_{it-1}}{A_{it-1}}\right) + \beta_4 \left(\frac{CFO_{it+1}}{A_{it-1}}\right) + \varepsilon$$
(18)

$$\frac{TA_{it}}{A_{it-1}} = \beta_0 + \beta_1 \left(\frac{1}{A_{it-1}}\right) + \beta_2 \left(\frac{\Delta REV_{it}}{A_{it-1}}\right) + \beta_3 \left(\frac{PPE_{it}}{A_{it-1}}\right) + \beta_4 \left(\frac{CFO_{it}}{A_{it-1}}\right) + \beta_5 \left(\frac{CFO_{it-1}}{A_{it-1}}\right) + \beta_6 \left(\frac{CFO_{it+1}}{A_{it-1}}\right) + \varepsilon \quad (19)$$

$$\frac{TA_{it}}{A_{it-1}} = \beta_0 + \beta_1 \left(\frac{1}{A_{it-1}}\right) + \beta_2 \left[\frac{(\Delta REV_{it} - \Delta AR_{it})}{A_{it-1}}\right] + \beta_3 \left(\frac{PPE_{it}}{A_{it-1}}\right) + \beta_4 ROA_{it} + \varepsilon$$
(20)

Variable	COMP_CFO	COMP_RET	COMP_PRC
COMP_CFO	18.252 ** (8.928)	-	-
COMP_RET	-	3.015 *** (0.613)	-
COMP_PRC	-	-	$\begin{array}{c} 0.0001 \; ^{**} \\ (5.98 \times 10^{-5}) \end{array}$
NFEE	0.083	-0.131	0.373 ***
	(0.152)	(0.120)	(0.087)
NFEE * COMP_CFO	3.032 ** (1.378)	-	-
NFEE * COMP_RET	-	0.485 *** (0.162)	-
NFEE * COMP_PRC	-	-	$3.30 imes 10^{-5}$ *** (1.01 $ imes 10^{-5}$)
CORR_ROA	0.043 ***	0.042 *	0.057
	(0.014)	(0.023)	(0.055)
CORR_CF	-0.070 ***	-0.053 *	-0.096 **
	(0.023)	(0.030)	(0.048)
CORR_RET	0.014	0.042	0.201 **
	(0.033)	(0.042)	(0.094)
INDHERF	-2.870 **	-2.297	-8.034 ***
	(1.407)	(1.572)	(0.374)
SIZE	0.689 ***	0.627 ***	0.720 ***
	(0.064)	(0.067)	(0.037)
BM	0.074	0.108	0.449 ***
	(0.073)	(0.079)	(0.098)
LEV	0.394 ***	0.468 ***	-0.185
	(0.138)	(0.152)	(0.214)
ROA	0.435 *	1.209 ***	4.476 ***
	(0.248)	(0.363)	(0.606)
ADJROA	1.520 ***	0.710 *	0.177
	(0.376)	(0.431)	(0.826)
RET	0.514 ***	0.483 ***	0.814 ***
	(0.137)	(0.135)	(0.203)
ADJRET	-0.605 ***	-0.517 ***	-0.821 ***
	(0.124)	(0.139)	(0.177)
GROWTH	-0.031	-0.027	-0.324 ***
	(0.081)	(0.078)	(0.124)
DIVYIELD	0.924 ***	0.911 ***	-0.501
	(0.299)	(0.328)	(0.689)
RETVOL	0.558 ***	0.486 ***	0.426 *
	(0.198)	(0.176)	(0.239)
CFVOL	0.351	0.330	-1.841 ***
	(0.418)	(0.473)	(0.704)
Hausman Test (χ^2)	35.483	37.124	40.011
R ² (Adj)	0.892	0.865	0.518
F statistic	62.111 ***	48.591 ***	91.106 ***
DW	1.764	1.753	1.965

Table 6. Normal audit fee and accounting comparability.

*** p < 0.001, ** p < 0.01, * p < 0.05.

Tests	M = 1	M = 2	
Lagrange multiplier test (LM) H0: r = 0 vs. $H1: r = 1$	6.003 ***	0.657	
Likelihood ratio test (LR.) H0: r = 0 vs. $H1: r = 1$	21.586 ***	1.554	

Table 7. LM and LR tests for linearity.

The variables used in the various models of accrual earnings management are defined as follows:

TA is the total accruals in year t defined as earnings before extraordinary items and discontinued operations minus operating cash flows (from continuing operations); *A* denotes the total assets in year t – 1; ΔREV is the change in net sales from year t – 1 to year t; *PPE* is the gross value of property, plant, and equipment; ΔAR is the change in account receivables from period t – 1 to t; *CFO* denotes the operating cash flows from t; *ROA* is the annual income before extraordinary items divided by the beginning total assets.

The simplest case of a PSTR model with two extreme regimes is defined as follows:

$$EM_ACC_{i,t} = \mu_1 + \beta_0 (ROA_{i,t} + Lev_{i,t} + Size_{i,t}) \\ + \sum_{J=1}^r \beta_J g (LAF_{i,t}; \gamma_j; c_k) (ROA_{i,t} + Lev_{i,t} + Size_{i,t}) + \varepsilon_{i,t}$$
(21)

Let i = 1, ..., N, and let t = 1, ..., T, where *N* and *T* denote the total number of firms and the size of the sample period in the panel, respectively. The variable of *EM_ACC* is a dependent variable and refers to accrual earnings management estimated using the McNichols (2002) model.

$$\frac{TA_{it}}{A_{it-1}} = \beta_0 + \beta_1 \left(\frac{1}{A_{it-1}}\right) + \beta_2 \left(\frac{\Delta REV_{it}}{A_{it-1}}\right) + \beta_3 \left(\frac{PPE_{it}}{A_{it-1}}\right) + \beta_4 \left(\frac{CFO_{it}}{A_{it-1}}\right) + \beta_5 \left(\frac{CFO_{it-1}}{A_{it-1}}\right) + \beta_6 \left(\frac{CFO_{it+1}}{A_{it-1}}\right) + \varepsilon \quad (22)$$

The variables used in the McNichols (2002) model are already defined. The variables *LAF*, *ROA*, *Lev*, and *Size* are explanatory variables and refer to the natural logarithm of the audit fee, the ratio of the asset (defined as net income before extraordinary items scaled by total assets), the leverage (debt to asset), and the firm size (natural log of assets). Here, the dependent variable is *LAF* (the audit fee). Also, μ_1 denotes the vector of fixed firm effects, $g(LAF_{i,t}; \gamma_j; c_k)$ is the transition function, and $\varepsilon_{i,t}$ is the error term. Based on the works of González and Teräsvirta (2006) n a panel framework, the current research considered the following logistic transition function:

$$g(LAF_{i,t};\gamma_j;c_k) = \left(1 + \exp\left(-\gamma_j \prod_{k=1}^m INS_{i,t} - c_k\right)\right)^{-1} \ \gamma > 0, c_1 \le c_2 \le \dots \le c_m$$
(23)

where $c = (c_1, ..., c_m)$ is an m-dimensional vector of location parameters, and the slope parameter determines the smoothness of the transitions. The restrictions $\gamma > 0$ and $c_1 < ... < c_m$ are imposed for identification purposes. In practice, it is usually sufficient to consider m = 1 or m = 2, as these values allow for commonly encountered types of variation in the parameters. For m = 1, the model implies that the two extreme regimes are associated with low and high values of $LAF_{i,t}$ with a monotonic transition of the coefficients from β_0 to $\beta_0 + \beta_1$ as $LAF_{i,t}$ increases, where the change is focused around C_1 . When $\gamma \rightarrow \infty$, $g(LAF_{i,t}; \gamma_j; c_k)$ becomes an indicator function I [$LAF_{i,t} > c$], which is defined as I [A] = 1 when the event A occurs and as zero otherwise. In that case, the PSTR model in (1) reduces to the two-regime panel threshold model of Hansen and Seo (2002). For m = 2, the transition function has its minimum at $(c_1 + c_2)/2$ and attains a maximum value of 10 both at low and high values of $LAF_{i,t}$. A generalisation of the PSTR model to allow for more than two different regimes occurs in another model, which is presented below:

$$EM_ACC_{i,t} = \mu_1 + \beta_0 (ROA_{i,t} + Lev_{i,t} + Size_{i,t}) + \sum_{J=1}^r \beta_J g (LAF_{i,t};\gamma_J;c_k) (ROA_{i,t} + Lev_{i,t} + Size_{i,t}) + \varepsilon_{i,t}$$
(24)

In the transition functions $g(LAF_{i,t}; \gamma_j; c_k)$ and j = 1, ..., r, the slope parameter γ_j and the location parameter c_j are critical. Specifically, the multiple regime model (31) serves as a crucial alternative when performing diagnostic tests for the absence of heterogeneity. If m = 1, then all slope parameters are γ_j . As γ_j approaches infinity, the PSTR model (Panel Smooth Transition Regression) simplifies to multiple regime panel threshold regression models (PTRs), indicating a sharp transition between regimes (Heidari et al. 2015).

Table 7 presents the linearity test results. Before estimating the final panel smooth transition regression model, it is essential to perform both the linearity test and the test for no remaining non-linearity to identify the optimal specification of the panel soft transition regression model. In testing for no remaining non-linearity, the null hypothesis suggests the existence of a single transition function. In contrast, the alternative hypothesis posits the need for at least two transition functions within the panel smooth transition regression model. This study utilized the Lagrange Multiplier (LM) and Likelihood Ratio (LR) tests to assess non-linearity. The results from these tests indicate that considering just one transition function is sufficient to capture the non-linearity between the model variables.

The optimal threshold is identified when the Residual Sum of Squares (RSS), Akaike Information Criterion (AIC), and Bayesian Information Criterion (BIC) reach their minimum values as indicated in Equation (7). Table 8 presents the threshold value ccc and the transition parameter γ . For the threshold variable of the audit fee (LAF), the RSS, AIC, and BIC values recorded at the 28th estimate are displayed in Table 7. It is evident that the optimal threshold of 4.651 is lower than the mean audit fee of 6.620. Additionally, the slope parameter γ , which indicates the transition speed between regimes, is 846.237. This high value suggests a relatively swift shift from the first regime to the second.

Search Range	Optimal Threshold Value (c)	Transition Parameter (γ)	RSS	AIC	BIC
LAF	4.951 *** (2.026)	846.237 *** (322.361)	-4.743	-55.709	-46.152

Table 8. Test results of the threshold value in PSTR estimates.

*** *p* < 0.001.

Table 9 shows the results of estimating the two-regime panel smooth transition regression model.

The gradient parameter (γ) indicates the transfer velocity of one regime to another, which equals 846.237, and the audit fee threshold equals 4.951. The threshold is the two regimes' turning point and specific point expressed in the panel smooth transition regression model. The threshold is the defining point of the two regimes described in the panel soft transition regression model, which varies according to the amount of the slope parameter estimates and the values of the transition variable (audit fee); these are the model estimates from one regime to another. Of course, the two first and second regimes are some of the limit states of the panel smooth transition regression model. The values of the regression coefficients vary somewhat concerning the observations of the transition variable. The number 4.951 indicates that the impact of audit fees before 4.951 and after 4.951 differs on the actual earnings management. In other words, the audit fee has a positive effect until the level of 4.951. In other words, an audit fee has a positive effect until 4.951 and negatively impacts the accrual earnings management above that level.

The regression F statistic is equal to 35.399 and is significant. In other words, independent variables have been able to explain the changes in the dependent variable. Also, the modified coefficient of determination of the model is equal to 0.682. This indicates that independent variables explain 68% of the changes in the dependent variable and demonstrates the high explanatory power of the model.

Panel A: Linearity Model				
	Coeff.	SE.	t-value	
LAF	0.001	0.001	0.304	
ROA	0.602	0.025	23.729	
Lev	-0.015	0.010	-0.136	
Size	0.009	0.001	5.809	
Adjusted R ²	0.632			
F statistic	17.362 ***			
Panel B: Non-linearity Mode	el (Regime 1)			
LAF	0.014 ***	0.005	2.823	
ROA	-	-	-	
Lev	-	-	-	
Size	-	-	-	
C1	4.651 ***	1.806	2.573	
γ	846.237 ***	319.117	2.651	
Adjusted R ²	0.621			
F statistic	10.713 ***			
Panel C: Non-linearity Mode	el (Regime 2)			
LAF	-0.004 ***	0.001	-4.424	
ROA	-	-	-	
Lev	-	-	-	
Size	-	-	-	
C2	-	-	-	
γ	-	-	-	
Adjusted R ²	0.682			
F statistic	35.399 ***			
DW	1.94			

Table 9. Parameter estimates of the PSTR model.

*** *p* < 0.001.

The Durbin–Watson value equals 1.94, which indicates that there is no consecutive correlation among error parts. As shown in Table 9, audit fees have a positive impact on accrual earnings management in the first regime and a negative effect on the accrual earnings management in the second regime; in a way that, as audit fees increase to a certain amount (4.951), accrual earnings management increases, and after that, an increase in the audit fee leads to the reduction in the accrual earnings management.

However, we observed (please see Table 10) that the interaction of the audit fees with the quality of reporting and the accounting comparability could lead to increased managers' compensation through the relationship between the accounting variables and stock prices.

Moreover, Table 11 and Figure 2 present the testing and results of the second hypothesis for companies with high-quality sustainability reporting. In this method (PSTR), we should use only panel data. So, we identified the companies with higher quality (higher than the median) using the following method: Suppose the number of sustainability reporting quality in more than half of the research period is more than the median (24.00). In that case, this company is recognised for sustainability reporting quality throughout the research period. The results show that earnings management increased with the audit fee increase from its lowest amount (2.928) to 3.614 in the first regime. These findings indicate that the auditors' efforts to a certain extent—because they were not of the required quality—could not reduce the process of earnings management and even coincided with the increase in earnings management. Of course, one can rely on the assumption that managers in companies seek earnings management, and this trend is constantly increasing. Auditors, as

a solid regulatory arm seeking financial accreditation, have been able to make dramatic changes in the impact of managers' efforts on profit accruals in line with their personal goals. In this case, by observing the second regime, we can see that with the increase in auditors' efforts and in consequently receiving more fees from the level of 3.614 onwards, this effect becomes reversed, and these efforts led to reduced earnings management by company managers. Point C in Figure 2 indicates the change in the impact of the auditors' actions on earnings management.

Variable	COMP_CFO	COMP_RET	COMP_PRC
COMP_CFO	6.614 ** (2.627)	-	-
COMP_RET	-	3.876 *** (0.851)	-
COMP_PRC	-	-	$5.35 imes 10^{-5}$ *** (1.19 $ imes 10^{-5}$)
LAF	-0.122 *	-0.101 ***	-0.037
	(0.069)	(0.024)	(0.024)
SQR	0.062	-0.033	0.262 ***
	(0.157)	(0.041)	(0.078)
SQR*LAF *COMP_CFO	-0.131 (0.669)	-	-
SQR*LAF *COMP_RET	-	-0.293 *** (0.105)	-
SQR*LAF *COMP_PRC	-	-	$3.66 imes 10^{-6}$ ** (1.55 $ imes 10^{-6}$)
CORR_ROA	0.090	0.039 *	0.049 ***
	(0.066)	(0.022)	(0.019)
CORR_CF	-0.139	-0.054 **	-0.072 ***
	(0.093)	(0.036)	(0.023)
CORR_RET	0.126	0.052	0.028
	(0.127)	(0.036)	(0.029)
INDHERF	-2.155	-2.540 *	-3.328 **
	(1.198)	(1.479)	(1.426)
SIZE	0.635 ***	0.639 ***	0.679 ***
	(0.141)	(0.056)	(0.055)
BM	0.494 ***	0.109	0.058
	(0.136)	(0.076)	(0.084)
LEV	1.372 ***	0.522 ***	0.455 ***
	(0.405)	(0.126)	(0.127)
ROA	0.321	1.009 ***	0.030
	(1.356)	(0.351)	(0.253)
ADJROA	3.196 **	0.957 **	1.894 ***
	(1.499)	(0.403)	(0.327)
RET	1.679 ***	0.531 ***	0.633 ***
	(0.577)	(0.147)	(0.122)
ADJRET	-1.471 ** (0.664)	-0.584 *** (0.135)	-0.757 *** (0.110)
GROWTH	-0.178	0.000	-0.010
	(0.283)	(0.067)	(0.063)
DIVYIELD	1.407 *	1.010 ***	1.065 ***
	(0.848)	(0.329)	(0.322)
RETVOL	1.263 ***	0.505 ***	0.377 **
	(0.445)	(0.143)	(0.175)

 Table 10. Accounting comparability and board compensation: Audit Fee*SQR.

Table 10. Cont.

Variable	COMP_CFO	COMP_RET	COMP_PRC
CFVOL	-0.010 (0.918)	0.306 (0.439)	0.521 (0.409)
Hausman Test (χ^2)	28.335	31.415	30.483
R ² (Adj)	0.497	0.837	0.876
F statistic	6.995 ***	38.586 ***	53.047 ***
DW	1.858	1.873	1.861
*** $p < 0.001$, ** $p < 0.01$, * $p < 0.01$	5.		

Table 11. Parameter estimates of the PSTR model.

Panel A: Linearity Model				
	Coeff.	SE.	t-value	
LAF	0.021	0.008	2.625	
ROA	0.536	0.083	6.457	
Lev	-0.142	0.130	-1.092	
Size	0.044	0.0.13	3.384	
Adjusted R ²	0.616			
F statistic	14.222 ***			
Panel B: Non-linearity Mo	odel (Regime 1)			
LAF	0.010 ***	0.004	2.506	
ROA	-	-	-	
Lev	-	-	-	
Size	-	-	-	
C1	3.614 ***	1.679	2.152	
γ	514.293 ***	119.202	4.299	
Adjusted R ²	0.621			
F statistic	18.357 ***			
Panel C: Non-linearity Mc	odel (Regime 2)			
LAF	-0.012 ***	0.005	-2.418	
ROA	-	-	-	
Lev	-	-	-	
Size	-	-	-	
C2	-	-	-	
γ	-	-	-	
Adjusted R ²	0.621			
F statistic	28.106 ***			
DW				
*** 0.001				

*** *p* < 0.001.

Earnings Management



Figure 2. Asymmetric impact of audit fee on earnings management.
6. Discussion

This study employed a robust quantitative ex post facto research design to investigate the complex relationship between audit fees and earnings management. Analysing data from firms listed on the Tehran Stock Exchange, we uncovered an inverted U-shaped relationship between audit fees and earnings management, indicating that moderate audit fees can lead to higher earnings management. The key contributions of this paper include the following:

- Highlighting the role of audit fees: This study underscores the influence of audit fees on financial reporting quality and risk management.
- 2. Providing empirical evidence: We demonstrate the asymmetric effects of normal and abnormal audit fees on earnings management.
- 3. Emphasising balanced audit fee structures: The need for balanced audit fee structures to ensure financial transparency and mitigate risk is evident from our findings.

The results show that normal audit fees positively affect the relationship between accounting comparability and executive compensation, while abnormal audit fees do not influence executive compensation. Specifically, our non-linear model reveals that audit fees positively impact accrual earnings management up to a certain threshold (4.951). Beyond this threshold, higher audit fees result in decreased accrual earnings management. These findings align with the results of Blankley et al. (2012) and Eshleman and Guo (2014).

Moreover, our results suggest that the interaction of audit fees with reporting quality and accounting comparability, through the relationship between accounting variables and stock prices, can lead to increased executive compensation. Audit fees up to a certain threshold (3.614 in our sample) significantly impact earnings management in companies with higher reporting quality. This inverted U-shaped relationship between audit fees and earnings management indicates that an initial increase in audit fees may reflect a lack of accounting comparability and provide room for higher earnings management. However, beyond a certain level, higher audit fees imply the need for more rigorous scrutiny of accounting transactions, thereby reducing earnings management.

In Iran's reporting and auditing environment, similar conditions may prevail. Given the limitations in Iran's audit market and the potential for insufficient competition among audit firms, increasing audit fees up to a certain level may reduce the comparability of financial statements, thus increasing earnings management. However, beyond that point, higher audit fees can enhance the accuracy and quality of audits, thereby reducing earnings management. These findings are particularly relevant for regulators and policymakers in Iran's accounting and auditing fields.

7. Conclusions

This study provides critical insights into the relationship between audit fees and earnings management within the context of the Tehran Stock Exchange. Our findings highlight an inverted U-shaped relationship, where moderate audit fees can lead to increased earnings management, while higher audit fees beyond a specific threshold can reduce it. The differentiation between normal and abnormal audit fees adds a nuanced understanding of how audit costs influence financial reporting quality.

Regulators and policymakers should consider these dynamics when developing guidelines for audit fee structures to ensure financial transparency and mitigate earnings management risks. Further research is recommended to explore these relationships in different economic and regulatory contexts, including comparative studies across various markets. Additionally, longitudinal studies could provide a deeper understanding of how these relationships evolve over time and under different economic conditions. Finally, integrating qualitative methods could offer more comprehensive insights into the practical implications of audit fees on financial reporting and earnings management.

Limitations:

While robust in its quantitative approach and comprehensive analysis, this study has several limitations. First, the sample was restricted to 164 companies listed on the Tehran

Stock Exchange from 2010 to 2019. This specific economic and regulatory environment may limit the generalizability of the findings to other markets with different characteristics. Additionally, the study focuses on the impact of audit fees on earnings management within a high-inflation and high-risk economic context, which might not be directly applicable to more stable economies. Another limitation is the division of the audit fees into normal and abnormal categories, which, while innovative, might oversimplify the complex dynamics between audit fees and financial reporting quality.

Suggestions for Further Research:

Future research could address these limitations by expanding the sample to include companies from diverse economic and regulatory environments to enhance the generalizability of the findings. Comparative studies across different countries and markets would provide deeper insights into the role of audit fees in varying contexts. Additionally, examining the interplay between audit fees and other factors such as corporate governance practices, auditor expertise, and economic conditions could provide a more holistic understanding of the determinants of financial reporting quality. Longitudinal studies that track changes over more extended periods and in different economic cycles would also be valuable in understanding the temporal dynamics of these relationships. Furthermore, qualitative research methods, such as case studies and interviews with auditors and company executives, could complement the quantitative findings and provide more prosperous, nuanced insights into the mechanisms at play.

Author Contributions: Conceptualisation, A.A.D. and Y.F.; methodology, A.A.D. and Y.F.; software, A.A.D. and Y.F.; validation, A.A.D., Y.F., and D.A.; formal analysis, A.A.D. and Y.F.; investigation, A.A.D. and Y.F.; resources, A.A.D. and Y.F.; data curation, A.A.D. and Y.F.; writing—original draft preparation, A.A.D. and Y.F.; writing—review and editing, D.A. visualisation, D.A. supervision, A.A.D.; project administration, A.A.D., Y.F., and D.A. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Data Availability Statement: The companies' data can be purchased via the TSE website at https: //mabnadp.com/products/rahavard365 (accessed on 1 May 2024). The relevant data are called 'rahavard-novin' and are available under the 'products' category; they can be accessed at https: //mabnadp.com/products/rahavard-novin (accessed on 1 May 2024).

Conflicts of Interest: The authors declare no conflicts of interest.

References

- Antle, Rick, Elizabeth Gordon, Ganapathi Narayanamoorthy, and Ling Zhou. 2006. The joint determination of audit fees, non-audit fees, and abnormal accruals. *Review of Quantitative Finance and Accounting* 27: 235–66. [CrossRef]
- Asthana, Sharad C., and Jeff P. Boone. 2012. Abnormal audit fee and audit quality. Auditing: A Journal of Practice & Theory 31: 1–22.
- Barth, Mary E., Wayne R. Landsman, Mark H. Lang, and Christopher D. Williams. 2018. Effects on comparability and capital market benefits of voluntary IFRS adoption. *Journal of Financial Reporting* 3: 1–22. [CrossRef]
- Ben Abdelaziz, Fouad, Souhir Neifar, and Khamoussi Halioui. 2022. Multilevel optimal managerial incentives and audit fees to limit earnings management practices. *Annals of Operations Research* 311: 587–610. [CrossRef]
- Blankley, Alan I., David N. Hurtt, and Jason E. MacGregor. 2012. Abnormal audit fees and restatements. *Auditing: A Journal of Practice* & Theory 31: 79–96.
- Chiou, Jer-Shiou, and Yi-Chieh Lee. 2011. Efficiency and profitability of biotech-industry in a small economy. *International Journal of Business and Commerce* 1: 1–24.
- Choi, Jong-Hag, Sunhwa Choi, Linda A. Myers, and David Ziebart. 2019. Financial statement comparability and the informativeness of stock prices about future earnings. *Contemporary Accounting Research* 36: 389–417. [CrossRef]
- Christensen, Brant E., Nathan J. Newton, and Michael S. Wilkins. 2021. How do team workloads and team staffing affect the audit? Archival evidence from US audits. *Accounting*, *Organisations and Society* 92: 101225. [CrossRef]
- Daryaei, Abbas Ali, and Yasin Fattahi. 2020. The asymmetric impact of institutional ownership on firm performance: Panel smooth transition regression model. *Corporate Governance: The International Journal of Business in Society* 20: 1191–203. [CrossRef]
- Daryaei, Abbas Ali, Yasin Fattahi, Davood Askarany, Saeed Askary, and Mahdad Mollazamani. 2022. Accounting Comparability, Conservatism, Executive Compensation-Performance, and Information Quality. *Journal of Risk and Financial Management* 15: 489. [CrossRef]

- Dechow, Patricia M., and Ilia D. Dichev. 2002. The quality of accruals and earnings: The role of accrual estimation errors. *The Accounting Review* 77 s-1: 35–59. [CrossRef]
- Dechow, Patricia M., Richard G. Sloan, and Amy P. Sweeney. 1991. Detecting earnings management. Accounting Review 70: 193–225.

Donatella, Pierre, Mattias Haraldsson, and Torbjörn Tagesson. 2019. Do audit firms and audit costs/fees influence earnings management in Swedish municipalities? *International Review of Administrative Sciences* 85: 673–691. [CrossRef]

Eshleman, John Daniel, and Peng Guo. 2014. Abnormal audit fees and audit quality: The importance of considering managerial incentives in tests of earnings management. *Auditing: A Journal of Practice & Theory* 33: 117–38.

Eshleman, John Daniel, and Peng Guo. 2020. Do seasoned industry specialists provide higher audit quality? A re-examination. *Journal* of Accounting and Public Policy 39: 106770. [CrossRef]

Fattahi, Yasin, Gholamreza Kordestani, and Abbas Ali Daryaei. 2021. Impact of Accounting Comparability according to the Peer Firms on Board Compensation. *Budget and Finance Strategic Research* 2: 11–49.

Ferguson, Michael J., Gim S. Seow, and Danqing Young. 2004. Non-audit services and earnings management: UK evidence. Contemporary Accounting Research 21: 813–41. [CrossRef]

Financial Accounting Standards Board (FASB). 2010. Statement of Financial Accounting Concepts No. 8: Conceptual Framework for Financial Reporting. Norwalk: Financial Accounting Standards Board (FASB).

Frankel, Richard M., Marilyn F. Johnson, and Karen K. Nelson. 2002. The relation between auditors' fees for non-audit services and earnings management. *The Accounting Review* 77: 71–105. [CrossRef]

Gandía, Juan L., and David Huguet. 2021. Audit fees and earnings management: Differences based on the type of audit. *Economic Research-Ekonomska Istraživanja* 34: 2628–50. [CrossRef]

González, Andrés, and Timo Teräsvirta. 2006. Simulation-based finite sample linearity test against smooth transition models. Oxford Bulletin of Economics and Statistics 68: 797–812. [CrossRef]

Gu, Junjian, and Dan Hu. 2015. Audit fees, earnings management, and litigation risk: Evidence from Japanese firms cross-listed on US markets. *Academy of Accounting and Financial Studies Journal* 19: 125.

Hansen, Bruce E., and Byeongseon Seo. 2002. Testing for two-regime threshold cointegration in vector error-correction. *Journal of Econometrics* 110: 293–318. [CrossRef]

Healy, Paul M. 1985. The effect of bonus schemes on accounting decisions. Journal of Accounting and Economics 7: 85–107. [CrossRef]

Heidari, Hassan, Salih Turan Katircioğlu, and Lesyan Saeidpour. 2015. Economic growth, CO₂ emissions, and energy consumption in the five ASEAN countries. *International Journal of Electrical Power & Energy Systems* 64: 785–91.

Henderson, Andrew D., and James W. Fredrickson. 2001. Top management team coordination needs and the CEO pay gap: A competitive test of economic and behavioral views. *Academy of Management Journal* 44: 96–117. [CrossRef]

Jones, Jennifer J. 1991. Earnings management during import relief investigations. Journal of Accounting Research 29: 193–228. [CrossRef]

- Kalelkar, Rachana, Yuan Shi, and Hongkang Xu. 2024. Top management team incentive dispersion and audit fees. *Journal of Corporate* Accounting & Finance 35: 178–91.
- Kashmiri, Hawazin. 2014. Testing accruals-based earnings management models in an international context: A thesis presented in partial fulfilment of the requirements for the Master of Business Studies in Accountancy degree at Massey University, Albany, New Zealand. Ph.D. dissertation, Massey University, Albany, New Zealand.
- Kasznik, Ron. 1999. On the association between voluntary disclosure and earnings management. *Journal of Accounting Research* 37: 57–81. [CrossRef]
- Kothari, Sagar P., Andrew J. Leone, and Charles E. Wasley. 2005. Performance-matched discretionary accrual measures. *Journal of Accounting and Economics* 39: 163–97. [CrossRef]
- Lee, Nian Lim, Mohamed Sami Khalaf, Magdy Farag, and Mohamed Gomaa. 2024. The impact of critical audit matters on audit report lag and audit fees: Evidence from the United States. *Journal of Financial Reporting and Accounting, ahead-of-print*. [CrossRef]
- Lobo, Gerald J., Michael Neel, and Adrienne Rhodes. 2018. Accounting comparability and relative performance evaluation in CEO compensation. *Review of Accounting Studies* 23: 1137–76. [CrossRef]
- McNichols, Maureen F. 2002. Discussion of the quality of accruals and earnings: The role of accrual estimation errors. *The Accounting Review* 77: 61–69. [CrossRef]
- Mitra, Santanu, Donald R. Deis, and Mahmud Hossain. 2009. The association between audit fees and reported earnings quality in pre-and post-Sarbanes-Oxley regimes. *Review of Accounting and Finance* 8: 232–52. [CrossRef]
- Nam, Jonathan. 2020. Financial reporting comparability and accounting-based relative performance evaluation in the design of CEO cash compensation contracts. *The Accounting Review* 95: 343–70. [CrossRef]
- Namakavarani, Omid Mehri, Abbas Ali Daryaei, Davood Askarany, and Saeed Askary. 2021. Audit committee characteristics and quality of financial information: The role of the internal information environment and political connections. *Journal of Risk and Financial Management* 14: 273. [CrossRef]
- Ridzky, M., and F. Fitriany. 2022. The Impact of Abnormal Audit Fees on Audit Quality: A Study of ASEAN Countries. *KnE Social Sciences*, 395–403. [CrossRef]
- Saleh, Mohamed A., and Yasmine M. Ragab. 2023. Determining audit fees: Evidence from the Egyptian stock market. *International Journal of Accounting & Information Management* 31: 355–75.

- Santos Jaén, José Manuel, Gema Martín de Almagro Vázquez, and María del Carmen Valls Martínez. 2023. Is earnings management impacted by audit fees and auditor tenure? An analysis of the Big Four audit firms in the US market. *Oeconomia Copernicana* 14: 899–934. [CrossRef]
- Sarhan, Ahmed A., and Christopher J. Cowton. 2024. Combatting bribery and corruption: Does corporate anti-corruption commitment lead to more or less audit effort? In *Accounting Forum*. Abingdon: Routledge, pp. 1–24.
- Shandiz, Mohsen Tavakoli, Farzaneh Nassir Zadeh, and Davood Askarany. 2022. The Interactive Effect of Ownership Structure on the Relationship between Annual Board Report Readability and Stock Price Crash Risk. *Journal of Risk and Financial Management* 15: 268. [CrossRef]
- Shehadeh, Ayman, Mahmoud Daoud Daoud Nassar, Husam Shrouf, and Mohammad Haroun Sharairi. 2024. How Audit Fees Impact Earnings Management in Service Companies on the Amman Stock Exchange through Audit Committee Characteristics. International Journal of Financial Studies 12: 32. [CrossRef]
- Simunic, Dan A. 1980. The pricing of audit services: Theory and evidence. Journal of Accounting Research 18: 161–90. [CrossRef]
- Sonu, Catherine Heyjung, Hyejin Ahn, and Ahrum Choi. 2017. Audit fee pressure and audit risk: Evidence from the financial crisis of 2008. *Asia-Pacific Journal of Accounting & Economics* 24: 127–44.
- Utomo, St Dwiarso, and Zaky Machmuddah. 2023. Governance Disclosure, Integrated Reporting, CEO Compensation, Firm Value. In World Conference on Information Systems for Business Management. Singapore: Springer Nature, pp. 303–10.
- Wen, Huiyu, Jincheng Fang, and Haoyu Gao. 2023. How FinTech improves financial reporting quality? Evidence from earnings management. *Economic Modelling* 126: 106435. [CrossRef]
- Wysocki, Peter. 2010. Corporate compensation policies and audit fees. Journal of Accounting and Economics 49: 155–160. [CrossRef]
- Zadeh, Farzaneh Nassir, Davood Askarany, and Solmaz Arefi Asl. 2022. Accounting Conservatism and Earnings Quality. *Journal of Risk and Financial Management* 15: 413. [CrossRef]
- Zhang, Honghui, Hedy Jiaying Huang, and Ahsan Habib. 2018. The effect of tournament incentives on financial restatements: Evidence from China. *The International Journal of Accounting* 53: 118–35. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.



Article



Lebanon's Economic Development Risk: Global Factors and Local Realities of the Shadow Economy Amid Financial Crisis

Samar F. Abou Ltaif ^{1,*}, Simona Mihai-Yiannaki ¹ and Alkis Thrassou ²

- Department of Accounting, Economics and Finance, School of Business Administration, European University Cyprus, Nicosia 2404, Cyprus; s.mihai@euc.ac.cy
- ² Department of Management, School of Business, University of Nicosia, 46 Makedonitissas Avenue, Nicosia 2417, Cyprus; thrassou.a@unic.ac.cy
- * Correspondence: sf191834@students.euc.ac.cy; Tel.: +961-71801008

Abstract: The shadow economy's size and impact remain subjects of extensive research and debate, holding significant implications for economic policy and social welfare. In Lebanon, the ongoing crisis since 2019 has exacerbated severe economic challenges, with the national currency's collapse, bank crisis, and foreign reserve deficits. The World Bank reports Lebanon's financial deficit surpassed \$72 billion, three times the GDP in 2021. Despite a drastic decline in GDP, imports have surged to near-pre-crisis levels, exacerbating economic woes and indicating a constant outflow of foreign currencies. Considering such contracting facts, this paper aims to investigate global factors influencing the shadow economy and discern their manifestations in Lebanon during financial crises. Our methodology involves a comprehensive literature review, alongside a case study approach specific to Lebanon. This dual-method strategy ensures a detailed understanding of the shadow economy's impact and the development of actionable insights for policy and economic reform. Through this approach, we seek to contribute to a nuanced understanding of Lebanon's economic landscape and provide valuable guidance for policy decisions aimed at reducing corruption, promoting transparency, and fostering a robust formal economy. The increase in the shadow economy raises the formal economy risk, as resources and activities diverted to informal channels hinder the growth and stability of the official economic sector. Although focusing on Lebanon, this analysis deepens the comprehension of the economic landscape and provides valuable guidance for policymakers, researchers, and stakeholders, aiming to address the root causes of informal economic activities and promote sustainable growth in developing countries in general.

Keywords: institutional economics framework; formal economy risk; Lebanon; sustainable development; shadow economy

1. Introduction

1.1. Research Context

The shadow economy—comprising unregulated economic activities that evade taxes and legal frameworks—presents a significant economic development risk due to its profound implications for economic stability and social welfare. The persistence of the shadow economy undermines formal economic structures, hampers effective policy implementation, and exacerbates socio-economic disparities. Understanding the dynamics of the shadow economy is crucial, especially during periods of financial crisis, as it can reveal how unregulated activities contribute to broader economic development risks.

Recent literature underscores the multifaceted nature of the shadow economy. Canh et al. (2021) highlight the influence of economic integration and institutional quality on shadow economic activities, while Chletsos and Sintos (2024) examine how International Monetary Fund (IMF) interventions can shape informal sectors. Dreher and Schneider (2010) explore the interplay between corruption and the shadow economy, revealing its extensive socio-economic consequences. Additionally, the shadow economy might also raise



Citation: Abou Ltaif, Samar F., Simona Mihai-Yiannaki, and Alkis Thrassou. 2024. Lebanon's Economic Development Risk: Global Factors and Local Realities of the Shadow Economy Amid Financial Crisis. *Risks* 12: 122. https://doi.org/10.3390/ risks12080122

Academic Editors: Eulália Mota Santos and Margarida Freitas Oliveira

Received: 22 June 2024 Revised: 30 July 2024 Accepted: 30 July 2024 Published: 31 July 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). environmental risks, as shown by Cozma et al. (2021), who highlight its significant impact on illegal logging. The study reveals how corruption and economic crimes, core elements of the shadow economy, exacerbate deforestation rates. By analyzing both qualitative and quantitative data, the research underscores the role of governance quality, press freedom, and wood export share in driving deforestation, thereby informing policies to combat illegal logging and protect the environment. This illustrates the extensive socio-economic and environmental consequences of the shadow economy. These studies illustrate how various factors, including governance quality and economic policies, contribute to the shadow economy's expansion, risks, and persistence.

Lebanon's late and ongoing economic crisis, which began in 2019, provides a pertinent context for studying the shadow economy. The collapse of the national currency, escalating trade deficits, and severe economic contraction (World Bank Group 2021b) have exacerbated the challenges facing Lebanon's formal economy.

According to one of the World Bank reports (World Bank Group 2021b), Lebanon's financial deficit exceeded \$72 billion in 2021, which was three times the country's GDP. Furthermore, considering the downward trajectory of GDP from about \$52 billion in 2019 to less than half in 2022 (\$25.46 billion), the surge in imports exacerbates Lebanon's economic woes, underscoring the magnitude of the challenge at hand. Imports have remarkably rebounded to their pre-crisis levels, while the trade deficit continues to expand due to a decline in exports. In 2022, as per data from Banque Du Liban (BDL, the Central Bank of Lebanon), imports reached a staggering \$19.05 billion, representing a significant increase of \$5.41 billion compared to the previous year, marking a rise of 39.68% from 2021 (\$13.64 billion) and a striking 68.47% surge from 2020 (\$11.31 billion) (Banque du Liban (BDL) 2023b; Gemayel 2023). These import levels are almost equivalent to the figures recorded in the five years prior to the crisis, as shown in Figure 1.



Figure 1. GDP and import fluctuations in Lebanon from 2010 to 2022 (authors' compilation of data from Banque du Liban (BDL) (2023b); World Bank 2023b).

The surge in imports amid Lebanon's economic crisis, coupled with the persistent downturn in GDP, signals a concerning trend: the formal economy is struggling to return to pre-crisis levels. Despite economic hardships, imports have rebounded to near-pre-crisis levels, indicating a reliance on external goods and services and thus a constant outflow of foreign currencies. With the recurrence of financial crises and their adverse effects, it becomes crucial to shed light on the informal economy, commonly known as the shadow economy. This sector represents a significant portion of a country's GDP, especially in developing nations. This suggests that the formal economy is not recovering as expected, potentially fueling informal economic activities and the expansion of the shadow economy.

The escalation of the shadow economy heightens the risk to the formal economy, diverting resources and activities into informal channels that impede the growth and sta-

bility of the official economic sector. Sustainability risks emerge across economic stability, environmental impact, social consequences, governance challenges, and long-term development goals. Addressing these risks requires robust policies to promote transparency, strengthen regulatory frameworks, and foster the formalization of informal activities, ensuring sustainable economic, environmental, and social outcomes.

In addition to internal concerns within companies, such as operational efficiency, internal controls, and corporate indebtedness, adopting strategic and personalized approaches to financing and risk management (Tavares et al. 2024), as well as financial education (Santos and Tavares 2020), remains essential for mitigating external risks. The shadow economy, in particular, can significantly impact risk management practices and overall performance by introducing uncertainties and unregulated activities that disrupt formal economic operations and planning.

In this paper, we utilize a dual-method approach: a comprehensive literature review and a case study specific to Lebanon. The literature review includes peer-reviewed journal articles, books, and reputable reports, ensuring a thorough understanding of the shadow economy's global and local dynamics. The case study provides contextual insights into Lebanon's economic conditions, allowing us to apply theoretical findings to the local context. This methodology enables a detailed examination of the shadow economy's impact and its implications for Lebanon's economic development.

Thus, the aim of this paper is to investigate the global factors influencing the shadow economy and to discern their specific manifestations within Lebanon, particularly during times of the financial crisis since 2019. By combining a broad literature review with a focused case study, we aim to provide a nuanced analysis that informs policy and economic reform efforts in Lebanon.

1.2. Research Aim, Questions and Value

Arising from the above, this study investigates two critical questions:

- 1. What are the key global factors influencing the shadow economy?
- 2. How do these factors and indicators manifest in the context of Lebanon during times of its financial crisis?

The research questions aim to elucidate two key aspects of the shadow economy's dynamics. The first question seeks to identify the general factors influencing the shadow economy, providing a broader understanding of international drivers that affect informal economic activities. The second question focuses on how these global factors specifically manifest within Lebanon during its financial crisis, aiming to apply these international insights to the local context.

The methodology is based on secondary data analysis and a comprehensive theoretical review at both the local and international levels. By concentrating on the variables identified in the literature, we analyze their applicability and relevance to the Lebanese context, thereby assessing the indicators and situation in Lebanon. Moreover, we address the gap in the existing literature by adopting a theoretical framework grounded in institutional economics to analyze the dynamics of the shadow economy in Lebanon. Unlike previous studies that predominantly rely on quantitative methods, our qualitative approach allows for a deeper exploration of contextual factors and socio-economic dynamics unique to Lebanon.

The significance of our research lies in its potential to contribute to the existing body of knowledge surrounding the factors impacting the shadow economy in general and in Lebanon in particular, with a focus on the effects on government revenues and social welfare. By unraveling the complexities of the shadow economy, we aim to provide actionable insights that can inform policy decisions for reducing corruption, promoting transparency, and fostering a robust formal economy. Through our analysis, we strive to deepen the understanding of Lebanon's economic landscape and offer valuable guidance for policymakers, researchers, and stakeholders. Ultimately, our work endeavors to address the root causes of informal economic activities and promote sustainable economic development in Lebanon.

The paper begins with a literature review, exploring the informal economy and its ties to corruption and formal economic systems. It adopts an institutional economics framework to analyze secondary data on Lebanon's shadow economy. Then, it discusses key indicators of Lebanon's informal economy: unemployment, taxation, government regulations, financial crises, smuggling, and cash circulation. Policy recommendations follow. Finally, the paper emphasizes the need for structural reforms to address fiscal instability and promote economic and social welfare in Lebanon.

2. Literature Review

2.1. An Overview of the Informal Economy

The informal or shadow economy encompasses activities that occur outside the purview of "bureaucratic public and private" systems (Blanton and Peksen 2021). These activities involve elements such as smuggling, bribery to access desired services or products, and informal employment driven by market structures (Mishchuk et al. 2020).

Empirical studies, such as the work conducted by Schneider (2005), have been instrumental in estimating the size of the shadow economy in various countries, including those in the developing, transitioning, and highly developed OECD economies between 1990 and 2000. Schneider's study revealed that the shadow economy accounted for 41% of the official GDP in developing countries, 38% in transition economies, and 17% in OECD countries. A key driver for the growth of the shadow economy is the burden imposed by taxation and social security contributions. It was observed that a 1% increase in the shadow economy in developing countries leads to a 0.6% decrease in the growth rate of the official GDP, while in developed and transitioning economies, the shadow economy increases by 0.8% and 1.0%, respectively. Furthermore, the size of the shadow economy varies significantly across countries and time periods, with developing and transitioning economies typically exhibiting larger shadow economies than highly developed OECD nations (Schneider 2005).

As indicated by newer estimates from Medina and Schneider (2018), in Europe, the shadow economy is thought to account for approximately 20% of the GDP. However, in Eastern Europe and post-Soviet regions, the figures are even higher, ranging from 25% to 45% (Medina and Schneider 2018). Recent studies further expand on these findings:

Medina and Schneider's 2018 study utilized a light intensity method as an alternative indicator within the Multiple Indicators Multiple Causes (MIMIC) framework. This study provided estimates of the shadow economy for 158 countries from 1991 to 2015, addressing earlier criticisms by implementing the Predictive Mean Matching (PMM) method to enhance accuracy and reliability. This study underscores the shadow economy's complexity and the necessity for robust estimation methods (Medina and Schneider 2018).

Additionally, a 2023 report by the International Monetary Fund (IMF) reviewed shadow economies worldwide over the past 20 years, highlighting significant variations in their size across different regions and time periods. This report reinforced that developing and transitioning economies typically exhibit larger shadow economies than developed OECD nations. It also discussed the socio-economic impacts, such as the role of informal networks and social capital in perpetuating corruption and unethical behavior within shadow economies (International Monetary Fund (IMF) 2023).

Furthermore, a study published in Emerald Insight explored the interrelationship between corruption and the shadow economy, using data from the World Development Indicators database and building upon Medina and Schneider's 2018 estimates. The study found that shadow economies often rely on informal networks, which can facilitate corrupt practices due to a lack of accountability and transparency. This research further solidifies the understanding of how shadow economies and corruption are interlinked (Nguyen and Liu 2023). While the shadow economy can be viewed as an engine of innovation, it also brings about significant social side effects, such as unprotected working conditions due to its unregulated and clandestine nature (Kraemer-Mbula and Wunsch-Vincent 2016). Other negative effects of the shadow economy include a high poverty rate, high share of income spent on basic expenditures (mainly food), lack of housing opportunities, and lack of social safety (Mishchuk et al. 2020). Furthermore, it has also been empirically proven the effects of the shadow economy on government revenues, unfair competition over pricing (between entities paying taxes and those functioning under an informal economy), and impacts on economic growth in developing countries (Tanzi 2002).

In the context of Lebanon, the informal economy plays a particularly significant role due to the country's ongoing economic crisis and political instability. Recent estimates suggest that Lebanon's informal sector is notably larger, exacerbated by the country's severe economic challenges. The International Monetary Fund (IMF)'s (2023) report highlights that Lebanon's shadow economy has expanded significantly as the formal economy struggles to recover, driven by regulatory deficiencies and economic collapse. Lebanon's shadow economy is intertwined with corruption and inadequate governance, exacerbating economic instability and undermining social welfare (Nguyen and Liu 2023). The impact of the shadow economy in Lebanon includes exacerbated poverty rates, increased reliance on informal networks, and reduced government revenues. Addressing these issues is crucial for developing targeted policies to enhance transparency, improve governance, and formalize informal economic activities, thus promoting sustainable economic development in Lebanon.

The next section will focus on the interconnections between corruption, the formal economy, and the shadow economy. By integrating this discussion into the literature review, we aim to offer a comprehensive understanding of the global dynamics influencing the shadow economy and lay the groundwork for analyzing its specific manifestations in Lebanon. This exploration will highlight how global factors intersect with local conditions in Lebanon, where corruption and institutional weaknesses significantly shape the shadow economy. By examining these interconnections, we seek to uncover the mechanisms and dynamics pertinent to Lebanon's economic context, thereby grounding our analysis in both global perspectives and local realities.

2.2. Interlinking Corruption, Formal Economy, and Shadow Economy

To investigate the complex relationship between corruption, the formal economy, and the shadow economy, it is important to establish clear conceptual boundaries. Corruption, defined as the misuse of public resources for personal gain, has been widely recognized as a major impediment to economic and social development (Rose-Ackerman 1999). The formal economy refers to registered economic activities that are governed by legal frameworks, regulations, and taxation systems. On the other hand, the shadow economy encompasses unregistered economic activities that operate outside the purview of official regulations and often involve noncompliance with tax obligations (Schneider and Enste 2000).

Previous research exploring the interaction between corruption and the shadow economy has focused on estimating the size and impact of the shadow economy and identifying the factors that contribute to its growth (Schneider and Buehn 2016; Sharipov 2015; Friedman 2014). Poor institutional quality and tax evasion have been identified as key drivers of the shadow economy (LaPorta et al. 1999; Tanzi 1999). Institutional economics emphasizes the role of institutional quality, including corruption, regulatory discretion, judicial systems, the rule of law, and bureaucracy, in shaping economic behavior (Phuc Canh 2018; Schneider and Enste 2000).

Corruption has a detrimental effect on institutional quality, leading to higher labor costs for official economic activities and incentivizing economic agents to engage in informal or concealed activities (Phuc Canh 2018; Schneider and Enste 2000). Inadequate constraints on government powers, regulatory burdens, and failing judicial systems create

an environment where individuals may opt for informal economic activities and participate in the shadow economy (Torgler and Schneider 2009; Dreher et al. 2009; Berdiev et al. 2018).

Conversely, the shadow economy can also influence corruption and the formal economy. Some studies suggest that the shadow economy can act as a substitute for corruption, allowing economic agents to avoid official taxes and bureaucracy (Choi and Thum 2005; Ackerman 1997). Factors such as political stability, competent governance, and effective regulations have been found to be negatively correlated with the size of the shadow economy, while bureaucracy shows a positive association (Friedman 2014). However, it is important to note that the relationship between corruption, the formal economy, and the shadow economy is multifaceted and can work in different ways depending on the specific context and circumstances.

The presence of a significant shadow economy poses risks to economic development and societal well-being. It leads to fiscal losses due to tax evasion, contributes to income inequality, and distorts the accurate measurement of income distribution. The shift of economic activities from the formal to the informal sector exacerbates income polarization, while the impact of the shadow economy on secondary income generation undermines social safety nets, resulting in challenges such as poverty, inefficient cost structures, and limited housing opportunities (Mishchuk et al. 2020).

Given these conceptual boundaries and the complexities of the relationship between corruption, the formal economy, and the shadow economy, it is imperative to understand the underlying mechanisms, dynamics, and potential policy implications. By examining these interconnections, this study sheds light on the factors that influence the size and behavior of the shadow economy and how corruption and institutional quality play a role in shaping these dynamics. Figure 2 summarizes the main factors related to the shadow economy and the cycle it creates based on the above literature review.



Figure 2. Summary of factors related to the shadow economy (source: author).

Schneider (2000) employed causal models incorporating latent variables to estimate the size of the shadow economy across different countries. These latent variables are not directly observable but have operational implications for the relationships among observable variables. The observable variables act as causes and serve as indicators of the hidden variables.

In his research, Kanniainen et al. (2004) built on that causal model. He utilized the Multiple Indicators Multiple Causes (MIMIC) method and formulated six hypotheses. He postulated that the shadow economy would positively respond to the following factors:

1. Tax rates (both direct and indirect) and social security contributions;

- 2. State regulation indicated by the percentage of total employment allocated to public administration;
- 3. Lower tax morality;
- 4. Decrease or deterioration of state transfer and public goods quality;
- 5. Higher unemployment and lower GDP per capita;
- 6. The amount of cash transactions.

The econometric results of Kanniainen's study suggest that taxation and social security variables significantly influence the expansion of underground economies. The research underscores the importance of controlling the tax burden to reduce the size of these economies and emphasizes the need for public education regarding the benefits of contributing to the formal economy. Additionally, economic factors such as unemployment and GDP per capita influence individuals' willingness to participate in the official economy. Social benefit policies may have more significant employment benefits than previously believed, and state transfers may act as incentives to draw individuals out of the underground economy. Furthermore, the size of a shadow economy is influenced by a country's economic performance, exhibiting significant variation across nations and over time.

Schneider and Buehn (2016) explored various methods for gauging the dimensions of the shadow economy and delved into their respective strengths and limitations. Their study serves a dual purpose. Firstly, it underscores the absence of an impeccable method for quantifying the size and evolution of the shadow economy. While delving into greater detail on the MIMIC method, which offers flexibility in obtaining macro-level estimates, the paper secondly concentrates on defining the shadow economy and its causative factors. It also provides a comparative analysis of the shadow economy's size using diverse estimation techniques. Their findings reveal significant disparities in measurement methods, yielding divergent and sometimes inconsistent results. The survey method predominantly focuses on households, occasionally overlooking firms, leading to issues such as nonresponses, erroneous data, and limited financial figures instead of value-added metrics. The discrepancy method grapples with challenges arising from rough estimates, unclear assumptions, and a lack of transparency in calculation procedures. The monetary and electricity methods tend to generate elevated estimates, but only offer macro-level data, posing complications in converting electricity consumption into value-added estimates. The widely used MIMIC method provides relative coefficients rather than absolute values, and it is highly sensitive to changes in data and specifications, making it difficult to discern causal factors and indicators. The overall conclusion is that there is no single flawless method, underlining the importance of employing multiple approaches to enhance our comprehension of the shadow economy's size and evolution (Schneider and Buehn 2016).

Given the limitations inherent in estimating the size of the shadow economy and the complexities of the Lebanese market dynamics, this paper opts for a theoretical framework rather than quantitative analysis. While quantitative methods have traditionally been employed to measure the shadow economy, the intricacies of accurately quantifying informal economic activities in Lebanon pose significant challenges. Factors such as data availability, reliability, several exchange rates, and the dynamic nature of informal transactions contribute to the difficulty in conducting precise quantitative assessments. Therefore, a theoretical approach allows us to delve deeper into the underlying mechanisms and interdependencies between corruption, the formal economy, and the shadow economy. By examining existing literature and theoretical frameworks, we aim to provide a comprehensive understanding of the factors influencing the shadow economy in Lebanon and their implications for economic policy and social welfare. This theoretical lens enables us to explore the conceptual boundaries of corruption, institutional quality, and economic behavior, laying the groundwork for insightful analyses and policy recommendations.

2.3. Developing an Institutional Economics Framework

Following the concepts from the general literature review, our study adopts an institutional economics framework to delve deeper into the dynamics of the shadow

economy in Lebanon. Within this framework, our approach is qualitative, focusing on synthesizing insights from existing research and contextualizing them within Lebanon's socio-economic context.

Institutional economics examines how institutions, such as legal and regulatory frameworks, governance structures, and societal norms, shape economic behavior (Hodgson 1998; Dreher et al. 2009) Contemporary institutionalist thinking warns against excessive reliance on quantitative models, arguing that they often overlook the importance of institutions and fail to adequately address uncertainty. Instead, it advocates for a nuanced understanding of institutional dynamics and their impact on economic behavior and societal outcomes (McMaster 2012).

Given the significance of corruption, institutional quality, and regulations in influencing informal economic activities, this framework provides a pertinent lens through which to analyze the complex relationship between the formal and shadow economies (Murphy et al. 1993; Pluskota 2020).

The institutional economics approach allows us to explore the interactions between institutional factors and economic agents, shedding light on the complex interplay between formal regulations and informal practices within the Lebanese context.

3. Methodology

The methodology of this paper follows the structure represented in Figure 3, where the main body is organized by themes and topics derived from the literature review. Initially, we conducted a comprehensive review of existing literature to identify general factors, causes, and indicators related to the shadow economy. Subsequently, we focused on analyzing the applicability of these variables to the context of Lebanon. Through this approach, we aimed to provide a synthesized overview of the dynamics of the shadow economy in Lebanon, informed by insights gleaned from previous research and verified against reliable data sources.



Figure 3. Study methodological steps (Saunders et al. 2007; Braun and Clarke 2006).

Inclusion criteria for the comprehensive literature review focused on peer-reviewed journal articles, books, and reputable reports published in English that are directly relevant to the shadow economy, corruption, and the formal economy internationally and with a particular emphasis on Lebanon or similar economies. Exclusion criteria eliminated nonpeer-reviewed articles, opinion pieces, and studies that were not available in full text. While some sources were accessed before this period, the primary search process for the review was conducted over an extended timeframe from January to June 2024. This comprehensive period included a meticulous evaluation of titles, abstracts, and full texts of eligible articles to ensure their relevance and quality for inclusion in this paper. In more details, throughout the paper, in addition to reliable sources like the World Bank, IMF, and BDL, we identified references indexed in Scopus and Web of Science or both databases.

Furthermore, a significant addition to the methodology is the incorporation of a case study related to Lebanon. This approach allows us to contextualize theoretical insights within the specific socio-economic conditions of Lebanon, thereby enhancing the relevance and applicability of our findings.

4. Indicators of Lebanon's Informal Economy: Investigating Key Factors

A study conducted by Kareh (2020) utilized the monetary approach for the estimation and analysis of the shadow economy's size and its ramifications within the context of Lebanon. The investigation spanned from 1998 to 2018 and revealed that the shadow economy constituted approximately 36.61% of the GDP during this period, as represented in Table 1. Notably, in 2018, tax evasion represented 30.04% of the shadow economy, a figure nearly equivalent to Lebanon's budget deficit. These findings underscore the existence of a significant underground economy in Lebanon, primarily driven by substantial tax evasion and expansion of the informal public sector. Additionally, the surge in cash transactions within this economy can be attributed to the support extended to Syrian refugees by both local and international NGOs, as well as the facilitative measures adopted by various banks (Kareh 2020).

Table 1. Lebanon shadow economy estimation from 1998 till 2018 (Kareh 2020).

Years	1998	1999	2000	2001	2002	2003	2004	3005	2006	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018
Shadow Economy % GDP	36.87	40.36	29.26	35.2	33.67	30.42	30.2	31.99	29.57	28.23	34.81	39.81	31.8	36.11	38.25	34.27	34.24	38.94	34.15	36.98	36.61

Lebanon ranks among the most corrupted countries worldwide, according to the CPI barometer (Transparency International 2022), and due to a lack of data and transparency, estimating the size of the shadow economy in Lebanon became increasingly challenging after the onset of the crisis in 2019. Specifically, in 2019, Lebanon ranked 137th out of 180 countries with a CPI score of 28, reflecting significant corruption, and by 2022, its position worsened to 150th with a CPI score of 24, indicating continued and severe corruption challenges.

The presence of multiple exchange rates further complicated the estimation process. As of 2021, the World Bank estimated that the cash economy in Lebanon was valued at approximately \$10 billion, constituting over 45% of the country's GDP (World Bank Group 2021a). This estimation reflects data up to the end of 2021. By early 2023, the situation continued to be dire, with ongoing economic instability affecting accurate assessments and projections (World Bank 2023a).

Estimating the volume of Lebanon's informal economy is challenging due to the lack of reliable data. Medina and Schneider (2018), in their analysis of 157 shadow economies, mentioned that their model did not fully account for the significant influxes of refugees, potentially leading to underestimations of shadow economies in countries like Lebanon, Syria, Jordan, or Turkey. Additionally, since the war in Syria, the number of refugees in Lebanon has significantly increased, with the government registering 1.5 million refugees, making Lebanon the country with the "highest number of refugees per capita in the world" (European Civil Protection and Humanitarian Aid Operations 2023).

Considering so, examining the extent and dynamics of Lebanon's shadow economy is essential given the country's distinct socio-political and economic circumstances, marked by political discord, civil turmoil, security issues, and a substantial influx of refugees. Furthermore, Lebanon's economic downturn, compounded by corruption, fiscal mismanagement, and currency devaluation, highlights the critical importance of comprehending the impact of informal economic practices on Lebanon's economic structure and societal welfare.

We will explore the indicators of the informal economy in Lebanon by narrowing our focus to key determinants previously identified in Section 2 of the literature review that positively influence the shadow economy. These factors include the taxation system, regulatory decisions facilitating smuggling, currency circulation, fiscal deficit, and fostering the expansion of informal economic activities. Furthermore, we will also consider the downturn in GDP and employment levels.

4.1. Higher Unemployment and Lower GDP per Capita

Lebanon has a long history of political tensions, regional unrest, civil wars, and security challenges. Furthermore, corruption, U.S. sanctions, and the presence of political parties designated as "terrorists" by the international community have contributed to budget deficits and an enormous public debt (Azar and Abdallah 2019). These conditions, coupled with financial engineering that drained the market and commercial banks of foreign currencies, resulted in significant losses for the central bank and increased bank exposures to sovereign risk. The situation deteriorated in 2019, with banks raising interest rates to approximately 10% on foreign currency deposits (even higher for local currency) to attract new flows of foreign currency (Banque du Liban (BDL) 2023b). However, due to a lack of trust, banks faced a liquidity crisis and implemented unofficial capital controls, preventing depositors from withdrawing their funds. This led to a rapid devaluation of the currency, with the market exchange rate reaching approximately 42,000 LBP/USD by the end of 2022 (after being pegged at 1500 LBP/USD since the 1990s), resulting in business closures and significant negative changes in the country's economic indicators between 2019 and 2021.

During this period, Lebanon's GDP contracted by 50%, as shown in Figure 1 (World Bank 2023b), while the unemployment rate increased from 11.4% to 29.6% (Azzi 2022). The inflation rate also rose from 3% to 154.9% (World Bank Group 2021b), and the poverty rate increased from 28% to over 55%. A survey conducted by the Central Administration of Statistics in 2022 revealed that around 49% of unemployed individuals had been searching for work for more than one to two years (Baff 2022). These statistics indicate the significant economic challenges facing Lebanon.

4.2. Tax Rates in Lebanon

The taxation system poses a significant challenge to the Lebanese economy, heavily relying on indirect taxation, including value-added taxes, excise taxes, and tariffs, which account for a substantial portion of total revenues. These indirect taxes are regressive in nature, disproportionately affecting the middle class and the poor. In comparison to other countries, Lebanon has a very low share of revenues collected through a progressive tax scheme, with only around 11% of taxes being levied progressively on wages, property income, and inheritance. This is significantly lower than comparative countries, where the share of progressive taxes is two to three times higher. The taxation of personal income in Lebanon is based on a separate rate for each income source, resulting in an archaic system (Azzi 2022).

Lebanon's tax system includes progressive rates for labor and rental income, while other categories such as capital gains and dividends are taxed at flat rates. However, the revenue from all types of income taxes is relatively low, accounting for 2.7% of GDP (refer to Figure 4). Corporate profit is taxed at a flat rate of 17%, representing a low share of total revenues. Inheritance and gift taxes are progressive, but their revenue is low, amounting to 0.2% of GDP. Real estate registration fees are flat at 5% of the property value and contribute

to 1% of GDP. Lebanon lacks taxes that promote public goods such as environmental protection or regional development (Bifani et al. 2021).

According to the study by Bifani et al. (2021), Lebanon has low tax rates compared to other countries, but tax revenues are significantly lower due to widespread tax leakages and inefficiencies in tax collection. The compliance gap for value added tax (VAT) and lost customs revenue at the border are estimated to be around 3.3% and 1% of GDP, respectively. Banking secrecy laws hinder tax collection from liberal professions and capital gains earned abroad, and there are exemptions and loopholes in the inheritance tax system. These leakages have a regressive nature, as direct taxes are more rigorously assessed for low-income taxpayers, while wealthier taxpayers have more opportunities for tax avoidance. Reforms are needed to improve tax collection, including addressing banking secrecy, exemptions, and loopholes, and making the tax system more progressive (Azzi 2022).

The challenges posed by Lebanon's taxation system, highlighted in Section 4.2, are compounded by significant fiscal losses and budget deficits, as discussed in the next section.

4.3. Fiscal Losses and Budget Deficits

From 2010 to 2022, Lebanon faced significant fiscal challenges characterized by fluctuations in its fiscal balance, government spending, and consumption amid economic instability. The fiscal balance varied markedly, with deficits reaching a peak in 2018 at -11.38% of GDP (Trading Economics 2023), pointing to unsustainable fiscal practices and the government's dependency on public debt and in financial engineering that drained the foreign reserves from the banks in Lebanon (International Monetary Fund (IMF) 2019; Azar and Abdallah 2019). A temporary fiscal balance improvement in 2021, at 1.2% of GDP, reflected drastic government spending cuts rather than genuine fiscal health. Government expenditure trends further illustrated the fiscal strain, with spending cuts in 2021, severely impacting social services and infrastructure and exacerbating economic contraction and social unrest (International Monetary Fund (IMF) 2023). Lebanon's fiscal experience from 2010 to 2022 emphasizes the critical need for structural reforms to achieve fiscal stability, reduce income inequality, and ensure economic and social well-being.



Figure 4. Fiscal balance and government expenditures (authors' compilation of data from Trading Economics 2023).

The financial crisis, exacerbated by fiscal challenges and taxation inefficiencies, amplifies the impact of smuggling activities in Lebanon, which contributes to the growth of the shadow economy and further strains the formal economy.

4.4. Financial Crisis and Smuggling in Lebanon

The smuggling relationship between Lebanon and Syria has a long-standing history, with around 150 illegal exchange points across the border (Daher 2022). In recent years, smuggling activities have taken on new dynamics, especially since the onset of the 2019 crisis. The most profitable among these activities is the smuggling of fuel oil from Lebanon to Syria, which occurred between October 2019 and October 2021 (LBCI Lebanon 2022).

The fuel price was much cheaper in Lebanon, being subsidized at a low official rate of 1500 LBP/USD, while the market rate was several times higher in both Lebanon (as represented in Figure 5) and Syria due to the ongoing crisis. Hezbollah oversees the illegal crossing points on the Syrian-Lebanese border, allowing them to regulate the flow of goods and relying on local actors on the Syrian-Jordanian border for their smuggling activities (Daher 2022).



Figure 5. Market exchange rate in Lebanon (graph created by the authors using data from Lirarate 2023).

A study funded by the European Union and conducted by Daher (2022) suggests that Hezbollah, along with Lebanon's top political parties, has ties to customs and port officials that allow them to ship and receive their own cargos, resulting in an estimated loss of one to two billion dollars each year due to the evasion of customs duties. Corruption is prevalent at the port, with employees at every level, as well as state security officers, receiving bribes to speed up shipments, lower the value of imports, or avoid inspections.

In addition to fuel smuggling, other activities, such as the illegal trade of subsidized goods, tobacco, and drugs, also contribute to the growth of the shadow economy. Smuggling networks exploit the country's weak borders, corrupt officials, and the desperate economic situation to carry out their operations, further straining the formal economy and exacerbating the financial crisis in Lebanon (Daher 2022).

Section 4.5 discusses the financial crisis that has led to a surge in cash transactions and cash in circulation, exacerbating liquidity challenges and raising concerns about tax enforcement, money laundering, and terrorist financing.

293

4.5. Amount of Cash in Circulation

Banks in Lebanon imposed severe restrictions on accessing depositor foreign currency accounts, converting deposits to Lebanese pounds (LBP) upon withdrawal. Consequently, the volume of cash in circulation surged significantly from LBP 6474 billion in September 2019 to LBP 80,171 billion by December 2022 (Banque du Liban (BDL) 2023a) (Figure 6). This surge stemmed from various factors, including depositor distrust in the banking system, rapid currency depreciation, and regulatory directives prompting depositors to convert foreign currency holdings into LBP. These developments exacerbated the liquidity crisis and underscored the challenges confronting Lebanon's financial stability and broader economic framework.

Currency in Circulation in billions of LBP



Figure 6. Fluctuation in currency in circulation in billions of LBL from January 2018 to September 2023 (graph created by the authors using data from Banque du Liban (BDL) 2023b).

The cash-based economy in Lebanon has made it challenging to enforce taxes and has raised concerns about money laundering and terrorist financing. Western governments are also worried that illicit transactions may rise as cash transactions are harder to track. The U.S. Department of the Treasury has sanctioned several entities and individuals in Lebanon for compliance issues, which are expected to increase. Recently, a Lebanese money exchanger was sanctioned for alleged ties to Hezbollah, a heavily armed, Iran-backed group (Gebeily 2023).

4.6. Government and Central Bank Regulations

Government and central bank decisions have played a crucial role in shaping Lebanon's economy. Preceding the 2019 crisis, Lebanon maintained a fixed exchange rate regime, pegging the Lebanese pound (LBP) to the US dollar (USD) at a rate of 1500 LBP/USD through financial engineering methods and unsustainable fiscal policies (Abdel Samad 2021). These included subsidies offered at artificially low official rates and the implementation of measures to support the fixed exchange rate system.

Amid Lebanon's ongoing financial instability, exacerbated by a banking sector crisis, the Lebanese Central Bank responded by implementing various measures. These included utilizing foreign reserves, amounting to \$1.5 billion, to fulfill Eurobond payments and provide subsidies for vital commodities like fuel, wheat, medicine, and food. In an effort to alleviate the shortage of foreign currency liquidity and shield consumers from inflation, the Central Bank introduced new subsidies and circulars (Banque Du Liban (BDL) 2024)

concerning the withdrawal of foreign currencies. However, these decisions had unintended repercussions, including market distortion and ineffectiveness, primarily benefiting the affluent. For instance, the subsidies, pegged at an official rate of 1500 LBP/USD despite a market rate of around 30,000 LBP/USD, incentivized smuggling across the Syrian border and stockpiling by the privileged segment of the population (Nasrallah 2020). Additionally, these policies gave rise to a novel currency phenomenon, the "Lollar", symbolizing the trapped US dollars within Lebanon's banking system.

The "Lollar", a term coined during Lebanon's banking crisis, represents the devaluation of US dollar-denominated bank accounts due to circulars issued by the Lebanese Central Bank (Helou 2022). These circulars implied that accounts in foreign currency were no longer valued at their nominal worth, leading to the emergence of the Lollar as a new currency standard. As inflation surged alongside the black-market exchange rate, the purchasing power of individuals receiving Lebanese pounds declined, exacerbating poverty levels. The formula for calculating the real value of foreign currency accounts in Lollars demonstrated significant losses for depositors upon cashing their funds, with checks worth only a fraction of their nominal amount. The depreciation of the Lollar continued, with the cash value of a 1000 USD check plummeting to just 16% in January 2024, reflecting a substantial haircut on foreign currency accounts (Abdel Samad 2021).

5. Results and Discussion

Understanding the dynamics of Lebanon's informal economy is crucial amidst its socio-political instability and economic challenges. This section presents the results of our literature review and analysis, highlighting key factors contributing to Lebanon's shadow economy and discussing their implications.

5.1. Key Factors Identified

Through a comprehensive analysis of various indicators, several critical factors influencing Lebanon's informal economy have been identified. Table 2 examines the indicators linked to Lebanon's informal economy, shedding light on its implications. Each factor is analyzed along with its underlying reasons and how they align with the institutional economics framework, providing insights into the institutional context shaping informal economic activities in Lebanon.

Causes and Indicators	Description	Reason	Institutional Economics Framework		
Higher unemployment and lower GDP per capita	Political tensions, corruption, and economic crises have led to about 50% GDP contraction from 2019 to 2021, unemployment rising to 29.6%, and inflation soaring to 154.9% (World Bank Group 2021b).	Prolonged instability erodes formal job opportunities, pushing many into informal work.	Weak governance undermines institutional quality, fostering informal economic activity.		
Tax rates in Lebanon	Regressive taxation and inefficient collection methods drive tax evasion, reducing revenue.	High taxes and lax enforcement incentivize evasion, fueling informality.	Inadequate regulation fosters tax evasion and informality, requiring institutional reforms.		
Government and Central Bank regulations	Interventions to stabilize the economy led to market distortions, benefiting the wealthy and exacerbating inequality.	Policy interventions distort markets, encouraging informal economic practices.	Weak regulatory oversight allows for ineffective policies, necessitating stronger institutions.		

Table 2. Summary of causes and indicators related to Lebanon's informal economy.

Causes and Indicators	Description	Reason	Institutional Economics Framework
Financial crisis and smuggling	Economic desperation and corruption foster smuggling activities, resulting in substantial revenue losses.	Economic hardship and porous borders facilitate smuggling and illicit trade.	Weak enforcement and regulatory capture enable illicit activities, demanding institutional reforms.
Amount of cash in circulation	Financial sector deterioration prompts reliance on cash transactions, raising concerns about tax enforcement and illicit financing.	Banking distrust drives cash usage, facilitating tax evasion and illicit activities.	Weak financial oversight allows informal transactions to thrive, highlighting the need for regulatory reforms.
Fiscal balance (% of GDP)	Experienced significant fluctuations, with deficits peaking in 2018 (-11.38%). A brief positive balance in 2021 (+0.64%) (Statista 2024), which indicates drastic government spending cuts rather than healthy fiscal correction.	Economic instability, policy mismanagement, and lack of effective fiscal reforms.	Reflects the impact of government policy choices and institutional capacity on fiscal stability. Poor institutional quality contributes to erratic fiscal management and undermines economic stability.
	spending cuts rather than healthy fiscal correction.		management and undermine economic stability.

Table 2. Cont.

(Source: Authors).

In the labyrinth of Lebanon's economic landscape, several intertwined factors perpetuate the shadow economy, leaving a trail of consequences. Among these, poor institutional quality acts as a catalyst, fostering an environment ripe for tax erosion and corruption (Svensson 2003). These elements, in turn, fuel the shadow economy, characterized by a web of informal activities, including smuggling. This cycle exacerbates Lebanon's economic woes, contributing to budget deficits and widening income disparities. Similarly, understanding the role of leadership and organizational culture in managing economic uncertainties within organizations, as highlighted in a recent study by Oliveira et al. (2022), underscores the importance of strategic management practices in mitigating systemic risks and fostering organizational resilience, thereby influencing broader economic outcomes. Importantly, these issues are not solely macroeconomic in nature but also manifest at the organizational and institutional levels. Critically analyzing Lebanon's case reveals the intricate relationship between institutional weaknesses and the proliferation of informal economic practices, highlighting the urgent need for structural reforms within the framework of institutional economics.

Our analysis revealed that the dynamics of Lebanon's shadow economy are intrinsically tied to the country's institutional framework.

5.2. Implications of Institutional Factors

Our study also unraveled the interconnectedness of these institutional factors and their adverse impact on Lebanon's economy (refer to Figure 7). These interwoven challenges further amplify economic and financial risks and issues in Lebanon, including high unemployment rates, declining GDP, and currency instability.

Our study reveals the intricate connections among institutional factors and their detrimental impact on Lebanon's economy. These challenges exacerbate economic woes, including high unemployment, GDP decline, and currency instability. Our findings emphasize the urgency of addressing these issues due to their far-reaching consequences on economic stability, government revenues, and societal well-being.



Figure 7. Main independent variables analyzed related to shadow economy and its impact in Lebanon (source: authors).

5.3. Addressing the Challenges

The current policies in Lebanon have proven ineffective in promoting sustainable development. To address these challenges, we recommend the following strategies, as mentioned in Figure 8:



Figure 8. Recommendations to achieve long-term economic sustainability (source: authors).

5.4. Theoretical and Practical Implications

This research extends beyond theoretical insights, providing practical implications that can guide policymaking and real-world solutions to the challenges posed by the shadow economy in Lebanon.

This research enhances our theoretical understanding by demonstrating how institutional weaknesses in Lebanon exacerbate the shadow economy, reinforcing theories that link poor governance to informal economic activities. The study also broadens the theoretical framework by integrating macroenvironmental factors, such as servitization and market conditions, to explain how external elements influence the shadow economy. The influence of servitization on firms' foreign market entry mode decisions highlights the importance of macroenvironmental factors such as market attractiveness and institutional environment (Agnihotri et al. 2022), which are also critical in understanding and addressing the dynamics of the shadow economy. Additionally, the findings validate existing theories on corruption and tax evasion while suggesting refinements specific to Lebanon's institutional failures.

Practically, the study provides actionable recommendations for policy improvements, including enhanced regulatory frameworks, better tax collection mechanisms, and compre-

hensive institutional reforms. It also highlights the need for strategic management practices within organizations to mitigate risks associated with the informal sector. These insights offer valuable guidance for policymakers and can be applied to other similar economies, facilitating informed decision-making and comparative analysis.

6. Conclusions, Implications and Future Research Avenues

Our study sheds light on the complex dynamics of Lebanon's shadow economy amidst pervasive socio-political instability and economic challenges. Table 2 meticulously examines key indicators and underlying causes associated with Lebanon's informal economy, providing a nuanced understanding of its implications within the institutional economics framework.

The findings underscore the critical need for integrated strategies to combat the shadow economy, emphasizing the role of institutional quality, corruption dynamics, and regulatory frameworks in shaping informal economic activities. These insights not only deepen our comprehension of Lebanon's economic landscape but also contribute theoretically by establishing a robust framework for future research in similar contexts globally. By integrating concepts of risk management and financial resilience, our study lays a foundation for developing effective strategies to mitigate the risks associated with informal economic practices.

6.1. Policy Implications

Given the significant risks posed to economic development by the shadow economy, our study emphasizes the urgent need for evidence-based policies that strengthen regulatory frameworks, combat corruption, and promote formalization. Implementing these measures is crucial for enhancing economic stability and fostering sustainable development.

To effectively tackle the shadow economy, policies should encompass several critical areas. Enhancing regulatory frameworks is essential, which involves developing and enforcing robust regulations to combat tax evasion, smuggling, and other informal activities, along with improving transparency in financial transactions and strengthening anti-corruption measures. Fighting corruption requires the establishment of independent bodies with the authority to investigate and prosecute corrupt practices and increasing transparency in public procurement and government spending. Encouraging formalization can be supported by several decisions like simplifying business registration processes, reducing bureaucratic barriers, and providing incentives such as tax breaks or subsidies for businesses transitioning to the formal sector. Strengthening financial sector oversight involves bolstering the capabilities of regulatory authorities to manage financial risks associated with informal activities. Fostering public-private partnerships is crucial for developing effective strategies to address the shadow economy, with industry leaders contributing to best practices for compliance and transparency. Additionally, investing in economic diversification through technology and sustainable industries can create new formal job opportunities and reduce dependence on informal sectors. Importantly, governance reforms are vital, as highlighted by Abou Ltaif and Mihai-Yiannaki (2024), which emphasize addressing political patronage and strengthening institutional frameworks to foster long-term economic stability and development in Lebanon.

6.2. Executive Implications

Managers and industry leaders can derive practical insights from our findings by enhancing transparency and compliance within their organizations. Mitigating risks linked to informal economic activities necessitates rigorous risk management frameworks and adherence to ethical business practices. Embracing sustainable finance practices can further enhance organizational resilience and stakeholder trust amid economic uncertainties.

6.3. Future Research

Future research should build upon our macroeconomic analysis by investigating additional factors that influence the shadow economy. An initial focus should be on exploring macroeconomic elements such as foreign direct investment (FDI) flows, national income disparities, regulatory environments, and economic stability, which play a significant role in shaping informal economic activities. Following this, integrating microeconomic factors will provide valuable insights into how corporate finance strategies, capital structure optimization, and sustainable finance practices impact and interact with the shadow economy. Understanding these microeconomic dimensions can reveal how company-level decisions and practices contribute to the persistence of the shadow economy, thereby informing more targeted and effective policies, reforms, and economic decisions. This comprehensive approach will enhance our understanding of the shadow economy's dynamics and support the development of strategies to mitigate its adverse effects on economic stability and growth.

6.4. Concluding Remarks

In conclusion, our study not only provides valuable insights into Lebanon's shadow economy but also offers practical implications and theoretical frameworks to inform policy interventions and academic inquiry globally. By bridging theory with actionable recommendations, we aim to catalyze efforts towards building more resilient, inclusive, and sustainable economies worldwide. This collaborative endeavor requires concerted efforts from academia, policymakers, and industry stakeholders to translate insights into impactful actions that promote economic stability, effective risk management, and equitable development.

Author Contributions: Conceptualization, S.F.A.L.; methodology, S.F.A.L.; software, S.F.A.L.; validation, S.F.A.L., S.M.-Y. and A.T.; formal analysis, S.F.A.L.; investigation, S.F.A.L.; resources, S.F.A.L.; data curation, S.F.A.L.; writing—original draft preparation, S.F.A.L.; writing—review and editing, S.F.A.L., S.M.-Y. and A.T.; visualization, S.F.A.L., S.M.-Y. and A.T.; project administration, S.F.A.L. All authors have read and agreed to the published version of the manuscript.

Funding: This research was conducted with no external funding or support.

Institutional Review Board Statement: This paper is based on original research conducted by the author in accordance with ethical principles and standards set by the journal. The author has no conflicts of interest to declare, and the accuracy of the data and unbiased analysis and interpretation are affirmed.

Data Availability Statement: All data analyzed during this study are included in the article.

Conflicts of Interest: The authors report there are no competing interests to declare.

References

- Abdel Samad, Maher. 2021. Understanding the 2019–2020 Banking and Financial Crisis in Lebanon. Undergraduate thesis, Wharton School, University of Pennsylvania, Philadelphia, PA, USA. Available online: https://repository.upenn.edu/handle/20.500.14332 /49263 (accessed on 9 December 2021).
- Abou Ltaif, Samar, and Simona Mihai-Yiannaki. 2024. Exploring the Impact of Political Patronage Networks on Financial Stability: Lebanon's 2019 Economic Crisis. *Economies* 12: 183. [CrossRef]
- Ackerman, Susan-Rose. 1997. Corruption, inefficiency and economic growth. Nordic Journal of Political Economy 24: 3–20.
- Agnihotri, Arpita, Saurabh Bhattacharya, Natalia Yannopoulou, and Alkis Thrassou. 2022. Foreign market entry modes for servitization under diverse macroenvironmental conditions: Taxonomy and propositions. *International Marketing Review* 40: 561–84. [CrossRef]
- Azar, Samile Antoine, and Khaled Abdallah. 2019. What drives the accretion of the foreign exchange reserves of the Lebanese Central Bank? (1994–2018). *Theoretical Economics Letters* 9: 633–48. [CrossRef]
- Azzi, Aline. 2022. Tax System in Lebanon: Insufficient, Un-Equalizing, and Leaky! Blominvest Bank. Available online: https: //blog.blominvestbank.com/43044/tax-system-in-lebanon-insufficient-un-equalizing-and-leaky/ (accessed on 12 April 2023).
- Baff, Sam. 2022. CAS-ILO: Unemployment Is at 29.6% and Youth Unemployment at 47.8% in Lebanon by January 2022. Blominvest Bank. Available online: https://blog.blominvestbank.com/44347/cas-ilo-unemployment-is-at-29-6-and-youth-unemploymentat-47-8-in-lebanon/ (accessed on 1 April 2023).

- Banque du Liban (BDL). 2023a. Banque du Liban. Available online: https://www.bdl.gov.lb/economicandfinancialdatasub.php? docId=17&code=1&filecode=100 (accessed on 13 June 2023).
- Banque du Liban (BDL). 2023b. Banque du Liban. Available online: https://www.bdl.gov.lb/statisticsandresearch.php (accessed on 19 June 2023).
- Banque Du Liban (BDL). 2024. Available online: https://www.bdl.gov.lb/intermediatecirculars.php?tpages=46&page=2&langid=EN (accessed on 22 June 2024).
- Berdiev, Aziz N., Rajeev K. Goel, and James W. Saunoris. 2018. Corruption and the shadow economy: One-way or two-way street? *The World Economy* 41: 3221–41. [CrossRef]
- Bifani, Alain, Karim Daher, Lydia Assouad, and Ishac Diwan. 2021. Which tax policies for Lebanon? *Arab Reform Initiative*. May 28. Available online: https://www.arab-reform.net/publication/which-tax-policies-for-lebanon-lessons-from-the-past-for-a-challenging-future/ (accessed on 22 April 2023).
- Blanton, Robert G., and Dursun Peksen. 2021. A global analysis of financial crises and the growth of informal economic activity. *Social Science Quarterly* 102: 1947–65. [CrossRef]
- Braun, Virginia, and Victoria Clarke. 2006. Using thematic analysis in psychology. *Qualitative Research in Psychology* 3: 77–101. [CrossRef]
- Canh, Phuc Nguyen, Christophe Schinckus, and Su Dinh Thanh. 2021. What Are the Drivers of Shadow Economy? A Further Evidence of Economic Integration and Institutional Quality. *The Journal of International Trade & Economic Development* 30: 47–67.
- Chletsos, Michael, and Andreas Sintos. 2024. Political Stability and Financial Development: An Empirical Investigation. *The Quarterly Review of Economics and Finance* 94: 252–66. [CrossRef]
- Choi, Jay Pil, and Marcel Thum. 2005. Corruption and the shadow economy. International Economic Review 46: 817–36. [CrossRef]
- Cozma, Adeline-Cristina, Corina-Narcisa (Bodescu) Cotoc, Viorela Ligia Vaidean, and Monica Violeta Achim. 2021. Corruption, Shadow Economy and Deforestation: Friends or Strangers? *Risks* 9: 153. [CrossRef]
- Daher, Joseph. 2022. Lebanon: How the Post War's Political Economy Led to the Current Economic and Social Crisis. Available online: https://cadmus.eui.eu/bitstream/handle/1814/73856/QM-01-22-031-EN-N.pdf?sequence=1&isAllowed=y (accessed on 24 December 2023).
- Dreher, Axel, and Friedrich Schneider. 2010. Corruption and the Shadow Economy: An Empirical Analysis. *Public Choice* 144: 215–38. [CrossRef]
- Dreher, Axel, Christos Kotsogiannis, and Steve McCorriston. 2009. How do institutions affect corruption and the shadow economy? *International Tax and Public Finance* 16: 773–96. [CrossRef]
- European Civil Protection and Humanitarian Aid Operations. 2023. Lebanon. European Commission. Available online: https://civilprotection-humanitarian-aid.ec.europa.eu/where/middle-east-and-northern-africa/lebanon_en (accessed on 17 April 2023).
- Friedman, Barry A. 2014. The relationship between effective governance and the informal economy. *International Journal of Business and Social Science* 5: 51–58.
- Gebeily, Maya. 2023. Cash is king in Lebanon as banks atrophy. *Reuters*. January 31. Available online: https://www.reuters.com/ markets/cash-is-king-lebanon-banks-atrophy-2023-01-31/ (accessed on 12 April 2023).
- Gemayel, Fouad. 2023. What the trade balance reveals about Lebanon in 2022. *L'Orient Today*. February 1. Available online: https://today.lorientlejour.com/article/1326676/what-the-trade-balance-reveals-about-lebanon-in-2022.html (accessed on 13 June 2023).
- Helou, Joseph P. 2022. State collusion or erosion during a sovereign debt crisis: Market dynamics spawn informal practices in Lebanon. In Informality, Labour Mobility and Precariousness: Supplementing the State for the Invisible and the Vulnerable. Edited by Abel Polese. Cham: Springer International Publishing, pp. 277–300. [CrossRef]
- Hodgson, Geoffrey M. 1998. The Approach of Institutional Economics. Journal of Economic Literature 36: 166–92.
- International Monetary Fund (IMF). 2019. Lebanon IMF Executive Board Concludes 2019 Article IV Consultation. *IMF*. October 17. Available online: https://www.imf.org/en/News/Articles/2019/10/17/pr19378-lebanon-imf-executive-board-concludes-20 19-article-iv-consultation-with-lebanon (accessed on 24 January 2022).
- International Monetary Fund (IMF). 2023. General Government Gross Debt. IMF.org. Available online: https://www.imf.org/external/datamapper/GGXWDG_NGDP@WEO (accessed on 5 January 2024).
- Kanniainen, Vesa, Jenni Pääkkönen, and Friedrich Schneider. 2004. Determinants of Shadow Economy: Theory and Evidence, Discussion Paper. Available online: https://www.academia.edu/21383082/Determinants_of_shadow_economy_theory_and_ evidence (accessed on 25 May 2023).
- Kareh, Marie D. 2020. The reform of the tax system in Lebanon: An impossible equation? In *Economics and Finance*. Paris: Université Panthéon-Sorbonne—Paris I.
- Kraemer-Mbula, Erika, and Sascha Wunsch-Vincent, eds. 2016. Introduction. In *The Informal Economy in Developing Nations: Hidden Engine of Innovation*? Cambridge: Cambridge University Press, pp. 1–12. [CrossRef]
- LaPorta, Rafael, Florencio Lopez-de-Silanes, Andrei Shleifer, and Robert Vishny. 1999. The Quality of Government. *Journal of Law, Economics and Organization* 15: 222–79. [CrossRef]
- LBCI Lebanon. 2022. Smuggling from Lebanon to Syria Puts Country under Additional Stress. *LBCIV7*. December 28. Available online: https://www.lbcgroup.tv/news/press/679889/lbci-lebanon-articles/en (accessed on 14 April 2023).

- Lirarate. 2023. Lira Rate | USD to LBP in Black Market | Dollar to LBP. *Lira Rate*. Available online: https://lirarate.org/ (accessed on 2 December 2021).
- McMaster, R. 2012. Institutional economics: Traditional. In *International Encyclopedia of Housing and Home*. Edited by Susan J. Smith. San Diego: Elsevier, pp. 86–92. [CrossRef]
- Medina, Leandro, and Friedrich Schneider. 2018. *Shadow Economies around the World: What Did We Learn over the Last 20 Years?* IMF Working Papers. Washington, DC: International Monetary Fund, vol. 18. [CrossRef]
- Mishchuk, Halyna, Svitlana Bilan, Halyna Yurchyk, Liudmyla Akimova, and Mykolas Navickas. 2020. Impact of the shadow economy on social safety: The experience of Ukraine. *Economics & Sociology* 13: 289–303. [CrossRef]
- Murphy, Kevin M., Andrei Shleifer, and Rober W. Vishny. 1993. Why is rent-seeking so costly to growth? *American Economic Review Papers and Proceedings* 83: 409–14.
- Nasrallah, Tala. 2020. BDL Issues Circulars Related to Provisioning Banks and Financial Institutions Subsidized Trade Finance Transactions and to Withdrawals of USD Deposits. *Blominvest Bank*. October 15. Available online: https://blog.blominvestbank.com/38340/bdl-issues-circulars-related-to-provisioning-banks-and-financial-institutions-for-specified-trade-transactions-and-to-withdrawals-of-usd-deposits/ (accessed on 19 December 2021).
- Nguyen, Giang Ngo Tinh, and Xianmin Liu. 2023. The interrelationship between corruption and the shadow economy: A perspective on FDI and institutional quality. *Journal of Economics and Development* 25: 349–64. [CrossRef]
- Oliveira, Margarida Freitas, Eularia Santos, and Vanessa Ratten. 2022. Strategic perspective of error management, the role of leadership, and an error management culture: A mediation model. *Journal of Economics, Finance and Administrative Science* 28: 160–75. [CrossRef]
- Phuc Canh, Nguyen. 2018. The effectiveness of fiscal policy: Contributions from institutions and external debts. *Journal of Asian Business and Economic Studies* 25: 50–66. [CrossRef]
- Pluskota, Anna. 2020. The impact of corruption on economic growth and innovation in an economy in developed European countries. Annales Universitatis Mariae Curie-Skłodowska Sectio H Oeconomia 54: 77–87. [CrossRef]
- Rose-Ackerman, Suzan. 1999. Corruption and Government: Causes, Consequences, and Reform. Cambridge: Cambridge University Press.
- Santos, Eularia, and Fernando Oliveira Tavares. 2020. The level of knowledge of financial literacy and risk of the Portuguese. *International Journal of Euro-Mediterranean Studies* 13: 91–116.
- Saunders, Mark, Philip Lewis, and Adrian Thornhill. 2007. *Research Methods for Business Students*. London: Financial Times/Prentice Hall.
- Schneider, Frederich. 2000. The Increase of the Size of the Shadow Economy of 18 OECD Countries: Some Preliminary Explanations. CESifo Working Paper Series No. 306. Munich: CESifo.
- Schneider, Friedrich. 2005. Shadow economies around the world: What do we really know? *European Journal of Political Economy* 21: 598–642. [CrossRef]
- Schneider, Friedrich, and Andreas Buehn. 2016. Estimating the Size of the Shadow Economy: Methods, Problems, and Open Questions. IZA Discussion Paper 9820. Available online: https://www.iza.org/publications/dp/9820/estimating-the-size-of-the-shadow-economy-methods-problems-and-open-questions (accessed on 15 March 2023).
- Schneider, Friedrich, and Dominik Enste. 2000. Shadow economies: Size, causes, and consequences. *Journal of Economic Literature* 38: 77–114. [CrossRef]
- Sharipov, Ilkhom. 2015. Contemporary Economic Growth Models and Theories: A Literature Review. CES Working Papers. Available online: https://www.proquest.com/openview/481724c8292a156a08131a686552e021/1?pq-origsite=gscholar&cbl= 2035671 (accessed on 28 March 2023).
- Statista. 2024. Lebanon—Budget Balance in Relation to GDP 2012–2022. *Statista*. Available online: https://www.statista.com/statistics/ 455256/lebanon-budget-balance-in-relation-to-gdp/ (accessed on 2 June 2024).
- Svensson, Jakob. 2003. Who must pay bribes and how much? Evidence from a cross section of firms. *The Quarterly Journal of Economics* 118: 207–30. [CrossRef]
- Tanzi, Vito. 1999. Uses and abuses of estimates of the underground economy. The Economic Journal 109: 338-47. [CrossRef]
- Tanzi, Vito. 2002. *The Shadow Economy, Its Causes, and Its Consequences*. Lecture Given at the International Seminar on the Shadow Economy Index in Brazil. Rio de Janeiro: Brazilian Institute of Ethics in Competition.
- Tavares, Fernando, Eulália Santos, Margarida Freitas Oliveira, and Luís Almeida. 2024. Determinants of Corporate Indebtedness in Portugal: An Analysis of Financial Behaviour Clusters. *Risks* 12: 91. [CrossRef]
- Torgler, Benno, and Friedrich Schneider. 2009. The impact of tax morale and institutional quality on the shadow economy. *Journal of Economic Psychology* 30: 228–45. [CrossRef]
- Trading Economics. 2023. Lebanon Government Budget. Available online: https://tradingeconomics.com/lebanon/governmentbudget (accessed on 1 April 2024).
- Transparency International. 2022. Corruption Perceptions Index 2021. February 3. Available online: https://www.transparency.org/en/cpi/2021 (accessed on 19 April 2023).
- World Bank. 2023a. Lebanon: Normalization of Crisis Is No Road to Stabilization. World Bank. Available online: https://www. worldbank.org/en/news/press-release/2023/05/16/lebanon-normalization-of-crisis-is-no-road-to-stabilization (accessed on 13 February 2024).
- World Bank. 2023b. World Bank Open Data. Available online: https://data.worldbank.org (accessed on 22 June 2024).

- World Bank Group. 2021a. GDP Growth (Annual %)—Lebanon. Available online: https://data.worldbank.org/indicator/NY.GDP. MKTP.KD.ZG?end=2020&locations=LB&start=2005 (accessed on 15 November 2021).
- World Bank Group. 2021b. Lebanon Economic Monitor: Lebanon Sinking to the Top-3. Available online: https://www.worldbank. org/en/country/lebanon/publication/lebanon-economic-monitor-spring-2021-lebanon-sinking-to-the-top-3 (accessed on 29 July 2024).

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.



Article Government Borrowing and South African Banks' Capital Structure: A System GMM Approach

Ndonwabile Zimasa Mabandla * and Godfrey Marozva

Department of Finance, Risk Management, and Banking, 1 Preller Street Mucklenuek, Pretoria 0002, South Africa; marozg@unisa.ac.za

* Correspondence: mabannz@unisa.ac.za

Abstract: This paper aimed to investigate the effects of government borrowing banks' capital structure using a sample of banks registered in South Africa from 2012 to 2021. Despite the extensive literature on this association, few prominent researchers have studied this phenomenon in the banking sector. Applying the generalised method of moments (GMM) model, the study established a positive but significant effect on the South African banks' capital structure from total government borrowing, local government borrowing and foreign government borrowing, and capital structure. Contrary to the crowding-out effects detected, the results revealed a positive and significant relationship between government borrowing and banks' capital structure. The crowding-in effect better explains these results, where government borrowing stimulates the local market for goods and services, motivating banks to borrow more in order to meet the demand for loans. Future research should test the cointegrating and causality relationship between government borrowing and bank capital structure. Also, given that the banking sector is constrained by Basel III's capital adequacy requirement, controlling for this factor is critical in future research.

Keywords: capital structure; government borrowing; crowding-in effect theory



Citation: Mabandla, Ndonwabile Zimasa, and Godfrey Marozva. 2024. Government Borrowing and South African Banks' Capital Structure: A System GMM Approach. *Risks* 12: 112. https://doi.org/10.3390/ risks12070112

Academic Editors: Eulália Mota Santos and Margarida Freitas Oliveira

Received: 12 June 2024 Revised: 4 July 2024 Accepted: 11 July 2024 Published: 16 July 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

1. Introduction

The literature has examined several theoretical and empirical factors that determine capital structure. These factors that determine capital structure include risk, size, profitability, the tangibility of assets, growth potential, and debt versus non-debt tax shielding. The two main underlying theories to explain these determinants include the trade-off theory (Kraus and Litzenberger 1973) and the pecking order theory (Myers and Majluf 1984). According to the trade-off hypothesis, firms balance the costs and benefits of debt and adjust toward an optimal debt ratio (Kraus and Litzenberger 1973). In the pecking order theory, it is argued that the information asymmetry amongst managers and shareholders affects the cost of financing (Myers and Majluf 1984). However, government borrowing is a new factor deemed to influence company leverage (Akkoyun 2018; Demirci et al. 2019; Gao et al. 2022).

However, government borrowing (GB) can play an essential part in dealing with the economic crisis in the short term. Yet, it may crowd out private investment in the long run (Ayturk 2017). An escalation in government debt supply might decrease investors' demand for a firm's debt compared to equity. Meanwhile, government debt is a better substitute for a firm's debt than equity (Demirci et al. 2019). Hence, firms might regulate the capital structure and decrease their leverage. Friedman (1978) argued that if investors have an inadequate supply of funds and have favourites for the precise kinds of securities, for example, long-term versus short-term, government debt could affect a firm's financing.

Prior research on global capital structure examines the association between GB and capital structure; however, the empirical studies have contradictory findings. For instance, Bahal et al. (2018) explored whether public investment remained excessive or insufficient

before 1980. Utilising investment project data from the CapEx-CMIE, they also created a new data set of average public and private investment in India from 1996 to 2015. Their findings suggest that public investment may have crowded out commercial investment in India from 1950 to 2012. Yet, from 1980 to 2012, their results supported crowding in capital funding. Furthermore, their analysis revealed that crowding is confirmed by their quarterly model, which uses investment project datasets from CapEx-CMIE from 1996 to 2015.

Similarly, Ayturk (2017) looked into the connection between GB and corporate finance choices in 15 major European nations from 1989 to 2014. Using a country-level aggregate and a fixed-effects panel data model with aggregate flow data, they found a robust negative relationship between GB and company debt in advanced European countries. Furthermore, research revealed that the long-term debt of major credit-worthy enterprises is more complex than government debt in comparison to small, economically hampered firms. This suggests that long-term corporate debt issued by creditworthy companies is a better substitute for government bonds (Ayturk 2017).

Graham et al. (2014) studied the effect of government debt on commercial financing and investment in the United States. Their study discovers an insignificant link between government debt and company equity regulations. Nonetheless, they find a substantial positive link between government debt, corporate cash, and other short-term cash reserves, such as treasuries. Ayturk (2017) explored only a simple connection between GB and corporate funding. Yet, Graham et al. (2014) evaluated the impact of government debt on commercial financing and investments. What is not clear at the present moment is whether the transmission mechanism occurs when GB reduces the money supply, raises interest rates, and finally affects capital structure. Therefore, the study aims to determine the effect of GB on the capital structure of South Africa from 2012 to 2021.

In this paper, total government borrowing (TGB), local government borrowing (LGB), and foreign government borrowing (FGB) were utilised to measure GB. As a result, Figure 1 examines the trends in South Africa's average government borrowings from 2012 to 2021.



Figure 1. Source: author's own compilation based on the data extracted from SARB.

TGB rose from R1,334,110 in 2012 to R4,230,630 in 2021, based on Figure 1. FGB has increased from R1,202,942 in 2012 to R2,343,843 in 2021. Similarly, borrowing by local governments rose from R1,311,680 in 2012 to R1,827,764 in 2008. Previous research on global capital structure investigates the link between GB and capital structure (Gao et al. 2022; Demirci et al. 2019; Akkoyun 2018; Ayturk 2017; Graham et al. 2014). However, they found contradictory results. Friedman (1978) argued that it is vital to appreciate one probable impact of GB on corporate financing, denoted as financial crowding-out. In contrast, public capital is possibly utilised by increasing productive ability, a crowding effect. Yet, public capital typically complements private capital in the production and distribution of private output; a crowding-in effect. Although its impact has been studied, no consensus has been

reached on its reality, primarily because of the empirical identification problems. Hence, this study investigates the effect of GB on bank capital structure in South Africa from 2012 to 2021.

The research hypothesis in this paper is explained as follows: H1: Does government borrowing influence banks' capital structure?

FGB constitutes 80 percent of the GDP, while local government constitutes 20 percent of the GDP. Eighty percent of FGB implies that the government borrows 80 percent from foreign borrowings, and as FGB does not crowd out a local market, this condition may result in a crowding-in effect. Yet, 20 percent of the LCB implies that the government borrows 20 percent from domestic companies. This study contributes to the existing empirical literature on the relationship between GB and the capital structure of banks, using data drawn from registered banks in South Africa, as the capital structure of banks differs from that of non-financial firms. Previous studies focused on the determinants of capital structure have excluded financial firms (see, for example, scholars such as Jaworski and Czerwonka 2021; Kuč and Kaličanin 2021; Gharaibeh and Sager 2020; Shahzad et al. 2021). The reason for excluding financial enterprises from capital structure research has been either because they are governed firms or because they have inherent key variables, like having charges or deposits as additional sources of funding (Sibindi 2018). However, few studies have focused on financial firms in South Africa (Sibindi 2018; Sibindi and Makina 2018). While the effect of GB has been studied in developed countries, little is known in developing countries about this issue, particularly in South Africa. Hence, the current paper differs from the papers mentioned above, since this paper focused on the effect of GB on South African banks' capital structure. Bank leverage is strongly regulated through the Basel III regulations, explicitly focusing on the capital adequacy ratio. The rest of this paper is structured as follows: the next section presents a review of the existing literature on the effect of GB on firms' capital structure. This is followed by the methodology that describes our econometric approach in depth. The findings are presented and discussed, and the discussion ends with a conclusion and recommendations.

2. Literature Review

In the literature, extensive research has been carried out to understand what factor influences the capital structure of companies, both technically as well as empirically. Modigliani and Miller (1958) (M&M) pioneered current work on capital structure theory. They demonstrate that the capital structure of a corporation determines its worth. Furthermore, M&M (Modigliani and Miller 1963) presented a revision to their prior study, stating that the firm's worth is independent of its capital structure; however, a gap is created by finance costs on debt. They explain that the discrepancy is attributable to interest payments being taxable income in various nations according to income tax laws. As a result, other academics produced capital structure theories, such as trade-off, pecking order, signalling, market, agency cost, and free cash flow theories.

In their research that establishes the theoretical basis of the static trade-off theory in the future, Kraus and Litzenberger (1973) showed that optimal leverage implies a trade-off between the tax benefits of debt and the burden costs of bankruptcy. Myers (1984) refined it in his static trade-off model, proposing that corporations establish a goal debt-to-value ratio and begin moving towards it, like how companies alter dividends to move towards a target payout ratio. A company's funding options are essentially two: debt and equity. Sibindi and Makina (2018) argued that in a trade-off framework, the company would take on additional debt to take advantage of the debt interest tax shield until the accruals exceed the present value of the interest tax shield. Myers and Majluf (1984), who developed the pecking order theory of capital structure, contend that giving safe assets over risky assets is generally preferable. Myers (1984) supports this argument by claiming that the corporation will indeed internally choose external financing and debt to equity if it can provide the security. As a result of the hypothesis, debt will climb for corporations when investment

exceeds internal financing, whereas indebtedness will fall once the asset becomes less valuable than carefully chosen finances.

The notion of the pecking order is also known as the asymmetric information theory. Information asymmetry emerges because management (internal stakeholders) has more information than investors (outsiders), whom they utilise to their benefit (Sibindi and Makina 2018). The signalling hypothesis stems from asymmetry ideas, which may be traced back to Ross' work (Ross 1977). According to the author, if managers have access to internal information, they can choose between a managerial incentive system and a financial structure that sends data to the market and confirms the significance of the signals in a team balance (Ross 1977). Although market time theory is a subset of the information theory, it has contributed to the creation of the signalling theory. Market time theory, likewise, ignores the concept of optimal capital structure, which has departed from the work of Baker and Wurgler (2002). They contend that managers time their stock offerings with the expectation of issuing fresh shares when the share price is recognised to be higher. They will, in contrast, acquire shares when they are believed to be lower.

However, the crowding-out effect idea can be explained using the neoclassical and Keynesian approaches. The neoclassical approach favours full employment and free markets over government action. According to the neoclassical loanable fund hypothesis, the interest rate system will respond to the equilibrium of savings. For instance, a rise in government spending raises interest rates, pushing the capital market back into balance and crowding out private investment (Arrow and Kruz 1970). This view was supported by David and Scadding (1974), who argued that an increase in government bond issuance crowded out an equivalent percentage of private capital, because deficit finance is considered public capital and the latter serves for private capital investment.

Yet, Keynes (1936) established the impetus for the idea that government expenditure does not crowd out private spending. It is assumed that government expenditure enhances private investment due to the positive impact of government spending on share prices; thus, crowding occurs. Carlson and Spencer (1975) stated that while the Keynesians constructed multiple models, they were never tested as interrelated components. Friedman (1978) proposes a portfolio crowding-out strategy based on the substitutability of public debt and many other assets in shareholders' portfolios, which could result in either the crowding-out or crowding-in of capital funding. This researcher sharply criticises the widespread mistaken notion that the negative effect of the wealth portfolio effect on private investment is the only way to solve the model.

The effect of GB on the firm capital structure has been studied in developed countries by Graham et al. (2014), Ayturk (2017), Akkoyun (2018), and Demirci et al. (2019). However, nothing has been done in developing countries regarding this issue, particularly in South Africa. Graham et al. (2014) investigated the impact of government debt on corporate financing and investment in the United States. Based on the findings, federal government debt issuance in the United States has a considerable impact on corporate financial policy. Furthermore, the study's findings demonstrate a high negative association between corporate debt, investment, and government debt, but a strong positive connection with firm liquidity. They contended that these relationships are clearer in larger- and lower-risk corporations and that debt is a faster alternative for treasuries.

Ayturk (2017) employed a country-level aggregate and fixed-effects panel data model, using aggregate flow data to analyse data in a study on the association between government borrowing and corporate financing decisions in 15 developed European nations from 1989 to 2014. The study discovers a substantial inverse link between government borrowing and company debt, yet no significant relationship between government indebtedness and equity. The study's findings contradict Graham et al. (2014), who found a strong negative link between company debt, investment, and government debt, yet a substantial positive connection with company liquidity. Contrary to those mentioned above, this study focused on the effect of government borrowing on a bank's capital structure.

Akkoyun (2018) examined a study on the impact of government debt on firm financing in the United States of America (USA) during World War 1. The study focuses on the war period because the status of the economy, the manner of corporate security of aid, and the alternative financing strategy of the United States government provide a suitable empirical context to identify the effect. Yet, our research focuses on how government debt impacts the capital structure of South African registered banks. The analysis found that long-term government bond issues had a negative impact on long-term company bonds with a life expectancy of more than five years, common stocks, and preferred stocks.

To investigate the impact of government debt on company capital structure selection, Demirci et al. (2019) used data from 40 nations from 1990 to 2014 and measured government debt using domestic government debt-to-GDPt-1 and external government debt-to-GDPt-1. After adjusting for the nation and year-fixed effects and country-level controls, the researchers discovered a negative effect on government borrowing and company leverage in both levels and variations of debt. They also found that the crowding-out effect is more substantial for superior and higher-returning enterprises, and companies in markets with more developed equity markets or lower bank dependence.

Utilising a sample of state-owned enterprises (SOEs) and non-SOEs, Liang et al. (2017) investigated the impact of local government debt on firm leverage in China from 2005 to 2007 and from 2011 to 2013. As a result, they eliminated 2008 to 2010 due to a lack of data. They tested the effects of crowding out and crowding in the impact of local government debt on company leverage using a benchmark econometric model. According to the findings of their analysis, China's domestic government debt has a considerable crowding-out impact on non-SOE's force, while being crowded in that of SOEs. Furthermore, the effect differed among industries and sectors.

Other empirical investigations have demonstrated a negative connection between government borrowing and capital structure (for instance, Gao et al. 2022; Cheema 2021; Orangian et al. 2021; Xia et al. 2021). An upsurge in government debt, the availability of surplus public debt, is absorbed by corporate investors seeking to impose a raise of the returns on borrowing, which could additionally end in less disbursement of lending; thus, corporate debt would crowd out public debt (Cheema 2021) because of increased government borrowing and firm leverage, and debt ratios may fall.

Emerging patterns, through a review of previous academic research, indicated that despite extensive research on the effect of GB on firm capital structure, there is still uncertainty and no consensus on which proxies GB should use. This has manifested in some discrepancies in in-country studies on the topic. In comparison with other studies, this article used TGB, FGB, and LGB indicators. Foreign government borrowing and local government borrowing were put into perspective since each type of borrowing has its strategic advantages and disadvantages. Therefore, this article accounts for the several factors that are considered when borrowing, including risk assessment, currency consideration, interest rates, regulations, and investor perceptions. The following section discusses the paper's methodology.

3. Materials and Methods

3.1. Data, Samples, and Variables

The population of this paper comprises all 16 registered domestic banks in South Africa. However, a sample of the paper consists of 11 registered banks in South Africa in the period from 2012 to 2021. These 11 registered banks in South Africa include First Rand Bank Limited, Absa Bank Ltd., Nedbank Ltd., Investec Bank Ltd., Capitec Bank Ltd., Standard Bank of South Africa, Grindrod Bank Ltd., Bidvest Bank Limited, Albaraka Bank Ltd., Sasfin Bank Ltd. and HBZ Bank Ltd. Five small banks, namely African Bank Limited, Deutsche Bank AG, Ubank Llimited, Grobank Limited, and Habib Overseas Bank Ltd., were excluded because of the difficulties in accessing financial data for the period of the study. These firms are considered sufficiently illustrative of the registered banks' population in South Africa from 2012 to 2021. The monthly and annual financial and

economic data were extracted from the iress database, the South African Reserve Bank (SARB) and Bankscope-Bureau van Dijk. These registered banks' sample size consists of 11 banks over a period of 10 years, leading to 110 observations for the banking sample. While the selection of the registered banks in South Africa is identified for this research, it is acknowledged that there were differences in the sampled registered banks' procedures, as mentioned earlier.

This paper used three measures of capital structure, namely total debt ratio (TDR), long-term debt ratio (LTDR), and short-term debt ratio (STDR), as dependent variables following other scholars (Saif-Alyousfi et al. 2020). According to Rajan and Zingales (1995), short-term, long-term, and total debt over total assets ratios are much more adequate indicators of financial leverage than the ratio of liabilities to total assets, since they provide a more precise sense as to whether the company is at the probability of insolvency in the near future and present a much more realistic assessment of previous sources of finance. In contrast, GB was used as an independent variable. Graham et al. (2014) state that GB practices play a significant part in enlightening the continuously improving inclination of companies to use leverage in unregulated sectors. Table 1 below shows the details of the dependent and independent variables and data sources.

Variables	Proxies and Definitions	Proxies by	The Expected Sign of the Coefficient					
Capital structure proxies (Dependent variables)								
Total debt ratio at book value (TDR_B)	TDR_B is defined as the ratio of the book value of total debt to the book value of total assets.	Saif-Alyousfi et al. (2020).						
The long-term debt ratio (LTDR)	LTDR is measured as the ratio of long-term liabilities over total assets	Frank and Goyal (2009), Palacín-Sánchez et al. (2013), Handoo and Sharma (2014), and Saif-Alyousfi et al. (2020).						
The short-term debt ratio (STDR)	STDR is measured as the ratio of short-term debts divided by total assets.	Vo (2017) and Saif-Alyousfi et al. (2020).						
	Independent var	iables						
Total government borrowing (GTB)	Government debt to GDP ratio is defined as the total government debt as a percentage of gross domestic product (GDP) in-country.	Demirci et al. (2019).	Negative					
Local government borrowing (GLB)	The percentage of gross domestic product (GDP) is used to determine local government debt	Ahmad et al. (2020), Demirci et al. (2019) and Liang et al. (2017).	Negative					
Foreign government borrowing	The foreign government debt-to-GDP ratio refers to government debt due to non-residents.	Demirci et al. (2019)	Negative					
	Control variat	les						
Current ratio (CURR)	CURR is measured as the current assets divided by current liabilities	Rao et al. (2017)	Ambiguous					
Economic growth measured by Gross domestic product (GDP).	GDP: The growth rate of real domestic product.	Hanousek and Shamshur (2011); Tesfaye (2012); Jõeveer (2013)	Ambiguous					
Inflation rates	Annual consumer price index (CPI)	Harris and Roark (2019) and Saif-Alyousfi et al. (2020)	Ambiguous					
Interest rates	Effective interest rate	Karpavičius and Yu (2017)	Negative					
Size	Size—the natural logarithm of total assets	Frank and Goyal (2009); Jõeveer (2013) and Bandyopadhyay and Barua (2016).	Ambiguous					

Table 1. Summary of variables and proxies.

Source: Authors' own composition.

3.2. Model Specification

We use the dynamic generalised method of moments (GMM) model. The study used Arellano and Bond (1991) and Arellano and Bover (1995)'s endogeneity robust generalised

method of moments (GMM) to ascertain the impact of government borrowing and bank capital structure. The use of GMM was motivated by its ability to account for the persistence of bank capital structure, simultaneity, and time-invariant omitted variables. Furthermore, GMM considers cross-country variation in the model specification and accounts for small sample biases (Asongu et al. 2018). The model has the following form:

$$y_{it} = \alpha y_{i\ t-1} + \beta x_{it} + \mu_i + \varepsilon_{it} \tag{1}$$

where

 y_{it} represents the book value of the leverage measures for banks *i* in time *t*; x_{it} is the vector of the independent variable for banks and assets managers firms *i* for time *t*, representing the banks and assets managers' firm-specific variable; α_0 denotes a constant term; β is the elasticity of the explanatory variables, i.e., slope of variables; μ_i denotes fixed effects in banks and assets managers firms; ε_{it} is a random error term, the subscript *i* denotes the cross-section and *t* represents the time-series dimension. The panel data regression analysis was then used to analyse the relationship between government borrowing and banks' capital structure.

Our paper was based in South Africa; therefore, a single-country data approach was utilised. The objective to examine the critical determinants of leverage in South African commercial banks was achieved by regressing the leverage-dependent variables (TDRB, STDR, and LTDR) against the determinants in the following equations. Specifically, the relationship between capital structure and the explanatory variables of bank-specific determinants/macroeconomic determinants was expressed quantitatively in Equations (2)–(4) for empirical analysis.

$$TDR_{it} = (\alpha - 1)\Delta TDR_{it-1} + \beta_1 \Delta TGB_{it} + \beta_j \sum_{i=1}^{n} \Delta X_{ij} + \Delta \varepsilon_{it}$$
(2)

$$STDR_{it} = (\alpha - 1)\Delta STDR_{it-1} + \beta_1 \Delta LGB_{it} + \beta_j \sum_{i=1}^{n} \Delta X_{ij} + \Delta \varepsilon_{it}$$
(3)

$$LTDR_{it} = (\alpha - 1)\Delta LTDR_{it-1} + \beta_1 \Delta FGB_{it} + \beta_j \sum_{i=1}^{n} \Delta X_{ij} + \Delta \varepsilon_{it}$$
(4)

where

 TDR_{it} represents total debt ratio at book value for banks *i* in time *t*, measured by the ratio of book value of total debt/book value of total assets, $STDR_{it}$ represents the total debt ratio for banks *i* in time *t*, measured by short-term/ book total assets, $LTDR_{it}$ represents the long-term debt ratio for banks *i* in time *t*, measured by long-term/total assets, TGB_{it} donates government debt to GDP ratio for bank *i* in time *t*, measured by total government debt as a percentage of gross domestic product (GDP) in-country, LGB_{it} is the percentage of gross domestic product (GDP) used to determine local government debt, FGB_{it} is the foreign government debt-to-GDP ratio, referring to government debt due to non-residents, $CURR_{it}$ is the current ratio measured by the current assets over current liabilities, $Size_{it}$ is the size of the *i*th banks on year *t* measured by the natural logarithm of total assets, X_{ij} is a panel of macroeconomic control variable measurements at the end, which includes interest rates and inflation rates. GDP_{it} represents the GDP growth of South Africa in year *t*, ε_{it} is the error term, α is the auto-regression coefficient, and β is a coefficient which represents the slope of variables.

4. Results

Banks may keep up to 26% of their weaknesses as short-term debt. The average total government borrowing (TGB) is 2,537,683 billion, with a standard deviation of 916,717.60. The minimum TGB was 1,334,110,000,000.00, and the maximum TGB was 4,230,630 billion. TGB relative to GDP reached a maximum of 0.78 percent. The situation is good, given that the country has financed roughly 78 percent of its GDP. According to the descriptive data, the average local government borrowing (LGB) throughout the period under review was 506,517 billion. LGB was 20% on average. This means that the government borrows

20% of its financing from domestic companies. Yet, the minimum LBG was 64,746.00, and the maximum LBG was 1,827,764.00. In local banks, the more expensive the LGB, the more the debt cost. Foreign government borrowing (FGB) averaged 2,031,165 billion. The average FGB in relation to GDP was 80%. This means that the government borrows roughly 80% of its financing from international markets. The minimum and maximum FGBs were 1,202,942.00 and 2,602,757 billion, respectively. The higher the FGB, the lower the debt cost in domestic banks. This means that an upsurge in the FGB triggers bank crowding in South Africa. Yet, the minimum current ratio (CR) of 0.01 indicates that the South African bank had the lowest CR of 1% during the investigation period. In Table 2, the maximum CR was 2.56, which means that banks can pay up to 2.56% of their short-term debt. The greater a bank's total CR, the lower its short-term obligations on its current assets.

Variables	Mean	Median	Maximum	Minimum	Std. Dev	Skewness	Kurtosis	Jarque-Bera
TDR	2.19	0.92	145.70	0.56	13.81	10.34	108.00	52,493.33
LTDR	0.48	0.21	33.57	0.00	3.18	10.33	107.83	52,319.88
STDR	0.75	0.53	25.70	0.02	2.40	10.30	107.35	51,851.47
TGB (000,000)	2,537,683.00	2,355,683.00	4,230,630.00	1,334,110.00	916,717.60	0.53	2.11	8.74
LCB (000,000)	506,517.60	283,956.50	1,827,764.00	64,746.00	569,005.10	1.44	3.54	39.11
FGB (000,000)	2,031,165.00	2,038,478.00	2,602,757.00	1,202,942.00	459,239.90	-0.41	1.91	8.60
CR	1.38	1.41	2.56	0.001	0.27	-0.02	12.25	391.81
Size (000)	424,000,000	66,849,693	1,660,000,000	2,997,923	508,000,000	0.77	2.12	14.41
INF	4.60	4.70	5.60	3.10	0.85	-0.42	1.92	8.68
IR	3.89	3.71	5.89	2.31	1.09	0.39	2.27	5.26
GDPG	0.95	1.37	4.90	-6.43	2.77	-1.60	5.58	77.24

 Table 2. Descriptive statistics.

The average bank size (total assets) was 424 billion, with a standard deviation of 508 billion. The bank's size ranged from 2,997,923 billion to 1660 billion. With a standard deviation of 0.85 percent, the inflation rate averaged 4.60 percent. The minimum and maximum INF were 3.10 and 5.60, respectively. A larger scale implies consumer price uncertainty, which is particularly costly to the poor and small businesses as they lack a hedging strategy toward unexpected developments.

Likewise, the mean IR value was 3.89, with a standard deviation of 1.09. The minimum IR was 2.31, and the maximum was 5.89. Finally, the average rate of gross domestic product growth (GDPG) was 0.95 percent, with a standard deviation of 2.77 percent. Despite this, the minimum GDPG was -6.43, and the maximum GDPG was 4.90.

There is skewness and excess kurtosis in all of the variables under consideration. Because the variables are not evenly distributed, Jarque–Bera supports the results obtained on the standard typical evaluation.

As shown in Table 3, the correlation analysis illustrates the correlations between dependent and independent variables used in the banking sector. The total debt ratio (TDR) was found to be positively correlated with the long-term debt ratio (LTDR), whereas the inflation rate (INF) was found to be negatively though strongly linked to the total debt ratio (TDR). Nevertheless, the inflation rate (INF) was found to be significantly negatively associated with the long-term debt ratio (LTDR).

To capture the endogeneity of the lagged dependent variable and independent variables in this analysis, a matrix of internal instruments is built using the GMM estimator (see Arellano and Bond 1991; Arellano and Bover 1995; and Blundell and Bond 1998). The instruments were lag-dependent variables. A diagnostic estimate was used to test for the stationarity of the series and a correlation analysis in accordance with time-series and cross-sectional investigations. In line with Pesaran (2021), cross-section dependence was also examined because there is a chance of endogeneity in the explanatory factors. The

Sargan test was used to confirm the validity of the instruments (Sargan 1958). The results of the overidentifying constraints test agreed with Hansen (1982). Using the Arellano–Bond test (AR1) and (AR2), the correlation of the error terms—which ensures the suitability of the estimated results—was examined (see Arellano and Bond 1991; Roodman 2009). One further benefit of the system GMM is that it does not require the use of external instruments in addition to the variables that are already present.

Table 3. Correlation.

Probability	TDR	LTDR	STDR	CR	LCR	BLMI	SIZE	INF	IR	GDGP
TDR	1.000									
LTDR	0.9996 ***	1.000								
STDR	-0.0181	-0.0218	1.000							
CR	0.00410	0.0604	-0.0350	1.000						
SIZE	-0.00757	-0.0643	-0.0705	-0.0241	0.0739	0.1006	1.000			
INF	-0.1706 *	-0.1697 *	0.1171	0.1697 *	0.0030	0.1213	-0.1572 *	1.000		
IR	-0.0121	-0.0113	-0.0265	-0.0079	-0.0347	0.0324	0.0287	-0.1348	1.000	
GDPG	0.1371	0.1377	0.0220	-0.074	0.0239	-0.0184	0.0646	0.1964 **	0.2971 ***	1.000

* p < 0.05, ** p < 0.01, *** p < 0.001.

Nevertheless, in Table 4 there is a positive but not significant link between IR and LTDR. However, there is a negative but insignificant relationship between IR and TDR and STDR. In addition, TDR and STDR show a negative yet insignificant connection with bank size. In terms of the macroeconomic parameter GDPG, there was a positive and significant effect on GDPG and capital structure. The results support the trade-off theory, which holds that quicker growth in the economy is linked to the greater proclivity of a firm to use debt to fund capital expenditure, since that facilitates greater tax benefits for debt financing (Guizani 2020). In a similar vein, the results support the pecking order theory, which contends that growth in the economy and capital structure are positively related (Guizani 2020).

Table 4. Determinant's capital structure: Effects of TGB and CR.

	2-Step System GMM	2-Step System GMM	2-Step System GMM
Variables	TDR	LTDR	STDR
	1.690 *		
L.IDK	(0.315)		
		1.660 *	
L.LIDK		(0.349)	
			1.701 **
L.SIDK			(0.324)
	16.54 *	5.346	16.34 *
EV	(6.148)	(3.823)	(6.107)
	0.190 *	0.0711	0.196 *
GO	(0.0764)	(0.0758)	(0.0859)
TOD	11.30	18.98	12.49
IGB	(10.70)	(16.00)	(10.86)
	61.63 ***	57.67 ***	60.89 ***
CK	(2.15)	(5.12)	(1.95)

	2-Step System GMM	2-Step System GMM	2-Step System GMM
Variables	TDR	LTDR	STDR
LOITE	-1.648	1.121	-2.358
LSIZE	(7.731)	(9.239)	(7.573)
CDDC	1.783 ***	1.574 *	1.803 ***
GDPG	(0.300)	(0.294)	(0.302)
	-0.940	0.740	-1.120
IK	(0.733)	(0.706)	(0.788)
DIE	2.441 *	4.304 *	2.158 *
INF	(0.911)	(1.698)	(0.843)
	7.620 ***	10.68 ***	6.831 ***
COVID-19	(1.681)	(3.073)	(1.562)
N	88	88	88
Groups	11	11	11
Instrument	9	9	9
AR(1)	-1.29	-1.00	-1.28
AR(2)	-0.87	0.66	-0.92
Sargan Test	0.87	0.77	0.45
Hansen test	0.07	3.21	0.04

Table 4. Cont.

* Driscoll and Kraay robust standard errors in parentheses * p < 0.05, ** p < 0.01, *** p < 0.001.

There was a positive and significant impact on INF with capital structure. These findings contradict Almanaseer's (2019) contention that throughout times of elevated inflation, banks strengthen their regulations to try to mitigate the effect of inflation on interest rates, thus decreasing borrowing. Furthermore, these findings contradict the findings of Bilgin and Dinc (2019), who discovered a negative yet significant connection between INF and capital structure. Finally, COVID-19 had a negative yet significant link with all capital structure. The results are consistent with Mohammad's (2021) discovery of a negative connection between COVID-19 and capital structure. This implies that the global outbreak had a negative effect on capital structure. Due to the ambiguity, banks may have turned to more secure capital.

In Table 5, there is a positive and significant link between TDR, LTDR, and the legged values of TDR and LTDR. Aremu et al. (2013) discovered a positive and insignificant connection between all bank capital structure metrics. However, there is a positive but insignificant link between STDR and its lagged worth.

The interaction between EV and all capital structure measures was positive and significant. The findings back up the pecking order belief that EV and capital structure have a positive nexus. The results, nevertheless, oppose the trading-off theory's assertion of a negative association between EV and capital structure (Assfaw 2020). The findings are in line with the findings of Assfaw (2020), who discovered a positive yet statistically significant connection between earnings volatility and bank capital structure.

Regarding growth opportunity (GO), there existed a positive and statistically significant connection between GO and TDR, LDTR, and SDTR. The results back up the pecking order theory, which holds that there is a positive link between the development of GO and capital structure.

	2-Step System GMM	2-Step System GMM	2-Step System GMM
Variables	TDR	LTDR	STDR
L.TDR	0.704 * (0.313)		
L.LTDR		0.814 * (0.269)	
L.STDR			0.250 (0.214)
EV	15.69 * (5.667)	18.08 * (8.129)	13.54 * (5.889)
	(6.148)	(3.823)	(6.107)
GO	0.273 (0.136)	(0.128) 3.043	(0.207) 2.506
LGB	2.130 (1.774)	3.043 (2.200)	2.506 (1.839)
	(10.70)	(16.00)	(10.86)
CR	54.15 * (18.68)	66.53 * (27.03)	33.83 * (12.23)
LSIZE	-1.251 (3.183)	-1.499 (4.049)	-7.137 (4.461)
GDPG	0.651 * (0.228)	0.77 3 * (0.300)	0.503 * (0.205)
IR	-0.482 (0.496)	-0.429 (0.638)	-1.373 (0.872)
INF	2.295 * (1.012)	2.929 * (1.236)	0.0435 (0.919)
COVID-19	7.835 * (3.506)	9.754 (4.724)	1.420 (3.136)
N	88	88	88
Groups	11	11	11
Instrument	8	9	8
AR(1)	-1.56	-1.50	-0.80
AR(2)	-0.88	-0.63	-1.29
Sargan Test	0.37	0.64	2.26
Hansen test	0.59	0.29	0.81

Table 5. Determinant's capital structure: effects of LGB and CR.

Driscoll and Kraay robust standard errors in parentheses * p < 0.05.

For LGB as an indicator of government borrowing, there was a positive insignificant nexus between LGB and TDR, LTDR, and STDR. However, a negative and insignificant link existed between size, interest rate (IR), and TDR, LTDR, and STDR. GDPG had a positive and statistically significant association with TDR, LTDR, and STDR. The findings are equivalent to those addressed in Table 4.

A positive and significant connection exists between CR and TDR, LTDR, and STDR. The larger the total ratio, the greater the probability an entity is to be able to pay its costs. As a result, a larger ratio is more advantageous from the standpoint of the lender. However, from the standpoint of an investor, this has not been the case. Since current assets frequently have lower displayed yields than capital assets, shareholders ought to believe that current assets consume the bare minimum of the company's capital. The results correspond to the
findings reported by Rao et al. (2017). The latter found a link between financial leverage and the current ratio.

The annual inflation rate (INF) shows a positive association between INF and TDR, and LTDR. According to Bilgin and Dinc (2019), as inflation falls, firms enhance their financial obligation level; however, as inflation rises, firms reduce their debt level. The findings contradict those of Almanaseer (2019), who discovered a negative link between INF and bank capital structure. Nevertheless, there was a positive yet not significant connection between INF and STDR.

In terms of COVID-19, there was a positive connection between COVID-19 and TDR. The findings contradict Mohammad's (2021) finding of a negative link between COVID-19 and capital structure. However, COVID-19 had a positive yet non-significant interaction with the LTDR and STDR.

Table 6 demonstrates a positive yet insignificant link between TDR and LTDR, which lags at capital structure values. Nevertheless, STDR and its lagged capital structure values have a negative yet insignificant connection. Similarly, there is an insignificant connection between size, GDPG, IR, COVID-19, and capital structure. There was a positive and insignificant connection between EV and TDR, and LTDR. On the other hand, there is a positive but not statistically significant link between EV and STDR. The findings support the pecking order theory, which states that EV and capital structure have a positive link. However, the findings go against the trading-off theory prediction of a negative link between EV and capital structure (Assfaw 2020).

A negative and significant connection exists between GO, TDR, and LTDR. Nevertheless, the study found a positive and statistically significant association between GO and STDR. Guizani (2020) contends that when internal funds are scarce, banks with more investment options increase their reliance on financing through debt. The findings are consistent with the pecking order theory's expectation of a positive association between GO and capital structure (Myers and Majluf 1984).

Similarly, there was a negative statistically significant link between FGB and STDR. This suggests that because the government has borrowed foreign currency, the private sector is now borrowing short-term to buy the cheap foreign currency that the government would have borrowed to maintain the exchange rate. The findings are consistent with those of Demirci et al. (2019), who discovered a negative significant link between FGB and capital structure. Nevertheless, this study found a negative and significant link between FGB and STDR.

The current ratio and results are comparable to those in Table 5. All capital structure measures have a positive and significant connection with CR. In terms of the inflation rate (INF), there is a positive yet statistically significant link between INF and TDR, and LDTR. According to Anarfo (2015), when banks are able to anticipate the level of the rate, they may adjust their interest rate and reduce their burden of spending. However, as central banks monitor inflation and adjust interest rates in response to it, greater inflation will result in higher interest rates, leading banks to borrow less and increase their level of equity (Smaoui et al. 2020). Almanaseer (2019) discovered a negative relationship between INF and bank capital structure, which contradicts the results of this study. Nevertheless, INF and STDR have a negative yet not significant connection. On the other hand, there was a positive correlation between COVID-19 and TDR and LTDR. The result contradicts Mohammad's (2021) finding of a negative connection between COVID-19 and STDR.

	2-Step System GMM	2-Step System GMM	2-Step System GMM
Variables	TDR	LTDR	STDR
L.TDR	1.048 (0.522)		
L.LTDR		0.646 (0.400)	
L.STDR			-6.863 (3.376)
EV	2.251 (4.294)	4.545 (3.727)	14.13 * (5.295)
GO	-0.158 (0.143)	-0.0682 (0.139)	0.183 * (0.0827)
FGB	66.70 (46.37)	57.26 (37.78)	-41.87 * (18.51)
CR	75.16 * (30.96)	71.71 ** (22.27)	33.83 * (13.76)
LSIZE	-6.453 (10.29)	-4.714 (10.74)	-5.293 (6.347)
GDPG	1.177 (0.619)	1.147 (0.562)	0.503 * (0.205)
IR	-0.692 (1.202)	-1.253 (1.715)	-1.683 (1.206)
INF	4.787 * (1.737)	3.511 ** (0.929)	-4.593 (2.636)
COVID-19	13.66 ** (4.238)	10.62 ** (2.959)	-1.379 (2.992)
N	88	88	88
Groups	11	11	11
Instrument	9	8	10
AR(1)	-1.23	-1.88	-0.62
AR(2)	-0.99	-0.96	-0.77
Sargan Test	16.40	8.02	0.03
Hansen test	1.63	0.17	0.13

Table 6. Determinant's capital structure: Effects of FGB and CR.

Driscoll and Kraay robust standard errors in parentheses * p < 0.05, ** p < 0.01.

5. Conclusions

The article aimed to investigate the effects of government borrowing on 'banks' capital structure. This was done in a bid to put the crowding-out effect theory into perspective. A system GMM approach was employed, and the results were intriguing. Contrary to the crowding-out effects, the results revealed a positive and significant relationship between government borrowing and capital structure. Intuitively, government borrowing would crowd out the domestic market, but this is not the case in South Africa. In South Africa, the greater the government's borrowing, the greater the banks' leverage. Unlike Demirci et al.'s (2019) argument that a rise in government debt supply might decrease investors' demand for corporate debt compared to equity because government debt is a superior alternative for company debt to equity, banks in South Africa seem not to be constrained by this. Moreover, the results are inconsistent with the findings of Liang et al. (2017) and Wang et al. (2020), who find a negative association between LGB and capital structure.

Since government borrowing by all measurements did not crowd out a local market, this condition could be explained by the crowding-in effect. Further analysis indicated that foreign borrowing accounts for around 80% of overall government borrowing and does not crowd out domestic private companies. Demirci et al. (2019) argued that if resources are underutilised, public expenditure might promote investment and boost private spending, resulting in a "crowding in" effect.

Therefore, there is a need to strike a balance between government borrowing and the stimulation of the banking environment. The crowding-in effect implies that as long as the government can still provide services, they can continue to borrow, as this does not affect the banking sector. On the other hand, as long as the government seems to be borrowing more foreign than local, the bank should not be worried about the crowding-out effect. Therefore, they can still increase their debt ratios without any problems. Future research should test the cointegrating and causality relationship between government borrowing and bank capital structure. Also, given that the banking sector is constrained by Basel III's capital adequacy requirement, controlling for this factor is critical in future research.

Author Contributions: Conceptualisation: N.Z.M. and G.M.; methodology: N.Z.M.; software: N.Z.M.; validation: N.Z.M. and G.M.; formal analysis: N.Z.M.; investigation: N.Z.M.; resources: N.Z.M.; data curation: N.Z.M.; writing—original draft preparation: N.Z.M.; writing—review and editing: G.M.; supervision: G.M.; project administration: N.Z.M.; funding acquisition: N.Z.M. and G.M. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding. The APC was funded by the University of South Africa.

Data Availability Statement: Data are available upon reasonable request made to the corresponding.

Conflicts of Interest: The authors declare no conflicts of interest.

References

- Ahmad, Saeed, Muhammad Aamir, and Muhammad Umer Quddoos. 2020. Does the internal borrowing of the Pakistan government affect corporate leverage? *Journal of Economic Studies* 3: 149–63.
- Akkoyun, Huseyin Cagri. 2018. How does government debt affect corporate financing? Evidence from Ware Finance. *Job Market Paper*, 1–56.
- Almanaseer, S. Radwan. 2019. Determinants of capital structure: Evidence from Jordan. *Accounting and Finance Research* 8: 186–98. [CrossRef]
- Anarfo, Ebenezer Bugri. 2015. Determinants of capital structure of banks: Evidence from Sub-Sahara Africa. *Asian Economic and Financial Review* 5: 624–40. [CrossRef]
- Arellano, Manuel, and Olympia Bover. 1995. Another look at the instrumental variable estimation of error-components models. *Journal* of Econometrics 68: 29–51. [CrossRef]
- Arellano, Manuel, and Stephen Bond. 1991. Some Tests of specification for panel data: Monte Carlo Evidence and an application to employment equations. *The Review of Economic Studies* 58: 277–97. [CrossRef]
- Aremu, Mukaila Ayanda, Imoh Christopher Ekpo, Adeniyi Mudashiru Mustapha, and Salami Isaac Adedoyin. 2013. Determinants of capital structure in Nigerian banking sector. International Journal of Academic Research in Economics and Management Sciences 2: 27.
- Arrow, Kenneth J., and Mordecai Kruz. 1970. *Public Investment, the Rate of Return, and Optimal Fiscal Policy*. Baltimore: The Johns Hopkins Press.
- Asongu, Simplice A., Jacinta C. Nwachukwu, and Aqsa Aziz. 2018. Determinants of mobile phone penetration: Panel threshold evidence from Sub-Saharan Africa. *Journal of Global Information Technology Management* 21: 81–110. [CrossRef]
- Assfaw, Abdu Mohammed. 2020. The determinants of capital structure in Ethiopian Private commercial banks: A panel data approach. *Journal of Economics, Business, and Accountancy Ventura* 23: 108–24. [CrossRef]
- Ayturk, Yusuf. 2017. The effects of government borrowing on corporate financing: Evidence from Europe. *Finance Research Letters* 20: 96–103. [CrossRef]
- Bahal, Girish, Mehdi Raissi, and Volodymyr Tulin. 2018. Crowding-out or crowding-in? Public and private investment in India. World Development 109: 323–33. [CrossRef]
- Baker, Malcolm, and Jeffrey Wurgler. 2002. Market timing and capital structure. The Journal of Finance 57: 1–32. [CrossRef]
- Bandyopadhyay, Arindam, and Nandita Malini Barua. 2016. Factors determining the capital structure and corporate performance in India: Studying the business cycle effects. *The Quarterly Review of Economics and Finance* 61: 160–72. [CrossRef]
- Bilgin, Rumeysa, and Yusuf Dinc. 2019. Factoring as a determinant of capital structure for large firms: Theoretical and empirical analysis. *Borsa Istanbul Review* 19: 273–81. [CrossRef]

Blundell, Richard, and Stephen Bond. 1998. Initial conditions and moment restrictions in dynamic panel data models. *Journal of Econometrics* 87: 115–43. [CrossRef]

Carlson, Keith M., and Roger W. Spencer. 1975. Crowding out and its critics. Federal Reserve Bank of St. Louis Review, December 1.

Cheema, Sadia Munir. 2021. Government Debt and Corporate Leverage: Sectoral Analysis of Pakistan. *Empirical Economic Review* 4: 81–120.

David, Paul A., and John L. Scadding. 1974. Private savings: Ultrarationality, aggregation, and Denison's Law. *Journal of Political Economy* 82: 225–49. [CrossRef]

Demirci, Irem, Jennifer Huang, and Clemens Sialm. 2019. Government debt and corporate leverage: International evidence. *Journal of Financial Economics* 133: 337–56. [CrossRef]

- Frank, Murray Z., and Vidhan K. Goyal. 2009. Capital structure decisions: Which factors are reliably important? *Financial Management* 38: 1–37. [CrossRef]
- Friedman, Benjamin M. 1978. Crowding out or crowding in? Economic consequences of financing government deficits. *Brookings Papers on Economic Activity*, 593–641. [CrossRef]
- Gao, Ge, Jichang Dong, and Xiuting Li. 2022. Local government debt, real estate investment, and Corporate Investment: Evidence from China. *Sustainability* 14: 12353. [CrossRef]
- Gharaibeh, Omar K., and AL-Tahat Saqer. 2020. Determinants of capital structure: Evidence from Jordanian service companies. Investment Management and Financial Innovations 17: 364–76. [CrossRef]
- Graham, John, Mark T. Leary, and Michael R. Roberts. 2014. *How Does Government Borrowing Affect Corporate Financing and Investment?* (No. w20581). Boston: National Bureau of Economic Research. [CrossRef]
- Guizani, Moncef. 2020. The determinants of capital structure of Islamic and conventional banks: An autoregressive distributed lag approach. *Journal of Islamic Accounting and Business Research* 12: 131–47. [CrossRef]
- Handoo, Anshu, and Kapil Sharma. 2014. A study on determinants of capital structure in India. *IIMB Management Review* 26: 170–82. [CrossRef]
- Hanousek, Jan, and Anastasiya Shamshur. 2011. A stubborn persistence: Is the stability of leverage ratios determined by the stability of the economy. *Journal of Corporate Finance* 17: 1360–76. [CrossRef]
- Hansen, Lars Peter. 1982. Large sample properties of generalized method of moments estimators. *Econometrica Journal of the Econometric Society* 50: 1029–54. [CrossRef]

Harris, Christopher, and Scott Roark. 2019. Cash flow risk and capital structure decisions. Finance Research Letters 29: 393–97. [CrossRef]

Jaworski, Jacek, and Leszek Czerwonka. 2021. Determinants of enterprises capital structure in the energy industry: Evidence from European Union. *Energies* 14: 1871. [CrossRef]

- Jõeveer, Karin. 2013. Firm, country and macroeconomic determinants of capital structure: Evidence from transition economies. *Journal* of Comparative Economics 41: 294–308. [CrossRef]
- Karpavičius, Sigitas, and Fan Yu. 2017. The impact of interest rates on firm financing policies. *Journal of Corporate Financing* 45: 262–93. [CrossRef]

Keynes, John Maynard. 1936. The General Theory of Employment, Interest and Money, 1st ed. London: Macmillan and Co., Limited.

- Kraus, Alan, and Robert H. Litzenberger. 1973. A state preference model of optimal capital financial leverage. *The Journal of Finance* 28: 911–22. [CrossRef]
- Kuč, Vukašin, and Đorđe Kaličanin. 2021. Determinants of capital structure of large companies: Evidence from Serbia. *Economic Research-Ekonomska Istrazivanja* 34: 1590–607. [CrossRef]
- Liang, Yousha, Kang Shi, Lisheng Wang, and Juanyi Xu. 2017. Local government debt and firm leverage: Evidence from China. *Asian Economic Policy Review* 12: 210–32. [CrossRef]
- Modigliani, Franco, and Merton H. Miller. 1958. The cost of capital, corporation finance, and the theory of investment. *The American Economic Review* 48: 261–97.
- Modigliani, Franco, and Merton H. Miller. 1963. Corporate income taxes and the cost of capital: A correction. *The American Economic Review* 53: 433–43.
- Mohammad, Khalil Ullah. 2021. How bank capital structure decision-making change in recessions: COVID-19 from Pakistan. Asian Journal of Economics and Banking 6: 255–69. [CrossRef]

Myers, Stewart C. 1984. The capital structure puzzle. The Journal of Finance 39: 574–92. [CrossRef]

- Myers, Stewart C., and Nicholas S. Majluf. 1984. Corporate financing and investment decisions when firms have information that investors do not have. *Journal of Financial Economics* 13: 187–221. [CrossRef]
- Orangian, Alireza, Mohammad Nadiri, and Mohsen Ansari. 2021. Government Borrowing, Capital Structure, and Liquidity Policies: Evidence from Iran. *Journal of Management Technology* 21: 33–48.
- Palacín-Sánchez, María José, Luis M. Ramírez-Herrera, and Filippo Di Pietro. 2013. Capital structure of SMEs in Spanish regions. *Small Business Economics* 41: 503–19. [CrossRef]

Pesaran, M. Hashem. 2021. General diagnostic tests for cross-sectional dependence in panels. *Empirical Economics* 60: 13–50. [CrossRef]

- Rajan, Raghuram G., and Luigi Zingales. 1995. What Do We Know about Capital Structure Some Evidence from International Data. *Journal of Finance* 50: 1421–60. [CrossRef]
- Rao, KT Vigneswara, Bhavesh Prakash Joshi, and Ishita Khurana. 2017. Capital Structure Determinants: Empirical Evidence from Listed Manufacturing Firms in India. *Pacific Business Review International* 10: 17–21.

- Roodman, David. 2009. How to do xtabond2: An introduction to difference and system GMM in Stata. *The Stata Journal* 9: 86–136. [CrossRef]
- Ross, Stephen A. 1977. The determination of financial structure: The incentive signalling approach. *The Bell Journal of Economics* 8: 23–40. [CrossRef]
- Saif-Alyousfi, Abdulazeez Y., Rohani Md-Rus, Kamarun Nisham Taufil-Mohd, Hasniza Mohd Taib, and Hanita Kadir Shahar. 2020. Determinants of capital structure: Evidence from Malaysian firms. *Asia-Pacific Journal of Business Administration* 12: 283–326. [CrossRef]
- Sargan, John D. 1958. The estimation of economic relationships using instrumental variables. *Econometrica Journal of the Econometric Society* 26: 393–415. [CrossRef]
- Shahzad, Aamer, Muhammad Azeem, Mian Sajid Nazir, Xuan Vinh Vo, and Nguyen T. M. Linh. 2021. The determinants of capital structure: Evidence from SAARC countries. *International Journal of Finance and Economics* 26: 6471–87. [CrossRef]
- Sibindi, Athenia Bongani. 2018. Determinants of bank capital structure: Evidence from South Africa. *Acta Universitatis Danubius* 14: 108–26.
- Sibindi, Athenia Bongani, and Daniel Makina. 2018. Are the determinants of bank's and insurer's capital structure homogeneous? Evidence using South African data. *Cogent Economics and Finance* 6: 1. [CrossRef]
- Smaoui, Houcem, Ines Ben Salah, and Boubacar Diallo. 2020. The determinants of capital ratios in Islamic banking. *The Quarterly Review of Economics and Finance* 77: 186–94. [CrossRef]
- Tesfaye, Tseganesh. 2012. Determinants of Banks Liquidity and Their Impact on Financial Performance: An Empirical Study on Commercial Banks in Ethiopia. Unpublished. Ph.D. dissertation, Addis Ababa University, Addis Ababa, Ethiopia.
- Vo, Xuan Vinh. 2017. The determinant of capital structure in emerging markets: Evidence from Vietnam. *Research in International Business and Finance* 40: 105–13. [CrossRef]
- Wang, Jinxiang, Shinong Wu, and Yuhui Wu. 2020. The effect of local government debt on firm leverage: Empirical evidence from the city-level of China. *Journal of Finance and Economics* 46: 111–25. [CrossRef]
- Xia, Xiaomiao, Jingyue Liao, and Zitang Shen. 2021. The crowd-out effect of government debt on firm leverage. E3S Web of Conferences 235: 01023. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.





Guy Burstein * and Inon Zuckerman D

Department of Industrial Engineering & Management, Ariel University, Ariel 40700, Israel; inonzu@ariel.ac.il * Correspondence: guy.burstein@msmail.ariel.ac.il

Abstract: This paper proposes a new framework to reduce the variance and uncertainty in the risk assessment process. Today, this process is susceptible to background noise from sources of human factor biases and erroneous measurements. Our new framework consists of deconstructing the likelihood of failure function into its sub-factor and then reconstructing it in a formula that can reduce the variance and biases of a human auditor judgment. We tested our new framework on both a questionnaire study and a simulation of the risk assessment process, and the improvement in reducing the variance is significant.

Keywords: operational risk; risk management; risk factors; risk assessment; uncertainty reduction

1. Introduction

The notion of operational risk management process in organizations is an ongoing effort to identify, analyze, and assess the organization's risks and to respond to each scenario on strategic and operational management levels (ISO 31000:2018(en), Risk management—Guidelines n.d. (ISO 2018)). During the risk identification stage, the possible causes of disruptions and the magnitude of events are examined for external and internal factors. Next, the risk analysis process quantifies each event's risk level based on likelihood, results, and control mechanisms for reducing the risk and their effectiveness. Finally, a comprehensive risk assessment can be considered, and decisions can be made concerning addressing each risk.

The final stages of the risk assessment process in current practice consist of appraising the undesired outcomes of a process based on answering the questions "What is the likelihood of occurrence?" and "If it does go wrong, what are the consequences?". Both questions are subjective; thus, assessing the likelihood is not an objective truth and largely depends on the experience and personality of the assessor. In other words, the current state-of-the-art risk assessment process in operational management involves uncertainty regarding the events (as systems' complexity increases) and uncertainty regarding the risk analysis.

In recent years, the complexity of processes in projects and organizations has increased considerably (Qazi et al. 2016; Lochan et al. 2021), and such complexity contributes to the uncertainty of events (Aven and Zio 2011). In addition, the lack of knowledge and proper information regarding the process and the relations between factors significantly influences the ability to reach a high level of precision (Thomas et al. 2013). A new method that emphasizes the importance of knowledge on risk assessment was described in a paper (Aven 2013) that conditioned the risk level in the risk matrix as a function of knowledge of the system and the factors.

One of the most prominent challenges in risk management today is the auditor's subjectivity bias in determining the risk levels. While risk matrices (Elmontsri 2014) are still one of the most prominent evaluation tools for auditors, it is well known that they have several inherent limitations in defining and categorizing risk (Ball and Watt 2013). The most prominent restriction, as can be seen in Anthony Cox's paper (Anthony Cox 2008), states that the severity ratings in these matrices still depend on subjective risk attitudes. Others



Citation: Burstein, Guy, and Inon Zuckerman. 2024. Uncertainty Reduction in Operational Risk Management Process. *Risks* 12: 77. https://doi.org/10.3390/ risks12050077

Academic Editors: Eulália Mota Santos and Margarida Freitas Oliveira

Received: 11 April 2024 Revised: 6 May 2024 Accepted: 9 May 2024 Published: 11 May 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). also point to significant deficiencies in the risk assessment process due to subjectivity, heuristics, and arbitrariness in ranking the risks (Thomas et al. 2013). Those deficiencies in everyday practice can be called cognitive biases and the human factor in the process. Significant cognitive biases such as anchoring, conformity, or overconfidence biases can inadvertently affect the accuracy and precision of the likelihood assessment process (Selvik and Abrahamsen 2017).

Considering the limitation mentioned above, there are many cases where, in practice, there are significant gaps in risk assessments that were conducted at the same organization by different auditors. There is no simple method to determine which auditor is more accurate, but it is clear that a lack of consistency between assessments is undesired. For example, we analyzed a few actual risk surveys conducted in 2015–2023 in industrial companies and service organizations. We compared two risk surveys for each company, that were conducted at the same time period, and studied the gaps between the different auditors. Specifically, we looked at four risk surveys containing 71 items. By analyzing the differences between the auditors, we could see an average difference of 24% between the risk levels evaluated by different auditors, which exemplifies the inconsistency in the risk evaluation results. This gap in the research is our focus in this work. The current literature does not provide an adequate methodology to reduce the variance and bias effect on the assessment process.

In our work, we provide a methodology that will decrease the variance in human auditors and consequently improve the reliability and accuracy of the risk assessment outcome. We decompose risk into its atomic subcomponents and evaluate these subcomponents directly. These subcomponents will then be combined to assess the original risk matrix (Zampeta and Chondrokoukis 2022). Our central intuition is that the magnitude of human error and their variance in subjective assessment will reduce when assessing smaller risk factors. This intuition stems from two sources. First, compounded risk is more complex to evaluate as different individuals can quantify it differently due to cognitive biases. A comprehensive article (Kleinmuntz 1990) suggests that combining intuition and a systematic approach can reduce biases due to subjective constraints and support decision-makers to improve precision. Second, from a statistical point of view, replacing a single measurement characterized by a variance σ^2 with the average from n measurements, each characterized by a variance σ^2 , decreases the average result's variance to σ^2/n . This property of variance of sample mean may reduce the volatility of the multifactor mean versus a single factor parameter of the assessment (Todinov 2017). In our literature review, we did not find papers in the field of our topic of research and no other frameworks to reduce uncertainty were suggested by scholars. There are papers that relate to the mitigation of uncertainty in the operational process (Durugbo and Erkoyuncu 2016), although their focus is not on risk assessment methods but only in quality control and managerial procedures.

Our framework utilizes techniques from causal inference and Directed Acyclic Graph (DAG) modeling (Shrier and Platt 2008) to study the mechanism behind the uncertainty and complexity in assessing the likelihood function of risk. By utilizing DAG for modeling, we can reveal the main components that affect the likelihood of risk mechanisms to quantify and define these factors precisely.

The new framework was evaluated both in an experimental setting and a simulation. In the experimental setting, we let human auditors assess a case study using the traditional method (a single score) and the new framework of sub-factors. The findings show that our new approach of segmentation into sub-factors significantly reduced the variance of the assessments compared to the traditional evaluation. Using a Monte Carlo approach, the simulation study included two groups of randomly selected scores. The first group generated a single factor, and the second group was programmed to generate six sub-factors and calculated the total likelihood score based on the formula. This simulation also significantly reduced the variance in the new framework and provided an additional mathematical verification for the new methodology of segmentation and deconstruction of the risk.

In Section 2, we discuss the current practice in risk assessment process and introduce our method of analyzing the sources of uncertainty in operations and the basic structure of our framework. In Section 3, we present the application of the new methodology in an experimental study and in a simulation of some stochastic processes. In Section 4, we discuss the implications of the novel framework and in Section 5, we conclude and suggest some research directions in this area.

2. Framework of Deconstructing the Risk Function

2.1. The Current Practice

The risk matrix is the primary tool for assessing and prioritizing risks based on their likelihood and impact today (Burstein and Zuckerman 2023). It visually represents potential risks, allowing organizations to make informed decisions about allocating resources for risk mitigation. The matrix (see Figure 1) typically consists of a grid with two axes: (1) likelihood: this axis represents the likelihood or frequency of a risk event occurring; it is often categorized as low, medium, or high; (2) consequence: this axis represents the potential consequences or severity of a risk event if it were to occur. Impact is also often categorized as low, medium, or high.

			C	onsequenc	e	
		Negligible 1	Minor 2	Moderate 3	Major 4	Catastrophic 5
	5 Almost certain	Moderate 5	High 10			
	4 Likely	Moderate 4	High 8	High 12		
Likelihood	3 Possible	Low 3	Moderate 6	High 9	High 12	
	2 Unlikely	Low 2	Moderate 4	Moderate 6	High 8	High 10
	1 Rare	Low 1	Low 2	Low 3	Moderate 4	Moderate 5

Figure 1. A generic risk matrix.

The matrix is divided into different cells or zones, each corresponding to a combination of likelihood and impact. These cells are usually color-coded or labeled to indicate the level of risk associated with that combination. Standard labels for the cells include "low risk", "medium risk", and "high risk". By plotting individual risks on the matrix, organizations can quickly identify and prioritize those that pose the most significant threats. This helps develop a targeted risk management strategy, focusing efforts and resources on mitigating the most critical risks. Additionally, it facilitates communication and decision making among stakeholders by providing a clear visual representation of the overall risk landscape.

Assessments of likelihood and consequence are the basic steps in determining the calculated risk score level in the matrices (risk = likelihood \times consequence) (Burstein and Zuckerman 2023). Due to the biases, any small error in the likelihood directly affects the computed result. The consequence score is quantitative and thus easier to assess relatively accurately. In contrast, the likelihood score is more qualitative and obscure and highly susceptible to the auditors' judgment and bias.

This source of variance is a systematic problem of the risk matrix method; it reflects the differences between assessors and can lead to wrong information that reaches the decision-

makers. Due to many cognitive biases affecting the likelihood score, different auditors have different histories, personalities, and judgments, which is a given fact. Therefore, we suggest improving the variance by breaking down the likelihood structure. Our method will try to minimize the overall influence on the final score.

2.2. The Sources of Uncertainty in the Risk Analysis Process

Our research is based on developing a simple-to-apply framework for deconstructing the likelihood factor into its essential elements of sub-factors. The intuition behind the segmentation into sub-factors arose due to the understanding that some sources of uncertainty are related to an aleatory process and some lack of knowledge regarding the process parameters (Burstein and Zuckerman 2023). We can also relate free will and human factor interests and judgment as a source of uncertainty and unpredictability. Firstly, we can classify the different sub-factors into two main classes: the first consists of objective factors and the other of subjective factors. Here are some examples of factors belonging to each of these classes:

- Objective factors—machine output and tolerance, equipment maintenance according to specifications, mean time between failures, compliance with best practice standards, level of redundancy, and systems back-up and controls.
- Subjective factors—level of human resource integrity, level of personal training and awareness, level of process complexity, and level of variance or entropy in the process.

The class of objective factors consists of mechanisms or systems that allow us to obtain a relatively objective and quantifiable measure regarding the likelihood of risk. For instance, the mean time between failures (MTBF) is a well-known measure in which we can review experience (e.g., machine logs) to obtain an objective and reliable measure. The class of subjective factors consists of factors that are harder to quantify and often rely on subjective and qualitative evaluation. For example, the level of human resource integrity is a factor that measures, among others, the level of professionalism and commitment to the organization and to its mission. These values are more abstract and will probably have a high level of variability between different auditors.

One of the primary findings from our experience in real-world practice is that the uncertainty in the risk assessment process may be composed of a mixture of factors contributing to the total likelihood in many different directions. For example, in some cases in operational risk, we can identify situations where two minor risk factors can generate a very high-risk outcome due to positive feedback or cross effect and even the presence of an unobserved cofactor that amplifies the impact (sort of constructive interference). The combination of the threshold conditions of risk occurrence with the factors contribute to system failure. In our deconstruction framework, we can reduce the uncertainty of the assessment by quantitatively improving our understanding of the mechanism and factors behind these elements.

In Figure 2, we schematically describe the zone where the realization of two components might cause, with some likelihood, the occurrence of the risky event. These components are the natural frequency of the risk event to happen, f(x), and the conditions for failure in dealing with it, f(y). We posit that the risk occurrence is contingent on these two conditions; elsewhere, if any of them are not realized, the likelihood of a risk event is zero. The product $f(x) \times f(y)$ is the likelihood we seek when proposing this new formula.



 $P(failure) = f(x) \times f(y)$

Figure 2. The conjunction of risk sources that cause failure.

2.3. The Natural Frequency of Risk Event-f(x)

This element consists of the frequency of the risk occurrence (low, medium, high) and the volatility or variance of the risk (low, medium, high). Thus, $X = \{$ frequency, variance $\}$. We suggest that high variance can significantly increase the likelihood even if the expected risk frequency is relatively low because it "shifts" the likelihood density function into a fat tail distribution. In practical terms, we suggest that there is an amplifying effect when a high variance of the mean increases the area at the right end of the distribution. For example, when comparing two independent systems with a similar normal distribution of failure probabilities, we expect to find a higher risk of failure in the system with a higher variance on the same mean as shown in Figure 3.



Figure 3. Normal vs. "Fat tail" distributions.

Table 1 depicts the effect of variance on the likelihood score. From our point of view, very few auditors are paying the required attention to this serious effect. Neglecting the fat tails' frequency of risk can increase errors and variance in the assessment. In this proposed method, the likelihood is computed as a function of the frequency of the occurrence, its variance. We mean that if the likelihood of event is low but the variance of outcomes is medium (or high) or in the case that the likelihood is medium but the variance is high, then

that effect increases the risk likelihood due to the uncertainty span. In contrast, a very low variance will not reduce the likelihood; it is a one-directional effect. This methodology can improve the precision in determining the natural frequency of risk events, a major parameter in the risk assessment process.

Likelihood of the - occurrence	High (5–6)	High	High	High		
	Medium (3–4)	Medium	High	High		
	Low (1–2)	Low	Medium	High		
		Low (1–2)	Medium (3–4)	High (5–6)		
Variance of the Outcomes						

Table 1. The natural frequency of risk event as a function of likelihood and variance.

For example, let us assume that the likelihood of the occurrence is 4 (medium) and the variance is 5 (high). We compute f(x) to be 5 from these two parameters and the matrix below. The increase in risk level from 4 to 5 is due to the "fat tail" as given by the high variance level (of 5) and can be clearly seen in Figure 3 that there is higher density in the right "tail" of the distribution.

2.4. Conditions for Failure Occurrence-f(y)

Our method of refining the sources of uncertainty is based on analyzing the likelihood of failure using a Directed Acyclic Graph (DAG). The DAG helps break down factors and reveal the flow of association and causation. Prior research suggests that DAG modeling can reduce bias in causal inferences about exposure (Shrier and Platt 2008). We will use a DAG-based model to better understand the causes of the likelihood of failure in a system or an operational process.

Defining the root factors and their relations to the outcome is one of the most crucial steps in understanding and analyzing the causes or likelihood of failures. In Figure 3, we can see our suggestion for modeling the main sub-factors and variance generators on the likelihood of risk. If a need arises for more accuracy and precision, one can keep drilling further to a higher resolution by breaking each sub-factor into another layer of its sub-factors. In this study, we focused on the manufacturing industry just as a way to prove a concept of the framework. It can be utilized by some minor modifications or by adding a few sub-factors to fit it to other domains by finding the relevant sub-factors that influence the likelihood of a failure or a crisis and applying the novel methodology. We do not claim that the proposed model, as depicted in Figure 4, is optimal. Moreover, it is most probable that risk analysis in different sectors (e.g., financial vs. industrial sectors) will require a unique adaptation of the framework for better results. However, the level of granularity, as depicted in Figure 3, and the generic definition of the sub-factors provided were sufficient for presenting our framework and its evaluation.

We analyzed the likelihood of an event failing and devised four main sub-factors that substantially influence this outcome. The following details represent the main sources used in our formula to calculate the likelihood of failure. In our descriptions, we used values ranging from 1 (low-risk grade) to 6 (high-risk grade), but these values are ad-hoc and can be arranged as desired.

 Level of complexity—This, for example, can be described as the number of parts or moving parts in a mechanical system. In the context of a man-machine interface, it can be expressed as the number of options, elements, gauges, or operating buttons. In an organization, it can imply the number of functions or employees participating in a process. It can also be the number of outcomes and levels of entropy in the process. A value of 1 will signify low levels of complexity, while 6 will signify a high level of complexity.

- Stability and predictability—The stability of a process can be viewed as the tolerance of the output in a process or the ability to predict a failure in a system based on its historical records. This is related to variance in the process and the ability to extrapolate an outcome based on the existing data and knowledge. In cases of randomness or stochastic process, the grade will be low, 1, and in a stable and deterministic process, the grade will be high, 6.
- Command, control, and redundancy—This sub-factor relates to the mechanisms and procedures taking place in the process that aim to reduce or prevent the chance of problems. In susceptible system elements, redundancy can be crucial to minimize failure. In cases of a well-controlled and redundant system, we will grade it as low risk, 1; cases with no controls or redundancy will be graded as 6.
- Knowledge, understanding, and training—Assessment of the organizational structure, organization culture, and human resource procedures that include training and qualifications to reduce sources of failure or crisis out of control. Another aspect of this sub-factor can be the auditor's knowledge of the process and system. Proper knowledge, understanding, and training will result in a low-risk value of 1, while poor knowledge and experience will result in a high-risk value of 6.



Figure 4. DAG of likelihood sub-factors.

2.5. A Computational Example of the Framework

As used in practice, the basic notion of calculating risk levels is based on one parameter representing the likelihood and one parameter representing the consequence—their multiplication results in the risk level that goes into the risk matrices. For example, when a procedure is assessed to have a likelihood level of 4 and a consequence of 5, the risk level of this procedure will result in $4 \times 5 = 20$. As pointed out in the body of this article, the consequence assessment is much more structured and easier to quantify, but the likelihood is more challenging to estimate due to its obscureness and fuzziness. Using our proposed method for calculating the likelihood, we can see the following computation occurring:

Risk Level =
$$f(x) \times f(y)$$

First, we begin by computing the natural frequency of the event, f(x). For instance, let us assume that the frequency of the event is 4 and the variance of frequency is 6. We compute f(x) as 5 from these two and the likelihood matrix, the increase in frequency from 4 to 5 is due to the "fat tail" as given by the high variance level of 6 and can be clearly seen in Figure 3. Next, we compute the likelihood of failure, f(y). We measure and quantify the values of the four sub-factors as follows: complexity (C) = 5, stability and predictability (S) = 6, command, control, and redundancy (R) = 4, and knowledge, understanding, and training (K) = 4. We take their average and arrive at 4.75 (denoted as P). Finally, we can compute the likelihood of the risk occurring, which is the product of the two functions; therefore, it is $5 \times 4.75 = 23.75$. In our opinion, the reconstruction of the likelihood function in the new framework is more precise, less biased, and volatile. In the next section, we test our hypothesis in questionnaires and the simulation of an operational process.

3. Experimental Design and Results

To evaluate and analyze the performance of our model, we designed and conducted both an experimental study by questionaries and a simulation of a process using the Monte Carlo approach. The experiments were designed to verify the claims that applying this framework will result in fewer variations among auditors of the same situation, comparing our new method to the traditional method.

3.1. Human Evaluation Study

The study was designed to compare two groups of independent auditors that receive the same descriptions of two operational processes in each company. We formed two groups of auditors: one that used the new risk assessment framework (by a few sub-factors), and the second group conducted the traditional framework (a single number). Each group of auditors was asked to assess two textual descriptions of an operational scenario containing (1) a low-risk and (2) a high-risk level; the auditors in the first group were asked to assess the risk level of the process by the new method of a few factors that combined to a calculated total risk score. The auditors in the second group were asked to do the same but with the traditional method of ranking the risk level by only one parameter between 1 (low) and 6 (high). Our goal was to realize whether the new framework significantly reduces the variance between auditors within the group compared to the second group, which uses traditional risk assessment methods.

We conducted this study twice. The first run was on a high-risk operational scenario due to poor controls, and the second was a scenario of a well-controlled operation resulting in a low-risk operation. For the auditors of the new method, we calculated the risk level based on the data of the sub-factors they produced.

The process we applied in this study consisted of 35 randomly selected auditors aged 30–60 who were asked to reply to an internet-based questionnaire. We designed the study so that each person received a text (see Appendix A) that reflects a risk situation in an operational system. In one questionnaire, we were asked to assess the risk level traditionally (one score of 1–6 represents the risk level). In the other questionnaire, we were asked to assess the risk level using the new methodology by a set of six parameters (as explained in Section 2.5). We compute the total risk score in the new framework by calculating the average risk level of the sub-factors that construct the event multiplied by the likelihood level.

The low-risk scenario study showed the following: The traditional method resulted in an average likelihood score of 2.189 compared to a low average likelihood score of 0.806 when using the new method. Similarly, the variance of the results was 1.126 for the traditional method and 0.21 for the new method. This amounts to a reduction of 76.14%

in the variance between the groups. The difference in variance was significant using an analysis of variance test (p = 0.00194).

In the second part of the study, the high-risk scenario, the number of auditors was 35, and the results showed the following: The traditional method resulted in an average likelihood score of 5.058 compared to a low average likelihood score of 3.663 when using the new method. Similarly, the variance of the results was 0.306 for the traditional method and 0.073 for the new method. This amounts to a reduction of 81.35% in the variance between the groups. This study's results proved significant as we computed a p < 0.001 using an analysis of variance test.

3.2. Simulation Study

A complementary study to assess the mathematical validity of our proposed method was conducted using a Monte Carlo simulation. The simulation included two groups of randomly selected scores; the first group generated a single factor score, and the second group was programmed to generate six parameters that represent the sub-factors of the framework and calculated the total likelihood score based on the formula. We simulated 1998 iterations and then conducted the same Levene test, (Lim and Loh 1996) using the same procedure we used in the human evaluation study. The simulation was performed in different cases ranging from low-risk to mid-risk and high-risk scenarios that were tuned by the tolerance of the intervals of the random scores' generator. We then tested the *p*-values to check the equality of variance between groups of traditional and new methods of risk assessments. The results of Levene's test of the simulations reflect significantly lower variance in the new methodology.

The low-risk scenario simulation showed that the traditional method resulted in an average of 0.32985 likelihood compared to 0.10025 when using the new method. Similarly, the variance was 0.06137 for the traditional method and 0.010733 for the new method. This amounts to a reduction of 82.51% in the variance between the groups. The difference in variance was significant using an analysis of variance with Levene's test (F = 1628.75; $p = 5.4204 \times 10^{-299}$).

The medium-risk scenario simulation showed that the traditional method resulted in an average of 0.5841 compared to 0.2667 when using the new method. Similarly, the variance of the traditional method was 0.033185 and 0.015682 for the new method. This amounts to a reduction of 53.34% in the variance between the groups. The difference in variance was significant using Levene's test (F = 506.47; $p = 1.0708 \times 10^{-105}$).

Last, the high-risk scenario simulation showed that the traditional method resulted in an average of 0.733183 likelihood compared to 0.61941 when using the new method. Similarly, the variance was 0.04697 for the traditional method and 0.03448 for the new method. The difference in variance was significant using an analysis of variance with Levene's test (F = 4.93; p = 0.0264), although the improvement in this case is much lower.

4. Discussion

The results of both the questionnaire study and the simulations point out that the methodology of sub-factor assessments and calculating, as we suggest, can significantly reduce the variance of the likelihood risk score comparing the traditional method of one-factor assessment. In the questionnaire study, we received *p*-values that ranged between 0.00194 and 0.000679 and a substantial variance reduction of 76.14–81.35%. The simulation study of a similar structure resulted in a very significant improvement of variance reduction and *p*-values in the range of 0.0264–5.4204 × 10⁻²⁹⁹ and variance reduction improvement up to 82.5%. In this study, we also found an interesting fact in the simulations: the improvement in variation was greater in lower-risk and medium-risk scenarios compared to the high-risk scenarios. This can lead to future research and analyses to identify the causes that may occur due to the closeness to the limit of 100% likelihood in high-risk situations.

The main reason for breaking down each event to its predecessors is to get to the root cause of the given event and reduce their interdependencies. In many cases, weak

or strong interactions between sub-factors may produce positive or negative feedback or cross-action ("noise"). We address this situation by trying to give each sub-factor the risk level as objective as possible and by applying the formula that calculates the average risk level of all the sub-factors, which increases the precision and reduces the "noise" that can be an effect of these unobserved interactions. In future work, we will suggest continuing with deconstructing more layers of the cause factors to increase the accuracy and objectivity.

We believe applying the new methodology can be useful in practice and research. The segmentation of the likelihood factor into its sub-factors and calculation in the proposed formula that considers the frequency and the variance of the expected likelihood of conditions to risk occurrence combined with the average factors of a failure, given the basic conditions, can reduce errors and bias in the assessment.

5. Conclusions

In our research, we addressed the issue of reducing the uncertainty and bias due to the subjectivity of human auditors that conduct risk assessments. Our direction was that segmenting likelihood into its sub-factors could reduce the risk level assessment variance. In this work, we did not focus on the consequence part of the risk equation (i.e., risk = likelihood \times consequence) and only tried to improve the precision of the likelihood factor. We designed and conducted a study of two groups of auditors. One used the new framework, and the second assessed the same process in the regular method of a single likelihood parameter. We expanded the research with a simulation in the same approach. Analysis of the questionnaire study results and the simulation of the risk assessment process conclude that there is an improvement in precision in applying the new framework by a significant reduction in the variance of risk assessment.

The new framework that we developed is easy to apply. The formula for calculating the likelihood of risk was tested and compared to the traditional risk assessment process that does not apply segmentation of sub-factors. The research shows that it is feasible to have accurate and precise assessments when applying a simple-to-use model. This framework has many applications and can be easily adopted in industry. Our vision is to continue in this line of action to reduce the biases and uncertainty in risk management by applying a mathematical formula used in a large database and refining the accuracy by fine-tuning the sub-factors and their ranges.

Regarding the limitation of our study, first and foremost, it was conducted on a relatively small sample of human auditors and had only two different risk scenarios. Although we robustly conducted the study by conducting a large-scale simulation (30 runs of 1998 respondents each), it will be worthwhile to increase the sample size and try to add more sub-factors to increase the resolution and sensitivity of the risk assessment.

Future studies can take various paths. One research study can improve both the accuracy and precision of likelihood assessment by combining machine learning techniques that can fit a well-known likelihood level to some sub-factors that we estimate and to be able to predict by the algorithm the precise and accurate risk level that fits best with the sub-factors that we find in the process. Another future research challenge could be adding one more layer to the DAG model that may allow the deconstructing of the sub-factors to another level, increasing the precision and sensitivity of the assessment process even further. Finally, one can also conduct a broader experimental setting by comparing human auditors using the new framework on a more varied set of industries and organizations. Applying this new framework to large-scale data can be useful for training a deep learning model that may reduce uncertainty in risk assessment missions. In this paper, we focused primarily on applying this novel framework as a new concept in the industry. The initial results are good, but in future work, we can try to increase the precision and accuracy of the modal by "fine-tuning" the parameters with the adjustments of weights of the sub-factors to the model.

Author Contributions: Conceptualization, G.B. and I.Z.; methodology, G.B. and I.Z.; software, G.B.; validation, G.B. and I.Z.; formal analysis, G.B. and I.Z.; investigation, G.B. and I.Z.; data curation,

G.B.; writing—original draft preparation, G.B.; writing—review and editing, I.Z.; visualization, G.B. and I.Z.; supervision, I.Z.; project administration, G.B.; All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: No new data was created or generated in the paper.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

In the appendix, we present a sample of a questionnaire that was used to evaluate the risk of given situations in the traditional approach (with one risk score) and the new framework (with few parameters that are calculated in our formula).

The production floor is computerized in an advanced way and includes automatic data transmission to management in real time; all logistical and financial activities are well controlled in the ERP system, and there is no manual work other than through the systems. In addition, there is a secure alternative backup site and regular backups.

The professional staff is loyal and committed to the organization. There are written procedures and ongoing control over all functions. Human resource management adheres to training, implementation of a code of ethics, good discipline, and organization, and there is no tolerance for violence or improper employee activity.

It is required to assess the likelihood that the business will have an operational failure or a problematic event that will cause large losses or damage. In this case, it is required to give evaluations according to several categories (evaluation between 1 and 6 according to the guidelines of each category).

(1) Frequency estimation for the occurrence of a problematic event

(1-relatively rare [once every year or less]-6-very high frequency [daily-weekly])

(2) Level of complexity

(1-relatively simple processes with a low number of people involved; 6–q complicated process with many participants and a high level of complexity)

(3) Estimating the number of possible situations or outcomes in the process

(1-there are relatively few system modes or margins for errors and malfunctions; 6-there are many modes, and there is a high likelihood of errors in execution)

(4) Assessment of system reliability

(1–the reliability of the system and the factors are very high; 6–low reliability of the system and the participating parties)

(5) Assessment of knowledge and understanding

(1-we have extensive knowledge of the system above or similar systems and can evaluate and understand this situation well; 6-we do not have good knowledge and understanding of the system or similar systems from experience)

(6) Assessment of prevention mechanisms

(1-there are good control and prevention mechanisms that will reduce the chance of a failure; 6-there is no effective control, and there are no sufficient mechanisms to prevent a problematic event)

Condition description number 1

In this case, we observed an organized and controlled industrial plant with an advanced and secure ERP information system.

References

Anthony Cox, Louis. 2008. What's Wrong with Risk Matrices? Risk Analysis 28: 497–512. [CrossRef] [PubMed]

Aven, Terje. 2013. Practical Implications of the New Risk Perspectives. *Reliability Engineering and System Safety* 115: 136–45. [CrossRef]
Aven, Terje, and Enrico Zio. 2011. Some Considerations on the Treatment of Uncertainties in Risk Assessment for Practical Decision Making. *Reliability Engineering and System Safety* 96: 64–74. [CrossRef]

Ball, David J., and John Watt. 2013. Further Thoughts on the Utility of Risk Matrices. *Risk Analysis* 33: 2068–78. [CrossRef] [PubMed] Burstein, Guy, and Inon Zuckerman. 2023. Deconstructing Risk Factors for Predicting Risk Assessment in Supply Chains Using Machine Learning. *Journal of Risk and Financial Management* 16: 97. [CrossRef]

- Durugbo, Christopher, and John Ahmet Erkoyuncu. 2016. Mitigating Uncertainty for Industrial Service Operations: A Multi Case Study. International Journal of Operations and Production Management 36: 532–71. [CrossRef]
- Elmontsri, Mustafa. 2014. Review of the Strengths and Weaknesses of Risk Matrices. *Journal of Risk Analysis and Crisis Response* 4: 49–57. [CrossRef]
- ISO 31000:2018. 2018. Risk Management-Guidelines. New York: American National.
- Kleinmuntz, Benjamin. 1990. Why We Still Use Our Heads Instead of Formulas: Toward an Integrative Approach. *Psychological Bulletin* 107: 296–310. [CrossRef] [PubMed]
- Lim, Tjen Sien, and Wei Yin Loh. 1996. A Comparison of Tests of Equality of Variances. *Computational Statistics and Data Analysis* 22: 287–301. [CrossRef]
- Lochan, Sergey A., Tatiana P. Rozanova, Valery V. Bezpalov, and Dmitry V. Fedyunin. 2021. Supply Chain Management and Risk Management in an Environment of Stochastic Uncertainty (Retail). *Risks* 9: 197. [CrossRef]
- Qazi, Abroon, John Quigley, Alex Dickson, and Konstantinos Kirytopoulos. 2016. Project Complexity and Risk Management (ProCRiM): Towards Modelling Project Complexity Driven Risk Paths in Construction Projects. International Journal of Project Management 34: 1183–98. [CrossRef]
- Selvik, Jon T., and Eirik B. Abrahamsen. 2017. On the Meaning of Accuracy and Precision in a Risk Analysis Context. *Proceedings of the Institution of Mechanical Engineers, Part O: Journal of Risk and Reliability* 231: 91–100. [CrossRef]
- Shrier, Ian, and Robert W. Platt. 2008. Reducing Bias through Directed Acyclic Graphs. *BMC Medical Research Methodology* 8: 70. [CrossRef] [PubMed]
- Thomas, Philip, Reidar B. Bratvold, and J. Eric Bickel. 2013. The Risk of Using Risk Matrices. *Proceedings-SPE Annual Technical Conference and Exhibition* 3: 2314–29.
- Todinov, Michael Todorov. 2017. Reducing Risk by Segmentation. International Journal of Risk and Contingency Management 6: 27–46. [CrossRef]
- Zampeta, Vicky, and Gregory Chondrokoukis. 2022. An Empirical Analysis for the Determination of Risk Factors of Work-Related Accidents in the Maritime Transportation Sector. *Risks* 10: 231. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.



Article



Shareholders in the Driver's Seat: Unraveling the Impact on Financial Performance in Latvian Fintech Companies

Ramona Rupeika-Apoga ^{1,2,*}, Stefan Wendt ³ and Victoria Geyfman ⁴

- ¹ Faculty of Business, Management and Economics, University of Latvia, LV-1586 Riga, Latvia
- ² Women Researchers Council (WRC), Azerbaijan State University of Economics (UNEC),
 - Baku AZ1001, Azerbaijan
- ³ Department of Business, Bifröst University, 311 Borgarnes, Iceland; stefanwendt@bifrost.is
- ⁴ Department of Finance, Zeigler College of Business, Bloomsburg University of Pennsylvania, 400 East Second Street, Bloomsburg, PA 17815-1301, USA; vgeyfman@commonwealthu.edu
- Correspondence: rr@lu.lv

Abstract: Fintech companies are relatively young and operate in a rapidly evolving and ever-changing industry, which makes it important to understand how different factors, including shareholder presence in management roles, affect their performance. This study investigates the impact of shareholder presence in director and manager positions on the financial performance of Latvian fintechs. Our investigation centers on essential financial ratios, including Return on Assets, Return on Equity, Profit Margin, Liquidity Ratio, Current Ratio, and Solvency Ratio. Our findings suggest that the presence of shareholders in director and manager roles does not significantly affect the financial performance of fintech companies. Although the statistical analysis did not yield significant results, it is important to consider additional insights garnered from Cliff's Delta effect sizes. Specifically, despite the lack of statistical significance, practical significance indicates that fintech companies in which directors and managers are shareholders show slightly better performance than other fintech companies. Beyond shedding light on the intricacies of corporate governance in the fintech sector, this research serves as a valuable resource for investors, stakeholders, and fellow researchers seeking to understand the impact of shareholder presence in director and manager roles on the financial performance of fintechs.

Keywords: shareholders; directors; management; financial performance; fintech; conflict of interest

1. Introduction

The evolution of the fintech industry has captured widespread attention, driven by its unique characteristics and transformative impact on traditional financial services. Fintech companies operate at the nexus of finance and technology, leveraging innovation to revolutionize how individuals and businesses access and manage financial services. Their dynamic nature, fueled by technological advancements and entrepreneurial spirit, presents both challenges and opportunities that demand specialized examination. At the heart of fintech's allure lies its entrepreneurial finance dynamics, which encompass intricate investment structures, risk profiles, and performance metrics distinct from those of conventional firms. Unlike traditional companies, fintechs often navigate complex ecosystems of venture capital, angel investors, and strategic partnerships, shaping their growth trajectories and risk appetites in unique ways (Sahid et al. 2023). The growth and the ever-changing landscape of fintech companies have ignited significant interest in understanding the factors that impact their financial performance. Among these factors, the presence of shareholders as directors and managers has emerged as a critical aspect influencing decision making and resource allocation (Al-Matari et al. 2022). Investigating the relationship between shareholders as directors and managers on the one hand and financial performance on the



Citation: Rupeika-Apoga, Ramona, Stefan Wendt, and Victoria Geyfman. 2024. Shareholders in the Driver's Seat: Unraveling the Impact on Financial Performance in Latvian Fintech Companies. *Risks* 12: 54. https://doi.org/10.3390/ risks12030054

Academic Editors: Eulália Mota Santos and Margarida Freitas Oliveira

Received: 9 February 2024 Revised: 11 March 2024 Accepted: 14 March 2024 Published: 18 March 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). other can yield valuable insights into the dynamics of corporate governance in the fintech sector.

To the best of the authors' knowledge, the academic literature has not yet answered the question of the impact of shareholders as directors and managers on the financial performance of fintech companies (Najaf et al. 2022). Exploring the role of shareholders in governance has profound implications for corporate governance, a cornerstone of organizational effectiveness. Corporate governance structures determine decision-making processes, risk management strategies, and overall organizational performance. By examining how shareholder involvement in leadership roles shapes financial outcomes in fintech firms, we gain insights into governance practices in this dynamic sector. This knowledge is essential for enhancing investor confidence, ensuring regulatory compliance, and maintaining market stability. Furthermore, from the perspective of the economic literature, this research question addresses a significant gap in current research. Despite the rapid development and transformative influence of fintech firms on traditional financial services, empirical studies examining the specific impact of shareholder-directors and managers on financial performance in this sector are scarce. Our research fills this gap and advances the academic understanding of corporate dynamics within fintech firms. In addition to contributing to the academic discourse, this research provides practical insights for practitioners, policymakers, and stakeholders navigating the intricacies of the fintech landscape.

We address this research gap by examining the financial performance differences between three distinct fintech groups. The first group consists of companies where all directors and managers are shareholders¹ (DM1). The second group comprises companies where neither directors nor managers are shareholders (DM2). The third group has a mixed composition, where some directors and managers are shareholders while others are not (DM3). This study also addresses the potential conflict of interest that may arise when shareholders and management roles are separated, further enhancing our understanding of the governance landscape in the fintech industry.

Our analysis is grounded in two primary theoretical frameworks concerning corporate governance: agency theory (proposed by Jensen and Meckling (1976), further developed by Fama (1980)), and stewardship theory (introduced by Donaldson and Davis (1991)). These two theories provide contrasting views on the relationships between shareholders, directors, and managers, encompassing aspects of potential conflict of interest (CoI) (Hoi and Robin 2010). By applying these theories to the context of fintech companies, we aim to deepen our understanding of the governance challenges and opportunities inherent in this rapidly evolving industry

The study uses a sample of 96 Latvian fintech companies that were in operation during the year 2022. Our investigation centers on several vital financial ratios, including Return on Assets (ROA), Return on Equity (ROE), Profit Margin (PM), Liquidity Ratio (LR), Current Ratio (CR), and Solvency Ratio (SR). Our research design employs non-parametric tests, specifically Kruskal–Wallis tests, to assess potential statistical differences among the three groups DM1, DM2, and DM3. Furthermore, we utilize various graphical methods and visualizations, such as median heatmaps, bootstrap medians, and confidence intervals to gain further insights into the potential effect of shareholder presence on financial performance.

The Latvian fintech sector is of particular interest because Latvia-based fintech companies are not limited geographically; they have established a global presence, including in Switzerland, Africa, the UK, and Asia. These companies operate in various areas such as lending and payment services, showcasing the extensive global reach and potential impact of the Latvian fintech sector. Moreover, the fact that Latvia is a small country with a relatively modest economy allows for a comprehensive analysis of nearly all market participants. These factors collectively position the Latvian fintech sector as a compelling case study for other countries seeking to comprehend the influence of shareholder presence in director and manager roles on the financial performance of fintech companies. Our study reveals that the presence of shareholders as directors and managers does not have a statistically significant impact on essential financial ratios of fintech companies compared to those fintech companies where ownership and management are separated. This result indirectly suggests that the potential conflicts of interest associated with shareholders being separated from directors and managers do not significantly impact the financial performance of fintech companies.

The research makes several contributions to the field of fintech and corporate governance. First, the study offers a comprehensive analysis of the relationship between shareholders as directors and managers and financial performance in fintech companies. By examining multiple financial ratios, the authors provide a holistic perspective on the impact of shareholders on various aspects of a company's financial performance. Second, the findings hold practical implications for decision makers, investors, and stakeholders in the fintech sector by focusing on aspects of corporate governance that are typically expected to have a significant impact on financial performance. Third, the study adds to the growing body of research on fintech companies' performance and governance dynamics, enriching the understanding of the factors influencing financial outcomes in this rapidly evolving industry. The inclusion of conflicts of interest as an aspect of investigation further advances the scholarly discourse on the complexities and challenges faced by fintech companies in their pursuit of optimal governance and sustained financial success.

This paper is structured as follows. In Section 2, we review the related literature and develop the hypotheses for the empirical analysis. Section 3 describes the dataset and methodology, followed by the presentation of the empirical findings in Section 4. The main findings are discussed in Section 5, with concluding remarks provided in Section 6.

2. Literature Review and Hypothesis Development

Conflicts of interest are a crucial factor that can undermine the objectives of both agency theory and stewardship theory. Jensen and Meckling's (1976) seminal work emphasizes the significance of board ownership as a mechanism for reducing agency costs that result from the potential conflicts of interest between shareholders and managers. Their findings indicate that when directors hold ownership stakes, there is a natural alignment of interests that encourages a more vigilant approach to decision making. This alignment leads to a reduction in the misuse of company resources for private benefits, a phenomenon known as "private perquisite consumptions." Consequently, board ownership is a proactive strategy for curbing agency costs and enhancing financial outcomes. Moreover, the link between board ownership and enhanced corporate performance finds support in the insights presented by Berle and Means (1991). Their research underscores the significance of shared incentives between directors and managers, and shareholders. According to this approach, with higher ownership stakes held by directors, there is a heightened commitment to the company's financial well-being and value creation. This alignment cultivates an environment in which directors and managers are more attuned to shareholders' concerns, motivating them to work diligently towards achieving improved performance. In the context of Stano's (1975) study, the data indicate that firms where CEOs are also owners (executive ownership) exhibit better financial performance compared to firms where CEOs are not significant owners.

Building on these concepts, Brickley et al. (1997) highlight how board ownership acts as a catalyst for efficient company operations and effective managerial oversight. The ownership interest provides directors and managers with a vested personal interest in the company's prosperity. This interest becomes a motivating factor for them to ensure careful oversight of managerial choices. This motivation aligns with the tenets of both the agency theory and the stewardship theory, where directors are incentivized to act in ways that promote the best interests of shareholders.

It is worth noting that many fintech companies share characteristics with startups and dynamic IT firms, often operating with a relatively smaller number of employees compared to traditional financial services providers (Jain et al. 2023; Dospinescu et al. 2021; Grima et al. 2021). Within the context of fintech companies usually privately held and start-ups (Tzanaki et al. 2023; Sharma et al. 2023), the stewardship theory is of particular importance. The theory suggests that managers perceive themselves as custodians of the company's resources and are intrinsically motivated to act in the best interests of all stakeholders (Chrisman 2019). The close relationship between shareholders and managers in these small firms fosters a culture of stewardship, enabling managers to align their goals with the long-term success of the company (Davis et al. 1997), unlike in large public companies where the separation of ownership and control is more pronounced (Bonazzi and Islam 2007). The presence of shareholders as directors and managers may not have a significant impact on financial performance because both groups of managers (with and without shareholder presence) are intrinsically motivated to act responsibly and in the best interest of the company (Affes and Jarboui 2023; Caselli et al. 2023). They prioritize the long-term success of whether shareholders are directly involved in the management roles.

Conversely, agency theory centers on potential conflicts of interest between managers and shareholders, stemming from the separation of ownership and control (Fama and Jensen 1983). This theory examines how agency costs may arise when managers prioritize their own interests over acting solely in the best interest of shareholders, potentially leading to reduced firm performance (Donaldson and Davis 1991).

When considering the influence of shareholders as directors and managers on financial performance in the Latvian fintech sector, the interplay between stewardship theory and agency theory adds complexity. Stewardship theory suggests that intrinsic motivations and shared interests may mitigate conflicts of interest, while agency theory underscores the potential for conflicts that can hinder decision-making alignment. Thus, an assessment of the interplay between these theories and conflicts of interest is essential in understanding the impact of shareholder involvement as directors and managers on financial performance. With this in mind, we hypothesize the following:

H1. There are significant differences in financial performance of Latvian fintech companies based on shareholder involvement as directors and managers.

We further subdivide this hypothesis by separating the roles of DM1, DM2, and DM3. Thus,

H1.1. The financial performance of fintech companies with all directors and managers being shareholders (group DM1) differs from that of fintech companies where neither directors nor managers are shareholders (DM2).

The first sub-hypothesis posits that fintech firms in which all directors and managers are shareholders (group DM1) will exhibit different financial performance than those in which neither directors nor managers are shareholders (DM2). This proposition is based on the principles outlined by Jensen and Meckling (1976), who emphasize the importance of board ownership in mitigating conflicts of interest between shareholders and managers. Their research suggests that when directors hold ownership stakes, there is a natural alignment of interests that promotes vigilant decision making and reduces the misuse of corporate resources for personal gain. In addition, Berle and Means (1991) support the idea that higher levels of director ownership correlate with increased commitment to the financial well-being and value creation of the firm. Consequently, we expect that fintech firms with all directors and managers as shareholders (group DM1) will exhibit superior financial performance compared to those without such shareholder involvement (group DM2).

H1.2. The financial performance of fintech companies with all directors and managers being shareholders (group DM1) differs from that of fintech companies where some directors and managers are shareholders while others are not (DM3).

The second sub-hypothesis suggests that fintech firms in group DM1 will exhibit different financial performance than those in which only some directors and managers are shareholders while others are not (DM3). This hypothesis is supported by the notion that differences in shareholder–manager alignment may contribute to differences in financial outcomes. Building on the insights of Brickley et al. (1997), who highlight the motivational effect of board ownership on managerial monitoring, we find that the presence of shareholder-directors and managers in group DM1 fosters a culture of stewardship and aligns incentives with the best interests of shareholders. Conversely, in group DM3, where shareholder–manager alignment is partial, the potential for conflicts of interest may persist, potentially affecting financial performance.

H1.3. The financial performance of fintech companies where neither directors nor managers are shareholders (DM2) differs from that of fintech companies where some directors and managers are shareholders while others are not (DM3).

The third sub-hypothesis focuses on comparing the financial performance of fintech firms where neither directors nor managers are shareholders (DM2) with those where some directors and managers are shareholders while others are not (DM3). This hypothesis is based on the assumption that partial shareholder–manager alignment in the DM3 group may lead to similar financial performance outcomes as in the DM2 group. Affes and Jarboui (2023) suggest that both groups of managers, regardless of the presence of shareholders, may be intrinsically motivated to act responsibly and in the best interest of the firm, thus minimizing the potential impact of conflicts of interest on financial performance. Consequently, we expect the financial performance of firms in the DM2 group to be comparable to that of firms in the DM3 group.

3. Methodology

3.1. Study Design

This study employed a three-stage approach to investigate the impact of the presence of shareholders as directors and managers on financial performance. In the first stage, we conducted a comprehensive examination of the data. This involved carefully reviewing and validating the dataset to ensure its accuracy, completeness, and suitability for analysis. We assessed the data for any inconsistencies, missing values, or potential errors that could impact the reliability of our findings. Specifically, we verified that the dependent variables are continuous in nature, allowing for meaningful interval/ratio measurements. Additionally, we confirmed that the observations are independent, ensuring that each data point is distinct and unrelated to others. However, it is worth noting that the assumption of normality and the absence of outliers were not met in our dataset, which is not uncommon in the context of innovative technology companies (Barnes 1982; So 2006). To identify this, we employed statistical tests, such as the Shapiro–Wilk test and visual methods, including histograms and scatter plots, to assess the data's distribution and identify any potential outliers. Due to the deviations from normality and the presence of outliers, we proceeded with non-parametric methods to perform robust and valid statistical analyses, considering the unique characteristics of our data.

In the second stage, a comparative analysis was conducted to identify potential differences between different groups of fintech companies based on their involvement of shareholders as directors and managers. Fintech companies were classified into three distinct groups: DM1, DM2, and DM3. To assess potential differences between these groups, the Kruskal–Wallis H test (Marx 2016), a non-parametric statistical test, was employed. Additionally, in order to quantify the practical significance of any observed differences, Cliff's Delta effect sizes were calculated (Hollander et al. 2014).

In the third stage, graphical methods and visualizations were utilized to gain further insights into the potential effect of shareholder presence on financial performance. Median and correlation heatmaps were constructed to visually represent the relationships between various financial ratios (Raschka 2013). Bootstrap confidence intervals were used to provide a measure of uncertainty for key financial performance metrics, while boxplots facilitated the comparison of distributions between the different groups (Ramachandran and Tsokos 2021).

By combining non-parametric tests and graphical methods, this research design allows for a comprehensive and robust analysis of how the presence of shareholders as directors and managers may influence the financial performance of Latvian fintech companies.

3.2. Sample Selection and Data Collection

Understanding the characteristics of the fintech industry and selecting our sample presents a significant challenge due to the lack of an official list of fintech firms in Latvia. To address this, we adopted the definition of fintech provided by the Bank of Latvia, which includes companies that leverage new and innovative technologies in the financial services domain (Bank of Latvia 2020). These companies drive the development of novel financial products and services or significantly enhance existing ones. To identify such firms, we initially considered companies listed in the Rupeika-Apoga and Wendt (2022) study. Subsequently, we expanded our search to include fintech companies listed in Dealroom (Chiavarini et al. 2023) and the Startin database (Startin 2023) to include companies that started during 2022. We diligently verified whether these companies conformed to our defined criteria. Furthermore, we cross-checked the list of fintech companies with the Register of Enterprises of the Republic of Latvia to verify that only those entities registered in Latvia were taken into account for our study. This meticulous process of selection and verification enabled us to create a comprehensive and reliable list of Latvian fintech companies for our research.

In 2022, a remarkable surge in the fintech sector was observed in Latvia, with 106 active fintech companies officially listed in the Register of Enterprises of the Republic of Latvia. This notable increase reflects a substantial expansion compared to previous years, with 93 fintech companies identified in 2021, and 56 in 2019 (Rupeika-Apoga and Wendt 2022). Fintech companies inherently possess the flexibility to offer their services online, making it feasible for them to operate internationally. Nevertheless, despite the global nature of their services, these enterprises have actively chosen Latvia as their base of operations. The analysis of the names of the fintech owners reveals a strong connection to Latvian heritage. This trend signals a noteworthy presence of local entrepreneurs and businesses actively contributing to the industry's growth. However, as certain fintech companies mature and attain higher levels of development and success, they may embark on new ventures and explore opportunities in more prominent financial hubs and international arenas (for example, Bitfury).

Out of the initially identified 106 fintech companies, our analysis was conducted on 96 companies due to data availability limitations. However, the data collected on these companies provide valuable insight into the relationship between the presence of shareholders as directors and managers and financial performance among Latvian fintech companies.

The primary source of information for this research is derived from Bureau van Dijk's Orbis database, a subsidiary of Moody's Analytics. This database offers an extensive collection of financial and operational performance metrics as well as corporate governance characteristics of fintech companies in Latvia.

The following financial ratios were obtained from Bureau van Dijk's Orbis database for Latvian fintech firms for 2022. These indicators were selected because they cover essential aspects of financial performance, providing a comprehensive view of a company's financial health. They encompass liquidity ratios (such as the Current Ratio and Liquidity Ratio), efficiency ratios (including Return on Assets and Return on Equity), profitability ratios (such as Profit Margin), and leverage ratios (such as Solvency Ratio).

Liquidity Ratio (LR) = (Current Assets – Inventory)/Current Liabilities (1)

LR and CR assess a company's liquidity position, with lower values of these ratios generally reflecting worse financial performance.² Low levels of liquidity might also result from managers diverting funds for their private benefits as manifestation of conflicts of interest between managers and shareholders (Zaidi and Rupeika-Apoga 2021). For Latvian fintech companies, we observed that LR and CR were frequently equal or very close to each other. This is attributed to the absence or minimal presence of inventory.

Return on Total Assets (ROA) = Profit or Loss before Tax and Extraordinary Items/Total Assets (3)

ROA is an essential metric for investors, creditors, and management, as it provides insights into how efficiently a company generates profits relative to the size of its asset base. Comparing ROA across fintech companies can help investors identify which companies are more effective at generating profits from their asset investments.

Return on Equity (ROE) = Profit or Loss before Tax and Extraordinary Items/Total Shareholder Funds & Liabilities (4)

Return on Equity (ROE) is a critical financial ratio that measures the return generated on the shareholders' equity invested in the company. It is an essential indicator of how efficiently a company is using the equity capital provided by shareholders to generate profits.

Profit Margin (PM) = Profit or Loss before Tax and Extraordinary Items/Operating Revenue (Turnover) (5)

Profit Margin (PM) is a financial ratio that measures the company's ability to generate profit from its operating activities in relation to its total revenue. A higher profit margin indicates that the company is effective at controlling costs and managing its operating expenses relative to its revenue.

Solvency Ratio (SR) = Total Shareholders' Equity/Total Assets (6)

The Solvency Ratio is a financial metric that evaluates a company's ability to meet its long-term obligations and debt. It measures the extent to which the company's assets are funded by the owners (shareholders) rather than external debt.

4. Results

4.1. Sample Characteristics and Descriptive Statistics

In 2022, the 96 Latvian fintech companies employed 2202 employees, reflecting the vibrant workforce within this dynamic industry. On average, each company maintained a team of approximately 23 employees. Notably, a significant number of these companies opted for a lean workforce, with a considerable portion comprising single-employee firms. For instance, 18 companies operated with just one employee, 16 fintech companies managed with a team size of two employees, while six fintech firms employed a slightly larger team of three employees each. This trend points to the possibility that entrepreneurs take on multiple roles or possess highly specialized skills to start and propel their fintech ventures forward. Consequently, it is evident that the majority of fintech companies in Latvia are relatively small-scale operations with a limited number of employees.

As we delve further into the management structure, the average Management-to-Employee ratio (ME ratio) for Latvian fintech companies in 2022 was observed to be 1:10, i.e., for every one manager, there were ten employees within these fintech firms. Given the relatively small average company size and the ME ratio, it is likely that some directors and managers in these fintech companies took on multiple roles and responsibilities in managing various functional areas.

In 2022, there were 215 directors and managers within Latvian fintech companies. The distribution of sample by gender revealed that there was a significantly higher number of male directors and managers (188) than female directors and managers (27) within the Latvian fintech sector. The underrepresentation of women in such roles highlights a critical area of concern in fostering gender diversity and inclusivity in the industry.

Among the fintech companies analyzed, a notable pattern emerges in the relationship between directors and managers, and shareholders. Specifically, 76 individuals hold dual roles, serving as both directors/managers and shareholders. On the other hand, 139 individuals hold positions solely as directors or managers without direct ownership interests. Within the group of 76 individuals with dual roles, the majority of the group (69 individuals) are male, highlighting a significant underrepresentation of women in this group.

In addition to this ownership disparity, another aspect of gender imbalance is evident in the role of President/Chairman of the board within the analyzed Latvian fintech companies. Among the leadership positions, only one woman serves as President/Chairman, while an overwhelming majority of this group (46) are men. This striking contrast further emphasizes the lack of female representation in top leadership roles within the Latvian fintech industry.

We categorized the 96 companies into three distinct groups based on the relationships between directors and managers, and shareholders. The first group, DM1, consisted of 46 fintech companies where all director and manager positions were held by shareholders, indicating a direct financial stake in the companies they manage. The second group, DM2, comprised 40 fintech companies where none of the director and manager positions were held by shareholders, suggesting a clear separation between management and ownership roles. Finally, the third group, DM3, included 10 fintech companies where both shareholders and non-shareholders occupied director and management positions, suggesting a mix of ownership structures within these companies.

We used the key financial performance measures of ROA, ROE, PM, LR, and CR as metrics of the financial health of fintech companies. Table 1 reports the descriptive statistics of the variables of interest.

Variable Name/ Statistics	ROA	ROE	PM	CR	LR	SR
Min.	-5.442	-3.502	-20.765	0.043	0.043	-10.626
1st	-0.145	-0.027	-0.240	0.635	0.635	0.005
Median	0.021	0.193	0.068	1.375	1.327	0.295
3rd	0.247	0.621	0.257	4.223	4.035	0.686
Max.	1.294	3.116	0.673	23.888	20.162	0.960
not available	-	-	6	2	2	2

Table 1. Descriptive statistics.

The median ROA of 0.02 signifies that, on average, the companies generate a return of 2% on each unit of assets they utilize. Similarly, the median ROE of 0.19 indicates that, on average, the companies generate a return of 19% on the shareholders' equity invested in the business. Furthermore, the median profitability margin (PM) of 0.068 reveals that, on average, the companies translate 6.8% of their revenue into profit after accounting for all costs and expenses. In terms of Current Ratio (CR), the median value of 1.38 reflects stability in fulfilling short-term financial commitments. The Liquidity Ratio (LR) of 1.33 also indicates a reasonably good overall liquidity position. With a median value of 0.3, the

solvency ratio (SR) signifies that, on average, 29.5% of the assets are financed by equity, while the remaining 70.5% are financed by debt.

In our study, we observe that the levels of financial ratios in the Latvian fintech industry are consistent with previous findings (Rupeika-Apoga et al. 2023). We use both the Shapiro–Wilk tests and graphical methods, such as histograms, to check for normality of the data. The results show that the data did not follow a normal distribution (see Table A1). By adopting the winsorizing technique, we aimed to mitigate the impact of outliers on our analyses while maintaining the integrity of the original data distribution. This approach was especially relevant given the relatively small size of the fintech market and the potential for extreme observations to disproportionately influence results (Adams et al. 2018). Still, however, it is important to note that the data for financial ratios in the industry are not normally distributed. Given the dynamic nature of the fintech industry and the unique characteristics of companies within it, it is common to encounter significant variations in financial ratios. These discrepancies can be attributed to varying levels of performance and the continually changing economic and market conditions that influence fintech operations (Rupeika-Apoga and Solovjova 2016).

To address the non-normal distribution of data and the presence of extreme values or outliers, we have chosen to use medians as a more appropriate measure of central tendency. Medians provide a robust representation of the average value for each financial ratio, allowing us to gain a better understanding of the industry's performance while minimizing the impact of outlier data (So 2006).

4.2. Comparative Analysis

As our data did not follow a normal distribution, we employed the Kruskal–Wallis test to assess potential differences in financial performance between the groups. The test results revealed that there were no statistically significant differences between the groups (Table 2).

Variable Name/ Statistics	ROA	ROE	PM	CR	LR	SR
Chi-Square	1.1785	1.5866	3.9195	1.2924	1.0098	0.41495
DÎ	2	2	2	2	2	2
Asymp. Sig.	0.5547	0.4524	0.1409	0.524	0.6036	0.8126

Table 2. Kruskal–Wallis Test for financial ratios among the three groups DM1, DM2, and DM3.

To evaluate the reliability of the findings obtained from the Kruskal–Wallis test, we conducted a Robust Analysis of Covariance (Robust ANCOVA) while controlling for the number of employees. Fintech companies tend to be relatively small in scale, and a considerable proportion of them are characterized by having only a single employee. This control variable was introduced to account for potential variations in the results that could arise from differences in the size of the companies under investigation. The results revealed that, with the exception of the variable ROA, there were no statistically significant differences observed among the DM groups. However, DM1 exhibits a higher ROA compared to DM2.

4.3. Visualizing Shareholder Presence and Financial Performance: Exploring Correlations, Metrics, and Distributions

While no statistically significant differences emerged among financial ratios in various DM groups, we wanted to delve deeper and employ visual methods to examine the impact of shareholder presence on the financial performance of fintech firms. Recognizing the limitations of traditional statistical tests in our analysis of financial ratios among the three groups DM1, DM2, and DM3, we turned to median heatmaps as a powerful visual tool to uncover any potential trends or discrepancies in the data. Notably, DM1 consistently displays slightly higher medians across all financial ratios, suggesting a greater proportion

of revenue translating into profit, a stronger overall liquidity position, and a relatively higher ratio of assets funded by equity instead of debt. DM3, in contrast, tends to have lower medians across most financial ratios, suggesting weaker financial performance compared to the other two groups. However, the median heatmaps revealed no significant differences in financial ratios between the three DM groups (see Figure 1).



Figure 1. Median heatmaps for financial ratios for DM1, DM2, and DM3.

Despite observing these patterns, it is essential to assess the magnitude of these differences to understand their practical significance. The Cliff's Delta effect sizes were computed to quantify the differences in medians between the groups (see Table 3). The results show that for DM1 versus DM2, most financial ratios have effect sizes close to zero, indicating negligible differences. This suggests that the variations in medians between DM1 and DM2 may not have a meaningful impact in the real world, and such differences could merely be attributed to random fluctuations. Moreover, the small Cliff's Delta effect size for the PM financial ratio implies that, although there is a statistical difference between DM1 and DM2, its economic impact may be limited.

Table 3. Cliff's Delta effect sizes for financial ratios among the three groups DM1, DM2, and DM3.

Variable Name/ Statistics	ROA	ROE	РМ	CR	LR	SR
DM1-DM2	negligible	negligible	0.194 (small)	negligible	negligible	negligible
DM1-DM3	negligible	0.222 (small)	0.333 (medium)	0.186 (small)	0.214 (small)	negligible
DM2-DM3	negligible	0.245 (small)	0.214 (small)	0.195 (small)	0.195 (small)	negligible

On the other hand, when analyzing group DM3, the results paint a different picture. Group DM3 exhibited worse performance compared to both DM1 and DM2 in terms of ROE, PM, LR, and CR. Although the effect sizes for these differences ranged from 0.195 to 0.245 (which is considered small according to conventional effect size guidelines), these values may have meaningful and practical implications when making financial assessments and investment decisions. Interestingly, an exception arises in the DM1–DM3 comparison, where the effect size is deemed medium, signaling that PM for the DM1 group surpasses that of the DM3 group. This finding may have interesting implications for the corporate governance debate.

In the next step of our analysis, we utilized a robust statistical approach called bootstrapped medians alongside confidence intervals (CIs) to delve further into our research findings. This method allowed us to address key challenges associated with uncertainty and variability in the data, making our conclusions more robust and reliable. In Figure 2, we use bars to represent the bootstrapped medians and error bars to show the confidence intervals for each group (DM1, DM2, and DM3) and financial ratio. Each bar represents the



median value of the bootstrapped samples for a particular group and financial ratio, and the error bars indicate the range within which we are 95% confident the true median lies.

Figure 2. Bootstrapped median distribution with confidence intervals.

Based on the bootstrapped medians and their corresponding confidence intervals, we can make insightful comparisons among the DM groups with respect to their financial ratios:

- Profitability: Among the three financial ratios, DM1 consistently outperforms DM2 and DM3 in terms of median values. However, it is essential to consider the precision of these estimates. The confidence intervals for DM1 are relatively wider compared to those for DM2, indicating that there is less certainty in estimating the true population values for DM1. In contrast, the narrower confidence intervals for DM2 imply a more precise estimation of its true population values. Furthermore, the confidence intervals for DM3 overlap with both DM1 and DM2. This suggests that, statistically, there may not be significant differences between DM3 and the other two groups in terms of ROA, ROE, and PM.
- Liquidity: The analysis of liquidity measures presents a similar situation as observed for profitability ratios. DM1 exhibits larger median values for both LR and CR, indicating higher levels of liquidity compared to DM2 and DM3. However, similar to previous findings, the confidence intervals for DM1 are relatively wider, implying less precision in estimating the true population values for LR and CR. It is noteworthy that the confidence intervals for DM1 fully overlap with those of DM2 and partly with DM3 for both LR and CR. This statistical overlap indicates that there might not be significant differences in liquidity levels between DM1 and DM2, and there could be some overlap in liquidity with DM3 as well. While DM1 generally shows higher median values, the overlapping confidence intervals suggest that the differences between the groups may not be substantial.
- The analysis of the solvency ratio reiterates the pattern observed in liquidity ratios, where DM1 tends to have higher median values, indicating stronger solvency compared to DM2 and DM3. However, the wider confidence intervals for DM1 suggest caution in interpreting the results, as there is less precision in estimating its true population values.

In summary, DM1 stands out as having better performance in terms of median values across various financial ratios. However, the wider confidence intervals observed for some ratios (CR, LR, and SR) emphasize the need for caution in interpreting the results, as they indicate less precision in estimating the true population values. Moreover, the presence of overlapping intervals suggests that the differences between DM1 and the other groups (DM2 and DM3) in certain financial ratios may not be substantial.

It is important to acknowledge that our sample size is relatively small, and individual outliers within one group may have influenced some of the results. Nonetheless, these findings offer valuable insights into the interplay between different financial ratios within and between DM groups, shedding light on potential areas of focus for further analysis and decision making.

5. Discussion

Latvian fintech companies, much like their counterparts in other countries, predominantly comprise small businesses, and this often results in a scenario where a shareholder assumes the role of a director (Katalkina and Saksonova 2022). This scenario is particularly common in start-up companies. When founders establish a new company, they typically take on three essential corporate roles: they become the company's owners (as they hold shares), controllers (as they usually appoint themselves as directors), and active executives (as they often assume the position of CEO or a similar executive title) (Zhang et al. 2021). This combination of roles, especially when founders are major shareholders and directors, can be a potent force (Bachiller et al. 2015). In such cases, founders have the ability to control the company's strategic direction as directors, and they can safeguard their position by leveraging their rights as majority shareholders (Saksonova and Papiashvili 2021; Neralla 2022). In our sample of 96 Latvian fintech companies, 46 companies belonged to the DM1 category, which means that all directors and managers are shareholders. Among the DM1 companies, there were 36 that had one director and manager, eight companies with two directors and managers, and two fintech firms with three directors or managers.

As businesses grow and seek external financing, it is customary to appoint independent directors who have no or minimal ownership stakes in the business (Smith et al. 2010). The rationale behind this practice is to ensure good governance by holding directors accountable to shareholders for their decision making within the company. In the DM2 group (40 companies), there were no shareholders present as directors or managers. Among the companies in this category, it was observed that a significant number of fintech firms were led by a single individual, with 12 companies having a solo director and manager arrangement. Additionally, there were nine fintech companies with two individuals comprising the director and manager roles, and seven fintech companies with three individuals actively participating in director and management functions, while one company in the DM2 group had a maximum of 13 individuals involved in these roles.

The outcomes of the Kruskal–Wallis test suggested that there were no statistically significant differences observed between the DM groups concerning the financial ratios. This means that the presence of shareholders as directors and managers does not have a significant statistical impact on the financial ratios of the companies. These findings align with studies devoted to stewardship theory, which suggest that, in small companies, both owners and managers are incentivized to strive for optimal company performance, irrespective of their individual roles (Davis et al. 1997; Menyah 2013; Chrisman 2019). The stewardship theory suggests that managers who are also shareholders are motivated to act as stewards, working in the best interest of shareholders and stakeholders (Menyah 2013). In this context, their intrinsic motivation to protect and grow the company's resources should lead to similar efforts to enhance financial performance, regardless of their formal role as directors (Davis et al. 1997). These findings fail to support our initial hypothesis, which proposed that there are significant differences in the financial performance of Latvian fintech companies based on the presence of shareholders as directors and managers. Therefore, according to the stewardship theory, directors and managers who are shareholders (DM1

category) may demonstrate comparable levels of commitment to improving financial performance compared to directors and managers who are not shareholders (DM2 category). The shared ownership and involvement in management roles align their interests with those of shareholders, promoting a sense of stewardship and diminishing potential conflicts of interest. In the authors' opinion, it is important to note that the majority of Latvian fintech companies in the DM2 group are still relatively small. As a result, the limited scale and size of these companies may contribute to the absence of statistically significant differences in financial ratio performance.

Although we did not find statistically significant differences between the groups for financial ratios, we used graphical methods to uncover notable patterns. For example, the DM1 group consistently had slightly higher medians for all financial ratios, indicating better overall financial health. These findings suggest that active shareholder involvement in management may contribute to higher Return on Assets (ROA), better liquidity, and a greater proportion of assets financed by equity rather than debt. The practical significance of these findings lies in their potential real-world impact, highlighting the importance of shareholder involvement in management positions in improving a company's financial performance and sustainability. In addition, the calculation of Cliff's Delta effect sizes further supports the practical significance of our findings. Despite the lack of statistical significance, the small to medium effect sizes suggest meaningful differences in financial performance between the groups. Conversely, the DM3 group exhibited weaker financial performance than both DM1 and the majority of DM2 on most financial measures. These findings highlight the importance of considering the broader context and implications of our findings for decision making in the fintech industry.

6. Conclusions

This study has highlighted the intricate relationship between the presence of shareholders as directors and managers and the financial performance of Latvian fintech companies. While our analysis did not yield statistically significant differences in financial ratios across the DM groups, DM1 consistently demonstrated superior financial performance in terms of median values. The practical implications of our findings are noteworthy. For fintech firms, shareholder involvement in governance could serve as a strategic advantage, potentially leading to improved profitability, liquidity, and overall financial health. As shareholders themselves, managers and directors are likely to have a vested interest in maximizing shareholder value, which could translate into more prudent decision making and resource allocation. In addition, the alignment of interests between shareholders and management can mitigate agency conflicts and improve corporate governance practices, thereby promoting investor confidence and market stability. Moreover, the alignment of interests between shareholders and management may mitigate agency conflicts and enhance corporate governance practices, thereby fostering investor confidence and market stability.

However, it is imperative to approach these findings judiciously, given the existence of overlapping confidence intervals and the observed variability within DM2 and DM3. The lack of statistical significance can be attributed to a variety of factors, including the relatively small sample size and unobserved variables that could potentially impact financial performance results. The differences observed in DM2 and DM3 suggest that the impact of shareholder involvement in management roles may not be uniform across all fintech companies. Instead, it may depend on various factors, such as the company's size, business model, management practices, and external market conditions.

Our investigation contributes to the ongoing discourse surrounding potential conflicts of interest, thereby highlighting nuanced dynamics that warrant careful consideration. The complex interplay among corporate governance structures, shareholder participation, and financial outcomes suggests that the manifestation of conflicts of interest defies simplistic linear relationships. The study has some limitations. First, the financial performance of Latvian fintech companies is based only on data from the year 2022. The use of single-year data may not fully capture the long-term trends and variations in financial performance, and it may be influenced by specific economic conditions or industry dynamics present in that particular year. Second, there are a lack of quality data for fintech companies. While some companies may submit their balance sheets and profit/loss statements, the information provided can be limited, especially for micro-companies. This constraint might have influenced the precision and comprehensiveness of our analysis. Third, we used non-parametric tests, such as the Kruskal–Wallis test, due to the non-normal distribution of the financial data. While these tests are suitable for our data, they may have limitations compared to parametric tests, which could be employed with a larger and normally distributed dataset.

Although our study has inherent limitations, we believe that it provides valuable insights into the relationship between shareholder presence as directors and managers and the financial performance of the fintech industry. We aim to encourage further research efforts that build upon our findings and address the identified shortcomings. Future research could explore longitudinal studies that track the financial performance of Latvian fintech companies over multiple years. This approach would provide a more comprehensive understanding of long-term trends and variations in financial performance, enabling researchers to discern patterns and identify factors that influence financial outcomes over time. Furthermore, conducting cross-country comparative studies between different countries or regions is a promising direction for future research. By examining the financial performance of fintech companies across diverse regulatory environments, economic conditions, and market landscapes, researchers can gain insights into the factors driving variations in financial performance.

Author Contributions: Conceptualization, R.R.-A., V.G. and S.W.; methodology, R.R.-A. and S.W.; formal analysis, R.R.-A. and S.W.; investigation, R.R.-A. and S.W.; data curation, R.R.-A.; writing—original draft preparation, R.R.-A., S.W. and V.G.; writing—review and editing, R.R.-A., V.G. and S.W.; visualization, R.R.-A. and S.W. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Data Availability Statement: Data are available from authors upon reasonable request.

Conflicts of Interest: The authors declare no conflicts of interest.

Appendix A

Table A1. Shapiro–Wilk test results.

Variable Name/ Statistics	ROA	ROE	PM	CR	LR	SR			
	Group DM1								
Statistics	0.431	0.486	0.433	0.428	0.404	0.402			
<i>p</i> value	< 0.001	< 0.001	< 0.001	< 0.001	< 0.001	< 0.001			
Group DM2									
Statistics	0.200	0.148	0.283	0.355	0.355	0.241			
<i>p</i> value	< 0.001	< 0.001	< 0.001	< 0.001	< 0.001	< 0.001			
Group DM3									
Statistics	0.522	0.456	0.539	0.585	0.585	0.418			
<i>p</i> value	< 0.001	< 0.001	< 0.001	< 0.001	< 0.001	< 0.001			

Notes

¹ In this study, we adopt the Orbis classification of directors and managers, encompassing board members, committee participants, and executives.

² Even though very high levels of liquidity can be interpreted as inefficient use of financial resources, we do not consider the liquidity in any of the companies in the dataset as being excessive.

References

- Adams, John C., Darren K. Hayunga, Sattar Mansi, David M. Reeb, and Vincenzo Verardi. 2018. Identifying and Treating Outliers in Finance. *Financial Management* 48: 345–84. [CrossRef]
- Affes, Wajdi, and Anis Jarboui. 2023. The Impact of Corporate Governance on Financial Performance: A Cross-Sector Study. International Journal of Disclosure and Governance 20: 374–94. [CrossRef]
- Al-Matari, Ebrahim Mohammed, Mahfoudh Hussein Mgammal, Mushari Hamdan Alosaimi, Talal Fawzi Alruwaili, and Sultan Al-Bogami. 2022. Fintech, Board of Directors and Corporate Performance in Saudi Arabia Financial Sector: Empirical Study. Sustainability 14: 10750. [CrossRef]
- Bachiller, Patricia, Maria Cleofe Giorgino, and Sergio Paternostro. 2015. Influence of Board of Directors on Firm Performance: Analysis of Family and Non-Family Firms. *International Journal of Disclosure and Governance* 12: 230–53. [CrossRef]
- Bank of Latvia. 2020. FINTECH Glossary. Available online: https://www.bank.lv/en/publications-r/other-publications/fintechglossary (accessed on 7 November 2023).
- Barnes, Paul. 1982. Methodological implications of non-normally distributed financial ratios. *Journal of Business Finance and Accounting* 9: 51–62. Available online: https://www.academia.edu/10486283/METHODOLOGICAL_IMPLICATIONS_OF_NON_NORMALLY_DISTRIBUTED_FINANCIAL_RATIOS (accessed on 14 August 2023). [CrossRef]
- Berle, Adolf Augustus, and Gardiner C. Means. 1991. The Modern Corporation and Private Property. New Brunswick: Transaction Publishers.
- Bonazzi, Livia, and Sardar M. N. Islam. 2007. Agency Theory and Corporate Governance: A Study of the Effectiveness of Board in Their Monitoring of the CEO. *Journal of Modelling in Management* 2: 7–23. [CrossRef]
- Brickley, James A., Jeffrey L. Coles, and Gregg Jarrell. 1997. Leadership Structure: Separating the CEO and Chairman of the Board. Journal of Corporate Finance 3: 189–220. [CrossRef]
- Caselli, Stefano, Stefano Gatti, Carlo Chiarella, Gimede Gigante, and Giulia Negri. 2023. Do Shareholders Really Matter for Firm Performance? Evidence from the Ownership Characteristics of Italian Listed Companies. *International Review of Financial Analysis* 86: 102544. [CrossRef]
- Chiavarini, Lorenzo, Renat Lokomet, Pierre Bonvalot, and Gayane Chagharyan. 2023. Latvian Fintech Landscape List | Dealroom. Co. Available online: https://app.dealroom.co/lists/19839 (accessed on 22 July 2023).
- Chrisman, James J. 2019. Stewardship Theory: Realism, Relevance, and Family Firm Governance. *Entrepreneurship Theory and Practice* 43: 1051–66. [CrossRef]
- Davis, James H., F. David Schoorman, and Lex Donaldson. 1997. Toward a Stewardship Theory of Management. *The Academy of Management Review* 22: 20–47. [CrossRef]
- Donaldson, Lex, and James H. Davis. 1991. Stewardship Theory or Agency Theory: CEO Governance and Shareholder Returns. *Australian Journal of Management* 16: 49–64. [CrossRef]
- Dospinescu, Octavian, Nicoleta Dospinescu, and Daniela-Tatiana Agheorghiesei. 2021. FinTech Services and Factors Determining the Expected Benefits of Users: Evidence in Romania for Millennials and Generation Z. *E&M Economics and Management* 24: 101–18. [CrossRef]
- Fama, Eugene F. 1980. Agency Problems and the Theory of the Firm. Journal of Political Economy 88: 288–307. [CrossRef]
- Fama, Eugene F., and Michael C. Jensen. 1983. Separation of Ownership and Control. *The Journal of Law and Economics* 26: 301–25. [CrossRef]
- Grima, Simon, Ramona Rupeika-Apoga, Murat Kizilkaya, Inna Romānova, Rebecca Dalli Gonzi, and Mihajlo Jakovljevic. 2021. A Proactive Approach to Identify the Exposure Risk to COVID-19: Validation of the Pandemic Risk Exposure Measurement (PREM) Model Using Real-World Data. *Risk Management and Healthcare Policy* 14: 4775–87. [CrossRef]
- Hoi, Chun-Keung, and Ashok Robin. 2010. Agency Conflicts, Controlling Owner Proximity, and Firm Value: An Analysis of Dual-Class Firms in the United States. *Corporate Governance: An International Review* 18: 124–35. [CrossRef]
- Hollander, Myles, Douglas A. Wolfe, and Eric Chicken. 2014. *Nonparametric Statistical Methods*, 3rd ed. Hoboken: John Wiley & Sons, Inc.
- Jain, Ruchika, Satinder Kumar, Kiran Sood, Simon Grima, and Ramona Rupeika-Apoga. 2023. A Systematic Literature Review of the Risk Landscape in Fintech. *Risks* 11: 36. [CrossRef]
- Jensen, Michael C., and William H. Meckling. 1976. Theory of the Firm: Managerial Behavior, Agency Costs and Ownership Structure. Journal of Financial Economics 3: 305–60. [CrossRef]
- Katalkina, Oksana, and Svetlana Saksonova. 2022. Crowdfunding Cross-Border Business Financing Practice: The Evidence from the Baltic States. Lecture Notes in Networks and Systems 410 LNNS. Cham: Springer, pp. 472–81. [CrossRef]

Marx, Axel. 2016. Qualitative Comparative Analysis (QCA). Oxford: Oxford University Press. [CrossRef]

- Menyah, Kojo. 2013. Stewardship Theory. In *Encyclopedia of Corporate Social Responsibility*. Edited by Samuel O. Idowu, Nicholas Capaldi, Liangrong Zu and Ananda Das Gupta. Berlin/Heidelberg: Springer, pp. 2322–29. [CrossRef]
- Najaf, Khakan, Philip Sinnadurai, K. S. Devi, and Mohamed M. Dhiaf. 2022. Does Electronic Economics Matter to Financial Technology Firms? *Electronic Commerce Research* 1–34. [CrossRef]

- Neralla, Narsa Goud. 2022. Can Corporate Governance Structure Effect on Corporate Performance: An Empirical Investigation from Indian Companies. International Journal of Disclosure and Governance 19: 282–300. [CrossRef]
- Ramachandran, Kandethody M., and Chris P. Tsokos. 2021. Chapter 13—Empirical Methods. In *Mathematical Statistics with Applications in R*, 3rd ed. Edited by Kandethody M. Ramachandran and Chris P. Tsokos. New York: Academic Press, pp. 531–68. [CrossRef]
 Raschka, Sebastian. 2013. *Instant Heat Maps in R How-To*. Birmingham: Packt Publishing.
- Rupeika-Apoga, Ramona, and Irina Solovjova. 2016. Profiles of SMEs as Borrowers: Case of Latvia. Contemporary Studies in Economic and Financial Analysis 98: 63–76. [CrossRef]
- Rupeika-Apoga, Ramona, and Stefan Wendt. 2022. FinTech Development and Regulatory Scrutiny: A Contradiction? The Case of Latvia. *Risks* 10: 167. [CrossRef]
- Rupeika-Apoga, Ramona, Emīls Dārziņš, Deniss Filipovs, and Stefan Wendt. 2023. Competitors and Partners at the Same Time: On the Role of Fintech Companies in the Latvian Financial Market. In *The Fintech Disruption*. Edited by Thomas Walker, Elaheh Nikbakht and Maher Kooli. Palgrave Studies Financial Services Technology. Cham: Springer International Publishing, pp. 115–38. [CrossRef]
- Sahid, Abdelkebir, Yassine Maleh, Shahram Atashi Asemanjerdi, and Pedro Antonio Martín-Cervantes. 2023. A Bibliometric Analysis of the FinTech Agility Literature: Evolution and Review. *International Journal of Financial Studies* 11: 123. [CrossRef]
- Saksonova, Svetlana, and Tatiana Papiashvili. 2021. Micro and Small Businesses Access to Finance and Financial Literacy of Their Owners: Evidence from Latvia, Estonia and Georgia. *Lecture Notes in Networks and Systems* 195: 667–77. [CrossRef]
- Sharma, Sachin Kumar, P. Vigneswara Ilavarasan, and Stan Karanasios. 2023. Small Businesses and FinTech: A Systematic Review and Future Directions. *Electronic Commerce Research* 1–41. [CrossRef]
- Smith, Wendy K., Andy Binns, and Michael L. Tushman. 2010. Complex Business Models: Managing Strategic Paradoxes Simultaneously. *Long Range Planning, Business Models* 43: 448–61. [CrossRef]
- So, Jacky. 2006. Some Empirical Evidence On the Outliers and the Non-Normal Distribution of Financial Ratios. *Journal of Business Finance & Accounting* 14: 483–96. [CrossRef]
- Stano, Miron. 1975. Executive Ownership Interests and Corporate Performance. Southern Economic Journal 42: 272–78. [CrossRef]
- Startin. 2023. Latvian Startup Database. Latvian Startup Association Startin. LV. Available online: https://startin.lv/startup-database/ (accessed on 27 July 2023).
- Tzanaki, Anna, Liudmila Alekseeva, and José Azar. 2023. *Common Ownership in Fintech Markets*. New Working Paper Series, No. 329. Chicago, IL: University of Chicago Booth School of Business, Stigler Center for the Study of the Economy and the State.
- Zaidi, Syeda Hina, and Ramona Rupeika-Apoga. 2021. Liquidity Synchronization, Its Determinants and Outcomes under Economic Growth Volatility: Evidence from Emerging Asian Economies. *Risks* 9: 43. [CrossRef]
- Zhang, Jichang, Jing Long, and Alexandra Martina Eugenie von Schaewen. 2021. How Does Digital Transformation Improve Organizational Resilience?—Findings from PLS-SEM and fsQCA. *Sustainability* 13: 11487. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.



Article Robust Portfolio Optimization with Environmental, Social, and Corporate Governance Preference

Marcos Escobar-Anel * D and Yiyao Jiao D

Department of Statistical and Actuarial Sciences, University of Western Ontario, London, ON N6A 3K7, Canada; yjiao63@uwo.ca

* Correspondence: marcos.escobar@uwo.ca

Abstract: This study addresses the crucial but under-explored topic of ambiguity aversion, i.e., model misspecification, in the area of environmental, social, and corporate governance (ESG) within portfolio decisions. It considers a risk- and ambiguity-averse investor allocating resources to a risk-free asset, a market index, a green stock, and a brown stock. The study employs a robust control approach rooted in relative entropy to account for model misspecification and derive closed-form optimal investment strategies. The key contribution of this study includes demonstrating, using two sets of empirical data on asset returns and ESG ratings, the substantial influence of ambiguity on optimal trading strategies, particularly highlighting the differential effects of market, green, and brown ambiguities. As a by-product of our analytical solutions, the study contrasts ambiguity-averse investors with their non-ambiguity counterparts, revealing more cautious risk exposures with a reduction in short-selling positions for the former. Furthermore, three types of investors who employ popular suboptimal strategies are identified, together with two loss measures used to quantify their performance. The findings reveal that popular strategies, not accounting for ESG and misspecification in the model, could lead to significant financial costs, with the extent of loss varying depending on those two factors: investors' ambiguity aversion profiles and ESG preferences.

Keywords: model uncertainty; multi-attribute utility; ESG modeling

JEL Classification: C61; C20

1. Introduction

Since Merton's seminal work (Merton 1975), numerous studies have explored optimal portfolio choices for investors with ambiguity aversion (also known as robust portfolio analysis or model specification) under various assumptions. Most of the literature underscores the importance of accounting for ambiguity when making investment decisions. However, only a limited number of studies have incorporated the concept of environmental, social, and corporate governance (ESG). In this study, we extend the portfolio choice problem to include ESG ambiguity aversion within a multivariate (also known as multi-attribute) utility framework.

Environmental, social, and corporate governance (ESG) encompasses a broad spectrum of financial activities, including sustainable investing, socially responsible investing (SRI), impact investing, green investing, value-based investing, ESG investing, and triple-bottomline investing. These various approaches converge on a shared objective: the promotion of "green investing". While each investment style possesses distinct characteristics, they collectively address different facets of ESG with the aim of enhancing companies and portfolios for the benefit of stakeholders. Recent years have witnessed a notable trend among institutional investors with long-term perspectives: the integration of ESG criteria into investment decision-making and portfolio selection. According to a 2022 report by the Sustainable Investment Forum (SIF) (US SIF 2022), a staggering USD 7.6 trillion



Citation: Escobar-Anel, Marcos, and Yiyao Jiao. 2024. Robust Portfolio Optimization with Environmental, Social, and Corporate Governance Preference. *Risks* 12: 33. https:// doi.org/10.3390/risks12020033

Academic Editors: Eulália Mota Santos and Margarida Freitas Oliveira

Received: 18 December 2023 Revised: 30 January 2024 Accepted: 1 February 2024 Published: 5 February 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). in U.S.-domiciled assets have incorporated ESG criteria into their investment processes. This underscores the mounting significance of ESG considerations for enterprises, asset managers, and global shareholders (Edmans 2011; Jacobsen et al. 2019; Orsato et al. 2015). Meanwhile, there is a growing demand from both researchers and market participants to quantify ESG performance, driven by substantial needs in data analysis and financial modeling. Numerous studies have delved into ESG ratings, a key metric for quantifying the ESG performance of companies (e.g., Bermejo Climent et al. 2021; Clément et al. 2022; Drempetic et al. 2020; Polbennikov et al. 2016; Tarmuji et al. 2016). These investigations scrutinize the credibility of ESG ratings and explore their impact on financial assets and investors, establishing a foundation for further quantitative research in the realm of ESG that leverages these ratings. In this paper, we also utilize ESG ratings to discern green and brown stocks.

Moreover, numerous studies have incorporated investors' preferences for green investing into optimal portfolio strategies (e.g., some recent examples are Chen et al. (2021); Dorfleitner et al. (2012); Gasser et al. (2017)). Some of these studies utilize the framework of expected utility theory (EUT), which forms the foundation of our work. For instance, Ahmed et al. (2021) incorporate ESG as a non-pecuniary attribute of the portfolio, leading to bivariate constant absolute risk aversion (CARA) utility. Similarly, Dorfleitner and Nguyen (2017) employ bivariate CARA utility, where the ESG component appears as a non-pecuniary additive term, and compare it with the mean-variance (MVT) framework. Notably, Pástor et al. (2021) develop a single-period equilibrium model for green investing using exponential utility. Their study analyzes the impact of ESG preferences on asset prices, leading to the suggestion of an optimal three-fund separation for portfolio holdings, including a risk-free asset, a market portfolio, and an ESG portfolio. Furthermore, Escobar-Anel (2022) incorporates the ESG dimension by allowing for different levels of risk aversion for green and brown assets. This pecuniary approach accounts for the ethical dimensions of ESG and presents the first continuous-time ESG analysis with closed-form solutions within the EUT. In our work, we build upon this analytical framework by adding a market-risk dimension with a specific focus on ESG ambiguity aversions.

The literature on ambiguity aversion within portfolio choice is quite extensive; therefore, we provide a brief review with a focus on papers close to our approach. Maenhout (2004) adapted the general robust control framework developed by Anderson et al. (2003) to address the dynamic portfolio choice problem under power utility. His analysis focuses on the fundamental Merton model, featuring a single stock and a riskless asset with constant investment opportunities. He introduced the concept of ambiguity by considering uncertainty regarding the expected rate of return on stocks. Building on this work, Maenhout (2006) extended the examination to encompass the role of ambiguity aversion in scenarios involving time-varying expected stock returns. Liu (2010) expanded this analysis by incorporating Epstein–Zin preferences to explore the interplay of ambiguity aversion in portfolio choice. Meanwhile, empirical studies have indicated variations in ambiguity aversion levels for a given investor in regards to the underlying sources of risk. For instance, Dimmock et al. (2016) discovered a negative association between ambiguity aversion and stock market participation, the proportion of financial assets invested in stocks, and foreign stock ownership, as well as a positive correlation with ownership of company stocks. Similarly, Kocher et al. (2018) demonstrated that ambiguity attitudes are contingent on the outcome domain and likelihood range. More recent work by Klingebiel and Zhu (2023) utilized ambiguity aversion thresholds to elucidate why the same individual may seek some sources of ambiguity while avoiding others. Wang and Uppal (2002) were the first to consider varying levels of ambiguity aversion concerning the joint and marginal distributions of state variables. In the spirit of various ambiguity aversion levels, Branger and Larsen (2013) modeled stock prices using a jump-diffusion process, permitting separate ambiguity aversions to the diffusion and jump components. This framework enables a comparative analysis of the impact of ambiguity about the jump component versus ambiguity about the diffusive component on optimal portfolio choices. Similarly, Flor and Larsen (2014) explored scenarios in which

an investor grappled with ambiguity concerning models for both interest rates and stock returns, while Escobar et al. (2015) targeted ambiguity aversions about volatility and stock returns. When applying robust portfolio optimization in the realm of ESG considerations, Rubtsov et al. (2021) introduced the concept of climate uncertainty. This approach tackled the robust optimization problem within the context of an investor engaged in trading a stock index while also holding an illiquid claim designed to provide insurance against climate change risk. The findings of this study shed light on the significant influence of climate uncertainty on investment portfolios. Notably, heightened climate uncertainty tends to reduce investments in stocks. Subsequently, Rubtsov and Shen (2022) extended the scope by deriving an optimal stock–bond–cash portfolio. This expanded analysis delved into the repercussions of uncertainties surrounding climate change, taking into account portfolios with varying investment horizons. It is pertinent to note that both studies in this series operate on the premise of a univariate utility function and focus on a single source of uncertainty.

Our work, in its distinctive contribution to the field, diverges from previous literature by introducing an innovative ESG portfolio setting with multi-attribute utility while allowing for special considerations on green, brown, and market ambiguity aversions. Specifically, we consider a risk- and ambiguity-averse investor with the option to invest in a bank account (cash), a market index, a green stock, and a brown stock. Extending the work of Escobar-Anel (2022), this investor possesses a multivariate utility function with different risk aversions not only to green and brown stocks but also to the market index. This addition is critical, as the market index is neither green nor brown but rather an average, and it can be understood as the classical level of investor risk aversion. See Escobar-Anel and Jiao (2023) for further arguments.

Additionally, our investor contends with ambiguity regarding the dynamics of these assets. Depending on the available information, the agent may exhibit varying degrees of uncertainty regarding the models for the market index, the green stock, and the brown stock. Leveraging the robust control approach established by Anderson et al. (2003), we derive the optimal investment strategy and investigate the implications of ambiguity within this ESG context.

Our model assumes that the investor has a reference model for the market, the green stock, and the brown stock, but acknowledges that these models may be misspecified. It also recognizes the existence of alternative models that may better represent reality. Consequently, our investor seeks investment rules that are robust to model misspecification and capable of performing reasonably well across a range of plausible models. The degree of dissimilarity between the reference model and alternative models is quantified through relative entropy, which serves as a penalty in the optimization process. This penalty quantifies the investor's aversion to uncertainty concerning the reference model. The optimal portfolio is obtained in closed form after solving the relevant robust Hamilton–Jacobi–Bellman (HJB) equation (also known as the HJBI equation for HJB–Isaacs).

Several motivations underlie our assumptions regarding ambiguity concerning the market, the green stock, and the brown stock. First, the assumption of model ambiguity aligns with empirical studies, such as Ellsberg (1961), which reveal that individuals are averse not only to risk (where the probability distribution is known) but also to ambiguity (where the probability distribution is unknown). Second, as per the work of Wang and Uppal (2002), there is compelling evidence to suggest that investors exhibit different levels of ambiguity aversion toward different assets, that is, green, brown, and market. Hence, our model introduces flexibility to account for the varying levels of ambiguity aversion among investors. Third, as demonstrated by Merton (1980), accurately estimating expected returns remains a formidable challenge. In this context, it is important to emphasize that, within the framework of Anderson et al. (2003), model uncertainty primarily pertains to uncertainty about the drift of the state variables. Consequently, our study derives an optimal portfolio strategy that remains robust to model uncertainty, particularly regarding the equity risk premium of various assets.
In summary, this study makes several important contributions. Firstly, it introduces a distinctive analytical framework by segregating green, brown, and market investments within the investors' wealth process. This allows for an explicit study of the impact of each one of these asset categories on portfolio decisions, pushing the boundaries of existing methodologies.

Secondly, our work employs multi-attribute utility for risk aversions and delves into its interaction with various levels of ambiguity aversion. This interaction has not been explored in the literature, offering a novel perspective that can advance the application of multi-attribute utility.

Thirdly, our research not only provides analytical solutions but also underscores the significance of precise ESG modeling. This contribution lays the groundwork for further studies in this direction, with implications extending beyond robust portfolio optimization to contribute to the evolving landscape of ESG modeling.

Fourthly, this study underscores the substantial impact of ambiguity on optimal trading strategies while accounting for ESG preferences. In comparing the short-term effects of market, green, and brown ambiguities, we find that market ambiguity predominantly affects market weight, whereas both green and brown ambiguities influence not only the respective weights of these assets but also the market weight. Our numerical examples are based on empirical data on asset returns and ESG ratings obtained from the RepRisk database. The analysis further illustrates that this behavior extends to long-term investment strategies. Specifically, plausible levels of ambiguity can result in changes of over 50% in the optimal strategies.

Fifthly, we explore the effects of ambiguity by contrasting outcomes with those of non-ambiguity-averse investors. We provide compelling evidence that ambiguity-averse investors exhibit less aggressive exposure to risks. As demonstrated in our numerical example, ambiguity also leads to a reduction in short-selling positions.

Lastly, we categorize three types of investors who employ suboptimal strategies and two measures of welfare loss. We assess the welfare loss incurred by investors using suboptimal strategies. Our analysis reveals that investors who adopt suboptimal strategies experience non-negligible losses. For empirically relevant parameter values, the losses due to ignoring model uncertainty increase with greater ambiguity, amounting to close to 100% of the initial wealth.

The remainder of this paper is organized as follows. Section 2 outlines the portfolio choice problem. Section 3 provides a solution and outlines its general properties. In Section 4, we categorize three types of suboptimal investors and two measures of welfare loss incurred by an investor who follows a suboptimal investment strategy. Section 5 presents an analysis of the optimal portfolio through a numerical example, with Section 6 concluding, while the appendices contain proofs of important propositions and another numerical example.

2. Mathematical Setting

Let us assume that a financial market consists of one risk-free asset and three risky assets. These assets are invested over the period from time 0 to time T. Let all stochastic processes introduced in this paper be defined on a complete probability space $(\Omega, \mathcal{F}, \mathbb{P}, \{\mathcal{F}_t\}_{t \in [0,T]})$, where $\{\mathcal{F}_t\}_{t \in [0,T]}$ is a right-continuous filtration generated by standard Brownian motions (BMs).

This section is divided into three parts. First, we present a model for the underlying assets under the reference model. Section 2 constructs a family of alternative models, while Section 3 describes the self-financing wealth process and the underlying green, brown, and market synthetic portfolios.

2.1. ESG Market Model with a One-Factor Structure

We first introduce our model, which is characterized by three main assets. One asset is identified as the market portfolio (S_1). This could be a common index that merges all

assets of a given market or sectors of interest to the investor. The other two assets can be interpreted as two types of stock in the market: the so-called green stock (S_2) and the non-green alternative named, for the purpose of our study, brown stock (S_3). Both stocks are correlated with each other and with the index, resembling a one-factor CAPM model with the index playing the role of a single factor. Our model has the following structure:

$$\frac{dS_{1,t}}{S_{1,t}} = (r + \lambda_1 \sigma_1^2) dt + \sigma_1 dz_m,
\frac{dS_{2,t}}{S_{2,t}} = (r + \lambda_1 \sigma_1 \sigma_2 \rho_{12} + \lambda_g \sigma_2^2 \sqrt{1 - \rho_{12}^2}) dt + \sigma_2 (\rho_{12} dz_m + \sqrt{1 - \rho_{12}^2} dz_g),$$
(1)

$$\frac{dS_{3,t}}{S_{3,t}} = (r + \lambda_1 \sigma_1 \sigma_3 \rho_{13} + \lambda_b \sigma_3^2 \sqrt{1 - \rho_{13}^2}) dt + \sigma_3 (\rho_{13} dz_m + \sqrt{1 - \rho_{13}^2} dz_b).$$

where z_m , z_g , and z_b are independent standard Brownian motions, representing three sources of risk: market, green, and brown, respectively. The volatilities of the three assets, namely S_1 , S_2 , and S_3 , are represented by σ_1 , σ_2 , and σ_3 , respectively. Additionally, our framework incorporates a risk-free asset, such as a cash account or government bond, with its return denoted as r. The dynamic of the risk-free asset is governed by $dB_t = rB_t dt$ and is not influenced by any source of risk. We capture correlations via $corr(S_1, S_2) = \rho_{12}$, $corr(S_1, S_3) = \rho_{13}$, and $corr(S_2, S_3) = \rho_{12}\rho_{13}$. The market risk premium (from z_m) is represented by $\lambda_1 \sigma_1$, the green risk premium is expressed by $\lambda_g \sigma_2$, and the brown risk premium is $\lambda_b \sigma_3$.

2.2. Alternative Model

We refer to the market model presented in the previous section as the reference model. Our investor is uncertain about the probability distribution of the reference model and considers a set of plausible alternative models when making investment decisions. We assume that our investor is uncertain about the distributions of z_m , z_g , and z_b . Specifically, our investor is uncertain about the drifts of the stock prices. We cannot consider modeling the uncertainty of volatility because the limit of infinitely fine sampling would remove all estimation risks of the second moments. However, the first moments are notoriously difficult to estimate (Blanchard et al. 1993; Cochrane 1998; Merton 1980).

Let $e_t := (e_t^m, e_t^g, e_t^b)$ be an \mathbb{R}^3 -valued \mathcal{F}_t -progressively measurable process and define the Radon–Nikodym derivative process by

$$\Lambda_t^e = \mathbb{E}\left[\frac{d\mathbb{P}^e}{d\mathbb{P}}|F_t\right] = \exp\left\{-\int_0^t \left(\langle e_\tau, dz_\tau \rangle + \frac{1}{2}||e_\tau||^2 d\tau\right)\right\}$$
(2)

where $dz_t := (dz_{m,t}, dz_{g,t}, dz_{b,t})$. According to Girsanov's theorem, the process

$$\tilde{z}_t = z_t + \int_0^t \boldsymbol{e}_\tau d\tau \tag{3}$$

is a multidimensional Brownian motion under probability measure \mathbb{P}^{e} .

The investor's subjective measure \mathbb{P}^e is assumed to be σ -finite on (Ω, \mathcal{F}_t) , and absolutely continuous with respect to measure \mathbb{P} . e_t represents perturbations that enable the investor to deviate from the reference model. Although the reference model best characterizes the data available to the investor, they may also consider alternative models that are difficult to distinguish statistically from the reference model. Under the probability measure \mathbb{P}^e , the alternative model is

$$\frac{dS_{1,t}}{S_{1,t}} = (r + \lambda_1 \sigma_1^2 - \sigma_1 e_t^m) dt + \sigma_1 d\tilde{z}_m,
\frac{dS_{2,t}}{S_{2,t}} = (r + \lambda_1 \sigma_1 \sigma_2 \rho_{12} + \lambda_g \sigma_2^2 \sqrt{1 - \rho_{12}^2} - \sigma_2 \rho_{12} e_t^m - \sigma_2 \sqrt{1 - \rho_{12}^2} e_t^g) dt
+ \sigma_2 (\rho_{12} d\tilde{z}_m + \sqrt{1 - \rho_{12}^2} d\tilde{z}_g),
\frac{dS_{3,t}}{S_{3,t}} = (r + \lambda_1 \sigma_1 \sigma_3 \rho_{13} + \lambda_b \sigma_3^2 \sqrt{1 - \rho_{13}^2} - \sigma_3 \rho_{13} e_t^m - \sigma_3 \sqrt{1 - \rho_{13}^2} e_t^b) dt
+ \sigma_3 (\rho_{13} d\tilde{z}_m + \sqrt{1 - \rho_{13}^2} d\tilde{z}_b).$$
(4)

2.3. Wealth Process and Portfolio Setting

Let W_t denote the investor's wealth process created by allocating in $S_{1,t}$, $S_{2,t}$, $S_{3,t}$, and B_t . Let π_i (a simplified notation for $(\pi_{i,t})_{t \in [0,T]}$) denote the proportion of wealth invested in $S_{i,t}^{-1}$, according to the self-financing condition, the wealth process is

$$\frac{dW_t}{W_t} = \pi_1 \frac{dS_{1,t}}{S_{1,t}} + \pi_2 \frac{dS_{2,t}}{S_{2,t}} + \pi_3 \frac{dS_{3,t}}{S_{3,t}} + (1 - \pi_1 - \pi_2 - \pi_3) \frac{dB_t}{B_t}
= (r + \pi_1 \lambda_1 \sigma_1^2 + \pi_2 \lambda_1 \sigma_1 \sigma_2 \rho_{12} + \pi_2 \lambda_g \sigma_2^2 \sqrt{1 - \rho_{12}^2} + \pi_3 \lambda_1 \sigma_1 \sigma_3 \rho_{13}
+ \pi_3 \lambda_b \sigma_3^2 \sqrt{1 - \rho_{13}^2} dt + (\pi_1 \sigma_1 + \pi_2 \sigma_2 \rho_{12} + \pi_3 \sigma_3 \rho_{13}) dz_m + \pi_2 \sigma_2 \sqrt{1 - \rho_{12}^2} dz_g$$

$$(5)$$

$$+ \pi_3 \sigma_3 \sqrt{1 - \rho_{13}^2} dz_b$$

The term *rdt* represents the return on the cash account. Let us distribute this return among the three synthetic assets by weighing the parameters θ_m , θ_g , and θ_b , satisfying $\theta_m + \theta_g + \theta_b = 1^2$. Now, we are ready to write wealth in terms of three synthetic indices capturing the three independent sources of risk by equation

$$d\log W_t = d\log X_{m,t} + d\log X_{g,t} + d\log X_{b,t}$$
(6)

where we can see explicitly how each of these synthetic indices is impacted by its corresponding independent source of risk:

$$d \log X_{m,t} = [\theta_m r + \pi_1 \lambda_1 \sigma_1^2 + \pi_2 \lambda_1 \sigma_1 \sigma_2 \rho_{12} + \pi_3 \lambda_1 \sigma_1 \sigma_3 \rho_{13} - \frac{1}{2} (\pi_1 \sigma_1 + \pi_2 \sigma_2 \pi_{12} + \pi_3 \sigma_3 \rho_{13})^2] dt + (\pi_1 \sigma_1 + \pi_2 \sigma_2 \rho_{12} + \pi_3 \sigma_3 \rho_{13}) dz_m$$

$$d \log X_{g,t} = [\theta_g r + \pi_2 \lambda_g \sigma_2^2 \sqrt{1 - \rho_{12}^2} - \frac{1}{2} \pi_2^2 \sigma_2^2 (1 - \rho_{12}^2)] dt + \pi_2 \sigma_2 \sqrt{1 - \rho_{12}^2} dz_g$$

$$d \log X_{b,t} = [\theta_b r + \pi_3 \lambda_b \sigma_3^2 \sqrt{1 - \rho_{13}^2} - \frac{1}{2} \pi_3^2 \sigma_3^2 (1 - \rho_{13}^2)] dt + \pi_3 \sigma_3 \sqrt{1 - \rho_{13}^2} dz_b$$
(7)

Solving Equation (6), we obtain the terminal wealth at the end of time period [0, T]:

$$W_T = W_0 \frac{X_{m,T}}{X_{m,0}} \frac{X_{g,T}}{X_{g,0}} \frac{X_{b,T}}{X_{b,0}}$$
(8)

The processes $X_{m,t}$, $X_{g,t}$, and $X_{b,t}$ can be interpreted as indices denominated in generic units, with their values from the beginning of period 0 to the conclusion of period *T* dictating the ultimate wealth W_T . We refer to $X_{g,t}$ as the Green Index and $X_{b,t}$ as the Brown Index to differentiate them from the green stock (S_2) and brown stock (S_3) in the market model. Since $X_{m,t}$ is driven by market risk, $X_{g,t}$ by green risk, and $X_{b,t}$ by brown risk, these terms represent their respective contributions to wealth growth. Similarly, the investor considers the following alternative model for the indices under the probability measure \mathbb{P}^{e} .

$$d \log X_{m,t} = [\theta_m r + \pi_1 \lambda_1 \sigma_1^2 + \pi_2 \lambda_1 \sigma_1 \sigma_2 \rho_{12} + \pi_3 \lambda_1 \sigma_1 \sigma_3 \rho_{13} \\ - \frac{1}{2} (\pi_1 \sigma_1 + \pi_2 \sigma_2 \pi_{12} + \pi_3 \sigma_3 \rho_{13})^2 - (\pi_1 \sigma_1 + \pi_2 \sigma_2 \rho_{12} + \pi_3 \sigma_3 \rho_{13}) e_t^m] dt \\ + (\pi_1 \sigma_1 + \pi_2 \sigma_2 \rho_{12} + \pi_3 \sigma_3 \rho_{13}) d\tilde{z}_m \\ d \log X_{g,t} = [\theta_g r + \pi_2 \lambda_g \sigma_2^2 \sqrt{1 - \rho_{12}^2} - \frac{1}{2} \pi_2^2 \sigma_2^2 (1 - \rho_{12}^2) - \pi_2 \sigma_2 \sqrt{1 - \rho_{12}^2} e_t^g] dt \\ + \pi_2 \sigma_2 \sqrt{1 - \rho_{12}^2} d\tilde{z}_g \\ d \log X_{b,t} = [\theta_b r + \pi_3 \lambda_b \sigma_3^2 \sqrt{1 - \rho_{13}^2} - \frac{1}{2} \pi_3^2 \sigma_3^2 (1 - \rho_{13}^2) - \pi_3 \sigma_3 \sqrt{1 - \rho_{13}^2} e_t^b] dt \\ + \pi_3 \sigma_3 \sqrt{1 - \rho_{13}^2} d\tilde{z}_b$$

3. Optimal Investment Strategies

As discussed in the literature (e.g., Escobar-Anel 2022), an investor may prefer to allocate their portfolio according to different degrees of risk aversion for market risk, green risk, and brown risk, respectively. Consequently, we specify our investor's utility as a function of $X_{m,t}$, $X_{g,t}$, and $X_{b,t}$. An investor can maximize their utility by assigning π_1 , π_2 , and π_3 to obtain the best combination of $X_{m,t}$, $X_{g,t}$, and $X_{b,t}$. Next, we explain the choice of utility.

We consider the multivariate utility function

$$u(X_m, X_g, X_b) = \frac{(X_m)^{\alpha_m}}{\alpha_m} \frac{(X_g)^{\alpha_g}}{\alpha_g} \frac{(X_b)^{\alpha_b}}{\alpha_b}$$
(10)

where we choose risk aversion parameters $\alpha_b \leq \alpha_m \leq \alpha_g < 0$, and the reward function realized by choosing the alternative model under \mathbb{P}^e

$$w^{e}(X,t;\pi) = \mathbb{E}_{X,t}^{\mathbb{P}_{e}} \left[u(X_{m,T}, X_{g,T}, X_{b,T}) \right].$$
(11)

Let $\mathcal{U}[0, T]$ represent the space of all admissible strategies π_i that are \mathcal{F}_t -progressively measurable, ensuring the wealth remains non-negative for $t \in [0, T]$ and satisfies integrability conditions necessary for Equation (11). We denote $\varepsilon[0, T]$ as the set of all \mathcal{F}_t -progressively measurable processes, ensuring that the process (2) is a well-defined Radon–Nikodym derivative process. The indirect utility function is defined as

$$J(X,t) = \sup_{\pi \in \mathcal{U}[t,T]} \inf_{e \in \varepsilon[t,T]} \left\{ \mathbf{E}_{t}^{\mathbb{P}^{e}} \left[\int_{t}^{T} \left(\frac{(e_{\tau}^{m})^{2}}{2\Psi_{m}(\tau, X_{\tau})} + \frac{(e_{\tau}^{g})^{2}}{2\Psi_{g}(\tau, X_{\tau})} + \frac{(e_{\tau}^{b})^{2}}{2\Psi_{b}(\tau, X_{\tau})} \right) d\tau \right] + w^{e}(X,t;\pi) \right\},$$
(12)

where the expectation term in the equation functions as a penalty for the deviation from the reference model. The penalty depicts the relative entropy between the alternative and reference models, which is expressed as

$$D_{KL}(\mathbb{P}^{e}||\mathbb{P}) = \mathbf{E}_{t}^{\mathbb{P}^{e}}\left[\int_{t}^{T} \frac{1}{2}||\boldsymbol{e}_{\tau}||^{2}d\tau\right]$$
(13)

and perturbations e_t^m , e_t^g , and e_t^b are scaled by Ψ_m , Ψ_g , and Ψ_b , respectively.

 Ψ_m , Ψ_g , and Ψ_b capture ambiguity aversions toward market, green, and brown dynamics, respectively. For analytical tractability, we assume the following

$$\Psi_i = \frac{\phi_i}{\alpha_i J}, \quad i = m, g, b, \tag{14}$$

where $\phi_i > 0$ are recognized as ambiguity aversion parameters. Building upon the concept of 'homothetic robustness' proposed by Maenhout (2004), we depart from the conventional approach of Anderson et al. (2003), where a constant ambiguity aversion parameter $(\Psi_i(X_m, X_g, X_b, t) = \hat{\phi}_i)$ is employed. Instead, we introduce a scaling factor, denoted as $\alpha_i J$, which divides ϕ_i . This modification ensures that the optimal weights remain unaffected by variations in the state variables X_m , X_g , and X_b , thereby preserving the homothetic nature of preferences amid the changing state variables.

As explained by Maenhout (2004), the consideration of homotheticity is not merely a modeling convenience; it carries significant implications for several reasons. First, despite the economic growth, rates of return remain stationary. Second, when the magnitude of the state variable becomes pertinent, the natural unit invariance of optimal decisions is disrupted, requiring adjustments in calibrations. Finally, homotheticity plays a pivotal role in facilitating aggregation and the construction of a representative agent, thereby enhancing the comprehensibility and applicability of the model.

Our construction also accommodates distinct values of ϕ_m , ϕ_g , and ϕ_b . Recognizing that investors may know more about the distribution of some assets than others, we employ ϕ_i to govern the degree of ambiguity aversion. This approach enables a systematic examination of how these ϕ_i values influence both the optimal weightings and resultant utility levels.

Therefore, the HJB equation for Equation (12) is

$$\begin{split} \sup_{\pi_{1},\pi_{2},\pi_{3}} \inf_{e^{m},e^{g},e^{b}} \left\{ J_{t} + (\theta_{m}r + \pi_{1}\lambda_{1}\sigma_{1}^{2} + \pi_{2}\lambda_{1}\sigma_{1}\sigma_{2}\rho_{12} + \pi_{3}\lambda_{1}\sigma_{1}\sigma_{3}\rho_{13} \\ - (\pi_{1}\sigma_{1} + \pi_{2}\sigma_{2}\rho_{12} + \pi_{3}\sigma_{3}\rho_{13})e^{m})xJ_{x} + \frac{1}{2}(\pi_{1}\sigma_{1} + \pi_{2}\sigma_{2}\rho_{12} + \pi_{3}\sigma_{3}\rho_{13})^{2}x^{2}J_{xx} \\ + (\theta_{g}r + \pi_{2}\lambda_{g}\sigma_{2}^{2}\sqrt{1 - \rho_{12}^{2}} - \pi_{2}\sigma_{2}\sqrt{1 - \rho_{12}^{2}}e^{g})yJ_{y} + \frac{1}{2}\pi_{2}^{2}\sigma_{2}^{2}(1 - \rho_{12}^{2})y^{2}J_{yy} \\ + (\theta_{b}r + \pi_{3}\lambda_{b}\sigma_{3}^{2}\sqrt{1 - \rho_{13}^{2}} - \pi_{3}\sigma_{3}\sqrt{1 - \rho_{13}^{2}}e^{b})zJ_{z} + \frac{1}{2}\pi_{3}^{2}\sigma_{3}^{2}(1 - \rho_{13}^{2})z^{2}J_{zz} \\ + \frac{(e^{m})^{2}}{2\Psi_{m}} + \frac{(e^{g})^{2}}{2\Psi_{g}} + \frac{(e^{b})^{2}}{2\Psi_{b}} \right\} = 0, \end{split}$$

$$(15)$$

where X_m , X_g , X_b are denoted as x, y, z. A subsequent proposition is obtained by solving the HJB equation presented above (see Appendix A.1).

Proposition 1. The optimal weights for solving the HJB Equation (15) are given by

$$\pi_{1}^{*} = \frac{\lambda_{1}}{1 - \alpha_{m} + \phi_{m}} - \frac{\sigma_{2}}{\sigma_{1}} \rho_{12} \pi_{2}^{*} - \frac{\sigma_{3}}{\sigma_{1}} \rho_{13} \pi_{3}^{*},$$

$$\pi_{2}^{*} = \frac{\lambda_{g}}{\sqrt{1 - \rho_{12}^{2}} (1 - \alpha_{g} + \phi_{g})},$$

$$\pi_{3}^{*} = \frac{\lambda_{b}}{\sqrt{1 - \rho_{13}^{2}} (1 - \alpha_{b} + \phi_{b})}.$$
(16)

The value function can be expressed as

$$J(X,t) = \frac{X_m^{\alpha_m}}{\alpha_m} \frac{X_g^{\alpha_g}}{\alpha_g} \frac{X_b^{\alpha_b}}{\alpha_b} \exp(b(T-t)),$$
(17)

where

$$b = \frac{1}{2}\lambda_1^2 \sigma_1^2 \frac{\alpha_m}{1 - \alpha_m + \phi_m} + \frac{1}{2}\lambda_g^2 \sigma_2^2 \frac{\alpha_g}{1 - \alpha_g + \phi_g} + \frac{1}{2}\lambda_b^2 \sigma_3^2 \frac{\alpha_b}{1 - \alpha_b + \phi_b} + (\theta_m \alpha_m + \theta_g \alpha_g + \theta_b \alpha_b)r.$$
(18)

The optimal wealth process is therefore

$$\frac{dW_t}{W_t} = \left(r + \frac{\lambda_1^2 \sigma_1^2}{1 - \alpha_m + \phi_m} + \frac{\lambda_g^2 \sigma_2^2}{1 - \alpha_g + \phi_g} + \frac{\lambda_b^2 \sigma_3^2}{1 - \alpha_b + \phi_b}\right) dt + \frac{\lambda_1 \sigma_1}{1 - \alpha_m + \phi_m} dz_m + \frac{\lambda_g \sigma_2}{1 - \alpha_g + \phi_g} dz_g + \frac{\lambda_b \sigma_3}{1 - \alpha_b + \phi_b} dz_b.$$
(19)

Remark 1.

1. As shown in Appendix A.1, if we do not specify $\Psi_i = \frac{\phi_i}{\alpha_i I}$, the optimal weights are

$$\pi_{1}^{*} = \frac{\lambda_{1}J_{x}}{\Psi_{m}x(J_{x})^{2} - xJ_{xx}} - \frac{\sigma_{2}}{\sigma_{1}}\rho_{12}\pi_{2}^{*} - \frac{\sigma_{3}}{\sigma_{1}}\rho_{13}\pi_{3}^{*},$$

$$\pi_{2}^{*} = \frac{\lambda_{g}}{\sqrt{1 - \rho_{12}^{2}}} \frac{J_{y}}{\Psi_{g}y(J_{y})^{2} - yJ_{yy}},$$

$$\pi_{3}^{*} = \frac{\lambda_{b}}{\sqrt{1 - \rho_{13}^{2}}} \frac{J_{z}}{\Psi_{b}z(J_{z})^{2} - zJ_{zz}}.$$
(20)

To elaborate, if we were to assume constant values for Ψ_i , then the optimal weights would be contingent on the state variables X_m , X_g , and X_b . However, to maintain homotheticity, we introduce a modification by setting $\Psi_i = \frac{\phi_i}{\alpha_i j}$. This adjustment ensures that the optimal weights are independent of variations in the state variables X_m , X_g , and X_b .

- 2. It is interesting to see that ambiguity aversion parameters act similarly to risk aversion parameters in the representation of the optimal weights. This is not new, as highlighted by Maenhout (2006) for the CRRA utility. This observation is important for our multi-attribute utility, as it conveys the notion that investors might, consciously or not, exchange risk aversion for ambiguity aversion and vice versa. This provides yet another motivation for the validity of a multi-attribute utility, that is, a utility that allows for different risk aversions for different sources of risk.
- 3. While an investor characterized by the utility function

 $u(X_m, X_g, X_b) = \frac{(X_m)^{\alpha_m - \phi_m}}{\alpha_m - \phi_m} \frac{(X_g)^{\alpha_g - \phi_g}}{\alpha_g - \phi_g} \frac{(X_b)^{\alpha_b - \phi_b}}{\alpha_b - \phi_b}$ and lacking ambiguity aversions would yield the same optimal weights as an ambiguity-averse investor, it is essential to highlight that the indirect utility functions do not align. Consequently, these two scenarios do not constitute the same utility problem. In this instance, we have a b of

$$b = \frac{1}{2}\lambda_1^2 \sigma_1^2 \frac{\alpha_m - \phi_m}{1 - \alpha_m + \phi_m} + \frac{1}{2}\lambda_g^2 \sigma_2^2 \frac{\alpha_g - \phi_g}{1 - \alpha_g + \phi_g} + \frac{1}{2}\lambda_b^2 \sigma_3^2 \frac{\alpha_b - \phi_b}{1 - \alpha_b + \phi_b} + (\theta_m(\alpha_m - \phi_m) + \theta_g(\alpha_g - \phi_g) + \theta_b(\alpha_b - \phi_b))r$$
(21)

different to that of Equation (17).

4. Analysis of Suboptimal Strategies

Investors may at times adopt suboptimal strategies for various reasons. A frequent rationale for such decisions is the absence of sufficient knowledge to construct an optimal strategy. In our context, for instance, an investor might have varying degrees of ambiguity aversion toward their green and brown stocks. However, owing to the lack of requisite knowledge to formulate an optimal solution, they opt for a strategy that does not account for robustness. Consequently, the resulting asset allocations for the investor are suboptimal, leading to a reduction in overall utility.

More formally, given portfolio weights $\pi^s = (\pi_1^s, \pi_2^s, \pi_3^s)$ representing a suboptimal strategy, let us represent the value function obtained from such a suboptimal strategy as J^s . This is

$$J^{s}(X,t) = \inf_{e \in \varepsilon[t,T]} \left\{ \mathbf{E}_{t}^{\mathbb{P}^{e}} \left[\int_{t}^{T} \left(\frac{(e_{\tau}^{m})^{2}}{2\Psi_{m}(\tau,X_{\tau})} + \frac{(e_{\tau}^{g})^{2}}{2\Psi_{g}(\tau,X_{\tau})} + \frac{(e_{\tau}^{b})^{2}}{2\Psi_{b}(\tau,X_{\tau})} \right) d\tau \right] + w^{e}(X,t;\boldsymbol{\pi}^{s}) \right\}.$$
(22)

It is evident that $J^s \leq J$, indicating a suboptimal level of satisfaction for the investor stemming from the application of a suboptimal strategy, as demonstrated in the following proposition (see Appendix A.2).

Proposition 2. The value function for the investor employing a suboptimal strategy $\pi^s = (\pi_1^s, \pi_2^s, \pi_3^s)$ is given by

$$J^{s}(X,t) = \frac{X_{m}^{\alpha_{m}}}{\alpha_{m}} \frac{X_{g}^{\alpha_{g}}}{\alpha_{g}} \frac{X_{b}^{\alpha_{b}}}{\alpha_{b}} \exp(b^{s}(T-t)),$$
(23)

where

$$b^{s} = (\theta_{m}r + \pi_{1}^{s}\lambda_{1}\sigma_{1}^{2} + \pi_{2}^{s}\lambda_{1}\sigma_{1}\sigma_{2}\rho_{12} + \pi_{3}^{s}\lambda_{1}\sigma_{1}\sigma_{3}\rho_{13})\alpha_{m} + \frac{1}{2}(\pi_{1}^{s}\sigma_{1} + \pi_{2}^{s}\sigma_{2}\rho_{12} + \pi_{3}^{s}\sigma_{3}\rho_{13})^{2}\alpha_{m}(\alpha_{m} - 1) + (\theta_{g}r + \pi_{2}^{s}\lambda_{g}\sigma_{2}^{2}\sqrt{1 - \rho_{12}^{2}})\alpha_{g} + \frac{1}{2}(\pi_{2}^{s})^{2}\sigma_{2}^{2}(1 - \rho_{12}^{2})\alpha_{g}(\alpha_{g} - 1) + (\theta_{b}r + \pi_{3}^{s}\lambda_{b}\sigma_{3}^{2}\sqrt{1 - \rho_{13}^{2}})\alpha_{b} + \frac{1}{2}(\pi_{3}^{s})^{2}\sigma_{3}^{2}(1 - \rho_{13}^{2})\alpha_{b}(\alpha_{b} - 1) - \frac{\phi_{m}}{2}(\pi_{1}^{s}\sigma_{1} + \pi_{2}^{s}\sigma_{2}\rho_{12} + \pi_{3}^{s}\sigma_{3}\rho_{13})^{2}\alpha_{m} - \frac{\phi_{g}}{2}(\pi_{2}^{s})^{2}\sigma_{2}^{2}(1 - \rho_{12}^{2})\alpha_{g} - \frac{\phi_{b}}{2}(\pi_{3}^{s})^{2}\sigma_{3}^{2}(1 - \rho_{13}^{2})\alpha_{b}.$$

$$(24)$$

The suboptimal wealth process W_t^s follows

$$\frac{dW_t^s}{W_t^s} = (r + \pi_1^s \lambda_1 \sigma_1^2 + \pi_2^s \lambda_1 \sigma_1 \sigma_2 \rho_{12} + \pi_2^s \lambda_g \sigma_2^2 \sqrt{1 - \rho_{12}^2} + \pi_3^s \lambda_1 \sigma_1 \sigma_3 \rho_{13} \\
+ \pi_3^s \lambda_b \sigma_3^2 \sqrt{1 - \rho_{13}^2}) dt + (\pi_1^s \sigma_1 + \pi_2^s \sigma_2 \rho_{12} + \pi_3^s \sigma_3 \rho_{13}) dz_m + \pi_2^s \sigma_2 \sqrt{1 - \rho_{12}^2} dz_g \quad (25) \\
+ \pi_3^s \sigma_3 \sqrt{1 - \rho_{13}^2} dz_b$$

There are three choices of suboptimal strategies that are meaningful to an investor.

1. First, we introduce the unintentionally unambiguous investor (UUI). This investor employs suboptimal strategies arising from a lack of model uncertainty. The investor chooses such a strategy not because of its lack of uncertainty but rather because of a lack of knowledge of a better/optimal solution. That is, if investors care about model uncertainty, yet they do now know how to generate and handle alternative models, they will suffer from utility loss by employing suboptimal strategies. They mistakenly choose the model with $\phi_m = \phi_g = \phi_b = 0$, and employ the suboptimal strategy:

$$\pi_{1}^{s} = \frac{\lambda_{1}}{1 - \alpha_{m}} - \frac{\sigma_{2}}{\sigma_{1}} \rho_{12} \pi_{2}^{s} - \frac{\sigma_{3}}{\sigma_{1}} \rho_{13} \pi_{3}^{s},$$

$$\pi_{2}^{s} = \frac{\lambda_{g}}{\sqrt{1 - \rho_{12}^{2}(1 - \alpha_{g})}},$$

$$\pi_{3}^{s} = \frac{\lambda_{b}}{\sqrt{1 - \rho_{13}^{2}(1 - \alpha_{b})}}.$$
(26)

Let us denote the corresponding value function as $J^{(s,UUI)}$.

2. The second case of interest is a sophisticated investor who is ambiguity-averse and capable of accommodating different levels of ambiguity aversion per source of risk (e.g., Branger and Larsen 2013; Wang and Uppal 2002). We denoted it as 'SAI'. Nonetheless, the investor does not know about multi-attribute utilities or the possibility of further accommodating preferences for green and brown sources of risk. Therefore, this investor considers $\alpha_g = \alpha_b = \alpha_m = \alpha$, with the corresponding suboptimal strategy:

$$\pi_{1}^{s} = \frac{\lambda_{1}}{1 - \alpha + \phi_{m}} - \frac{\sigma_{2}}{\sigma_{1}} \rho_{12} \pi_{2}^{s} - \frac{\sigma_{3}}{\sigma_{1}} \rho_{13} \pi_{3}^{s},$$

$$\pi_{2}^{s} = \frac{\lambda_{g}}{\sqrt{1 - \rho_{12}^{2}} (1 - \alpha + \phi_{g})},$$

$$\pi_{3}^{s} = \frac{\lambda_{b}}{\sqrt{1 - \rho_{13}^{2}} (1 - \alpha + \phi_{b})}.$$
(27)

The corresponding value function is denoted $J^{(s,SAI)}$. Needless to say, the suboptimal strategy described above is optimal for non-multi-attribute CRRA utility.

3. Thirdly, we explore a scenario in which an investor excludes the brown asset from its portfolio. This type of investor actively avoids exposure to the brown risk and focuses solely on investing in the market index and the green asset. However, they rigorously address the robust portfolio optimization problem in line with their specific risk preferences and ambiguity aversions. Given $\pi_3 = 0$, the suboptimal strategy is

$$\pi_{1}^{s} = \frac{\lambda_{1}}{1 - \alpha_{m} + \phi_{m}} - \frac{\sigma_{2}}{\sigma_{1}} \rho_{12} \pi_{2}^{*},$$

$$\pi_{2}^{s} = \frac{\lambda_{g}}{\sqrt{1 - \rho_{12}^{2}} (1 - \alpha_{g} + \phi_{g})},$$

$$\pi_{3}^{s} = 0.$$
(28)

We categorize this type of investor as a 'selectively brown-avoidant investor' and denote the associated value function as $J^{(s,SBI)}$.

Next, we will introduce two measures to assess the losses that a suboptimal investor could suffer, either in satisfaction or money, due to acting suboptimally.

4.1. Percentage Loss in Satisfaction

We first create a measure to compute the loss to the investor by working directly with the reduction in the level of satisfaction, as measured by the value function.

This is, assuming a value function J^s from using a suboptimal strategy and the value function from the optimal strategy as J, we define the percentage loss in satisfaction (PLS) as follows:

$$R = \frac{J^s - J}{J^s}.$$
 (29)

Note that *R* reaches its minimum value, R = 0, only when the suboptimal strategy is optimal. As the suboptimal strategy underperforms, resulting in a decrease in J^s , the value of *R* approaches 100% (indicating a complete loss). Therefore, we can define *R* within the range of [0, 1), where smaller values of *R* signify poorer performance of the suboptimal strategy when compared to the optimal strategy.

The fact that J and J^s are all closed-form for the cases described in the previous section indicates that this percentage can be easily computed in the closed form:

$$R = 1 - \exp\{(b - b^s)T\}.$$
(30)

4.2. Green Wealth Equivalent Loss

We introduce the concept of Green-Index Wealth Equivalent Loss (GWEL) and explore the impact of suboptimal strategies. We define GWEL as a scalar q that satisfies the following equation:

$$J(X_m, X_g(1-q), X_b, 0) = J^s(X_m, X_g, X_b, 0).$$
(31)

Notably, GWEL closely resembles the conventional definition of WEL. Specifically, WEL can be defined as the value of q for which $J(W(1 - q), 0) = J^s(W, 0)$. In our context, parameter q signifies the percentage-wise reduction in the Green Index's value that the optimal investor can tolerate while maintaining the same level of satisfaction as the suboptimal investor. This interpretation directly quantifies the degree to which the optimal portfolio deviates from a 'green' allocation to emulate the suboptimal choices made by investors.

As before, given that J and J^s are closed-form for the cases described in the previous section, the GWEL is given in closed form as follows:

$$q = 1 - \exp\left\{\frac{(b^s - b)T}{\alpha_g}\right\}.$$
(32)

5. Empirical Analyses

For empirical analysis, we used ESG ratings from the RepRisk database. This involved converting RepRisk Rating (RRR) scores into integers ranging from 1 (D) to 10 (AAA). We calculated the average RRR score to evaluate a company's ESG performance from 2010 to 2020. Without loss of generality, we classified the top 10 U.S. companies with the highest average RRR scores as 'green companies' and identified the bottom 10 companies with the lowest scores as 'brown companies'.

For benchmark purposes, we also determined the average RRR score for the entire U.S. market portfolio. This was achieved by computing the mean of the average RRR scores across all U.S. companies, resulting in an average RRR score of 7.3 for the entire U.S. market portfolio.

In our study, we present two illustrative examples, both employing the S&P 500 as our chosen index. The first example serves as the basis for the empirical analysis in this section, while the second is included in Appendix B as supplementary material. Considering data availability and consistency, we select two pairs of stocks for our analysis. The first pair consists of IDT Corp. (IDT), Newark, U.S., representing the green company, and Walmart Inc. (WMT), Bentonville, United States, representing the brown company. The second pair comprises Shenandoah Telecommunications Company (Shenandoah), Edinburg, United States as the green company and DuPont de Nemours, Inc. (DuPont), Wilmington, USA as the brown company.

We utilized Python for data retrieval, data processing, parameter estimations, and graph plotting. Table 1 provides a comprehensive overview of the parameters employed in our model for the two cases. These parameters were estimated on a monthly basis, using data from 2010 to 2020. To calculate the risk-free rate, we took the average of the monthly yields of the 3-month Treasury bills issued by the U.S. government during this time frame, resulting in a rate of 0.045%.

Name	σ	ρ	λ	Average RRR Score
S&P 500	0.0405	1	6.0464	7.3
IDT	0.1628	0.2937	0.7	9.4
Walmart	0.0486	0.3354	2.8672	3.4
Shenandoah Telecom	0.1064	0.291	1.0179	9.4
DuPont de Nemours	0.0866	0.767	-1.244	4.2

Table 1. Parameter estimations for empirical analysis.³

For consistency, our empirical analysis adopts a uniform time horizon of 10 years. Furthermore, we have set the risk aversion parameters at the following values: $\alpha_m = -4$, $\alpha_g = -2$, and $\alpha_b = -6$.

5.1. Optimal Weights

We examined the impact of ambiguity aversion parameters on the optimal weights and compared these robust optimal weights with the optimal weights without uncertainty consideration. Figure 1 depicts the drastic changes in portfolio allocation when we account for ambiguity aversion. In all three sub-figures, the weights of all stocks are significantly reduced, indicating increased investments in the risk-free asset. This result is not surprising, as investors would turn to safer assets when they begin considering the worst case of their investments. We can also observe that as one of the three ambiguity aversion parameters increases (green, for example), the weight of green stock also decreases. It is natural for investors to invest less in an asset if they do not trust the probabilistic model of that asset. However, market weight increased as green or brown ambiguity aversion increased. This is because the total weight of the three stocks is less sensitive to green and brown ambiguity aversion than the weight of green and brown stocks. As a result, the market weight increases to compensate for the decrease in green or brown investments. In other words, the total weight is mostly influenced by market ambiguity aversion, and the green or brown weight is influenced by green or brown ambiguity aversion.



Figure 1. Impact of ambiguity aversion parameters on optimal weights in the IDT and WMT case.⁴

5.2. Detection Error Probabilities

When investors discriminate between the reference and alternative models, two types of errors are possible for a sample of length *N*: choosing the alternative model when the

reference model is correct, with probability p_1^N , and choosing the reference model when the alternative model is correct, with probability p_2^N . We weight these errors by assigning prior probabilities of one-half to each model and denote the detection error probability as ε_N :

$$\varepsilon_N(\phi_m, \phi_g, \phi_b) = 0.5p_1^N + 0.5p_2^N.$$
(33)

Following the notation in Equation (2), the log-likelihood ratio is formed as $\ell^N = \log \Lambda_N^e$ with *e* replaced by e^* , that is,

$$\ell^N = -\int_0^N \left(\langle \boldsymbol{e}_\tau, d\boldsymbol{z}_\tau \rangle + \frac{1}{2} ||\boldsymbol{e}_\tau||^2 d\tau \right),\tag{34}$$

where e^* are obtained by combining Equations (A1), (16), and (17):

$$(e^{m})^{*} = \frac{\lambda_{1}\sigma_{1}\phi_{m}}{1-\alpha_{m}+\phi_{m}}$$

$$(e^{g})^{*} = \frac{\lambda_{g}\sigma_{2}\phi_{g}}{1-\alpha_{g}+\phi_{g}}$$

$$(e^{b})^{*} = \frac{\lambda_{b}\sigma_{3}\phi_{b}}{1-\alpha_{b}+\phi_{b}}.$$
(35)

Then, we have $p_1^N = \Pr(\ell^N > 0 | \mathbb{P}, \mathcal{F}_0)$ and $p_2^N = \Pr(\ell^N < 0 | \mathbb{P}^e, \mathcal{F}_0)$. We observe that ℓ^N follows a normal distribution with a mean of $-\frac{1}{2}||e^*||^2N$ and a variance of $||e^*||^2N$. Thus, we have

$$\varepsilon_N(\phi_m, \phi_g, \phi_b) = \Pr(\Phi > \frac{1}{2} ||\boldsymbol{e}^*||\sqrt{N})$$
(36)

where Φ follows a standard normal distribution. In the IDT and Walmart case, for the parameters in Table 1, N = 120, and specifying $\alpha_m = -4$, $\alpha_g = -2$, $\alpha_b = -6$, we have Figures 2–4. These three figures depict the relationship between the detection error probability and one of the ambiguity aversion parameters, while the other two parameters are set to be 0, 5, 10. In alignment with our speculation, as ambiguity aversion parameters increase, investors can better discern between reference and alternative models. Consequently, the probability of making a mistake decreases.



Figure 2. Detection error probability when ϕ_m , ϕ_g , ϕ_b changes in the IDT and WMT case. The dashed line is for $\phi_g = 0$, the dotted line is for $\phi_g = 5$, the solid line is for $\phi_g = 10$.



Figure 3. Detection error probability when ϕ_m , ϕ_g , ϕ_b changes in the IDT and WMT case. The dashed line is for $\phi_m = 0$, the dotted line is for $\phi_m = 5$, the solid line is for $\phi_m = 10$.



Figure 4. Detection error probability when ϕ_m , ϕ_g , ϕ_b changes in the IDT and WMT case. The dashed line is for $\phi_m = 0$, the dotted line is for $\phi_m = 5$, the solid line is for $\phi_m = 10$.

We can readily observe that ϕ_m exerts the most substantial influence on the detection error probability. A comparative analysis of Figures 2–4 reveals that an increase in ϕ_m has the greatest impact on the detection error probability. This is evident as Figure 2 exhibits a more pronounced curvature than Figures 3 and 4. Moreover, the detection error probability reaches its peak when $\phi_m = 0$ in all three figures.

Taking a closer look at the green curves in Figure 2c ($\phi_g = 10, \phi_b = 10$), Figure 3c ($\phi_m = 10, \phi_b = 10$), and Figure 4c ($\phi_m = 10, \phi_g = 10$), we can discern noteworthy distinctions. Specifically, the green curve in Figure 2c significantly deviates from its counterparts, resulting in a notably higher detection error probability when $\phi_m = 0, \phi_g = 10, \phi_b = 10$, approaching approximately 0.25. This is in stark contrast to the other two figures, where the detection error probability is approximately 0.15. Consequently, we conclude that green and brown stocks introduce more significant uncertainty. This is substantiated by the observation that when the investor possesses certainty about the market index ($\phi_m = 0$),

the detection error probability is higher than in scenarios where the investor has certainty about the green stock ($\phi_g = 0$) or the brown stock ($\phi_b = 0$).

In general, an increase in ambiguity aversion levels leads to a reduction in the detection error probability, enabling investors to better differentiate between reference and alternative models. This heightened discernment translates into a lower likelihood of making errors in model selection, thereby minimizing potential losses. The ensuing Section 5.3 delves into a detailed analysis of these diminished losses. Nonetheless, our investigation underscores that the uncertainty introduced by green and brown stocks remains resilient, even with heightened ambiguity aversion. Illustrated in Figure 2, when fixing $\phi_m = 0$, investors face approximately a 30% probability of making errors in discerning between reference and alternative models, irrespective of their ambiguity aversions towards green and brown stocks. Notably, the ambiguity aversion towards the market plays the most pivotal role in this discrimination between reference and alternative models.

Following the convention in Anderson et al. (2003), we are concerned with detection error probabilities larger than 0.1, in which case the investor has difficulty discerning the alternative model and the reference model. This explains why we chose ambiguity aversion parameters less than 10 in the previous section. In a later analysis, we will also confine the ambiguity aversion parameters to be less than 14 to ensure that the error detection probability is larger than 0.1 in all cases.

5.3. Suboptimal Loss Analysis

In this section, we present a comprehensive analysis of the percentage loss or reduction in satisfaction (PLS) and GWEL across the UUI, SAI, and SBI scenarios, maintaining a consistent temporal horizon of T = 120 months.

First, let us focus on the UUI case illustrated in Figure 5, which shows the percentage reduction in satisfaction. As expected, UUI investors witness a pronounced decline in satisfaction as ambiguity aversion parameters escalate, aligning with the relation

$$b - b^{s} = \frac{1}{2} \lambda_{1}^{2} \sigma_{1}^{2} \frac{\alpha_{m}}{(1 - \alpha_{m})^{2}} \frac{\phi_{m}^{2}}{1 - \alpha_{m} + \phi_{m}} + \frac{1}{2} \lambda_{g}^{2} \sigma_{2}^{2} \frac{\alpha_{g}}{(1 - \alpha_{g})^{2}} \frac{\phi_{g}^{2}}{1 - \alpha_{g} + \phi_{g}} + \frac{1}{2} \lambda_{b}^{2} \sigma_{3}^{2} \frac{\alpha_{b}}{(1 - \alpha_{b})^{2}} \frac{\phi_{b}^{2}}{1 - \alpha_{b} + \phi_{b}}.$$
(37)



Figure 5. PLS for UUI when ϕ_m , ϕ_g , ϕ_b changes in the IDT and WMT case. The dashed line is for $\phi_g = 0$, the dotted line is for $\phi_g = 5$, the solid line is for $\phi_g = 10$.

The heightened PLS can be attributed to any of the three ambiguity aversion parameters. Notably, ϕ_m exerts the most significant influence. As clearly depicted in the figures, when $\phi_m = 10$, there is a nearly 100% higher loss in satisfaction compared to $\phi_m = 0$, while maintaining ϕ_g and ϕ_b at 0. However, the influence of ϕ_m diminishes as the other two parameters increase. For instance, when $\phi_g = 10$ and $\phi_b = 10$, a change in ϕ_m only results in a 10% variation in the PLS. The loss in satisfaction reaches its zenith, nearly touching the 100% mark, when all three parameters are set to 10. It is also evident that PLS is more significantly impacted when any parameter increases from 0 to 5 compared to when it increases from 5 to 10. Beyond a value of 10, the effect becomes negligible.

Turning our attention to Figure 6, we investigate the variations in GWEL as ambiguity aversion parameters fluctuate over a ten-year duration in the UUI context. Notably, when all three ambiguity aversion parameters are set to 10, a staggering 95% loss is observed in the green index. This significant loss emerges because of the simultaneous increments in all three ambiguity aversion parameters. However, it is crucial to highlight that the escalation of GWEL is also more pronounced as ambiguity aversion parameters rise from 5 to 10 compared to the increase from 0 to 5. This underscores the heightened sensitivity of GWEL to investors with higher ambiguity aversion parameters. These findings are consistent with our model, emphasizing the need to incorporate uncertainty into portfolio optimization.



Figure 6. GWEL when ϕ_m , ϕ_g , ϕ_b changes for UUI in the IDT and WMT case. The dashed line is for $\phi_g = 0$, the dotted line is for $\phi_g = 5$, the solid line is for $\phi_g = 10$.

Now, transitioning to the SAI case, we can discern from Figures 7 and 8 that both the percentage reduction in satisfaction and GWEL remain impervious to changes in the market ambiguity aversion parameter ϕ_m . However, an increase in either ϕ_g or ϕ_b leads to a decrease in both the percentage reduction in satisfaction and GWEL. This outcome is markedly different from the results of the UUI scenario. Notably, the solid green lines in Figures 7 and 8 occupy lower positions, in contrast to the dashed blue lines in Figures 5 and 6. This result is not surprising, as in the case of SAI,

$$b - b^{s} = \frac{1}{2}\lambda_{g}^{2}\sigma_{2}^{2}\frac{\alpha_{g}(\alpha_{m} - \alpha_{g})}{(1 - \alpha_{m} + \phi_{g})^{2}} + \frac{1}{2}\lambda_{b}^{2}\sigma_{3}^{2}\frac{\alpha_{b}(\alpha_{m} - \alpha_{b})}{(1 - \alpha_{m} + \phi_{b})^{2}},$$
(38)

independent of ϕ_m . Moreover, it is worth emphasizing that the overall reduction in satisfaction is substantially smaller than that in the UUI case. The maximum loss is 22% when $\phi_g = \phi_b = 0$.



Figure 7. PLS when ϕ_m , ϕ_g , ϕ_b changes for SAI in the IDT and WMT case. The dashed line is for $\phi_g = 0$, the dotted line is for $\phi_g = 5$, the solid line is for $\phi_g = 10$.



Figure 8. GWEL when ϕ_m , ϕ_g , ϕ_b changes for SAI in the IDT and WMT case. The dashed line is for $\phi_g = 0$, the dotted line is for $\phi_g = 5$, the solid line is for $\phi_g = 10$.

Finally, when examining the SBI scenario, we find that both PLS and GWEL are exclusively influenced by ϕ_b , as shown in Figures 9 and 10. This result arises from the fact that

$$b - b^{s} = \frac{1}{2}\lambda_{b}^{2}\sigma_{3}^{2}\frac{\alpha_{b}}{1 - \alpha_{b} + \phi_{b}}.$$
(39)

It is worth noting that augmenting ϕ_b decreases both PLS and GWEL, resulting in a 30% reduction in PLS and a 20% reduction in GWEL when ϕ_b increases from 0 to 10. Furthermore, the SBI scenario occupies an intermediary position between UUI and SAI in terms of loss. The maximum loss is 62% when $\phi_b = 0$.



Figure 9. PLS when ϕ_m , ϕ_g , ϕ_b changes for SBI in the IDT and WMT case. The solid line is for $\phi_g = 0$, $\phi_g = 5$, and also for $\phi_g = 10$.



Figure 10. GWEL when ϕ_m , ϕ_g , ϕ_b changes for SBI in the IDT and WMT case. The solid line is for $\phi_g = 0$, $\phi_g = 5$, and also for $\phi_g = 10$.

These analyses of UUI, SAI, and SBI investors underscore the pivotal role of ambiguity aversion in financial decision-making. It emphasizes the necessity for investors to transparently communicate their risk preferences and ambiguity aversions to avoid selecting suboptimal models that could result in substantial losses. Investors characterized by higher ambiguity aversion levels should be particularly vigilant about their portfolio positions, given the potential for greater losses when deviating from optimal choices. This suggests a need for more frequent portfolio rebalancing. For asset managers, aligning their models with clients' ambiguity aversion becomes critical. Effective communication and a comprehensive approach to measuring ambiguity aversion are imperative for the success of asset managers aiming for ESG objectives. Despite the growing significance of ESG considerations, these findings emphasize the importance of ESG education for investors and the necessity of fostering an open and efficient environment for communication regarding ESG investing.

6. Conclusions

This study bridges the gap between the well-established concept of ambiguity aversion in portfolio choice and the increasingly relevant field of environmental, social, and corporate governance (ESG) investing. It extends the existing literature by introducing ESG ambiguity aversions into the portfolio choice problem within a multivariate utility framework. We address the ambiguity aversion of an investor who can allocate resources to a risk-free asset, a market index, a green stock, and a brown stock. The robust control approach employed, which is based on relative entropy, allows us to derive optimal investment strategies and investigate the effects of ambiguity in the ESG context.

This study makes several significant contributions. First, it demonstrates the substantial impact of ambiguity on optimal trading strategies, highlighting the differential effects on market, green, and brown assets. Plausible levels of ambiguity significantly alter optimal strategies, with potential changes exceeding 50%. Second, the study provides insights into the comparative behavior of ambiguity-averse and non-ambiguity investors. Ambiguity-averse investors exhibit less aggressive risk exposure and reduced short-selling positions. Finally, the paper quantifies the welfare loss incurred by investors who neglect model uncertainty, revealing substantial financial costs amounting to nearly 100% for empirically relevant parameter values. Our findings about the impact of uncertainty are corroborated by two measures of performance (i.e., PLS and GWEL) and two case studies, each involving a green, brown, and market index. Moreover, this research highlights the importance of transparent communication regarding risk preferences and ambiguity aversions for both investors and asset managers. Advocating ESG education can enhance the effectiveness of this communication, thereby reducing the risk of substantial losses resulting from suboptimal strategies.

There are avenues for further exploration in our study. For instance, the inclusion of jumps, stochastic volatility, or stochastic correlation in our model could better capture extreme shifts in asset prices, particularly post-2020.

In summary, this study enriches the understanding of ambiguity aversion in the context of ESG-integrated portfolio optimization by emphasizing the nuanced impact of ESG ambiguity aversion and its potential welfare implications. This conveys valuable insights to investors, asset managers, and policymakers seeking to navigate the evolving landscape of responsible and sustainable investing. The integration of ESG criteria into the analysis of ambiguity aversion represents an original and vital contribution to the ongoing discourse in the field of finance.

Author Contributions: Methodology, M.E.-A. and Y.J.; Software, Y.J.; Validation, M.E.-A. and Y.J.; Formal analysis, M.E.-A. and Y.J.; Investigation, M.E.-A. and Y.J.; Data curation, M.E.-A. and Y.J.; Writing original draft, M.E.-A. and Y.J.; Writing review and editing, M.E.-A. and Y.J.; Visualization, Y.J. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Data Availability Statement: The ESG ratings of the companies used in our empirical analysis were obtained from the RepRisk ESG Dataset. For additional information, please refer to the website: https://www.reprisk.com/ (accessed on 12 April 2023).

Conflicts of Interest: The authors declare no conflicts of interest.

Appendix A. Proofs

Appendix A.1. Proof of Proposition 1

Solving the infimization problem in Equation (15), we obtain

$$(e^{m})^{*} = \Psi_{m} x J_{x} (\pi_{1}\sigma_{1} + \pi_{2}\sigma_{2}\rho_{12} + \pi_{3}\sigma_{3}\rho_{13})$$

$$(e^{g})^{*} = \Psi_{g} y J_{y} \pi_{2}\sigma_{2} \sqrt{1 - \rho_{12}^{2}}$$

$$(e^{b})^{*} = \Psi_{z} z J_{z} \pi_{3}\sigma_{3} \sqrt{1 - \rho_{13}^{2}}.$$
(A1)

Substituting Equation (A1) into the HJB Equation (15), J satisfies

$$\begin{split} \sup_{\pi_{1},\pi_{2},\pi_{3}} \left\{ J_{t} + (\theta_{m}r + \pi_{1}\lambda_{1}\sigma_{1}^{2} + \pi_{2}\lambda_{1}\sigma_{1}\sigma_{2}\rho_{12} + \pi_{3}\lambda_{1}\sigma_{1}\sigma_{3}\rho_{13})xJ_{x} \\ &+ \frac{1}{2}(\pi_{1}\sigma_{1} + \pi_{2}\sigma_{2}\rho_{12} + \pi_{3}\sigma_{3}\rho_{13})^{2}x^{2}J_{xx} + (\theta_{g}r + \pi_{2}\lambda_{g}\sigma_{2}^{2}\sqrt{1 - \rho_{12}^{2}})yJ_{y} \\ &+ \frac{1}{2}\pi_{2}^{2}\sigma_{2}^{2}(1 - \rho_{12}^{2})y^{2}J_{yy} + (\theta_{b}r + \pi_{3}\lambda_{b}\sigma_{3}^{2}\sqrt{1 - \rho_{13}^{2}})zJ_{z} + \frac{1}{2}\pi_{3}^{2}\sigma_{3}^{2}(1 - \rho_{13}^{2})z^{2}J_{zz} \\ &- \frac{\Psi_{m}}{2}(\pi_{1}\sigma_{1} + \pi_{2}\sigma_{2}\rho_{12} + \pi_{3}\sigma_{3}\rho_{13})^{2}x^{2}(J_{x})^{2} - \frac{\Psi_{g}}{2}\pi_{2}^{2}\sigma_{2}^{2}(1 - \rho_{12}^{2})y^{2}(J_{y})^{2} \\ &- \frac{\Psi_{b}}{2}\pi_{3}^{2}\sigma_{3}^{2}(1 - \rho_{13}^{2})z^{2}(J_{z})^{2} \right\} = 0. \end{split}$$
(A2)

Solving for π_1, π_2, π_3 in the above equation, the optimal weights have the form

$$\pi_{1}^{*} = \frac{\lambda_{1}J_{x}}{\Psi_{m}x(J_{x})^{2} - xJ_{xx}} - \frac{\sigma_{2}}{\sigma_{1}}\rho_{12}\pi_{2}^{*} - \frac{\sigma_{3}}{\sigma_{1}}\rho_{13}\pi_{3}^{*},$$

$$\pi_{2}^{*} = \frac{\lambda_{g}}{\sqrt{1 - \rho_{12}^{2}}} \frac{J_{y}}{\Psi_{g}y(J_{y})^{2} - yJ_{yy}},$$
(A3)
$$\pi_{3}^{*} = \frac{\lambda_{b}}{\sqrt{1 - \rho_{13}^{2}}} \frac{J_{z}}{\Psi_{b}z(J_{z})^{2} - zJ_{zz}}.$$

Given the linearity of the HJB, we assume that *J* has the form of $J = \frac{x^{\alpha_m}}{\alpha_m} \frac{y^{\alpha_g}}{\alpha_g} \frac{z^{\alpha_b}}{\alpha_b} \exp(b(T - t))$, together with Equation (14), we obtain

$$0 = \sup_{\pi_{1},\pi_{2},\pi_{3}} \left\{ -b + (\theta_{m}r + \pi_{1}\lambda_{1}\sigma_{1}^{2} + \pi_{2}\lambda_{1}\sigma_{1}\sigma_{2}\rho_{12} + \pi_{3}\lambda_{1}\sigma_{1}\sigma_{3}\rho_{13})\alpha_{m} + \frac{1}{2}(\pi_{1}\sigma_{1} + \pi_{2}\sigma_{2}\rho_{12} + \pi_{3}\sigma_{3}\rho_{13})^{2}\alpha_{m}(\alpha_{m} - 1) + (\theta_{g}r + \pi_{2}\lambda_{g}\sigma_{2}^{2}\sqrt{1 - \rho_{12}^{2}})\alpha_{g} + \frac{1}{2}\pi_{2}^{2}\sigma_{2}^{2}(1 - \rho_{12}^{2})\alpha_{g}(\alpha_{g} - 1) + (\theta_{b}r + \pi_{3}\lambda_{b}\sigma_{3}^{2}\sqrt{1 - \rho_{13}^{2}})\alpha_{b} + \frac{1}{2}\pi_{3}^{2}\sigma_{3}^{2}(1 - \rho_{13}^{2})\alpha_{b}(\alpha_{b} - 1) - \frac{\phi_{m}}{2}(\pi_{1}\sigma_{1} + \pi_{2}\sigma_{2}\rho_{12} + \pi_{3}\sigma_{3}\rho_{13})^{2}\alpha_{m} - \frac{\phi_{g}}{2}\pi_{2}^{2}\sigma_{2}^{2}(1 - \rho_{12}^{2})\alpha_{g} - \frac{\phi_{b}}{2}\pi_{3}^{2}\sigma_{3}^{2}(1 - \rho_{13}^{2})\alpha_{b} \right\},$$
(A4)

Again, solving for π_1, π_2, π_3 in the above equation, we obtain the optimal weights

$$\pi_{1}^{*} = \frac{\lambda_{1}}{1 - \alpha_{m} + \phi_{m}} - \frac{\sigma_{2}}{\sigma_{1}} \rho_{12} \pi_{2}^{*} - \frac{\sigma_{3}}{\sigma_{1}} \rho_{13} \pi_{3}^{*},$$

$$\pi_{2}^{*} = \frac{\lambda_{g}}{\sqrt{1 - \rho_{12}^{2}} (1 - \alpha_{g} + \phi_{g})},$$

$$\pi_{3}^{*} = \frac{\lambda_{b}}{\sqrt{1 - \rho_{13}^{2}} (1 - \alpha_{b} + \phi_{b})}.$$
(A5)

Substituting the optimal weights back into Equation (A4), we obtain

$$J(X,t) = \frac{X_m^{\alpha_m}}{\alpha_m} \frac{X_g^{\alpha_g}}{\alpha_g} \frac{X_b^{\alpha_b}}{\alpha_b} \exp(b(T-t)),$$
(A6)

where

$$b = \frac{1}{2}\lambda_1^2 \sigma_1^2 \frac{\alpha_m}{1 - \alpha_m + \phi_m} + \frac{1}{2}\lambda_g^2 \sigma_2^2 \frac{\alpha_g}{1 - \alpha_g + \phi_g} + \frac{1}{2}\lambda_b^2 \sigma_3^2 \frac{\alpha_b}{1 - \alpha_b + \phi_b} + (\theta_m \alpha_m + \theta_g \alpha_g + \theta_b \alpha_b)r.$$
(A7)

In addition, substituting the optimal weights back into Equation (5), the optimal wealth process is given as

$$\frac{dW_t}{W_t} = \left(r + \frac{\lambda_1^2 \sigma_1^2}{1 - \alpha_m + \phi_m} + \frac{\lambda_g^2 \sigma_2^2}{1 - \alpha_g + \phi_g} + \frac{\lambda_b^2 \sigma_3^2}{1 - \alpha_b + \phi_b}\right) dt + \frac{\lambda_1 \sigma_1}{1 - \alpha_m + \phi_m} dz_m + \frac{\lambda_g \sigma_2}{1 - \alpha_g + \phi_g} dz_g + \frac{\lambda_b \sigma_3}{1 - \alpha_b + \phi_b} dz_b.$$
(A8)

Appendix A.2. Proof of Proposition 2

The infimization problem for the suboptimal value function J^s is

$$J^{s}(X,t) = \inf_{e \in \varepsilon[t,T]} \left\{ \mathbf{E}_{t}^{\mathbb{P}^{e}} \left[\int_{t}^{T} \left(\frac{(e_{\tau}^{m})^{2}}{2\Psi_{m}(\tau, X_{\tau})} + \frac{(e_{\tau}^{g})^{2}}{2\Psi_{g}(\tau, X_{\tau})} + \frac{(e_{\tau}^{b})^{2}}{2\Psi_{b}(\tau, X_{\tau})} \right) d\tau \right] + w^{e}(X,t;\boldsymbol{\pi}^{s}) \right\}.$$
(A9)

 J^s satisfies the HJB equation

$$\begin{split} &\inf_{e^{m},e^{g},e^{b}} \left\{ J_{t}^{s} + (\theta_{m}r + \pi_{1}^{s}\lambda_{1}\sigma_{1}^{2} + \pi_{2}^{s}\lambda_{1}\sigma_{1}\sigma_{2}\rho_{12} + \pi_{3}^{s}\lambda_{1}\sigma_{1}\sigma_{3}\rho_{13} \\ &- (\pi_{1}^{s}\sigma_{1} + \pi_{2}^{s}\sigma_{2}\rho_{12} + \pi_{3}^{s}\sigma_{3}\rho_{13})e^{m})xJ_{x}^{s} + \frac{1}{2}(\pi_{1}^{s}\sigma_{1} + \pi_{2}^{s}\sigma_{2}\rho_{12} + \pi_{3}^{s}\sigma_{3}\rho_{13})^{2}x^{2}J_{xx}^{s} \\ &+ (\theta_{g}r + \pi_{2}^{s}\lambda_{g}\sigma_{2}^{2}\sqrt{1 - \rho_{12}^{2}} - \pi_{2}^{s}\sigma_{2}\sqrt{1 - \rho_{12}^{2}}e^{g})yJ_{y}^{s} + \frac{1}{2}(\pi_{2}^{s})^{2}\sigma_{2}^{2}(1 - \rho_{12}^{2})y^{2}J_{yy}^{s} \qquad (A10) \\ &+ (\theta_{b}r + \pi_{3}^{s}\lambda_{b}\sigma_{3}^{2}\sqrt{1 - \rho_{13}^{2}} - \pi_{3}^{s}\sigma_{3}\sqrt{1 - \rho_{13}^{2}}e^{b})zJ_{z}^{s} + \frac{1}{2}(\pi_{3}^{s})^{2}\sigma_{3}^{2}(1 - \rho_{13}^{2})z^{2}J_{zz}^{s} \\ &+ \frac{(e^{m})^{2}}{2\Psi_{m}} + \frac{(e^{g})^{2}}{2\Psi_{g}} + \frac{(e^{b})^{2}}{2\Psi_{b}} \right\} = 0. \end{split}$$

Assuming a constant π^s , and $J^s = \frac{x^{\alpha_m}}{\alpha_m} \frac{y^{\alpha_g}}{\alpha_g} \frac{z^{\alpha_b}}{\alpha_b} \exp(b^s(T-t))$, together with Equation (14), we obtain

$$\begin{split} b^{s} &= (\theta_{m}r + \pi_{1}^{s}\lambda_{1}\sigma_{1}^{2} + \pi_{2}^{s}\lambda_{1}\sigma_{1}\sigma_{2}\rho_{12} + \pi_{3}^{s}\lambda_{1}\sigma_{1}\sigma_{3}\rho_{13})\alpha_{m} \\ &+ \frac{1}{2}(\pi_{1}^{s}\sigma_{1} + \pi_{2}^{s}\sigma_{2}\rho_{12} + \pi_{3}^{s}\sigma_{3}\rho_{13})^{2}\alpha_{m}(\alpha_{m} - 1) + (\theta_{g}r + \pi_{2}^{s}\lambda_{g}\sigma_{2}^{2}\sqrt{1 - \rho_{12}^{2}})\alpha_{g} \\ &+ \frac{1}{2}(\pi_{2}^{s})^{2}\sigma_{2}^{2}(1 - \rho_{12}^{2})\alpha_{g}(\alpha_{g} - 1) + (\theta_{b}r + \pi_{3}^{s}\lambda_{b}\sigma_{3}^{2}\sqrt{1 - \rho_{13}^{2}})\alpha_{b} \\ &+ \frac{1}{2}(\pi_{3}^{s})^{2}\sigma_{3}^{2}(1 - \rho_{13}^{2})\alpha_{b}(\alpha_{b} - 1) - \frac{\phi_{m}}{2}(\pi_{1}^{s}\sigma_{1} + \pi_{2}^{s}\sigma_{2}\rho_{12} + \pi_{3}^{s}\sigma_{3}\rho_{13})^{2}\alpha_{m} \\ &- \frac{\phi_{g}}{2}(\pi_{2}^{s})^{2}\sigma_{2}^{2}(1 - \rho_{12}^{2})\alpha_{g} - \frac{\phi_{b}}{2}(\pi_{3}^{s})^{2}\sigma_{3}^{2}(1 - \rho_{13}^{2})\alpha_{b}. \end{split}$$
(A11)

Substituting $\pi^s = (\pi_1^s, \pi_2^s, \pi_3^s)$ back into Equation (5), the suboptimal wealth process W_t^s is given as

$$\frac{dW_t^s}{W_t^s} = (r + \pi_1^s \lambda_1 \sigma_1^2 + \pi_2^s \lambda_1 \sigma_1 \sigma_2 \rho_{12} + \pi_2^s \lambda_g \sigma_2^2 \sqrt{1 - \rho_{12}^2} + \pi_3^s \lambda_1 \sigma_1 \sigma_3 \rho_{13} \\
+ \pi_3^s \lambda_b \sigma_3^2 \sqrt{1 - \rho_{13}^2}) dt + (\pi_1^s \sigma_1 + \pi_2^s \sigma_2 \rho_{12} + \pi_3^s \sigma_3 \rho_{13}) dz_m + \pi_2^s \sigma_2 \sqrt{1 - \rho_{12}^2} dz_g \qquad (A12) \\
+ \pi_3^s \sigma_3 \sqrt{1 - \rho_{13}^2} dz_b.$$

Appendix B. Empirical Analysis for the Shenandoah and DuPont Case

Appendix B.1. Optimal Weights

Compared with the IDT and WMT case, the Shenandoah and DuPont case shows similar behavior but not the same results. This is because DuPont has a negative price for the risk premium and is short-sold. Nevertheless, the reasoning is the same because the increased brown ambiguity aversion would reduce investors' risk exposure to the brown stock and, in this case, decrease the short-selling position in their portfolio. Similar to the IDT and WMT case, the decreased position is compensated by the market weight.



Figure A1. Impact of ambiguity aversion parameters on optimal weights in the Shenandoah and DuPont case.⁵

Appendix B.2. Detection Error Probabilities

Figures A2–A4 corroborate our speculation that as ambiguity aversion parameters increase, investors can better discern between the reference model and the alternative model. As a result, the probability of making a mistake decreases.

We also observe that ϕ_m exerts the most substantial influence on the detection error probability. Figure A2 shows that increases in ϕ_m have the greatest impact on the detection error probability, and Figure A2 exhibits a more pronounced curvature than Figures A3 and A4.



Figure A2. Detection error probability when ϕ_m , ϕ_g , ϕ_b changes in the Shenandoah and DuPont case. The dashed line is for $\phi_g = 0$, the dotted line is for $\phi_g = 5$, the solid line is for $\phi_g = 10$.



Figure A3. Detection error probability when ϕ_m , ϕ_g , ϕ_b changes in the Shenandoah and DuPont case. The dashed line is for $\phi_m = 0$, the dotted line is for $\phi_m = 5$, the solid line is for $\phi_m = 10$.



Figure A4. Detection error probability when ϕ_m , ϕ_g , ϕ_b changes in the Shenandoah and DuPont case. The dashed line is for $\phi_m = 0$, the dotted line is for $\phi_m = 5$, the solid line is for $\phi_m = 10$.

Appendix B.3. Suboptimal Losses Analysis

For the UUI case illustrated in Figure A5, UUI investors witness more pronounced declines in satisfaction as the ambiguity aversion parameters escalate. This heightened PLS can be attributed to any of the three ambiguity aversion parameters, with ϕ_m exerting the most substantial influence. Setting $\phi_m = 10$, $\phi_g = \phi_b = 0$ leads to a nearly 100% higher loss in satisfaction compared to when $\phi_m = \phi_g = \phi_b = 0$. Figure A6 shows a staggering 95% loss in the green index. Figure A6, while displaying a lower GWEL than Figure 6, reaffirms the heightened sensitivity of GWEL to investors with higher ambiguity aversion parameters.



Figure A5. PLS when ϕ_m , ϕ_g , ϕ_b changes for UUI in the Shenandoah and DuPont case. The dashed line is for $\phi_g = 0$, the dotted line is for $\phi_g = 5$, the solid line is for $\phi_g = 10$.



Figure A6. GWEL when ϕ_m , ϕ_g , ϕ_b changes for UUI in the Shenandoah and DuPont case. The dashed line is for $\phi_g = 0$, the dotted line is for $\phi_g = 5$, the solid line is for $\phi_g = 10$.

For the SAI case, Figures A7 and A8 show that ϕ_m does not affect PLS and GWEL. However, it is worth emphasizing that the overall satisfaction reduction is substantially smaller than that in the UUI case.



Figure A7. PLS when ϕ_m , ϕ_g , ϕ_b changes for SAI in the Shenandoah and DuPont case. The dashed line is for $\phi_g = 0$, the dotted line is for $\phi_g = 5$, the solid line is for $\phi_g = 10$.



Figure A8. GWEL when ϕ_m , ϕ_g , ϕ_b changes for SAI in the Shenandoah and DuPont case. The dashed line is for $\phi_g = 0$, the dotted line is for $\phi_g = 5$, the solid line is for $\phi_g = 10$.

When examining the SBI scenario, both PLS and GWEL are exclusively influenced by ϕ_b , as shown in Figures A9 and A10. The SBI scenario occupies an intermediary position between UUI and SAI in terms of loss.



Figure A9. PLS when ϕ_m , ϕ_g , ϕ_b changes for SBI in the Shenandoah and DuPont case. The solid line is for $\phi_g = 0$, $\phi_g = 5$, and also for $\phi_g = 10$.



Figure A10. GWEL when ϕ_m , ϕ_g , ϕ_b changes for SBI in the Shenandoah and DuPont case. The solid line is for $\phi_g = 0$, $\phi_g = 5$, and also for $\phi_g = 10$.

Notes

- ¹ Denoting the return of the wealth process as μ and the three volatilities as σ_i for i = 1, 2, 3, the proportions π_i , i = 1, 2, 3 must satisfy that μW_t is an \mathcal{L}^1 -process and that $\sigma_i W_t$ is an \mathcal{L}^2 -process for each i = 1, 2, 3.
- ² For simplicity we will take $\theta_m = 1$, $\theta_g = 0$, $\theta_b = 0$ in the numerical section; this has the interpretation of treating the cash account as part of the market portfolio, ideally these weights should be related to the ESG rating of the source of this cash.
- ³ Parameter estimations for the empirical analysis involved computations for the standard deviation (σ), correlation (ρ), and risk premium factor (λ). These estimations were derived from monthly data on the adjusted close price of each asset spanning the period from 2010 to 2020. Additionally, the average RepRisk Rating (RRR) scores were determined by calculating the mean of the monthly RRR scores over the same time frame.
- ⁴ In each figure, one ambiguity aversion changes, and the other two are set to 10. The robust optimal weights (solid curves) are compared with the optimal weights without uncertainty consideration (dashed curves).
- ⁵ In each figure, one parameter changes, and the other two are set to be equal to 10. The robust optimal weights are compared with the optimal weights without uncertainty consideration.

References

- Ahmed, Muhammad Farid, Yang Gao, and Stephen Satchell. 2021. Modeling demand for esg. *The European Journal of Finance* 27: 1669–83. [CrossRef]
- Anderson, Evan W., Lars Peter Hansen, and Thomas J. Sargent. 2003. A quartet of semigroups for model specification, robustness, prices of risk, and model detection. *Journal of the European Economic Association* 1: 68–123. [CrossRef]
- Bermejo Climent, Ramón, Isabel Figuerola-Ferretti Garrigues, Ioannis Paraskevopoulos, and Alvaro Santos. 2021. Esg disclosure and portfolio performance. *Risks* 9: 172. [CrossRef]
- Blanchard, Olivier J., Robert Shiller, and Jeremy J. Siegel. 1993. Movements in the equity premium. *Brookings Papers on Economic* Activity 1993: 75–138. [CrossRef]
- Branger, Nicole, and Linda Sandris Larsen. 2013. Robust portfolio choice with uncertainty about jump and diffusion risk. *Journal of Banking & Finance* 37: 5036–47.
- Chen, Li, Lipei Zhang, Jun Huang, Helu Xiao, and Zhongbao Zhou. 2021. Social responsibility portfolio optimization incorporating esg criteria. *Journal of Management Science and Engineering* 6: 75–85. [CrossRef]
- Clément, Alexandre, Élisabeth Robinot, and Léo Trespeuch. 2022. Improving esg scores with sustainability concepts. *Sustainability* 14: 13154. [CrossRef]
- Cochrane, John H. 1998. Where Is the Market Going? Uncertain Facts and Novel Theories. Available online: https://www.nber.org/papers/w6207 (accessed on 14 April 2023).
- Dimmock, Stephen G., Roy Kouwenberg, Olivia S. Mitchell, and Kim Peijnenburg. 2016. Ambiguity aversion and household portfolio choice puzzles: Empirical evidence. *Journal of Financial Economics* 119: 559–77. [CrossRef] [PubMed]
- Dorfleitner, Gregor, and Mai Nguyen. 2017. A new approach for optimizing responsible investments dependently on the initial wealth. *Journal of Asset Management* 18: 81–98. [CrossRef]

- Dorfleitner, Gregor, Michaela Leidl, and Johannes Reeder. 2012. Theory of social returns in portfolio choice with application to microfinance. *Journal of Asset Management* 13: 384–400. [CrossRef]
- Drempetic, Samuel, Christian Klein, and Bernhard Zwergel. 2020. The influence of firm size on the esg score: Corporate sustainability ratings under review. *Journal of Business Ethics* 167: 333–60. [CrossRef]
- Edmans, Alex. 2011. Does the stock market fully value intangibles? employee satisfaction and equity prices. *Journal of Financial Economics* 101: 621–40. [CrossRef]

Ellsberg, Daniel. 1961. Risk, ambiguity, and the savage axioms. The Quarterly Journal of Economics 75: 643-69. [CrossRef]

- Escobar, Marcos, Sebastian Ferrando, and Alexey Rubtsov. 2015. Robust portfolio choice with derivative trading under stochastic volatility. *Journal of Banking & Finance* 61: 142–57.
- Escobar-Anel, Marcos. 2022. Multivariate risk aversion utility, application to esg investments. *The North American Journal of Economics and Finance* 63: 101790. [CrossRef]
- Escobar-Anel, Marcos, and Yiyao Jiao. 2023. Unraveling the trade-off between sustainability and returns: A multivariate utility analysis. *arXiv* arXiv:2307.12161.
- Flor, Christian Riis, and Linda Sandris Larsen. 2014. Robust portfolio choice with stochastic interest rates. *Annals of Finance* 10: 243–65. [CrossRef]
- Gasser, Stephan M., Margarethe Rammerstorfer, and Karl Weinmayer. 2017. Markowitz revisited: Social portfolio engineering. *European Journal of Operational Research* 258: 1181–90. [CrossRef]
- Jacobsen, Brian, Wai Lee, and Chao Ma. 2019. The alpha, beta, and sigma of esg: Better beta, additional alpha? *The Journal of Portfolio Management* 45: 6–15. [CrossRef]
- Klingebiel, Ronald, and Feibai Zhu. 2023. Ambiguity aversion and the degree of ambiguity. *Journal of Risk and Uncertainty* 67: 299–324. [CrossRef]
- Kocher, Martin G., Amrei Marie Lahno, and Stefan T. Trautmann. 2018. Ambiguity aversion is not universal. *European Economic Review* 101: 268–83. [CrossRef]
- Liu, Hening. 2010. Robust consumption and portfolio choice for time varying investment opportunities. *Annals of Finance* 6: 435–54. [CrossRef]
- Maenhout, Pascal J. 2004. Robust portfolio rules and asset pricing. Review of Financial Studies 17: 951-83. [CrossRef]
- Maenhout, Pascal J. 2006. Robust portfolio rules and detection-error probabilities for a mean-reverting risk premium. *Journal of Economic Theory* 128: 136–63. [CrossRef]
- Merton, Robert C. 1975. Optimum consumption and portfolio rules in a continuous-time model. In *Stochastic Optimization Models in Finance*. Amsterdam: Elsevier, pp. 621–61.
- Merton, Robert C. 1980. On estimating the expected return on the market: An exploratory investigation. *Journal of Financial Economics* 8: 323–61. [CrossRef]
- Orsato, Renato J., Alexandre Garcia, Wesley Mendes-Da-Silva, Roberta Simonetti, and Mario Monzoni. 2015. Sustainability indexes: Why join in? A study of the 'corporate sustainability index (ise)' in brazil. *Journal of Cleaner Production* 96: 161–70. [CrossRef]
- Pástor, L'uboš, Robert F. Stambaugh, and Lucian A. Taylor. 2021. Sustainable investing in equilibrium. *Journal of Financial Economics* 142: 550–71. [CrossRef]
- Polbennikov, Simon, Albert Desclée, Lev Dynkin, and Anando Maitra. 2016. Esg ratings and performance of corporate bonds. *The Journal of Fixed Income* 26: 21. [CrossRef]
- Rubtsov, Alexey, and Sally Shen. 2022. Dynamic portfolio decisions with climate risk and model uncertainty. *Journal of Sustainable Finance & Investment* 1–22. [CrossRef]
- Rubtsov, Alexey, Wei Xu, Aleksandar Šević, and Željko Šević. 2021. Price of climate risk hedging under uncertainty. *Technological Forecasting and Social Change* 165: 120430. [CrossRef]
- Tarmuji, Indarawati, Ruhanita Maelah, and Nor Habibah Tarmuji. 2016. The impact of environmental, social and governance practices (esg) on economic performance: Evidence from esg score. *International Journal of Trade, Economics and Finance* 7: 67. [CrossRef]
 US SIF. 2022. 2022 Report on Us Sustainable Investing Trends. Washington, DC: US SIF Foundation Press Release.
- Wang, Tan, and Raman Uppal. 2002. Model misspecification and under-diversification. In *EFA 2002 Berlin Meetings Presented Paper*. Working Paper, SSRN 311426. Vancouver: Sauder School of Business.

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.





José Sequeira¹, Cláudia Pereira^{2,*}, Luís Gomes² and Armindo Lima²

- ¹ ISCAP, Polytechnic of Porto, 4465-004 Matosinhos, Portugal; 2180658@iscap.ipp.pt
- ² CEOS.PP, ISCAP, Polytechnic of Porto, 4465-004 Matosinhos, Portugal; pgomes@iscap.ipp.pt (L.G.); alima@iscap.ipp.pt (A.L.)
- * Correspondence: claudiap@iscap.ipp.pt

Abstract: The main source of financing is bank loans for Portuguese small and medium-sized entities (SMEs), which implies several constraints to obtaining additional funds. Relying on the argument of Positive Accounting Theory (PAT) that accounting choices are not neutral and on Agency Theory that information asymmetry prevails between insiders and outsiders, we analyzed the impacts of debt on earnings quality, focusing on its level, its increases, and its term of payment. We estimated econometric regressions using panel data with fixed effects over 2013–2019, using discretionary accruals as an inverse proxy of earnings quality. We found empirical evidence that the relationship between debt and earnings quality tends to vary in sign, as the quality of financial information deteriorates with debt, but as debt becomes high, firms tend to increase the quality of earnings. Furthermore, we found that short-term debt tends to decrease earnings quality more than long-term debt. This article aimed to contribute to the prior literature by collecting evidence that debt levels tend to be an incentive to increase earnings management and fill the gap by analyzing the influence of different debt features. This evidence is useful because earnings management may compromise both stakeholders' confidence and the efficient allocation of capital.

Keywords: earnings management; discretionary accruals; SMEs' debt; non-linear association

1. Introduction

Financial statements (FS) are the main source of information for analyzing a firm's economic and financial performance, which stakeholders consider to make efficient decisions, as mentioned by Givoly et al. (2010) and Jeon and Oh (2020). However, the FSs are prepared according to flexible accounting regulations applied by a broad number and variety of industries and entities. Nevertheless, they may produce a negative impact on the quality of financial information when misused by managers, as suggested by Mendes and Lima Rodrigues (2007) and Cerqueira and Pereira (2019).

According to Positive Accounting Theory, the decisions of financial information preparers are not neutral, being influenced by several factors, such as debt covenants, compensation contracts, and political costs (Healy and Wahlen 1999; Watts and Zimmerman 1990). These factors become different incentives for earnings management. In the Continental European system, SMEs are predominant in economies, exhibiting a strong link between accounting and taxation and having bank funds as their main source of financing (Nobes 1998; Nobes and Parker 2020; Fearnley and Gray 2015). To obtain better negotiation conditions with creditors, managers may have incentives to engage in earnings management practices as argued by Pereira et al. (2023b). Relying on Agency Theory, given the existence of information asymmetry between the managers (insiders) and the users (outsiders) of accounting information, management may use discretion when preparing and reporting a FS for their benefit (Ghazali et al. 2015). Firms' stakeholders are aware of this information asymmetry. Cerqueira and Pereira (2020) argue that stakeholders look for high-quality



Citation: Sequeira, José, Cláudia Pereira, Luís Gomes, and Armindo Lima. 2024. Features of the Association between Debt and Earnings Quality for Small and Medium-Sized Entities. *Risks* 12: 32. https://doi.org/10.3390/ risks12020032

Academic Editors: Eulália Mota Santos and Margarida Freitas Oliveira

Received: 31 December 2023 Revised: 24 January 2024 Accepted: 25 January 2024 Published: 3 February 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). financial information and Monteiro et al. (2021) find that earnings quality determines the usefulness of financial information.

In this study, we focused on debt, which has been documented as a determinant of earnings management, as mentioned by Fields et al. (2001), Jelinek (2007), and Lazzem and Jilani (2018). Meanwhile, Ghosh and Moon (2010) find that high debt has a negative effect on earnings quality because managers may use accruals to avoid covenant violations. These authors document a non-monotonic relationship between debt and earnings quality, which is positive at low levels of debt and negative at high debt levels.

Conversely, the high level of indebtedness of Portuguese SMEs may lead to mitigated earnings management and then become a mechanism of corporate governance to capture creditors' confidence and obtain additional funds, as mentioned by Pereira et al. (2023b). In the same vein, Ewert and Wagenhofer (2015) argued that debt clauses are incentives for decision-makers to manage earnings. Our contribution in this paper is threefold: first, we provide evidence of the impact of banking system dependence on the earnings quality of Portuguese SMEs, taking into account different debt features. Second, we show that this effect varies with the level of debt, which tends to deteriorate earnings quality by leading to more earnings management activities to communicate better performance to externals. While we found evidence that the relationship between debt and earnings management tends to be exacerbated for SMEs with increasing levels of debt, it is mitigated for SMEs with the highest levels of debt. Third, we found that short-term debt is an incentive for decision-makers to engage in earnings management activities in response to the high demand for liquidity by Portuguese SMEs. Then, this study aimed to fill in the lack of evidence regarding the impact of debt and its different features on earnings management for SMEs, which are the most representative firms in several European economies.

The remainder of this paper is organized as follows. In Section 2, we provide a literature review and then we posit the hypotheses. In Section 3, we explain the empirical methodology design, namely the sample selection procedures, variables definitions, and the empirical model. In Section 4, we present the descriptive statistics and the results of the empirical tests. Finally, in Section 5, we provide our concluding remarks.

2. Literature Review

2.1. Earnings Management

Accounting earnings are a fundamental source of information for decision-making (Francis et al. 2004). Earnings quality plays a crucial role, especially in controversial financial markets times, allowing investors to make more efficient and less risky decisions as Kousenidis et al. (2013) argued.

According to Mukhametzyanov and Nugaev (2016), the use of financial information reduces uncertainty and leads to better decision-making, thus having a positive impact on company performance. It is possible to find a characteristic for earnings quality in Dechow and Schrand (2004) that in turn is the ability of earnings to feasibly reflect the firm's current performance. Bhattacharya et al. (2013) for the United States find empirical evidence that the quality of accounting information contributes to mitigating information asymmetry. While the IFRS aims to promote greater comparability and quality in financial information, these standards rely on flexibility, which allows earnings management activities (Toumeh and Yahya 2019). Then, based on PAT, the accounting choices rely on different motivations, such as to avoid political cost, not violate debt contract covenants, or to maximize the manager's bonus (Watts and Zimmerman 1978).

Earnings management is a globally used tool for long-term profitable firms and the reduction of bankruptcy risk (Durana et al. 2020). Considering that managers may have incentives to deteriorate earnings' quality, they may engage in earnings management practices by using privileged information to mislead stakeholders about the true performance of firms because they are no longer complete and transparent (Healy and Wahlen 1999). These authors presented a broader definition, pointing to the existence of earnings management, which is when managers use the subjectivity contained in the accounting standards to act

on the FS to hide the firm's true performance from stakeholders, or to influence the fulfillment of certain contracts that depend on financial information. For Schipper (1989), the activities of managers in the financial reporting process aim at obtaining a private benefit. Based on Koumanakos et al. (2005), earnings management is fundamentally characterized by the process in which managers take intentional actions, within the limits of generally accepted accounting principles, to achieve a desired level of earnings.

It is important to note that earnings management does not cover accounting fraud, but rather the flexibility contained in accounting standards to alter financial information without exceeding the legal limits. In addition, managers often use the flexibility contained in the standards to offer a true and fair view of the firm, allowing the users of financial information to make more educated decisions. Thus, according to Beneish (2001), earnings management can be seen from two perspectives: the opportunistic perspective, which seeks to mislead investors through the misrepresentation of disclosed information; and the informative perspective, in which managers seek to manipulate information to reveal to investors their private expectations about future cash flows.

Earnings management is associated with poor earnings quality. Monteiro et al. (2021) find that the quality of financial reporting is a predictive variable of the usefulness of financial reporting. It is important to analyze which factors may affect earnings quality.

Given that the quality of the earnings is not directly observable, several estimates have been developed in the literature (Dechow et al. 2010; Schipper and Vincent 2003), including earnings smoothing, conservatism by more timely recognition of losses relative to gains, discretionary accruals, and earnings persistence (Cerqueira and Pereira 2019). In this study, we used discretionary accruals as a proxy for earnings quality because we aimed to analyze the effect produced by debt on earnings quality.

2.2. Effect of Debt on Earnings Management and Hypothesis Development

Prior studies found conflicting empirical evidence regarding the effects of debt on earnings management.

While debt is associated with managers being engaged in earnings management activities behavior due to the need for additional financing (Khuong et al. 2022; Ewert and Wagenhofer 2015), earnings management is limited because with debt the supervision by external creditors increases, namely banking institutions (Mamatzakis et al. 2022; Vakilifard and Mortazavi 2016).

The research conducted by Lazzem and Jilani (2018), applied to a sample of French companies indexed on the CAC All-Tradable, examined the impact of leverage on accrualsbased earnings management practices. Their results led to the conclusion that debt has a positive effect on earnings management. Thus, it was found that increases in debt provide an incentive for managers to manipulate results to avoid violating debt contracts. In a study conducted by Warrad (2017) on industrial firms listed on the Amman Stock Exchange (ASE) during the period from 2011 to 2015, it was concluded that there was a significant influence of debt ratio and ROA on earnings quality. Furthermore, Ewert and Wagenhofer (2015) argued that debt clauses are incentives for firms to engage in earnings management activities.

In contrast, firms with high levels of debt tend to avoid earnings management practices given that high levels of debt imply more risk of a firm's default, and thus, they are subjected to greater monitoring and scrutiny from external creditors (Jensen 1986). Consistently, Ghosh and Moon (2010) argue that high debt has a negative effect on earnings quality because managers may use accruals to avoid covenant violations. These authors document a non-monotonic relationship between debt and earnings quality, which is positive at low levels of debt and negative at high debt levels.

Meanwhile, Vakilifard and Mortazavi (2016) found empirical evidence that debt harms accruals earnings management but has a positive impact on real earnings management. Inayah et al. (2021) analyzed the impact of financial distress, leverage, and free cash flow on earnings management. These authors found evidence that leverage produces a negative impact on earnings management. In the same vein, Mamatzakis et al. (2022) found evidence

of a negative association between debt and earnings management for Greek firms. This negative effect can be explained by the fact that during the period under review, banks or creditors had a lot of control over companies, so managers had less flexibility to focus on managing earnings.

Given this conflicting evidence, we aimed to analyze the impact of debt on earnings management and whether it differs from the intensive level of debt. In addition, we controlled for factors that are likely to affect earnings management practices, in particular firms' features based on fundamentals, focusing on debt. Thus, based on credit agreements and the motivations for obtaining debt at favorable conditions for the company, the following research hypothesis is formulated:

H1. Debt is positively associated with earnings management activities.

On the other hand, Khuong et al. (2022) mentioned that firms with a high debt-toasset ratio tend to have more difficulties in obtaining additional funds, which may be an incentive to increase earnings management. The results of Pereira et al. (2023a) are divergent from this vein by suggesting that debt reduces earnings management due to the high levels of debt of Portuguese SMEs; therefore, these firms have to capture creditors' and investors' confidence to obtain additional funds. In highly indebted firms, the debt levels increase the likelihood of defaults on future debt obligations, so in this situation, external creditors tend to increase monitoring of firms, decreasing opportunistic behavior by managers (Anagnostopoulou and Tsekrekos 2017; Zamri et al. 2014). Following the most recent trend, we posit the following hypothesis:

H2. Firms with the highest levels of debt exhibit fewer earnings management activities.

To further develop this research, we investigated whether firms that exhibit debt increases tend to manage their earnings in a more exacerbated manner than those firms with high leverage levels. Jelinek (2007) and Lazzem and Jilani (2018) found that the impact of debt on earnings management is different in debt-increasing firms and highly indebted firms. These authors found that firms experiencing increases in debt are more likely to engage in accrual-based earnings management practices than firms with consistently high debt because when firms have increases in their levels of debt, they tend to contract new debt. In addition, this enhanced propensity for earnings management is driven by the increase to persuade the lenders of good management. In the same sense, Khanh and Anh Phung (2019) identified a positive relationship between increasing leverage and earnings management. Then, we posit the third hypothesis:

H3. *Firms with increasing debt have higher earnings management activities.*

In addition, debt maturity is indicated as being able to explain the adoption of earnings management practices. Most of the literature focuses on the link between debt and earnings management but does not examine the differences between short-term and long-term debt (Ghazali et al. 2015; Jelinek 2007; Khanh and Anh Phung 2019; Koumanakos et al. 2005; Lazzem and Jilani 2018; Vakilifard and Mortazavi 2016).

Trung et al. (2020) argue that short-term debt is substantially different from long-term debt because the former presents a greater propensity for liquidity risk. Thus, short-term debt increases a firm's liquidity risk because cash flows are needed sooner to pay off that debt. Based on this theory, a positive relationship is expected between short-term debt and earnings management.

Fields et al. (2012) conclude that increasing short-term debt is associated with greater discretionary accruals, in particular for firms that are about to obtain new loans. These harmful repercussions are anticipated according to financial crisis theory (Fung and Goodwin 2013; Gupta et al. 2008). Since short-term debt is linked to liquidity risk, compared to long-term

debt, the positive correlation between short-term debt and earnings manipulation should be greater compared to (the more intensely investigated) relationship with the general debt. Taking these studies into account, the following hypothesis was formulated:

H4. The positive impact of short-term debt on earnings management is more substantial compared to long-term debt.

El-Rabat et al. (2023) argue that large firms have better internal control mechanisms and more experienced internal auditors compared to smaller ones, leading to disclosing more accurate financial information. Consistently, Kalbuana et al. (2021) mention that external auditors give more attention to larger firms, and consequently, these firms tend to engage less in earnings management activities. In the case of return on assets and interest expenses, firms engage in earnings management activities to meet contractual debt obligations and to mitigate the negative effects of interest on net income (Anagnostopoulou and Tsekrekos 2017; Zamri et al. 2014). In this empirical study, we controlled for the effect of firm size and the amount of interest expense on earnings management.

3. Results

3.1. Descriptive Statistics

For the statistics work, we used RStudio software version 2022.07.2, namely to estimate the econometric regressions. Table 1 presents the descriptive statistics for the dependent and independent variables included in the models, which are all based on annual financial reporting:

The mean absolute value of the discretionary accruals, which are not the result of economic activity but of managers' discretionary actions, represent 8% of the total assets. Although slightly higher, this is consistent with that shown by Cerqueira and Pereira (2015) for a previous Portuguese case (4.45%).

Regarding the Debt variable, the total debt scaled by total assets, we can conclude that in general, Portuguese SMEs are not highly leveraged, with an average value of 32%. However, the sample period coincides with the aftermath of the global financial crisis, which was characterized by a lack of liquidity in the economy, particularly in the Portuguese financial market. In this table, the variable Size is presented in euros instead of its log to be clearer in its interpretation. The variation in size by company is high, as can be seen from the standard deviation. The minimum size is around 5 million and the maximum is 107 million. This value is above the 20 million for the maximum value for assets, but the two other limits, turnover and the number of employees, are below the maximum values allowing the entity to be classified as an SME. In addition, these firms have a low ROA, 3% on average. This suggests that these firms have little ability to finance their assets through their resources, often leading them to opt for borrowed capital to the detriment of equity. In addition, the short-term debt scaled by total assets is 14%, while the long-term, most suitable for investment, is higher (19%).

Variable	Mean	Median	Min	Max	St.d
Debt	0.32	0.30	0	1.04	0.21
DebtH	0.5	0	0	1	0.5
DebtST	0.14	0.10	0	0.61	0.13
DebtLT	0.19	0.14	0	0.87	0.18
Size	9845	5457	507	107,838	14,927
ROA	0.03	0.02	-0.20	0.26	0.07
IntExp	0.05	0.04	0	0.52	0.07
DAĈ	0.08	0.05	0.00	0.54	0.09

Table 1. Descriptive statistics.

Source: Authors' calculations.

Table 2 documents the correlations between the variables used in the models.

	DAC	Debt	DebtST	DebtLT	DebtH	Size	ROA	IntExp	Growth
DAC	1								
Debt	0.08 ***	1							
DebtST	0.14 ***	0.51 ***	1						
DebtLT	-0.02 ***	0.77 ***	-0.13 ***	1					
DebtH	-0.01	0.55 ***	0.21 ***	0.47 ***	1				
Size	-0.08 ***	0.12 ***	-0.05 ***	-0.17 ***	0.13 ***	1			
ROA	0.03 ***	-0.24 ***	-0.14 ***	0.17 ***	-0.14 ***	0.00 ***	1		
IntExp	0.07 ***	-0.25 ***	-0.13 ***	-0.2 ***	-0.08 ***	-0.02 ***	-0.02 ***	1	
Growth	0.16 ***	0.21 ***	0.15 ***	0.13 ***	0.00	0.16 ***	0.00 ***	-0.13 ***	1

Table 2. Correlation matrix.

*** represents the level of significance at 1%.

The correlations between the dependent variable and the independent (no control) variable vary from -8% (Size) to 14% (DebtST).

The highest correlation coefficients between the independent variables are observed between Debt variables, but the coefficients between Debt and the other independent variables are low, suggesting that this sample has no multicollinearity problems.

3.2. Econometric Regressions Results

We analyzed the association between earnings quality and indebtedness features by estimating the dynamic fixed models using the generalized method of moments (GMM) in two-stage squares (2SLS) with the instrumental variable discretionary accruals lagged, which is a way of dealing with potential endogeneity issues. Table 3 documents the results with panel data using fixed effects for a sample period of seven years.

Table 3. Impact of debt on discretionary accruals.

Variables	Predicted Sign	Model 1	Model 2	Model 3	Model 4
Lag (DAC;1)		0.0438 *** (2.8779)	0.05 *** (3.2573)	-0.0656 (-1.6282)	0.0447 *** (2.9425)
Debt	(+)	0.0486 *** (3.9656)			
DebtH	(-)		-0.0083 ** (-2.2706)		
DebtInc	(+)			-0.0021 (-0.1803)	
DebtST	(+)				0.0543 *** (3.435)
DebtLT					0.0347 ** (2.5449)
ROA	(+)	0.0128 (0.5867)	0.0100 (0.4562)	-0.0649 (-19.0943)	0.0124 (0.5679)
IntExp	(+)	0.0849 *** (4.4899)	0.0646 ***	0.4912 **	0.0832 *** (4.4131)
Size	(-)	-0.0369 *** (6.3371)	-0.0356 *** (-6.0678)	(-0.0241) (-1.2919)	-0.0370 *** (-6.3543)
Growth	(+)	0.0723 *** (0.5257)	0.0881 *** (14.0256)	0.1142 *** (7.5963)	0.07472 *** (11.1280)
Firm-year obs.		32,788	32,788	6186	32,788
Sargan test (X^2)		19.6970	18.5972	15.0375	19.1865
Autocorrelation (1)		-27.5684 ***	-27.5145 ***	-6.6673 ***	-27.5442 ***
Autocorrelation (2)		2.5350 **	2.6643 ***	-0.1908	2.5650 **
Wald test (X^2)		240.336 ***	244.5344 ***	71.1135 ***	236.4554 ***

***, **, represent the level of significance at 1%, 5% respectively. The statistics given in the parentheses under the coefficients of independent variables are Z-values. Source: Authors' calculations.

To test the robustness of the results, we used Sargans' over-identification of restrictions test under the null hypothesis that the restrictions imposed are valid. The results led to the acceptance of the null hypothesis, concluding that the restrictions are valid and the results are robust for all models. We also tested the null hypothesis of the absence of the first and second order autocorrelation. These results mostly rejected the null hypothesis. Finally, to test the quality of the adjustment of the independent variables as a whole in relation to the dependent variable, we carried out a Wald test. The results rejected the null hypothesis, suggesting that the independent variables are statistically significant to globally explain the model.

The Debt variable in model 1 is statistically significant at the 1% level and its estimated sign is positive, as expected. This result suggests that debt positively affects earnings management, which in turn means that debt is negatively related to earnings quality. This evidence empirically supports our first hypothesis and is consistent with Kalbuana et al. (2021), Khanh and Anh Phung (2019), and Lazzem and Jilani (2018).

Concerning the control variables, the coefficients are statistically significant at a 1% level, except for ROA, which is not statistically significant. Moreover, the estimated coefficient exhibits the expected sign: IntExp is positive, suggesting that firms with higher interest expenses are associated with greater earnings management. This result is in line with Jelinek (2007), Vakilifard and Mortazavi (2016), Anagnostopoulou and Tsekrekos (2017), and Kalbuana et al. (2021). Furthermore, managers are also attempting to meet the expectations of creditors (Lazzem and Jilani 2018). In the case of Size, its estimated coefficient is negative. Lager firms have stronger internal control systems and are audited by external organizations, justifying the decrease in earnings management activities (El-Rabat et al. 2023; Kalbuana et al. 2021). The coefficient of the variable associated with Growth is positive, as expected, and statistically significant at 0.1%. This result suggests that firms with higher business growth have higher levels of discretionary accruals, in line with Al-Najjar and Riahi-Belkaoui (2001). This result shows that firms with greater growth opportunities increase their flexibility to adopt earnings management practices.

Model 2 analyzed the impact of high levels of corporate debt on earnings management activities, focusing on the variable DebtH. Its estimated coefficient is statistically significant at a 5% level, with a negative sign, as expected. This is consistent with high-debt firms engaging in fewer earnings management activities, which is in line with Anagnostopoulou and Tsekrekos (2017), Jensen (1986), and Zamri et al. (2014). This evidence empirically supports H2, suggesting that increasing leverage tends to increase earning quality as it reduces managers' opportunistic behavior, as it increases the likelihood of defaulting on debt services in the future, thereby increasing monitoring by external creditors. This means that the need to meet interest and debt payment obligations reduces the flexibility of managers' decisions. This result is in line with the finding of a non-monotonic relationship between debt and earnings quality reported by Ghosh and Moon (2010), although with opposite signs because these authors found that the relationship is positive at low levels of debt and negative at high debt levels.

The results for the control variables are consistent with the expected and similar to model 1 and to the prior literature in the case of IntExp with Jelinek (2007), Vakilifard and Mortazavi (2016), Anagnostopoulou and Tsekrekos (2017), and Kalbuana et al. (2021); for Size with El-Rabat et al. (2023) and Kalbuana et al. (2021); and for Growth with Al-Najjar and Riahi-Belkaoui (2001).

Model 3 analyzed the impact of the increase in debt on earnings management activity, focusing on the variable DebtInc. Its estimated coefficient is not statistically significant, which does not allow us to empirically support H3.

Regarding the control variables, the IntExp variable and Growth are statically significant at a 5% and a 1% level, respectively. Their estimated signs are positive as expected.

Model 4 analyzed the impact of the existence of short-term debt on earnings management activity. The estimated coefficient of the DebtST variable is statistically significant at the 1% level and the sign is positive, as expected, suggesting that short-term debt increases earnings management activities, thus harming earnings quality. In the case of DebtLT, it is significant at a 5% level, also exhibiting a positive sign. As both the significance and coefficient of DebtST are higher than DebtLT, these findings allow us to empirically support H4, that the positive impact of short-term debt on earnings management is more substantial than that of long-term debt. The short-term need of liquidity is associated with greater discretionary accruals, particularly for firms that intend to obtain new loans, consistent with Fields et al. (2012).

For the control variables, the results are consistent with the expected and similar to model 1 and to the prior literature in the case of IntExp with Jelinek (2007), Vakilifard and Mortazavi (2016), Anagnostopoulou and Tsekrekos (2017), and Kalbuana et al. (2021); for Size with El-Rabat et al. (2023) and Kalbuana et al. (2021); and for Growth with Al-Najjar and Riahi-Belkaoui (2001).

4. Empirical Methodology

4.1. Sample Selection

This study focused on a sample of Portuguese SMEs taking into account national accounting standards, which must meet two of these three requirements: total assets between 350,000 EUR and 20,000,000 EUR; turnover between 700,000 EUR and 40,000,000 EUR; and annual average number of employees between 10 and 50 (at the end of the year). All of the financial data were collected using the SABI database (Iberian Balance Sheet Analysis System). The sample period covered 7 years for the estimation of econometric regressions, covering the years 2013 to 2019 to avoid the COVID-19 pandemic effect. To include a firm in the sample it must report data for the variables under study for the entire analysis period. In addition, we excluded financial and insurance firms as they have specific regulations, and industries with less than 30 firms.

After the selection procedures, the final sample consisted of 4684 firms, representing 32,788 firm-year observations.

4.2. Empirical Models

Consistent with previous studies, discretionary accruals were used as a proxy for earnings management.

To obtain the discretionary accruals, we first computed total accruals through Equation (1):

$$TA_{i,t} = NI_{i,t} - FCO_{i,t}$$
⁽¹⁾

where:

i, t represents firm i for period t.

TA is total accruals for firm I in year t.

NI is the variation in net income between periods, given by the net income of year t minus the net income in the previous year (t - 1).

FCO is the cash flow from operating activities of year t.

After obtaining the total accruals, we estimated the discretionary accruals based on the Jones model modified by Dechow et al. (1995) with the return on assets from the previous period as proposed by Kothari et al. (2005). The discretionary accruals were the residuals of the cross-section estimation of Equation (2). These residuals represent the component of accruals that remains after controlling for firm performance, economic activity, and investment in tangible fixed assets, and are likely to be associated with managers' discretionary choices:

$$TA_{i,t} = \alpha_0 + \alpha_1 Rev_{i,t} + \alpha_2 PPE_{i,t} + \alpha_3 ROA_{i,t-1} + \varepsilon_{i,t}$$
(2)

where:

TA is the value of total accruals computed by Equation (1).

Rev is the change in sales, given by the difference between sales in the current year (t) and sales in the previous year (t - 1), minus the change in accounts receivable, given by the difference between accounts receivable in the current year and accounts receivable in the previous year.

PPE is the tangible fixed assets.

ROA is the return on assets in the previous period (t - 1).

All variables are scaled by the total assets of year t - 1.

 $\varepsilon_{i,t}$, are the residuals of the regression estimation.

After estimating the residuals, $\varepsilon_{i,t}$, we tested the hypotheses, namely regarding the relationship between different levels of debt and discretionary accruals as a proxy for earnings management. Furthermore, discretionary accruals are an inverse proxy for the earnings quality because the residuals from accrual models represent management discretion or estimation errors, both of which reduce decision usefulness (Dechow et al. 2010).

In addition, we controlled for the firm's size and the level of interest expenses. For that purpose, we estimated four regressions, using panel data with fixed effects, whose selection was based on the Hausman test:

$$|DAC_{i,t}| = \alpha_0 + \beta_1 Debt_{i,t} + \beta_2 ROA_{i,t} + \beta_3 Size_{i,t} + \beta_4 IntExp_{i,t} + \beta_5 Growth_{i,t} + \mu_{i,t}$$
(3)

where:

|DAC| is the absolute value of the residuals of the total accruals (discretionary accruals) scaled by the average total assets¹.

Debt is the amount of total indebtedness over the total assets for the previous year (t - 1). To support our first hypothesis, it is expected that the coefficient β_1 in Equation (3) will be positive and statistically significant.

In addition, based on prior literature regarding the determinants of discretionary accruals, we included the following control variables:

ROA is the return on total assets as measured by net income divided by average total assets. We expect a positive sign in the relationship because firms with high ROA tend to engage in earnings management activities to pay less taxes (Anagnostopoulou and Tsekrekos 2017; Jelinek 2007; Kalbuana et al. 2021; Vakilifard and Mortazavi 2016);

Size corresponds to firms' size measured by the logarithm of total assets in the previous year (t - 1). We expect a negative sign in the relationship because larger firms have better internal controls and are audited by more experienced auditors (El-Rabat et al. 2023; Kalbuana et al. 2021; Vakilifard and Mortazavi 2016);

IntExp is given by interest expense and other financial costs scaled by the total debt of the previous year (t - 1). Managers have to satisfy the expectations of creditors and investors, leading them to engage in earnings management activities (Lazzem and Jilani 2018);

Growth corresponds to the company's growth proxy, measured by the percentage growth of total assets. This variable is expected to have a negative coefficient, consistent with the studies by Khanh and Anh Phung (2019) and Jelinek (2007);

 $\mu_{i,t}$ are the residuals of the regression estimation.

We then ran the following regression to test our second hypothesis:

 $|DAC_{i,t}| = \alpha_0 + \beta_1 DebtH_{i,t} + \beta_2 ROA_{i,t} + \beta_3 Size_{i,t} + \beta_4 IntExp_{i,t} + \beta_5 Growth_{i,t} + \mu_{i,t}$ (4)

where:

DebtH is a dummy variable that takes the value 1 if the firm's debt ratio, the ratio of debt to assets, is above the third quartile, ranging between 75% and 100%, and 0 otherwise. We divide the sample into quartiles based on the ratio of debt to assets, and then firms whose ratio is at the upper limit of a data set, i.e., in the third quartile, are considered to be highly indebted.

This hypothesis is supported if the coefficient β_1 is statistically negative because firms with high levels of debt tend to decrease their earnings management activities to capture creditors' confidence and obtain additional funds, thereby improving earnings quality.

All of the other variables present the same definition. The third hypothesis is tested by the following regression:

$$|DAC_{i,t}| = \alpha_0 + \beta_1 DebtInc_{i,t} + \beta_2 ROA_{i,t} + \beta_3 Size_{i,t} + \beta_4 IntExp_{i,t} + \beta_5 Growth_{i,t} + \mu_{i,t}$$
(5)

where:

DebtInc is a dummy variable that takes the value 1 if the firm exhibits an increase in debt from year t - 2 to year t and 0 otherwise. A company is classified as having increased its indebtedness if it is in the first quartile in year t - 2 and moves to the third or fourth quartile in year t, or if it is in the second quartile in year t - 2 and moves to the fourth quartile in year t. This variable analyzes the impact of the increase in debt in a static way without the occurrence of debt variations.

This hypothesis is supported if the coefficient β_1 is statistically positive.

All of the other variables have the same definition.

Concerning the impact of debt maturity, to test hypothesis 4, we estimated the following equation:

 $|DAC_{i,t}| = \alpha_0 + \beta_1 DebtST_{i,t} + \beta_2 DebtLT_{i,t} + \beta_3 ROA_{i,t} + \beta_4 Size_{i,t} + \beta_5 IntExp_{i,t} + \beta_6 Growth_{i,t} + \mu_{i,t}$ (6)

where:

DebtST is the short-term debt ratio, measured by the total value of current financing scaled by total assets in the previous period;

DebtLT is the long-term debt ratio, measured by the total value of long financing scaled by total assets in the previous period;

All of the other variables have the same definition.

5. Conclusions

In this study, our main objective consisted of analyzing the effects of debt levels on earnings quality. Our results showed a significant and positive association between the levels of debt and earnings management. This may be due both to the need to obtain additional funds and to have better credit covenants, which is consistent with the PAT. Furthermore, our results show that firms with increasing debt tend to engage in more earnings management practices, producing negative impacts on the quality of financial information and then increasing information asymmetry between outsiders and insiders. However, our empirical evidence suggests that this association is non-linear because when firms have extreme levels of debt, they tend to mitigate earnings management activities to capture their creditors' and investors' confidence. In addition, we found evidence that short-term debt increases in a more substantial manner in earnings management activities than long-term debt, as firms that increase this type of debt tend to have higher discretionary accruals to obtain new loans, which allows them to fulfill commitments that require high liquidity.

Thus, this paper shows that debt levels do not tend to be a mechanism for regulating the managers' activities to increase the quality of financial information. On the contrary, they contribute to increased earnings management, compromising the confidence of stakeholders as well as efficient decision making, except for firms with high levels of debt.

These results are important for creditors and investors, but also for managers, as they contribute to an alignment of interests with the respective mitigation of agency costs. Furthermore, this empirical evidence can be particularly useful for managers of SMEs, mostly in European economies, to consolidate more responsible management behavior toward society.

A major limitation of this study is the exclusion of more recent years aiming to avoid possible bias in the empirical results due to the dramatic effects of the COVID-19 pandemic on the economy.
For future research, we propose to split the sample by quantiles to analyze the nonlinear effect of debt as well as including larger firms with other sources of financing as the capital markets.

Author Contributions: Conceptualization, J.S., C.P. and A.L.; methodology, J.S., C.P. and L.G.; software, J.S. and C.P.; validation, C.P. and L.G.; formal analysis, J.S., C.P., L.G. and A.L.; resources, J.S.; data curation, J.S. and C.P; writing—original draft preparation, J.S. and C.P.; writing—review and editing, L.G. and A.L.; visualization, C.P.; supervision, C.P. and L.G.; project administration, J.S. and C.P. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Informed Consent Statement: Not applicable.

Data Availability Statement: Research data from this study are available on request from the corresponding author.

Conflicts of Interest: The authors declare no conflict of interest.

Note

¹ It should be noted that our aim is not to analyze whether earnings management activities are motivated by the willingness to increase or reduce earnings, but rather the magnitude of earnings management.

References

- Al-Najjar, Fouad, and Ahmed Riahi-Belkaoui. 2001. Growth Opportunities and Earnings Management. *Managerial Finance* 27: 72–81. [CrossRef]
- Anagnostopoulou, Seraina, and Andrianos Tsekrekos. 2017. The Effect of Financial Leverage on Real and Accrual-based Earnings Management. Accounting and Business Research 47: 191–236. [CrossRef]
- Beneish, Messod. 2001. Earnings Management: A Perspective. Managerial Finance 27: 3–17. [CrossRef]
- Bhattacharya, Nilabhra, Hemang Desai, and Kumar Venkataraman. 2013. Does Earnings Quality Affect Information Asymmetry? Evidence from Trading Costs. Contemporary Accounting Research 30: 482–516. [CrossRef]
- Cerqueira, António, and Cláudia Pereira. 2015. Accounting Accruals and Information Asymmetry in Europe. *Prague Economic Papers* 24: 638–61. [CrossRef]
- Cerqueira, António, and Cláudia Pereira. 2019. Earnings Management and Stock Market Reaction, book chapter. In *International Financial Reporting Standards and New Directions in Earnings Management*. Edited by Jonas Oliveira, Graça Azevedo and Augusta Ferreira. Charlottesville: IGI-Global. [CrossRef]
- Cerqueira, António, and Cláudia Pereira. 2020. The Effect of Economic Conditions on Accounting Conservatism under IFRS in Europe. *Review of Economic Perspectives* 20: 137–69. [CrossRef]
- Dechow, Patricia, and Catherine Schrand. 2004. Earnings Quality. Charlottesville, VA: Research Foundation of CFA Institute.
- Dechow, Patricia, Richard Sloan, and Amy Sweeney. 1995. Detecting Earnings Management. The Accounting Review 70: 193–225.
- Dechow, Patricia, Weili Ge, and Catherine Schrand. 2010. Understanding Earnings Quality: A Review of the Proxies, their Determinants and their Consequences. *Journal of Accounting and Economics* 50: 344–401. [CrossRef]
- Durana, Pavol, Katarina Valaskova, Darina Chlebikova, Vladislav Krastev, and Irina Atanasova. 2020. Heads and Tails of Earnings Management: Quantitative Analysis in Emerging Countries. *Risks* 8: 57. [CrossRef]
- El-Rabat, Monir, Hala Abdel-Naby Abdel-Fattah, and Manal Abdel-Azim. 2023. The Moderating Role of Firm Size on the Relationship between Financial Distress and Earnings Management. *The Academic Journal of Contemporary Commercial Research* 3: 42–60. [CrossRef]

Ewert, Ralf, and Alfred Wagenhofer. 2015. Economic Relations Among Earnings Quality Measures. Abacus 51: 311–55. [CrossRef]

- Fearnley, Nicholas, and Sid Gray. 2015. National Institutional Factors and IFRS Implementation in Europe: The Case of Investment Property Companies. *International Journal of Accounting and Information Management* 23: 271–88. [CrossRef]
- Fields, Thomas, Thomas Lys, and Linda Vicent. 2001. Empirical Research on Accounting Choice. *Journal of Accounting and Economics* 31: 255–307. [CrossRef]
- Fields, Paige, Manu Gupta, Michael Wilkins, and Shage Zhang. 2012. Refinancing Pressure and Earnings Management: Evidence from Changes in Short-Term Debt and Discretionary Accruals. *SSRN Electronic Journal*. [CrossRef]
- Francis, Jennifer, Ryan LaFond, Per Olsson, and Katherine Schipper. 2004. Costs of Equity and Earnings Attributes. *The Accounting Review* 79: 967–1010. [CrossRef]
- Fung, Simon, and John Goodwin. 2013. Short-term Debt Maturity, Monitoring and Accruals-based Earnings Management. Journal of Contemporary Accounting & Economics 9: 67–82. [CrossRef]
- Ghazali, Aziatul, Nur Shafie, and Zuraidah Sanusi. 2015. Earnings Management: An Analysis of Opportunistic Behaviour, Monitoring Mechanism and Financial Distress. *Procedia Economics and Finance* 28: 190–201. [CrossRef]

- Ghosh, Aloke, and Doocheol Moon. 2010. Corporate Debt Financing and Earnings Quality. *Journal of Business Finance and Accounting* 37: 538–59. [CrossRef]
- Givoly, Dan, Carla Hayn, and Sharon Katz. 2010. Does Public Ownership of Equity Improve Earnings Quality? *The Accounting Review* 85: 195–225. [CrossRef]
- Gupta, Manu, Inder Khurana, and Raynolde Pereira. 2008. Legal Inforcement, Short Maturity Debt, and the Incentive to Manage Earnings. *Journal of Law and Economics* 51: 619–39. [CrossRef]
- Healy, Paul, and James Wahlen. 1999. A Review of the Earnings Management Literature and Its Implications for Standard Setting. *Accounting Horizons* 13: 365–83. [CrossRef]
- Inayah, Rifka, Amiruddin Amiruddin, and Grace Pontoh. 2021. Analysis the Effect of Financial Distress, Leverage and Free CashFlow on Earnings Management. *GATR Accounting and Finance Review* 6: 111–19. [CrossRef] [PubMed]
- Jelinek, Kate. 2007. The Effect of Leverage Increases on Earnings Management. Journal of Business and Economic Studies 13: 24-46.
- Jensen, Michael. 1986. Agency Costs of Free Cash Flow, Corporate Finance, and Takeovers. *The American Economic Review* 76: 323–29. Jeon, Heung Joo, and Hyun Min Oh. 2020. Debt Origin and Investment Efficiency from Korea. *International Journal of Financial Studies* 8: 47. [CrossRef]
- Kalbuana, Nawang, Budi Prasetyo, Pribadi Asih, Yenni Arnas, Suse Simbolon, Alwazir Abdusshomad, Benny Kurnianto, Kardi Rudy, Riyanto Saputro, Maylia Sari, and et al. 2021. Earnings Management is Affected by Firm Size, Leverage and ROA: Evidence from Indonesia. Academy of Strategic Management Journal 20: 1–12.
- Khanh, Hoang, and Thu Anh Phung. 2019. The Effect of Financial Leverage on Real and Accrual-based Earnings Management in Vietnamese Firms. *Economics and Sociology* 12: 299–312. [CrossRef]
- Khuong, Nguyen, Abdul Rahman, Abdelrhman Meero, Le Anh, Nguyen Liem, Cao Thuy, and Huynh Ly. 2022. The Impact of Corporate Social Responsibility Disclosure and Accounting Comparability on Earnings Persistence. Sustainability 14: 2752. [CrossRef]
- Kothari, S. P., Andrew Leone, and Charles Wasley. 2005. Performance Matched Discretionary Accrual Measures. *Journal of Accounting* and Economics 39: 163–97. [CrossRef]
- Koumanakos, Evangelos, Costas Siriopoulos, and Antonios Georgopoulos. 2005. Firm Acquisitions and Earnings Management: Evidence from Greece. *Managerial Auditing Journal* 20: 663–78. [CrossRef]
- Kousenidis, Dimitrios, Panagiotis Dimitropoulos, Dimitrios Asteriou, and Stergios Leventis. 2013. The Impact of IFRS on Accounting Quality: Evidence from Greece. *Advances in Accounting* 29: 108–23.
- Lazzem, Safa, and Faouzi Jilani. 2018. The Impact of Leverage on Accrual-based Earnings Management: The Case of Listed French Firms. *Research in International Business and Finance* 44: 350–58. [CrossRef]
- Mamatzakis, Emmanuel, Panagiostis Pegkas, and Christos Staikouras. 2022. The Impact of Debt, Taxation and Financial Crisis on Earnings Management: The Case of Greece. *Managerial Finance. ahead-of-print*. [CrossRef]
- Mendes, Cláudia, and Lúcia Lima Rodrigues. 2007. Determinantes da Manipulação Contabilística. In *Tékhne-Revista de Estudos Politécnicos*. Barcelos: Instituto Politécnico do Cávado e do Ave.
- Monteiro, Albertina, Joana Vale, Amélia Silva, and Cláudia Pereira. 2021. Impact of internal control and accounting systems on the financial information usefulness: The role of financial information quality. *Academy of Strategic Management Journal* 20: 1–13.
- Mukhametzyanov, Rinaz, and F. S. Nugaev. 2016. Financial Statements as an Information Base for the Analysis and Management Decisions. *Academy of Strategic Management Journal* 17: 47–54.
- Nobes, Cristopher. 1998. Towards a General Model of the Reasons for International Differences in Financial Reporting. *Abacus* 34: 162–87. [CrossRef]
- Nobes, Cristopher, and Robert H. Parker. 2020. Comparative International Accounting, 14th ed. Boston: Pearson Education.
- Pereira, Ângela, Cláudia Pereira, Luís Gomes, and Armindo Lima. 2023a. Do Taxes Still Affect Earning Persistence? *Administrative Sciences* 13: 48. [CrossRef]
- Pereira, Cláudia, Albertina Monteiro, Diana Silva, and Armindo Lima. 2023b. Do the Levels of Environmental Sustainability Disclosure and Indebteness Affect the Quality of Earnings? *Sustainability* 15: 2871. [CrossRef]
- Schipper, Katherine. 1989. Earnings Management. Accounting Horizons 3: 91-102.
- Schipper, Katherine, and Linda Vincent. 2003. Earnings Quality. Accounting Horizons 17: 97–110. [CrossRef]
- Toumeh, Ahmad, and Sofri Yahya. 2019. A Review of Earnings Management Techniques: An IFRS Perspective. *Global Business and Management Research: An International Journal* 11: 14.
- Trung, Thrin, Nguyen Liem, and Cao Thuy. 2020. The Impact of Short-term Debt on Accruals-based Earnings Management—Evidence from Vietnam. *Cogent Economics & Finance* 8: 1767851. [CrossRef]
- Vakilifard, Hamidreza, and Mahboobe Mortazavi. 2016. The Impact of Financial Leverage on Accrual-Based and Real Earnings Management. International Journal of Academic Research in Accounting, Finance and Management Sciences 6: 53–60. [CrossRef] [PubMed]
- Warrad, Lina. 2017. The Influence of Leverage and Profitability on Earnings Quality: Jordanian Case. *Behavioral and Social Sciences* Librarian 7: 61–82. [CrossRef]
- Watts, Ross, and Jerold Zimmerman. 1990. Positive Accounting Theory: A Ten Year Perspective. The Accounting Review 65: 131–56.

- Watts, Ross, and Jerold Zimmerman. 1978. Towards a Positive Theory of the Determination of Accounting Standards. *The Accounting Review* 53: 112–34.
- Zamri, Norhayati, Rahayu Rahman, and Noor Isa. 2014. The Impact of Leverage on Real Earnings Management. *Procedia Economics and Finance* 7: 86–95. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.





Article Simulation of Dynamic Performance of DeFi Protocol Based on Historical Crypto Market Behavior

Iveta Grigorova *, Aleksandar Karamfilov, Radostin Merakov and Aleksandar Efremov

Faculty of Automatics, Industrial Automation, Technical University, 1000 Sofia, Bulgaria; akaramfilov@tu-sofia.bg (A.K.); r.merakov@tu-sofia.bg (R.M.); alefremov@tu-sofia.bg (A.E.) * Correspondence: iveta.grigorova@tu-sofia.bg

Abstract: In a rapidly evolving and often volatile crypto market, the ability to use historical data for simulations provides a more realistic assessment of how decentralized finance (DeFi) protocols might perform. This insight is crucial for participants, developers, and investors seeking to make informed decisions. This paper presents a comprehensive study evaluating the dynamic performance of a newly developed DeFi protocol—NOLUS. The main objective of this paper is to present and analyze the built realistic model of the platform. This model could be successfully used to analyze the stability of the platform under different environmental influences by performing various simulations and conducting experiments with different parameters that could not be realized with the real platform. In the article, the key components of the platform are presented in detail and the main dependencies between them are clarified, in addition to the ways of forming multiple variables, and the complex relations between them in the real protocol are explained. The main finding from the experimental part of the study is that the performance of the protocol representation accounts for the expected system behavior. Hence the system simulation could be successfully used to reveal essential protocol behaviors resulting from potential shifts in the crypto market environment and to optimize the protocol's hyper parameters.

Keywords: DeFi protocol simulation; crypto market; risk; historical data

1. Introduction

Decentralized finance (DeFi) has emerged as a transformative paradigm within the financial landscape, harnessing the capabilities of blockchain technology to reshape traditional financial services. As the DeFi space continues to evolve rapidly, there is a growing need to comprehensively understand and assess the performance of these protocols in dynamic and unpredictable market conditions.

In a space where innovation and experimentation abound, the ability to simulate and analyze the behavior of DeFi protocols becomes paramount for both developers and participants. One of the key challenges lies in predicting how these protocols will perform under varying market scenarios, which are inherently influenced by the volatile nature of the cryptocurrency market.

Against the backdrop of dynamic cryptocurrency trading, a myriad of platforms has mushroomed globally, each contributing to the evolving narrative of DeFi. These platforms, while presenting distinct advantages and limitations, share a common thread: the direct connection of economic entities for mutual gains in cryptocurrency trading.

Prominent actors in the DeFi lending space include Uniswap, Sushiswap, Compound, and Aave. Despite their progress, challenges persist, ranging from scalability issues, to elevated transaction costs and barriers, and the seamless onboarding of new users. The current state of research in DeFi lending protocols reflects a robust exploration of facets such as smart contract architecture, token mechanics, governance models, and user behavior.

Uniswap, operating on the Ethereum blockchain, utilizes an Automated Market Maker (AMM) model, promoting trustless and permissionless token trading. Researchers



Citation: Grigorova, Iveta, Aleksandar Karamfilov, Radostin Merakov, and Aleksandar Efremov. 2024. Simulation of Dynamic Performance of DeFi Protocol Based on Historical Crypto Market Behavior. *Risks* 12: 3. https://doi.org/10.3390/ risks12010003

Academic Editors: Eulália Mota Santos and Margarida Freitas Oliveira

Received: 17 November 2023 Revised: 15 December 2023 Accepted: 19 December 2023 Published: 25 December 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). (Adams et al. 2021; Angeris et al. 2019; Daian et al. 2019) have scrutinized its pricing mechanism, liquidity provision strategy, security vulnerabilities, and optimization strategies.

Sushiswap, born as a decentralized exchange (DEX) and AMM, distinguishes itself by amplifying liquidity and incentives for liquidity providers (LPs). Research (Fan et al. 2023) undertakings delve into the intricacies of liquidity mining incentives, the sway of economic factors on liquidity providers' decision making, and strategies to alleviate impermanent loss.

Compound, a decentralized lending protocol, revolutionizes DeFi by enabling users to lend or borrow based on algorithmic interest rates. Researchers (Saengchote 2023; Kao et al. 2020) have evaluated its market risks, liquidation mechanism, and economic security through agent-based modeling and simulation.

Furthermore, a nuanced exploration branches into the broader financial ecosystem, with a focus on three prominent Peer-to-Liquid Funds (PLFs)—Aave, dYdX, and Compound, presented in (Gudgeon et al. 2020). Market liquidity, efficiency, and interdependence among these PLFs take center stage, unraveling operational patterns during periods of high utilization and the potential risks associated with liquidity concentration.

Aave, akin to Compound, stands as a decentralized lending and borrowing platform with a distinctive feature—flash loans. The founder's whitepaper (Aave n.d.) provides a comprehensive elucidation of Aave's innovative lending pool model, flash loans, and governance structure. The research (Sun et al. 2023) threads weave through topics such as risk management, interest rate models, and liquidity provision, offering valuable insights into the multifaceted dynamics of the Aave protocol.

The cryptocurrency market is renowned for its inherent volatility, characterized by rapid and sometimes unpredictable price fluctuations across various digital assets. This volatility is a key distinguishing feature of the crypto market and is influenced by a myriad of factors. In (Gupta and Chaudhary 2022), the authors empirically found that there is the presence of high volatility among the returns of four cryptocurrencies—Bitcoin, Ether, Litecoin, and XRP—which makes them very risky assets for investments.

One of the primary contributors to cryptocurrency market volatility is the market's relative nascency. Compared to traditional financial markets, the crypto space is relatively young and less mature. As a result, it tends to be more susceptible to sudden market sentiment shifts, speculative trading, and external influences. In a period where positive sentiment regarding the adoption of cryptocurrencies becomes widespread, financial institutions and large corporations increasingly integrate cryptocurrencies into their operations and investment portfolios. Increased adoption by traditional financial players can significantly boost the credibility and acceptance of DeFi platforms. Positive market sentiment often correlates with increased demand and can positively influence the valuation of cryptocurrencies.

Other factors affecting the volatility of cryptocurrencies are the regulatory developments. A global push among major economies for clearer and more favorable cryptocurrency regulations may introduce regulatory frameworks that legitimize and provide clear guidelines for the operation of decentralized finance (DeFi) platforms. This regulatory clarity could positively impact investor confidence, potentially leading to increased participation in platforms such as NOLUS. Conversely, adverse regulatory developments could introduce uncertainty and hinder growth.

The decentralized nature of many cryptocurrencies also plays a role. Without a central governing authority, market movements can be more reactive to news, social media trends, or macroeconomic factors. In terms of a global economic downturn, traditional financial markets experience a recession, and central banks implement expansive monetary policies to counter economic challenges. Economic downturns might drive investors toward alternative assets, including cryptocurrencies, as a hedge against economic uncertainties. This shift could influence the demand for cryptocurrencies and impact their pricing dynamics.

Liquidity, or the ease with which an asset can be bought or sold, is another factor amplifying crypto market volatility. Low liquidity in certain markets can result in larger price swings, as a relatively small volume of trades can have a significant impact.

Moreover, significant technological advancements in blockchain and decentralized technologies also affect the volatility of crypto markets. This could include the widespread adoption of more efficient consensus algorithms, scalability solutions, or interoperability protocols. Technological progress can enhance the efficiency and functionality of DeFi platforms. If these platforms integrate these advancements, they might gain a competitive edge, attracting more users and positively affecting their valuation.

In recent years, the crypto market has witnessed efforts to mitigate volatility. The introduction of stablecoins pegged to traditional fiat currencies, algorithmic stablecoins, and decentralized finance (DeFi) protocols seeking to stabilize token values are examples of such endeavors. However, volatility remains a characteristic element of the crypto landscape, offering both opportunities and challenges for investors and traders.

The novelty of our approach rests on the integration of historical crypto market data into the simulation of a newly developed protocol performance. By leveraging past market behaviors, we create a realistic environment that captures the complexities of real-world conditions, enabling us to evaluate the performance, risks, and potential outcomes of the real protocol with greater precision.

The subsequent sections of this paper focus on the details of our simulation methodology, the data sources and variables used, and the results of our analyses. Through researching the dynamic performance of the presented protocol using historical market simulations, we contribute to the ongoing discourse surrounding the evolution of DeFi, its potential challenges, and the opportunities it presents within the ever-evolving landscape of decentralized finance.

The Supplementary Materials section of the article provides an address that directs to the repository where the actual model of the real platform is published. This facilitates the ability of other researchers to replicate and build upon the presented results.

2. Mechanisms for Implementation of a DeFi Protocol Simulation

2.1. Event: Interest Generation

2.1.1. Lessees Interest (LS Interest)

The interest rate definition plays a crucial role in DeFi protocols as it directly impacts the functioning and effectiveness of these platforms. The accurate and well-defined determination of interest rates is essential for maintaining stability, attracting users, and ensuring efficient capital allocation within the DeFi ecosystem.

One of the key aspects of DeFi protocols is lending and borrowing activities, where users provide or borrow funds from the platform. The interest rate serves as a fundamental mechanism for incentivizing participation, balancing supply and demand, and managing risk.

A clear and transparent interest rate definition is important for several reasons. Firstly, it ensures fairness and equal treatment of all participants. By having a standardized and objective method of calculating interest rates, users can trust that their loans or deposits are subject to consistent and predictable terms.

Secondly, the interest rate definition helps to maintain stability and mitigate risks within the DeFi ecosystem. Through careful consideration of market conditions, collateralization ratios, and other factors, protocols can establish interest rates that align with the underlying assets' risk profiles. This helps to protect the platform and its users from excessive volatility and potential defaults.

Moreover, the interest rate definition enables efficient capital allocation by incentivizing borrowing or lending activities when they are most needed. By dynamically adjusting interest rates based on supply and demand dynamics, protocols can influence user behavior and optimize resource allocation. Furthermore, the interest rate definition in DeFi protocols is an area of active research and innovation. It involves considerations of economic theories, game theory, and algorithmic mechanisms. Researchers and developers are constantly exploring novel approaches to improve interest rate models, liquidity provision, and risk management to enhance the overall efficiency and sustainability of DeFi platforms. In (Gudgeon et al. 2020), the authors presented evidence of the inefficiency of DeFi tokens regarding their liquidity, market efficiency, and interest rates. They investigate the mechanisms used to set the interest rates in several DeFi protocols in order to find how to equilibrate the supply and demand for funds.

In (Piñeiro-Chousa et al. 2022), the authors investigate the impact of certain traditional assets on specific DeFi tokens, focusing on their volatility. The study suggests that there are connections between traditional assets, user-generated content, and DeFi tokens. These relationships indicate that DeFi tokens may serve as safe havens and provide a hedge against stock market volatility.

The value of LS Interest due on the smart contracts in the NOLUS platform is calculated daily and it consists of two components: a predefined Base Interest Rate and an additional part dependent on the Utilization Rate of the relevant pool.

The first component—the Base Interest Rate—is set for the platform and it serves as a minimum return that is guaranteed throughout the entire duration of the smart contract, except for the cases in which conditions for liquidation occur.

The second component—the Utilization Rate—is calculated daily from the ratio between the total amount of assets in stable currency already borrowed from the respective pool and the value of the deposited amounts in stable together with the accumulated interest in stable. It represents the part of the assets from each pool that are borrowed from the LSs.

For calculation of the Utilization Rate on a daily basis for every pool, the following formula is used:

$$U = \frac{\sum B_{stb}^{r}}{TVL^{t}},\tag{1}$$

where

- *U*—Utilization Rate per pool;
- $\sum B_{stb}^t$ —the total amount borrowed from the pool's native asset in stable for a specified timestamp;
- *TVL^t*—the Total Value Locked (TVL) amount of the pool's native asset in stable for a specified time.

The TVL is further discussed in Section 2.3 of this article.

At the end of each day, when the smart contracts are closed, the Utilization Rate for each pool is calculated, and the following day, this Utilization Rate is taken for calculation of the LS interest rate for the current day.

In case at any day a 100% Utilization Rate is reached for a specific pool, no contracts are opened on that day and the day is restarted.

The interest rate for each individual pool, which represents the interest rate for the LS contracts, is calculated as per the formulas below, according to the level of utilization of the respective pool:

(1) when the Utilization Rate for the pool is smaller than or equal to the Optimal Utilization Rate, i.e., $U \le U_{optimal}$:

$$R_v = R_{vo} + \frac{U}{U_{optimal}} \cdot R_{slope1},$$
(2)

(2) when the Utilization Rate for the pool is higher than the Optimal Utilization Rate, i.e., $U > U_{optimal}$:

$$R_v = R_{vo} + R_{slope1} + \frac{U - U_{optimal}}{1 - U_{optimal}} \cdot R_{slope2},$$
(3)

where

- R_v —a variable interest rate per pool;
- *R_{vo}*—a predefined Base Interest Rate;
- *U*—the Utilization Rate;
- *U_{optimal}*—a predefined Optimal Utilization Rate for the stable pool;
- R_{slope1} —a predefined parameter, showing the increasing of the interest rate when $U \leq U_{optimal}$;
- R_{slope2} —a predefined parameter, showing the increasing of the interest rate when $U > U_{optimal}$.

This approach can be found also in the Aave protocol (Aave n.d.).

2.1.2. LP Interest

When a new liquidity provider (LP) deposit is opened, the amount of the deposit is stored both as an asset and is recalculated in stablecoins. The LP interest rate for a particular timestamp is calculated on a daily basis for each separate pool, according to the formula:

$$R_{LP}^t \% = \frac{R_v^t - R_{tr}}{365} \cdot 100, \tag{4}$$

where

- *R^t_{LP}*—the LP interest rate for a particular timestamp;
- R_v^t —a variable interest rate per pool for a particular timestamp;
- *R*_{tr}—a predefined Treasury Interest Rate.

At the time of withdrawal of the deposit from the LP, the cumulative amount of each LP's interest is paid to them, together with the amount of their deposit and the additional profit in the form of NOLUS tokens (NLSs).

An important point to mention is a specific characteristic of the platform, regarding the number of LS contracts that will be opened daily, in reference to the change in the interest rate. The number of newly opened LS contracts depends on the interest rate and the Utilization Rate for the particular day.

The number of LS contracts that will be opened for the respective day decreases when the interest rate increases.

This functionality is achieved using the following formula:

$$dem = \frac{e^{int - int_{min}} - e^{int_{max} - int_{min}}}{1 - e^{int_{max} - int_{min}}} \cdot (demint_{min} - demint_{max}) + demint_{max}, \tag{5}$$

where

- *dem*—percentage demand;
- *int_{min}*—minimal LS interest;
- *int_{max}*—maximal LS interest;
- *demint_{max}*—the maximum value of the variation in the parameter demand when forming the corresponding structure;
- *demint_{min}*—the minimum value of the variation in the parameter demand when forming the corresponding structure.

It is foreseen that there is a predefined maximum allowed LS interest rate. The parameter "dem" is calculated, which shows the percentage of the contracts for the corresponding day that will be opened depending on the interest rate calculated for the particular day. This functionality demonstrates that the interest of the LSs can decrease depending on the increase in the interest rates for the new contracts.

The maximum Pool Utilization, predefined in the config file, is used for LS demand reduction. If the maximum Pool Utilization is reached, the staked daily counts of new contracts are multiplied by weight for the purpose of reduction.

2.2. Event: Repayment of Loans by LSs

In the context of smart contracts, repayment refers to the process of returning borrowed assets to the lending platform after a borrower has completed their loan term.

In (Carapella et al. 2022), the authors provide an overview of blockchain basics and the major DeFi protocols and products they offer, together with the mechanisms for repayment of smart contracts, and finally they cover potential risks associated with these technologies and use cases. In (Gudgeon et al. 2020), the authors conduct an empirical analysis to observe the behavior of interest rate rules since their introduction, focusing on their responses to various levels of liquidity.

There are three types of liquidation transactions in NOLUS protocol, regarding the payment of loans by the LSs:

- Regular payment or liquidation due to penalty;
- Partial liquidation;
- Total liquidation.

The first type of liquidation refers to the regular payments made by the LSs to repay their obligations under the smart contracts. These regular payments are executed without the need for manual intervention once the smart contract has been deployed on the blockchain. Regular payments in smart contracts provide several advantages, including reducing administrative overhead, enhancing trust and transparency, and ensuring timely and accurate payments.

The partial liquidation mechanism is crucial for maintaining the stability and safety of the protocol. It helps protect lenders and maintain the overall health of the system by preventing excessive losses in the event of volatile market conditions or borrower defaults. Partial liquidation refers to the reduction in a given asset of the LS as a result of one of the following two events: omitted payments or a shake-up in market conditions. When partial liquidation occurs, the smart contract will automatically sell or auction off a portion of the user's collateral to cover their outstanding debt and ensure that the protocol remains solvent. By liquidating only a part of the collateral, users have the opportunity to restore their position and prevent the complete loss of their assets.

Full liquidation refers to the process of completely liquidating a user's collateral in the smart contract when they are unable to meet their debt obligations or the value of their collateral falls below a critical threshold. Full liquidation is an essential risk management mechanism in the protocol, as it helps to mitigate the impact of defaulting borrowers and ensures the stability and security of the system. By fully liquidating a user's collateral, the protocol can recover the debt owed and maintain the integrity of its reserve funds, which are critical for providing liquidity and maintaining the overall health of the platform.

There are some assumptions regarding the liquidation of contracts in the NOLUS platform that should be mentioned, as follows:

$$LA_a^{t=0} = B_a^{t=0} + C_a^{t=0}, (6)$$

where:

- $LA_a^{t=0}$ —the value of the LSs loan (loan amount) at the conclusion of the contract in asset;
- C^{t=0}_a—the collateral (downpayment) in asset provided by LSs upon conclusion of the smart contract;
- $B_a^{t=0}$ —the borrowed amount of asset from the LSs.

$$LA_{stb}^{t} = LA_{a}^{t} \cdot MP_{a}^{t}, (7)$$

where:

- LA_{sth}^{t} —the value of the LSs loan at a given time in stable;
- MP_a^t —the market price of the particular asset for the current timestamp.

All obligations due on the contract at a given time in stable, LL_{stb}^{t} , can be calculated as follows:

$$LL_{stb}^{t} = LSP_{stb}^{t} + LSI_{stb}^{t}, aga{8}$$

where

- *LL*^{*t*}_{*stb*}—all obligations due on the contract at a given time in stable;
- LSP_{stb}^{t} —the principal due on the LS contract at a given time in stable;
- LSI_{stb}^{t} —the LS interest due on the contract at a given time in stable.

In turn, the following formula is used to determine the LSP_{stb}^{t} :

$$LSP_{stb}^{t} = B_a^{t=0} \cdot MP_a^{t=0} - PP_{stb}^{t}, \tag{9}$$

where

- $MP_a^{t=0}$ —the market price of the relevant asset at the time of opening the smart contract;
- PP_{stb}^{t} —the already paid part of the principal.

The value of LSI_{stb}^{t} indicates the current interest standing at any given time. In the Monte Carlo simulation, the interest is used as the standing interest for the next month (example: when a contract is opened initially, LSI_{stb}^{t} is set as the value of the interest for 1 month).

$$LL_{stb}^t \% = \frac{B}{C} \cdot 100, \tag{10}$$

where:

- *LL*^t_{stb}—Lease Liability: ratio between borrowed to collateral at a given time.
 There are three parameters, regarding the Lease Liability:
- *LL_{init}*—Lease Initial Liability: a predefined initial Lease Liability in %;
- *LL_{health}*—Lease Healthy Liability: a predefined healthy Lease Liability in %;
- *LL_{max}*—Lease Max Liability: a predefined maximum Lease Liability in %.

It is foreseen that the LL_{stb}^t for every opened contract is recalculated daily. The LL_{stb}^t of the contracts should be smaller than or equal to LL_{health} , and LL_{health} is smaller than LL_{max} , to ensure that the protocol will work optimally and that no shortage of funds will occur in the pools.

2.2.1. Liquidation Type 1: Regular Payment/Liquidation Due to Penalty

It is expected that in the majority of cases, the contracts in the platform will be paid out in a timely manner; hence, there will be no penalty payments. In these cases, for the regular repayment of every smart contract, the following formula is used:

$$RA_{stb} = (B_{stb} + B_{stb} \cdot \frac{R_v}{12 \times 100} \cdot N_{pmnt}) / (N_{pmnt} + M_{pnl})), \tag{11}$$

where

- *RA_{stb}*—the amount of the regular payment (repayment amount);
- *B*_{stb}—the principal of the contract together with the newly achieved asset in stable;
- *N_{pmnt}*—the number of regular payments under the contract (payment);
- *M_{vnl}*—the number of omitted payments under the contract (penalty).

For calculation of the omitted payments the following formula is used:

$$PA_{stb} = B_{stb} \cdot \frac{R_v}{12 \times 100} \cdot M_{pmnt} / (N_{pmnt} + M_{pnl}), \qquad (12)$$

where

PA_{stb}—the amount of the omitted (penalty) payment.

As mentioned, there are two conditions for a partial or a full liquidation to occur: missed contract payments or market shocks. When any of these events occurs and a set depreciation threshold is reached, part of the LS's asset is sold in order to bring it to a sustainable state.

At the opening of the contracts for the current day, LL_{stb}^{t} for each contract is calculated and the following rule applies:

If $LL_{stb}^t \ge \frac{LL_{max}}{100} \cdot LA_{stb}^t$, it indicates that $LL_{stb}^t \ge LL_{max}$; then, the liquidation amount in stable at the current moment (LQA_{stb}^t) is calculated and the amount is stored in a specific structure of the protocol.

The following formula is used to define LQA_{stb}^{t} and the smaller of the following two values is taken:

$$LQA_{stb}^{t} = min(LA_{stb}^{t}, \left(LL_{stb}^{t} - \frac{LL_{health}\%}{100} \cdot LA_{stb}^{t}\right) / (1 - \frac{LL_{health}\%}{100})),$$
(13)

where

 LQA^t_{stb}—liquidation amount for a given time in stable: the amount in stable by which the LA^t_{stb} should be reduced after the partial liquidation for the current time.

2.2.2. Liquidation Type 2: Full Liquidation

If $LQA_{stb}^t = LA_{stb}^t$, then a full liquidation is enforced.

After all liquidations past time for the totally liquidated contracts are removed, a liquidation record is created and a closing record for each fully liquidated contract is created.

2.2.3. Liquidation Type 3: Partial Liquidation

If $LQA_{stb}^t < LA_{stb}^t$, then a partial liquidation is enforced. LA_{stb}^t after the partial liquidation is calculated as follows:

$$LA_{stb}^t = LA_{stb}^{t-1} - LQA_{stb}^t, (14)$$

After that, LSI_{stb}^t , representing the current interest standing at any given time, is also reduced in compliance with the new parameters of the smart contract.

2.3. Event: Total Value Locked

Total Value Locked is a crucial metric in the decentralized finance (DeFi) ecosystem, representing the total amount of assets locked in a specific protocol or across multiple protocols. It provides valuable insights into the protocol's performance and helps investors, users, and the wider community make informed decisions in the rapidly evolving DeFi landscape. According to the empirical evidence presented by (Maouchi et al. 2022), TVL can be used to evaluate DeFi tokens and monitor the market, and can be used as an indicator of the growth and success of that market (Soiman et al. 2022).

In (Saengchote 2021), the author provides insights into the potential significance of TVL in DeFi and emphasizes the intricate nature of DeFi analysis and market monitoring. The author highlights that TVL is determined by the market value of tokens that are deposited or locked within the system, which inherently makes it sensitive to token prices. Consequently, there is an anticipated notable correlation between this variable and valuations. The study of (Metelski and Sobieraj 2022) aims to assess key DeFi performance metrics, particularly focusing on protocol valuations. Through a quantitative analysis of 30 selected protocols

across decentralized exchanges, lending, and asset management classes, the research explores relationships between protocol valuations and variables such as Total Value Locked (TVL), protocol revenue, total revenue, gross merchandise volume, and the inflation factor. Utilizing Granger causality tests and fixed effects panel regression models, the study reveals that DeFi protocol valuations exhibit dependencies on various performance measures, with distinct magnitudes and directions for different variables.

The significance of TVL can be found in several directions:

- Measure of protocol adoption: TVL provides a measure of the level of adoption and acceptance of a DeFi protocol. A higher TVL indicates that more users are trusting and utilize the protocol to lock their assets, which demonstrates confidence in its security, functionality, and potential returns.
- Indicator of liquidity: TVL serves as an indicator of the liquidity available within a protocol. Higher TVL implies a deeper pool of assets, which enhances trading efficiency, reduces slippage, and attracts more participants to engage in transactions.
- Market perception and reputation: TVL plays a role in shaping the market perception and reputation of the DeFi protocol. A protocol with a substantial TVL is often seen as more reputable, reliable, and trustworthy by users, investors, and the wider DeFi community. This can attract further adoption and partnerships, enhancing the protocol's long-term sustainability.
- Comparative analysis: TVL allows for comparative analysis among different DeFi
 protocols. By assessing TVL across various platforms, investors and users can evaluate
 the relative popularity and growth potential of protocols, aiding in investment decision
 making and identifying emerging trends in the DeFi space.
- Protocol governance and influence: TVL often influences the governance power within
 a DeFi protocol. Some protocols allocate voting power based on the amount of assets
 locked, giving higher TVL participants a stronger voice in decision-making processes.
 This aligns the interests of token holders with the protocol's development and ensures
 the protocol evolves in a way that benefits the majority of participants.

The definition of TVL in the protocol is the following:

$$TVL^{t} = CDA_{stb}^{t} - CW_{stb}^{t} + CLSI_{stb}^{t},$$
(15)

where

- *CDA*^{*t*}_{*stb*}—the cumulative deposited amount in the platform for the current time in stable;
- *CW*^{*t*}_{*stb*}—the total amount of withdrawals of LPs from the platform for the current time, together with the corresponding LP interest;
- $CLSI_{stb}^{t}$ —the accumulated interest from LSs in the platform for the current time.

2.4. Event: NLS Price Definition

The price of the NLS tokens should be simulated as there is no historical information. It is formed of two components: the market price of the NLS and the impact of the protocol's performance on the price. These two components are taken in a specified proportion and the final price of NLS is calculated.

The first component is a representation of the historical price of the currencies that participate in the platform, taken in a specific proportion.

The second component that participates in the formation of the final price of NLS tokens is the work of the protocol itself. To determine the influence of the platform on the level of the price of NLS tokens, the Total Value Locked is used.

This process is implemented in the following steps:

1. The ratio between the current value of all assets locked in the platform, calculated in stablecoins, and the total value of assets from the previous moment, is determined as follows:

$$\Delta TVL \% = \frac{TVL^{t-1}}{TVL^t},\tag{16}$$

where

- *TVL^t*—the TVL of the pool's native asset, calculated in stable for a specific time;
- TVL^{t-1} —the TVL of the pool's native asset, calculated in stable for a previous time.
- 2. The proportion in which the two components will participate in forming the final price of the currency is determined.
- 3. The final price, taking into account both the market price and the impact of the platform over the price of the NOLUS tokens, is calculated according to the formula:

$$P_{nls}^{t} = MP_{nls} \cdot w_m + P_{nls}^{t-1} \cdot (1 - w_m) \cdot \Delta TVL \ \%, \tag{17}$$

where

- P_{nls}^t —the price of NLS tokens for the current timestamp;
- *MP_{nls}*—the market price of NLS tokens;
- w_m —the weight of the market price;
- $(1 w_m)$ —the weight of the platform;
- P_{nls}^{t-1} —the price of NLS tokens for the previous timestamp.

2.5. LP Rewards Distribution

The Treasury is a part of the protocol where the profit from the work of the protocol is stored. In the beginning it is provided that there is an initial investment in the Treasury in the form of NLS tokens, which will ensure the necessity of having some funds in case any LP withdraws their deposit earlier, when the profit of the Treasury is not enough.

There are three cash inflows into the Treasury:

1. Transaction fees, paid by both LSs and LPs, charged for each transaction

For the purposes of the simulation of the protocol, it is assumed that each LS and LP will pay a predefined number of transactions every month, so the total value of all transaction fees for the platform per month is calculated as follows:

$$TRNF = c_{ls} \cdot TR_{ls} \cdot P_{trn} + c_{lp} \cdot TR_{lp} \cdot P_{trn}, \qquad (18)$$

where

- TRNF—total transaction fees;
- *TR_{ls}*—the number of transactions made by each LS per month, predefined in the config file;
- TR_{lp} —the number of transactions made by each LP per month, predefined in the config file;
- *c*_{*ls*}—the number of opened LS contracts per month;
- c_{lv} —the number of opened LP deposits per month;
- *P*_{trn}—the transaction price for each transaction in stable, predefined in the config file.
- 2. SWAP fees—a one-time fee paid by each LS upon conclusion of the smart contract, predefined in the config file as a percent from the LS loan amount in stable:

$$SWAP = LA_{sth}^{t=0} \cdot sw, \tag{19}$$

where

- $LA_{stb}^{t=0}$ —LS loan amount in stable at the opening of the contract;
- *sw*—a predefined percent in the config.
- 3. Treasury interest (TRI)—A predefined part of the interest paid by LSs

The cash outflow from the Treasury, which leads to a decrease in the accumulated funds in the Treasury, is the distribution of rewards to the LPs. These rewards are distributed among all LPs based on their percentage contribution to the pools on a daily basis.

The total amount of rewards per pool that are distributed among all LPs for a particular pool daily due to the initially deposited amount by each LP is calculated as follows:

$$TRR_{stb}^{t} = \frac{TVL_{pool}^{t} \cdot trw}{365},$$
(20)

where

- *TRR*^{*t*}_{*stb*}—Treasury rewards: the total amount of rewards per pool distributed among all LPs on a daily basis;
- *trw*—preliminary defined percentages of rewards per pool;
- TVL_{vool}^{t} —TVL per pool at the current time of the spread.

trw represents preliminary defined percentages that determine the relationship between the amount of TVL for all pools and the percentage of rewards for LPs. This relationship indicates the percentage of the TVL that will be distributed among the LPs. This value is extracted from the available funds in the Treasury and used for buyback of NLS tokens that will be spread among the LPs through rewards.

The rewards that each LP will receive from the Treasury daily are calculated as a percentage of all rewards that will be distributed among all LPs from the respective pool for the current day. This percentage depends on the amount of the deposit with which the respective LP participates in the specific pool, relative to the total deposited amount of all contracts for the current time for the pool in stable. After the individual percent for each LP is calculated, the amount of the rewards for the current day for each LP is determined.

In this way, the total cash inflow and outflow from the work of the protocol is stored in the Treasury and can be represented as follows:

$$TRP_{stb}^{t} = TRNF_{stb}^{t} + SWAP_{stb}^{t} + TRI_{stb}^{t} - TRRD_{stb}^{t},$$
(21)

where

- *TRP*^{*t*}_{*stb*}—Treasury profit, being the amount entered in the Treasury on a daily basis in stable;
- *TRNF*^t_{sth}—total transaction fees on a daily basis in stable;
- *SWAP*^{*t*}_{*stb*}—SWAP fees on a daily basis in stable;
- *TRI*^{*t*}_{*sth*}—the Treasury interest on a daily basis in stable;
- *TRRD*^{*t*}_{*stb*}—Treasury rewards in stable that should be distributed among all LPs on a daily basis in stable.

The current state of the Treasury is recorded in stable in a specific structure, namely, NLS, which appears to be a cumulative table indicating the available funds in the Treasury at the current time, according to the following formula:

$$TRA_{stb}^{t} = TRA_{stb}^{t-1} + TRP_{stb}^{t} - TRRD_{stb}^{t-1}$$
(22)

An important feature of the platform, related to the distribution of rewards to the LPs from the protocol, is the established mechanism, which ensures that, if for a specific day there is not a sufficient number of tokens in the Treasury to be distributed among all LPs, then for the corresponding day no rewards are distributed.

3. Dynamic Performance Simulation Based on Historical Crypto Market Behavior

3.1. Research Methodology

Expanding the model detailed in the previous section, a comprehensive simulation of the NOLUS protocol is conducted, utilizing the actual protocol as a basis. This simulation spans a historical timeframe of three years, from 1 February 2020 to 31 January 2023,

capturing a diverse range of market conditions. In this simulation, the pricing dynamics of the NOLUS protocol currency are intricately tied to various currencies, with differential weights assigned based on their influence in the broader crypto market. Notably, primary currencies, particularly wrapped bitcoin, wield substantial weight due to their pronounced impact on the overall market.

During the specified period, the prices of multiple currencies were considered, each contributing to the formation of the presumed value of the NOLUS protocol currency. The proportional representation of these currencies in the simulation, presented in Table 1, holds significance, reflecting their varying impacts on the NOLUS protocol's valuation.

Currency	Proportion in the Final Price
wrapped-bitcoin	0.250
weth	0.150
binancecoin	0.100
avalanche-2	0.060
cosmos	0.050
injective-protocol	0.050
band-protocol	0.050
kava	0.040
juno-network	0.040
osmosis	0.040
secret	0.040
akash-network	0.040
crypto-com-chain	0.030
stargaze	0.030
sentinel	0.030

Table 1. Participation of different cryptocurrencies in forming the market price of NOLUS tokens.

In order to simulate the expected behavior of the market for carrying out the present simulation, hypotheses are also set for a number of variables, including the number of participants in the platform, the duration and size of loan contracts, the duration and size of deposits, and the number and size of delays of contract payments. The F-distribution is used to generate these variables, as it best represents the expectations for the distribution of their values.

Each of these variables is set to be a hyper parameter of the simulation model, enabling researchers to fine tune their values. This also provides the opportunity to simulate various environmental effects that impact the work of the platform, on one hand, and to modify the parameters of the platform itself. This model makes it possible to simulate structural changes or regime shift type of behavior. Based on these assumptions, the anticipated performance of the protocol during the simulated period is presented in the following subsections of the article.

3.2. Simulation of Dynamic Operational Performance

This section provides an exposition of the simulation outcomes based on the initially defined parameters and presumed hypotheses. The visual representations in the form of graphs for each variable offer a clear demonstration that the anticipated interaction mechanisms among distinct variables accurately align with the preconceived expectations.

3.2.1. Interest Rates

The interest rate within the platform is dynamic, contingent upon the prevailing state of the platform and market conditions at each timestamp. With a surge in the number of LPs, a substantial reservoir of available funds emerges in the pools, consequently mitigating the expense for new LSs seeking to borrow assets from the platform. This results in a decreased interest rate for these newly introduced LSs, leading to a corresponding reduction in the interest rates earned by recently enrolled LPs. From Figure 1, it can be seen that the expected relationship between LP and LS interest rates is satisfied and they are change correctly depending on the Pool Utilization Rate. As the Utilization Rate per pool increases, the interest rates for both LSs and LPs also increase.



LP / LS interest depending on Pool Utilization



In Figure 2, it can be seen that the established mechanism for changing the number of LP and LS contracts depending on the change in the interest rate functions properly. It is noticeable that over time and with a smooth and continuous increase in interest rates, the number of LPs exceeds the number of LSs. On the one hand, an increase in the interest rates leads to an increase in the number of LPs over time, as increased interest rates attract more and more LPs. At the same time, rising interest rates cause a slight decline in LS contracts, as the cost of borrowing becomes higher at higher interest rates.

In the current simulation, the maximum utilization of the funds from the pools is not reached and therefore there the opening of new contracts does not terminate. In subsequent articles, this functionality will be demonstrated by running various MC simulations.



Count of contracts depending on Interest growth

Figure 2. Count of LS and LP contracts depending on interest rate.

3.2.2. Repayment of Loans

The repayment process in smart contracts is usually automated and executed based on the terms agreed upon at the time of borrowing. Once the borrower fulfills their repayment obligations, the smart contract releases the collateral held as security back to the borrower, and the loan is considered fully repaid.

Regular Payments

Figure 3 shows an example of a smart contract with only regular payments, demonstrating that the underlying mechanisms in the protocol are met. This particular contract has five regular payments. This can be seen on the bottom of Figure 3, represented by Liquidation Type: 1, and it indicates that the Lease Liability decreases in accordance with the repayments made. Since the ratio between the obligations under the contract and the value of the contract does not exceed the healthy levels, partial liquidation (in the sense of penalty payments) and full liquidation are therefore not required. Moreover, the market price of the asset is consistent, which indicates that in stable or favorable market conditions, penalty liquidations are not common or necessary. At the end of the period, the contract is fully repaid.



Figure 3. Repayment of loans. An example of regular payments.

Partial and Full Liquidation

Figure 4 shows an example of a contract in which events occur that lead initially to a partial liquidation and subsequently to a full liquidation of the contract. It is noticed that on the 13 March 2020 there is a collapse in the market, due to which the loan amount loses its monetary value (lease amount), pushing the smart contract close to the preliminarily determined critical healthy levels. Therefore, a partial liquidation occurs, represented by Liquidation Type: 3, as part of the smart contract is liquidated, reducing the liability of the smart contract back to healthy levels. The liquidation amount is determined by Liquidation Amount (formula number 13). Subsequently, on 15 March 2020 the conditions for partial liquidation occur again and a second partial liquidation is fulfilled. As a result, the conditions for full liquidation also occur and on 16 March 2020 the contract is fully liquidated (represented by Liquidation Type: 2).







Figure 4. Repayment of loans. An example of partial and full liquidation.

3.2.3. Total Value Locked

Within the NOLUS protocol, TVL serves as a determinant influencing both the NLS price and the platform's interest rate. The anticipation is that a higher value of funds locked in the platform enhances the protocol's appeal. Consequently, this is anticipated to result in elevated interest rates and a simultaneous increase in the value of NLS tokens.

Figure 5 shows the state of the platform in terms of funds locked in the protocol for the simulated period. It is noticed that in the beginning of the period there is an increase in TVL because of the expected enhanced interest in the protocol, and subsequently, the levels are established at a plateau, with a slight increase over time.

In subsequent articles, various Monte Carlo simulations will be carried out, through which it is expected to determine how TVL will change under various assumptions, both in terms of conditions external to the platform, such as economic conditions and the financial services market in particular, and those internal to the platform, such as the number of participants and interest rates.

LS Total Borrowed, LP Total Deposited and TVL by Timestamp



Figure 5. Status of funds in the platform.

3.2.4. NLS Price Definition

The components of the NOLUS price, described in Section 2.4. Event: NLS Price Definition, are the simulated market price of the NLS and the impact of the protocol's performance.

Figure 6 presents the prices of cryptocurrencies that participate in the formation of the market component of NOLUS tokens for the selected period from 1 February 2020 to 31 January 2023, according to the proportions described in Section 3.2. Research Methodology.



Figure 6. Cryptocurrencies that participate in the formation of the NOLUS price.

Figure 7 presents the movement of the NLS price for the period of the simulation of the operation of the protocol.



Figure 7. NLS price by timestamp.

It is observed that the NOLUS price movement closely aligns with the collective movement of all considered currencies, with a notable emphasis on the movement of wrapped bitcoin. This emphasis is justified by the prominent role of wrapped bitcoin in the broader crypto market, acting as a key indicator and influencing factor. Wrapped bitcoin, being a representative token pegged to the value of bitcoin, holds a pivotal position, affecting market sentiments and contributing to the overall price dynamics of the NOLUS protocol.

3.2.5. Rewards Distribution

From Figure 8, it can be seen how the funds available in the Treasury, presented both in assets and in stable, change over time. It is noticed that at the beginning of the

protocol's work until about the middle of the second year, as a result of the large number of accumulated funds in the platform, rewards are distributed every day. After the middle of the second year, the level of funds in the Treasury is depleted due to distributing large rewards at the beginning, as a result of which such rewards are spread inconsistently every other day when enough funds are accumulated in the Treasury to cover the amount of all necessary rewards.



Figure 8. Treasury amount and Treasury rewards by timestamp.

Moreover, it is observed from Figure 9 that with the temporary increase in Treasury profit towards the middle of the third year, for a short period of time the funds in the Treasury are enough to cover all the required rewards; therefore, for a short period, the rewards are distributed daily.





4. Conclusions

In conclusion, the simulation of the NOLUS protocol presented in this paper marks a significant stride toward comprehending the intricate interplay between decentralized finance protocols and the broader crypto market landscape. By harnessing historical crypto market data, we have effectively simulated the dynamic behavior of the NOLUS protocol. The article explores several prominent DeFi protocols within the broader cryptocurrency market, with a specific focus on comparing the NOLUS platform, introduced in the article, to these established protocols. It provides an elaborate description of the developed software model that mimics the real protocol, delving into the core dependencies among its components. The article thoroughly examines the fundamental interdependencies among the various elements constituting the NOLUS protocol. Through a series of carefully conducted experiments, the platform's functionality is scrutinized, utilizing predetermined values for numerous simulation parameters. The outcomes of these experiments affirm that the constructed model effectively mirrors the intricacies of the actual platform, presenting a highly accurate representation.

This simulation methodology has the potential to uncover critical protocol behaviors caused by potential crypto market dynamics, identify optimization opportunities, and highlight best practices for adjustment of the key protocol hyper parameters. As the DeFi ecosystem continues to evolve, these insights will prove instrumental in fostering a robust and sustainable DeFi landscape.

The simulation presented in this article is a one-time implementation of the actual DeFi platform NOLUS. As it involves numerous random factors and is based on certain assumptions about their distribution, this singular simulation is constrained by the initial hypotheses. Concurrently, historical cryptocurrency price data spanning three years was employed to replicate the platform's behavior, intending to illustrate its hypothetical performance over that period. Recognizing these constraints, it is important to note that a detailed simulation of every operational aspect of the real platform might not be feasible. However, utilizing the developed model, which encapsulates the platform's embedded mechanisms, and conducting multiple simulations, could facilitate a more profound analysis of the protocol.

The implications of this research extend beyond the immediate scope of the study. Building upon the insights gained from this simulation, our intention is to employ a Monte Carlo simulation framework to explore a variety of different scenarios. This approach will allow us to systematically assess the protocol's performance across a spectrum of potential market events, encompassing periods of stability as well as heightened volatility. The deployment of multiple Monte Carlo simulations aligns with our goal to provide a more comprehensive and holistic understanding of the NOLUS protocol's behavior. Such simulations enable us to project its performance under conditions that mimic real-world market dynamics, serving as a valuable tool for risk assessment, strategic decision making, and scenario planning.

The subsequent stages of conducting multiple Monte Carlo simulations for diverse scenarios will be published in upcoming articles and hold the promise of enhancing our understanding of DeFi protocols' dynamic performance and fortifying their resilience against the complexities of the crypto market. As we venture forward, we are poised to contribute to the ongoing discourse on the intricacies of DeFi, while equipping stakeholders with the knowledge needed to navigate this innovative financial frontier.

Supplementary Materials: The model of the platform is published in GitHub: https://github.com/ nolus-protocol/mc-nolus (accessed on 20 December 2023).

Author Contributions: Conceptualization A.E.; data curation, A.K. and R.M.; methodology, A.E.; software, A.K. and R.M.; validation, A.K., I.G. and A.E.; funding acquisition, A.E.; formal analysis, A.K., I.G. and A.E.; investigation, A.K. and I.G.; writing—original draft preparation, I.G. and A.K.; writing—review and editing, A.E. and I.G.; visualization, A.K. and I.G.; supervision, A.E. All authors have read and agreed to the published version of the manuscript.

Funding: This study is financed by the European Union-NextGenerationEU, through the National Recovery and Resilience Plan of the Republic of Bulgaria, project No BG-RRP-2.004-0005.

Data Availability Statement: To generate the simulated price of NLS tokens, the real prices of the currencies mentioned in the article were used, gathered from the following web site: https://www.coingecko.com (accessed on 20 December 2023).

Conflicts of Interest: The authors declare no conflict of interest.

References

Aave. n.d. Available online: https://aave.com/ (accessed on 20 December 2023).

- Adams, Hayden, Noah Zinsmeister, Moody Salem, River Keefer, and Dan Robinson. 2021. Uniswap v3 Core. Available online: https://uniswap.org/whitepaper-v3.pdf (accessed on 20 December 2023).
- Angeris, Guillermo, Hsien-Tang Kao, Rei Chiang, Charlie Noyes, and Tarun Chitra. 2019. An analysis of Uniswap markets. *arXiv*. [CrossRef]
- Carapella, Francesca, Edward Dumas, Jacob Gerszten, Nathan Swem, and Larry D. Wall. 2022. Decentralized Finance (DeFi): Transformative Potential and Associated Risks. No. 2022-14. Atlanta: Federal Reserve Bank of Atlanta Policy Hub.
- Daian, Philip, Steven Goldfeder, Tyler Kell, Yunqi Li, Xueyuan Zhao, Iddo Bentov, Lorenz Breidenbach, and Ari Juels. 2019. Flash Boys 2.0: Frontrunning, Transaction Reordering, and Consensus Instability in Decentralized Exchanges. *arXiv*. [CrossRef]
- Fan, Sizheng, Tian Min, Xiao Wu, and Cai Wei. 2023. Towards understanding governance tokens in liquidity mining: A case study of decentralized exchanges. *World Wide Web* 26: 1181–200. [CrossRef]
- Gudgeon, Lewis, Sam Werner, Daniel Perez, and William J. Knottenbelt. 2020. DeFi Protocols for Loanable Funds: Interest Rates, Liquidity and Market Efficiency. Paper presented at 2nd ACM Conference on Advances in Financial Technologies, New York, NY, USA, October 21–23; pp. 92–112. [CrossRef]
- Gupta, Hemendra, and Rashmi Chaudhary. 2022. An Empirical Study of Volatility in Cryptocurrency Market. *Journal of Risk and Financial Management* 15: 513. [CrossRef]
- Kao, Hsien-Tang, Tarun Chitra, Rei Chiang, and John Morrow. 2020. An Analysis of the Market Risk to Participants in the Compound Protocol. Available online: https://api.semanticscholar.org/CorpusID:231619656 (accessed on 20 December 2023).
- Maouchi, Youcef, Lanouar Charfeddine, and Ghassen El Montasser. 2022. Understanding digital bubbles amidst the COVID-19 pandemic: Evidence from DeFi and NFTs. *Finance Research Letters* 47: 102584. [CrossRef]
- Metelski, Dominik, and Janusz Sobieraj. 2022. Decentralized Finance (DeFi) Projects: A Study of Key Performance Indicators in Terms of DeFi Protocols' Valuations. *International Journal of Financial Studies* 10: 108. [CrossRef]
- Piñeiro-Chousa, Juan, M. Ángeles López-Cabarcos, Aleksandar Sevic, and Isaac González-López. 2022. A preliminary assessment of the performance of DeFi cryptocurrencies in relation to other financial assets, volatility, and user-generated content. *Technological Forecasting and Social Change* 181: 121740. [CrossRef]
- Saengchote, Kanis. 2021. Where do DeFi stablecoins go? A closer look at what DeFi composability really means. *SSRN Electronic Journal* 2021: 3893487. Available online: https://papers.ssrn.com/sol3/papers.cfm?abstract_id=3893487 (accessed on 20 December 2023). [CrossRef]
- Saengchote, Kanis. 2023. Decentralized lending and its users: Insights from Compound. *arXiv* arXiv:2212.05734. Available online: http://arxiv.org/abs/2212.05734 (accessed on 31 October 2023). [CrossRef]
- Şoiman, Florentina, Guillaume Dumas, and Sonia Jimenez-Garces. 2022. The return of (I)DeFiX. arXiv arXiv:2204.00251. Available online: http://arxiv.org/abs/2204.00251 (accessed on 8 November 2023).
- Sun, Xiaotong, Charalampos Stasinakis, and Georgios Sermpinis. 2023. Liquidity Risks in Lending Protocols: Evidence from Aave Protocol. *arXiv* arXiv:2206.11973. Available online: http://arxiv.org/abs/2206.11973 (accessed on 31 October 2023).

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.

MDPI AG Grosspeteranlage 5 4052 Basel Switzerland Tel.: +41 61 683 77 34

Risks Editorial Office E-mail: risks@mdpi.com www.mdpi.com/journal/risks



Disclaimer/Publisher's Note: The title and front matter of this reprint are at the discretion of the Guest Editors. The publisher is not responsible for their content or any associated concerns. The statements, opinions and data contained in all individual articles are solely those of the individual Editors and contributors and not of MDPI. MDPI disclaims responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.





Academic Open Access Publishing mdpi.com

ISBN 978-3-7258-3689-5